

HANDBOOK OF GEOPHYSICAL EXPLORATION SEISMIC EXPLORATION

Klaus Helbig and Sven Treitel, Editors

VOLUME 38

Wave Fields in Real Media Wave Propagation in Anisotropic, Anelastic, Porous and Electromagnetic Media

by JOSÉ M. CARCIONE

Second Edition, Revised and Extended

HANDBOOK OF GEOPHYSICAL EXPLORATION

SECTION I. SEISMIC EXPLORATION

VOLUME 38

WAVE FIELDS IN REAL MEDIA:

Wave Propagation in Anisotropic, Anelastic, Porous and Electromagnetic Media (SECOND EDITION, REVISED AND EXTENDED)

HANDBOOK OF GEOPHYSICAL EXPLORATION

SECTION I. SEISMIC EXPLORATION

Editors: Klaus Helbig and Sven Treitel

Volume

- 1. Basic Theory in Reflection Seismology
- 2. Seismic Instrumentation, 2nd Edition
- 3. Seismic Field Techniques
- 4A. Seismic Inversion and Deconvolution: Classical Methods
- 4B. Seismic Inversion and Deconvolution: Dual-Sensor Technology
- 5. Seismic Migration (Theory and Practice)
- 6. Seismic Velocity Analysis
- 7. Seismic Noise Attenuation
- 8. Structural Interpretation
- 9. Seismic Stratigraphy
- 10. Production Seismology
- 11. 3-D Seismic Exploration
- 12. Seismic Resolution
- 13. Refraction Seismics
- Vertical Seismic Profiling: Principles 3rd Updated and Revised Edition
- 15A. Seismic Shear Waves: Theory
- 15B. Seismic Shear Waves: Applications
- 16A. Seismic Coal Exploration: Surface Methods
- 16B. Seismic Coal Exploration: In-Seam Seismics
 - 17. Mathematical Aspects of Seismology
 - 18. Physical Properties of Rocks
 - 19. Shallow High-Resolution Reflection Seismics
- 20. Pattern Recognition and Image Processing
- 21. Supercomputers in Seismic Exploration
- 22. Foundations of Anisotropy for Exploration Seismics
- 23. Seismic Tomography
- 24. Borehole Acoustics
- 25. High Frequency Crosswell Seismic Profiling
- 26. Applications of Anisotropy in Vertical Seismic Profiling
- 27. Seismic Multiple Elimination Techniques
- 28. Wavelet Transforms and Their Applications to Seismic Data Acquisition, Compression, Processing and Interpretation
- 29. Seismic Signatures and Analysis of Reflection Data in Anisotropic Media
- 30. Computational Neural Networks for Geophysical Data Processing
- Wave Fields in Real Media: Wave Propagation in Anitsotropic, Anelastic and Porous Media
- 32. Nuclear Magnetic Resonance Petrophysical and Logging Applications
- 33. Seismic Amplitude Inversion in Reflection Tomography
- 34. Seismic Waves and Rays in Elastic Wave Media
- 35. Seismic While Drilling: Fundamentals of Drill-Bit Seismic for Exploration
- 36. Information-based Inversion and Processing with Applications
- 37. Seismic Stratigraphy, Basin Analysis and Reservoir Characterisation
- Wave Fields in Real Media: Wave Propagation in Anisotropic, Anelastic Porous and Electromagnetic Media (Second Edition, Revised and Extended)

SECTION I. SEISMIC EXPLORATION

Volume 38

WAVE FIELDS IN REAL MEDIA:

Wave Propagation in Anisotropic, Anelastic, Porous and Electromagnetic Media

(SECOND EDITION, REVISED AND EXTENDED)

by

José M. CARCIONE Istituto Nazionale di Oceanografia e di Geofisica Sperimentale (OGS), Borgo Grotta Gigante 42c, 34010 Sgonico, Trieste, Italy



Amsterdam – Boston – Heidelberg – London – New York – Oxford Paris – San Diego – San Francisco – Singapore – Sydney – Tokyo Elsevier The Boulevard, Langford Lane, Kidlington, Oxford OX5 1GB, UK Radarweg 29, PO Box 211, 1000 AE Amsterdam, The Netherlands

First edition 2007

Copyright © 2007 Elsevier Ltd. All rights reserved

No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means electronic, mechanical, photocopying, recording or otherwise without the prior written permission of the publisher

Permissions may be sought directly from Elsevier's Science & Technology Rights Department in Oxford, UK: phone (+44) (0) 1865 843830; fax (+44) (0) 1865 853333; email: permissions@elsevier.com. Alternatively you can submit your request online by visiting the Elsevier web site at http://elsevier.com/locate/permissions, and selecting *Obtaining permission to use Elsevier material*

Notice

No responsibility is assumed by the publisher for any injury and/or damage to persons or property as a matter of products liability, negligence or otherwise, or from any use or operation of any methods, products, instructions or ideas contained in the material herein. Because of rapid advances in the medical sciences, in particular, independent verification of diagnoses and drug dosages should be made

British Library Cataloguing in Publication Data

A catalogue record for this book is available from the British Library

Library of Congress Cataloging-in-Publication Data

A catalog record for this book is available from the Library of Congress

ISBN-13: 978-0-08-046408-4 ISBN-10: 0-08-046408-4 ISSN: 0950-1401

For information on all Elsevier publications visit our website at books.elsevier.com

Printed and bound in The Netherlands

```
07 \ 08 \ 09 \ 10 \ 11 \quad 10 \ 9 \ 8 \ 7 \ 6 \ 5 \ 4 \ 3 \ 2 \ 1
```



Contents

	Prej	face	xiii				
	About the author						
	Bas	ic notat	ion xx				
	Glo	ssary oj	main symbols xxi				
1	Ani	sotrop	ic elastic media 1				
	1.1	Strain	-energy density and stress-strain relation				
	1.2	Dynamical equations					
		1.2.1	Symmetries and transformation properties	ř.			
			Symmetry plane of a monoclinic medium	1			
			Transformation of the stiffness matrix	į.			
	1.3	Kelvir	-Christoffel equation, phase velocity and slowness	È			
		1.3.1	Transversely isotropic media				
		1.3.2	Symmetry planes of an orthorhombic medium 13	ľ			
		1.3.3	Orthogonality of polarizations				
	1.4	Energ	y balance and energy velocity				
		1.4.1	Group velocity	ł			
		1.4.2	Equivalence between the group and energy velocities	l			
		1.4.3	Envelope velocity	ĺ			
		1.4.4	Example: Transversely isotropic media	į			
		1.4.5	Elasticity constants from phase and group velocities	ľ			
		1.4.6	Relationship between the slowness and wave surfaces 24				
			SH-wave propagation				
	1.5	Finely	layered media				
	1.6	Anom	alous polarizations	1			
		1.6.1	Conditions for the existence of anomalous polarization 29	1			
		1.6.2	Stability constraints				
		1.6.3	Anomalous polarization in orthorhombic media 33	1			
		1.6.4	Anomalous polarization in monoclinic media	1			
		1.6.5	The polarization				
		1.6.6	Example				
	1.7	The b	est isotropic approximation				
	1.8	Analy	tical solutions for transversely isotropic media	ľ			
		1.8.1	2-D Green's function				

LONTEND	÷
	6
	,

		1.8.2	3-D Green's function					
	1.9	Reflec	tion and transmission of plane waves					
		1.9.1	Cross-plane shear waves 45					
2	Vis	coelast	icity and wave propagation 51					
_	21	Energ	v densities and stress-strain relations 52					
		211	Fading memory and symmetries of the relayation tensor 54					
	00	Stewart	etrain relation for 1 D viscoalastic modia					
	4.4	0.0.1	Complex modulus and storage and less moduli					
		2.2.1	Complex modulus and storage and loss moduli					
		2.2.2	Energy and significance of the storage and loss moduli 57					
		2.2.3	Non-negative work requirements and other conditions					
		2.2.4	Consequences of reality and causality					
		2.2.5	Summary of the main properties					
			Relaxation function					
			Complex modulus					
	2.3	Wave	propagation concepts for 1-D viscoelastic media 61					
		2.3.1	Wave propagation for complex frequencies					
	2.4	Mecha	inical models and wave propagation					
		2.4.1	Maxwell model					
		2.4.2	Kelvin-Voigt model					
		2.4.3	Zener or standard linear solid model					
		2.4.4	Burgers model					
		2.4.5	Generalized Zener model					
			Nearly constant Q 80					
		246	Nearly constant_O model with a continuous spectrum 82					
	9.5	Const	ant O model and wave equation 82					
	2.0	0.5.1	Disco valority and attenuation factor					
		2.0.1	Wass spectra in differential form Excitant descriptions					
		2.0.2	Wave equation in differential form. Fractional derivatives 85					
		-	Propagation in Pierre snale					
	2.6	The c	oncept of centrovelocity					
		2.6.1	1-D Green's function and transient solution					
		2.6.2	Numerical evaluation of the velocities					
		2.6.3	Example					
	2.7	Memo	ry variables and equation of motion					
		2.7.1	Maxwell model					
		2.7.2	Kelvin-Voigt model					
		2.7.3	Zener model					
		2.7.4	Generalized Zener model					
3	Isot	Isotropic anelastic media 97						
	3.1	Stress	-strain relation					
	3.2	2 Equations of motion and dispersion relations						
	3.3	Vector plane wayes						
		3.3.1	Slowness, phase velocity and attenuation factor 100					
		332	Particle motion of the P wave 102					
		333	Particle motion of the S waves					
		3 9 4	Polarization and orthogonality 100					
		0.0.4	a oran contraction and oranogonality					

	3.4	Energy	y balance, energy velocity and quality factor
		3.4.1	P wave
		3.4.2	S waves
	3.5	Bound	lary conditions and Snell's law
	3.6	The co	prrespondence principle
	3.7	Raylei	gh waves
		3.7.1	Dispersion relation
		3.7.2	Displacement field
		3.7.3	Phase velocity and attenuation factor
		3.7.4	Special viscoelastic solids
			Incompressible solid
			Poisson solid
			Hardtwig solid
		3.7.5	Two Rayleigh waves
	3.8	Reflect	tion and transmission of cross-plane shear waves
	3.9	Memo	ry variables and equation of motion
	3.10	Analyt	tical solutions
		3.10.1	Viscoacoustic media
		3.10.2	Constant-Q viscoacoustic media
		3.10.3	Viscoelastic media
	3.11	The el	astodynamic of a non-ideal interface
		3.11.1	The interface model
			Boundary conditions in differential form
		3.11.2	Reflection and transmission coefficients of SH waves
			Energy loss
		3.11.3	Reflection and transmission coefficients of P-SV waves 133
			Energy loss
			Examples
33	0.012	en s	1 V 11 424 010
4	Ani	sotrop	ic anelastic media 139
	4.1	Stress-	strain relations
		4.1.1	Model 1: Effective anisotropy
		4.1.2	Model 2: Attenuation via eigenstrains
		4.1.3	Model 3: Attenuation via mean and deviatoric stresses 144
	4.2	Wave	velocities, slowness and attenuation vector
	4.3	Energy	y balance and fundamental relations
		4.3.1	Plane waves. Energy velocity and quality factor
		4.3.2	Polarizations
	4.4	The pl	hysics of wave propagation for viscoelastic SH waves
		4.4.1	Energy velocity
		4.4.2	Group velocity
		4.4.3	Envelope velocity
		4.4.4	Perpendicularity properties
		4.4.5	Numerical evaluation of the energy velocity
		4.4.6	Forbidden directions of propagation
	4.5	Memo	ry variables and equation of motion in the time domain

		4.5.1	Strain memory variables
		4.5.2	Memory-variable equations
		4.5.3	SH equation of motion
		4.5.4	qP-qSV equation of motion
	4.6	Analy	tical solution for SH waves in monoclinic media
5	The	recip	rocity principle 171
	5.1	Source	es, receivers and reciprocity
	5.2	The re	eciprocity principle
	5.3	Recip	rocity of particle velocity. Monopoles
	5.4	Recip	rocity of strain
		5.4.1	Single couples
			Single couples without moment
			Single couples with moment
		5.4.2	Double couples
			Double couple without moment. Dilatation
			Double couple without moment and monopole force
			Double couple without moment and single couple
	5.5	Recip	rocity of stress
6	Ref	lection	and transmission of plane waves 183
	6.1	Reflec	tion and transmission of SH waves
		6.1.1	Symmetry plane of a homogeneous monoclinic medium
		6.1.2	Complex stiffnesses of the incidence and transmission media 186
		6.1.3	Reflection and transmission coefficients
		6.1.4	Propagation, attenuation and energy directions
		6.1.5	Brewster and critical angles
		6.1.6	Phase velocities and attenuations
		6.1.7	Energy-flux balance
		6.1.8	Energy velocities and quality factors
	6.2	Reflec	tion and transmission of oP-oSV waves
		6.2.1	Pronagation characteristics
		6.2.2	Properties of the homogeneous wave 207
		6.2.3	Reflection and transmission coefficients
		6.2.4	Propagation, attenuation and energy directions
		6.2.5	Phase velocities and attenuations
		6.2.6	Energy-flow balance
		6.2.7	Umov-Poynting theorem, energy velocity and quality factor
		6.2.8	Reflection of seismic waves
		6.2.9	Incident inhomogeneous waves
			Generation of inhomogeneous waves
			Ocean bottom
	6.3	Reflec	tion and transmission at fluid/solid interfaces
	1.10	6.3.1	Solid/fluid interface
		6.3.2	Fluid/solid interface
		6.3.3	The Rayleigh window
	6.4	Reflec	tion and transmission coefficients of a set of lavers

7	Biot	's the	ory for porous media 235
	7.1	Isotrop	oic media. Strain energy and stress-strain relations
		7.1.1	Jacketed compressibility test
		7.1.2	Unjacketed compressibility test
	7.2	The co	oncept of effective stress
		7.2.1	Effective stress in seismic exploration
			Pore-volume balance
			Acoustic properties
		7.2.2	Analysis in terms of compressibilities
	7.3	Anisot	ropic media. Strain energy and stress-strain relations
		7.3.1	Effective-stress law for anisotropic media
		7.3.2	Summary of equations
		20,000,00	Pore pressure
			Total stress
			Effective stress
			Skempton relation 256
			Undrained-modulus matrix 256
		7.3.3	Brown and Korringa's equations 256
		1.0.0	Transversely isotronic medium 257
	7.4	Kineti	c energy 257
		7.4.1	Anisotropic media 260
	7.5	Dissing	ation notential 260
	1.0	7.5.1	Anisatronic media 263
	7.6	Lagrar	remotions and equation of motion 263
	1.0	7.6.1	The viscodynamic operator 265
		762	Fluid flow in a plane slit 265
		763	Anisatronic madia 270
	77	Plane	Amsonopic media
	4+4	7 7 1	Compressional waves 271
		1.1.1	Polotion with Towards's low 274
			The diffusion clear mode 276
		770	The dimusive slow mode
	= 0	1.1.2	The shear wave
	1.8	Strain	energy for inhomogeneous porosity
		7.8.1	Complementary energy theorem
	-	7.8.2	Volume-averaging method
	7.9	Bound	ary conditions
		7.9.1	Interface between two porous media
			Deresiewicz and Skalak's derivation
			Gurevich and Schoenberg's derivation
		7.9.2	Interface between a porous medium and a viscoelastic medium 288
		7.9.3	Interface between a porous medium and a viscoacoustic medium 289
		7.9.4	Free surface of a porous medium
	7.10	The m	esoscopic loss mechanism. White model
	7.11	Green	s function for poro-viscoacoustic media
		7.11.1	Field equations
		7.11.2	The solution

	7.12	Green's function at a fluid/porous medium interface	299
	7.13	Poro-viscoelasticity	303
	7.14	Anisotropic poro-viscoelasticity	307
		7.14.1 Stress-strain relations	308
		7.14.2 Biot-Euler's equation	309
		7.14.3 Time-harmonic fields	309
		7.14.4 Inhomogeneous plane waves	312
		7.14.5 Homogeneous plane waves	314
		7.14.6 Wave propagation in femoral bone	316
8	The	acoustic-electromagnetic analogy	321
	8.1	Maxwell's equations	323
	8.2	The acoustic-electromagnetic analogy	. 324
		8.2.1 Kinematics and energy considerations	329
	8.3	A viscoelastic form of the electromagnetic energy	
		8.3.1 Umov-Povnting's theorem for harmonic fields	. 332
		8.3.2 Umov-Poynting's theorem for transient fields	333
		The Debye-Zener analogy	337
		The Cole-Cole model	341
	8.4	The analogy for reflection and transmission	349
	0.4	8.4.1 Reflection and refraction coefficients	349
		Propagation attenuation and ray angles	343
		Energy-flux balance	343
		8.4.2 Application of the analogy	344
		Defraction index and Freenal's formulae	944
		Representation index and rreshers formulae	245
		Critical angle. Total reflection	946
		Deflectivity and transmicrivity	340
		Dual fields	349
		Dual fields	349
		Sound waves	330
		6.4.5 The analogy between 1M and 1E waves	331
		Green's analogies	352
		8.4.4 Brief historical review	355
	8.0	3-D electromagnetic theory and the analogy	356
		8.5.1 The form of the tensor components	357
	0.0	8.5.2 Electromagnetic equations in differential form	358
	8.0	Plane-wave theory	359
		8.6.1 Slowness, phase velocity and attenuation	361
		8.6.2 Energy velocity and quality factor	363
	8.7	Analytical solution for anisotropic media	366
		8.7.1 The solution	368
	8.8	Finely layered media	369
	8.9	The time-average and CRIM equations	372
	8.10	The Kramers-Kronig dispersion relations	373
	8.11	The reciprocity principle	374
	8.12	Babinet's principle	375

	rotation	. 376					
	8.14	Poro-a	coustic and electromagnetic diffusion	. 378			
		8.14.1	Poro-acoustic equations	. 378			
		8.14.2	Electromagnetic equations	. 380			
			The TM and TE equations	. 380			
			Phase velocity, attenuation factor and skin depth	. 381			
			Analytical solutions	. 381			
	8.15	Electro	p-seismic wave theory	. 382			
9	Nur	nerical	methods	385			
	9.1	Equati	on of motion	. 385			
	9.2	Time i	ntegration	. 386			
		9.2.1	Classical finite differences	. 388			
		9.2.2	Splitting methods	. 389			
		9.2.3	Predictor-corrector methods	. 390			
			The Runge-Kutta method	. 390			
		9.2.4	Spectral methods	. 390			
		9.2.5	Algorithms for finite-element methods	. 392			
	9.3	Calcul	ation of spatial derivatives	. 392			
		9.3.1	Finite differences	. 392			
		9.3.2	Pseudospectral methods	. 394			
		9.3.3	The finite-element method	. 396			
	9.4	Source	implementation	. 397			
	9.5	Bound	ary conditions	398			
	9.6	Absort	bing boundaries	. 400			
	9.7	Model	and modeling design – Seismic modeling	401			
	9.8	Conch	ulu mouenny acaign oceanie mouenny	404			
	0.0	Annen	dix	405			
	0.0	991	Electromagnetic-diffusion code	405			
		002	Finite-differences code for the SH-wave equation of motion	400			
		0.0.2	Finite-differences code for the SH-wave and Maxwell's equations	415			
		0.0.4	Pseudospectral Fourier Method	499			
		0.0.5	Pseudospectral Chebyshev Method	424			
	220.00	3.3.9	r seudospectral chebysnev method	, 121			
	Exar	ninatio	ns	427			
	Chre	mology	of main discoveries	431			
	Leonardo's manuscripts						
	A list of scientists						
	Bibliography						
	Name Index						
	a i	ae mae	ал. А.	491			
	Sub	ject in	dex	503			

({ L'impeto }) cioè la propagazione della perturbazione del mezzo o, più in generale, di un qualsiasi elemento saliente ({ è molto più veloce che ll'acqua, perché molte sono le volte che l'onda fuggie il locho della sua creatione, e ll'acqua non si muove di sito, a ssimilitudine delle onde fatte il maggio nelle biade dal corso de venti, che ssi vede correre l'onde per le campagnie, e le biade non si mutano di lor sito)).

({ The impetus }) that is, the propagation of the perturbation of the medium or, more generally, of any salient element ({ is much faster than the water, because many are the times that the wave escapes the place of its creation, and water stays in place, as the waves made in May in the corn by the blowing of the wind, so that one can see the running waves in the fields and the corn does not change place)).

Leonardo da Vinci (Del moto e misura dell'acqua)

Preface

(SECOND EDITION, REVISED AND EXTENDED)

This book presents the fundamentals of wave propagation in anisotropic, anelastic and porous media. I have incorporated in this second edition a chapter about the analogy between acoustic waves (in the general sense) and electromagnetic waves. The emphasis is on geophysical applications for seismic exploration, but researchers in the fields of earthquake seismology, rock acoustics, and material science, – including many branches of acoustics of fluids and solids (acoustics of materials, non-destructive testing, etc.) – may also find this text useful. This book can be considered, in part, a monograph, since much of the material represents my own original work on wave propagation in anisotropic, viscoelastic media. Although it is biased to my scientific interests and applications, I have, nevertheless, sought to retain the generality of the subject matter, in the hope that the book will be of interest and use to a wide readership.

The concepts of porosity, anelasticity¹ and anisotropy in physical media have gained much attention in recent years. The applications of these studies cover a variety of fields, including physics and geophysics, engineering and soil mechanics, underwater acoustics, etc. In particular, in the exploration of oil and gas reservoirs, it is important to predict the rock porosity, the presence of fluids (type and saturation), the preferential directions of fluid flow (anisotropy), the presence of abnormal pore-pressures (overpressure), etc. These microstructural properties and in-situ rock conditions can be obtained, in principle, from seismic and electromagnetic properties, such as travel times, amplitude information, and wave polarization. These measurable quantities are affected by the presence of anisotropy and attenuation mechanisms. For instance, shales are naturally bedded and possess intrinsic anisotropy at the microscopic level. Similarly, compaction and the presence of microcracks and fractures make the skeleton of porous rocks anisotropic. The presence of fluids implies relaxation phenomena, which causes wave dissipation. The use of modeling and inversion for the interpretation of the seismic response of reservoir rocks requires an understanding of the relationship between the seismic and electromagnetic properties and the rock characteristics, such as permeability, porosity, tortuosity, fluid viscosity, stiffness, dielectric permittivity, etc.

Wave simulation is a theoretical field of research that began nearly three decades ago, in close relationship with the development of computer technology and numerical algo-

¹The term anelasticity seems to have been introduced by Zener (1948) to denote materials in which "strain may lag behind stress in periodic vibrations", in which no permanent deformation occurs and wherein the stress-strain relation is linear. Viscoelasticity combines the classical theories of elasticity and Newtonian fluids, but is not restricted to linear behavior. Since this book deals with linear deformations, anelasticity and viscoelasticity will be synonymous herein.

rithms for solving differential and integral equations of several variables. In the field of research known as computational physics, algorithms for solving problems using computers are important tools that provide insight into wave propagation for a variety of applications.

This book examines the differences between an ideal and a real description of wave propagation, where ideal means an elastic (lossless), isotropic and single-phase medium, and real means an anelastic, anisotropic and multi-phase medium. The first realization is, of course, a particular case of the second, but it must be noted that in general, the real description is not a simple and straightforward extension of the ideal description.

The analysis starts by introducing the constitutive equation (stress-strain relation) appropriate for the particular rheology². This relation and the equations of conservation of linear momentum are combined to give the equation of motion, a second-order or a first-order matrix differential equation in time, depending on the formulation of the field variables. The differential formulation for lossy media is written in terms of memory (hidden) variables or alternatively, fractional derivatives. Biot's theory is essential to describe wave propagation in multi-phase (porous) media from the seismic to the ultrasonic frequency range, representative of field and laboratory experiments, respectively. The acoustic-electromagnetic analogy reveals that different physical phenomena have the same mathematical formulation. For each constitutive equation, a plane-wave analysis is performed in order to understand the physics of wave propagation (i.e., calculation of phase, group and energy velocities, and quality and attenuation factors). For some cases, it is possible to obtain an analytical solution for transient wave fields in the spacefrequency domain, which is then transformed to the time domain by a numerical Fourier transform. The book concludes with a review of the so-called direct numerical methods for solving the equations of motion in the time-space domain. The plane-wave theory and the analytical solutions serve to test the performance (accuracy and limitations) of the modeling codes.

A brief description of the main concepts discussed in this book follows.

Chapter 1: Anisotropic elastic media. In anisotropic lossless media, the directions of the wavevector and Umov-Poynting vector (ray or energy-flow vector) do not coincide. This implies that the phase and energy velocities differ. However, some ideal properties prevail: there is no dissipation, the group-velocity vector is equal to the energy-velocity vector, the wavevector is normal to the wave-front surface, the energy-velocity vector is normal to the slowness surface, plane waves are linearly polarized and the polarization of the different wave modes are mutually orthogonal. Methods used to calculate these quantities and provide the equation of motion for inhomogeneous media are shown. We also consider finely layered and anomalously polarized media and the best isotropic approximation of anisotropic media. Finally, the analysis of a reflection-transmission problem and analytical solutions along the symmetry axis of a transversely isotropic medium are discussed.

Chapter 2: Anelasticity and wave propagation. Attenuation is introduced in the

²From the Greek $\rho e \tilde{\omega}$ – to flow, and $\lambda \sigma \gamma \delta \varsigma$ – word, science. Today, rheology is the science concerned with the behavior of real materials under the influence of external stresses.

PREFACE

form of Boltzmann's superposition law, which implies a convolutional relation between the stress and strain tensors through the relaxation and creep matrices. The analysis is restricted to the one-dimensional case, where some of the consequences of anelasticity become evident. Although phase and energy velocities are the same, the group velocity loses its physical meaning. The concept of centrovelocity for non-harmonic waves is discussed. The uncertainty in defining the strain and rate of dissipated-energy densities is overcome by introducing relaxation functions based on mechanical models. The concepts of memory variable and fractional derivative are introduced to avoid time convolutions and obtain a time-domain differential formulation of the equation of motion.

Chapter 3: Isotropic anelastic media. The space dimension reveals other properties of anelastic (viscoelastic) wave fields. There is a distinct difference between the inhomogeneous waves of lossless media (interface waves) and those of viscoelastic media (body waves). In the former case, the direction of attenuation is normal to the direction of propagation, whereas for inhomogeneous viscoelastic waves, that angle must be less than $\pi/2$. Furthermore, for viscoelastic inhomogeneous waves, the energy does not propagate in the direction of the slowness vector and the particle motion is elliptical in general. The phase velocity is less than that of the corresponding homogeneous wave (for which planes of constant phase coincide with planes of constant amplitude); critical angles do not exist in general, and, unlike the case of lossless media, the phase velocity and the attenuation factor of the transmitted waves depend on the angle of incidence. There is one more degree of freedom, since the attenuation vector is playing a role at the same level as the wavenumber vector. Snell's law, for instance, implies continuity of the tangential components of both vectors at the interface of discontinuity. For homogeneous plane waves, the energy-velocity vector is equal to the phase-velocity vector.

Chapter 4: Anisotropic anelastic media. In isotropic media there are two well defined relaxation functions, describing purely dilatational and shear deformations of the medium. The problem in anisotropic media is to obtain the time dependence of the relaxation components with a relatively reduced number of parameters. Fine layering has an "exact" description in the long-wavelength limit. The concept of eigenstrain allows us to reduce the number of relaxation functions to six; an alternative is to use four or two relaxation functions when the anisotropy is relatively weak. The analysis of SH waves suffices to show that in anisotropic viscoelastic media, unlike the lossless case: the groupvelocity vector is not equal to the energy-velocity vector, the wavevector is not normal to the energy-velocity surface, the energy-velocity vector is not normal to the slowness surface, etc. However, an energy analysis shows that some basic fundamental relations still hold: for instance, the projection of the energy velocity onto the propagation direction is equal to the magnitude of the phase velocity.

Chapter 5: The reciprocity principle. Reciprocity is usually applied to concentrated point forces and point receivers. However, reciprocity has a much wider application potential; in many cases, it is not used at its full potential, either because a variety of source and receiver types are not considered or their implementation is not well understood. In this chapter, the reciprocity relations for inhomogeneous, anisotropic, viscoelastic solids, and for distributed sources and receivers are obtained. In addition to the usual relations involving directional forces, it is shown that reciprocity can also be applied to a variety of source-receiver configurations used in earthquake seismology and seismic reflection and refraction methods.

Chapter 6: Reflection and transmission coefficients. The SH and qP-qSV cases illustrate the physics of wave propagation in anisotropic anelastic media. In general, the reflected and transmitted waves are inhomogeneous, i.e., equiphase planes do not coincide with equiamplitude planes. The reflected wave is homogeneous only when the symmetry axis is perpendicular to the interface. If the transmission medium is elastic and the incident wave is homogeneous, the transmitted wave is inhomogeneous of the elastic type, i.e., the attenuation vector is perpendicular to the Umov-Poynting vector. The angle between the attenuation vector and the slowness vector may exceed 90°, but the angle between the attenuation and the Umov-Poynting vector is always less than 90°. If the incidence medium is elastic, the attenuation of the transmitted wave is perpendicular to the interface. The relevant physical phenomena are not related to the propagation direction (slowness vector), but rather to the energy-flow direction (Umov-Povnting vector) for instance, the characteristics of the elastic type inhomogeneous waves, the existence of critical angles, and the fact that the amplitudes of the reflected and transmitted waves decay in the direction of energy flow despite the fact that they grow in the direction of phase propagation.

Chapter 7: Biot's theory for porous media. Dynamic porous media behavior is described by means of Biot's theory of poroelasticity. However, many developments in the area of porous media existed before Biot introduced the theory in the mid 50s. These include, for instance, Terzaghi's law, Gassmann's equation, and the static approach leading to the concept of effective stress, much used in soil mechanics. The dynamical problem is analyzed in detail using Biot's approach: that is, the definition of the energy potentials and kinetic energy and the use of Hamilton's principle to obtain the equation of motion. The coefficients of the strain energy are obtained by the so-called jacketed and unjacketed experiments. The theory includes anisotropy and dissipation due to viscodynamic and viscoelastic effects. A short discussion involving the complementary energy theorem and volume-average methods serves to define the equation of motion for inhomogeneous media. The interface boundary conditions and the Green function problem are treated in detail, since they provide the basis for the solution of wave propagation in inhomogeneous media. The mesoscopic loss mechanism is described by means of White's theory for planelayered media developed in the mid 70s. An energy-balance analysis for time-harmonic fields identifies the strain- and kinetic-energy densities, and the dissipated-energy densities due to viscoelastic and viscodynamic effects. The analysis allows the calculation of these energies in terms of the Umov-Povnting vector and kinematic variables, and the generalization of the fundamental relations obtained in the single-phase case (Chapter 4). Measurable quantities, like the attenuation factor and the energy velocity, are expressed in terms of microstructural properties such as tortuosity and permeability.

Chapter 8: The acoustic-electromagnetic analogy. The two-dimensional Maxwell's equations are mathematically equivalent to the SH-wave equation based on a Maxwell stress-strain relation, where the correspondence is magnetic field/particle velocity, electric field/stress, dielectric permittivity/elastic compliance, resistivity/viscosity and magnetic permeability/density. It is shown that Fresnel's formulae can be obtained from the re-

PREFACE

flection and transmission coefficients of shear waves. The analogy is extended to three dimensions. Although there is not a complete correspondence, the material properties are mathematically equivalent by using the Debye-Zener analogy. Moreover, an electromagnetic energy-balance equation is obtained from viscoelasticity, where the dielectric and magnetic energies are equivalent to the strain and kinetic energies. Other analogies involve Backus averaging for finely layered media, the time-average equation, the Kramers-Kronig dispersion relations, the reciprocity principle, Babinet's principle, Alford rotation, and the diffusion equation describing electromagnetic fields and the behaviour of the Biot quasi-static mode (the second slow wave) at low frequencies.

Chapter 9: Numerical methods. In order to solve the equation of motion by direct methods, the model (the geological layers in exploration geophysics and seismology) is approximated by a numerical mesh; that is, the model is discretized in a finite numbers of points. These techniques are also called grid methods and full-wave equation methods, since the solution implicitly gives the full wave field. Direct methods do not have restrictions on the material variability and can be very accurate when a sufficiently fine grid is used. They are more expensive than analytical and ray methods in terms of computer time, but the technique can easily handle the implementation of different strain-stress laws. Moreover, the generation of snapshots can be an important aid in interpretation. Finite-differences, pseudospectral and finite-element methods are considered in this chapter. The main aspects of the modeling are introduced as follows: (a) time integration, (b) calculation of spatial derivatives, (c) source implementation, (d) boundary conditions, and (e) absorbing boundaries. All these aspects are discussed and illustrated using the acoustic and SH wave equations. The pseudospectral algorithms are discussed in more detail.

This book is aimed mainly at graduate students and researchers. It requires a basic knowledge of linear elasticity and wave propagation, and the fundamentals of numerical analysis. The following books are recommended for study in these areas: Love (1944), Kolsky (1953), Born and Wolf (1964), Pilant (1979), Auld (1990a,b), Celia and Gray (1992), Jain (1984) and Slawinski (2003). At the end of the book, I provide a list of questions about the relevant concepts, a chronological table of the main discoveries and a list of famous scientists, regarding wave propagation and its related fields of research.

Slips and errors that were present in the first edition have been corrected in the present edition. This extends the scope of the book to electromagnetism by including Chapter 8. Other additions to the first edition include: the extension of anomalous polarization to monoclinic media (Chapter 1), the best isotropic approximation of an anisotropic elastic medium (Chapter 1), the analysis of wave propagation for complex frequencies (Chapter 2), Burgers's mechanical model (Chapter 2), White's mesoscopic-attenuation theory (Chapter 7), the Green function for surface waves in poroelastic media (Chapter 7), a Fortran code for the diffusion equation based on spectral methods, a Fortran code for the numerical solution of Maxwell's equations, and other minor additions and relevant recent references. Also, the history of science has been expanded by including researchers and discoveries related to the theory of light and electromagnetic wave propagation. Errata for the first edition may be found on author's homepage, currently at: http://www.ogs.trieste.it Errata and comments may be sent to the author at the following: jcarcione@libero.it jcarcione@inogs.it Thank you!

xviii

ABOUT THE AUTHOR

José M. Carcione was born in Buenos Aires, Argentina in 1953. He received the degree "Licenciado in Ciencias Físicas" from Buenos Aires University in 1978, the degree "Dottore in Fisica" from Milan University in 1984, and the degree Ph.D. in Geophysics from Tel-Aviv University in 1987. In 1987 he was awarded the Alexander von Humboldt scholarship for a position at the Geophysical Institute of Hamburg University, where he stayed from 1987 to 1989. From 1978 to 1980 he worked at the "Comisión Nacional de Energía Atómica" at Buenos Aires. From 1981 to 1987 he worked as a research geophysicist at "Yacimientos Petrolíferos Fiscales", the national oil company of Argentina. Presently, he is a senior geophysicist at the "Istituto Nazionale di Oceanografia e di Geofisica Sperimentale (OGS)" (former "Osservatorio Geofisico Sperimentale") in Trieste, where he was Head of the Department of Geophysics from 1996 to 2000. He is Editor of Geophysics, Near Surface Geophysics, and Bolletino di Geofisica Teorica ed Applicata. His current research deals with numerical modeling, the theory of wave propagation in acoustic and electromagnetic media, and their application to geophysical problems.



Basic notation

We denote the spatial variables x, y and z of a right-hand Cartesian system by the indices i, j, ... = 1, 2 and 3, respectively, the position vector by x or by r, a partial derivative with respect to a variable x_i with ∂_i , and a first and second time derivative with ∂_t and ∂^2_{tt} . For clarity in reading and ease in programming, the use of numbers to denote the subindices corresponding to the spatial variables is preferred. The upper case indices $I, J, \ldots = 1, \ldots, 6$ indicate the shortened matrix notation (Voigt's notation) where pairs of subscripts (i, j) are replaced by a single number (I or J) according to the correspondence $(11) \rightarrow 1$, $(22) \rightarrow 2$, $(33) \rightarrow 3$, $(23) = (32) \rightarrow 4$, $(13) = (31) \rightarrow 5$, (12) $= (21) \rightarrow 6$. Matrix transposition is denoted by the superscript " \top " (it is not indicated in two- and three-components vectors), $\sqrt{-1}$ by i, complex conjugate by the superscript "*", the scalar and matrix products by the symbol ".", the vector product by the symbol "×", the dyadic product by the symbol " \otimes ", and unit vectors by $\hat{\mathbf{e}}_i$, i = 1, 2, 3 if referring to the Cartesian axes. The identity matrix in n-dimensional space is denoted by I_n . The gradient, divergence, Laplacian and curl operators are denoted by grad [·], div [·], Δ [·] and curl [·], respectively. The components of the Levi-Civita tensor ϵ_{iik} are 1 for cyclic permutations of 1,2 and 3, -1 if two indices are interchanged and 0 if an index is repeated. The operators $Re(\cdot)$ and $Im(\cdot)$ take the real and imaginary parts of a complex quantity (in some cases, the subindices R and I are used). The Fourier-transform operator is denoted by $\mathcal{F} [\cdot]$ or a tilde above the function. The convention is

$$\tilde{f}(\omega) = \int_{-\infty}^{\infty} f(t) \exp(-i\omega t) dt, \quad f(t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} \tilde{f}(\omega) \exp(i\omega t) dt,$$

where t is the time variable and ω is the angular frequency. The Einstein convention of repeated indices is assumed, but the notation I(I) or i(i) implies no summation. In general, we express vectors and column matrices (arrays) by bold and lower case letters and matrices and tensors by bold and upper case letters.

Glossary of main symbols

u	displacement vector.	T	kinetic energy.
v	particle-velocity vector.	V	strain energy.
eni. eni	strain components ¹ .	E	total energy.
0. 1	dilatation.	Ď	rate of dissipated energy.
d^2	deviator.	0	quality factor.
e (e1. e11). e	strain array (tensor).	Un	phase velocity.
σ (σ_1, σ_{ii}), Σ	stress array (tensor).	U.	group velocity.
0	density.	Vene	envelope velocity.
C (c11)	elasticity matrix.	17e	energy velocity.
w .	relaxation function ² .	6	porosity.
M	complex modulus.	n	viscosity.
Te. Te	relaxation times.	R	permeability.
e (e)	memory variable.	u ^(s)	displacement of the solid.
$\Psi(\psi_{ij})$	relaxation matrix.	u(1)	displacement of the fluid.
$\mathbf{P}(nu)$	complex stiffness matrix.	11 ^(m)	displacement of the matrix.
A	eigenstiffnesses.	C	variation of fluid content.
s	slowness vector.	w	relative fluid displacement.
ĸ	real wavenumber vector.	K	dry-rock moduli.
k	complex wavenumber vector.	Ke	Gassmann's modulus.
0	attenuation vector ³	C	nore compressibility
<u>a</u>	attenuation factor	τ	tortuosity
F (F.,)	Kelvin-Christoffel matrix	v	viscodynamic operator
P (- 0)	Umov-Poynting vector	R.	critical angle
G	Green's function.	θ_B	Brewster angle.
Е	electric vector		refraction index
н	magnetic vector ⁴	E	dielectric energy
D	electric displacement.	E	conductive energy.
B	magnetic induction	E	magnetic energy
2	dielectric-permittivity tensor ⁵	E	internal (hidden) variable
à	conductivity tensor	P	reflection coefficient
0 A	magnetic normashility tensor	T	transmission coefficient ⁶
μ	magnetic-permeability tensor.	1	transmission coemcient".

 $\label{eq:eigenvector} \begin{array}{l} {}^1(e_{ij}=2\;\epsilon_{ij},\,i\neq j.) \\ {}^2\text{Also used to denote the angle of the energy-velocity vector.} \\ {}^3\text{Also used to denote the effective-stress-coefficient matrix.} \end{array}$

Also used to denote the encoder success contains matrix, 4 Also used to denote the propagation matrix, ${}^{5}\hat{e}^{0}_{ij}$ and \hat{e}^{∞}_{ij} : static and optical components; \hat{e}_{0} : dielectric permittivity of free space, 6 Also used to denote the kinetic energy.

This page intentionally left blank

Chapter 1

Anisotropic elastic media

Wood two years since I printed this Denny in an Annymon of the read of any Book of the Descriptions of Rehavings science illumission defield. Ut travia sincis, That is the Power of any Spring is in the same progention with the Travian theory, Dayt is, if one power structures in bondy it into space, two will be of it two, and three will have differed on there, and so formade.

Heterogeneous actions from articlect on propagated within the solid on a direct love if they but perpendicular to the superfaces or beyods, but if videpady in constant direct, but different and differted, according to the particular incluention of the body straking, and according to the proparties of the Particle's straking and being strack.

Robert Rocke (Rooke, 4678)

For stress-strain law and/or wave propagation in anisotropic elastic (hostless) modia are discussed in several books, notably, Luve (1914). Musgrave (1970). Federav (1988) Belizer (1988), Paytin (1983), Nye (1985), Hanyga (1985). Aboudi (1991), Add (1990a.b), Holbig (1994) and Tang (1996). Crampin (1981), Winterstein (1990), Mavko, Mukerji and Dvorkin (1998). Eventsin (2004) and Cervený (2004) provide a comprehensive review of the subject with respect to sensitic apple atoms. In this chapter, we review the more features of anisotropy in order to guiderstand the physics of wave propagation in unsotropic elastic media, and to provide the basis for the theoretical developments regarding more emplex theologies, discussed in the next chapters.

1.1 Strain-energy density and stress-strain relation

Defining strain energy is the first step in determining the constitutive equations or stressstrain relations, which provide the basis for the description of static and dynamic deformations of physical media. Invoking the symmetry of the stress and strain tensors), the most general form of the strain energy volume density is

$$2V \le \sum_{I=0}^{n} \sum_{\ell=0}^{n} a_{II} \rho_{II}$$
 (1.1)

New Yild (1996).cc and Klauster (1991) and New are 1991) for a theory of reacyptic trackness and enable-tensory.

According to Voigt stantation.

$$c_1 = c_1, \quad (\partial_t y_1, \dots, T = 1, 2, 3, \dots, c_1 = e_1, \dots, (\partial_t y_1 + (\partial_t y_1, \dots, e_T))(T = 1, 5, 6), \quad (1, 2)$$

where u_i are the displacement components, and a_{IJ} are 21 coefficients related to the elasticity constants c_{IJ} as $a_{IJ} = c_{IJ}$ and $a_{IJ} = 2c_{IJ}$ for $I \neq J$ (howe, 1994, p. 100, 159). Note that the strains in standard use are

$$e_{ij} = e_{Ii} - I = 1, 2, 3, \quad e_{ij} = \frac{1}{2} e_{Ij} - \frac{1}{2} e_{Ij} - i \neq j \ (I = 1, 5, 6), \quad (1.3)$$

Alternatively, using the Cartesian components, the strons-energy density can be expressed in terms of a fronth-order elasticity tensor ϕ_{AL} as

where the symmetries

$$c_{00} = c_{0k} - c_{0k} - c_{0n}$$
 (1.5)

reduce the number of independent elasticity constants from SU to 21. The first and second equalities arise from the symmetry of the strain and stress tensors. The first organity is obtained by noting that the second partial derivatives of U are rulependent of the order of differentiation with respect to the strain components (Auld. 1990a, p. 138–114). Fing 1996, p. 320.

The strain traser can be expressed as $\mathbf{c} = \sum c_0 | \mathbf{c}_0 \otimes \mathbf{c}_1 |$ Let us consider a medium that possesses at each point the (x, z)-plane as its plane of symmetry. This medium has monoclude symmetry. A reflection with respect to this plane $(y_1 + \cdots + y_i)$ should have the strain energy gradiened. Such a transformation implies $e_{y_1} = e_{y_2}$ and $e_{y_1} = e_{y_2}$, which implies $c_{y_1} = c_{y_2} = c_{y_3} = c_{y_3} = c_{y_3} = c_{y_4} = 0$ (see have (944) $\mathbf{p} = 173$). The result is

$$||\mathcal{X}|| = c_{12}c_{12}^{2} + c_{22}c_{12}^{2} + c_{31}c_{12}^{2} + 2c_{12}c_{12} + 2c_{13}c_{13} + 2c_{13}c_{13}c_{3}$$

$$+ c_{32}c_{32}^{2} + c_{32}c_{32}^{2} + 2c_{12}c_{32} + c_{22}c_{23}c_{33} + c_{33}c_{34}c_{34} + 2c_{33}c_{34}c_{34} + (1.6)$$

Similar reflections with respect to the other Cartesian planes of symmetry imply that other coefficients become equal to each other. Thus, the number of coefficients required to describe a molium possessing orthorhombic symmetry — three nuturally orthogonal planes of symmetry — is reduced. The result is

If the material possesses an axis of rotational symmetry – as in a transvetsely isotropic mediate – the strain energy should be invariant to totations about that axis. Then,

$$21 = c_{11}(e_{11}^{2} + e_{12}^{2}) + c_{11}e_{11}^{2} + 2(c_{11} - 2c_{10})e_{10}e_{12} + 2c_{10}(e_{11} + e_{12})e_{31} + c_{10}(e_{11}^{2} + e_{11}^{2}) + c_{11}e_{12}^{2}$$

$$(1.8)$$

(Love, 1944) p. 152-160; Helbig, 1994, p. 87

If the medium is isotropic, every plane is a plane of symmetry, and every axis is an axis of symmetry. Consequently, some of the coefficients vanish, and we obtain

$$2V = e_{11}(e_{11}^{+} + e_{22}^{+} + e_{11}^{+}) + 2(e_{11}^{+} + 2e_{22}^{+}) + e_{22}(e_{12}^{+} + e_{23}^{+}) + e_{22}(e_{12}^{+} + e_{2$$

where $c_{11} = 3 + 2\mu$, and $c_{22} = \mu$, with λ and μ being the Lamé constants.

We matively the strain energy for isotropic media can be expressed in terms of vavariants of strain – up to the scenal-order. These invariants can be identified in equation (2.9). In fact, this equation can be rewritten as

$$2V = c_0 u^T - h_{00}c_0$$
 (1.40)

where

$$v = v_0 + v_0 + v_0 \qquad (1.1)$$

aml

$$w = e_{13}e_{23} + e_{10}e_{33} + e_{23}e_{33} - \frac{1}{4}(e_{33}^2 + e_{13}^2 + e_{13}^2)$$
(1.32)

īυĭ

are invariants of strain clove, 1911, \mathbf{p}_{i} (4). These invariants are the coefficients of the second and first powers of the polynomial in $v_{i} \det(\mathbf{c} - v\mathbf{L}_{i})$, where $\mathbf{c} = \sum_{i} \mathbf{e}_{i} \geq \mathbf{e}_{i}$. The roots of this polynomial are the principal strains that define the strain quadric – on ellipsoid (Love, 1914) $\mathbf{p}_{i} = 0.5$

We know a prime 'for instance, from experiments) that a boungemons isotropic medium 'supports' two prior deformation modes, i.e., a dilatational one and a shear one. These correspond to a change of volume, without a change in shape, and a change in shape without a change of volume, respectively. It is, therefore, reasonable to follow the physics of the problem and write the strain energy in terms of the dilatation d and the deviator

$$d^2 = d_{10}d_{10}$$
 (1.13)

where

$$d_0 \le \epsilon_0 - \frac{1}{3}\omega_0$$
 (1.35)

are the components of the deviatoric strain tensor, with β_{ij} being the components of the Kronecker matrix. Since,

$$u^{i} = e_{11}^{i} + e_{22}^{i} + e_{11}^{i} + \frac{1}{2}(e_{22}^{i} + e_{13}^{i} + e_{11}^{i}) - \frac{\theta^{i}}{3}$$
 (1.16)

and $\varpi = (d^2/3) + d^2/2$, we have the following expression:

$$2V = \left(c_{11} - \frac{1}{3}c_{20}\right)d^2 + 2c_0d^2, \qquad (1.17)$$

This firm is used in Chapter 7 to derive the dynamical equations of pornelasticity.

Having obtained the strain energy expression, we now consider stress. The stresses are given by

$$\frac{\partial \Gamma}{\partial c_{0}} = \frac{11.380}{0}$$

(Love, 1914, p. 95) or, using the shortened matrix notation.

$$\sigma_j = \frac{\partial 1}{\partial c_1}$$
. (1.19)

where:

$$\sigma = (\sigma_1, \sigma_2, \sigma_3, \sigma_4, \sigma_5, \sigma_6, \sigma_6) = (\sigma_{11}, \sigma_{12}, \sigma_{13}, \sigma_{13}, \sigma_{13}, \sigma_{13}, \sigma_{13}, \sigma_{13})$$
 (11.20)

Having much use of the standard strain components from the outset, and having calculated the stresses as $\sigma_{ij} = \partial 1/\partial \epsilon_{ij}$ from equation (1, k), we are required to distinguish between (i), and (i)) components |i| < j. However, the use of Love's notation, to express both the strain components ϵ_{ij} and the form (1, 1), avoids the measiby of this distinction

Using the Cartesian and shortened notations, we can write Hooke's law for the anisotropic elastic case as

$$\sigma_0 = c_0 \alpha_0$$
 (1.21)

and

$$\sigma_{I} = c_{II}c_{I}$$
, (11.22)

responsively.

1.2 Dynamical equations

In this section, we derive the differential equations describing wave propagation it, terms of the displacements of the material. The conservation of linear momentum implies

$$\partial_t \sigma_{ii} + f_i = \rho \partial_0^2 \sigma_i$$
 (1.23)

(Aubi (1992a) p. (3)) where n_i are the components of the displacement vector $|\rho|$ is the mass density and f_i are the components of the body forces per unit volume. Assuming a volume Ω bounded by a surface S_i the volume integral of equation (1.23) is the balance between the surface tractions on S_i - obtained by applying the divergence theorem to $\partial_i \sigma_{ij}$

) and the body forces with the mertra term $pd_{i}a_{i}$. Equations (1.23) are known as Euler's equations for elasticity, corresponding to Newton's law of motion for particles.

The substitution of Hooke's law (1/21) into equation (1.23) yields

$$d_{\mu}(r_{\mu\nu}, q_{\mu\nu}) = f_{\mu\nu} - i\partial_{\mu}^{\mu}q_{\mu} \qquad (1.24)$$

In order to use the shurtened matrix mutation, we introduce Aubl's mutation (Auld, 1990)a.h) for the differential operators. The symmetric gradient operator has the following matrix representation

$$\nabla = \begin{pmatrix} \partial & 0 & 0 & 0 & \partial_s & \partial_f \\ 0 & \partial_s & 0 & \partial_1 & 0 & \partial_s \\ 0 & 0 & \partial_s & \partial_f & d & 0 \end{pmatrix},$$
(1.25)

The strain-displacement relation (1.2) can then be written as

$$\mathbf{e} = \nabla [\langle \mathbf{u}_i \rangle - \langle i \rangle - \nabla j_i \sigma_i),$$
 (1.26)

1.2 Dynamical equations

with

$$(\sigma_{12}, \sigma_{22}, \sigma_{$$

The divergence of the stress tensor $\partial_t \sigma_0$ can be expressed as $\nabla \cdot \boldsymbol{\sigma}_i$ and equation (1.23) becomes

$$\nabla \cdot \boldsymbol{\sigma} \sim \mathbf{f} = \rho \partial_{\boldsymbol{\sigma}}^{2} \mathbf{u},$$
 (1.28)

where σ is defined in equation (1.20), and where

$$\mathbf{u} = (u_1, u_2, u_3)$$
 (1.29)

and

$$f = (f_0, f_0, f_0)$$
 (1.30)

Similarly using the matrix notation, the stress-strain relation (1.22) reads

$$\sigma = C = e_{c}$$
 (1.31)

with the clasticity matrix given he

$$\mathbf{C} = \begin{pmatrix} c_{11} & c_{22} & c_{33} & c_{33} & c_{33} & c_{33} \\ c_{12} & c_{23} & c_{23} & c_{23} & c_{33} & c_{23} \\ c_{13} & c_{23} & c_{33} & c_{33} & c_{33} & c_{33} \\ c_{13} & c_{23} & c_{13} & c_{23} & c_{33} & c_{33} \\ c_{13} & c_{23} & c_{13} & c_{13} & c_{23} & c_{33} \\ c_{13} & c_{23} & c_{23} & c_{13} & c_{23} & c_{23} \\ c_{13} & c_{23} & c_{23} & c_{13} & c_{23} & c_{23} \\ c_{13} & c_{23} & c_{23} & c_{13} & c_{23} & c_{23} \\ c_{13} & c_{23} & c_{23} & c_{13} & c_{23} & c_{23} \\ c_{13} & c_{23} & c_{23} & c_{23} & c_{23} \\ c_{13} & c_{23} & c_{23} & c_{23} & c_{23} \\ c_{13} & c_{23} & c_{23} & c_{23} & c_{23} \\ c_{13} & c_{23} & c_{23} & c_{23} & c_{23} \\ c_{13} & c_{13} & c_{13} & c_{13} & c_{13} \\ c_{13} & c_{13} \\ c_{13} & c_{13} \\ c_{13} & c$$

The zero strain state corresponds to static equilibrium with minimum strain energy (1) = 0. Because this energy must always increase when the medium is deformed, we have $e_{IPPP} \ge 0$. Mathematically, this expression involving non-zero components e_{I} defines a positive definite quadratic function, which, by definition, imposes some constraints on the elasticity constants (stability condition) size Auk (1966a, p. 147, and Tueg, 1996, p. 56), namely, all principal determinants should be greater than zero.

$$\epsilon_{1:I} \simeq 0, \quad \det\left(\frac{|c_{I,I}| - |c_{I,I}|}{|c_{I,I}| - |c_{I,I}|}\right) > 0, \quad \dots \quad \det(r_{I,I}) \gg 0, \tag{1.33}$$

Alternatively, the strain-energy density can be expressed in terms of the eigenvalues of matrix **C**, namely, Λ_t , t = 1, ..., 6, called eigenstiffnesses (Kelvin, 1856) (see Section 4.1 m Chapter 4)) that is, $21 = \Lambda_t \mathbf{e}_1 \cdot \mathbf{e}_1$, wherein \mathbf{e}_1 are the eigenvectors or eigenstrains. It as clear that a possive strain energy implies the condition $\Lambda_t > 0$ (see Explan, 1976).

Equations (1.26), (1.28) and (1.33) combine to give

$$\nabla \in \mathbf{C} \cdot (\nabla \to \mathbf{u}_{1}^{*}) * \mathbf{f} = \rho \partial_{\mu}^{*} \mathbf{u}_{1}$$
 (1.34)

112

$$\Gamma_{N} \cdot \mathbf{u} + \mathbf{f} = p \partial_{0}^{*} \mathbf{u}, \quad (\Gamma_{N}, u_{1} + f_{0} - p \partial_{0}^{*} u_{0}), \quad (1.35)$$

where

$$\Gamma_{\mathbf{x}} = \nabla \left(\mathbf{C} \left(\nabla_{(\alpha)} - (\Gamma_{Vij} - \nabla_{ii})_{\alpha} \nabla_{\alpha} \right) \right)$$
(1.36)

is the 3 % 3 symmetric Kelvin-Christoffel differential operator matrix

1.2.1 Symmetries and transformation properties

Differentiation of the strain energies (1.6), (1.7) and (1.8) in accordance with equation (1.18) yields the elasticity matrices for the monorlinic, orthorhombic and transversely isotropic media. Hence, we obtain

$$\mathbf{C}(\text{instruction}) = \begin{pmatrix} c_{11} + c_{12} + c_{13} + 0 + c_{2} + 0 \\ c_{23} + c_{23} + c_{33} + 0 + c_{2} + 0 \\ c_{33} + c_{23} + c_{33} + 0 + c_{23} + 0 \\ 0 + 0 + 0 + c_{33} + 0 + c_{33} + 0 \\ c_{33} + c_{34} + c_{33} + 0 + c_{34} + 0 \\ 0 + 0 + 0 + c_{34} + 0 + c_{34} + 0 \\ c_{44} + c_{44} + c_{44} + 0 + 0 + 0 \\ c_{44} + c_{44} + c_{44} + 0 + 0 + 0 \\ c_{44} + c_{44} + c_{44} + 0 + 0 + 0 \\ 0 + 0 + 0 + c_{34} + 0 + 0 \\ 0 + 0 + 0 + 0 + c_{34} + 0 \\ 0 + 0 + 0 + 0 + c_{34} + 0 \\ 0 + 0 + 0 + 0 + c_{44} \end{pmatrix}$$
(1.38)

and

$$\mathbf{C}^{\text{transversely isotropic}}_{\mathbf{c}_{1}} = \begin{pmatrix} \epsilon - \epsilon_{12} - \epsilon_{23} & 0 & 0 & 0 \\ \epsilon_{11} - \epsilon_{12} & 0 & 0 & 0 \\ \epsilon_{11} - \epsilon_{13} & 0 & 0 & 0 \\ 0 & 0 & 0 & \epsilon_{1} & 0 & 0 \\ 0 & 0 & 0 & 0 & \epsilon_{13} & 0 \\ 0 & 0 & 0 & 0 & \epsilon_{13} & 0 \\ 0 & 0 & 0 & 0 & \epsilon_{13} & 0 \\ 0 & 0 & 0 & 0 & 0 & \epsilon_{13} \end{pmatrix}, \quad 2c_{2} = \epsilon_{3} - c_{42} \quad (1.39)$$

which imply 13, 9 and 5 independent clasticity constants, respectively. In the monoclinic case, the symmetry plane is the 17, 15-plane. A isotation by an angle θ_{\perp} with $\tan(2\theta)$ $2e_{int}(r_{\theta_{\perp}} = r_{in})$ — about the g-axis regulars e_{int} so that the medium can actually be described by 12 elasticity constants. The isotropic case is obtained from the transversely isotropic case, where $r_{\perp} = r_{int} = \lambda - 2\mu$, $r_{int} = r_{int} = \mu$ and $r_{\perp t} = \lambda$, in terms of the Lagritudied systems. The alot mentioned material symmetries are control to describe most of the geological systems at different scales. For example, matrix (1.30) may represent a finely layered medical (see Section 1.5), matrix (1.48) may represent two sets of cracks with crack normals at 90°, or a vertical set of cracks in a finely layered medium, and matrix (1.37) may represent two sets of marks with crack normals other than 30° m 50° (Winterstein, 1990).

Let us consider the conditions of existence for a transversely isotropic medium according to equations (1.33). The first condition implies $c_{22} > 0$, $c_{23} > 0$, $c_{33} > 0$ and $c_{34} > 0$, the second-order determinants maply $e_1^2 = e_1^2 > 0$ and $c_{34} = e_3^2 > 0$, and the relevant third-order determinant implies $te_1^2 = e_1^2 > t_{34} = -c_{34} > 0 = X^2$ these conditions can be combined into

 $\epsilon \sim > 1 + \epsilon_{\rm e} (\epsilon_{\rm e} + \epsilon_{\rm ext}) + 2\epsilon_{\rm e}^2 (\epsilon_{\rm e} > 0)$ (1.40)

he isotrupic media, expressions (1, 10) reduce to

$$3\lambda + 2\mu \pm 0$$
, $-2\mu > 0$, (1.11)

where these sufficiesses are the eigenvalues of reatrix C_{i} the second eigenvalue having a multiplicity of two

It is useful to express explicitly the equations of motion for a particular symmetry that are suitable for informical simulation of wave propagation in inhomogeneous media. The particle-velocity/stress formulation is widely used for this purpose. Consider, for instance, the case of a motion exhibiting monoclinic symmetry. From equations (1.34) and (1.37), we obtain the following expressions.

Particle velocity:

$$\begin{split} \partial_t v_i &= \rho^{-1} \left(\partial_t \sigma_i - \partial_t \sigma_{ij} - \partial_s \sigma_{ij} - f_j \right) \\ \partial_t v_j &= \rho^{-1} \left(\partial_t \sigma_{ij} - \partial_t \sigma_{jj} - \partial_s \tau_{jk} - f_j \right) \\ \partial_t v_j &= \rho^{-1} \left(\partial_t \sigma_{ij} + \partial_t \sigma_{ij} + \partial_s \tau_{ik} - f_j \right), \end{split}$$
(1.42)

Stress

$$\begin{aligned} \partial_t \sigma_1 &= c_0 \partial_t c_1 + c_0 \partial_t c_1 + c_0 \partial_t c_1 + c_0 \partial_t \partial_t c_1 + \partial_t c_1 \\ \partial_t c_{12} &= c_0 \partial_t c_1 + c_2 \partial_t c_1 + c_2 \partial_t c_1 + c_{20} \partial_t c_2 + \partial_t c_1 \\ \partial_t \sigma_{21} &= c_0 \partial_t c_1 + c_0 \partial_t c_2 + c_0 \partial_t c_1 + c_{20} \partial_t c_2 + \partial_t c_1 \\ \partial_t \sigma_{21} &= c_0 \partial_t c_1 + \partial_t c_2 + c_0 \partial_t c_1 + \partial_t c_1 \\ \partial_t \sigma_{21} &= c_0 \partial_t c_1 + c_0 \partial_t c_2 + c_0 \partial_t c_1 + c_0 \partial_t c_1 \\ \partial_t \sigma_{21} &= c_0 \partial_t c_1 + c_0 \partial_t c_1 + c_0 \partial_t c_1 + c_0 \partial_t c_1 \\ \partial_t \sigma_{21} &= c_0 \partial_t c_1 + c_0 \partial_t c_1 + c_0 \partial_t c_1 + c_0 \partial_t c_1 \\ \partial_t \sigma_{21} &= c_0 \partial_t c_1 + c_0 \partial_t c_1 + c_0 \partial_t c_1 + c_0 \partial_t c_1 \end{aligned}$$
(1.43)

where the particle-velocity vector is

$$\mathbf{v} = \{v_1, v_2, v_3\} = i \mathbf{A} \mathbf{u} = (i \mathbf{A} u_1, i \mathbf{A} u_2, i \mathbf{A} u_3, i \mathbf{A} u_4\}$$
 (1.14)

Symmetry plane of a monoclinic medium

In the (x, z)-plane $(\partial_y = 0)$, we identify two sets of incoupled differential equations

$$\begin{aligned} \partial_{t} &= \rho - \left(d |\sigma_{11} + \partial_{1} \sigma_{12} + f \right) \\ \partial_{t} v_{1} &= \rho - \left(d |\sigma_{12} + \partial_{1} \sigma_{11} + f_{1} \right) \\ \partial_{t} \sigma_{1} &= c_{12} d |v_{1} + c_{12} d |v_{1} + c_{22} (\partial_{1} \phi_{1} + \partial_{2} v_{1}) \\ \partial_{t} \sigma_{21} &= c_{22} d |v_{1} + c_{23} d |v_{1} + c_{22} (\partial_{1} \phi_{1} + \partial_{2} v_{1}) \\ \partial_{t} \sigma_{21} &= c_{22} d |v_{1} + c_{23} \partial_{2} v_{1} + c_{22} (\partial_{1} \phi_{2} + \partial_{2} v_{1}) \\ \partial_{t} \sigma_{21} &= c_{22} d |v_{2} + c_{22} \partial_{2} v_{1} + c_{22} (\partial_{1} \phi_{2} + \partial_{2} v_{1}) \end{aligned}$$

and

$$\begin{aligned} \partial \phi_{ij} &= \rho^{-1} \left(\partial_i \sigma_{ij} + \partial_j \sigma_{ij} + f_j \right) \\ \partial \sigma_{ij} &= r_{ij} \partial_i \phi_i + \phi_{ij} \partial_j v_j \\ \partial \phi_{ij} &= r_{ij} \partial_j \psi_j + \phi_{ij} \partial_j v_j \end{aligned} \tag{1.46}$$

The first set describes in-plane particle motion while the second set describes cross-plane particle motion, that is, the propagation of a pure shear wave. A sing the appropriate classicity constants, equations (1,15) and (1,45) hold in the three symmetry planes of an orthorhouslag medium, and at every point of a transversely isotropic medium, by virtue of the azimuthal symmetry around the gravis. The uncoupling implies that a cross-plane shear wave exists at a plane of minor symmetry (Helbig, 2004, p. 142).

Equations (1,12) and (1,43) can be restated as a matrix equation

$$\partial_t \mathbf{v} = \mathbf{H} \cdot \mathbf{v} + \mathbf{f},$$
 (1.47)

where

$$\mathbf{r} = (r_1, r_2, r_3, \sigma_1, \sigma_2, \sigma_3, \sigma_4, \sigma_5, \sigma_4, \sigma_5)$$
 (1.18)

is the 9 × 1 column matrix of the unknown field.

$$\mu \Gamma \sim (f_{11} f_{12} f_{13} 0.0000, 0.000)$$
 (1.49)

and H is the 9 × 9 differential-operator matrix. The formal solution of equation (1.47) is

$$(0) = \exp(10) \cdot y_c + \exp(10) \cdot fy_c$$
 (1.50)

where \mathbf{v}_t is the initial condition. A numerical solution of equation (1.50) requires a polynomial expansion of the so-called evolution operator $\exp(\mathbf{H}t)$ in powers of $\mathbf{H}t$. This is shown in Chapter 9, where the numerical methods are presented.

It is important to distinguish between the principal axes of the material and the Cartesian axis. The principal axes is called crystal axis in crystallography is are entries in axis, that define the symmetry of the mathem. For instance, to ultrain the strain energy (1.7), we have chosen the Cartesian axis in such a way that they entried with the three principal axes defined by the three mutually orthogonal planes of symmetry of the orthorhomble medium. The Cartesian axis may be arbitrarily oriented with respect to the principal axes. It is, therefore, necessary to analyze how the form of the elasticity matrix may be transformed for use in other coordinate systems.

The displacement vector and the strain and stress tensors transform from a system (x, y, z) to a system (x', y', z') as

$$u_1' = u_1 u_1, \quad v_{11}' = u_3 v_0 v_{31}, \quad \sigma_{11}' = u_1 v_0 \sigma_{11}, \quad (1.51)$$

where

$$\mathbf{a} = \begin{pmatrix} a_{11} & a_{22} & a_{33} \\ a_{21} & a_{22} & a_{23} \\ a_{11} & a_{32} & a_{33} \end{pmatrix}$$
(1.52)

is the introgonal transformation matrix. Orthogonality implies $\mathbf{a} \to -\mathbf{a}^+$ and $\mathbf{a} < \mathbf{a} \to -\mathbf{l}_0^+$ det $\mathbf{a} = 1$ for rotations and det $\mathbf{a} = -1$ for inflections. For instance, a clockwise rotation through an angle *H* about the transformations (1,51) provide the reasonal character for the respective physical quantities — first rank in the case of the displacement vector, and second rank in the case of the strain and stress tensors

After converting the stress components to the shortened notation, each component of equation (1.51) must be analyzed individually. Using the symmetry of the stress tensor, we have

$$\sigma' = \mathbf{M} \cdot \sigma_s - (\sigma'_1 + M_{IJ}\sigma_J),$$
 (1.53)

where

$$\mathbf{M} = \begin{pmatrix} a_{11}^{2} & a_{12}^{2} & a_{13}^{2} & 2a_{22}a_{23} & 2a_{23}a_{23} & 2a_{23}a_{23} \\ a_{21}^{2} & a_{22}^{2} & a_{23}^{2} & 2a_{22}a_{23} & 2a_{23}a_{23} & 2a_{23}a_{23} \\ a_{1}^{2} & a_{22}^{2} & a_{23}^{2} & 2a_{22}a_{23} & 2a_{23}a_{23} & 2a_{23}a_{23} \\ a_{2}^{2} u_{31} & a_{22}u_{32} & u_{23}a_{33} & a_{22}a_{33} & a_{32}a_{33} & 2a_{33}a_{3} & 2a_{33}a_{3} \\ a_{2}^{2} u_{31} & a_{22}u_{32} & u_{23}a_{33} & a_{22}a_{33} & a_{33}a_{33} & a_{33}a_{33} & a_{23}a_{33} \\ a_{3}^{2} u_{31} & a_{22}u_{32} & u_{32}a_{33} & a_{22}a_{33} & a_{31}u_{32} & a_{32}u_{33} & u_{32}a_{33} & u_{32}a_{33} \\ a_{3}^{2} u_{31} & a_{31}u_{33} & a_{31}u_{33} & a_{31}u_{33} & a_{31}u_{33} & a_{32}u_{33} & u_{32}u_{33} \\ a_{3}^{2} u_{32} & a_{31}u_{33} & a_{31}u_{33} & a_{31}u_{33} & a_{31}u_{33} & u_{32}u_{33} & u_{32}u_{33} \\ a_{3}^{2} u_{32} & a_{31}u_{33} & a_{31}u_{33} & a_{31}u_{33} & a_{31}u_{33} & u_{32}u_{33} & u_{32}u_{33} \\ a_{3}^{2} u_{32} & a_{31}u_{33} & a_{31}u_{33} & a_{31}u_{33} & u_{31}u_{33} & u_{31}u_{33} & u_{31}u_{33} & u_{32}u_{33} \\ a_{3}^{2} u_{32} & a_{31}u_{33} & a_{31}u_{33} & u_{31}u_{33} & u_{31}u_{33} & u_{31}u_{33} & u_{31}u_{32} & u_{32}u_{33} \\ a_{3}^{2} u_{31} & a_{31}u_{33} & u_{31}u_{33} & u_{31}u_{33} & u_{31}u_{33} & u_{31}u_{32} & u_{31}u_{32} & u_{31}u_{32} \\ a_{3}^{2} u_{31} & u_{31}u_{33} & u_{31}u_{33} & u_{31}u_{33} & u_{31}u_{32} & u_{31}u_{32} & u_{31}u_{32} & u_{31}u_{32} \\ a_{3}^{2} u_{31} & u_{31}u_{33} & u_{31}u_{33} & u_{31}u_{31}u_{32} & u_{31}u_{32} &$$

(Auld, 1990a, p. 74). Due to the 1/2 factor in ϵ_{ij} (secondation (1.3)), the transformation matrix for the strain component is different from **M**. We have

$$e^{i} = N_{c}e_{c} - b_{T}^{i} - N_{D}(j),$$
 (1.55)

where

$$\mathbf{N} = \begin{pmatrix} a_{1}^{(i)} & a_{21}^{(i)} & a_{21}^{(i)} & u_{12}a_{11} & a_{12}u_{11} & u_{12}a_{21} \\ a_{21}^{(i)} & a_{22}^{(i)} & a_{21}^{(i)} & u_{12}a_{21} & a_{21}u_{12} & u_{22}a_{21} \\ a_{21}^{(i)} & a_{22}^{(i)} & a_{21}^{(i)} & u_{12}a_{21} & a_{22}u_{12} & u_{12}a_{22} \\ a_{21}^{(i)} & a_{22}^{(i)} & a_{22}^{(i)} & u_{22}a_{21}u_{22} & u_{22}a_{21}u_{22} & u_{22}a_{22}u_{21} & u_{22}a_{22}u_{21} & u_{22}a_{22}u_{22} & u_{22}a_{22}u_{22} & u_{22}a_{22}u_{22} & u_{22}a_{22}u_{22} & u_{22}a_{22}u_{22} & u_{22}u_{22}u_{22} & u_{22}u_{22}u_{22} & u_{22}u_{22}u_{22}u_{22}u_{22} & u_{22}u_{22$$

(Ankl. 1990a, p. 75). Matrices M and N are called Bond matrices after W. L. Bond who developed the approach from which they are obtained.

Let us now find the transformation law for the elasticity tensor from one system to the other. From equations (1.31), (1.53) and (1.55), we have

$$\sigma' = C' \cdot \phi', \quad C' = M \cdot C \cdot N^{-1}, \quad (1.57)$$

Because neutric π in (1.52) is orthogonal, the matrix N $_{\odot}$ can be found by transposing all subscripts in equation (1.55). The result is simply \mathbf{M}_{\odot} and (1.55) because

$$C' \simeq M \cdot C \cdot M$$
 (1.58)

Transformation of the stiffness matrix

In the current sensible terminology, a transversely isotropic modum refers to a medium represented by the elasticity matrix (1.39), with the symmetry axis along the vertical direction (i.e., the plaxis). By performing appropriate rotations of the conducate system the midman may become azimuthally anisatiophe (e.g., Thomson, 1988). An example is a transversity isotropic medium whose symmetry axis is invitantal and makes an angle θ with the *x*-axis. To obtain this medium, we perform a clockwise rotation by $\tau/2$ about the *y*-axis followed by a connect lockwise rotation by θ about the new t-axis. The corresponding rotation matrix is given by

$$\mathbf{a} = \begin{pmatrix} \cos\theta & \sin\theta & 0\\ \sin\theta & \cos\theta & 0\\ 0 & 0 & 1 \end{pmatrix} \cdot \begin{pmatrix} 0 & 0 & 1\\ 0 & 1 & 0\\ 1 & 0 & 0 \end{pmatrix} = \begin{pmatrix} 0 & \sin\theta & \cos\theta\\ 0 & \cos\theta & \sin\theta\\ 1 & 0 & 0 \end{pmatrix}, \quad (1.59)$$

The corresponding Road transformation matrix is

$$\mathbf{M} = \begin{pmatrix} \mathbf{0} & \sin^2\theta & \cos^2\theta & \sin(2\theta) & 0 & 0 \\ \mathbf{0} & \cos^2\theta & \sin^2\theta & \sin(2\theta) & 0 & 0 \\ 1 & 0 & 0 & 0 & 0 & 0 \\ \mathbf{0} & 0 & 0 & 0 & \sin\theta & \cos\theta \\ \mathbf{0} & 0 & 0 & 0 & \cos\theta & \sin\theta \\ \mathbf{0} & (\sin(2\theta) & \sin(2\theta) & \cos(2\theta) & 0 & 0 \end{pmatrix}, \quad (1.60)$$

Using (1.58), we note that the elasticity constants in the new system are

$$\begin{aligned} c_{11}^{2} &= s_{12}\cos^{2}\theta + z_{11}(z_{2} + 2\cos^{2}\theta) + z_{12}\sin^{2}\theta \\ c_{22}^{2} &= z_{12}\cos^{2}\theta + z_{12}\sin^{2}\theta \\ c_{31}^{2} &= s_{12}\cos^{2}\theta + z_{12}\sin^{2}\theta \\ c_{31}^{2} &= z_{31}\cos^{2}\theta + z_{12}\sin^{2}\theta \\ c_{32}^{2} &= z_{41}\cos^{2}\theta + z_{12}\sin^{2}\theta \\ c_{31}^{2} &= z_{41}\cos^{2}\theta + z_{12}\sin^{2}\theta \\ c_{31}^{2} &= z_{41}\cos^{2}\theta + z_{12}\sin^{2}\theta \\ c_{31}^{2} &= z_{41}\cos^{2}\theta + z_{13}\sin^{2}\theta \\ c_{31}^{2} &= z_{41}\cos^{2}\theta + z_{41}\sin^{2}\theta \\ c_{31}^{2} &= z_{41}\cos^{2}\theta + z_{41}\sin^{2}\theta \\ c_{31}^{2} &= z_{41}\cos^{2}\theta + z_{42}\sin^{2}\theta \\ c_{31}^{2} &= z_{41}\cos^{2}\theta + z_{52}\sin^{2}\theta \\ c_{31}^{2} &= z_{41}\cos^{2}\theta + z_{52}\cos^{2}\theta \\ c_{31}^{2} &= z_{41}\cos^{2}\theta + z_{52}\cos^{2}\theta \\ c_{31}^{2} &= z_{42}\cos^{2}\theta + z_{53}\cos^{2}\theta \\ c_{32}^{2} &= z_{42}\cos^{2}\theta + z_{53}\cos^{2}\theta \\ c_{31}^{2} &= z_{42}\cos^{2}\theta + z_{53}\cos^{2}\theta \\ c_{32}^{2} &= z_{42}\cos^{2}\theta + z_{53}\cos^{2}\theta \\ c_{32}^{2} &= z_{42}\cos^{2}\theta + z_{53}\cos^{2}\theta \\ c_{32}^{2} &= z_{53}\cos^{2}\theta \\ c_{33}^{2} &= z_{53}\cos^{2}$$

and the other components are equal to zero

1.3 Kelvin-Christoffel equation, phase velocity and slowness

A plane-wave analysis yields the Kelvin-Christoffel equations and the expressions for the phase velocity and slowness of the different wave randow. A general phane-wave subting for the displacement vector of hody waves is

$$\mathbf{u} = \mathbf{u} \exp[(\mathbf{u} \mathbf{x} + \mathbf{x})] \qquad (1.62)$$

where **u**, represents a constant complex vector, ω is the angular frequency and κ is the wavenumber vector or wavevector. We recall that when using complex notation for plane waves, the field variables are obtained as the real part of the corresponding wave fields. The particle velocity is given by

$$\mathbf{v} = \partial_t \mathbf{u} = i_0 u_0$$
 (1.63)

In the absence of body forces $(\mathbf{f} = 0)$, we consider plane waves propagating along the direction

$$\kappa = l_1 \mathbf{e} \rightarrow l_2 \mathbf{e}_1 + G \mathbf{e}_3$$
, (1.64)

(or (l_1, l_2, l_4)), where l_1, l_2 and l_3 are the direction cosines. We have

$$\kappa = (\mu_1, \mu_2, c_3) = \kappa (l_1, l_2, l_3) = n\kappa, \qquad (1.65)$$

where s is the magnitude of the wavevector. In this case, the time derivative and the spatial differential operator (1.25) can be replaced by

and

$$\nabla = i \cdot \left[i \cdot \left[\begin{array}{cccc} i & 0 & 0 & 0 & l_{0} & i_{0} \\ 0 & l_{0} & 0 & l_{0} & 0 & l_{0} \\ 0 & 0 & l_{0} & l_{0} & l_{0} & 0 \end{array} \right] = -i i \cdot \mathbf{L}, \tag{1.67}$$

respectively

Substitution of these operators into the equation of motion (1.35) yields

$$n^{\mu} \mathbf{\Gamma} \cdot \mathbf{u} = \rho \omega^{\mu} \mathbf{u}, \quad (\kappa^{\mu} V_{\mu} u_{\mu} - \rho \omega^{\mu} u_{\mu}), \tag{1.68}$$

where

$$\mathbf{\Gamma} = \mathbf{I}_{0} \cdot \mathbf{C} \cdot \mathbf{I}_{0} = - (\mathbf{\Gamma}_{0} - J_{0} c_{0} d_{0}) \qquad (1.60)$$

is the symmetric Kelvin Christoffel matrix . Defining the phase velocity vector as

$$\mathbf{v}_{p} = v_{p} \mathbf{\kappa}, \qquad v_{p} = \frac{\pi}{\kappa}, \qquad (1.70)$$

we had that equation (1.68) becames an "eigenoquation" (the Kelvin-Christoffel equation)

$$\langle \Gamma - \rho r \langle \Gamma_i \rangle \cdot \mathbf{u} = 0 \qquad (1.71)$$

for the eigenvalues $(\rho c_p)_m$ and eigenvectors $(\mathbf{u})_m$, m = 1, 2, 3. The dispersion relation is given by

$$det(\Gamma - \rho_0 (\Gamma_1) = 0)$$
 (1.72)

In explicit form, the components of the Kelvin Christoffel matrix are

$$\begin{split} \Gamma_{1} &= c_{1} l_{1}^{2} + c_{2} l_{1}^{2} + c_{2} l_{1}^{2} + 2 c_{3} l_{4} l_{5} + 2 c_{3} l_{4} l_{5} + 2 c_{3} l_{4} l_{5} \\ \Gamma_{12} &= c_{2} l_{4}^{2} + c_{3} l_{2}^{2} + c_{3} l_{4}^{2} + 2 c_{3} l_{4} l_{5} + 2 c_{3} l_{4} l_{5} + 2 c_{3} l_{4} l_{5} \\ \Gamma_{13} &= c_{2} l_{3}^{2} + c_{3} l_{5}^{2} + c_{4} l_{4}^{2} + 2 c_{3} l_{4} l_{5} + 2 c_{3} l_{4} l_{5} + 2 c_{3} l_{4} l_{5} \\ \Gamma_{13} &= c_{3} l_{3}^{2} + c_{3} l_{5}^{2} + c_{4} l_{4}^{2} + l_{4} l_{5} l_{5} + l_{5} l_{5} l_{5} + l_{5} l_{5} l_{5} \\ \Gamma_{13} &= c_{3} l_{3}^{2} + c_{3} l_{5}^{2} + c_{4} l_{5}^{2} + l_{4} l_{5} l_{5}^{2} + l_{4} l_{5} l_{5} + l_{5} l_{5} l_{5} l_{5} \\ \Gamma_{13} &= c_{3} l_{5}^{2} + c_{3} l_{5}^{2} + c_{4} l_{5}^{2} + l_{4} l_{5} l_{5} + l_{6} l_{5} + c_{4} l_{5} l_{5} l_{5} \\ \Gamma_{13} &= c_{3} l_{5}^{2} + c_{5} l_{5}^{2} + c_{4} l_{5}^{2} + l_{6} l_{5} l_{5} + c_{4} l_{5} l_{5} l_{5} \\ \Gamma_{14} &= c_{3} l_{5}^{2} + c_{5} l_{5}^{2} + c_{4} l_{5}^{2} + l_{6} l_{5} l_{5} + l_{6} l_{5} l_{5} + l_{6} l_{5} l_{5} l_{5} \\ \Gamma_{14} &= c_{4} l_{5}^{2} + c_{5} l_{5}^{2} + c_{4} l_{5}^{2} + l_{6} l_{5} l_{5} + l_{6} l_{5} l_{5} l_{5} l_{5} \\ \Gamma_{14} &= c_{4} l_{5}^{2} + c_{5} l_{5}^{2} + c_{4} l_{5}^{2} + l_{6} l_{5} l_{5} l_{5} l_{5} l_{5} \\ \Gamma_{14} &= c_{4} l_{5} l_{5}$$

The three solutions obtained from considering m = 1, 2, 3 correspond to the three body waves propagating in an unbounded homogeneous medium. At a given frequency ω , $v_p(t_1, b_2, t_3)$ defines a surface in the wavenumber space as a function of the direction cosines. The slowness is defined as the inverse of the phase vehicity, namely as

$$s = \frac{\kappa - 1}{- - \gamma}$$
. (1.73)

Similarly, we can define the slawness surface $s(l_1, l_2, l_3)$. The slawness vector is clusely related to the wavevector by the expression

$$\mathbf{s} = \frac{\mathbf{\kappa}}{s} = s\mathbf{\kappa} \tag{1.75}$$

1.3.1 Transversely isotropic media

Let us consider wave propagation in a plane containing the scenarity axis () savis) of a transversely isotropic medium. This problem illustrates the effects of anisotropy on the velocity and polarization of the hody waves. For propagation in the (x_{ij}) (plane, $l_{ij} \in 0$ and equation (1.7)) (refines to

$$\begin{pmatrix} \Gamma_{1} & \rho e_{\mu}^{2} & 0 & \Gamma_{13} \\ 0 & \Gamma_{10} & \rho e_{\mu}^{2} & 0 \\ -\Gamma_{11} & 0 & \Gamma_{13} & -\rho e_{\mu}^{2} \end{pmatrix} + \begin{pmatrix} u \\ v_{1} \\ v_{1} \end{pmatrix} = 0.$$
(1.76)

чľ

$$\begin{pmatrix} c_{13}r_{1}^{2} + c_{23}J_{1}^{2} - \rho c_{p}^{2} & 0 & (c_{23}r_{1}^{2} + c_{23}d_{1}^{2}) \\ 0 & (c_{23}r_{1}^{2} + c_{23}J_{1}^{2} - \rho c_{p}^{2}) & 0 \\ - (c_{13}r_{1}^{2} + c_{23}J_{1}^{2}) & 0 & (c_{33}J_{3}^{2} + c_{23}J_{1}^{2} + c_{23}J_{1}^{2}) \end{pmatrix} + \begin{pmatrix} a_{1} \\ a_{2} \\ a_{3} \\ a_{4} \end{pmatrix} = 0, (c_{13}r_{1}^{2})$$

We obtain two moonpled dispersion relations

$$\begin{aligned} c_{\mu}F + c_{\mu}f &= \rho f = 0\\ (c_{\mu}F + c_{\mu}f)(c_{\mu}f + c_{\mu}f) &= (c_{\mu} + c_{\mu})(f)F = 0, \end{aligned} \tag{1.78}$$

giving the phase velocities

$$\begin{aligned} v_{\mu\nu} &= \sqrt{(\mu_{\nu}^{-1}) c_{\mu} b^{\nu} + c_{\mu} b^{\nu}}, \\ v_{\mu\nu} &= 2\mu b^{\nu} + \sqrt{(\mu_{\nu} b^{\nu} + c_{\mu} b^{\nu}_{\mu} + c_{\mu} b^{\nu}_{\mu} + c_{\mu} b^{\nu}_{\mu})}, \\ v_{\mu\nu} &= 2\mu b^{\nu} + \sqrt{(\mu_{\nu} b^{\nu}_{\mu} + c_{\mu} b^{\nu}_{\mu} + c_{\mu} b^{\nu}_{\mu} + c_{\mu} b^{\nu}_{\mu})}, \\ C &= \sqrt{(\mu_{\nu}^{-1} + c_{\mu} b^{\nu}_{\mu} + c_{\mu} b^{\nu}_{\mu} + c_{\mu} b^{\nu}_{\mu})}, \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{\mu}), \\ b^{\nu} &= c_{\mu\nu} b^{\nu}_{\mu} + (c_{\mu\nu}^{-1} + c_{\mu\nu} b^{\nu}_{$$

From reptation (1.77), we see that the first solution has a displacement for polarization; given by $(0, v_0, 0)$, which is normal to the 4x, 0-plane of propagation. Therefore, this solution describes a pure shear wave – termed SH wave in the graphysical iterature with H denoting horizontal polarization of the graves is oriented in the vertical direction. Note that the dispersion relation (1.78), can be written as

$$\frac{S_{\rm eff}^2}{p/m_{\rm eff}} = \frac{S_{\rm eff}^2}{p/m_{\rm eff}} = 1,$$
 (1.80)

where $s_1 = s^{-1}l_1$ and $s_2 = s^{-1}l_2$ with $s^{-1} = 1/r_p$. Hence, the slowness surface is an ellipse, with semiaxes p/r_{th} and p/r_{th} along the x_2 and collipsetions, respectively.

For the coupled waves, the normalized polarizations are obtained from equation (4.76) by using the dispersion relation (4.78). Hence, we obtain

$$\begin{pmatrix} u_{i} \\ u_{i} \end{pmatrix} = \frac{1}{\sqrt{\Gamma_{i}} - \Gamma_{ii} - 2\rho v_{j}^{2}} \begin{pmatrix} \sqrt{\Gamma_{ii}} - \rho v_{j}^{2} \\ \sqrt{\Gamma_{ii}} - \rho v_{j}^{2} \end{pmatrix}, \qquad (1.81)$$

Using the fact that $E_i + E_i = 1$, as well as equations (1.79) and (1.81), we can identify the wave modes along the *x* and - axes, which may be written as

$$\begin{aligned} i &= \operatorname{axis} (l_1 = 1), \quad pv_{0,1}^{(i)} = v_{1,1}, \quad u_1 = 0, \quad s \in \operatorname{S} \text{ ware} \\ i &= \operatorname{axis} (l_1 = 1), \quad pv_{0,2}^{(i)} = v_{1,2}, \quad u_1 = 0, \quad s \in \operatorname{P} \text{ wave} \\ \operatorname{axis} (l_1 = 0), \quad pv_{1,2}^{(i)} = e_{2,2}, \quad u_2 = 0, \quad s \in \operatorname{S} \text{ wave} \\ \operatorname{axis} (l_1 = 0), \quad pv_{1,2}^{(i)} = e_{2,2}, \quad u_3 = 0, \quad s \in \operatorname{P} \text{ wave}, \end{aligned}$$
(1.82)

These expressions denote pure mode directions for which the palarization of the P waveconnides with the wavevector direction and the S wave palarization is normal to this direction. There exists another pure mode direction defined by

$$r_{80}(\theta = \frac{c_0}{c_0} + \frac{2c_0}{c_0} + \frac{c_0}{c_0}, \quad b = \operatorname{arcsn}_0 l_0$$
 (1.83)

(Brugger, 1965), which extends azimuthally about the t-axis. The polarizations along the other directions are not parallel or perpendicular to the propagation directions and, therefore, the waves are termed quasi P and quasi S. The latter is usually called the qSV wave, with V denoting the vertical plane of the varies is oriented in the vertical direction. The (x,y)-plane of a transversely isotropic medime is a plane of isotropy, where the velocity of the SV wave is $\sqrt{e_0}$, p and the velocity of the SII wave is $\sqrt{e_0}/p$. The velocity of the compressional wave is $\sqrt{e_0}/p$.

1.3.2 Symmetry planes of an orthorhombic medium

In the symmetry planes of an orthorhomble medium, the physics of wave propagation is similar to the previous case, i.e., there is a pure shear wave dabeled 1 below) and two coupled waves.

The respective slowness surfaces are:

$$(x,y)$$
 plane $(T = \sin \theta / t_y - \cos \theta)$:

$$\begin{aligned} \epsilon_{N} J_{1}^{\prime} &= \epsilon_{N} J_{2}^{\prime} - \rho r_{p}^{\prime} = 0 \\ \epsilon_{N} J_{1}^{\prime} &= \epsilon_{N} J_{2}^{\prime} - \rho r_{p}^{\prime} (\epsilon_{N} J_{2}^{\prime} + c_{N} J_{1}^{\prime} - \rho r_{p}^{\prime}) - (\epsilon_{N} - \epsilon_{N})^{\prime} J_{1}^{\prime} J_{p}^{\prime} = 0; \end{aligned} \tag{1.84}$$

(*i.e.*) optime $|I_1| = \sin \theta, |I_1| = \cos \theta$ (:

$$\frac{c_{\mu}l^{\mu} + c_{\lambda}d^{\mu}_{\mu} - \rho c_{\mu}^{\mu} = 0}{(r_{\mu}l^{\mu} + c_{\lambda}l^{\mu}_{\mu} + c_{\lambda}l^{\mu}_{\mu} - \rho c_{\mu}^{\mu}) + (c_{\mu}r_{\mu} + c_{\lambda})^{\mu}Fl^{\mu}_{\mu} = 0}$$
(1.85)

(y, z)-plane $(I_{z} = \sin \theta, I_{z} = \cos \theta)$

$$e_{\mu\nu}l_{\mu}^{\mu} + e_{\mu}l_{\mu}^{\mu} = 0$$

 10.86 $e_{\mu}l_{\mu}^{\mu} = 0, \qquad (1.86)$
 $e_{\mu}l_{\mu}^{\mu} + e_{\mu}l_{\mu}^{\mu} + e_{\mu}l_{\mu}^{\mu} = 0,$

The corresponding phase velocities are:

ta syt plants

$$\begin{aligned} c_{12} &= \sqrt{(\mu^{1/2}) \cos((-e\cos k))} \\ c_{13} &= (2\mu^{1/2}) \sqrt{e_{1}(1+e_{2}k)} + c_{2} + C \\ c_{13} &= (2\mu^{1/2}) \sqrt{e_{1}(k)} + e_{2}k) + c_{2} + C \\ C &= \sqrt{(1+e_{1})} k^{2} + (1+e_{2}k)^{2} + c_{2} + C \\ C &= \sqrt{(1+e_{1})} k^{2} + (1+e_{2}) k^{2} + (1+e_{2}) k^{2} + C \end{aligned}$$
(1.87)

tratification (http://www.com/

$$\begin{aligned} v_{ij} &= \sqrt{(\mu^{-1}) (i_0 d_1^j + c_0 d_0^j)} \\ v_{ij} &= (2i_0^j - (\sqrt{c_0} d_1^j + c_0 d_1^j + c_0^j - C) \\ v_{ij} &= (2i_0^j - (\sqrt{c_0} d_1^j + c_0 d_1^j + c_0^j + C) \\ C &= \sqrt{(c_0^j - (c_0^j d_1^j + c_0^j d_1^j + c_0^j d_1^j))} \end{aligned}$$

$$(1.88)$$

the deplaces

$$\begin{aligned} v_{11} &\approx \sqrt{(\rho^{1/2})} v_{02} l_{1}^{2} + c_{22} l_{1}^{2} \\ v_{12} &= (2\rho_{1} + \sqrt{c_{22}} l_{1}^{2} + c_{32} l_{1}^{2} + c_{32} l_{1}^{2} + c_{32} - C \\ v_{13} &= (2\rho_{1} + \sqrt{c_{22}} l_{1}^{2} + c_{32} l_{1}^{2} + c_{32} + C \\ C &= \sqrt{(c_{33} + c_{32}) l_{1}^{2} + (c_{33} + c_{33}) l_{1}^{2} l_{1}^{2} + (c_{33} + c_{33}) l_{1}^{2} l_{1}^{2}, \end{aligned}$$
(1.89)
Angle θ is measured from the y-axis in the nz. y -plane, and from the t-axis in the following (y,z)-planes

The vehicities along the principal axes are:

(2.5g)-planes

$$\begin{aligned} c_{\mu}(0) &= c_{\mu}(0) = \sqrt{c_{\mu}}\rho \\ c_{\mu}(0) &= c_{\mu\nu}(0) = \sqrt{c_{\mu\nu}}\rho \\ c_{\mu}(0) &= c_{\mu\nu}(0) = \sqrt{c_{\mu\nu}}\rho \\ c_{\mu}(0) &= c_{\mu\nu}(0) = \sqrt{c_{\mu\nu}}\rho \end{aligned}$$
(130)

exist-plane:

$$\begin{aligned} v_{\mu}(W) &= v_{\mu}(W) = \sqrt{v_{\mu}(\rho)} \\ v_{\mu}(W) &= v_{\mu\nu}(W) = \sqrt{v_{\mu}(\rho)} \\ v_{\mu}(W) &= v_{\mu\nu}(W) = \sqrt{v_{\mu}(\rho)} \\ v_{\mu}(W) &= v_{\mu\nu}(W) = \sqrt{v_{\mu}(\rho)} \end{aligned}$$

(q. , oplane:

$$\begin{aligned} v_{\mu}(0) &= v_{\mu}(0) = \sqrt{c_{\mu}/\rho} \\ v_{\mu}(0) &= v_{\mu\nu}(0) = \sqrt{c_{\mu}/\rho} \\ \end{aligned}$$
(1.92)

1.3.3 Orthogonality of polarizations

In order to determine if the polarizations of the waves are orthogonal, we consider two solutions "a" and "b" of the eigensystem (1.71).

$$\mathbf{\Gamma} \cdot \mathbf{u}_{0} = \rho c_{ps}^{2} \mathbf{u}_{0}, \quad \mathbf{\Gamma} \cdot \mathbf{u}_{b} = \rho c_{ps}^{2} \mathbf{u}_{b}, \quad (1.93)$$

and take the scalar product from the left hand side with the displatements u_i and u_a , respectively.

$$\mathbf{u}_{b} \cdot \mathbf{\Gamma} \cdot \mathbf{u}_{b} = \rho v_{jb}^{2} \mathbf{u}_{b} \cdot \mathbf{u}_{b}, \quad (\mathbf{u}_{b} \cdot \mathbf{\Gamma} \cdot \mathbf{u}_{b} = \rho v_{jb}^{2} \mathbf{u}_{b} \cdot \mathbf{u}_{b}$$
(1.94)

Since Γ is symmetric, we have $u_n:\Gamma \cdot u_n = u_n \cdot \Gamma \cdot u_n$. Subtracting one equation from the other, we get

$$\rho(r_{tac}^2 = r_p^2) \mathbf{u}_{tac} (\mathbf{u}_{tac} = 0)$$
 (1.95)

If the phase velocities are different, we have $\mathbf{u}_{ij}(\mathbf{u}) = 0$ and the polarizations are orthogonal. Note that this property is a consequence of the symmetry of the Kelvin-Christoffel matrix

1.4 Energy balance and energy velocity

Energy-balance expressions are important for characterizing the energy stored and the transport properties in a field. In particular, the concept of energy velocity is useful in determining how the energy transferred by the wave held is related to the strength of the field, i.e., the location of the wave front. Although, in bashess medic, this velocity can be ultrained from "kinematic" considerations – we shall see that the group and the energy velocities are the same – an analysis of this media provides a basis to study more complex situations, such as wave propagation in anchestic and porous media.

The equation of motion (1.28) corresponding to the plane wave (1.62) is

$$i_{\ell} L + \sigma \approx i_{\ell} \rho v_{\ell}$$
 (1.96)

where we assumed no body forms and used equations (1.66) and (1.67). The scalar product of $-x^2$ and equation (1.69) is

$$\kappa \mathbf{v}^{*} + \mathbf{I}_{\mathbf{r}} \cdot \boldsymbol{\sigma} = \omega \rho \mathbf{v}^{*} + \mathbf{v}$$
 (1.97)

Moreover, the strain-displacement relation (1.26) is replaced by

$$p_{\rm e} = r_{\rm L} + v_{\rm e}$$
 (198)

The scalar product of the complex conjugate of equation (1.98) and σ_{\pm} gives

$$s\sigma = (\mathbf{L} + \mathbf{v}^{\dagger}) \otimes \sigma^{\dagger} + e^{t}$$
 (199)

The left-hard sides of equations (1.97) and (1.99) commite and can be written in terms of the Uniov-Proteing vertex lags power-flow vectors.

$$\mathbf{p} = -\frac{1}{2} \begin{pmatrix} \sigma & \sigma_{12} & \sigma_{13} \\ \sigma_{12} & \sigma_{22} & \sigma_{23} \\ \sigma_{13} & \sigma_{34} & \sigma_{13} \end{pmatrix} \cdot \mathbf{v}^{\prime}$$
(1.106)

(up

$$2\kappa \cdot \mathbf{p} = \omega_0 \mathbf{v}^* \cdot \mathbf{v}$$
 (1.00)

and

$$2\kappa \cdot p \approx \pm \sigma \to e^{\epsilon}$$
. (1.102)

Adding equations (1,101) and (1,102), we get

$$\mathbf{t} \mathbf{c} \cdot \mathbf{p} = \omega_Q \mathbf{v}^* \cdot \mathbf{v} + \boldsymbol{\sigma} = (\mathbf{e}^*),$$
 (1.104)

m, using the stress strain relation (1.31) and the symmetry of C_{γ} we obtain

$$4\kappa \cdot p \ge \omega_0 \kappa^2 \cdot \kappa + c \ge C \cdot c^2$$
 (1.104)

For generic field variables **a** and **b**, and a symmetry matrix **D**, the time average over a cycle of period $2\pi/\omega$ has the following properties:

$$\langle \operatorname{Re}(\mathbf{a}_{-1}) | \operatorname{Re}(\mathbf{b}) \rangle = \frac{1}{2} \operatorname{Re}(\mathbf{a}_{-1}) | \mathbf{b}^* \rangle$$
 (1.105)

Assetue of the density of energy flax introduced independently ity N=1 in or in 1871 and [1] R vieting in 1881 (Meksey, 1986).

(Booker, 1992), and

$$\begin{aligned} &[\operatorname{Reta}^{+}(\cdot \operatorname{Re}(\mathbf{D}) + \operatorname{Re}(\mathbf{a})) &\simeq \frac{1}{2}\operatorname{Reta}^{+}(\mathbf{D} + \mathbf{a}^{*}), \\ &[\operatorname{Reta}^{+}(\cdot \operatorname{Im}_{\mathbf{D}}\mathbf{D}) + \operatorname{Re}(\mathbf{a})) &= [\operatorname{Im}_{\mathbf{D}}\mathbf{a}^{+} + \mathbf{D} + \mathbf{a}^{*}). \end{aligned} \tag{1.1080}$$

(Carcinum and Cavallini, 1993). Using equation (1/105), we obtain the firm overage of the real Uniov Privating vector (1,100), namely

$$Re(\Sigma) \cdot Re(v)$$
, (1.107)

where

$$\Sigma = \begin{pmatrix} \sigma_{11} & \sigma_{12} & \sigma_{13} \\ \sigma_{12} & \sigma_{22} & \sigma_{23} \\ \sigma_{13} & \sigma_{13} & \sigma_{33} \end{pmatrix}$$
(1.108)

is.

$$|\mathbf{p}\rangle = \text{Re}(\mathbf{p}).$$
 (1.109)

which represents the conjustule and direction of the transactorized power line

We identify, in equation (1.404), the time averages of the kinetic (and strain energy densities, namely,

$$\langle T \rangle \approx \frac{1}{2} \langle \operatorname{Re} \mathbf{v} (\cdot \operatorname{Re} (\mathbf{v}) \rangle \approx \frac{1}{4} \operatorname{Re} (\mathbf{v}' \cdot \mathbf{v})$$
 (1.110)

and

$$\langle V \rangle = \frac{1}{2} (\operatorname{Re}(\mathbf{e}_{-}) \cdot \mathbf{C} \cdot \operatorname{Re}(\mathbf{e})) = -\frac{1}{4} \operatorname{Re}(\mathbf{e}_{-} \cdot \mathbf{C} \cdot \mathbf{e}^{*}),$$
 (1.111)

The substitution of opportions (1.111) and (1.111) into the real part of equation (1.104) yields the energy-balance equation

$$\kappa_{\gamma}(\mathbf{p}) = \omega(T + V) (-\omega_{\gamma}(T + V)) - \omega(E),$$
 (1.112)

where $\langle E \rangle$ is the time-averaged energy density.

The wave surface is the heirs of the end of the energy-velocity vector multiplied by one mut of propagation time, with the energy-velocity vector defined as the ratio of the time averaged power flow vector (\mathbf{p}_{\perp} to the total energy density (E). Because this is equal to the sum of the time-averaged kinetics and strain-energy densities (T) and (V), the energy-velocity vector is

$$\mathbf{v}_{t} = \frac{\langle \mathbf{p} \rangle - \langle \mathbf{p} \rangle}{\langle E \rangle - \langle T + V \rangle}$$
, (1.113)

Using this definition, we note that equation (1.142) gives

$$\kappa (v_i = i_{p_i} = (s (v_i = 1)))$$
 (1.114)

where v_{μ} and s are the phase velocity and slowness vector defined in equations (1.75), and (1.75), respectively. Relation (1.111) means that the phase velocity is equal to the projection of the energy velocity onto the propagation direction. The wave front is associated with the higher energy velocity. Since, in the elastic case, all the wave surfaces have the same velocity – there is no velocity dispersion – the concepts of wave front and wave surface are the same – In anclastic media, the wave front is the wave surface associated with the nucleaved energy velocity.

Equation (1.113) allows further soupliheations. Let us calculate the time averages of the kinetic and strain energies explicitly. The substitution of equation (1.62) into equation (2.110) yields

$$\langle T \rangle > \frac{1}{4} \rho c^2 |\mathbf{u}_1|^2,$$
 (1.135)

From equations (1.26) and (1.67), we have

which implies

$$\mathbf{e} = (\mathbf{C} \cdot \mathbf{e}^{*} - \kappa^{*} \mathbf{u} \cdot \mathbf{L} \cdot \mathbf{C} \cdot \mathbf{L} - (\mathbf{u}^{*} - \kappa^{*} \mathbf{u} \cdot \mathbf{U} \cdot \mathbf{u}^{*})$$

$$(1.117)$$

where we have used equation (1.09). In view of the complex emgigate of equation (1.08), equation (1.117) can be written as

$$\mathbf{e} = \mathbf{C} \cdot \mathbf{e}^* - \rho \omega^* \mathbf{u} \cdot \mathbf{u}^* - \rho \omega^* \mathbf{u} \cdot \hat{\mathbf{u}}^*$$
 (1.118)

Using this relation, we find that the time-averaged strain-energy density (1,111) becomes

$$\langle V \rangle = \frac{1}{4} \rho \omega^2 |\mathbf{u}_{c}|^2 - \langle T \rangle$$
 (1.19)

Hence, in elastic media, the time averages of the strain, and kinetic energy densities are optal and the chergy-velocity vector (1.113) can be simplified to

$$\mathbf{x}_{i} = \frac{\langle \mathbf{p} \rangle}{2\langle T \rangle}$$
(1.120)

It can be shown that for a traveling wave, whose angument is $t = s \cdot \mathbf{x}$, the plane wave (1.62) is a particular case. The instantaneous kinetics and strain-energy densities are the same. On the other hand, an evolving of kinetic and potential energies docurs in forced oscillators tegerous left to the reader.

1.4.1 Group velocity

A wave packet can be seen as a superposition of harmonic components. In general, each component may travel with a different phase velocity. This is not the case in humageneous clastic media, since the phase velocity is frequency hulependent (see, for first-acc) the transversely isotropic case, equation (1.79)). Following the superposition principle, the wave packet propagates with the same velocity as each harmonic component. However, the relation between the group and the energy velocities, as well as the velocity of propagation of the pulse as a function of the propagation direction, merus excelible consideration

fat us consider two hormonic components fall and the given by

$$u = u_0[\cos(\omega_d - u_0 r) + \cos(\omega_d - u_0 r)]$$

$$(1.121)$$

and assume that the frequencies are slightly different

$$\omega_{\delta} = \omega_{c} + \delta_{c}\omega_{c} - \kappa_{b} + \delta_{\delta}$$

$$(1.122)$$

Equation (1.121) can then be written as

$$u = 2u_t \cos\left[\frac{1}{2}\left(c(z)t - ck(x)\right)\cos(\omega t - \kappa x)\right] - 1.123)$$

where

$$\kappa = \frac{1}{2}(\kappa_n + \epsilon_N), \quad (\omega = \frac{1}{2}(\omega_n + \omega_N))$$
(1.121)

The first term in equation (1.123) is the modulation envelope and the second term is the carrier wave, which has a phase velocity equal to ω/ω . The velocity of the modulation wave is equal to ω/ω , which by taking the built $\lambda \to 0$, gives the group velocity

$$v_n = \frac{\partial \varphi}{\partial v}, \qquad (1.125)$$

Generalizing this oppation to the 3-D case, we obtain the group-velocity victor

$$\mathbf{v}_{a} = \frac{\partial \omega}{\partial \kappa} \mathbf{v}_{1} + \frac{\partial \omega}{\partial \kappa_{1}} \mathbf{v}_{2} + \frac{\partial \omega}{\partial \kappa_{3}} \mathbf{v}_{3}, \quad \begin{pmatrix} v_{s0} & \partial \omega \\ -\partial \kappa_{1} & \partial \kappa_{1} \end{pmatrix}$$
(1.126)

(Eglohill, 1964; 1978, p. 312).

In general, the dispersion relation $\omega = \omega(n_i)$ is not available in explicit form. For instance, using equations (1.65) and (1.70), we note that equation (1.78), has the form

$$(r_{10}s^{2} + c_{3}s^{2}) = \rho\omega^{2}(r_{10}s^{2} + c_{3}s^{2} - \rho\omega^{2}) = (c_{11} + c_{32})(a_{1}^{2}s^{2}) = 0.$$
(1.127)

In general, we have from (1.71).

$$d\sigma_{1} \kappa [\Gamma - \rho_{+}(\Gamma_{1})] + F(\omega, e_{1}) = 0.$$
 (1.128)

Using implicit differentiation, we have for each component

$$\frac{\partial F}{\partial \omega}\delta\omega = \frac{\partial F}{\partial v_0}\delta s_0 = 0,$$
 (1.126)

which is obtained by keeping the other components constant. Thus, the final expression of the group velocity is

$$\mathbf{v}_{n} = \left(\frac{\partial F}{\partial \omega}\right)^{-1} \left(\frac{\partial F}{\partial w_{i}} \mathbf{e}_{i} + \frac{\partial F}{\partial w_{i}} \mathbf{e}_{i} + \frac{\partial F}{\partial w_{i}} \mathbf{e}_{i}\right)^{-1} \left(\frac{\partial F}{\partial \omega}\right)^{-1} \nabla_{R} F,$$

$$\left[\mathbf{v}_{q} = \left(\frac{\partial F}{\partial \omega}\right)^{-1} \left(\frac{\partial F}{\partial w_{i}} \mathbf{e}_{i}\right)\right], \qquad (1.130)$$

1.4.2 Equivalence between the group and energy velocities

In order to find the relation between the group and energy velocities, we use Cartosian octation. Rewriting the Kelvin Christoffel matrix (1.69) in terms of this notation, we get

$$\Gamma_{11} = c_{120} l_k l_l$$
 (1.131)

We have, from equation (1.68), after using (1.62) and (1.65).

$$p_{\pm}(a_0) = r_{i,\mu}a_{ij}r_{j}a_{0i}$$
 (1.132)

Differentiating this equation with respect to κ_0 , we obtain

$$2\rho\omega \frac{\partial\omega}{\partial r_{c}} a_{cc} \approx 2\epsilon_{cd}a_{cc} a_{cc} \qquad (1.133)$$

since $\partial(\mu_i \mu_i)/\partial \mu_i = \mu_i + u_i \delta_{ik} = 2\mu_i$. Taking the scalar product of equation (1.133) and u_{ik}^* and using the definition of group velocity (1.126), we obtain

$$v_{st} = \frac{\partial \omega}{\partial v_s} = \frac{v_{st} v s_t v_{st} u_s^*}{\rho \omega |\mathbf{u}_s|^2} \tag{1.134}$$

On the other hand, the Cartesian components of the complex power-flow vector (1,100) ran be expressed as

$$\mu_{i} = \frac{1}{2} \sigma_{i} \sigma_{i}^{*},$$
 (1.135)

Using the stress-strain relation (1.21) and of the most, we have

$$\sigma_{ab} f = 4 \omega_{ab} \phi_{ab} \mu_{b}^{*}$$
(1.136)

The strainalispharement relations (1.2) and (1.3) imply

$$a_{\mu}v_{\mu}^{*} = -\frac{1}{2}i_{\nu}v_{\mu\nu}(\partial_{\mu}u_{\nu} + \partial_{\nu}v_{\nu})u_{\nu}^{*} = -\frac{1}{2}e_{\mu\nu}(a_{\mu}u_{\nu} + a_{\nu}u_{\nu})u_{\nu}^{*}, \qquad (1.137)$$

when we have used the property $\partial_{t} a_{t} = -\alpha_{t} a_{t}$ (see equation (1.671). Using the symmetry properties (1.5) of $\alpha_{t} a_{t}$, we note that equation (1.147) becomes

$$\sigma_{\mu}v_{\nu}^{*} = -\omega v_{\mu\nu}\kappa_{0}u_{1}u_{\nu}^{*} = -\omega v_{\mu\nu}\kappa_{1}u_{\nu}u_{\nu}^{*}$$
(1.138)

The Cartesian components of the energy velocity (1.120) car be obtained by using equations (1.32), (1.109), (1.119), (1.135) and (1.138). Thus, we obtain

$$\frac{\omega_{1,1}\omega_{2}u_{0}u_{0}^{*}}{\rho\omega^{2}|\mathbf{n}_{0}|^{2}} = \frac{\omega_{1,2}\rho\omega_{0}u_{0}^{*}}{\rho\omega^{2}|\mathbf{n}_{0}|^{2}} = \frac{11.139}{\rho\omega^{2}|\mathbf{n}_{0}|^{2}} = \frac{11.139}{\rho\omega^{2}|\mathbf{n}_{0}|^{2}}$$

which which compared to explain a 11.135), shows that, in clastic media, the energy velocity is equal to the group velocity, namely,

$$v_{0} = v_{0}$$
 (1.140)

This fact simplifies the calculations since the group velocity is easily to compute than the energy velocity

1.4.3 Envelope velocity

The spatial part of the phase of the plane wave (1.62) can be written as $\kappa \cdot \mathbf{x} = n l_1 c + l_2 q + l_3 c$. An equivalent definition of wave surface in anisotropic clastic methans given by the manippe of the plane.

$$l[x + l_{2l} + h_{1}^{2}] = l_{1}x_{1} - l_{2}x_{2} - l_{2}x_{3}$$
 (1.111)

(flow), 1944, $p_{c}(260)$, because the velocity of the envelope of plane waves at unit propagation trace, which we call y_{co} , has the components

$$v_{in}|_{0} = v_{i} = \frac{\partial v_{p}}{\partial t}$$
(1.1.12)

and

$$c_n = \sqrt{r^2 + g^2 + \gamma^2}$$
(1.14)

To compute the components of the envelope velocity, we need the function $v_{\mu} = v_{\mu}(t_{\mu})$, which is available only in simple cases, such as those describing the symmetry planes (see equations (1.87)-(1.89)). However, note that $v_{\mu} = \omega/\mu$ and l_{μ} are related by the function T defined in equation (1.128), since using (1.13) and dividing by $s^{(1)}$ we obtain

$$T(\omega, \kappa_i) = \kappa^{-\alpha} \det(r_{ijkl}r_j \kappa_l - \mu \omega^2 \lambda_j) = \det(r_{ijkl}h_l^2 - \mu c_j^2 \delta_{ij})$$
(1.11)

and \perp and s_i , and v_i and l_i are related by the same function. Hence,

$$c_{ij1} = \frac{\partial v_i}{\partial t_i} - \frac{\partial z_i}{\partial r_i} + v_{gi} - c_i$$
 (1.145)

from (1.1.01), and in anisotropic elastic media, the covelope vehicity is equal to the group and energy velocities.

If we restrict out analysis to a viven plane, say the (x, β) -plane $(t_{y} = 0)$, we obtain mother well-known expression of the envelope velocity (Postana, 1955) Berryman, 1979). In this case, the wavevector directions can be defined by $t_{y} = \sin \theta$ and $t_{y} = \cos \theta$, where θ is the angle between the wavevector and the cases. Differentiating equation (1.131) with respect to θ (squaring it and adding the results to the square of equation (1.131), we get

$$v_{row} = \sqrt{v_p^2 + \left(\frac{dv_p}{d\theta}\right)^2}, \qquad (1.146)$$

Postma (1955) obtained this equation for a transversely isotropic medium. Although the group velocity is commonly called the envelope velocity in literature, we show in Chapter 1 that they are not the same in attentiating media. Bather, the envelope velocity is equal to the energy velocity in isotropic anelastic media. In anisotropic anelastic media: the three velocities are different.

1.4.4 Example: Transversely isotropic media

The phase velocity of SE waves is given in equation (1.79). The calculation of the group velocity makes use of the dispersion relation (1.78), in the form

$$F(n_1, n_3, \omega) = (q_0 n_1^2 + n_2 n_3^2 + q_2 \omega^2 + 0)$$
 (1.147)

From oppation (1/130), and using (1.70), we obtain

$$\mathbf{v}_{\mathbf{v}} \sim \frac{1}{\rho \omega} (c_0 \sigma_1 \mathbf{e}_1 + c_0 \sigma_1 \mathbf{e}_3) \sim \frac{1}{\rho c_0} (c_0 \sigma_1^T \mathbf{e}_1 + c_0 \delta_1 \mathbf{e}_1), \tag{1.118}$$

and

$$v_{\mu} = \frac{1}{m_{\mu}} \sqrt{c_{\mu}^{2} E + c_{\mu}^{2} P_{\mu}^{2}}$$
 (1.1.80)

It is rather easy to show that, using the dispersion relation (1.781), we obtain the same result from equations (1.112) and (1.146).

To compute the energy vehicity, we use equation (1.120). Thus, we need to calculate the complex Uniov Poynting vector (1.100), which for SH wave propagation or the (2.21) plate can be expressed as

$$\mathbf{p} = \frac{1}{2} [\sigma_{10} \mathbf{e}_{10} + \sigma_{10} \mathbf{e}_{10}]$$
 (1.150)

Frum equations (1.26) and (1.31), we note that

$$\sigma_{12} = \epsilon_{12} \partial_{1} \sigma_{11} = \sigma_{11} = c_{12} \partial_{1} g_{12} \qquad (1.151)$$

and using equations (1.62) and (1.67), we have

$$\sigma_{12} = i \kappa_1 \alpha_2 \sigma_2 + \sigma_{12} = -i \kappa_1 c_2 \sigma_2$$
 (1.152)

Since $(z) = -i \pm a_{z}^{2}$ and $a_{z}a_{z}^{2} = -\mathbf{u}_{0}^{-1}$, we use equation (1.152) to obtain

$$\mathbf{p} = \frac{1}{2}\omega(\phi_1 \mathbf{e}_1 + \phi_2 \phi_3 \mathbf{e}_1) u_i u_j = \frac{1}{2}\omega(\mathbf{e}_1 u_j)^2 (\phi_2 d_1 \mathbf{e}_1 + \phi_2 d_3 \mathbf{e}_3)$$
(1.153)

Substituting equation (1.115) into equation (1.120), and using expressions (1.109) and (1.153), we get

$$\mathbf{v}_{i} = \frac{1}{\rho^{n}_{i}}(c_{i}J_{i}\mathbf{e}_{i} + c_{i}J_{i}\mathbf{e}_{i}) = \mathbf{v}_{a}, \qquad (1.154)$$

Note that because $v_0 = c_0 l_{1/2} \rho_0^2$ and $v_0 = c_0 l_0 / \rho_0^2$, we have

$$\frac{M_{\rm eff}}{M_{\rm eff}} = \frac{M_{\rm eff}}{M_{\rm eff}} = 1.$$
 (1.155)

where we have used equation (1.79). Hence, the energy-volucity curve - and the wave front - is an ellipse, with semiaxes $c_{s,j}\mu$ and c_{M}/μ along the μ - and β directions, respectively. We have already demonstrated that the slowness surface for SH waves is an ellipsi (see equation (1.89)).

To obtain the energy velocity for the coupled qP and qS waves, we compute, for simplicity, the group velocity using equation (1.130) by rewriting the dispersion relation (1.78), as

$$F(\kappa_1, \kappa_1, \omega) \simeq (c_{11}\kappa_1^2 + c_{23}\kappa_1^2 - \mu_{-}^2)(c_{13}\kappa_1^2 + c_{23}\kappa_1^2 - \mu_{-}^2) = (c_{11} + c_{23}c_1^2\kappa_1^2\epsilon_2^2 \simeq 0, (1.150)$$

Hen, after some calculations.

$$v_{t} = \begin{pmatrix} i \\ v_{p} \end{pmatrix} \frac{(\Gamma_{ts} - \rho v_{p}^{2})v_{ts} + (\Gamma_{ts} - \rho v_{p}^{2})v_{ts} + v_{ts}v_{ts}^{2}}{\rho(\Gamma_{ts} + \Gamma_{ts} - 2\rho v_{p}^{2})}$$
(1.157)



Figure 1.1: Shows so an and group value by many (b) for a particular 1.67 GPa ($g_{22} = 13.1$ GPa, $g_{33} = 16$ GPa ($g_{33} = 110$ GPa ($g_{33} = 3603$ GPa (and $g_{23} = 3200$ kg/s⁻¹). The constant is transversely consepts, The convex septement we done of the respective showness and group value it statistically plane containers the symmetry over . The polarization showning are indicated in the convex the SII polarization showning period of the page z.

$$c_{ex} = \left(\frac{l_x}{r_p}\right) \frac{(\Gamma_{ex} - \rho c_p')c_{ex} + (\Gamma_{ex} - \rho c_p')c_{xy} \cdots (r_{ex} - c_{yy})^{dp'}}{\rho(\Gamma_{ex} + \Gamma_{ey})^2} \frac{2\rho c_p'}{2\rho c_p'}, \qquad (1.158)$$

where Γ_{\pm} and Γ_{38} are defined in equations (1.76) and (1.77). The phase and energy velocities of each mode coincide at the principal axes – the Cartesian axes in these examples.

Figure 1.1 shows the slowness (a) and group-volucity curves (b) for aparter (Pavtor, 1983, p. 3: Carcium, Kusloff and Kusloff, 1988a). Only uniquarter of the curves are displayed because of symmetry considerations. The cusps, fulls of lactmas, on the q8 wave are due to the presence of inflection points in the slowness surface. This plactometors implies three q8 waves around the cusps. One of the remarkable effects of anisotropy on accuste waves is the possible appearance of these folds (tripheations) in wave fronts. Frequency slices taken through anisotropic field data exhibit rings of interference patterns (Ohanian, Sayder and Carrione, 1997). The phenomenon by which a single adjustropic wave front interfrees with itself was reported by Maris (1983) in his study of the effect of finite photon wavelength on phonon focusing. The phenomenon by which shear waves have different velocities along a given direction is termed shear-wave splitting in seismic wave propagation

1.4.5 Elasticity constants from phase and group velocities

Elasticity constants can be obtained from five phase velocity measurements. For typical (ranscherer widths (\approx 10 mm), for which the measured signal in ultrasonic experiments is a plane wave, the travel mass correspond to the phase velocity (Dellinger and Vernik, 1992). Let us consider the (x, z)-plane of an orthorhorible medium. The corresponding phase velocities are given megatives (188) and (1.97). Moreover, using the despersion

relations (1.85) with $\theta = 45^{\circ} \theta_{1} = 6 = 1/\sqrt{2}$, we obtain

$$\begin{aligned} v_{1} &= \rho v_{p_{-}}^{2}(190)^{*} \\ v_{0} &= \rho v_{p_{-}}^{2}(07) \\ v_{0} &= \rho v_{p_{-}}^{2}(07) \\ v_{0} &= \rho v_{p_{-}}^{2}(07) \\ v_{0} &= \rho v_{p_{-}}^{2}(90) \\ v_{0} &= \rho v_{p_{-}}^{2}(90) \\ v_{0} &= v_{0} + \sqrt{\frac{1}{2}} \rho v_{p_{0}}^{2}(45) + 2\rho v_{p_{0}}^{2}(15) \\ v_{0} &= v_{0} + 2v_{0} + v_{0} \\ (1.150) \end{aligned}$$

When the signal is not a plane wave fort a localized wave packet - transducers less than 2 mm wide - the measured travel time is related to the energy velocity not to the phase velocity. In this case, c_{13} can be obtained as follows. If the receiver is located at 15°, equation (1.114) implies

$$v_{\rm c} \cos((\gamma \tau_{\rm c})) \cos(\theta = 15^{\circ}) = v_{\rm p}$$
 (1.160)

нт

$$v_p = \frac{6}{\sqrt{2}}(l + l_1),$$
 (1.061)

where y is the angle between the ray and propagation directions (see Figure 1.2).



Figure 1.2: Relation between the phase velocity static # and the group velocity sugles.

Now, noting that reprations (1.157) and (1.158) for transversely isotropic model are also the energy velocity components for our case, we perform the scalar product between (v_{i}, v_{i}) and $(l_{i} - l_{i})$ and use $v_{i} = v_{i1} = v_{i2} \cos 4V = v_{i1}\sqrt{2}$. Hence, we obtain

$$\begin{aligned} \rho e_{p} v_{i}(t) &= I_{0}(1) + |V_{1}| + |2\rho v_{p}|) + \sqrt{2} [(V_{1,k} - \rho e_{p}^{*})) e_{j}t' + |e_{j}t'_{1}| \\ &= e_{1} + |\rho v_{p}^{*}| (e_{k}t'_{1} - e_{j}t'_{1})], \end{aligned}$$
(1.162)

where

$$\Gamma_{1} = (\gamma_{1} P_{1}^{\prime} + (\gamma_{1} P_{1}^{\prime} - 1)_{A} + (\gamma_{2} P_{1}^{\prime} + (\gamma_{1} P_{1}^{\prime} - 1)_{A})$$

(1.163)

By substituting replation (1.161), we can solve (quation (1.162) for $\theta = \arcsin(l_1)$ more that $l' + l'_1 = 11$ as a function of the elasticity constants $e_1 + e_2$ and e_{N_2} and e_3 . Thus, the elasticity constant e_2 can be obtained from the dispersion relation (1.85) as

$$c_{11} = c_{22} + \frac{1}{i \beta_x} \sqrt{(\Gamma_x - \rho c_y^*)(\Gamma_{13} - \rho c_y^*)}$$
 (1.163)

1.4.6 Relationship between the slowness and wave surfaces

Die normal to the slowness surface $F(\kappa_i, \omega) = F(\kappa_i)$ – use $s_i = c_i/\omega$ in equation (1.128)

is $\nabla_{\mathbf{x}} F_{\mathbf{x}}$ where $\nabla_{\mathbf{x}} = (\partial_{\mathbf{x}} \partial_{\mathbf{x}} - \partial_{\mathbf{x}} \partial_{\mathbf{x}} - \partial_{\mathbf{x}} \partial$

On the other band, since the energy velocity, which defines the wave front, is equal to the envelope velocity in lossless media (see equation (1.147)), the wave surface can be defined by the human

$$W(x_0) = \kappa_0 x_0 = \omega_0$$
 (1.165)

in accordance with equation (1.111), and using (1.70) and $w_i = w_i^*$. The normal vector to the wave surface is grad 11. But grad W = $(w_i, w_i, w_i) = \kappa$. Therefore, the wavevector is normal to the wave surface, a somehow obvious fact, because the wave surface is the envelope of the plane waves . A geometrical illustration of these perpendicularity properties is shown in Figure 1.3.



Figure 1.3: Relationships between the slowness and the ray on waver surfaces, perpendicularity prepresences .

SH-wave propagation

We obtain the showness and wave surfaces from equations (1.80) and (1.155), namely

$$F(s_1, s_2) \approx \frac{s_1^2}{p_1 \phi_{\mu}} + \frac{s_1^2}{p_1^2 \phi_{\mu}} = 1.$$
(1.166)

and

$$W(r_{C}, r_{C}) \simeq \frac{r_{C}^{2}}{r_{C}/\sigma} + \frac{r_{C}^{2}}{r_{C}/\sigma} = 0.$$
 (1.167)

Taking the respective gradients, and using equation (1.154), we have

$$\nabla_{\mathbf{x}} T = \frac{2}{\rho} (c_0 s_1, c_0 s_1) = \frac{2}{\rho s_p} (c_0 J_1, c_0 J_1) = 2 \mathbf{v}_{c_0}$$
(1.168)

and

$$\nabla_{\mathbf{y}_{1}}W = 2\left(\frac{(s_{1}, s_{1})}{s_{0}}\right) + \frac{2}{s_{p}}(t_{1}, t_{2}) + \frac{2}{\omega}\mathbf{\kappa} - 2\mathbf{s}, \qquad (1.169)$$

which agree with the statements demonstrated earlier in this section.

1.5 Finely layered media

Most geological systems can be modeled as fine lowering, which refers to the case where the dominant wavelength of the pulse is much larger than the thicknesses of the individual layers. When this occurs, the medium is effectively transversely isotropic. The first to obtain a subtion for this problem was Burgground (1937). Later, other intestigators studied the problem using different approaches, e.g., Burnichenko (1930) and Postma (1955). Isothestrate the averaging process and obtain the equivalent transversely isotropic medium, we consider a two-constraint periodically layered reachem, as illustrated in Figure 1.4 and follow Postma's mastering (Postma, 1956). We assume that all the stress and strain components in planes parallel to the layering are the same in all layers. The other components may differ from layer to layer and are represented by average values.

+ 0



Figure 1.4. Representative volume of stratified mediate.

Assume that a stress σ_{13} is applied to the faces perpendicular to the p-axis, and that there are no tangential components, namely σ_{13} and σ_{23} . This stress does not generate sheat strains. On the faces perpendicular to the *x*-axis, we impose

$$\sigma_{11}^{(1)}$$
 or molecure f,
 $\sigma_{11}^{(2)}$ and medians 2. (1.170)
and $\sigma_{11}^{(2)} = \sigma_{11}^{(2)} = \sigma_{12}^{(2)}$.

Similarly, on the faces perpendicular to the gausis, we topize

$$\sigma_{ij}$$
 = on molecum 1,
 $\sigma_{ij}^{(i)}$ = on molecum 2, (1.171)
and $\phi_{ij}^{(i)} = \phi_{ij}^{(i)} = \phi_{ij}^{(i)}$.

We also must have

$$\sigma_{11} = \sigma_{11}^{(2)} = \sigma_{13},$$
 (1.172)

The proceeding equations guarantee the continenty of displacements and normal stress across the interfaces. This changes in the kness in media 1 and 2 are $e_{13}^{(2)}d_1$ and $e_{13}^{(2)}d_2$ respectively.

The stress-struct relations of each isotropic wedget can be obtained from equations (1.17) and (1.18). We obtain for medium 1.07 = 1) and medium 2.97 = 20.

$$\begin{split} a_{14}^{(3)} &= E_0 e_1 + \lambda_0 (e_{10} + e_{20}) \\ \sigma_{10}^{(3)} &= E_0 (e_0 + \lambda_0 (e_{10} + e_{10})) \\ \sigma_{13} &= E_0 (e_0 + \lambda_0 (e_1 + e_{10})), \end{split}$$
(1.173)

where $E_t = \lambda_t - 2\mu_t$. The average stresses on the facts perpendicular to the x_{-} and y axes are

$$\sigma := \frac{d_i \sigma_{ij} + d_j \sigma_{ij}}{d} = \sigma_{ij} + \frac{d_j \sigma_{ij} + d_j \sigma_{ij}}{d}, \qquad (1.171)$$

where $d = d_1 + d_2$. Eliminating the stresses $\sigma_1^{(i)}$ and $\sigma_2^{(i)}$ from equations (1.174) and (1.174), we obtain

$$\frac{d(\sigma_{12}) - (F_1d_1 + F_2d_2)e_{11} + (\lambda_1d_1 + \lambda_2d_2)e_{22} + \lambda_2d_2e_{23}^2 + \lambda_2d_2e_{24}^2}{d(\sigma_{12}) - (F_2d_1 + F_2d_2)e_{12} + (\lambda_2d_2)e_{12} + \lambda_2d_2e_{12}^2 + \lambda_2d_2e_{24}^2} = (1.175)$$

$$\frac{d(\sigma_{13}) - (\lambda_2d_2 + \lambda_2d_2)(e_{11} + e_{22}) + F_2d_2e_{23} + F_2d_2e_{23}^2}{d(\sigma_{13}) - (\lambda_2d_2 + \lambda_2d_2)(e_{12} + e_{22}) + F_2d_2e_{23}} = (1.175)$$

The strain along the paxis is the average given by

$$\frac{d}{dr_{11}} + d_{\theta} \frac{d}{d} \qquad (1.176)$$

then, we can compute the normal strains along the $z_{\rm stars}$ by using (1.173), and (1.176). Hence, we obtain

$$\frac{e_{ij}}{e_{ik}} = \frac{d(E_1e_{ik} - \lambda_i)(\lambda_i - \lambda_j)(e_{ij} - e_{ij})d_j}{d_1E_2 + d_2E_1} = \frac{d(E_1e_{ij} - (\lambda_i - \lambda_j))(- e_{ij})d_j}{d_1E_1 + d_2E_1}$$
(1777)

Substituting these results into equations (1.175), we obtain a stress-strain relation for an effective transversely isotropic medium, for which

$$\begin{aligned} a_{12} &= c_{12}c_{13} + c_{22}c_{23} + c_{12}c_{13}, \\ a_{22} &= c_{12}c_{13} + c_{22}c_{22} + c_{12}c_{13}, \\ \sigma_{33} &= c_{12}c_{13} + c_{22}c_{23} + c_{12}c_{13}, \end{aligned}$$
(1.178)

1.5 Finely layered media

where

$$\begin{aligned} c_{13} &= [d^* F_1 F_2 + 4d_1 d_2(\mu_1 - \mu_1)(\lambda_1 + \mu_1 - \lambda_2 - \mu_3)(\mathbf{D})]^3 \\ c_{13} &= [d^* \lambda_1 \lambda_2 + 2(\lambda_1 d_1 + \lambda_3 d_2)(\mu_2 d_1 + \mu_3 d_3)]\mathbf{D}]^3 \\ c_{13} &= d(\lambda_1 d_1 E_1 + \lambda_3 d_2 E_3)\mathbf{D}]^3 \\ c_{14} &\approx d^* E_1 E_2 \mathbf{D} \quad . \end{aligned}$$

$$(1.179)$$

and

$$\ell = d(d_1 E_1 + d_2 E_1)$$
 (1.180)

Next, we apply a stress $\sigma_{i,1}$ to the faces perpendicular to the (saxis). Contantity of cargeneral stresses implies $\sigma_{i,1} = \sigma_{i,1}^{(2)} = \sigma_{i,2}^{(2)}$, the resulting strain is shown in Figure 1.5.

I



Figure 1.76. Tooget hid stress and show

We have, in this case,

$$\phi_{N} = d \phi_{A}^{(1)} + d \phi_{A}^{(2)} (d^{(1)}, -\sigma_{B}) - \mu \sigma_{B}^{(1)}$$
(1.181)

Hence, eliminating c_{11}^{\dagger} , we obtain a relation between σ_{23} and c_{14} . Similarly, we find the relation between σ_{15} and c_{14} . Thus, we obtain

$$|v_{13} = v_{13}|_{14}, \quad |v_{13} = v_{13}|_{14}$$
(1.182)

with

$$c_{11} = \frac{d\mu_1 \mu_2}{d_1 \mu_2 + d_2 \mu_1}$$
, (1.183)

To obtain n_{s} , we apply a stress $\sigma_{ij}^{(l)}$ to the faces perpendicular to the q-axis and note that $r_{ij}^{(l)} = r_{ij}^{(l)} = r_{ij}^{(l)}$, because for thin layers the displacement inside a layer cannot differ greatly from the displacement at its binmularity. Then

$$\sigma_{ij}^{ij} = \mu_0 \omega_0$$
 (1.183)

and, some the average stress satisfies

$$d[\sigma_{12} - d[\sigma_{12}^2 + d_2\sigma_{21}^2] = e_0(d_1p + d_2p_1)$$
(1.185)

we have

$$(\sigma_{11} - \sigma_{21})_{12}$$
 (1.186)



Figure 1.6: Stowness so for the and group velocity soften to corresponding to the median longwavelength quarket to or opercosplass segment of layers with equal compositions (eq.). 2013 GPA (eq. (2)) GPA (q.) is S.GPA (q.) 13.1 GPA (q.) GPA (add p.) 1815 kg/mm. (Ed.) some quarter of the corresponded by layer boots of synchronic considerations. The polarization directions are indicated.

as the curves of a SE gold rotation is perpendicular to the plane of the page .

where

$$c_{m} = (d_{1}p_{1} + d_{2}p_{2})d^{-1}$$
(1.187)

Note the relation $\alpha_0 = 1_{12} = c_0/(2)$. The equivalent anisotropic media possess four enspitial transities at 43. from the principal coses

Figure 1.6 shows an example where the showness and group-velocity sections can be appreciated (Carcinne, Kosloff and Behle, 1991). The medium is an epoxy-glass sequence of layers with equal composition. We may infer from equations (1.183) and (1.187) that $\epsilon_{11} \leq \epsilon_{21}$ and Postma (1955) shows that $\epsilon_{11} \geq \epsilon_{21}/2$.

Backus (1962) ubtained the average elasticity constants in the case where the single layers are transversely isotropic with the symmetry axis perpendicular to the layering plane. Moreover, he assumed stationarity, that is, in a given length of composite medium much smaller than the wavelength, the proportion of each material is constant (periodicity is not required). The equations were further generalized by Schoenberg and Mur-(1989) for anisotropic single constituents. The transversely isotropic equivalent medium is described by the following constants:

$$\begin{aligned} v_{11} &= (e_{11} - e_{12}^{*} e_{13}^{*}) + (e_{11}^{*} e_{13}^{*} e_{13}^{*})^{2} \\ v_{11} &= (e_{12})^{-1} \\ e_{13} &= (e_{13})^{-1} (e_{13} e_{13}) \\ e_{13} &= (e_{13})^{-1} (e_{13} e_{13}) \\ e_{13} &= (e_{13})^{-1} \\ e_{13} &= (e_{13})^{-1} \\ e_{13} &= (e_{13})^{-1} \end{aligned}$$
(1.188)

where the weighted average of a quantity or is defined as

$$\langle n \rangle = \sum_{i=1}^{l} \rho n_{i},$$
 (1.189)

where p_i is the proportion of material k_i . More details about these media (for instance, constraints in the values of the different elasticity constants, are given by Helbig (1994), $p_i(315)$.

Fire dispersive effects are investigated by Norris (1992). Carriene, Koshiff and Behle (1991) evaluate the long wavelength approximation using manufeld modeling experiments. An acceptable rule of themb is that the wavelength must be larger than eight times the layer thickness. A complete theory, for all frequencies, is given in Barridge, de Hoop, Le and Norris (1993) and Shapiro and Habral (1999). This theory, which includes Backus averaging in the low-frequency hunt and not theory in the high-frequency built can be used to study velocity dispersion and frequency-dependent anisotropy for place waves purpositing at nov angle in a layered medium. The extension of the low-frequency theory to potoelastic media can be found in Nottis (1993). Bakulin and Mobikrov (1997) and Gelinsky and Shapiro (1997).

1.6 Anomalous polarizations

In this system?, we show that there are media with the same phase velocity or slowness surface that exhibit drasmally different polarization behaviors. Such media are kinemathally admitted but dynamically different. Therefore, classification of the media according to wave volucity along is not sufficient, and the identification of the wave type should be based on both velocity and polarization

"Anomalous Polarization," refers to the situation whole the slowness and wave surfaces of two elastic media are identical, but the polynzation fields are different. Examples of anomalous polarization have been discussed for transverse isotropy by Relbig and Scheenherg (1987), and for orthorhomble symmetry by Coreane and Relbig (2000). In this note we determine without prior restriction of the symmetry class index what conditions the phenomenon can occur. Since the three shownesses in a given direction are the square roots of the eigenvalues of the Kelvize-Christoffel matrix, while the polarizations are the corresponding eigenvectors, the condition for the existence of anomalous polarizations are the formulated as: Two module with different stiffness matrices are canomalous compannois," if the characteristic equations (1.72) of their respective Kelvine for isotoffel matrices \mathbf{F} and \mathbf{F}' are identical in c_{ij} if $\det(\mathbf{F} = \mathbf{M}_{ij}) = \det(\mathbf{F}' = \mathbf{M}_{ij})$ where $\lambda = \mu c_{ij}$.

1.6.1 Conditions for the existence of anomalous polarization

Without loss of generality we assume that the elastic fourth-rank stiffness tensors (and the corresponding 6×6 stiffness matrices) are referred to a natural coordinate system of the media. Inspection of the Kelvin-Christoffel dispersion relation

 $\det(\mathbf{\Gamma} = \lambda \mathbf{I}_1) = -\lambda^2 + (\mathbf{I}_0 \rightarrow \Gamma_1) + \mathbf{U}_{33}(\lambda^2 - (\mathbf{U}_2)\Gamma_3) - \Gamma_{24}^2 + \Gamma_1(\Gamma_3) - \Gamma_{13}^2 + \Gamma_1(\Gamma_2) - \mathbf{U}_{12}^2(\lambda)$

[&]quot;The sector has been or ten to still stration with Kiers Helbar

indicates that two stillness tensors have identical characteristic equations it for all propagation dues tions the following three conditions hold:

- 1. The diagonal terms of the Kelvin-Ouristoffel matrices P and P¹ are identical:
- The squares of their off-diagonal terms are identical; and
- 3. The products of their three off-diagonal terms are identical.

The second and third conditions can be satisfied simultaneously if all corresponding offchagonal terms have the same magnitude, and precisely two corresponding terms have opposite sign.

Let us consider the three conditions:

 The diagonal terms of the Kelvin-Christoffel matrices of an anamalous companion pair are equal for all propagation directions if they share the 15 stiffnesses occurring in equations (1.53), (1.73), and (1.73), i.e..

Ewo anomalius companion materies can thus differ only in-

The position of these elasticity constants in the stiffness matrix are

•		·		·	•	•	•	·
•			٢,	· ۲۰	· " · ·	•	-	
	c_{2}	-		100		100	-	
	C_{11}		$C_{\rm eff}$	_		_	6.65	-
'	c_{ij}	'						•
•		'	c ~	•		•	•	•
•		•	• /	۰	•	•	•	•
				с.,				

2. Two of the three off-diagonal terms of the Kelvin-Christoffel matrices for an anomalous comparion pair must be of equal magnitude but opposite sign for all propagation directions, thus for these terms all coefficients of the product of direction cosines must change sign. The off-diagonal terms of the Kelvin-Chrostoffel matrix are given by equations (1.73), (1.73), and (1.73). The nine stiffnesses $c_1, -c_3, -c_4, -c_5, -c_5, -c_5, -c_6, -c_7, -c_6, -c_7, -c_6, -c_7, -c_6, -c_6, -c_7, -c_6, -c_6,$

$$\begin{split} & \Gamma_{21} = (c_{23} + c_{33}) \ell_2 l_3 + c_{10} \ell_1 l_3 + c_{22} \ell_2 l_3 \\ & \Gamma_{11} = c_{23} l_2 \ell_3 + (c_{13} + c_{23}) \ell_1 l_3 + c_{23} \ell_1 \ell_2 \\ & \Gamma_{12} = (c_{23} \ell_2 \ell_3 + c_{13}) \ell_1 \ell_2 + (c_{23} + c_{13}) \ell_1 \ell_2 , \end{split}$$

$$(1.193)$$

and

$$\begin{aligned} \Gamma_{ij}^{*} &= (r_{ij}^{*} + r_{ij}) l_{j} l_{s} + r_{jj}^{*} l_{s} + r_{ij}^{*} l_{s} \\ \Gamma_{ij}^{*} &= r_{ij}^{*} l_{s} l_{s} + (r_{ij}^{*} + r_{ij}) l_{s} + r_{s}^{*} l_{s}^{*} l_{s} \\ \Gamma_{ij}^{*} &= r_{ij}^{*} l_{s} l_{s} + r_{ij}^{*} l_{s}^{*} + r_{ij}^{*} l_{s}^{*} + r_{ij}^{*} l_{s}^{*} \end{aligned}$$
(1.191)

1.6 Anomalous polarizations

with

$$\Gamma_{23}^{i} = (\Gamma_{23} - \Gamma_{23}^{i} - (\Gamma_{13} - \Gamma_{23}^{i} - (\Gamma_{13} - \Gamma_{23}^{i} - (\Gamma_{13} - \Gamma_{13}^{i} - (\Gamma_{13}$$

There are eight sign combinations of off-diagonal terms of the Kelver-Christoff-duraters, each corresponding to a characteristic equation (1.190) with identical coefficients for the terms with λ_m , m = 1, ..., 3. The condition $\Gamma_{0}\Gamma_{0}^{*}\Gamma_{0}^{*} = \Gamma_{0}\Gamma_{0}^{*}\Gamma_{0}^{*}\Gamma_{0}^{*}$ divides the corresponding eight slowness surfaces into two classes containing each four elements with the same product $\Gamma_{0}\Gamma_{0}^{*}\Gamma_{0}^{*}$. The two classes share the intersections with the coordinate planes, but differ outside these planes. The following table shows the sign conductions for the two sets of four slowness rache

This table shows that any two anomalous companion mecha differ in the algebraic signs of precisely two off-diagonal terms of the Kelvin-Christoffel matrix. Inspection of equations (1.193)(1.195) shows that this is possible only if either all three or precisely two of the three stafficies is $\{r_{i,i}, r_{i'}, r_{i'}\}$ vanishe if two of these stiffnesses would not vanish all three off-diagonal terms would be allected and would have to charge sign. The two slowless surfaces would share the metric tions with the coordinate plates, but would not be identical outside these planes. It follows that anomalous polarization is possible for any stiffness matrix that can be brought – through rotation of the coordinate system and for exchange of subscripts – into the following forms:

Medium with one (x, y)-symmetry plane;

$$\begin{pmatrix} c_{12} & c_{22} & c_{23} & 0 & 0 & 0 \\ c_{12} & c_{23} & c_{23} & 0 & 0 & 0 \\ c_{13} & c_{21} & c_{33} & 0 & 0 & c_{33} \\ 0 & 0 & 0 & c_{33} & 0 & 0 \\ 0 & 0 & 0 & 0 & c_{33} & 0 \\ 0 & 0 & 0 & 0 & c_{33} & 0 \\ 0 & 0 & c_{22} & 0 & 0 & c_{34} \end{pmatrix} ;$$

$$(1.197)$$

ii. Medium with an (7,) (symmetry plane

$$\begin{pmatrix} e_{11} & e_{22} & e_{13} & 0 & 0 & 0 \\ e_{12} & e_{23} & e_{23} & 0 & e_{23} & 0 \\ e_{13} & e_{14} & e_{23} & 0 & 0 & 0 \\ 0 & 0 & 0 & e_{13} & 0 & 0 \\ 0 & e_{23} & 0 & 0 & e_{23} & 0 \\ 0 & 0 & 0 & 0 & 0 & e_{23} \end{pmatrix},$$
(1.198)

Median with a cy, to-symmetry plane.

$$\begin{pmatrix} e_{11} & e_{23} & e_{23} & e_{34} & 0 & 0 \\ e_{12} & e_{23} & e_{23} & 0 & 0 & 0 \\ e_{13} & e_{24} & e_{35} & 0 & 0 & 0 \\ e_{13} & 0 & 0 & e_{33} & 0 & 0 \\ 0 & 0 & 0 & 0 & e_{33} & 0 \\ 0 & 0 & 0 & 0 & e_{33} & 0 \\ 0 & 0 & 0 & 0 & e_{33} & 0 \\ \end{pmatrix},$$

$$(1.199)$$

The media defined by these matrices have normal polarization in the symmetry plane and anomalous polarization in the other orthogonal planes.

1.6.2 Stability constraints

In the previous section the formal conditions for the existence of anomalous companion pairs were drived without regard to the stability of the corresponding rando. Only stable media can exist index the laws of physics. An elastic median is stable if and only if every deformation requires energy. This means that all principal minors of the stillness matrix must be positive for this terminology, a trained its the determinant of the corresponding sub-matrix: the main diagonal of the sub-matrix corresponding to a "principal runner" is a non-empty subset of the main diagonal of the matrix. This is equivalent with the requirement that the stiffness matrix must be positive definite. The condition for positive definitences can be relayed to "all leading principal minuts must be positive" (see equation (1.33)). The sub-matrix corresponding to a leading principal infinite is contiguents and contains the leading relation of the matrix.

Let us consider the matrix defined in (1.199). The first-order principal minors are positive if

$$c_{1,1} \ge 0, c_{21} \ge 0, c_{11} \ge 0, c_{21} \ge 0, c_{22} \ge 0, c_{23} \ge 0, c_{24} \ge 0$$
 (1.200)

The scenal-order principal minors are positive if

$$\frac{\sqrt{1+0}}{\sqrt{1+0}} = \frac{\sqrt{1+0}}{\sqrt{1+0}} = -\frac{\sqrt{1+0}}{\sqrt{1+0}} = \frac{\sqrt{1+0}}{\sqrt{1+0}} = \frac{\sqrt$$

The last modulity tenastraint of e_{13} is easily changed to the constraints on e_{23} and e_{23} . The leading principal third-order minor

$$D_{0} \leq c_{+} v_{0} c_{0} c_{0} + 2 c_{0} c_{0} \rho_{-\infty} - c_{0} c_{0}^{2} c_{0} - v_{0} c_{0}^{2} c_{0} - v_{1} c_{0}^{2}$$
(1.202)

is positive if equal or and equivatisfy

$$1 + 2 \frac{c_{23}c_{32}c_{33}c_{32}}{c_{33}c_{33}c_{33}} + \frac{c_{23}}{c_{33}c_{33}} + 0.$$
(1.203)

The leading principal fourth order minor is obtained by development about the fourth coheant:

$$D_{t} = c_{0}D_{t} - c_{0}^{2}(c_{0}c_{0} - c_{0}),$$
 (1.204)

If inequalities (1.200), (1.201) and (1.203) are setisfied, D_3 is positive of and only if

$$r_{A}^{2} + \frac{c_{A}D_{1}}{c_{B}c_{A}} + \frac{c_{A}D_{1}}{\sqrt{c_{B}c_{A}} + C_{1}} \leq r_{A} + \frac{c_{A}D_{2}}{\sqrt{c_{B}c_{A}} + C_{2}}$$
(1.205)

with obvious generalizations to the constraints on c_{23} and c_{23}

$$C_{\alpha\beta} = \frac{c_{\beta}D_{\beta}}{c_{i}|v_{\alpha} - v_{\beta}|} \rightarrow - \frac{c_{\beta}D_{\beta}}{\sqrt{c_{j}|v_{\alpha} - v_{\beta}|}} + \frac{c_{\beta}D_{\beta}}{\sqrt{c_{j}|v_{\alpha} - v_{\beta}|}} = - \frac{(1.205)}{(1.205)}$$

and

$$c_{\mu}^{2} = \frac{c_{\mu}D_{\mu}}{c_{\mu}c_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}c_{\mu}c_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}c_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}c_{\mu}c_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}c_{\mu}c_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}c_{\mu}c_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}c_{\mu}c_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}c_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}D_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}C_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{\mu}D_{\mu}} + \frac{c_{\mu}D_{\mu}}{\int c_{$$

1.6.3 Anomalous polarization in orthorhombic media

It follows from equations (1/1935-1/195) that for orthochembre modia the off-diagonal terms of the Kelvin-Christoffel matrices of a pair of companion matrices are

$$\begin{split} \Gamma_{0n} &= (c_{11} + c_{11}) l_{2n} \\ \Gamma_{0n} &= (c_{11} + c_{21}) l_{2n} \\ \Gamma_{0n} &= (c_{22} + c_{21}) l_{2n} \end{split} \tag{1.208}$$

ward

$$\begin{aligned}
 \Gamma_{12} &= (c_{11}^2 + c_{11}) l_1 l_1 \\
 \Gamma_{11} &= (c_{11}^2 + c_{21}) l_1 l_1 \\
 \Gamma_{12} &= (c_{11}^2 + c_{22}) l_1 l_2,
 \end{aligned}$$
(1.209)

with

 $\Gamma_{1} = \pm \Gamma_{3} + \Gamma_{4} = \pm \Gamma_{4} + \Gamma_{5}^{2} + \pm \Gamma_{45}$ (1.20)

where in the last line precisely two of the minus signs must be taken. We obtain the clasticity constants of the anomalous companions as:

i. Modinic with an (1), gl-symmetry plane

$$\frac{d^2}{d^2} = \frac{d^2}{d^2} \left[\frac{d^2}{d^2} + \frac{d^2}{d^2} +$$

ii. Mediana with an task to symmetry plane

$$\frac{c_{ij}^2 - c_{ij}}{c_{ij}^2 + c_{ij}} = \frac{(c_{ij} - c_{ij}) + c_{ij}^2}{(c_{ij} + c_{ij}) - (c_{ij} + c_{ij}) + c_{ij}^2} = \frac{(c_{ij} - 2c_{ij})}{(c_{ij} + 2c_{ij})}$$
(1.212)

iii. Medium with a (g.) symmetry plane

$$\frac{c_{ij}^{2} - c_{ij}}{c_{ij}^{2} - c_{ij}} = -(c_{ij} - c_{ij}) \rightarrow c_{ij}^{2} = -(c_{ij} - 2c_{ij})$$

$$\frac{c_{ij}^{2} - c_{ij}}{c_{ij}^{2} - c_{ij}} \approx -(c_{ij} - c_{ij}) \rightarrow c_{ij}^{2} \approx -(c_{ij} - 2c_{ij}).$$
(1.213)

where the polarization is normal in the symmetry planes

Only companion pairs where $\{c_{ik}, c_{ik}, c_{ik}\}$ and $\{c'_{ik}, c'_{ik}, c'_{ik}\}$ satisfy the stability conditions are meaningful.

1.6.4 Anomalous polarization in monoclinic media

It follows from equations (1 1931)(1,195) and (1 197) that for monoclinic media with the (x,y)-plane as symmetry plans, the off-magonal terms of the Kelvin-Christoffel matrices of a pair of comparison matrices are

$$\begin{split} \Gamma_{A1} &= (1, 1, \dots, n_1, 0) f_{A} = (n_1 d_2 f_A + \dots + n_N d_N f_A + \dots + n_N d_N f_A + \dots + 1, 214) \\ \Gamma_{A1} &= (n_1 f_A + \dots + n_N) f_{A} f_A + \dots + (n_N) f_{A} + \dots +$$

and

$$\begin{split} \Gamma_{i,n}^{*} &= -\Gamma_{i,1} = i c_{i,n}^{*} + c_{i,n} (l_i l_i - c_{i,n} l_i l_i) \\ \Gamma_{i,n}^{*} &= -\Gamma_{i,1} = -c_{i,n} (l_i l_i - c_{i,n}^{*} - c_{i,n} (l_i l_i)) \\ \Gamma_{i,n}^{*} &= \Gamma_{i,n}^{*} :: (c_{i,n} - c_{i,n}) (l_i l_i) \end{split}$$

$$(1.255)$$

This is easily satisfied if the orthorhomble "root" medium that is obtained by setting ϵ_{θ}

It has an aroundous compariant. Then, $(x_0^* + x_0) = (x_0^* + x_0)$ and $(x_0^* + x_0) = (x_0^* + x_0)$ and because the leading third-order minor $D_0 \ge 0$, the interval (1.207) for the addition of (x,y) is not empty.

Therefore, the anomalius companions of monoclinic andia are

i. Medium with an (x, y)-symmetry planet

$$\begin{aligned} c_{ijk}^{*} &= c_{ijk} \\ c_{ijk}^{*} &= -(c_{ijk} + 2c_{ijk}) \\ c_{ijk}^{*} &= +(\epsilon_{ijk} + 2c_{ijk}); \end{aligned} \tag{1.216}$$

in Medium with an accel-symmetry planet

$$c_{25}^{\prime} = -\epsilon_{25}^{\prime}$$

 $c_{12}^{\prime} = +(\epsilon_{12} + 2\alpha_{15})$ (1.217)
 $c_{12}^{\prime} = -(\epsilon_{12} + 2\alpha_{15})$

iii. Medium with a (9,)1-symmetry plane:

$$C_{11} = C_{12}$$

 $C_{12} = 1c_{12} + 2c_{13}$ (1.218)
 $C_{23} = 1c_{13} + 2c_{13}$

1.6.5 The polarization

The components of the polarization vector, $u_{0,i}$ corresponding to propagation direction l_i and one of the three eigenvalues λ_i , stand in the same ratio as the encospicaling collations of $(\Gamma - M_{i})_i$ in the development of det $(\Gamma - M_{i})$ for an arbitrary j_i i.e.

 $\begin{aligned} & \eta \in \mathcal{H}_{1} \quad & \eta \in \mathcal{H}_{1} \\ & (1, \gamma) \in \mathcal{M}(\Gamma_{12} - \mathcal{A}) = [\Gamma_{11}^{-1} \cap [\Gamma_{12} - \Gamma_{12}(\Gamma_{12} - \mathcal{A})] \cap [\Gamma_{21} \cap \Gamma_{22} - \Gamma_{21}(\Gamma_{22} - \mathcal{A})], \quad & (1, \gamma) \in \mathcal{H}_{1} \\ & \Gamma_{21}^{-1} \cap [\gamma - \Gamma_{22} \cap \Gamma_{22} - \mathcal{A}) \cap [\Gamma_{22}^{-1} - \mathcal{A}) \cap [\Gamma_{22}^{-1} \cap \Gamma_{22}^{-1} \cap \Gamma_{22}^{-$

It follows from the expressions for $j \ge 3$ that

$$\left[(\Gamma_{3} - \lambda)(\Gamma_{22} \otimes \lambda) \otimes \Gamma_{23}^{2}\right] = \frac{\left[(-)\Gamma_{13} - \Gamma_{23}(\Gamma_{1} - \lambda)\right]\left[\Gamma()\Gamma_{33} - \Gamma()(\Gamma_{22} - \lambda)\right]}{\Gamma_{23}\Gamma_{33} - \Gamma()(\Gamma_{32} - \lambda)}, \quad (1.226)$$

After substitution of this equation into (1.219) and division of all three terms by $|\mathbf{1}_{1,2}|_{1,2} = 1_{1,2} (|\mathbf{1}_{1,2}| - \lambda)[|\mathbf{1}_{1,2}|_{1,2} - |\mathbf{1}_{1,2}|_{1,2} - |\mathbf{1}_{2,2}|_{2,2})$, one obtains the symmetric expression

$$\begin{split} & u_{1} - u_{2} + u_{3} = \\ & -1 & -1 & -1 & -1 \\ & \Gamma_{12} \Gamma_{13} - \Gamma_{23} \Gamma_{13} - \Lambda_{11} \Gamma_{12} \Gamma_{23} - \Gamma_{13} (\Gamma_{23} - \lambda_{11})^{-1} \left[\Gamma_{23} \Gamma_{13} - \Gamma_{12} (\Gamma_{13} - \lambda_{11})^{-1} \right] \end{split}$$

For a pair of companion media with $\Gamma_{11}^{\prime} = \Gamma_{23}$ and $\Gamma_{13}^{\prime} = \Gamma_{13}$ one has

i.e..

$$[n_{1}^{2} + n_{1}^{2} + n_{2}^{2}] = n_{1}^{2} + n_{2}^{2} + n_{3}^{2}$$
(1.223)

 u_{i} since $u_{i}^{i} = u_{i}^{i}$.

$$\mathbf{u}' = \begin{pmatrix} -1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{pmatrix} \mathbf{u}.$$
 (11.223)

1.6.6 Example

We consider an orthorhor bic medium. The four polarization distributions corresponding to the "normal" showness surface — with the sign conducations (i) 1961 — are shown in Figure 1.7.—This figure shows the intersections of the showness surface with the three planes of symmetry, and the polarization vector for the fastest (innermost) sheet wherever it makes an angle greater than $\pi/4$ with the propagation vector. The "zones" of atomatons polarization are clearly visible in Figure 1.7.

In the following example, we assume a simultaneous charge of sign of c_1, \dots, c_n and c_1, \dots, c_n . The stiffness matrix of the orthorhombic medium with pointal polarization is

$$\mathbf{C} = \begin{pmatrix} c_{11} & c_{12} & c_{13} & 0 & 0 & 0 \\ c_{12} & c_{23} & c_{23} & 0 & 0 & 0 \\ c_{12} & c_{23} & c_{23} & 0 & 0 & 0 \\ 0 & 0 & 0 & c_{23} & 0 & 0 \\ 0 & 0 & 0 & 0 & c_{23} & 0 \\ 0 & 0 & 0 & 0 & 0 & c_{23} \end{pmatrix} = \begin{pmatrix} 10 & 2 & 1.5 & 0 & 0 & 0 \\ 2 & 9 & 1 & 0 & 0 & 0 \\ 1.5 & 1 & 8 & 0 & 0 & 0 \\ 0 & 0 & 0 & 3 & 0 & 0 \\ 0 & 0 & 0 & 0 & 3 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 2 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 & 1 \end{pmatrix}$$
(1.225)

enormalized by $p \in MPa$, where p is the density in kg/m³. Then, according to equation (1.213), c' v = -1 GPa and c' v = -5.5 GPa

Figure 1.8 shows the group velocities and corresponding snapshots of the wave field in the three symmetry planes of the normal and anomalous media. The polarization is indicated on the curves, when it is not plotted, the particle motion is perpendicular to the respective plane (cross-plane shear waves). Only one ortant is shown due to symmetry considerations.

As each be seen, a sign change in $\phi \to \phi_0$ and $\phi_1 \to \phi_0$ only affects the (x, y)- and (x, 1)-phases, leaving the polarizations in the (y, z)-phase fundation. The anomaly is more pronounced about 45° where the polarization of the lastest wave is quasi-transverse and the cusp hid is essentially longitudinal. Moreover, the cross-phase shear wave with polarization perpendicular to the respective symmetry plane can be clearly seen in Figures 1.8c and d. More details about this example and anomalous polarization can be fixed in firmd in Carciane and Holbig (2001).





Figure 1.8. Ray velocity sections and statishes. If the distribution to vector at the symmetry planes of an order thereine magnetic lugares do and the correction date the article in equival. Or be one obtain affine model space is displayed due to symmetry is reached one.

1.7 The best isotropic approximation

We address in this section the problem of finding the last isotropic approximation of the axisotropic stress-strain relation and quantifying anisotropy with a single numerical index. Federov (1968) and Backus (1970) obtained the bulk and shear wordali of the last isotropic medium using component notation. Here, we follow the approach of Cavallin (1999), who used a shorter and coordinate-free derivation of equivalent results. The reader may refer to Gartin (1981) for background material on the correspondency mathematical methods.

Let Σ be any real finite-dimensional vector space, with a scalar product $\mathbf{a} \cdot \mathbf{b}$ for \mathbf{n} , \mathbf{b} in \mathbb{X} . The tensor (dyadic) product $\mathbf{a} \cdot \mathbf{b}$ is the linear operator such that

$$(\mathbf{a} \otimes \mathbf{b}) \mathbf{x} = (\mathbf{b} \cdot \mathbf{x}) \mathbf{a} \qquad (1.2.9)$$

for all \mathbf{x} in X_n . The space L(X) of linear operators over \mathbf{X} inherits from \mathbf{X} a scalar product, which is defined by

$$\mathbf{a} \cdot \mathbf{b} = (\mathbf{r}(\mathbf{a} - \mathbf{r})\mathbf{b}) = -\mathbf{f}(\mathbf{r}|\mathbf{a}, |\mathbf{b}|) \cdot L(\mathbf{X}),$$
 (1.227)

where to denotes the trace (the sum of all eigenvalues, each count of with its multiplicity), and symbol β denotes the composition of maps. We denote by \mathbb{R}^n the obdimensional Euclidean space. Moreover, Lin is the space of linear operators over \mathbb{R}^n , sym is the subspace of him forward by symmetric operators. So is the subspace of Sym formed by \mathbb{R}^n the operators proportional to the identity operator \mathbf{I}_n . D is the subspace of Sym formed by all the operators with zero trace. The operators \mathbf{S} (spin-rical) and \mathbf{D} (deviatoric) defined by

and

$$\mathbf{D} \sim \mathbf{L} = \mathbf{S} \approx \frac{1}{3} \begin{pmatrix} -2 & -1 & -1 & 0 & 0 & 0 \\ -1 & 2 & -1 & 0 & 0 & 0 \\ -1 & -1 & 2 & 0 & 0 & 0 \\ 0 & 0 & 0 & 3 & 0 & 0 \\ 0 & 0 & 0 & 0 & 3 & 0 \\ 0 & 0 & 0 & 0 & 0 & 3 \end{pmatrix}$$
 (1.229)

are orthogonal projections from Sym into S and D, respectively.

We consider the stress-strain relation (1.31), written in tensorial notation instead of the Vorger correct notation. Using tensor notation, also termed "Kelvin's notation", the stress-strain relation reads

$$\begin{pmatrix} -\sigma_{11} \\ -\sigma_{22} \\ -\sigma_{23} \\ -\sigma_{23} \\ -\sqrt{2}\sigma_{11} \\ \sqrt{2}\sigma_{23} \\ \sqrt{2}\sigma_{23} \\ \sqrt{2}\sigma_{23} \end{pmatrix} = \begin{pmatrix} -c_{11} & -c_{13} & -\sqrt{2}c_{11} & \sqrt{2}c_{22} & \sqrt{2}c_{23} \\ -c_{12} & -c_{23} & -\sqrt{2}c_{11} & \sqrt{2}c_{23} & \sqrt{2}c_{23} \\ -\sqrt{2}c_{23} & -\sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} \\ \sqrt{2}\sigma_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & -2c_{23} & -2c_{23} \\ -\sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & \sqrt{2}c_{23} & \sqrt{2}c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} & -2c_{23} \\ \sqrt{2}c_{23} & -2c_{23} & -2$$

(Meltrabad: and Cowm, 1990; Helbig, 1994, p. 406). The three arrays in equation (1/230) are true tensions in 6-D space. Using the same symbols for simplicity equation (1.230) is similar to (1.34) $[\sigma = \mathbf{C} \cdot \mathbf{e})$ where $|\mathbf{C} \cdot \mathbf{S} \cdot \mathbf{y}_{1}| \rightarrow \mathbf{S} \mathbf{y}_{2}$ is a linear operator.

Accordingly, isotropy is a special case of waisotropy, and the isotropic stiffness operator has the form

$$C_{aa} = -3k |S| + 2p |D|$$
 (1.231)

where K and μ are the bulk and shear moduli, respectively.

For norms of **S** and **D** are 1 and $\sqrt{5}$, respectively, where the norm has the usual definition $|\mathbf{x}| \sim \sqrt{\mathbf{x}} \cdot \mathbf{x}|$. Therefore, **S** and $(1/\sqrt{5})$ **D** constitute an orthonormal pair, and the projector only the space of isotropic elasticity rensors is³.

$$P_{ps} = \mathbf{S} \otimes \mathbf{S} + \frac{1}{2} \mathbf{D} \otimes \mathbf{D}$$
 (1.232)

Thus, given an anisotropic stiffness tensor C, its best isotropic approximation is

$$P_{\alpha\beta}\mathbf{C} = (\mathbf{S} \cdot \mathbf{C})\mathbf{S} + \frac{1}{5}(\mathbf{D} \cdot \mathbf{C})\mathbf{D},$$
 (1.233)

where we have used equation (1.226). Now, comparing (1.23) and (1.233), the dilatational term is $d\mathbf{X} = \mathbf{S} - \mathbf{C} = \mathrm{tr}(\mathbf{S} = \mathbf{C}) = \mathrm{tr}(\mathbf{C} - \mathbf{I}_1)/3$, according to equations (1.227) and (1.228). The shear term is obtavated in the same way by using equation (1.229). Hence, the corresponding bulk and shear moduli are

$$\mathbf{\lambda} + \frac{1}{9} \operatorname{tr}(\mathbf{C} \cdot \mathbf{I}_{1})$$
 and $\mathbf{\mu} + \frac{1}{40} \operatorname{tr}(\mathbf{C} - \frac{1}{30}) \operatorname{tr}(\mathbf{C} \cdot \mathbf{I}_{1})$. (1.234)

In Vingt's notation, we have

$$tr(\mathbf{C} \cdot \mathbf{I}_{3}) = \sum_{I,I=1}^{N} c_{II} \quad \text{and} \quad tr(\mathbf{C} = \sum_{I=1}^{n} c_{II} + 2\sum_{I=0}^{n} c_{II}, \qquad (1.235)$$

to obtain

$$\mathcal{K} = \frac{1}{9} \left[c_{12} + c_{13} + c_{13} + 2(c_{13} + c_{13} + c_{24}) \right]$$
(1.236)

anal

$$\mu = \frac{1}{15} (c_0 + c_0 + c_0 + 3) (c_0 + c_0 + c_0) = (c_0 + c_0 + c_0), \qquad (1.237)$$

Note that the bulk modulus (1.236) can be obtained by assuming an isotropic strain state, i.e. $c_{11} = c_{22} = c_{13} = c_{13}$ and $c_{24} = c_{32} = c_{32} = 0$. Then, the mean stress $\sigma = (\sigma_{11} - \sigma_{22} - \sigma_{13})/3$ can be expressed as $\sigma = 3W_{13}$.

The eigenvalues of the isotropic stiffness operator (1.231) are 4X and 2μ , with corresponding eigenspaces S (of dimension 1) and D to followersion 5), respectively (see Chapter 5, Section 3.1.2). Then, from the symmetry of the stiffness operator we mendately get the introgonality between the corresponding projectors S and **D**, as mentioned before

The order to get a generational protocol of the projector C_{ini} to a join that S and D represent two orthon stand unit vectors along the Cartes are every standly, respectively. To project a general vector **x** onto the *x* axis, we perform the scalar protocol S extand obtain the proceeded vector as $|\mathbf{S}| = \mathbf{x}(\mathbf{S})$, which is equal to $|\mathbf{S}| \leq \mathbf{S} \leq \mathbf{S} \cdot \mathbf{x}$ care every structure (1.22b)

In order to quantify with a single number the level of unisotropy present in a natorial, we introduce for anasotropy unlex

$$F_{V} = \frac{\mathbf{C} - \frac{P_{m} \mathbf{C}}{|\mathbf{C}|}}{|\mathbf{C}|} = \sqrt{\frac{P_{m} \mathbf{C}^{-1}}{|\mathbf{C}|^{2}}}, \quad (1.248)$$

where the second identity follows from the *n*-dimensional Pythagoras' theorem. We obviously have $0 \le T_0 \le 1$, with $T_1 = 0$ corresponding to isotropic materials.

The quantity $||B_{\mu\nu} \mathbf{C}|^2$ that appears in equation (1.235) is easily computed using equation (1.231) and the orthonormality of the pair $\{\mathbf{S}_{\mu}(\mathbf{t})\sqrt{5}(\mathbf{D})\}$:

$$T_{1...}^{*} C^{*,i} = 2K^{i} + 20 \mu^{i}$$

(1.239)

To compute $||\mathbf{C}||^2$, we need to resort to component notation: for example, in Voiges notation we have

$$[\mathbf{C}_{ij}^{nj} = \sum_{I=0}^{3} c_{II}^{ij} + 4 \sum_{I=0,j}^{n} c_{II}^{ij} + 1 \sum_{l=0}^{3} \sum_{I=0}^{n} c_{II}^{ij} = (1.240)$$

As an example, let us consider the orthorhombic elastic matrix (1.225) and its corresponding anomalously-polarized medium whose matrix is obtained by using the relations (1.213). Both media have the same slowness surfaces, but their ansotropy indices are 0.28 (normal polarization) and 0.57 (morealous polarization). Thus, polarization along has a significant influence on the degree of unsotropy. Regarding geological media, Cavallisi (1999) computed the axis propionization of 0.323. The median is 0.17,

A general axisotropic stress-strait, relation can also be approximated by symmetries lower than isotropy, such as transversely (sotropic and orthorhombic media). This has been done by Arts (1993) using Federov's approach

1.8 Analytical solutions for transversely isotropic media

2-D and 3-D analytical solutions are available for the Green function — the response to $\delta(t\phi)(\mathbf{x})$ — in the symmetry axis of a transversely isotropic medium. This section shows how these exact solutions can be obtained. The complete Green's tensor for ellipsoidal slowness surfaces has been obtained by Burridge, Charlwick and Nords (1993).

1.8.1 2-D Green's function

Payton (1983, p, 38) provides a classification of the wavesfront curves on the basis of the location of the cusps. We consider here class IV materials, for which there are four cusps, two of these centered on the symmetry axis. Let us consider the (x, z)-plane and define

$$\alpha = c_{12} \delta v_{22} + \beta = c_{11} \delta v_{22} + \beta = 1 + \alpha \beta + (c_{13} v_{23} + 1)^2,$$
 (1.241)

the dimensionless variable

$$\zeta = \begin{pmatrix} 1 \\ 0 \end{pmatrix}^{T} \begin{pmatrix} n \\ e_{TT} \end{pmatrix}$$
, (1.242)

and

$$t_F = -\sqrt{\frac{p}{r_B}}, \quad t_S = -\sqrt{\frac{p}{r_B}}, \quad (1.233)$$

The following Green's function is given in Payton (1983, p. 78) and is valid for materials satisfying the conditions



Fighte 1.9. Two dimensional Green's function net and three dimensional response to the wavafe's function. To as a function of time. The source is a codificational point force. The mechanics operate and the control overlap distance is 5 cm.

Due to a force directed in the reduction, the Green function is

$$u_1(z, t) = 0.$$
 (1.245)

$$u_{0}(x_{t}t) = \begin{cases} 0, & 0 \le t \le t_{D}, \\ G_{1}(\zeta_{t}) = t_{0} \le t \le t_{0}, \\ 0, & t_{0} \le t \le t_{0}, \\ G_{1}(\zeta_{t}) = t \le t_{1}, \\ G_{1}(\zeta_{t}) = t \le t_{1}, \end{cases}$$
(1.246)

with

$$G_{1}(\zeta) = \frac{\sqrt{\zeta}}{4\pi i} \left[1 - \frac{2(1+\zeta)}{\sqrt{\zeta}} - \frac{(\beta-1)\zeta}{\sqrt{\zeta}} \right] \frac{\sqrt{(\beta-1)\zeta} - \sqrt{\zeta}}{\sqrt{(-2)\alpha - \zeta'(1-\zeta)}} , \qquad (1.2)7)$$

(1.244)

Chapter 1: Anisotropic elastic media

$$G_{Q,\zeta}(-\frac{\sqrt{s}}{2\pi}) \left[\sqrt{\ell} - \frac{1}{\sqrt{n}} \sum_{i=1}^{n} \left\{ s_{i+1}(\ell-1)s_{i} - 2\sqrt{3(n-\zeta)(1-\zeta)} \right\}^{-1} \right]$$
(1.2)8)

where

$$r = \frac{t_s}{a}$$
, (1.249)

$$q = \sqrt{\gamma (\beta + 1)} - 2\beta \alpha + 0 + 2\sqrt{\beta (1 + \alpha \beta - \gamma) (\alpha + \beta - \gamma))} (\beta - 1), \qquad (1.250)$$

and

$$\zeta = [\gamma - (\beta + 1)\zeta] = 1.0\sigma - \zeta (1 - \zeta)$$
 (1.251)

Figure 1.24 shows the Green function for a parity at 8 cm from the source location (see Figure 1.1 for an illustration of the slowness and group-velocity sections). The singularities are located at times t_{t} and t_{t} and the burning due to the cusps of an besseen between times t_{s} and t_{t} . The bast singularity is not present in an isotropic machine because $t_{t} \to \infty$. Fin more details are Carcinus, Kusloff and Kosloff (1988a).

1.8.2 3-D Green's function

In the 3-D case, the response to Heaviside's function H(t) is available (Payton, 1983, p. 1985). (Condition (1.235) must be satisfied in the following solution.) Let us consider a force along the collicection, that is

$$\mathbf{f} = (0, 0, 1)\delta(x)\delta(y)\delta(z)H(t),$$
 (1.252)

The solution is

$$u_1(z,t) = u_2(z,t) = 0,$$
 (1.253)

$$(q(z,t)) = \frac{1}{4\pi \pi} \begin{pmatrix} p \\ e_{N_{1}} \end{pmatrix} \begin{cases} 0, & 0 \le t \le t_{N_{1}} \\ b(\zeta), & tp \le t \le t_{N_{1}} \\ 2b(\zeta), & ts \le t \le t_{N_{1}} \\ 1, & t \ge t_{N_{1}} \end{cases}$$
(1.251)

with

$$b(\zeta) = \frac{1}{2} - \frac{2(1-\zeta)}{2\sqrt{\zeta}} + \frac{((\ell+1)\zeta}{2\sqrt{\zeta}},$$
 (1.255)

where the involved quantities have been introduced in the previous section. The Green function is the time activative of (1.254).

Figure 1.9b shows the response to Hownsde's function for operate at 8 cm from the source lucation. A seisengram can be obtained by time consolution of (1.254) with the time derivative of the corresponding source wavelet. For more details see Carcione, Kosloff, Belle and Seriaxi (1992).

1.9 Reflection and transmission of plane waves

An analysis of the reflection-transmission problem in anisotropic clastic media can be found in Musgrave (1960). Henneke II (1971). Daley and then (1977). Kerth and Crampu (1977). Rokhlin, Bolland and Adher (1986). Geochner (1992). Schoenberg and Protozio

12

(1992). Chapman (1991). Psencik and Vavryou's (1998) and Ursin and Hauger (1996). In the anisotropic case, we study the problem in terms of energy flow rather their amplitude since the energy-flow direction, in general, thes not caloride with the propagation evolvector edimetical. Critical angles or on when the my (energy flow) direction is paallel to the interface.

In this section, we formally introduce the problem for the general 3-D case and discuss in detail the reflection-transmission problem of cross-plane waves in the symmetry plane of a monochine medium (Schoenberg and Costa, 1991, Carnone, 1997a). This problem, rousidered in the context of a single wave made, illustrates most of the phenomena related to the presence of anisotropy.

Let us consider a plane wave of the firm (1.1.2), incident from the approximation on a plane boundary between two anisotropic media. The incident wave generates three reflected waves and three transmitted waves. For a webbei contact, the boundary conditions are continuity of displacement for particle velocity) and stresses on the interface:

$$\mathbf{u}^{t} \sim \mathbf{u}_{op}^{R} \sim \mathbf{u}_{op}^{R} \sim \mathbf{u}_{op}^{R} = \mathbf{u}_{op}^{T} \sim \mathbf{u}_{op}^{T} \sim \mathbf{u}_{op}^{T} = \mathbf{u}_{op}^{T}$$
(1.256)

$$(\boldsymbol{\sigma}^{T} + \boldsymbol{\sigma}^{R}_{q^{k}} + \boldsymbol{\sigma}^{R}_{q^{k}} + \boldsymbol{\sigma}^{R}_{q^{k}}) \cdot \mathbf{n} = (\boldsymbol{\sigma}^{T}_{q^{k}} + \boldsymbol{\sigma}^{T}_{q^{k}} + \boldsymbol{\sigma}^{T}_{q^{k}}) \cdot \mathbf{n}, \qquad (1.257)$$

where I, R and T denote the incident, reflected and transmitted waves, and \mathbf{n} is a unit vector normal to the interface. These are six boundary conditions, constituting a system of six algebraic equations in terms of the six unknown amplitudes of the reflected and transmitted waves.

Surff's law implies the following:

- All slowness vectors should be in the plane formed by the slowness vector of the incident wave and the mirriral to the interface.
- The projections of the slowness vectors on the interface coincide.

Since the slowness vectors lie in the same plane, it is convenient to choose this plane as one of the Cattesian planes, say, the pretty-plane. Once the elasticity constants are transformed into this system, the slowness vectors have two components. Let the inreflace be in the reduction. The generalized the slowness vectors is zero, and the generation pts are equal to that of the meident wave,

$$s_{ij}^{\mu} = s_{ijk}^{\mu} = s_{ijk}^{\mu} = s_{ajk}^{\mu} = s_{ajk}^{\mu} = s_{ijk}^{\mu} = s_{ijk}^{\mu} = s_{ijk}^{\mu} = (1.258)$$

The unknown s_s components are found by solving the dispersion relation (1.72), which ran be rewritten as

$$de(0) = \rho e_0 + 0, \quad (1.259)$$

since $s_{a} = h_{c}/r_{ac}$. This sixuader repution is solved for the upper and lower media. Of the 12 solutions = 6 for the lower media and 6 for the upper media – we should solvet there physical solutions for each half-space. The efficience medium for the effected waves and into the transmission medium for the transmitted waves. At erroral angles and for evanescent waves, the energy-flow vector is parallel to the interface. Calculation of the energy-flow vector requires the calculations of the eigenvectors for each reflected and transmitted wave special treatment is required along the symmetry axes. A root of the dispersion relation, can be real or complex. In the latter case, the chosen sign of its imaginary part must be such that it has an exponential decay away from the interface. However, this criterion is not always valid in lossy media (see Chapter 6). It is then encoedent to check the solutions by computing the energy balance normal to the interface.

Our analysis is simplified when the incidence plane coincides with a plane of symmetry of both nacha, because the maident wave does not generate all the reflected-transmitted modes at the interface. For example, at a symmetry plane, there always exists a crossplane shear wave (Hellog, 1994, p = 111). An incident cross-plane shear wave generates a reflected and a transmitted wave of the same nature. Incidence of qP and qS waves generates qP and qS waves. Most of the examples found in the literature correspond to these cases. Figure 140 shows an example of analysis for normally polarized media. A qP wave reaches the interface, generating two reflected waves and two transmitted waves. The projection of the wavevector onto the interface is the same for all the waves, and the group-velocity vector is perpendicular to the slowness curve.



Figure 1.10. Type plessformly is using the slowness surfaces of the relevance transmission problem between two anisotropic module 1 and 10. The module or transversity isotropic with the symmetry axes along the directions perpendendar to the interface. The curves are derived or we are the qU and qS slowness settings, respectively. Tail acrows correspond to the wavevector and or procorrows to the group-velocity vector.

Strange effects are caused by the deviation of the energy flow vector from the wavevector direction. In the phenomenon of external conteal refraction, the Uniov-Doyating vector may be contrad to the showness surface at an infinite set of points. If the sympletty axis of the incidence mediate is normal to the incertace and the transmission median is isotropic. Shell's law implies the existence of a divergent curoilar cone of transmitted rays (Musgrave, 1970, p. 111).

1.9.1 Cross-plane shear waves

Equations [1] 46) describe cross-plane shear mutton in the plane of symmetry of a monoclinic medium. Let us introduce the plane wave

$$z = z + \mu y_0 \exp (i\omega t - s/x - s_1 t),$$
 (1.260)

where ρ_i is a constant complex displacement and $s_i = \epsilon_{i,i,k}$ are the slowness components. Substitution of this plane wave into equations (1, 0) gives the slowness relation

$$T[s_{\pm}, s_{4}) = c_{0s}s[+2c_{0}s_{\pm}s_{3}+c_{3}s] = p = 0.$$
 (1.261)

which increal (s₁, s₂) space, is an ellipse due to the positive definite conditions

$$c_{11} > 0, \quad c_{00} > 0, \quad e^{2} \equiv e_{11}c_{00} + e^{2}_{01} > 0.$$
 (1.262)

which can be defined from equations (1.33).



Figure 1.11: Characteristics of the strongest statice corresponding to an SII wave in the plane of symmetry of a monordinic resolution

Eighte 1.11 allocated the characteristics of the slowness curve. The group or energy velocity can be calculated by using equation (1.130).

$$(1, -1)(r_{0,2}, -r_{0,2}, d)(p_{0,1}, r_{0,1}, 4r_{0,2}, -r_{0,2,3})/n$$
 (1.363)

Solving for s_{i} and s_{i} in terms of r_{i} , and r_{i} , and substituting the result into equation (1.261), we obtain the energy-velocity surface

$$|\psi_0(t) \rangle = 2c_{00}\psi_0(t_0 + c_0)t_0^2 - c_{10}^2\rho = 0,$$
 (1.364)

which is also an ellipse. In other to distinguish between down and up propagating waves the slowness relation (1.261) is solved for s_0 , given the horizontal slowness s_0 . It yields

$$s_{0} = -\frac{1}{c_{0}} \left(-c_{0} s_{0} + \sqrt{\rho} \epsilon_{0} - c' s' \right).$$
 (1.265)

In principle, the β sign corresponds to downward or β , propagating waves, while the sign corresponds to upward or β , propagating waves

Substituting the plane wave (1/260) into equations (1/465) and (1/465, we get

$$\sigma_{12} = (r_0, s_0 + c_0, s_1), \text{ and } \sigma_{21} = (r_0, s_1 + r_0, s_1)$$
 (1.268)

The Uniov-Poventing vector (1,100) is given by

$$\mathbf{p} = -\frac{1}{2} \left[\sigma_{ij} \mathbf{e}_{ij} - \sigma_{ji} \mathbf{e}_{ij} \right]^{2}, \qquad (1.267)$$

Substituting the plane wave (1.260) and the stress-strain relations (1.266) into regulation (1.267), we obtain

$$\mathbf{p} \approx \frac{1}{2} \omega^2 |u_0|^2 (X \mathbf{e}) - Z \mathbf{e}_0), \qquad (1.268)$$

where

$$V = c_0 s_0 + c_0 s_0$$
, and $Z = c_0 s_0 + c_0 s_0$. (1.269)

Using equation (12955) we have

$$Z = 2 \sqrt{\rho_{AB}} - e^{i} s_{B}^{2}$$
 (1.270)

The particle velocity of the incident wave can be written as

$$c^{1} = i\omega \exp(i\omega t - s_{0}c - s_{0}^{2}z)$$
, (1.271)

where

$$s_i = \sin(\theta^i \beta_{i_i}) \theta^i \beta_i = s_i^4 = \cos(\theta^i \beta_{i_i}) \theta^i \beta_i$$
(1.272)

where θ^{t} is the inclusion propagation angle (see Figure 1.11), and

$$c_p(\theta) \geq \sqrt{(c_0)\cos^2\theta + c_p(\sin^2\theta + c_b(\sin 2\theta))/\rho}$$
(1.273)

is the phase velocity.

Snell's law, i.e., the continuity of the horizontal slowness.

$$s_1^{d'} = s_1^{d'} = s_{12}^{d'}$$
 (1.274)

is a necessary condition to satisfy the boundary conditions.

Denoting the reflection and transmission coefficients by R_{sc} and I_{ssc} the particle velocities of the reflected and transmitted waves are given by

$$r^{R} = i_{\theta} \mathcal{H}_{\theta} \exp[i_{\theta}(t - s_{1}r - s_{2}^{R}))] \qquad (1.275)$$

and

$$v^{T} = i\omega T_{ss} \exp[i\omega T - s/r - s_{t}^{T}]^{2}_{t}$$
 (1.276)

respectively.

Then, continuity of r and σ_D at z = 0 gives

$$T_{cs} = 1 + R_{S}$$
 (1.277)

and

$$Z^{I} = R_{ss}Z^{I} = T_{ss}Z^{I}$$
 (1.278)

which have the following solution:

$$R_{s} = \frac{Z^{T} - Z^{T}}{Z^{T} - Z^{T}} = \frac{Z^{T} - Z^{T}}{Z^{T} - Z^{T}}$$
 (1.373)

where Z is defined in opportune (6.10). Some both the maident and tellected waves satisfy the slowness relation (1.261), the vertical shavness s_s^R can be obtained by subtracting $F(s_1, s_s^R)$ from $F(s_1, s_1^R)$ and assuming $s_1^R \neq s_1^R$. This yields

$$s_1^E = \left(s_1^E + \frac{2r_{10}}{r_{11}}s_1\right),$$
 (1.280)

Then, using opportune (1.278) we obtain

$$Z^{g} = Z^{I}$$
 (1.281)

and the reflection and transmission coefficients (1.959) because

$$R_{ii} = \frac{Z^{I} - Z^{I}}{Z^{I} - Z^{I}} = \frac{2Z^{I}}{Z^{I} - Z^{I}}$$
(1.282)

Denoting the material properties of the lower modium by promp quantities, we see that the slowness relation (1.994) of the transmission modium gives s_s^4 in terms of s_s :

$$d_{1}^{I} = \frac{1}{c_{11}^{I}} \left(-c_{12}^{I} s_{1} + \sqrt{d} c_{11}^{I} - c_{2}^{I} s_{1}^{I} \right)$$
(1.383)

with

$$e^{i\theta} = e_{10}^{\dagger} e_{00}^{\dagger} = e_{00}^{\dagger}$$
 (1.284)

Alternatively, from equation (1.270).

$$\vec{s}_{1}^{j} = \frac{1}{\vec{c}_{11}} \left(Z^{j} - \vec{c}_{p, N_{1}} \right)$$
 (1.285)

Let us consider an isotropic medium above a monoclinic medium, with the wavevector of the currelence wave lying in the [2, 1)-plane, which is assumed to be the monoclinemedium available. Then, or incident mass-plane shear wave generates only reflected and transmitted cross-plane shear waves. This case is discussed by Schoenberg and Costa (1991).

The isotropic medium has elasticity constants $r_{ch} = 0$ and $r_{1,c} = c - \mu$. The vertical slowness is ______

$$s_1^2 = \sqrt{n/p - s_1^2}$$
, (1.286)

and

$$Z^{*} = \mu_{2} \left\{ -11.287 \right\}$$

From equation (1/270), we have

$$Z^{T} = \pm \sqrt{g} \delta^{2} s_{0}^{\dagger} - \delta^{2} s_{0}^{\dagger}, \qquad (1.288)$$



Figure 1.12: The reflection transmission problem for an SII wave meldone on an interface between an is dropic medium and contended in accumum coordination that she shows as volters when the transmitted wave has a dewrywerd-printer, showness vector and dy shows the showness vectors when the transmitted wave has an upward-printer, showness vector and dy shows the showness vectors when the transmitted wave has an upward-printer, showness vector. The numerical demographic burry vector on pressures to shall to the showness statice and points downeeds.

I wo different situations are shown in Figure 1.12a-b. The slowness sections are shown, agether with the respective wavevectors (full arrows) and energy-velocity vectors (empty arrows). In Figure 1.12a, the transmitted slowness vector has a positive value of s₁ and points downwards. In Figure 1.12b, it has a regative value of s₁ and points downwards. In Figure 1.12b, it has a regative value of s₁ and points downess. The energy velocity vector points downwards and the solution is a valid transmitted wave. The transmitted slowness vector must be a point of the lower section isolid fine to the slowness surface since there the energy-velocity vector points downwards.

Example: Let us consider the following properties:

aad

$$|v_{12}'| \leq 15 \text{ GPa}, ||v_{22}'| = 7 \text{ GPa}, ||v_{22}'| \geq 22 \text{ GPa}, ||p'| = 2700 \text{ kg/m}^3$$

The absolute values of the reflection and transmission coefficients versus the incidence angle are shown in Figure 1.13.

According to equation (1.268), the condition $Z^1 = 0$ yields the critical angle $\theta_{\rm eb}$. From equation (1.288), we obtain

$$\theta_{ci} = \arcsin\left(\frac{1}{c^2}\frac{d^2}{\sqrt{p}}c_{ci}^2\mu\right), \qquad (1.289)$$

In this case $b_{ij} = 10^{\circ}$. Beyond the critical angle, the time-averaged power-flow vector of the transmitted wave is parallel to the interface because $\operatorname{Ret}Z^{1} = -0$ (see equation



Figure 1/13: Misolate values of the cellecton and transforssing coefficients versus the encodence propagation include for an SR wave incident strain interface separative in isotropic mediate and a monochrid mediate.

(1.268)) and the wave becauses evanescent. This problem is discussed in more distail in Chapter 6, when dissipation is considered.

Note that any lower medium with constants e_{10} , e_{20} and density p satisfying

$$\rho c_{11} = \rho c_{12}^{\dagger} = c_{1}^{\dagger} - c_{1}^{\dagger}^{\dagger}$$
, (1.290)

will have the same R_{S} and E_{S} for all s. If we choose the material projectics of the isotropic medium to satisfy

$$p_D = p_{C,D}^{(1)} = p_{C,D}^{(2)}$$
 (1.291)

then $Z^{T} = Z^{T}$, $R_{ss} = 0$ and $T_{ss} = 1$ for all s. In such a case, there is no reflected wave and, thus the interface would be impossible to detect using a reflection method based on cross-plate shear waves.
This page intentionally left blank

Chapter 2

Viscoelasticity and wave propagation

The quantity ET, by which the rate of displacement must be multiplied to get the force, may be called the coefficient of viscosity. It is the product of a coefficient of elasticity, E, and a time T, which may be called the "time of relaxation" of the elastic force. In the case of a collection of moving molecules such as we suppose a gas to be, there is also a resistance to change of form, constituting what may be called the linear elasticity, or "rigidity" of the gas, but this resistance gives way and diminishes at a rate depending on the amount of the force and on the nature of the gas.

James Clerk Maxwell (Maxwell, 1867)

The basic formulation of linear (infinitesimal) viscoelasticity has been developed by several scientists, including Maxwell (1867), Voigt (1892), Lord Kelvin (William Thomson (Kelvin, 1875)), Boltzmann (1874), Volterra (1909, 1940) and Graffi (1928). Boltzmann (1874), in particular, introduced the concept of memory, in the sense that at a fixed point of the medium, the stress at any time depends upon the strain at all preceding times. Viscoelastic behavior is a time-dependent, mechanical non-instantaneous response of a material body to variations of applied stress. Unlike a lossless elastic medium, a viscoelastic solid once set into vibration would not continue to vibrate indefinitely. Because the response is not instantaneous, there is a time-dependent function that characterizes the behavior of the material. The function embodies the stress or strain history of the viscoelastic body. The strength of the dependence is greater for events in the most recent past and diminishes as they become more remote in time: it is said that the material has memory. In a linear viscoelastic material, the stress is linearly related to the strain history until a given time. The strain arising from any increment of the stress will add to the strain resulting from stresses previously created in the body. This is expressed in mathematical form by Boltzmann's superposition principle or Boltzmann's law.

Notation: Let f and g be scalar time-dependent functions. The time convolution of f and g is defined by

$$f * g = \int_{-\infty}^{\infty} f(\tau)g(t - \tau)d\tau. \qquad (2.1)$$

Hooke's law can be expressed in 3-D space or 6-D space depending on whether the stress and the strain are tensors or column matrices. In the shortened matrix notation, the definition of convolution may be extended easily to include 6×1 column matrices (a) and 6×6 tensors (or matrices) (A):

$$f * \mathbf{a} = \int_{-\infty}^{\infty} f(\tau) \mathbf{a}(t - \tau) d\tau, \qquad (2.2)$$

$$\mathbf{A} * \mathbf{a} = \int_{-\infty}^{\infty} \mathbf{A}(\tau) \cdot \mathbf{a}(t - \tau) d\tau. \qquad (2.3)$$

As a convention, any function f(t) is said to be of the Heaviside type if the past history of f up to time t = 0 vanishes. That is,

$$f(t) = \tilde{f}(t)H(t),$$
 (2.4)

where H(t) is Heaviside's or step function, and there is no restriction on \tilde{f} . If f and gare of the Heaviside type, we can write

$$f * g = \int_{0}^{t} f(\tau)g(t - \tau)d\tau,$$
 (2.5)

If f is of the Heaviside type, we define the Boltzmann operation as

$$f \odot g = f(0)g + (\partial_t f H) * g,$$
 (2.6)

corresponding to the time derivative of the convolution between f and g, that is $f * (\partial_t g)$.

2.1 Energy densities and stress-strain relations

In order to obtain the stress-strain relation for anisotropic elastic media, we defined the strain-energy function (1.1) and used equation (1.21) (or equation (1.22)) to calculate the stress components in terms of the strain components. In materials with dissipation, a unique free-energy density function (the strain energy here) cannot be defined (e.g., Morro and Vianello, 1990). There are cases where the strain energy is unique, such as that of viscoelastic materials with internal variables based on exponential relaxation functions (Fabrizio and Morro, 1992, p. 61). The uniqueness holds when the number of internal variables is less than the number of physical (observable) variables (Graffi and Fabrizio, 1982).

We assume that the properties of the medium do not vary with time (non-aging material), and, as in the lossless case, the energy density is quadratic in the strain field. We introduce the constitutive equation as a convolutional relation between stress and strain, with the assumption of isothermal conditions. However, as stated above, it is important to note that the form of the strain-energy density is not unique (see Rabotnov, 1980, p. 72). Analogy with mechanical models provides a quite general description of anelastic phenomena. The building blocks are the spring and the dashpot. In these elements, it is assumed that energy is "stored" in the springs and "dissipated" in the dashpots. An arbitrary – series and parallel – connection of these elements provides a good phenomenological model to describe the behavior of many materials, from polymers to rocks. Christensen (1982, p. 86), Hunter (1983, p. 542), and Golden and Graham (1988, p. 12) define appropriate forms of the strain energy in the linear viscoelastic case (see also Carcione, 1999a).

2.1 Energy densities and stress-strain relations

A form of the strain-energy density, which can be made consistent with the mechanical model description, is

$$V(t) = \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} G_{ijkl}(t - \tau_1, t - \tau_2) \partial_{\tau_1} \epsilon_{ij}(\tau_1) \partial_{\tau_2} \epsilon_{kl}(\tau_2) d\tau_1 d\tau_2$$
(2.7)

(Christensen, 1982, p. 79; Golden and Graham, 1988, p. 12). As we shall see below, the general expression of the strain-energy density is not uniquely determined by the relaxation function.

Differentiation of V yields

$$\partial_t V = \partial_t \epsilon_{ij} \int_{-\infty}^t G_{ijkl}(t - \tau_2, 0) \partial_{\tau_2} \epsilon_{kl}(\tau_2) d\tau_2$$

+ $\frac{1}{2} \int_{-\infty}^t \int_{-\infty}^t \partial_t G_{ijkl}(t - \tau_1, t - \tau_2) \partial_{\tau_1} \epsilon_{ij}(\tau_1) \partial_{\tau_2} \epsilon_{kl}(\tau_2) d\tau_1 d\tau_2.$ (2.8)

We define the stress-strain relation

$$\sigma_{ij} = \psi_{ijkl} * \partial_t \epsilon_{kl},$$
 (2.9)

where ψ_{ijkl} are the components of the relaxation tensor, such that

$$\psi_{ijkl}(t) = G_{ijkl}(t, 0)H(t),$$
 (2.10)

where H(t) is Heaviside's function. Then,

$$\int_{-\infty}^{t} G_{ijkl}(t-\tau_2,0)\partial_{\tau_2}\epsilon_{kl}(\tau_2)d\tau_2 = \sigma_{ij}$$
(2.11)

and (2.8) becomes

$$\sigma_{ij}\partial_t \epsilon_{ij} = \partial_t V + D,$$
 (2.12)

where

$$\dot{D}(t) = -\frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} \partial_t G_{ijkl}(t - \tau_1, t - \tau_2) \partial_{\tau_1} \epsilon_{ij}(\tau_1) \partial_{\tau_2} \epsilon_{kl}(\tau_2) d\tau_1 d\tau_2$$
(2.13)

is the rate of dissipated-energy density. Note that the relation (2.10) does not determine the stored energy, i.e., this cannot be obtained from the stress-strain relation. However, if we assume that

$$G_{ijkl}(t, \tau_1) = \bar{\psi}_{ijkl}(t + \tau_1),$$
 (2.14)

such that

$$\psi_{ijkl}(t) = \bar{\psi}_{ijkl}(t)H(t),$$
 (2.15)

this choice will suffice to determine G_{ijkl} , and

$$V(t) = \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} \bar{\psi}_{ijkl} (2t - \tau_1 - \tau_2) \partial_{\tau_1} \epsilon_{ij}(\tau_1) \partial_{\tau_2} \epsilon_{kl}(\tau_2) d\tau_1 d\tau_2, \qquad (2.16)$$

$$\dot{D}(t) = -\int_{-\infty}^{t} \int_{-\infty}^{t} \partial \tilde{\psi}_{ijkl} (2t - \tau_1 - \tau_2) \partial_{\tau_1} \epsilon_{ij}(\tau_1) \partial_{\tau_2} \epsilon_{kl}(\tau_2) d\tau_1 d\tau_2, \qquad (2.17)$$

where ∂ denotes differentiation with respect to the argument of the corresponding function. Equation (2.14) is consistent with the corresponding theory implied by mechanical models (Christensen, 1982, p. 120; Hunter, 1983, p. 542), i.e., these expressions describe the energy stored in the springs and the energy dissipated in the dashpots (Cavallini and Carcione, 1994).

The strain-energy density must be positive; therefore $V \ge 0$. Substituting the strain function $\epsilon_{ij}(t) = \check{\epsilon}_{ij}H(t)$ into equation (2.7), we obtain the condition $G_{ijkl}(t, t)\check{\epsilon}_{ij}\check{\epsilon}_{kl} \ge 0$, which from (2.10) and (2.14) implies

$$\psi_{ijkl}(t)\tilde{\epsilon}_{ij}\tilde{\epsilon}_{kl} \ge 0.$$
 (2.18)

Similarly, since $\dot{D}(t) \ge 0$, the same test implies

$$\partial_t \psi_{ijkl}(t) \check{\epsilon}_{ij} \check{\epsilon}_{kl} \leq 0.$$
 (2.19)

The definitions of stored-(free-)energy and energy-dissipation rate are controversial, both in electromagnetism (Oughstun and Sherman, 1994, p. 31) and viscoelasticity (Caviglia and Morro, 1992, p. 53-57). The problem is particularly intriguing in the time domain, since different definitions may give the same time-average value for harmonic fields. Although the forms (2.16) and (2.17) may lead to ambiguous partitions of the rate of work (equation (2.12) is one of these possibilities), this ambiguity is not present when the stress-strain relation can be described in terms of springs and dashpots (Hunter, 1983, p. 542; Cavallini and Carcione, 1994).

2.1.1 Fading memory and symmetries of the relaxation tensor

On the basis of observations and experiments, we may postulate the fading memory hypothesis, which states that the value of the stress depends more strongly upon the recent history than upon the remote history of the strain (Christensen, 1982, p. 9). It is then sufficient that the magnitude of each component of the relaxation tensor be a decreasing function of time,

$$|\partial_t \psi_{ijkl}|_{t=t_1} \le |\partial_t \psi_{ijkl}|_{t=t_2}, \quad t_1 > t_2 > 0.$$
 (2.20)

As in the lossless case, the symmetry of the stress and strain tensors gives

$$\psi_{ijkl} = \psi_{jikl} = \psi_{ijlk}$$
, (2.21)

implying 36 independent components. In the shortened matrix notation, the stress-strain relation (2.9) has the form

$$\sigma = \Psi * \partial_t e, \quad (\sigma_I = \psi_{IJ} \partial_t e_J),$$
 (2.22)

where σ and e are defined in equations (1.20) and (1.27), respectively. In general, under the assumption that the stress-strain relation is given by Boltzmann's law, and without a precise definition of a strain-energy function, it can be shown that Ψ is a symmetric matrix in the low- and high-frequency limits only, that is

$$\psi_{ijkl}(t = \infty) = \psi_{klij}(t = \infty), \quad \psi_{ijkl}(t = 0) = \psi_{klij}(t = 0),$$
 (2.23)

(Christensen, 1982, p. 86; Fabrizio and Morro, 1992, p. 46). The number of components of the relaxation matrix can be reduced to 21 if we consider that the matrix is symmetric, i.e.,

$$\psi_{II}(t) = \psi_{II}(t), \quad (\psi_{ijkl}(t) = \psi_{klij}(t)).$$
 (2.24)

There is no rigorous demonstration of this property¹, and equation (2.24) is generally assumed to be valid (e.g., Golden and Graham, 1988, p. 37).

2.2 Stress-strain relation for 1-D viscoelastic media

The complex modulus is the key quantity in the following analysis. We determine its properties – closely related to those of the relaxation function – and its significance in terms of stored and dissipated energies. To introduce the basic concepts, it is simplest to start in one dimension.

2.2.1 Complex modulus and storage and loss moduli

Hooke's law in the lossless case is

$$\sigma = M_e \epsilon_i$$
 (2.25)

where M_e is the elastic modulus. ($M_e = \lambda$ is the Lamé constant if we assume $\mu = 0$). According to equation (2.9), the relaxation function in this case is

$$\psi(t) = M_e H(t),$$
 (2.26)

because

$$\tau = \psi * \partial_t \epsilon = \partial_t \psi * \epsilon = M_e \delta(t) * \epsilon = M_e \epsilon.$$
 (2.27)

In the lossy case,

$$\sigma = \psi * \partial_t \epsilon$$
, (2.28)

where

$$\psi = \bar{\psi}H(t)$$
. (2.29)

The Fourier transform of equation (2.28) gives

$$\mathcal{F}[\sigma(\omega)] = M(\omega)\mathcal{F}[\epsilon(\omega)] \quad (\bar{\sigma} = M\bar{\epsilon}),$$
 (2.30)

where F is the Fourier-transform operator, and

$$M(\omega) = \mathcal{F}[\partial_t \psi(\omega)] = \int_{-\infty}^{\infty} \partial_t \psi(t) \exp(-i\omega t) dt$$
(2.31)

¹The symmetry can be proved if one can show that the Hermitian (H) and antihermitian (A) parts of the relaxation matrix are even and odd functions, respectively. Any complex matrix can be written as $\psi_{IJ}(\omega) = \psi_{IJ}^{H}(\omega) + \psi_{IJ}^{A}(\omega)$, where $\psi_{IJ}^{H} = \frac{1}{2}[\psi_{IJ}(\omega) + \psi_{JI}^{*}(\omega)]$ and $\psi_{IJ}^{A} = \frac{1}{2}[\psi_{IJ}(\omega) - \psi_{JI}^{*}(\omega)]$. Moreover, since $\psi_{IJ}(\omega)$ is the Fourier transform of a real quantity, it must satisfy the reality condition $\psi_{IJ}^{*}(\omega) =$ $\psi_{IJ}^{*}(-\omega)$. The first statement implies $\psi_{IJ}^{H}(-\omega) = \psi_{IJ}^{H}(\omega)$ and $\psi_{IJ}^{A}(-\omega) = -\psi_{IJ}^{A}(\omega)$. Combining these relations into one by using the reality condition implies $\psi_{IJ}(\omega) = \psi_{IJ}(\omega)$. Melrose and McPhedran (1991, p. 83) justify the first statement for the dielectric-permittivity tensor by invoking the time-reversal invariance of the equation of motion, under certain transformations of the field variables (Onsager's relations).

is the complex modulus. Since $\partial_t \psi = \delta(t) \tilde{\psi} + \partial_t \tilde{\psi} H(t)$,

$$M(\omega) = \psi(0^+) + \int_0^\infty \partial_t \tilde{\psi}(t) \exp(-i\omega t) dt, \qquad (2.32)$$

because $\tilde{\psi}(0) = \psi(0^+)$. Equation (2.32) becomes

$$M(\omega) = \psi(\infty) + i\omega \int_0^\infty [\psi(t) - \psi(\infty)] \exp(-i\omega t) dt, \qquad (2.33)$$

since $\tilde{\psi}(t) = \psi(t)$ for t > 0. (To demonstrate (2.33) it is convenient to derive (2.32) from (2.33) using integration by parts²).

We decompose the complex modulus into real and imaginary parts

$$M(\omega) = M_1(\omega) + iM_2(\omega),$$
 (2.34)

where

$$M_1(\omega) = \psi(0^+) + \int_0^\infty \partial_t \tilde{\psi}(t) \cos(\omega t) dt = \psi(\infty) + \omega \int_0^\infty [\psi(t) - \psi(\infty)] \sin(\omega t) dt, \quad (2.35)$$

or,

$$M_1(\omega) = \omega \int_0^\infty \psi(t) \sin(\omega t) dt$$
(2.36)

is the storage modulus, and

$$M_2(\omega) = -\int_0^\infty \partial_t \tilde{\psi}(t) \sin(\omega t) dt = \omega \int_0^\infty [\psi(t) - \psi(\infty)] \cos(\omega t) dt \qquad (2.37)$$

is the loss modulus. To obtain equation (2.36), we have used the property

$$\omega \int_{0}^{\infty} \sin(\omega t) dt = 1 \qquad (2.38)$$

(Golden and Graham, 1988, p. 243).

In the strain-stress relation

$$\epsilon = \chi * \partial_t \sigma$$
, (2.39)

the function χ is referred to as the creep function. Since

$$\sigma = \partial_t \psi * \epsilon = \partial_t \psi * (\partial_t \chi * \sigma) = (\partial_t \psi * \partial_t \chi) * \sigma,$$
 (2.40)

we have

$$\partial_t \psi(t) * \partial_t \chi(t) = \delta(t),$$
 (2.41)

and

$$M(\omega)J(\omega) = 1,$$
 (2.42)

where

$$J(\omega) = F[\partial_t \chi]$$
 (2.43)

is the complex creep compliance.

²Proof: $i\omega \int_{0}^{\infty} [\psi(t) - \psi(\infty)] \exp(-i\omega t) dt = -\int_{0}^{\infty} [\psi(t) - \psi(\infty)] d\exp(-i\omega t) = - [[\psi(t) - \psi(\infty)] \exp(-i\omega t)] \exp(-i\omega t) dt = - [\psi(\infty) - \psi(\infty)] \exp(-i\omega \infty) + [\psi(0) - \psi(\infty)] \exp(-i\omega 0) + \int_{0}^{\infty} \partial_t \psi(t) \exp(-i\omega t) dt = \psi(0) - \psi(\infty) + \int_{0}^{\infty} \partial_t \psi(t) \exp(-i\omega t) dt.$

2.2.2 Energy and significance of the storage and loss moduli

Let us calculate the time-averaged strain-energy density (2.16) for harmonic fields of the form [+] exp(i ωt). The change of variables $\tau_1 \rightarrow t - \tau_1$ and $\tau_2 \rightarrow t - \tau_2$ yields

$$V(t) = \frac{1}{2} \int_0^\infty \int_0^\infty \tilde{\psi}(\tau_1 + \tau_2) \partial \epsilon(t - \tau_1) \partial \epsilon(t - \tau_2) d\tau_1 d\tau_2.$$
(2.44)

We now average this equation over a period $2\pi/\omega$ using the property (1.105) and obtain

$$\langle \partial \epsilon(t-\tau_1) \partial \epsilon(t-\tau_2) \rangle = \frac{1}{2} \operatorname{Re} \{ \partial \epsilon(t-\tau_1) [\partial \epsilon(t-\tau_2)]^* \} = \frac{1}{2} \omega^2 |\epsilon|^2 \cos[\omega(\tau_2-\tau_1)], \quad (2.45)$$

Then, the time average of equation (2.44) is

$$\langle V \rangle = \frac{1}{4} \omega^2 |\epsilon|^2 \int_0^\infty \int_0^\infty \check{\psi}(\tau_1 + \tau_2) \cos[\omega(\tau_1 - \tau_2)] d\tau_1 d\tau_2.$$
 (2.46)

A new change of variables $\zeta = \tau_1 + \tau_2$ and $\varsigma = \tau_1 - \tau_2$ gives

$$\langle V \rangle = \frac{1}{8}\omega^2 |\epsilon|^2 \int_0^\infty \int_{-\zeta}^{\zeta} \check{\psi}(\zeta) \cos(\omega\zeta) d\zeta d\zeta = \frac{1}{4}\omega |\epsilon|^2 \int_0^\infty \check{\psi}(\zeta) \sin(\omega\zeta) d\zeta.$$
(2.47)

Using equation (2.36), we finally get

$$\langle V \rangle = \frac{1}{4} |\epsilon|^2 M_1.$$
 (2.48)

A similar calculation shows that

$$\langle \dot{D} \rangle = \frac{1}{2} \omega |\epsilon|^2 M_2.$$
 (2.49)

These equations justify the terminology used for the storage and loss moduli M_1 and M_2 . Moreover, since the time-averaged strain and dissipated energies should be non-negative, it follows that

$$M_1(\omega) \ge 0$$
, $M_2(\omega) \ge 0$. (2.50)

2.2.3 Non-negative work requirements and other conditions

The work done to deform the material from the initial state must be non-negative

$$\frac{1}{t} \int_{0}^{t} \sigma(\tau) \partial_{\tau} \epsilon(\tau) d\tau \ge 0 \tag{2.51}$$

(Christensen, 1982, p. 86). Let us consider oscillations in the form of sinusoidally time variations

$$\epsilon(\tau) = \epsilon_0 \sin(\omega \tau),$$
 (2.52)

and let $t = 2\pi/\omega$ be one period, corresponding to a cycle. Using equation (2.31) (see equations (2.36) and (2.37)), we note that the stress-strain relation (2.28) becomes

$$\sigma(t) = \epsilon_0 \int_0^\infty \partial_\tau \psi \sin[\omega(t-\tau)] d\tau = \epsilon_0 [M_1 \sin(\omega t) + M_2 \cos(\omega t)].$$
(2.53)

Substitution of (2.53) into the inequality (2.51) gives

$$\frac{\omega^2}{2\pi}\epsilon_0^2 \left[M_1 \int_0^{2\pi/\omega} \sin(\omega\tau)\cos(\omega\tau)d\tau + M_2 \int_0^{2\pi/\omega}\cos^2(\omega\tau) \right] d\tau \ge 0.$$
 (2.54)

We now make use of the primitive integral $\int \cos^2(ax)dx = (x/2) + [\sin(2ax)/(4a)]$. The first integral vanishes, and the second integral is equal to π/ω . The condition is then

$$\frac{\omega}{2}M_2\epsilon_0^2 \ge 0$$
, or $M_2 \ge 0$, (2.55)

as found earlier (equation $(2.50)_2$). This result can also be obtained by using complex notation and the time-average formula (1.105).

We have shown, in addition, that $\langle \sigma \partial_t \epsilon \rangle = \langle \dot{D} \rangle$, if we compare our results to equation (2.49). From equation (2.12), this means that the time average of the strain-energy rate, $\langle \partial_t V \rangle$, is equal to zero.

Equation (2.37) and condition (2.55) imply that

$$\psi(t) - \psi(\infty) \ge 0.$$
 (2.56)

Then,

$$\psi(t = 0) \ge \psi(t = \infty).$$
 (2.57)

Note that from (2.33), we have

$$M(\omega = 0) = \psi(t = \infty),$$
 (2.58)

i.e., a real quantity. Moreover, from (2.33) and using $i\omega \mathcal{F}[f(t)] = f(t = 0)$, for $\omega \to \infty$ (Golden and Graham, 1988, p. 244), we have

$$M(\omega = \infty) = \psi(t = 0),$$
 (2.59)

also a real quantity. We then conclude that $M_2 = 0$ at the low- and high-frequency limits, and

$$M(\omega = \infty) \ge M(\omega = 0).$$
 (2.60)

(As shown in the next section, the validity of some of these properties requires $|M(\omega)|$ to be a bounded function).

Additional conditions on the relaxation function, based on the requirements of positive work and positive rate of dissipation, can be obtained from the general conditions (2.18) and (2.19),

$$\psi(t) \ge 0$$
, (2.61)

$$\partial_t \psi(t) \le 0.$$
 (2.62)

2.2.4 Consequences of reality and causality

Equation (2.28) can also be written as

$$\sigma = \psi * \epsilon$$
, (2.63)

where

$$\psi \equiv \partial_t \psi$$
. (2.64)

Note that $M = \mathcal{F}[\hat{\psi}]$ (equation (2.31)). Since $\hat{\psi}(t)$ is real, $M(\omega)$ is Hermitian (Bracewell, 1965, p. 16); that is

$$M(\omega) = M^{*}(-\omega),$$
 (2.65)

or

$$M_1(\omega) = M_1(-\omega), \quad M_2(\omega) = -M_2(-\omega).$$
 (2.66)

Furthermore, $\dot{\psi}$ can split into even and odd functions of time, $\dot{\psi}_e$ and $\dot{\psi}_o$, respectively, as

$$\dot{\psi}(t) = \frac{1}{2}[\dot{\psi}(t) + \dot{\psi}(-t)] + \frac{1}{2}[\dot{\psi}(t) - \dot{\psi}(-t)] \equiv \dot{\psi}_e + \dot{\psi}_e.$$
 (2.67)

Since $\dot{\psi}$ is causal, $\dot{\psi}_o(t) = \operatorname{sgn}(t)\dot{\psi}_e(t)$, and

$$\dot{\psi}(t) = [1 + \text{sgn}(t)]\dot{\psi}_e(t),$$
 (2.68)

whose Fourier transform is

$$\mathcal{F}[\dot{\psi}(t)] = M_1(\omega) - \left(\frac{i}{\pi\omega}\right) * M_1(\omega),$$
 (2.69)

because $\mathcal{F}[\dot{\psi}_e] = M_1$ and $\mathcal{F}[\text{sgn}(t)] = -i/(\pi\omega)$ (Bracewell, 1965, p. 272). Equation (2.69) implies

$$M_2 = -\left(\frac{1}{\pi\omega}\right) * M_1 = -\frac{1}{\pi} \operatorname{pv} \int_{-\infty}^{\infty} \frac{M_1(\omega')d\omega'}{\omega - \omega'}.$$
 (2.70)

Similarly, since $\dot{\psi}_{\epsilon}(t) = \operatorname{sgn}(t)\dot{\psi}_{o}(t)$,

$$\dot{\psi}(t) = [\operatorname{sgn}(t) + 1]\dot{\psi}_o(t)$$
 (2.71)

and since $\mathcal{F}[\hat{\psi}_o] = iM_2$, we obtain

$$M_1 = \left(\frac{1}{\pi\omega}\right) * M_2 = \frac{1}{\pi} \operatorname{pv} \int_{-\infty}^{\infty} \frac{M_2(\omega')d\omega'}{\omega - \omega'}.$$
 (2.72)

Equations (2.70) and (2.72) are known as Kramers-Kronig dispersion relations (Kronig, 1926; Kramers, 1927). In mathematical terms, M_1 and M_2 are Hilbert transform pairs (Bracewell, 1965, p. 267-272). Causality also implies that M has no poles (or is analytic) in the lower half complex ω -plane (Golden and Graham, 1988, p. 48). In the case of dispersive lossless media, $M_1(\omega)$ can depend on ω only through functions of ω whose Hilbert transform is zero.

Equations (2.70) and (2.72) are a consequence of linearity, causality and squareintegrability of $M(\omega)$ along the real axis of the ω -plane, i.e.,

$$\int_{-\infty}^{\infty} |M(\omega)|^2 d\omega < C, \qquad (2.73)$$

where C is a constant (Weaver and Pao, 1981). Square-integrability is equivalent to $M(\omega) \rightarrow 0$, for $|\omega| \rightarrow \infty$ ($\pi \ge \arg(\omega) \ge 0$). In most cases, the square-integrability condition cannot be satisfied, but rather the weaker condition that $|M(\omega)|$ is bounded is satisfied, i.e., $|M(\omega)|^2 < C$ is bounded. A lossless medium and the indexNiMaxwell-Maxwell and Zener models satisfy this weak condition (see Section 2.4.1, equation (2.147), and Section 2.4.3, equation (2.170)), but the elvin-Voigt and constant-Q models do not (see Section 2.4.2, equation (2.161), and Section 2.5, equation (2.212)). For models satisfying the weak condition, we may construct a new function

$$H(\omega) = \frac{M(\omega) - M(\omega_0)}{\omega - \omega_0}, \quad \text{Im}(\omega_0) \ge 0.$$
 (2.74)

This function is square-integrable and has no poles in the upper half plane, and, hence, satisfies equations (2.70) and (2.72). Substituting $H(\omega)$ as defined above for $M(\omega)$ in equations (2.70) and (2.72) and taking ω_0 to be real, we obtain

$$M_1(\omega) = M_1(\omega_0) + \left(\frac{\omega - \omega_0}{\pi}\right) \operatorname{pv} \int_{-\infty}^{\infty} \operatorname{Im} \left[\frac{M(\omega') - M(\omega_0)}{\omega' - \omega_0}\right] \frac{d\omega'}{\omega - \omega'}, \quad (2.75)$$

$$M_2(\omega) = M_2(\omega_0) - \left(\frac{\omega - \omega_0}{\pi}\right) \operatorname{pv} \int_{-\infty}^{\infty} \operatorname{Re} \left[\frac{M(\omega') - M(\omega_0)}{\omega' - \omega_0}\right] \frac{d\omega'}{\omega - \omega'}.$$
 (2.76)

(Weaver and Pao, 1981). These are known as dispersion relations for $M(\omega)$ with one subtraction. Further subtractions may be taken if $M(\omega)$ is bounded by a polynomial function of ω .

2.2.5 Summary of the main properties

Relaxation function

- 1. It is causal.
- It is a positive real function.
- 3. It is a decreasing function of time.

Complex modulus

- It is an Hermitian function of ω.
- Its real and imaginary parts are greater than zero, since the strain-energy density and the rate of dissipated-energy density must be positive.
- Its low- and high-frequency limits are real valued and coincide with the relaxed and instantaneous (unrelaxed) values of the relaxation function.
- 4. Its real and imaginary parts are Hilbert-transform pairs.
- It is analytic in the lower half complex ω-plane.

2.3 Wave propagation concepts for 1-D viscoelastic media

We note that the frequency-domain stress-strain relation (2.30) has the same form as the elastic stress-strain relation (2.25), but the modulus is complex and frequency dependent. The implications for wave propagation can be made clear if we consider the displacement plane wave

$$u = u_0 \exp[i(\omega t - kx)],$$
 (2.77)

where k is the complex wavenumber, and the balance between the surface and inertial forces is

$$\partial_1 \sigma = \rho \partial_{tt}^2 u$$
 (2.78)

(see equation (1.23)). Assuming constant material properties, using $\epsilon = \partial_1 u$ and equations (2.28) and (2.31), we obtain the dispersion relation

$$Mk^2 = \rho \omega^2$$
, (2.79)

which, for propagating waves (k complex, ω real), gives the complex velocity

$$v_e(\omega) = \frac{\omega}{k} = \sqrt{\frac{M(\omega)}{\rho}}.$$
 (2.80)

Expressing the complex wavenumber as

$$k = \kappa - i\alpha$$
, (2.81)

we can rewrite the plane wave (2.77) as

$$u = u_0 \exp(-\alpha x) \exp[i(\omega t - \kappa x)], \qquad (2.82)$$

meaning that κ is the wavenumber and α is the attenuation factor. We define the phase velocity

$$v_p = \frac{\omega}{\kappa} = \left[\operatorname{Re}\left(\frac{1}{v_c}\right) \right]^{-1},$$
 (2.83)

the real slowness

$$s_R = \frac{1}{v_p} = \operatorname{Re}\left(\frac{1}{v_c}\right),\tag{2.84}$$

and the attenuation factor

$$\alpha = -\omega \operatorname{Im}\left(\frac{1}{v_c}\right). \quad (2.85)$$

We have seen in Chapter 1 (equation (1.125)) that the velocity of the modulation wave is the derivative of the frequency with respect to the wavenumber. In this case, we should consider the real wavenumber κ ,

$$v_g = \frac{\partial \omega}{\partial \kappa} = \left(\frac{\partial \kappa}{\partial \omega}\right)^{-1} = \left[\operatorname{Re}\left(\frac{\partial k}{\partial \omega}\right)\right]^{-1}.$$
 (2.86)

Let us assume for the moment that u(t) is not restricted to the form (2.77). Since the particle velocity is $v = \partial_t u$, we multiply equation (2.78) on both sides by v to obtain

$$v\partial_t \sigma = \rho v \partial_t v.$$
 (2.87)

Multiplying $\partial_1 v = \partial_t \epsilon$ by σ and using equation (2.28), we have

$$\sigma \partial_1 v = (\partial_t \psi * \epsilon) \partial_t \epsilon.$$
 (2.88)

In the lossless case, $\psi = M_e H(t)$ and

$$\sigma \partial_1 v = M_e \epsilon \partial_t \epsilon.$$
 (2.89)

Adding equations (2.87) and (2.89), we obtain the energy-balance equation for dynamic elastic fields

$$-\partial_1 p = \partial_t (T + V) = \partial_t E,$$
 (2.90)

where

$$p = -\sigma v$$
 (2.91)

is the Umov-Poynting power flow,

$$T = \frac{1}{2}\rho v^2 \qquad (2.92)$$

is the kinetic-energy density, and

$$V = \frac{1}{2}M_e\epsilon^2$$
(2.93)

is the strain-energy density.

The balance equation in the lossy case is obtained by adding equations (2.87) and (2.88),

$$-\partial_1 p = \partial_t T + (\partial_t \psi * \epsilon) \partial_t \epsilon.$$
 (2.94)

In general, the partition of the second term in the right-hand side in terms of the rates of strain and dissipated energies is not unique (Caviglia and Morro, 1992, p. 56). The splitting (2.12) is one choice, consistent with the mechanical-model description of viscoelasticity – this is shown in Section 2.4.1 for the Maxwell model. A more general demonstration is given by Carcione (1999a) for the Zener model and Hunter (1983, p. 542) for an arbitrary array of springs and dashpots. We then can write

$$-\partial_1 p = \partial_t (T + V) + \dot{D}, \qquad (2.95)$$

where, from equations (2.16) and (2.17),

$$V(t) = \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} \check{\psi}(2t - \tau_1 - \tau_2) \partial_{\tau_1} \epsilon(\tau_1) \partial_{\tau_2} \epsilon(\tau_2) d\tau_1 d\tau_2, \qquad (2.96)$$

$$\dot{D}(t) = -\int_{-\infty}^{t} \int_{-\infty}^{t} \partial \breve{\psi}(2t - \tau_1 - \tau_2) \partial_{\tau_1} \epsilon(\tau_1) \partial_{\tau_2} \epsilon(\tau_2) d\tau_1 d\tau_2.$$
(2.97)

Let us consider again the form (2.77) for the displacement field. In order to compute the balance equation for average quantities, we obtain the complex versions of equations (2.87) and (2.88) by multiplying (2.78) by v^* and $(\partial_1 v)^* = (\partial_t \epsilon)^*$ by σ . We obtain

$$v^* \partial_1 \sigma = \rho v^* \partial_t v$$
, (2.98)

2.3 Wave propagation concepts for 1-D viscoelastic media

$$\sigma(\partial_t v)^* = (\partial_t \psi * \epsilon)(\partial_t \epsilon)^*.$$
 (2.99)

Because for the harmonic plane wave (2.77), $\partial_t \rightarrow i\omega$ and $\partial_1 \rightarrow -ik$, we can use equation (2.30) – omitting the tildes – to obtain

$$-kv^*\sigma = \omega \rho |v|^2$$
, (2.100)

and

$$-k^* \sigma v^* = \omega M |\epsilon|^2$$
. (2.101)

Now, using equations (1.105) and (1.106), we introduce the complex Umov-Poynting energy flow

$$p = -\frac{1}{2}\sigma v^*$$
, (2.102)

the time-averaged kinetic-energy density

$$\langle T \rangle = \frac{1}{2} \langle \rho[\operatorname{Re}(v)]^2 \rangle = \frac{1}{4} \rho \operatorname{Re}(vv^*) = \frac{1}{4} \rho |v|^2,$$
 (2.103)

the time-averaged strain-energy density

$$\langle V \rangle = \frac{1}{2} \langle \operatorname{Re}(\epsilon) \operatorname{Re}(M) \operatorname{Re}(\epsilon) \rangle = \frac{1}{4} \operatorname{Re}(\epsilon M \epsilon^*) = \frac{1}{4} |\epsilon|^2 M_1,$$
 (2.104)

and the time-averaged rate of dissipated-energy density

$$\langle \dot{D} \rangle = \omega \langle \operatorname{Re}(\epsilon) \operatorname{Im}(M) \operatorname{Re}(\epsilon) \rangle = \frac{1}{2} \omega \operatorname{Im}(\epsilon M \epsilon^*) = \frac{1}{2} \omega |\epsilon|^2 M_2,$$
 (2.105)

in agreement with equations (2.48) and (2.49). We can, alternatively, define the timeaveraged dissipated-energy density $\langle D \rangle$ as

$$(D) = \omega^{-1} (\dot{D}), \quad \omega > 0$$
 (2.106)

(there is no loss at zero frequency). Thus, in terms of the energy flow and energy densities, equations (2.100) and (2.101) become

$$kp = 2\omega(T)$$
, (2.107)

and

$$k^* p = 2\omega \langle V \rangle + i \langle \hat{D} \rangle.$$
 (2.108)

Because the right-hand side of (2.107) is real, kp is also real. Adding equations (2.107) and (2.108) and using $k + k^* = 2\kappa$ (see equation (2.81)), we have

$$\kappa p = \omega \langle E \rangle + \frac{i}{2} \langle \dot{D} \rangle,$$
 (2.109)

where

$$\langle E \rangle = \langle T + V \rangle$$
 (2.110)

is the time-averaged energy density. Separating equation (2.109) into real and imaginary parts, we obtain

$$\kappa(p) = \omega(E)$$
 (2.111)

63

and

$$\kappa \operatorname{Im}(p) = \frac{1}{2} \langle \dot{D} \rangle,$$
 (2.112)

where

$$(p) = \text{Re}(p)$$
 (2.113)

is the time-averaged power-flow density. The energy velocity is defined as

$$v_e = \frac{\langle p \rangle}{\langle E \rangle}$$
. (2.114)

Now, note from equations (2.102)-(2.104) that

$$\langle p \rangle = \operatorname{Re}(p) = -\frac{1}{2}\operatorname{Re}(v^*\sigma) = -\frac{1}{2}\operatorname{Re}[(-i\omega u^*)(M\epsilon)]$$

$$= -\frac{1}{2}\operatorname{Re}[(-i\omega u^*)(-ikMu)] = \frac{1}{2}\omega |u|^2\operatorname{Re}(kM), \qquad (2.115)$$

and

$$\langle E \rangle = \langle T \rangle + \langle V \rangle = \frac{1}{4}\rho |v|^2 + \frac{1}{4}M_1 |\epsilon|^2 = \frac{1}{4}\rho \omega^2 |u|^2 + \frac{1}{4}M_1 |k|^2 |u|^2.$$
(2.116)

Substituting these expression into equation (2.114) and using (2.80), we obtain

$$v_e = \frac{2 \text{Re}(M v_e^{-1}) |v_e|^2}{\rho |v_e|^2 + M_1}$$
 (2.117)

Because $M_1 = \text{Re}(M)$, $M = \rho v_e^2$, and using properties of complex numbers (in particular, $[\text{Re}(v_e)]^2 = (|v_e|^2 + \text{Re}(v_e^2)]/2)$, we finally obtain

$$v_e = |v_e|^2 [\operatorname{Re}(v_e)]^{-1} = v_p,$$
 (2.118)

where v_p is the phase velocity (2.83). We then have the result that the energy velocity is equal to the phase velocity in 1-D viscoelastic media. (Note that this result confirms the relation (2.111)). In the next chapter, we show that this is also the case for homogeneous viscoelastic plane waves in 2-D and 3-D isotropic media. We have seen in Chapter 1 that phase and energy velocities differ in anisotropic elastic media, and that the group velocity is equal to the energy velocity. This result proved very useful in the computation of the wave-front surfaces. However, the group velocity loses its physical meaning in viscoelastic media due to the dispersion of the harmonic components of the signal. We investigate this in some detail in Section 2.6 (1-D case), and Section 4.4.5, where we discuss wave propagation in anisotropic viscoelastic media.

Dissipation can also be quantified by the quality factor Q, whose inverse, Q^{-1} , is called the dissipation factor. Here, we define the quality factor as twice the time-averaged strainenergy density divided by the time-averaged dissipated-energy density (2.106). Hence, we have

$$Q = \frac{2\langle V \rangle}{\langle D \rangle}, \qquad (2.119)$$

which, by virtue of equations (2.80), (2.104) and (2.105) becomes

$$Q = \frac{M_1}{M_2} = \frac{\text{Re}(M)}{\text{Im}(M)} = \frac{\text{Re}(v_c^2)}{\text{Im}(v_c^2)}.$$
 (2.120)

64

Another form of the quality factor can be obtained from the definition of complex velocity (2.80). It can be easily shown that

$$Q = -\frac{\text{Re}(k^2)}{\text{Im}(k^2)}.$$
 (2.121)

Since $k^2 = \kappa^2 - \alpha^2 - 2i\kappa\alpha$, it follows from (2.85) and (2.121) that the quality factor is related to the magnitudes of the attenuation factor and the wavenumber by

$$\alpha = \left(\sqrt{Q^2 + 1} - Q\right)\kappa. \tag{2.122}$$

For low-loss solids, it is $Q \gg 1$, and using (2.83) and $f = \omega/(2\pi)$, we note that a Taylor expansion yields

$$\alpha = \frac{\kappa}{2Q} = \frac{\omega}{2Qv_p} = \frac{\pi f}{Qv_p},$$
(2.123)

Another common definition of quality factor is

$$Q = \frac{\langle E \rangle}{\langle D \rangle}$$
(2.124)

(Buchen, 1971a). It can be easily shown that this definition leads to a relation similar to (2.123), without approximations; that is $\alpha = \pi f/(Qv_p)$.

2.3.1 Wave propagation for complex frequencies

The analysis of wave propagation can also be performed for complex frequencies and real wavenumbers. Let us consider the 1-D case and the displacement plane wave

$$u = u_0 \exp[i(\Omega t - \kappa x)],$$
 (2.125)

where $\Omega = \omega + i\omega_I$ is the complex frequency, and κ is the real wavenumber. It is clear that the phase velocity is equal to ω/κ .

The balance between the surface and inertial forces is given by

$$\partial_1 \sigma = \rho \partial_{tt} u_i$$
 (2.126)

where σ is the stress. Since $\sigma = M\epsilon = M\partial_1 u$, where ϵ is the strain, we obtain the dispersion relation

$$M\kappa^2 = \rho \Omega^2$$
, (2.127)

which gives the complex velocity

$$v_c(\kappa) = \frac{\Omega}{\kappa} = \sqrt{\frac{M[\Omega(\kappa)]}{\rho}}.$$
 (2.128)

The phase velocity is

$$v_p = \frac{\operatorname{Re}(\Omega)}{\kappa} = \frac{\omega}{\kappa} = \operatorname{Re}(v_c).$$
 (2.129)

In order to compute the balance equation for average quantities, we note that

$$-\kappa v^* \sigma = \Omega \rho |v|^2 \qquad (2.130)$$

and

$$-\kappa v^* \sigma = \Omega^* M |\epsilon|^2$$
, (2.131)

where the asterisk indicates complex conjugate. These equations were obtained by multiplying (2.126) by v^* and $(\partial_1 v)^* = (\partial_t \epsilon)^*$ by σ , respectively.

We introduce the complex Umov-Poynting energy flow

$$p = -\frac{1}{2}\sigma v^*$$
, (2.132)

the time-averaged kinetic-energy density

$$\langle T \rangle = \frac{1}{2} \langle \rho[\text{Re}(v)]^2 \rangle = \frac{1}{4} \rho \text{Re}(vv^*) = \frac{1}{4} \rho |v|^2,$$
 (2.133)

the time-averaged strain-energy density

$$\langle V \rangle = \frac{1}{2} \langle \operatorname{Re}(\epsilon) \operatorname{Re}(M) \operatorname{Re}(\epsilon) \rangle = \frac{1}{4} \operatorname{Re}(\epsilon M \epsilon^*) = \frac{1}{4} |\epsilon|^2 \operatorname{Re}(M) = \frac{1}{4} |\epsilon|^2 M_1, \quad (2.134)$$

and the time-averaged dissipated-energy density

$$\langle D \rangle = \langle \operatorname{Re}(\epsilon) \operatorname{Im}(M) \operatorname{Re}(\epsilon) \rangle = \frac{1}{2} \operatorname{Im}(\epsilon M \epsilon^*) = \frac{1}{2} |\epsilon|^2 \operatorname{Im}(M) = \frac{1}{2} |\epsilon|^2 M_2.$$
 (2.135)

Thus, in terms of the energy flow and energy densities, equations (2.130) and (2.131) become

$$\frac{\kappa p}{\Omega} = 2\langle T \rangle$$
(2.136)

and

$$\frac{\kappa p}{\Omega^*} = 2\langle V \rangle + i \langle D \rangle.$$
 (2.137)

Adding these equations, we have

$$2\kappa p \operatorname{Re}\left(\frac{1}{\Omega}\right) = 2\omega \langle E \rangle + i \langle D \rangle,$$
 (2.138)

where

$$\langle E \rangle = \langle T + V \rangle$$
 (2.139)

is the time-averaged energy density.

Separating equation (2.138) into real and imaginary parts and using equation (2.113), the energy velocity is given by

$$v_e = \frac{\langle p \rangle}{\langle E \rangle} = \left[\kappa \operatorname{Re} \left(\frac{1}{\Omega} \right) \right]^{-1} = \left[\operatorname{Re} \left(\frac{1}{v_e} \right) \right]^{-1},$$
 (2.140)

i.e., the energy velocity has the same expression as a function of the complex velocity, irrespective of the fact that the frequency is complex or the wavenumber is complex. On the contrary, the phase velocity is given by equation (2.83) for real frequencies and by equation (2.129) for real wavenumbers and complex frequencies.

66



Figure 2.1: Creep function of aluminum and typical relaxation spectrum (after Zener, 1948).

2.4 Mechanical models and wave propagation

A typical creep function versus time, as well as a dissipation factor versus frequency are shown in Figure 2.1. These behaviors can be described by using viscoelastic constitutive equations based on mechanical models. To construct a mechanical model, two types of basic elements are required: weightless springs – no inertial effects are present – that represent the elastic solid, and dashpots, consisting of loosely fitting pistons in cylinders filled with a viscous fluid. The simplest are the Maxwell and Kelvin-Voigt models. The Maxwell model was introduced by Maxwell (1867) when discussing the nature of viscosity in gases. Meyer (1874) and Voigt (1892) obtained the so-called Voigt stress-strain relation by generalizing the equations of classical elasticity. The mechanical model representation of the Voigt solid (the Kelvin-Voigt model) was introduced by Lord Kelvin (Kelvin, 1875).

The relaxation function can be obtained by measuring the stress after imposing a rapidly constant unit strain in a relaxed sample of the medium, i.e., $\epsilon = H(t)$, such that (2.28) becomes

$$\sigma(t) = \partial_t \psi(t) * H(t) = \psi(t) * \delta(t) = \psi(t).$$
 (2.141)

A constant state of stress instantaneously applied to the sample ($\sigma = H(t)$), with the resulting strain being measured as a function of time, describes the creep experiment. The resulting time function is the creep function. That is

$$\epsilon(t) = \partial_t \chi(t) * H(t) = \chi(t) * \delta(t) = \chi(t).$$
 (2.142)

There are materials for which creep continues indefinitely as time increases. If the limit $\partial_t \chi(t = \infty)$ is finite, permanent deformation occurs after the application of a stress field. Such behavior is akin to that of viscoelastic fluids. If that quantity is zero, the material is referred to as a viscoelastic solid. If χ increases indefinitely, the relaxation function ψ must tend to zero, according to (2.41). This is another criterion to distinguish between fluid and solid behavior: that is, for fluid-like materials ψ tends to zero; for solid-like materials, ψ tends to a finite value.

2.4.1 Maxwell model

The simplest series combination of mechanical models is the Maxwell model depicted in Figure 2.2. A given stress σ applied to the model produces a deformation ϵ_1 on the spring and a deformation ϵ_2 on the dashpot. The stress-strain relation in the spring is

$$\sigma = M_U \epsilon_1$$
, (2.143)

where M_U is the elasticity constant of the spring (M_e in equation (2.25)). The subindex U denotes "unrelaxed". Its meaning will become clear in the following discussion. The stress-strain relation in the dashpot is

$$\sigma = \eta \partial_t \epsilon_2, \quad \eta \ge 0,$$
 (2.144)

where η is the viscosity. Assuming that the total elongation of the system is $\epsilon = \epsilon_1 + \epsilon_2$, the stress-strain relation of the Maxwell element is

$$\frac{\partial_t \sigma}{M_U} + \frac{\sigma}{\eta} = \partial_t \epsilon.$$
 (2.145)



The Fourier transform of equation (2.145), or equivalently, the substitution of a harmonic wave $[\cdot] \exp(i\omega t)$, yields

$$\sigma = M \epsilon$$
, (2.146)

where

$$M(\omega) = \frac{\omega \eta}{\omega \tau - i} \qquad (2.147)$$

is the complex modulus, with

$$\tau = \frac{\eta}{M_U}$$
 (2.148)

being a relaxation time.

The corresponding relaxation function is

$$\psi(t) = M_U \exp(-t/\tau)H(t).$$
 (2.149)

This can be verified by performing the Boltzmann operation (2.6),

$$\partial_t \psi = \psi \odot \delta = M_U \delta(t) - \frac{M_U}{\tau} \exp(-t/\tau)H(t),$$
 (2.150)

and calculating the complex modulus (2.31),

$$\mathcal{F}[\partial_t \psi] = \int_{-\infty}^{\infty} \partial_t \psi \exp(-\mathrm{i}\omega t) dt = M_U - \frac{M_U}{1 + \mathrm{i}\omega\tau} = \frac{\omega\eta}{\omega\tau - \mathrm{i}}.$$
 (2.151)

The complex modulus (2.147) and the relaxation function (2.149) can be shown to satisfy all the requirements listed in Section 2.2.5. Using equations (2.41) and (2.42), we note that the creep function of the Maxwell model is

$$\chi(t) = \frac{1}{M_U} \left(1 + \frac{t}{\tau}\right) H(t). \qquad (2.152)$$

The creep and relaxation functions are depicted in Figure 2.3a-b, respectively. As can be seen, the creep function is not representative of the real creep behavior in real solids. Rather, it resembles the creep function of a viscous fluid. In the relaxation





Figure 2.3: Creep (a) and relaxation (b) functions of the Maxwell model ($M_U = 2.16$ GPa, $\tau = 1/(2\pi f)$, f = 25 Hz). The creep function resembles the creep function of a viscous fluid. The system does not present an asymptotical residual stress as in the case of real solids.

experiment, both the spring and the dashpot experience the same force, and because it is not possible to have an instantaneous deformation in the dashpot, the extension is initially in the spring. The dashpot extends and the spring contracts, such that the total elongation remains constant. At the end, the force in the spring relaxes completely and the relaxation function does not present an asymptotical residual stress, as in the case of real solids. In conclusion, the Maxwell model appears more appropriate for representing a viscoelastic fluid. We can see from Figure 2.3a that M_U represents the instantaneous response of the system, hence, the name unrelaxed modulus.

We have seen in Section 2.3 that the partition of the second term in the right-hand side of equation (2.94) in terms of the rate of strain-energy density and rate of dissipatedenergy density is, in general, not unique. We have claimed that the splitting (2.12) is consistent with the mechanical-model description of viscoelasticity. As an example, we verify the correctness of the general form (2.16) (or (2.96)) for the Maxwell model. Substituting the relaxation function (2.149) into that equation, we obtain

$$V(t) = \frac{1}{2M_U} \left\{ \int_{-\infty}^t M_U \exp[-(t-\tau_1)/\tau] \partial_{\tau_1} \epsilon(\tau_1) d\tau_1 \right\}^2 = \frac{1}{2M_U} \left\{ \int_{-\infty}^\infty \psi(t-\tau_1) \partial_{\tau_1} \epsilon(\tau_1) d\tau_1 \right\}^2 = \frac{1}{2M_U} (\psi * \partial_t \epsilon)^2 = \frac{\sigma^2}{2M_U}.$$
 (2.153)

But this is precisely the energy stored in the spring, since, using (2.143) and the form (2.93), we obtain

$$V = \frac{1}{2}M_U \epsilon_1^2 = \frac{\sigma^2}{2M_U}.$$
 (2.154)

Note that because $\psi = \bar{\psi}H$, the second term in the right-hand side of (2.94) can be written as

$$(\partial_t \psi * \epsilon)\partial_t \epsilon = \psi(0)\epsilon \partial_t \epsilon + (\partial_t \bar{\psi} * \epsilon)\partial_t \epsilon,$$
 (2.155)

This is one possible partition and one may be tempted to identify the first term with the rate of strain-energy density. However, a simple calculation using the Maxwell model shows that this choice is not consistent with the energy stored in the spring.

The wave propagation properties are described by the phase velocity (2.83), the attenuation factor (2.85) and the quality factor (2.120). The quality factor has the simple expression

$$Q(\omega) = \omega \tau$$
. (2.156)



Figure 2.4: Phase velocity (a) and dissipation factor (b) of the Maxwell model ($M_U = \rho c^2$, $\rho = 2.4$ gr/cm³, c = 3 km/s, $\tau = 1/(2\pi f)$, f = 25 Hz). The system acts as a high-pass filter because low-frequency modes dissipate completely. The velocity for lossless media is obtained at the high-frequency limit. At low frequencies there is no propagation.

The phase velocity and dissipation factors are shown in Figures 2.4a-b, respectively. When $\omega \to 0$, then $v_p \to 0$, and $\omega \to \infty$ implies $v_p \to \sqrt{M_U/\rho}$, i.e., the velocity in the unrelaxed state. This means that a wave in a Maxwell material travels slower than a wave in the corresponding elastic material – if this is represented by the spring. The dissipation is infinite at zero frequency and the medium is lossless at high frequencies.

2.4.2 Kelvin-Voigt model

A viscoelastic model commonly used to describe anelastic effects is the Kelvin-Voigt stressstrain relation, which consists of a spring and a dashpot connected in parallel (Figure 2.5).

The total stress is composed of an elastic stress

$$\sigma_1 = M_R \epsilon$$
, (2.157)

where M_R is the spring constant – the subindex R denotes "relaxed" – and a viscous stress

$$\sigma_2 = \eta \partial_t \epsilon$$
, (2.158)

where ϵ is the total strain of the system. The stress-strain relation becomes

$$\sigma = \sigma_1 + \sigma_2 = M_R \epsilon + \eta \partial_t \epsilon. \qquad (2.159)$$



Figure 2.5: Mechanical model for a Kelvin-Voigt material. The strain on both elements is the same, but the forces are different.

The Fourier transform of (2.159) yields

$$\sigma = (M_R + i\omega\eta)\epsilon, \qquad (2.160)$$

which identifies the complex modulus

$$M(\omega) = M_R + i\omega\eta. \qquad (2.161)$$

The relaxation and creep functions are

$$ψ(t) = M_R H(t) + η δ(t), \qquad (2.162)$$

and

$$\chi(t) = \frac{1}{M_R} [1 - \exp(-t/\tau)] H(t), \qquad (2.163)$$

where $\tau = \eta / M_R$.

The calculation of the relaxation function from (2.159) is straightforward, and the creep function can be obtained by using (2.41) and (2.42) and Fourier-transform methods. The two functions are represented in Figure 2.6a-b, respectively.

The relaxation function does not show any time dependence. This is the case of pure elastic solids. The delta function implies that, in practice, it is impossible to impose an instantaneous strain on the medium. In the creep experiment, initially the dashpot extends and begins to transfer the stress to the spring. At the end, the entire stress is on the spring. The creep function does not present an instantaneous strain because the dashpot cannot move instantaneously. This is not the case of real solids. The creep function tends to the relaxed modulus M_R at infinite time.

The quality factor (2.120) is

$$Q(\omega) = (\omega \tau)^{-1}$$
. (2.164)

Comparing this equation to equation (2.156) shows that the quality factors of the Kelvin-Voigt and Maxwell models are reciprocal functions.



Figure 2.6: Creep (a) and relaxation (b) functions of the Kelvin-Voigt model ($M_R = 2.16$ GPa, $\tau = 1/(2\pi f)$, f = 25 Hz). The creep function lacks the instantaneous response of real solids. The relaxation function presents an almost elastic behavior.



Figure 2.7: Phase velocity (a) and dissipation factor (b) of the Kelvin-Voigt model ($M_R = \rho c^2$, $\rho = 2.4$ gr/cm³, c = 3 km/s, $\tau = 1/(2\pi f)$, f = 25 Hz). The system acts as a low-pass filter because high-frequency modes dissipate completely. The elastic (lossless) velocity is obtained at the low-frequency limit. High frequencies propagate with infinite velocity.

The phase velocity and dissipation factor are displayed in Figure 2.7a-b.

The Kelvin-Voigt model can be used to approximate the left slope of a real relaxation peak (see Figure 2.1). The phase velocity $v_p \rightarrow \sqrt{M_R/\rho}$ for $\omega \rightarrow 0$, and $v_p \rightarrow \infty$ for $\omega \rightarrow \infty$, which implies that a wave in a Kelvin-Voigt material travels faster than a wave in the corresponding elastic material.

2.4.3 Zener or standard linear solid model

A series combination of a spring and a Kelvin-Voigt model gives a more realistic representation of material media, such as rocks, polymers and metals. The resulting system, called the Zener model (Zener, 1948) or standard linear solid, is shown in Figure 2.8. This model was introduced by Poynting and Thomson (1902).



Figure 2.8: Mechanical model for a Zener material.

The stress-strain relations for the single elements are

$$\sigma = k_1 \epsilon_1,$$

 $\sigma_1 = \eta \partial_t \epsilon_2,$ (2.165)
 $\sigma_2 = k_2 \epsilon_2,$

with $k_1 \ge 0$, $k_2 \ge 0$ and $\eta \ge 0$. Moreover,

$$\sigma = \sigma_1 + \sigma_2$$
, $\epsilon = \epsilon_1 + \epsilon_2$. (2.166)

The solution of these equations for σ and ϵ gives the stress-strain relation

$$\sigma + \tau_{\sigma}\partial_t \sigma = M_R(\epsilon + \tau_{\epsilon}\partial_t \epsilon),$$
 (2.167)

where

$$M_R = \frac{k_1 k_2}{k_1 + k_2}, \quad (2.168)$$

is the relaxed modulus, and

$$\tau_{\sigma} = \frac{\eta}{k_1 + k_2}, \quad \tau_{\epsilon} = \frac{\eta}{k_2} \ge \tau_{\sigma}$$
(2.169)

are the relaxation times.

As in the previous models, the complex modulus is obtained by performing a Fourier transform of the stress-strain relation (2.167),

$$M(\omega) = M_R \left(\frac{1 + i\omega\tau_{\epsilon}}{1 + i\omega\tau_{\sigma}}\right).$$
 (2.170)

The relaxed modulus M_R is obtained for $\omega = 0$, and the unrelaxed modulus

$$M_U = M_R \left(\frac{\tau_e}{\tau_\sigma}\right), \quad (M_U \ge M_R)$$
 (2.171)

for $\omega \rightarrow \infty$.

The stress-strain and strain-stress relations are

$$\sigma = \psi * \partial_t \epsilon, \quad \epsilon = \chi * \partial_t \sigma,$$
 (2.172)

where the relaxation and creep functions are

$$\psi(t) = M_R \left[1 - \left(1 - \frac{\tau_{\epsilon}}{\tau_{\sigma}} \right) \exp(-t/\tau_{\sigma}) \right] H(t)$$
(2.173)

and

$$\chi(t) = \frac{1}{M_R} \left[1 - \left(1 - \frac{\tau_\sigma}{\tau_\epsilon} \right) \exp(-t/\tau_\epsilon) \right] H(t).$$
 (2.174)

(As an exercise, the reader may obtain the complex modulus (2.170) by using equations (2.31) and (2.173)). Note that by the symmetry of the strain-stress relation (2.167), exchanging the roles of τ_{σ} and τ_{ϵ} and substituting M_R for M_R^{-1} in equation (2.173), the creep function (2.174) can be obtained.



Figure 2.9: Creep (a) and relaxation (b) functions of the Zener model ($M_R = 2.16$ GPa, $M_U = 29.4$ GPa, $\tau_0 = 1/(2\pi f)$, f = 25 Hz). The creep function presents an instantaneous response and a finite asymptotic value as in real solids. The relaxation function presents an instantaneous unrelaxed state, and at the end of the process, the system has relaxed completely to the relaxed modulus M_R . The curve in (a) is similar to the experimental creep function shown in 2.1.

The relaxation and creep functions are represented in Figure 2.9a-b, respectively. In the creep experiment, there is an instantaneous initial value $\chi(0^+) = M_U^{-1}$, and an asymptotic strain $\chi(\infty) = M_R^{-1}$, determined solely by the spring constants. After the first initial displacement, the force across the dashpot is gradually relaxed by deformation therein, resulting in a gradual increase in the observed overall deformation; finally, the asymptotic value is reached. Similarly, the relaxation function exhibits an instantaneous unrelaxed state of magnitude M_U . At the end of the process, the system has relaxed completely to the relaxed modulus M_R . Such a system, therefore, manifests the general features of the experimental creep function illustrated in Figure 2.1a. The relaxation function and complex modulus can be shown to satisfy all the requirements listed in Section 2.2.5.

The quality factor (2.120) is

$$Q(\omega) = \frac{1 + \omega^2 \tau_{\epsilon} \tau_{\sigma}}{\omega (\tau_{\epsilon} - \tau_{\sigma})}, \qquad (2.175)$$

where we have used equation (2.170).



Figure 2.10: Phase velocity (a) and dissipation factor (b) of the Zener model. $(M_R = \rho c_R^2, \rho = 2.4$ gr/cm³, $c_R = 3$ km/s, $M_U = \rho c_R^2$, $c_U = 3.5$ km/s, $\tau_0 = 1/(2\pi f)$, f = 25 Hz).

The phase velocity and dissipation factor Q^{-1} are shown in Figure 2.10a-b. The model has a relaxation peak at $\omega_0 = 1/\tau_0$, where

$$\tau_0 = \sqrt{\tau_e \tau_\sigma}$$
(2.176)

The phase velocity increases with frequency. (The same happens for the Maxwell and Kelvin-Voigt models). The type of dispersion in which this happens is called anomalous dispersion in the electromagnetic terminology. In electromagnetism, the index of refraction – defined as the velocity of light in a vacuum divided by the phase velocity – decreases with frequency for anomalously dispersive media (Born and Wolf, 1964, p. 18; Jones, 1986, p. 644).

The Zener model is suitable to represent relaxation mechanisms such as those illustrated in Figure 2.8b. Processes such as grain-boundary relaxation have to be explained by a distribution of relaxation peaks. This behavior is obtained by considering several Zener elements in series or in parallel, a system which is described in the next section. The phase velocity ranges from $\sqrt{M_R/\rho}$ at the low-frequency limit to $\sqrt{M_U/\rho}$ at the high-frequency limit, and the system exhibits a pure elastic behavior ($Q^{-1} = 0$) at both limits.

2.4.4 Burgers model

A unique model to describe both the transient and steady-state creep process is given by the Burgers model, which is formed with a series connection of a Zener element and a dashpot, or equivalently, a series connection of a Kelvin-Voigt element and a Maxwell element (Klausner, 1991). The model is shown in Figure 2.11, and the constitutive equations of the single elements are

$$\sigma_1 = k_2 \epsilon_2$$

$$\sigma_2 = \eta_2 \partial_t \epsilon_2 = i \omega \eta_2 \epsilon_2$$

$$\sigma = \eta_1 \partial_t \epsilon_3 = i \omega \eta_1 \epsilon_3$$

$$\sigma = k_1 \epsilon_1,$$
(2.177)

where a time Fourier transform is implicit.



Figure 2.11: Burgers's viscoelastic model. The response of the Burgers model is instantaneous elasticity, delayed elasticity (or viscoelasticity) and viscous flow, the latter described by the series dashpot. On removal of the perturbation, the instantaneous and delayed elasticity are recovered, and it remains the viscous flow. The viscoelastic creep – with steady-state creep – of rocksalt can be described by the Burgers model which includes the transient creep of the Zener model, which does not exhibit steady-state creep, and the steady-state creep of a Maxwell model. (Carcione, Helle and Gangi, 2006).

Since

$$\epsilon = \epsilon_1 + \epsilon_2 + \epsilon_3$$

 $\sigma = \sigma_1 + \sigma_2,$
(2.178)

we have

$$\sigma = \sigma_1 + \sigma_2 = (k_2 + i\omega\eta_2)\epsilon_2 \qquad (2.179)$$

and

$$\epsilon = \epsilon_1 + \epsilon_2 + \epsilon_3 = \frac{\sigma}{k_1} + \frac{\sigma}{i\omega\eta_1} + \frac{\sigma}{k_2 + i\omega\eta_2} \equiv J(\omega)\sigma, \qquad (2.180)$$

where

$$J(\omega) = \frac{1}{M(\omega)} = \frac{1}{k_1} + \frac{1}{i\omega\eta_1} + \frac{1}{k_2 + i\omega\eta_2}$$
(2.181)

is the complex creep compliance (2.43).

An inverse Fourier transforms of (2.181) and a time integration of the result leads to

$$\partial_t \chi(t) = \frac{\delta(t)}{k_1} + \frac{H(t)}{\eta_1} + \frac{1}{\eta_2} \exp(-t/\tau_t) H(t)$$
(2.182)

and

$$\chi(t) = \left\{ \frac{1}{k_1} + \frac{t}{\eta_1} + \frac{1}{k_2} \left[1 - \exp(-t/\tau_{\epsilon}) \right] \right\} H(t), \quad (2.183)$$

where τ_{ϵ} is given by equation (2.169)₂. Equation (2.183) can also be obtained by adding the creep functions of the Maxwell (M) and Kelvin-Voigt (KV) models (equations (2.152) and (2.163), respectively), because $\epsilon_2 = \chi_{KV} * \partial_t \sigma$ and $\epsilon_1 + \epsilon_3 = \chi_M * \partial_t \sigma$.

The calculation of the relaxation function is more tricky. The model obeys a timedomain differential equation, which can obtained by combining equations (2.177) and (2.178):

$$\partial_{tt}^2 \sigma + \left(\frac{k_1}{\eta_1} + \frac{k_1}{\eta_2} + \frac{k_2}{\eta_2}\right) \partial_t \sigma + \frac{k_1 k_2}{\eta_1 \eta_2} \sigma = k_1 \partial_{tt}^2 \epsilon + \frac{k_1 k_2}{\eta_2} \partial_t \epsilon.$$
(2.184)

The relaxation function $\psi(t) = \sigma(t)$ is obtained for $\epsilon(t) = H(t)$. Then, factorizing the left-hand side, equation (2.184) can be rewritten as

$$(\omega_1 \delta - \delta') * (\omega_2 \delta - \delta') * \psi = k_1 \delta' + \frac{k_1 k_2}{\eta_2} \delta,$$
 (2.185)

where $\delta' = \partial_t \delta$, and

$$(2\eta_1\eta_2)\omega_{1,2} = -b \pm \sqrt{b^2 - 4k_1k_2\eta_1\eta_2}, \quad b = k_1\eta_1 + k_1\eta_2 + k_2\eta_1.$$
 (2.186)

Hence, the relaxation function is

$$\psi = (\omega_1 \delta - \delta')^{-1} * (\omega_2 \delta - \delta')^{-1} * \left(k_1 \delta' + \frac{k_1 k_2}{\eta_2} \delta\right), \qquad (2.187)$$

where here ()⁻¹ denotes the inverse with respect to convolution. Since³

$$(\omega_{1,2}\delta - \delta')^{-1} = -\exp(\omega_{1,2}t)H(t),$$
 (2.188)

we finally obtain

$$\psi(t) = [A_1 \exp(-t/\tau_1) - A_2 \exp(-t/\tau_2)]H(t), \qquad (2.189)$$

where

$$\tau_{1,2} = -\frac{1}{\omega_{1,2}}$$
 and $A_{1,2} = \frac{k_1 k_2 + \omega_{1,2} \eta_2 k_1}{\eta_2 (\omega_1 - \omega_2)}$. (2.190)

78

³Equation (2.188) is equivalent to $(\omega_{1,2}\delta - \delta') * [-\exp(\omega_{1,2}t)H(t)]$, i.e., $(\omega_{1,2} - \partial_t)[-\exp(\omega_{1,2}t)H(t)] = \delta$, which is identically true.

2.4 Mechanical models and wave propagation

The models studied in the previous sections are limiting cases of the Burgers model. The Maxwell creep function (2.152) is obtained for $k_2 \rightarrow \infty$ and $\eta_2 \rightarrow 0$, where $M_U = k_1$, $\tau = \eta_1/k_1$ and $\tau_{\epsilon} = 0$. The Kelvin-Voigt creep function (2.163) is obtained for $k_1 \rightarrow \infty$ and $\eta_1 \rightarrow \infty$, where $M_R = k_2$ and $\tau = \tau_{\epsilon}$. The Zener creep function (2.174) is obtained for $\eta_1 \rightarrow \infty$, where $\tau_1 = \infty$, $\tau_2 = \tau_{\sigma}$, $A_1 = M_R$ and $A_2 = M_R(\tau_{\epsilon}/\tau_{\sigma} - 1)$.

An example of the use of the Burgers model to describe borehole stability is given in Carcione, Helle and Gangi (2006).

2.4.5 Generalized Zener model

As stated before, some processes, as for example, grain-boundary relaxation, have a dissipation factor that is much broader than a single relaxation curve. It seems natural to try to explain this broadening with a distribution of relaxation mechanisms. This approach was introduced by Liu, Anderson and Kanamori (1976) to obtain a nearly constant quality factor over the seismic frequency range of interest. Strictly, their model cannot be represented by mechanical elements, since it requires a spring of negative constant (Casula and Carcione, 1992). Here, we consider the parallel system shown in Figure 2.12, with LZener elements connected in parallel. The stress-strain relation for each single element is

$$\sigma_l + \tau_{\sigma l} \partial_l \sigma_l = M_{Rl}(\epsilon + \tau_{\epsilon l} \partial_l \epsilon), \quad l = 1, ..., L,$$
 (2.191)

where the relaxed moduli are given by

$$M_{Rl} = \frac{k_{1l}k_{2l}}{k_{1l} + k_{2l}},$$
 (2.192)

and the relaxation times by

$$\tau_{\sigma t} = \frac{\eta_l}{k_{1l} + k_{2l}}, \quad \tau_{el} = \frac{\eta_l}{k_{2l}}.$$
 (2.193)

According to (2.170), each complex modulus is given by

$$M_l(\omega) = M_{Rl} \left(\frac{1 + i\omega \tau_{el}}{1 + i\omega \tau_{el}} \right). \qquad (2.194)$$

The total stress acting on the system is $\sigma = \sum_{l=1}^{L} \sigma_l$. Therefore, the stress-strain relation in the frequency domain is

$$\sigma = \sum_{l=1}^{L} M_l \epsilon = \sum_{l=1}^{L} M_{Rl} \left(\frac{1 + i\omega \tau_{el}}{1 + i\omega \tau_{\sigma l}} \right) \epsilon.$$
(2.195)

We can choose $M_{Rl} = M_R/L$, and the complex modulus can be expressed as

$$M(\omega) = \sum_{l=1}^{L} M_l(\omega), \quad M_l(\omega) = \frac{M_R}{L} \left(\frac{1 + i\omega\tau_{cl}}{1 + i\omega\tau_{\sigma l}}\right), \quad (2.196)$$

thereby reducing the number of independent constants to 2L + 1.



Figure 2.12: Mechanical model for a generalized Zener material.

The relaxation function is easily obtained from the time-domain constitutive equation

$$\sigma = \sum_{l=1}^{L} \sigma_l = \sum_{l=1}^{L} \psi_l * \partial_t \epsilon \equiv \psi * \partial_t \epsilon, \qquad (2.197)$$

where ψ_l has the form (2.173), and

$$\psi(t) = M_R \left[1 - \frac{1}{L} \sum_{l=1}^{L} \left(1 - \frac{\tau_{el}}{\tau_{\sigma l}} \right) \exp(-t/\tau_{\sigma l}) \right] H(t).$$
 (2.198)

The unrelaxed modulus is obtained for t = 0,

$$M_U = M_R \left[1 - \frac{1}{L} \sum_{l=1}^{L} \left(1 - \frac{\tau_{cl}}{\tau_{\sigma l}} \right) \right] = \frac{M_R}{L} \sum_{l=1}^{L} \frac{\tau_{cl}}{\tau_{\sigma l}}.$$
 (2.199)

The relaxation function obtained by Liu, Anderson and Kanamori (1976) lacks the factor 1/L.

Nearly constant Q

In oil prospecting and seismology, constant-Q models are convenient to parameterize attenuation in rocks, since the frequency dependence is usually not known. Moreover, there is physical evidence that attenuation is almost linear with frequency – therefore Qis constant – in many frequency bands (McDonal, Angona, Milss, Sengbush, van Nostrand and White, 1958). The technique to obtain a nearly constant Q over a given frequency range is to consider equispaced relaxation mechanisms in a log(ω) scale (Liu, Anderson and Kanamori, 1976). We show, in the following discussion, how to obtain a constant-Qmodel for low-loss solids by using a simple algorithm, without curve fitting of the Q factor.

A more physical parameterization of a single Zener element can be obtained with the center frequency $\omega_0 = \tau_0^{-1}$, and the value of the quality factor at this frequency,

$$Q_0 = \frac{2\tau_0}{\tau_\epsilon - \tau_\sigma}.$$
 (2.200)

The quality factor (2.175) becomes

$$Q(\omega) = Q_0 \left(\frac{1 + \omega^2 \tau_0^2}{2\omega \tau_0}\right).$$
 (2.201)

Solving for τ_{σ} and τ_{ϵ} in equations (2.176) and (2.200), we obtain

$$\tau_{\epsilon} = \frac{\tau_0}{Q_0} \left(\sqrt{Q_0^2 + 1} + 1 \right) \quad \text{and} \quad \tau_{\sigma} = \frac{\tau_0}{Q_0} \left(\sqrt{Q_0^2 + 1} - 1 \right). \tag{2.202}$$

Now, the problem is to find a set of relaxation times τ_{el} and $\tau_{\sigma l}$ that gives an almost constant quality factor Q in a given frequency band centered at $\omega_{0m} = 1/\tau_{0m}$. This is the location of the mechanism situated at the middle of the band, which, for odd L, has the index m = L/2 - 1. As mentioned above, single relaxation peaks should be taken equidistant in a log(ω) scale. The quality factor of the system is

$$Q(\omega) = \frac{\text{Re}(M)}{\text{Im}(M)} = \frac{\text{Re}(\sum_{l=1}^{L} M_l)}{\text{Im}(\sum_{l=1}^{L} M_l)},$$
(2.203)

where M_l is given in equation (2.196)₂. Since $Q_l = \text{Re}(M_l)/\text{Im}(M_l)$ is the quality factor of each element, equation (2.203) becomes

$$Q(\omega) = \frac{\sum_{l=1}^{L} Q_l \text{Im}(M_l)}{\sum_{l=1}^{L} \text{Im}(M_l)},$$
(2.204)

where

$$Q_t(\omega) = Q_{0t} \left(\frac{1+\omega^2 \tau_{0t}^2}{2\omega \tau_{0t}}\right). \qquad (2.205)$$

Using equation (2.200) and assuming the low-loss approximation ($\tau_{\sigma l} \approx \tau_{0l}$), we have

$$\operatorname{Im}(M_l) = \frac{M_R}{L} \left[\frac{\omega(\tau_{el} - \tau_{\sigma l})}{1 + \omega^2 \tau_{\sigma l}^2} \right] \approx \frac{M_R}{L} \left[\frac{2\omega \tau_{0l}}{Q_{0l}(1 + \omega^2 \tau_{0l}^2)} \right] = \frac{M_R}{LQ_l}.$$
 (2.206)

We now choose $Q_{0l} = Q_0$, and substitute equation (2.206) into equation (2.204) to obtain

$$Q(\omega) = LQ_0 \left(\sum_{l=1}^{L} \frac{2\omega\tau_{0l}}{1+\omega^2\tau_{0l}^2}\right)^{-1}.$$
 (2.207)

We choose τ_{0l} regularly distributed in the log(ω) axis, and $Q(\omega_{0m}) = \bar{Q}$, the desired value of the quality factor.



Figure 2.13: Phase velocity (a) and dissipation factor (b) of the generalized Zener model.

Thus, the choice

$$Q_0 = \frac{\bar{Q}}{L} \sum_{l=1}^{L} \frac{2\omega_{0m}\tau_{0l}}{1 + \omega_{0m}^2\tau_{0l}^2} \qquad (2.208)$$

gives a constant Q (equal to \bar{Q}), as can be verified by substitution of (2.208) into (2.207).

Figure 2.13 shows the phase velocity (a) and the dissipation factor (b) versus frequency, for five dissipation mechanisms – each with a quality-factor parameter $Q_0 = 15$, such that $\bar{Q} = 30$. The dotted curves are the quality factor of each single mechanism, and the vertical dotted line indicates the location of the third relaxation peak. The relaxation function of the nearly constant-Q model is shown in Figure 2.14.



Figure 2.14: Relaxation function of the generalized Zener model.

2.4.6 Nearly constant-Q model with a continuous spectrum

A linear and continuous superposition of Zener elements, where each element has equal weight, gives a continuous relaxation spectrum with a constant quality factor over a given frequency band (Liu, Anderson and Kanamori, 1976; Ben-Menahem and Singh, 1981, p. 911). The resulting relaxation function exhibits elastic (lossless) behavior in the low- and high-frequency limits. Its frequency-domain form is

$$M(\omega) = M_R \left[1 + \frac{2}{\pi \bar{Q}} \ln \left(\frac{1 + i\omega \tau_2}{1 + i\omega \tau_1} \right) \right]^{-1}, \qquad (2.209)$$

where τ_1 and τ_2 are time constants, with $\tau_2 < \tau_1$, and \bar{Q} defines the value of the quality factor, which remains nearly constant over the selected frequency band. The low-frequency limit of M is M_R , and we can identify this modulus with the elastic modulus. Alternatively, we may consider

$$M(\omega) = M_U \left[1 + \frac{2}{\pi \bar{Q}} \ln \left(\frac{\tau_2^{-1} + i\omega}{\tau_1^{-1} + i\omega} \right) \right]^{-1}, \qquad (2.210)$$

whose high-frequency limit is the elastic modulus M_U . These functions give a nearly constant quality factor in the low-loss approximation. Figure 2.15 represents the dissipation factor $Q^{-1} = \text{Im}(M)/\text{Re}(M)$ for the two functions (2.209) and (2.210) (solid and dashed lines, respectively).



Figure 2.15: Dissipation factors for the nearly constant-Q model, corresponding to the two functions (2.209) and (2.210) (solid and dashed lines, respectively). The curves correspond to $\tilde{Q} = 40$, $\tau_1 = 1.5$ s and $\tau_2 = 8 \times 10^{-5}$ s. The dotted line represents \tilde{Q}^{-1} .

2.5 Constant-Q model and wave equation

A perfect constant-Q model can be designed for all frequencies. Bland (1960), Caputo and Mainardi (1971), Kjartansson (1979), Müller (1983) and Mainardi and Tomirotti (1998) discuss a linear attenuation model with the required characteristics, but the idea is much older (Nutting, 1921; Scott Blair, 1949). The so-called Kjartansson's constant-Q model – in seismic prospecting literature – is based on a creep function of the form $t^{2\gamma}$, where t is time and $\gamma \ll 1$ for seismic applications. This model is completely specified by two parameters, i.e., phase velocity at a reference frequency and Q. Therefore, it is mathematically much simpler than any nearly constant Q, such as, for instance, a spectrum of Zener models (Carcione, Kosloff and Kosloff, 1988b,c,d). Due to its simplicity, Kjartansson's model is used in many seismic applications, mainly in its frequency-domain form.

The relaxation function is

$$\psi(t) = \frac{M_0}{\Gamma(1 - 2\gamma)} \left(\frac{t}{t_0}\right)^{-2\gamma} H(t), \qquad (2.211)$$

where M_0 is a bulk modulus, Γ is Euler's Gamma function, t_0 is a reference time and γ is a dimensionless parameter. The parameters M_0 , t_0 and γ have precise physical meanings that will become clear in the following analysis.

Using equation (2.31) and after some calculations, we get the complex modulus,

$$M(\omega) = M_0 \left(\frac{i\omega}{\omega_0}\right)^{2\gamma},$$
 (2.212)

where $\omega_0 = 1/t_0$ is the reference frequency.

2.5.1 Phase velocity and attenuation factor

The complex velocity is given by equation (2.80),

$$v_c = \sqrt{\frac{M}{\rho}}, \qquad (2.213)$$

and the phase velocity can be obtained from equation (2.83),

$$v_p = c_0 \left| \frac{\omega}{\omega_0} \right|^\gamma$$
(2.214)

with

$$c_0 = \sqrt{\frac{M_0}{\rho}} \left[\cos\left(\frac{\pi\gamma}{2}\right) \right]^{-1}. \tag{2.215}$$

The attenuation factor (2.85) is given by

$$\alpha = \tan\left(\frac{\pi\gamma}{2}\right) \operatorname{sgn}(\omega) \frac{\omega}{v_p},$$
 (2.216)

and the quality factor, according to equation (2.120), is

$$Q = \frac{1}{\tan(\pi\gamma)}.$$
 (2.217)

Firstly, we have from equation (2.214) that c_0 is the phase velocity at $\omega = \omega_0$ (the reference frequency), and that

$$M_0 = \rho c_0^2 \cos^2\left(\frac{\pi \gamma}{2}\right).$$
 (2.218)

Secondly, it follows from equation (2.217) that Q is independent of frequency, so that

$$\gamma = \frac{1}{\pi} \tan^{-1} \left(\frac{1}{Q} \right) \tag{2.219}$$

parameterizes the attenuation level. Hence, we see that Q > 0 is equivalent to $0 < \gamma < 1/2$. Moreover, $v_p \rightarrow 0$ when $\omega \rightarrow 0$, and $v_p \rightarrow \infty$ when $\omega \rightarrow \infty$. It follows that very high frequencies of the signal propagate at almost infinite velocity, and the differential equation describing the wave motion is parabolic (e.g., Prüss, 1993).

2.5.2 Wave equation in differential form. Fractional derivatives.

Let us consider propagation in the (x, z)-plane and a 2-D wave equation of the form

$$\frac{\partial^{\beta} w}{\partial t^{\beta}} = b\Delta w + f_w$$
, (2.220)

where w(x, z, t) is a field variable, β is the order of the time derivative, b is a positive parameter, Δ is the 2-D Laplacian operator

$$\Delta = \partial_1^2 + \partial_3^2, \qquad (2.221)$$

and f_w is a forcing term. Consider a plane wave

$$\exp[i(\omega t - k_1 x - k_3 z)],$$
 (2.222)

where ω is real and (k_1, k_3) is the complex wavevector. Substitution of the plane wave (2.222) in the wave equation (2.220) with $f_w = 0$ yields the dispersion relation

$$(i\omega)^{\beta} + bk^2 = 0$$
, (2.223)

where $k = \sqrt{k_1^2 + k_3^2}$ is the complex wavenumber. Equation (2.223) is the Fourier transform of equation (2.220). The properties of the Fourier transform when it acts on fractional derivatives are well established, and a rigorous treatment is available in the literature (e.g., Dattoli, Torre and Mazzacurati, 1998). Since $k^2 = \rho \omega^2/M$, a comparison of equations (2.223) and (2.212) gives

$$\beta = 2 - 2\gamma$$
, and $b = \left(\frac{M_0}{\rho}\right) \omega_0^{-2\gamma}$. (2.224)

Equation (2.220), together with (2.224), is the wave equation corresponding to Kjartansson's stress-strain relation (Kjartansson, 1979). In order to obtain realistic values of the quality factor, which correspond to wave propagation in rocks, $\gamma \ll 1$ and the time derivative in equation (2.220) has a fractional order.

Kjartansson's wave equation (2.220) is a particular version of a more general wave equation for variable material properties. The convolutional stress-strain relation (2.28) can be written in terms of fractional derivatives. In fact, it is easy to show, using equations (2.212) and (2.224), that it is equivalent to

$$\sigma = \rho b \frac{\partial^{2-\beta} \epsilon}{\partial t^{2-\beta}}. \qquad (2.225)$$
Coupled with the stress-strain relation (2.225) are the momentum equations

$$\partial_1 \sigma = \rho \partial_{tt}^2 u_1,$$
 (2.226)

$$\partial_3 \sigma = \rho \partial_{tt}^2 u_3,$$
 (2.227)

where u_1 and u_3 are the displacement components. By redefining

$$\epsilon = \partial_1 u_1 + \partial_3 u_3 \qquad (2.228)$$

as the dilatation field, differentiating and adding equations (2.226) and (2.227), and substituting equation (2.225), we obtain

$$\Delta_{\rho} \left(\rho b \frac{\partial^{2-\beta} \epsilon}{\partial t^{2-\beta}} \right) = \frac{\partial^2 \epsilon}{\partial t^2}, \qquad (2.229)$$

where

$$\Delta_{\rho} = \partial_1 \rho^{-1} \partial_1 + \partial_3 \rho^{-1} \partial_3. \qquad (2.230)$$

Multiplying by $(i\omega)^{\beta-2}$ the Fourier transform of equation (2.229), we have, after an inverse Fourier transform, the inhomogeneous wave equation

$$\frac{\partial^{\beta} \epsilon}{\partial t^{\beta}} = \Delta_{\rho} \left(\rho b \epsilon \right) + f_{\epsilon},$$
(2.231)

where we included the source term f_{ϵ} . This equation is similar to (2.220) if the medium is homogeneous.

A more general stress-strain relation is considered by Müller (1983), where the quality factor is proportional to ω^a , with $-1 \leq a \leq 1$. The cases a = -1, a = 0 and a = 1 correspond to the Maxwell, Kelvin-Voigt and constant-Q models, respectively (see equations (2.156), (2.164) and (2.217)). Müller derives the viscoelastic modulus using the Kramers-Kronig relations, obtaining closed-form expressions for the cases $a = \pm 1/n$, with n a natural number. Other stress-strain relations involving derivatives of fractional order are the Cole-Cole models (Cole and Cole, 1941; Bagley and Torvik, 1983, Caputo; 1998; Bano, 2004), which are used to describe dispersion and energy loss in dielectrics (see Section 8.3.2), anelastic media and electric networks.

Propagation in Pierre shale

Attenuation measurements in a relatively homogeneous medium (Pierre shale) were made by McDonal, Angona, Milss, Sengbush, van Nostrand and White (1958) near Limon, Colorado. They reported a constant-Q behavior with attenuation $\alpha = 0.12f$, where α is given in dB per 1000 ft and the frequency f in Hz. Conversion of units implies α (dB/1000 ft) = 8.686 α (nepers/ 1000 ft) = 2.6475 α (nepers/km). For low-loss solids, the quality factor is, according to (2.123),

$$Q = \frac{\pi f}{\alpha v_p}$$
,

with α given in nepers per unit length (Toksöz and Johnston, 1981). Since c is approximately 7000 ft/s (2133.6 m/s), the quality factor is $Q \simeq 32.5$. We consider a reference frequency $f_0 = \omega_0/(2\pi) = 250$ Hz, corresponding to the dominant frequency of the seismic

2.6 The concept of centrovelocity

source used in the experiments. Then, $\gamma = 0.0097955$, $\beta = 1.980409$, and $c_0 = \sqrt{M_0/\rho} = 2133.347$ m/s. The phase velocity (2.214) and attenuation factor (2.216) versus frequency $f = \omega/2\pi$ are shown in Figures 2.16a-b, respectively, where the open circles are the experimental points. Carcione, Cavallini, Mainardi and Hanyga (2002) solve the wave equation by using a numerical method and compute synthetic seismograms in inhomogeneous media. (The dotted and dashed lines in Figures 2.16a-b correspond to finite-difference approximations of the differential equations.) This approach finds important applications for porous media as well, since fractional derivatives appear in Biot's theory, which are related to memory effects at seismic frequencies (Gurevich and Lopatnikov, 1995; Hanyga and Seredyńska, 1999).



Figure 2.16: Phase velocity and attenuation factor versus frequency in Pierre shale (solid line). The open circles are the experimental data reported by McDonal, Angona, Milss, Sengbush, van Nostrand and White (1958).

2.6 The concept of centrovelocity

The velocity of a pulse in an absorbing and dispersive medium is a matter of controversy. The concept of velocity, which is relevant in the field of physics of materials and Earth sciences, has been actively studied under the impetus provided by the atomic theory on the one hand, and by radio and sound on the other (Eckart, 1948). In seismology, the concept of velocity is very important, because it provides the spatial location of an earthquake hypocenter and geological strata (Ben-Menahem and Singh, 1981). Similarly, groundpenetrating-radar applications are based on the interpretation of radargrams, where the travel times of the reflection events provide information about the dielectric permittivity and ionic conductivity of the shallow geological layers (Daniels, 1996; Carcione, 1996c).

The three velocities, strictly defined for a plane harmonic wave, are the phase velocity (2.83), the group velocity (2.86) and the energy velocity (2.114). As we have seen in Section 2.3, the latter is equal to the phase velocity in 1-D media. Sommerfeld and Brillouin (Brillouin, 1960) clearly show the breakdown of the group-velocity concept, which may exceed the velocity of light in vacuum and even become negative. They introduced the concept of signal velocity, which has been analyzed in detail for the Lorentz model. For non-periodic (non-harmonic) waves with finite energy, the concept of centrovelocity has been introduced (Vainshtein, 1957; Smith, 1970; Gurwich, 2001). Smith (1970) defines the centrovelocity as the distance travelled divided by the centroid of the time pulse. van Groesen and Mainardi (1989), Derks and van Groesen (1992) and Gurwich (2001) define the centrovelocity as the velocity of the "mass" center, where the integration is done over the spatial variable instead of the time variable. That is, on the "snapshot" of the wave field instead of the pulse time history. Unlike the phase (energy) and group velocities, the centrovelocity depends on the shape of the pulse, which changes as a function of time and travel distance. Therefore, an explicit analytical expression in terms of the medium properties cannot be obtained.

In order to investigate the concept of wave velocity in the presence of attenuation, we consider a 1-D medium and compare the energy (phase) and group velocities of a harmonic wave to the velocity obtained as the distance divided by the travel time of the centroid of the energy, where by energy we mean the square of the absolute value of the pulse time history. This concept is similar to the centrovelocity introduced by Smith (1970), in the sense that it is obtained in the time domain. Smith's definition is an instantaneous centrovelocity, as well as Gurwich's velocity (Gurwich, 2001), which is defined in the space domain. The travel times corresponding to the "theoretical" energy and group velocities are evaluated by taking into account that the pulse dominant frequency decreases with increasing travel distance. Thus, the dominant frequency depends on the spatial variable and is obtained as the centroid of the power spectrum. A similar procedure is performed in the spatial domain by computing a centroid wavenumber.

2.6.1 1-D Green's function and transient solution

The 1-D Green's function (impulse response) of the medium is

$$G(\omega) = \exp(-ikx) \qquad (2.232)$$

(e.g., Eckart, 1948; Pilant, 1979, p. 52), where k is the complex wavenumber and x is the travel distance. We consider that the time history of the source is

$$f(t) = \exp\left[-\frac{\Delta\omega^2 (t-t_0)^2}{4}\right] \cos[\bar{\omega}(t-t_0)], \qquad (2.233)$$

whose frequency spectrum is

$$F(\omega) = \frac{\sqrt{\pi}}{\Delta\omega} \left\{ \exp\left[-\left(\frac{\omega + \bar{\omega}}{\Delta\omega}\right)^2 \right] + \exp\left[-\left(\frac{\omega - \bar{\omega}}{\Delta\omega}\right)^2 \right] \right\} \exp(-i\omega t_0), \quad (2.234)$$

where t_0 is a delay, $\bar{\omega}$ is the central angular frequency, and $2\Delta\omega$ is the width of the pulse, such that $F(\bar{\omega} \pm \Delta\omega) = F(\bar{\omega})/e$. $(\Delta\omega = \bar{\omega}/2$ in the example below).

Then, the frequency-domain response is

$$U(\omega) = F(\omega)G(\omega) = F(\omega) \exp(-ikx)$$
 (2.235)

and its power spectrum is

$$P(\omega) = |U(\omega)|^2 = \frac{\pi}{\Delta\omega^2} \left\{ \exp\left[-\left(\frac{\omega + \bar{\omega}}{\Delta\omega}\right)^2 \right] + \exp\left[-\left(\frac{\omega - \bar{\omega}}{\Delta\omega}\right)^2 \right] \right\}^2 \exp(-2\alpha x),$$
(2.236)

where we have used equations (2.80) and (2.81), and α is given by equation (2.85). A numerical inversion by the discrete Fourier transform yields the desired time-domain (transient) solution.

2.6.2 Numerical evaluation of the velocities

In this section, we obtain expressions of the energy and group velocities and two different centrovelocities.

The energy of a signal is defined as

$$E = \int_{0}^{\infty} |u(t)|^{2} dt = \frac{1}{2\pi} \int_{-\infty}^{\infty} |U(\omega)|^{2} d\omega, \qquad (2.237)$$

where u(t) is the Fourier transform of $U(\omega)$, and Parseval's theorem has been used (Bracewell, 1965, p. 112).

We define "location of energy" as the time t_c corresponding to the centroid of the function $|u|^2$ in the time domain (time history) (Bracewell, 1965, p. 139). That is

$$t_c(x) = \frac{\int_0^\infty t |u(x,t)|^2 dt}{\int_0^\infty |u(x,t)|^2 dt}.$$
(2.238)

Then, the first centrovelocity, defined here as the mean velocity from 0 to x, is

$$\bar{c}_1(x) = \frac{x}{t_e(x)}$$
(2.239)

Smith's centrovelocity is

$$c_1(x) = \left(\frac{dt_e(x)}{dx}\right)^{-1} \tag{2.240}$$

(Smith, 1970).

The group and energy velocities (2.86) and (2.114) are evaluated at the centroid ω_c of the power spectrum. Since the medium is lossy, frequency ω_c depends on the position x', where $0 \le x' \le x$. We have

$$\omega_c(x') = \frac{\int_0^\infty \omega P(\omega, x') d\omega}{\int_0^\infty P(\omega, x') d\omega} = \frac{\int_0^\infty \omega |F|^2 \exp(-2\alpha x') d\omega}{\int_0^\infty |F|^2 \exp(-2\alpha x') d\omega},$$
(2.241)

where we have used equations (2.235) and (2.236).

The energy and group travel times are then obtained as

$$t_e(x) = \int_0^x \frac{dx'}{v_e[\omega_c(x')]} \quad \text{and} \quad t_g(x) = \int_0^x \frac{dx'}{v_g[\omega_c(x')]},$$
(2.242)

and the respective mean velocities are

$$\bar{v}_e(x) = \frac{x}{t_e(x)}$$
 and $\bar{v}_g(x) = \frac{x}{t_g(x)}$. (2.243)

We define a second centrovelocity as the mean velocity computed from the snapshots of the field, from 0 to time t,

$$\bar{c}_2(t) = \frac{x_c(t)}{t}$$
, (2.244)

where the "location of energy" is

$$x_e(t) = \frac{\int_0^{\infty} x |u(x, t)|^2 dx}{\int_0^{\infty} |u(x, t)|^2 dx},$$
 (2.245)

i.e., the centroid of the function $|u|^2$ in the space domain (snapshot). Gurwich's centrovelocity is

$$c_2(t) = \frac{dx_c(t)}{dt}$$
 (2.246)

(Gurwich, 2001). In this case, it is possible to compute the energy and group velocities if we assume a complex frequency $\Omega = \omega + i\omega_I$ and a real wavenumber, as in Section 2.3.1. The dispersion relation is given by equation (2.128). Generally, this equation has to be solved numerically for Ω to obtain $\omega(\kappa) = \text{Re}(\Omega)$. Then, the energy and group velocities are evaluated at the centroid κ_e of the spatial power spectrum. As before, the centroid wavenumber κ_e depends on the snapshot time t', where $0 \le t' \le t$. We have

$$\kappa_c(t') = \frac{\int_0^{\infty} \kappa P(\kappa, t') d\kappa}{\int_0^{\infty} P(\kappa, t') d\kappa}, \qquad (2.247)$$

where $P(\kappa, t')$ is the spatial power spectrum obtained by an inverse spatial Fourier transform. The phase, energy and group locations are then obtained as

$$x_p(t) = \int_0^t \frac{dt'}{v_p[\omega(\kappa_c(t'))]}, \quad x_e(t) = \int_0^t \frac{dt'}{v_e[\omega(\kappa_c(t'))]} \quad \text{and} \quad x_g(t) = \int_0^t \frac{dt'}{v_g[\omega(\kappa_c(t'))]},$$
(2.248)

where

$$v_p[\omega(\kappa)] = \frac{\omega(\kappa)}{\kappa} = \operatorname{Re}(v_e), \quad v_e[\omega(\kappa)] = \left[\operatorname{Re}\left(\frac{1}{v_e}\right)\right]^{-1} \text{ and } v_g[\omega(\kappa)] = \operatorname{Re}\left[\frac{\partial\Omega(\kappa)}{\partial\kappa}\right].$$

(2.249)

An energy velocity that differs from the phase velocity arises from the energy balance (see Section 2.3.1). The respective mean velocities are

$$\bar{v}_p(t) = \frac{x_p(t)}{t}, \quad \bar{v}_e(t) = \frac{x_e(t)}{t} \quad \text{and} \quad \bar{v}_g(t) = \frac{x_g(t)}{t}.$$
 (2.250)

In the next section, we consider an example of the first centrovelocity concept (equation (2.239)).

2.6.3 Example

We consider a Zener model whose complex modulus is given by equation (2.170), and we use equations (2.200) and (2.202). We assume $\omega_0 = 1/\tau_0 = 157/s$ and $M_U = \rho c_U^2$, with $c_U = 2 \text{ km/s}$. (The value of the density is irrelevant for the calculations.)

Figure 2.17 shows the energy and group velocities as a function of frequency (a), the initial spectrum (dashed line) and the spectrum at x = 50 m (solid line) (b), and the absolute value of the pulse in a lossless medium (dashed line) and for $Q_0 = 5$ (solid line) (c) (the travel distance is x = 1 km). The group velocity is greater than the energy (phase) velocity, mainly at the location of the relaxation peak. The amplitude of the spectrum for $Q_0 = 5$ is much lower than that of the initial spectrum, and the dominant frequency has decreased. From (c), we may roughly estimate the pulse velocity by taking the ratio travel distance (1 km) to arrival time of the maximum amplitude. It gives 2 km/s (1 km/0.5 s) for $Q_0 = \infty$ (dashed-line pulse) and 1.67 km/s (1 km/0.6 s) for $Q_0 =$ 5. More precise values are obtained by using the centrovelocity.



Figure 2.17: (a) Energy (solid line) and group (dashed line) velocities as a function of frequency for Q_0 = 5. (b) Initial spectrum (dashed line) and spectrum for x = 50 m (solid line). (c) Absolute value of the normalized displacement in a lossless medium (dashed line) and pulse for $Q_0 = 5$ (the travel distance is x = 1 km). The relation between the pulse maximum amplitudes is 194. The relaxation mechanism has a peak at $\omega_0 = 157/s$ ($f_0 = \omega_0/2\pi = 25$ Hz) and the source (initial) dominant frequency is $\bar{\omega} = 628/s$ ($\bar{f} = \bar{\omega}/2\pi = 50$ Hz).

The comparison between the energy and group velocities (see equation (2.243)) to the centrovelocity \bar{c}_1 (equation (2.239)) is shown in Figure 2.18. In this case $Q_0 = 10$. The relaxation mechanism has a peak at $f_0 = 25$ Hz and (a) and (b) correspond to source (initial) dominant frequencies of 50 Hz and 25 Hz, respectively. As can be seen, the centrovelocity is closer to the group velocity at short travel distances, where the wave packet keeps its shape. At a given distance, the centrovelocity equals the energy velocity and beyond that distance this velocity becomes a better approximation, particularly when the initial source central frequency is close to the peak frequency of the relaxation mechanism (case (b)). The problem is further discussed in Section 4.4.5.



Figure 2.18: Centrovelocity (solid line), and energy (dashed line) and group (dotted line) velocities as a function of travel distance and $Q_0 = 10$. The relaxation mechanism has a peak at $f_0 = 25$ Hz and the source (initial) dominant frequency is $\bar{f} = 50$ Hz (a) and $\bar{f} = 25$ Hz (b).

2.7 Memory variables and equation of motion

It is convenient to recast the equation of motion for a viscoelastic medium in the particlevelocity/stress formulation. This allows the numerical calculation of wave fields without the explicit differentiation of the material properties, and the implementation of boundary conditions, such as free-surface boundary conditions. Moreover, the equation of motion is more efficiently solved in the time domain, since frequency-domain methods are expensive because they involve the solutions of many Helmholtz equations.

In order to avoid the calculation of convolutional integrals, which can be computationally expensive, the time-domain formulation requires the introduction of additional field variables. Applying the Boltzmann operation (2.6) to the stress-strain relation (2.28), we have

$$\sigma = \partial_t \psi * \epsilon = \psi \odot \epsilon = \psi(0^+)(\epsilon + \varphi * \epsilon),$$
 (2.251)

where φ is the response function, defined as

$$\varphi = \tilde{\varphi}H, \quad \tilde{\varphi} = \frac{\partial_t \tilde{\psi}}{\psi(0^+)}.$$
 (2.252)

2.7.1 Maxwell model

For the Maxwell model (see equation (2.149)),

$$\bar{\psi} = M_U \exp(-t/\tau)$$
 (2.253)

and

$$\tilde{\phi} = -\frac{1}{\tau} \exp(-t/\tau).$$
 (2.254)

Equation (2.251) yields

$$\sigma = M_U(\epsilon + e), \qquad (2.255)$$

2.7 Memory variables and equation of motion

where

$$e = \varphi * \epsilon$$
 (2.256)

is the strain memory variable. (The corresponding stress memory variable can be defined as M_Ue – the term memory variable to describe hidden field variables in viscoelasticity being introduced by Carcione, Kosloff and Kosloff (1988b,c,d)). Note that the response function obeys the following first-order equation

$$\partial_t \tilde{\varphi} = -\frac{1}{\tau} \tilde{\varphi}.$$
 (2.257)

If we apply the Boltzmann operation to equation (2.256), we obtain a first-order differential equation in the time variable,

$$\partial_t e = \varphi(0)\epsilon + (\partial_t \tilde{\varphi} H) * \epsilon = \varphi(0)\epsilon - \frac{1}{\tau}\varphi * \epsilon, \qquad (2.258)$$

or,

$$\partial_t e = -\frac{1}{\tau}(\epsilon + e) = -\frac{\sigma}{\tau M_U}.$$
 (2.259)

The equation of motion (2.78), including a body-force term f_u , can be rewritten as

$$\partial_t v = \frac{1}{\rho} \partial_1 \sigma + f_u,$$
 (2.260)

where we used $\partial_t u = v$. Differentiating (2.255) with respect to the time variable and using $\epsilon = \partial_1 u$, we obtain

$$\partial_t \sigma = M_U(\partial_1 v + e_1),$$
 (2.261)

where $e_1 = \partial_t e$ obeys equation (2.259), that is

$$\partial_t e_1 = -\frac{1}{\tau} (\partial_1 v + e_1).$$
 (2.262)

Equations (2.260), (2.261) and (2.262) can be recast as a first-order matrix differential equation of the form

$$\partial_t \mathbf{y} = \mathbf{H} \cdot \mathbf{y} + \mathbf{f},$$
 (2.263)

where

$$\underline{v} = (v, \sigma, e_1)^{\top}$$
(2.264)

is the unknown field 3 × 1 array,

$$\mathbf{f} = (f_u, 0, 0)^\top$$
(2.265)

is the source 3×1 array, and

$$\mathbf{H} = \begin{pmatrix} 0 & \rho^{-1}\partial_1 & 0 \\ M_U\partial_1 & 0 & M_U \\ -\tau^{-1}\partial_1 & 0 & -\tau^{-1} \end{pmatrix}.$$
 (2.266)

In this case, the memory variable can be avoided if we consider equations (2.259) and (2.260), and the stress-strain relation (2.261):

$$\mathbf{y} = (v, \sigma)^{\top}$$
, (2.267)

$$\mathbf{f} = (f_u, 0)^\top$$
 (2.268)

as well as

$$\mathbf{H} = \begin{pmatrix} 0 & \rho^{-1}\partial_1 \\ M_U\partial_1 & -\tau^{-1} \end{pmatrix}. \qquad (2.269)$$

2.7.2 Kelvin-Voigt model

In the Kelvin-Voigt model, the strain ϵ plays the role of a memory variable, since the strain-stress relation (2.39) and the creep function (2.163) yield

$$\epsilon = \varphi_{\sigma} * \sigma$$
, (2.270)

(note that $\chi(0^+) = 0$), with

$$\phi_{\sigma} = \frac{1}{\tau M_R} \exp(-t/\tau).$$
 (2.271)

Then, as with the Maxwell model, there is no need to introduce an additional field variable. To express the equation as a first-order differential equation in time, we recast equation (2.260) as

$$\partial_t \partial_1 v = \partial_1 \rho^{-1} \partial_1 \sigma + \partial_1 f_u = \Delta_\rho \sigma + \partial_1 f_u, \qquad \Delta_\rho = \partial_1 \rho^{-1} \partial_1, \qquad (2.272)$$

and redefine

$$\epsilon_1 = \partial_t \epsilon.$$
 (2.273)

Noting that $\partial_1 v = \epsilon_1$, and using the stress-strain relation (2.160), we obtain

$$\partial_t \epsilon_1 = \Delta_\rho (M_R \epsilon + \eta \epsilon_1) + \partial_1 f_u.$$
 (2.274)

The matrix form (2.263) is obtained for

$$\mathbf{y} = (\epsilon_1, \epsilon)^\top, \tag{2.275}$$

$$\mathbf{f} = (\partial_1 f_u, 0)^\top$$

(2.276)

and

$$\mathbf{H} = \begin{pmatrix} \Delta_{\rho} \eta & \Delta_{\rho} M_R \\ 1 & 0 \end{pmatrix}. \tag{2.277}$$

Another approach is the particle-velocity/stress formulation. Using $v = \partial_t u$, the time derivative of the stress-strain relation (2.160) becomes

$$\partial_t \sigma = M_R \partial_1 v + \eta \partial_1 \partial_t v.$$
 (2.278)

Substituting (2.260) into (2.278) yields

$$\partial_t \sigma = M_R \partial_1 v + \eta \partial_1 \left(\frac{1}{\rho} \partial_1 \sigma + f_u \right).$$
 (2.279)

This equation and (2.260) can be recast in the matrix form (2.263), where

$$\mathbf{y} = (v, \sigma)^{\top}$$
, (2.280)

$$\mathbf{f} = (0, \eta \partial_1 f_u)^\top \qquad (2.281)$$

and

$$\mathbf{H} = \begin{pmatrix} 0 & \rho^{-1}\partial_1 \\ M_R \partial_1 & \eta \partial_1 \eta^{-1} \partial_1 \end{pmatrix}. \quad (2.282)$$

Carcione, Poletto and Gei (2004) generalize this approach to the 3-D case and develop a numerical algorithm to solve the differential equation for isotropic inhomogeneous media, including free-surface boundary conditions. The modeling simulates 3-D waves by using the Fourier and Chebyshev methods to compute the spatial derivatives along the horizontal and vertical directions, respectively (see Chapter 9). The formulation, based on one Kelvin-Voigt element, models a linear quality factor as a function of frequency.

2.7.3 Zener model

The stress-strain relation (2.251) is based on the relaxation function (2.173) and, after application of the Boltzmann operation (2.6), becomes

$$\sigma = M_U(e + e),$$
 (2.283)

where M_U is given by equation (2.171),

$$e = \phi * \epsilon$$
, (2.284)

is the strain memory variable, and

$$\tilde{\varphi} = \frac{1}{\tau_{\epsilon}} \left(1 - \frac{\tau_{\epsilon}}{\tau_{\sigma}} \right) \exp(-t/\tau_{\sigma}).$$
 (2.285)

Equation (2.285) obeys a differential equation of the form (2.257). The memory variable satisfies

$$\partial_t e = \varphi(0)\epsilon - \frac{e}{\tau_\sigma} = -\frac{1}{\tau_\sigma} \left[\left(1 - \frac{\tau_\sigma}{\tau_\epsilon} \right) \epsilon + e \right].$$
 (2.286)

First-order differential equations of the form (2.286) were introduced by Day and Minster (1984) to simulate wave propagation in anelastic media. Defining $e_1 = \partial_t e$ and differentiating (2.283) and (2.286) with respect to the time variable, we obtain

$$\partial_t \sigma = M_U(\partial_1 v + e_1)$$
 (2.287)

and

$$\partial_t e_1 = -\frac{1}{\tau_\sigma} \left[\left(1 - \frac{\tau_\sigma}{\tau_\epsilon} \right) \partial_1 v + e_1 \right]. \tag{2.288}$$

These equations and the equation of motion (2.260) can be written in the matrix form (2.263), with $\underline{\mathbf{y}}$ and $\underline{\mathbf{f}}$ given by equations (2.264) and (2.265), and

$$\mathbf{H} = \begin{pmatrix} 0 & \rho^{-1}\partial_1 & 0 \\ M_U\partial_1 & 0 & M_U \\ \varphi(0)\partial_1 & 0 & -\tau_{\sigma}^{-1} \end{pmatrix}. \quad (2.289)$$

2.7.4 Generalized Zener model

In this case, the stress-strain relation (2.251) (see (2.198)) is expressed in terms of L memory variables e_l ,

$$\sigma = M_U \left(\epsilon + \sum_{l=1}^{L} e_l \right), \qquad (2.290)$$

which, after defining $e_{1l} = \partial_l e_l$ and differentiating with respect to the time variable, becomes

$$\partial_t \sigma = M_U \left(\partial_1 v + \sum_{l=1}^L e_{1l} \right). \tag{2.291}$$

The memory variables satisfy

$$\partial_t e_{il} = \varphi_l(0)\partial_1 v - \frac{e_{il}}{\tau_{\sigma l}},$$
 (2.292)

with

$$\tilde{\varphi}_{l} = \frac{1}{\tau_{\sigma l}} \left(\sum_{l=1}^{L} \frac{\tau_{el}}{\tau_{\sigma l}} \right)^{-1} \left(1 - \frac{\tau_{el}}{\tau_{\sigma l}} \right) \exp(-t/\tau_{\sigma l}).$$
(2.293)

The matrix differential equation (2.263) has

$$\underline{\mathbf{v}} = (v, \sigma, e_{11}, e_{12}, \dots, e_{1L})^{\top},$$
 (2.294)

$$\mathbf{f} = (f_u, 0, 0, 0, \dots, 0)^{\top}$$
 (2.295)

and

$$\mathbf{H} = \begin{pmatrix} 0 & \rho^{-1}\partial_1 & 0 & 0 & \dots & 0 \\ M_U \partial_1 & 0 & M_U & M_U & \dots & M_U \\ \varphi_1(0)\partial_1 & 0 & -\tau_{\sigma_1}^{-1} & 0 & \dots & 0 \\ \varphi_2(0)\partial_1 & 0 & 0 & -\tau_{\sigma_2}^{-1} & \dots & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots & \vdots \\ \vdots & \vdots & \vdots & \vdots & \ddots & \vdots \\ \varphi_L(0)\partial_1 & 0 & 0 & 0 & \dots & -\tau_{\sigma_L}^{-1} \end{pmatrix}. \quad (2.296)$$

This formulation for the generalized Zener model is appropriate to simulate wave propagation in inhomogeneous viscoelastic media, with a general dependence of the quality factor as a function of frequency.

Alternatively, we can solve for the dilatation field (the strain ϵ in 1-D space) or the pressure field ($-\sigma$ in 1-D space). These formulations for the viscoacoustic equation of motion are convenient for 3-D problems where memory storage is demanding. We substitute the stress-strain relation (2.290) into equation (2.272) to obtain

$$\partial_t \epsilon_1 = \Delta_\rho \left[M_U \left(\epsilon + \sum_{l=1}^L e_l \right) \right] + \partial_1 f_u,$$
 (2.297)

where $\epsilon_1 = \partial_t \epsilon$. Then, the unknown field in equation (2.263) is

$$\mathbf{y} = (\epsilon, \epsilon_1, e_1, e_2, \dots, e_L)^\top, \tag{2.298}$$

the force term is

$$\mathbf{f} = (0, \partial_1 f_u, 0, 0, \dots, 0)^\top$$
(2.299)

and

$$\mathbf{H} = \begin{pmatrix} 0 & 1 & 0 & 0 & \dots & 0 \\ \Delta_{\rho}M_U & 0 & \Delta_{\rho}M_U & \Delta_{\rho}M_U & \dots & \Delta_{\rho}M_U \\ \varphi_1(0) & 0 & -\tau_{\sigma 1}^{-1} & 0 & \dots & 0 \\ \varphi_2(0) & 0 & 0 & -\tau_{\sigma 2}^{-1} & \dots & 0 \\ \vdots & \vdots & \vdots & \ddots & \vdots & \vdots \\ \vdots & \vdots & \vdots & \vdots & \ddots & \vdots \\ \varphi_L(0) & 0 & 0 & 0 & \dots & -\tau_{\sigma L}^{-1} \end{pmatrix}, \quad (2.300)$$

where $\Delta_{\rho} = \partial_i \rho^{-1} \partial_i$ in 3-D space (Carcione, Kosloff and Kosloff, 1988d).

96

Chapter 3

Isotropic anelastic media

When the velocity of transmission of a wave in the second medium, is greater than that in the first, we may, by sufficiently increasing the angle of incidence in the first medium, cause the refracted wave in the second to disappear [critical angle]. In this case, the change in the intensity of the reflected wave is here shown to be such that, at the moment the refracted wave disappears, the intensity of the reflected [wave] becomes exactly equal to that of the incident wave. If we moreover suppose the vibrations of the incident wave to follow a law similar to that of the cycloidal pendulum, as is usual in the Theory of Light, it is proved that on farther increasing the angle of incidence, the intensity of the reflected wave remains unaltered whilst the phase of the vibration gradually changes. The laws of the change of intensity, and of the subsequent alteration of phase, are here given for all media, elastic or non-elastic. When, however, both the media are elastic, it is remarkable that these laws are precisely the same as those for light polarized in a plane perpendicular to the plane of incidence.

George Green (Green, 1838)

The properties of viscoelastic plane waves in two and three dimensions are essentially described in terms of the wavevector bivector. This can be written in terms of its real and imaginary parts, representing the real wavenumber vector, and the attenuation vector, respectively. When these vectors coincide in direction, the plane wave is termed homogeneous; when these vectors differ in direction, the plane wave is termed an inhomogeneous body wave. Inhomogeneity has several consequences that make viscoelastic wave behavior particularly different from elastic wave behavior. These behaviors differ mainly in the presence of both inhomogeneities and anisotropy, as we shall see in Chapter 4.

In the geophysical literature, the main contributors to the understanding of wave propagation in isotropic viscoelastic media are Buchen (1971a,b), Borcherdt (1973, 1977, 1982), Borcherdt, Glassmoyer and Wennerberg (1986), and Krebes (1983a,b). The thermodynamical and wave-propagation aspects of the theory are briefly reviewed by Minster (1980) and Chin (1980), respectively. Bland (1960), Beltzer (1988), Christensen (1982), Pipkin (1972), Leitman and Fisher (1984), Caviglia and Morro (1992) and Fabrizio and Morro (1992) provide a rigorous treatment of the subject. In this chapter, we follow the "geophysical" approach to develop the main aspects of the theory of viscoelasticity.

3.1 Stress-strain relation

Let us denote the dimension of the space by n and consider n = 2 and n = 3 in the following. By n = 2 we strictly mean a two-dimensional world and not a plane-strain problem in 3-D space. Therefore, most of the equations lose their tensorial character and should be considered with caution.

The most general isotropic representation of the fourth-order relaxation tensor (2.10)in *n*-dimensional space is

$$\psi_{ijkl}(t) = \left[\psi_{\mathcal{K}}(t) - \frac{2}{n}\psi_{\mu}(t)\right]\delta_{ij}\delta_{kl} + \psi_{\mu}(t)(\delta_{ik}\delta_{jl} + \delta_{il}\delta_{jk}),$$
 (3.1)

where ψ_{κ} and ψ_{μ} are independent relaxation functions. Substitution of equation (3.1) into the stress-strain relations (2.9) gives

$$\sigma_{ij} = \left(\psi_{\kappa} - \frac{2}{n}\psi_{\mu}\right) * \partial_t \epsilon_{kk}\delta_{ij} + 2\psi_{\mu} * \partial_t \epsilon_{ij}. \qquad (3.2)$$

Taking the trace on both sides of this equation yields

$$\sigma_{ii} = n\psi_{k} * \partial_{t}\epsilon_{ii}$$
(3.3)

On the other hand, computing the deviatoric components of stress and strain gives

$$s_{ij} = 2\psi_{\mu} * \partial_t d_{ij}$$
, (3.4)

where

$$s_{ij} = \sigma_{ij} - \frac{1}{n} \sigma_{kk} \delta_{ij} \qquad (3.5)$$

and

$$d_{ij} = \epsilon_{ij} - \frac{1}{n} \vartheta \delta_{ij} \qquad (3.6)$$

are the components of the deviatoric strain. It is clear that $\psi_{\mathcal{K}}$ describes dilatational deformations, and ψ_{μ} describes shear deformations; $\psi_{\mathcal{K}}$ is the generalization of the bulk compressibility in the lossless case, and $\psi_{\mathcal{K}} - 2\psi_{\mu}/n$ and ψ_{μ} play the role of the Lamé constants λ and μ .

3.2 Equations of motion and dispersion relations

The analysis of wave propagation in homogeneous isotropic media is simplified by the fact that the wave modes are not coupled, as they are in anisotropic media. Applying the divergence operation to equation (1.23), and assuming constant material properties and $f_i = 0$, we obtain

$$\partial_i \partial_j \sigma_{ij} = \rho \partial_{tt}^2 \partial_i$$
(3.7)

where

$$\vartheta = \partial_i u_i = \epsilon_{ii} = \text{div } \mathbf{u}$$
(3.8)

is the dilatation field defined in equation (1.11). Using (3.2), we can write the left-hand side of (3.7) as

$$\partial_t \left(\psi_{\mathcal{K}} - \frac{2}{n}\psi_{\mu}\right) * \partial_i\partial_i\vartheta + 2\partial_t\psi_{\mu} * \partial_i\partial_j\epsilon_{ij}.$$
 (3.9)

Because $2\partial_i\partial_j\epsilon_{ij} = \partial_i\partial_j\partial_ju_i + \partial_i\partial_j\partial_iu_j = 2\partial_i\partial_j\partial_ju_i = 2\partial_i\partial_i\partial_-$ with the use of (1.2) and (3.8) – we obtain for equation (3.7),

$$\partial_t \left[\psi_{\mathcal{K}} + 2\psi_{\mu} \left(1 - \frac{1}{n} \right) \right] * \partial_t \partial_i \vartheta = \rho \partial_{tt}^2 \vartheta,$$
 (3.10)

or

$$\partial_t \psi_{\varepsilon} * \Delta \vartheta = \rho \partial_{tt}^2 \vartheta,$$
 (3.11)

where $\Delta = \partial_i \partial_i$ is the Laplacian, and

$$\psi_{\mathcal{E}}(t) = \psi_{\mathcal{K}}(t) + 2\psi_{\mu}(t)\left(1 - \frac{1}{n}\right)$$

(3.12)

is the P-wave relaxation function that plays the role of $\lambda + 2\mu$.

Applying the curl operator to equation (1.23), and assuming constant material properties and $f_i = 0$, we obtain

$$\epsilon_{lik}\partial_l\partial_j\sigma_{ij}\hat{\mathbf{e}}_k = \rho\partial_{ti}^2\Omega,$$
 (3.13)

where

$$\Omega = \epsilon_{lik} \partial_l u_i \hat{\mathbf{e}}_k = \text{curl } \mathbf{u}$$
(3.14)

and ϵ_{lik} are the components of the Levi-Civita tensor. Substitution of the stress-strain relation (3.2) gives

$$2\partial_t \psi_\mu * \epsilon_{tik} \partial_l \partial_j \epsilon_{ij} \hat{\mathbf{e}}_k = \rho \partial_{tt}^2 \Omega.$$
 (3.15)

Because $2\epsilon_{lik}\partial_l\partial_j\epsilon_{ij} \hat{\mathbf{e}}_k = \epsilon_{lik}\partial_l\partial_j(\partial_i u_j + \partial_j u_i) \hat{\mathbf{e}}_k = \epsilon_{lik}\partial_l\partial_l\vartheta\hat{\mathbf{e}}_k + \partial_j\partial_j(\epsilon_{lik}\partial_l u_i) \hat{\mathbf{e}}_k = \partial_j\partial_j\Omega$ (since $\epsilon_{lik}\partial_l\partial_l\vartheta = \text{curl }\vartheta = \text{curl }\text{grad }\mathbf{u} = 0$), we finally have

$$\partial_t \psi_\mu * \Delta \Omega = \rho \partial_{tt}^2 \Omega.$$
 (3.16)

Fourier transformation of (3.11) and (3.16) to the frequency domain gives the two Helmholtz equations

$$\Delta \vartheta + \frac{\omega^2}{v_P^2} \vartheta = 0, \quad \Delta \Omega + \frac{\omega^2}{v_S^2} \Omega = 0,$$
 (3.17)

where

$$\rho v_P^2 = \mathcal{F}[\partial_t \psi_{\mathcal{E}}] = \frac{\rho \omega^2}{k_P^2}, \quad \rho v_S^2 = \mathcal{F}[\partial_t \psi_{\mu}] = \frac{\rho \omega^2}{k_S^2}, \quad (3.18)$$

with v_P and v_S being the complex and frequency-dependent P-wave and S-wave velocities, and k_P and k_S being the corresponding complex wavenumbers.

The fact that the P- and S-wave modes satisfy equations (3.17) implies that the displacement vector admits the representation

$$\mathbf{u} = \operatorname{grad} \Phi + \operatorname{curl} \Theta$$
, div $\Theta = 0$, (3.19)

where Φ and Θ are a scalar and a vector potential, which satisfy $(3.17)_1$ and $(3.17)_2$, respectively:

$$\Delta \Phi + \frac{\omega^2}{v_P^2} \Phi = 0, \quad \Delta \Theta + \frac{\omega^2}{v_S^2} \Theta = 0.$$
 (3.20)

These equations can be easily verified by substituting the expression of the displacement into the equation of motion (1.23). The rigorous demonstration for viscoelastic media is given by Edelstein and Gurtin (1965) (cf. Caviglia and Morro, 1992, p. 42).

Let us introduce the complex moduli as in the 1-D case (see equations (2.31), (2.36) and (2.37)),

$$\mathcal{K}(\omega) = \mathcal{F}[\partial_t \psi_{\mathcal{K}}(t)] = \mathcal{K}_R(\omega) + i\mathcal{K}_I(\omega),$$

 $\mu(\omega) = \mathcal{F}[\partial_t \psi_{\mu}(t)] = \mu_R(\omega) + i\mu_I(\omega),$
(3.21)

where

$$\mathcal{K}_R(\omega) = \omega \int_0^\infty \tilde{\psi}_{\mathcal{K}}(t) \sin(\omega t) dt, \quad \mathcal{K}_I(\omega) = \omega \int_0^\infty [\tilde{\psi}_{\mathcal{K}}(t) - \tilde{\psi}_{\mathcal{K}}(0)] \cos(\omega t) dt,$$

 $\mu_R(\omega) = \omega \int_0^\infty \tilde{\psi}_{\mu}(t) \sin(\omega t) dt, \quad \mu_I(\omega) = \omega \int_0^\infty [\tilde{\psi}_{\mu}(t) - \tilde{\psi}_{\mu}(0)] \cos(\omega t) dt.$
(3.22)

Using (3.12) and (3.18), we define \mathcal{E} and μ as

$$\mathcal{E} = \mathcal{F}[\partial_t \psi_{\mathcal{E}}] = \rho v_P^2 = \mathcal{K} + 2\mu \left(1 - \frac{1}{n}\right), \quad \mu = \rho v_S^2.$$
 (3.23)

Then, the complex dispersion relations are

$$k_P^2 = \frac{\rho \omega^2}{\mathcal{E}}, \quad k_S^2 = \frac{\rho \omega^2}{\mu}.$$
 (3.24)

3.3 Vector plane waves

In general, plane waves in anelastic media have a component of attenuation along the lines of constant phase, meaning that their properties are described by two vectors – the attenuation and propagation vectors, which do not point in the same direction. We analyze in the following sections, the particle motion associated with these vector plane waves.

3.3.1 Slowness, phase velocity and attenuation factor

We consider the viscoelastic plane-wave solution

$$\Phi = \Phi_0 \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})], \qquad (3.25)$$

where

$$\mathbf{k} = \kappa - i\alpha = \kappa \hat{\kappa} - i\alpha \hat{\alpha} \qquad (3.26)$$

with κ being the real wavevector and α being the attenuation vector. They express the magnitudes of both the wavenumber κ and the attenuation factor α , and the directions of the normals to planes of constant phase and planes of constant amplitude.

Figure 3.1 represents the plane wave (3.25), with γ indicating the inhomogeneity angle. When this angle is zero, the wave is called homogeneous. We note that

$$\mathbf{k} = (\kappa - i\alpha)\hat{\kappa} \equiv k\hat{\kappa},$$
 (3.27)

100



Figure 3.1: Inhomogeneous viscoelastic plane wave. The inhomogeneity angle γ is less than 90°.

only for $\gamma = 0$, i.e., for homogeneous waves. Defining the complex wavenumber

$$k = \frac{\omega}{v_c}$$
, (3.28)

where v_c is the complex velocity defined in (3.18), the wavenumber and the attenuation factor for homogeneous waves have the simple form

$$\kappa_H = \omega \operatorname{Re}\left(\frac{1}{v_c}\right)$$
(3.29)

and

$$\alpha_H = -\omega \operatorname{Im} \left(\frac{1}{v_c}\right), \qquad (3.30)$$

as in the 1-D case (see equations (2.83) and (2.85)). Substitution of the plane-wave solution (3.25) into equation (3.20)₁, and the use of (3.18)₁ yields

$$\mathbf{k} \cdot \mathbf{k} = k_P^2 = k^2 = \text{Re}(k^2)(1 - iQ_H^{-1}),$$
 (3.31)

where

$$Q_{H} = -\frac{\text{Re}(k^{2})}{\text{Im}(k^{2})}$$
(3.32)

is the quality factor for homogeneous plane waves (Section 3.4.1). This quantity is an intrinsic property of the medium. For inhomogeneous plane waves, the quality factor also depends on the inhomogeneity angle γ , which is a characteristic of the wave field. Separating real and imaginary parts in equation (3.31), we have

$$\kappa^2 - \alpha^2 = \text{Re}(k^2),$$

 $2\kappa\alpha \cos \gamma = -\text{Im}(k^2) = \text{Re}(k^2)Q_H^{-1}.$
(3.33)

Solving for κ and α , we obtain

$$2\kappa^2 = \text{Re}(k^2) + \sqrt{[\text{Re}(k^2)]^2 + [\text{Im}(k^2)]^2 \sec^2\gamma},$$

 $2\alpha^2 = -\text{Re}(k^2) + \sqrt{[\text{Re}(k^2)]^2 + [\text{Im}(k^2)]^2 \sec^2\gamma},$
(3.34)

or,

$$2\kappa^{2} = \operatorname{Re}(k^{2})\left(1 + \sqrt{1 + Q_{H}^{-2}\operatorname{sec}^{2}\gamma}\right),$$

$$2\alpha^{2} = \operatorname{Re}(k^{2})\left(-1 + \sqrt{1 + Q_{H}^{-2}\operatorname{sec}^{2}\gamma}\right).$$
(3.35)

We first note that if $\text{Im}(k^2) = 0$ ($Q_H \rightarrow \infty$), $\alpha = 0$, and $\gamma = \pi/2$. This case corresponds to an inhomogeneous elastic wave propagating in a lossless material, generated by refraction, for instance. In a lossy material, γ must satisfy

$$0 \le \gamma < \pi/2$$
. (3.36)

We may include the case $\gamma = \pi/2$, keeping in mind that this case corresponds to the limit of a lossless medium.

The phase-velocity and slowness vectors for inhomogeneous plane waves are

$$\mathbf{v}_p = \left(\frac{\omega}{\kappa}\right) \hat{\boldsymbol{\kappa}}, \quad \mathbf{s}_R = \left(\frac{\kappa}{\omega}\right) \hat{\boldsymbol{\kappa}}, \quad (3.37)$$

and the attenuation vector α is implicitly defined in (3.26). For homogeneous plane waves, equation (3.29) implies

$$\mathbf{v}_{pH} = \left[\operatorname{Re}\left(\frac{1}{v_c}\right) \right]^{-1} \hat{\boldsymbol{\kappa}}, \quad \mathbf{s}_{RH} = \omega \operatorname{Re}\left(\frac{1}{v_c}\right) \hat{\boldsymbol{\kappa}}, \quad (3.38)$$

where v_e represents the P-wave velocity v_P defined in (3.18)₁. We can infer, from equations (3.35) and (3.37)₁, that the phase velocity and attenuation factor of an inhomogeneous plane wave tend to zero and ∞ , respectively, as γ approaches $\pi/2$, and that they are less than and greater than the corresponding quantities for homogeneous plane waves.

3.3.2 Particle motion of the P wave

Equation (3.19) implies that the P-wave displacement vector can be expressed in terms of the scalar potential (3.25) as

$$\mathbf{u} = \text{grad } \Phi = \text{Re}\{-i\Phi_0 \mathbf{k} \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})]\}.$$
 (3.39)

Using equation (3.26) and $\Phi_0 k = |\Phi_0 k| \exp[i \arg(\Phi_0 k)]$, we obtain

$$\mathbf{u} = -|\Phi_0 k| \exp(-\boldsymbol{\alpha} \cdot \mathbf{x}) \operatorname{Re} \left[i \left(\frac{v_c}{\omega} \right) \mathbf{k} \exp(i\varsigma) \right],$$
 (3.40)

where

$$\varsigma(t) = \omega t - \kappa \cdot \mathbf{x} + \arg(\Phi_0 k), \quad (3.41)$$

Equation (3.28) has been used (v_c represents the P-wave complex velocity v_P defined in equation (3.18)₁). We introduce the real vectors ξ_1 and ξ_2 , such that

$$\left(\frac{v_e}{\omega}\right)\mathbf{k} = \left(\frac{v_e}{\omega}\right)(\boldsymbol{\kappa} - i\boldsymbol{\alpha}) = \boldsymbol{\xi}_1 + i\boldsymbol{\xi}_2$$
(3.42)

3.3 Vector plane waves

where

$$\omega \boldsymbol{\xi}_1 = v_R \boldsymbol{\kappa} + v_I \boldsymbol{\alpha}, \quad \omega \boldsymbol{\xi}_2 = v_I \boldsymbol{\kappa} - v_R \boldsymbol{\alpha}, \quad (3.43)$$

and v_R and v_I denote the real and imaginary parts of v_c . Now,

$$\omega^2 \boldsymbol{\xi}_1 \cdot \boldsymbol{\xi}_2 = v_I v_R (\kappa^2 - \alpha^2) + (v_I^2 - v_R^2) \boldsymbol{\kappa} \cdot \boldsymbol{\alpha}, \qquad (3.44)$$

which, using equations (3.28) and (3.33), implies

$$\boldsymbol{\xi}_{1} \cdot \boldsymbol{\xi}_{2} = v_{I} v_{R} \operatorname{Re}\left(\frac{1}{v_{c}^{2}}\right) + \frac{1}{2}(v_{I}^{2} - v_{R}^{2}) \operatorname{Im}\left(\frac{1}{v_{c}^{2}}\right) \propto v_{I} v_{R} \operatorname{Re}(v_{c}^{2^{*}}) + \frac{1}{2} \operatorname{Im}(v_{c}^{2^{*}})(v_{I}^{2} - v_{R}^{2}) = 0.$$

(3.45)

Thus, the vectors are orthogonal,

$$\xi_1 \cdot \xi_2 = 0.$$
 (3.46)

Moreover,

$$\omega^{2}(\xi_{1}^{2} - \xi_{2}^{2}) = (\kappa^{2} - \alpha^{2})(v_{R}^{2} - v_{I}^{2}) + 4v_{R}v_{I}(\boldsymbol{\kappa} \cdot \boldsymbol{\alpha}).$$
(3.47)

Again, using equations (3.28) and (3.33), we obtain

$$\begin{aligned} \xi_1^2 - \xi_2^2 &= \operatorname{Re}\left(\frac{1}{v_e^2}\right) \left(v_R^2 - v_I^2\right) - 2v_R v_I \operatorname{Im}\left(\frac{1}{v_e^2}\right) \\ &= \frac{1}{|v_e|^4} \left[(v_R^2 - v_I^2)^2 + 4v_R^2 v_I^2 \right] = \frac{1}{|v_e|^4} (v_R^2 + v_I^2)^2 = 1; \end{aligned} (3.48)$$

that is

$$\xi_1^2 - \xi_2^2 = 1.$$
 (3.49)

Since $\xi_1 > 0$ and $\xi_2 > 0$, equation (3.49) implies $\xi_1 > \xi_2$. Substitution of equation (3.42) into equation (3.40) gives

$$\mathbf{u} = -|\Phi_0 k| \exp(-\boldsymbol{\alpha} \cdot \mathbf{x}) \operatorname{Re} \left[\mathrm{i}(\boldsymbol{\xi}_1 + \mathrm{i}\boldsymbol{\xi}_2) \exp(\mathrm{i}\boldsymbol{\varsigma})\right], \quad (3.50)$$

or

$$\mathbf{u} = U_0[\boldsymbol{\xi}_1 \sin \varsigma + \boldsymbol{\xi}_2 \cos \varsigma], \quad (3.51)$$

with

$$U_0 = |\Phi_0 k| \exp(-\alpha \cdot \mathbf{x}). \qquad (3.52)$$

We write the following definition:

$$U_1 = \frac{\mathbf{u} \cdot \boldsymbol{\xi}_1}{\xi_1 U_0}, \quad U_2 = -\frac{\mathbf{u} \cdot \boldsymbol{\xi}_2}{\xi_2 U_0},$$
 (3.53)

and eliminate ς from (3.51) to obtain

$$\frac{U_1^2}{\xi_1^2} + \frac{U_2^2}{\xi_2^2} = 1. \quad (3.54)$$

Equation (3.54) indicates that the particle motion is an ellipse, with major axis ξ_1 and minor axis ξ_2 . The sense of rotation is from κ to α and the plane of motion is defined by these vectors (see Figure 3.2). The cosine of the angle between the propagation direction and the major axis of the ellipse is given by $\kappa \cdot \xi_1/(\kappa \xi_1)$.



Figure 3.2: Particle motion of an inhomogeneous P wave in an isotropic viscoelastic medium. The ellipse degenerates into a straight line for a homogeneous plane wave.

This means that the particle motion of an inhomogeneous plane P wave is not purely longitudinal. When $\gamma = 0$ (i.e., an homogeneous plane wave),

$$\omega \xi_2 = \hat{\kappa} (v_I \kappa - v_R \alpha), \qquad (3.55)$$

according to equations (3.27) and (3.43)₂. But from equations (3.29) and (3.30), we have

$$v_I \kappa - v_R \alpha = v_I \kappa_H - v_R \alpha_H = \omega \left[v_I \operatorname{Re} \left(\frac{1}{v_e} \right) + v_R \operatorname{Im} \left(\frac{1}{v_e} \right) \right] = 0.$$
 (3.56)

Hence, $\xi_2 = 0$, and the particle motion is longitudinal.

3.3.3 Particle motion of the S waves

We can define two types of S waves, depending on the location of the particle motion, with respect to the (κ, α) -plane. Let us consider a plane-wave solution for type-I S waves of the form

$$\Theta = \text{Re}\{\Theta_0 \hat{\mathbf{n}} \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})]\}, \qquad (3.57)$$

where Θ_0 is a complex constant and $\hat{\mathbf{n}}$ is a unit vector perpendicular to the $(\boldsymbol{\kappa}, \boldsymbol{\alpha})$ -plane. This is a consequence of equation $(3.19)_2$, which implies $\mathbf{k} \cdot \hat{\mathbf{n}} = 0$. Orthogonality, in this case, should be understood in the sense of complex vectors (see Caviglia and Morro, 1992, p. 8, 46), and the condition $\mathbf{k} \cdot \hat{\mathbf{n}} = 0$ does not imply that the polarization Re(\mathbf{u}) (see equation (3.60) below) is perpendicular to the real wavenumber vector $\boldsymbol{\kappa}$.

The solution for type-II S waves is obtained by considering a plane wave,

$$\Theta = \text{Re}\{(\theta_1 \hat{\mathbf{e}}_1 + \theta_3 \hat{\mathbf{e}}_3) \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})]\}, \quad (3.58)$$

3.3 Vector plane waves

with θ_1 and θ_3 being complex valued. Moreover, let us assume that the (κ, α) -plane coincides with the (x, z)-plane, implying $\partial_2 [\cdot] = 0$. From equation $(3.19)_1$, the displacement field for such a wave is given by

$$\mathbf{u} = \text{Re}\{\Gamma_0 \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})]\}\hat{\mathbf{e}}_2, \quad \Gamma_0 = (\theta_3\hat{\mathbf{e}}_1 - \theta_1\hat{\mathbf{e}}_3) \cdot (\boldsymbol{\alpha} + i\boldsymbol{\kappa}),$$
 (3.59)

and $(3.19)_2$ implies $\theta_1(\alpha + i\kappa) \cdot \hat{\mathbf{e}}_1 = -\theta_3(\alpha + i\kappa) \cdot \hat{\mathbf{e}}_3$. Equation (3.59) indicates that the particle motion is linear perpendicular to the (κ, α) -plane.



Figure 3.3: Linear particle motion of a type-II S wave (SH wave).

Figure 3.3 shows the particle motion of the type-II S wave. Type-I and type-II S waves are denoted by the symbols SV and SH in seismology (Buchen, 1971a,b; Borcherdt, 1977).

The particle motion of the type-I S wave shows similar characteristics to the P-wave particle motion. From (3.19), its displacement vector can be expressed in terms of vector Θ as

$$\mathbf{u} = \operatorname{curl} \Theta = \operatorname{Re}\{-i\Theta_0(\hat{\mathbf{n}} \times \mathbf{k}) \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})]\},$$
 (3.60)

which lies in the plane of κ and α .

For simplicity, we use the same notation as for the P wave, but the complex velocity v_c is equal here to v_S , defined in (3.18)₂. Using equation (3.26) and $\Theta_0 k =$ $|\Theta_0 k| \exp[i \arg(\Theta_0 k)]$, we get

$$\mathbf{u} = -|\Theta_0 k| \exp(-\boldsymbol{\alpha} \cdot \mathbf{x}) \operatorname{Re} \left[i \left(\frac{v_c}{\omega} \right) (\hat{\mathbf{n}} \times \mathbf{k}) \exp(i\varsigma) \right], \quad (3.61)$$

where

$$\varsigma(t) = \omega t - \kappa \cdot \mathbf{x} + \arg(\Theta_0 k), \qquad (3.62)$$

As before, $v_c \mathbf{k}/\omega$ can be decomposed into real and imaginary vectors as in equation (3.42). Let us define

$$\left(\frac{v_c}{\omega}\right) \hat{\mathbf{n}} \times \mathbf{k} = \boldsymbol{\zeta}_1 + i\boldsymbol{\zeta}_2,$$
 (3.63)

where

$$\zeta_1 = \hat{n} \times \xi_1, \quad \zeta_2 = \hat{n} \times \xi_2,$$
 (3.64)

since ξ_1 and ξ_2 , defined in equation (3.43), lie in the (κ, α) -plane. On the basis of equations (3.46), (3.49) and (3.64), these vectors have the properties

$$\zeta_1 \cdot \zeta_2 = 0, \quad \zeta_1^2 - \zeta_2^2 = 1.$$
 (3.65)

Substituting (3.63) into equation (3.61) gives

$$\mathbf{u} = U_0[\boldsymbol{\zeta}_1 \sin \varsigma + \boldsymbol{\zeta}_2 \cos \varsigma], \quad (3.66)$$

with

$$U_0 = |\Theta_0 k| \exp(-\alpha \cdot \mathbf{x}). \qquad (3.67)$$

The particle motion is an ellipse, whose major and minor axes are given by ζ_1 and ζ_2 , and whose direction of rotation is from κ to α . The cosine of the angle between the propagation direction and the major axis of the ellipse is given by $\kappa \cdot \zeta_1/(\kappa \zeta_1)$ (Buchen, 1971a).



Figure 3.4 shows a diagram of the S-wave particle motion. For a homogeneous plane wave, the ellipse degenerates into a straight line.

3.3.4 Polarization and orthogonality

We have seen in Section 1.3.3 that the polarizations of the three wave modes are orthogonal in anisotropic elastic media. Now, consider the case of anelastic isotropic media. From equations (3.40) and (3.61), the P and S-I polarizations have the form

$$\mathbf{u}_{\mathrm{P}} = \operatorname{Re}[a(\kappa_{\mathrm{P}} - i\alpha_{\mathrm{P}})]$$
 and $\mathbf{u}_{\mathrm{S}} = \operatorname{Re}(b\hat{\mathbf{n}} \times (\kappa_{\mathrm{S}} - i\alpha_{\mathrm{S}})]$, (3.68)



respectively, where a and b are complex quantities, and the indices P and S indicate that κ and α for P and S waves differ. The general form (3.68) implies that the P and S polarizations are not orthogonal in general, that is, when the plane waves are inhomogeneous. For homogeneous waves, $\hat{\alpha} = \hat{\kappa}$, and if the propagation directions coincide, equation (3.68) simplifies to

$$\mathbf{u}_{P} = \operatorname{Re}(a') \hat{\kappa}$$
 and $\mathbf{u}_{S} = \operatorname{Re}(b') \hat{\mathbf{n}} \times \hat{\kappa}$, (3.69)

where a' and b' are complex quantities. These two vectors are orthogonal, since \hat{n} is perpendicular to $\hat{\kappa}$.

3.4 Energy balance, energy velocity and quality factor

To derive the mechanical energy-balance equation, we follow the same steps as we did to obtain equation (2.95) in the 1-D case. Using $\partial_t u_i = v_i$ ($\partial_t \mathbf{u} = \mathbf{v}$) and performing the scalar product of equation (1.23) with \mathbf{v} on both sides, we get

$$v_i \partial_j \sigma_{ij} = \rho v_i \partial_t v_i$$
, (3.70)

where we assumed $f_i = 0$. Contraction of $\partial_j v_i + \partial_i v_j = 2\partial_t \epsilon_{ij}$ with σ_{ij} yields

$$\sigma_{ij}\partial_t \epsilon_{ij} = \frac{1}{2}\sigma_{ij}(\partial_j v_i + \partial_i v_j) = \sigma_{ij}\partial_j v_i, \qquad (3.71)$$

using the symmetry of the stress tensor. Adding equations (3.70) and (3.71) and substituting the stress-strain relation (3.2), we obtain the energy-balance equation, equivalent to (2.94),

$$-\partial_t p_i = \partial_t T + \left[\left(\partial_t \psi_{\mathcal{K}} - \frac{2}{n} \partial_t \psi_{\mu} \right) * \epsilon_{kk} \delta_{ij} + 2 \partial_t \psi_{\mu} * \epsilon_{ij} \right] \partial_t \epsilon_{ij}, \quad (3.72)$$

where

$$p_i = -v_j \sigma_{ij}$$
 (3.73)

are the components of the Umov-Poynting vector, and

$$T = \frac{1}{2}\rho v_i v_i \qquad (3.74)$$

is the kinetic-energy density.

The second term in the right-hand side is then partitioned in terms of the rate of strain and dissipated energies on the basis of expressions (2.16) and (2.17). We obtain

$$-\text{div } \mathbf{p} = \partial_t (T + V) + D, \qquad (3.75)$$

where, defining $\tau' = 2t - \tau_1 - \tau_2$ and using (3.1), we have that the strain energy is

$$V(t) = \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} \left\{ \left[\breve{\psi}_{\mathcal{K}}(\tau') - \frac{2}{n} \breve{\psi}_{\mu}(\tau') \right] \delta_{ij} \delta_{kl} + \breve{\psi}_{\mu}(\tau') (\delta_{ik} \delta_{jl} + \delta_{il} \delta_{jk}) \right\}$$

Chapter 3. Isotropic anelastic media

$$\partial_{\tau_1} \epsilon_{ij}(\tau_1) \partial_{\tau_2} \epsilon_{kl}(\tau_2) d\tau_1 d\tau_2.$$
 (3.76)

Equation (3.76) becomes

$$V(t) = \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} \check{\psi}_{\mathcal{K}}(\tau') \partial_{\tau_{1}} \epsilon_{ii}(\tau_{1}) \partial_{\tau_{2}} \epsilon_{kk}(\tau_{2}) d\tau_{1} d\tau_{2}$$
$$+ \int_{-\infty}^{t} \int_{-\infty}^{t} \check{\psi}_{\mu}(\tau') \left[\partial_{\tau_{1}} \epsilon_{ij}(\tau_{1}) \partial_{\tau_{2}} \epsilon_{ij}(\tau_{2}) - \frac{1}{n} \partial_{\tau_{1}} \epsilon_{ii}(\tau_{1}) \partial_{\tau_{2}} \epsilon_{kk}(\tau_{2}) \right] d\tau_{1} d\tau_{2}, \quad (3.77)$$

where we used the symmetry of the strain tensor. In terms of the deviatoric components of strain (3.6), equation (3.77) becomes

$$V(t) = \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} \check{\psi}_{\kappa}(\tau') \partial_{\tau_1} \epsilon_{ii}(\tau_1) \partial_{\tau_2} \epsilon_{kk}(\tau_2) d\tau_1 d\tau_2$$
$$+ \int_{-\infty}^{t} \int_{-\infty}^{t} \check{\psi}_{\mu}(\tau') \partial_{\tau_1} d_{ij}(\tau_1) \partial_{\tau_2} d_{ij}(\tau_2) d\tau_1 d\tau_2.$$
(3.78)

To obtain the last equation, we must be careful with terms of the form $\partial_{\tau_1} \epsilon_{ij} \partial_{\tau_2} \epsilon_{ij}$, when $i \neq j$, since they come in pairs; e.g., $\partial_{\tau_1} \epsilon_{12}(\tau_1) \partial_{\tau_2} \epsilon_{12}(\tau_2) + \partial_{\tau_1} \epsilon_{21}(\tau_1) \partial_{\tau_2} \epsilon_{21}(\tau_2) = 2\partial_{\tau_1} \epsilon_{12}(\tau_1) \partial_{\tau_2} \epsilon_{12}(\tau_2)$.

Similarly, the rate of dissipated-energy density can be expressed as

$$\dot{D}(t) = -\int_{-\infty}^{t} \int_{-\infty}^{t} \partial \breve{\psi}_{\mathcal{K}}(\tau') \partial_{\tau_{1}} \epsilon_{ii}(\tau_{1}) \partial_{\tau_{2}} \epsilon_{kk}(\tau_{2}) d\tau_{1} d\tau_{2}$$
$$-\int_{-\infty}^{t} \int_{-\infty}^{t} 2\partial \breve{\psi}_{\mu}(\tau') \partial_{\tau_{1}} d_{ij}(\tau_{1}) \partial_{\tau_{2}} d_{ij}(\tau_{2}) d\tau_{1} d\tau_{2}, \qquad (3.79)$$

where ∂ denotes the derivative with respect to the argument.

3.4.1 P wave

In this section, we obtain the mechanical energy-balance equation for P waves. The complex displacement and particle-velocity components are from (3.39)

$$u_i = -i\Phi_0 k_i \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})] \qquad (3.80)$$

and

$$v_i = \partial_t u_i = \omega \Phi_0 k_i \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})],$$
 (3.81)

with $k = k_P$ in this case.

The time-averaged kinetic-energy density (3.74) can be easily calculated by using equation (1.105),

$$\langle T \rangle = \frac{1}{4} \rho \operatorname{Re}(v_i v_i^*) = \frac{1}{4} \rho |\mathbf{v}|^2 = \frac{1}{4} \rho \omega^2 |\Phi_0|^2 (\mathbf{k} \cdot \mathbf{k}^*) \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x})$$
$$= \frac{1}{4} \rho \omega^2 |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) (|\boldsymbol{\kappa}|^2 + |\boldsymbol{\alpha}|^2), \qquad (3.82)$$

108

where equation (3.26) was used. By virtue of (3.34), we can recast the kinetic-energy density in terms of the inhomogeneity angle γ ,

$$\langle T \rangle = \frac{1}{4} \rho \omega^2 |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \sqrt{[\operatorname{Re}(k^2)]^2 + [\operatorname{Im}(k^2)]^2 \sec^2 \gamma}.$$
 (3.83)

Let us now consider the strain-energy density (3.78). From equations (3.19) and (3.25), we have

$$\partial_t \epsilon_{ii} = \partial_i v_i = -i\omega k^2 \Phi$$
 (3.84)

and

$$\partial_t d_{ij} = -\mathrm{i}\omega \left(k_i k_j - \frac{1}{n} \delta_{ij} k^2 \right) \Phi. \tag{3.85}$$

Since $\tau' = 2t - \tau_1 - \tau_2$, the change of variables $\tau_1 \rightarrow t - \tau_1$ and $\tau_2 \rightarrow t - \tau_2$ in equation (3.78) yields

$$V(t) = \frac{1}{2} \int_0^\infty \int_0^\infty \check{\psi}_{\mathcal{K}}(\tau_1 + \tau_2) \partial \epsilon_{ii}(t - \tau_1) \partial \epsilon_{kk}(t - \tau_2) d\tau_1 d\tau_2 + \int_0^\infty \int_0^\infty \check{\psi}_{\mu}(\tau_1 + \tau_2) \partial d_{ij}(t - \tau_1) \partial d_{ij}(t - \tau_2) d\tau_1 d\tau_2.$$
(3.86)

Averaging over a period $2\pi/\omega$ by using (1.105), we note that

$$\langle \partial \epsilon_{ii}(t-\tau_1) \partial \epsilon_{kk}(t-\tau_2) \rangle = \frac{1}{2} \operatorname{Re} \{ \partial \epsilon_{ii}(t-\tau_1) [\partial \epsilon_{kk}(t-\tau_2)]^* \} = \frac{1}{2} \omega^2 |k|^4 |\Phi|^2 \cos[\omega(\tau_2-\tau_1)],$$

(3.87)

where equation (3.84) has been used. Similarly, using (3.85), we obtain

$$\langle \partial d_{ij}(t-\tau_1)\partial d_{ij}(t-\tau_2)\rangle = \frac{1}{2}\omega^2 \left|k_ik_j - \frac{1}{n}\delta_{ij}k^2\right|^2 |\Phi|^2 \cos[\omega(\tau_2-\tau_1)],$$
 (3.88)

where implicit summation is assumed in the square of the absolute modulus. Now, by a new change of variables similar to the one used to obtain equation (2.47), we have

$$\langle V \rangle = \frac{1}{4} \omega |k|^4 |\Phi|^2 \int_0^\infty \check{\psi}_{\mathcal{K}} \sin(\omega\zeta) d\zeta + \frac{1}{2} \omega \left| k_i k_j - \frac{1}{n} \delta_{ij} k^2 \right|^2 |\Phi|^2 \int_0^\infty \check{\psi}_{\mu} \sin(\omega\zeta) d\zeta.$$
(3.89)

Substituting the expressions of the real moduli (3.22) into equation (3.89), we obtain

$$\langle V \rangle = \frac{1}{4} |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \left(|k|^4 \mathcal{K}_R + 2 \left| k_i k_j - \frac{1}{n} \delta_{ij} k^2 \right|^2 \mu_R \right), \quad (3.90)$$

where equations (3.25) and (3.26) were used. But,

$$\begin{aligned} |k_i k_j - \frac{1}{n} \delta_{ij} k^2|^2 &= (k_i k_j - \frac{1}{n} \delta_{ij} k^2) (k_i^* k_j^* - \frac{1}{n} \delta_{ij} k^{2^*}) \\ &= (\mathbf{k} \cdot \mathbf{k}^*)^2 - \frac{1}{n} k^2 k^{2^*} \\ &= (\kappa^2 + \alpha^2)^2 - \frac{1}{n} |k^2|^2 \\ &= [\operatorname{Re}(k^2)]^2 + [\operatorname{Im}(k^2)]^2 \sec^2 \gamma - \frac{1}{n} \{ [\operatorname{Re}(k^2)]^2 + [\operatorname{Im}(k^2)]^2 \} \\ &= \{ [\operatorname{Re}(k^2)]^2 + [\operatorname{Im}(k^2)^2] \} (1 - \frac{1}{n}) + [\operatorname{Im}(k^2)]^2 \tan^2 \gamma, \end{aligned}$$
(3.91)

where we have used (3.24) and (3.34). Substituting expression (3.91) into equation (3.90), we have

$$\langle V \rangle = \frac{1}{4} |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \left\{ \{ [\operatorname{Re}(k^2)]^2 + [\operatorname{Im}(k^2)]^2 \} \left[\mathcal{K}_R + \left(2 - \frac{2}{n}\right) \mu_R \right] + 2\mu_R [\operatorname{Im}(k^2) \tan \gamma]^2 \}.$$
(3.92)

From equation (3.23),

$$\mathcal{K}_R + \left(2 - \frac{2}{n}\right)\mu_R = \operatorname{Re}(\mathcal{E}) = \frac{\rho\omega^2 \operatorname{Re}(k^2)}{[\operatorname{Re}(k^2)]^2 + [\operatorname{Im}(k^2)]^2},$$
 (3.93)

such that (3.92) becomes

$$\langle V \rangle = \frac{1}{4} |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \{\rho \omega^2 \operatorname{Re}(k^2) + 2\mu_R [\operatorname{Im}(k^2) \tan \gamma]^2 \}, \quad (3.94)$$

where k must be replaced by k_P . This can be written in terms of the medium properties by using equation (3.24). Expression (3.94) is obtained by Buchen (1971a).

Note, from equation (3.33), that

$$Re(k^2) = \kappa^2 - \alpha^2 \qquad (3.95)$$

and

$$[\operatorname{Im}(k^2) \tan \gamma]^2 = 4\kappa^2 \alpha^2 \sin^2 \gamma = 4|\kappa \times \alpha|^2.$$
 (3.96)

Therefore, equation (3.94) can be rewritten as

$$\langle V \rangle = \frac{1}{4} |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) [\rho \omega^2 (|\boldsymbol{\kappa}|^2 - |\boldsymbol{\alpha}|^2) + 8\mu_R |\boldsymbol{\kappa} \times \boldsymbol{\alpha}|^2].$$
 (3.97)

This form is obtained by Borcherdt (1973), but with the factor 4 in the second term, instead of the factor 8. This and other discrepancies have given rise to a discussion between Krebes (1983a) and Borcherdt (see Borcherdt and Wennerberg, 1985) regarding the preferred definitions of strain and dissipated energies. As pointed out by Caviglia and Morro (1992, p. 57), in the general case, the ambiguities remain, even though the time averages are considered. We should emphasize, however, that the ambiguity disappears when we consider energy densities compatible with mechanical models of viscoelastic behavior, as in the approach followed by Buchen (1971a). This discrepancy does not occur for homogeneous waves, because $\kappa \times \alpha = 0$, but may have implications when calculating the reflection and transmission coefficients at discontinuities, since inhomogeneous waves are generated.

The same procedure can be used to obtain the time-averaged rate of dissipated-energy density. (The reader may try to obtain the expression as an exercise). A detailed demonstration is given by Buchen (1971a):

$$\langle \dot{D} \rangle = \frac{1}{2} \omega |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \operatorname{Im}(k^2) [-\rho \omega^2 + 2\mu_I \operatorname{Im}(k^2) \tan^2 \gamma], \qquad (3.98)$$

or, in terms of the wavenumber and attenuation vectors, time-averaged dissipated-energy density is

$$\langle D \rangle = |\Phi_0|^2 \exp(-2\alpha \cdot \mathbf{x}) [\rho \omega^2 (\mathbf{\kappa} \cdot \alpha) + 4\mu_I |\mathbf{\kappa} \times \alpha|^2],$$
 (3.99)

3.4 Energy balance, energy velocity and quality factor

where we have used equations (2.106), $(3.33)_2$ and (3.96).

We now note the following properties. Since $\langle V \rangle$ in (3.97) must be a positive definite quantity, it follows that $\text{Re}(k_P^2) > 0$, and from (3.24),

$$\text{Re}(\mathcal{E}) = \mathcal{E}_R > 0.$$
 (3.100)

In addition,

$$\mu_R > 0.$$
 (3.101)

Also, since $\langle D \rangle$ must be non-negative, it follows that $\text{Im}(k_P^2) < 0$ – this can also be deduced from (3.33), since $\kappa \cdot \alpha > 0$ – or

$$Im(\mathcal{E}) = \mathcal{E}_{l} > 0,$$
 (3.102)

according to (3.24). Furthermore, since $\kappa \cdot \alpha > 0$,

$$\mu_I > 0.$$
 (3.103)

The time average of the total energy density, from (3.82) and (3.97), is

$$\langle E \rangle = \langle T + V \rangle = \frac{1}{2} |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) [\rho \omega^2 |\boldsymbol{\kappa}|^2 + 4\mu_R |\boldsymbol{\kappa} \times \boldsymbol{\alpha}|^2].$$
(3.104)

When the motion is lossless $(Im(k_p^2) = 0)$, the time-averaged kinetic- and strain-energy densities are the same

$$\langle T \rangle = \langle V \rangle = \frac{1}{4} |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \rho \omega^2 k_P^2.$$
 (3.105)

Let us calculate now the time-averaged energy flow, or the time average of the Umov-Poynting vector. From equations (1.105) and (3.73), we have

$$\langle p_i \rangle = -\frac{1}{2} \operatorname{Re}(v_j^* \sigma_{ij}).$$
 (3.106)

From (3.2) and (3.21) and using the relation $\mathcal{K} = \mathcal{E} - 2\mu(1 - 1/n)$ (see (3.23)), we write the stress-strain relation as

$$\sigma_{ij} = \left(\mathcal{K} - \frac{2}{n}\mu\right)\epsilon_{kk}\delta_{ij} + 2\mu\epsilon_{ij} = \mathcal{E}\epsilon_{kk}\delta_{ij} + 2\mu(\epsilon_{ij} - \epsilon_{kk}\delta_{ij}).$$
(3.107)

The following expressions are obtained for the plane wave (3.25),

$$v_j^* = \omega \Phi_0^* k_j^* \exp[-i(\omega t - \mathbf{k}^* \cdot \mathbf{x})] = \omega k_j^* \Phi^*, \qquad (3.108)$$

$$\epsilon_{kk} = -k^2 \Phi$$
 (3.109)

and

$$\epsilon_{ij} = -k_i k_j \Phi.$$
 (3.110)

Substituting these expressions into equations (3.106) and (3.107) yields

$$\langle p_i \rangle = \frac{1}{2} \omega |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \operatorname{Re}[\mathcal{E}k_j^* k^2 \delta_{ij} + 2\mu k_j^* (k_i k_j - k^2 \delta_{ij})], \qquad (3.111)$$

or, using (3.24), we have

$$\langle p_i \rangle = \frac{1}{2} \omega |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \operatorname{Re}[\rho \omega^2 k_j^* \delta_{ij} + 2\mu k_j^* (k_i k_j - k^2 \delta_{ij})].$$
 (3.112)

We can now write

$$k_j^*(k_ik_j - k^2\delta_{ij}) = \mathbf{k} \cdot \mathbf{k}^*k_i - \mathbf{k} \cdot \mathbf{k} \ k_i^*$$

= $(\mathbf{\kappa} \cdot \mathbf{\kappa} + \mathbf{\alpha} \cdot \mathbf{\alpha})k_i - (\mathbf{\kappa} \cdot \mathbf{\kappa} - \mathbf{\alpha} \cdot \mathbf{\alpha} - 2\mathbf{i}\mathbf{\alpha} \cdot \mathbf{\kappa})k_i^*$
= $\mathbf{\kappa} \cdot \mathbf{\kappa}(k_i - k_i^*) + \mathbf{\alpha} \cdot \mathbf{\alpha}(k_i + k_i^*) + 2\mathbf{i}\mathbf{\alpha} \cdot \mathbf{\kappa} \ k_i^*$
= $2\mathbf{i}\mathbf{\kappa} \cdot \mathbf{\kappa} \ \operatorname{Im}(k_i) + 2\mathbf{\alpha} \cdot \mathbf{\alpha} \ \operatorname{Re}(k_i) + 2\mathbf{i}\mathbf{\alpha} \cdot \mathbf{\kappa} \ k_i^*$ (3.113)

or, in vector form

vector =
$$-2i(\kappa \cdot \kappa)\alpha + 2(\alpha \cdot \alpha)\kappa + 2i(\alpha \cdot \kappa)(\kappa + i\alpha)$$

= $2\kappa[\alpha \cdot (i\kappa + \alpha)] - 2\alpha[\kappa \cdot (i\kappa + \alpha)]$
= $2(i\kappa + \alpha) \times (\kappa \times \alpha) = -2(\kappa \times \alpha) \times (i\kappa + \alpha),$
(3.114)

where we have used the property $\mathbf{a} \times (\mathbf{b} \times \mathbf{c}) = (\mathbf{a} \cdot \mathbf{c})\mathbf{b} - (\mathbf{a} \cdot \mathbf{b})\mathbf{c}$ (Brand, 1957, p. 31).

Substituting equation (3.114) into equation (3.112) and taking the real part, we obtain the final form of the time-averaged power flow, namely,

$$\langle \mathbf{p} \rangle = \frac{1}{2} \omega |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) [\rho \omega^2 \boldsymbol{\kappa} + 4(\boldsymbol{\kappa} \times \boldsymbol{\alpha}) \times (\mu_I \boldsymbol{\kappa} - \mu_R \boldsymbol{\alpha})].$$
 (3.115)

We can infer that the energy propagates in the plane of κ and α , but not in the direction perpendicular to the wave surface, as is the case with elastic materials and homogeneous viscoelastic waves, for which $\kappa \times \alpha = 0$.

Let us perform the scalar product of the time-averaged energy flow $\langle \mathbf{p} \rangle$ with the attenuation vector $\boldsymbol{\alpha}$. The second term contains the scalar triple product $[(\boldsymbol{\kappa} \times \boldsymbol{\alpha}) \times (i\boldsymbol{\kappa} + \boldsymbol{\alpha})] \cdot \boldsymbol{\alpha}$ (see equation (3.114)). Using the property of the triple product $(\mathbf{a} \times \mathbf{b}) \cdot \mathbf{c} = (\mathbf{b} \times \mathbf{c}) \cdot \mathbf{a}$ (Brand, 1957, p. 33), we have

$$[(\boldsymbol{\kappa} \times \boldsymbol{\alpha}) \times (i\boldsymbol{\kappa} + \boldsymbol{\alpha})] \cdot \boldsymbol{\alpha} = [(i\boldsymbol{\kappa} + \boldsymbol{\alpha}) \times \boldsymbol{\alpha}] \cdot (\boldsymbol{\kappa} \times \boldsymbol{\alpha}) = i|\boldsymbol{\kappa} \times \boldsymbol{\alpha}|^2.$$
(3.116)

Using this equation, we obtain

$$\langle \mathbf{p} \rangle \cdot \boldsymbol{\alpha} = \frac{1}{2} \langle \dot{D} \rangle,$$
 (3.117)

or

$$\langle \hat{D} \rangle = 2 \langle \mathbf{p} \rangle \cdot \boldsymbol{\alpha},$$
 (3.118)

where $\langle \hat{D} \rangle$ is the time-averaged rate of dissipated energy (3.99). Moreover, since

$$2(\mathbf{p}) \cdot \boldsymbol{\alpha} = (D) = -(\operatorname{div} \mathbf{p}),$$
 (3.119)

we can infer from equations (3.75) and (3.104) that the mean value of the rate of total energy density vanishes

$$(\partial_t E) = 0.$$
 (3.120)

On the other hand, if we calculate the scalar product between the time-averaged energyflow vector $\langle \mathbf{p} \rangle$ and the wavenumber vector $\boldsymbol{\kappa}$, we obtain $|\boldsymbol{\kappa} \times \boldsymbol{\alpha}|^2$ for the corresponding triple product. Using this fact, we obtain the following relation

$$\omega(E) = \langle \mathbf{p} \rangle \cdot \boldsymbol{\kappa},$$
 (3.121)

As in the 1-D case (equation (2.114)) and the lossless anisotropic case (equation (1.113)), we define the energy-velocity vector as the ratio of the time-averaged energy-flow vector to the time-averaged energy density,

$$\mathbf{v}_e = \frac{\langle \mathbf{p} \rangle}{\langle E \rangle}$$
. (3.122)

Combining (3.121) and (3.122), we obtain the relation

$$\hat{\kappa} \cdot \mathbf{v}_e = v_p$$
, $(\mathbf{s}_R \cdot \mathbf{v}_e = 1)$, (3.123)

where v_p and \mathbf{s}_R are the phase velocity and slowness vector introduced in (3.37). This can be interpreted as the lines of constant phase traveling with velocity v_e in the direction of $\langle \mathbf{p} \rangle$. This property is satisfied by plane waves propagating in an anisotropic elastic medium (see equation (1.114)). Here, the relation also holds for inhomogeneous viscoelastic plane waves. For homogeneous waves, $\gamma = 0$, $\kappa \times \alpha = 0$, equations (3.104) and (3.115) become

$$\langle E \rangle = \frac{1}{2} |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \rho \omega^2 |\boldsymbol{\kappa}|^2$$
 (3.124)

and

$$\langle \mathbf{p} \rangle = \frac{1}{2} \omega |\Phi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) \rho \omega^2 \boldsymbol{\kappa},$$
 (3.125)

and we have $\mathbf{v}_e = \mathbf{v}_p$.

As in the 1-D case (equation (2.119)), we define the quality factor as

$$Q = \frac{2\langle V \rangle}{\langle D \rangle}$$
, (3.126)

where

$$\langle D \rangle \equiv \omega^{-1} \langle \dot{D} \rangle$$
 (3.127)

is the time-averaged dissipated-energy density. Substituting the time-averaged strainenergy density (3.97) and the time-averaged rate of dissipated-energy density (3.99), and using equations (3.18) and (3.33), we obtain

$$Q = \frac{\kappa^2 - \alpha^2}{2\kappa\alpha} = -\frac{\text{Re}(k^2)}{\text{Im}(k^2)} = \frac{\text{Re}(v_c^2)}{\text{Im}(v_c^2)},$$
 (3.128)

where we assumed homogeneous waves ($\gamma = 0$). As in the 1-D case, we obtain the relation

$$\alpha = \frac{\pi f}{Qv_p}$$
(3.129)

for $Q \gg 1$ and homogeneous plane waves. On the other hand, definition (2.124) and equations (3.118), (3.122) and (3.127) imply

$$Q = \frac{\langle E \rangle}{\langle D \rangle} = \frac{\omega \langle E \rangle}{\langle \dot{D} \rangle} = 2 \left(\frac{\omega \langle E \rangle}{\langle \mathbf{p} \rangle \cdot \boldsymbol{\alpha}} \right) = \frac{\omega}{2\mathbf{v}_{e} \cdot \boldsymbol{\alpha}}.$$
(3.130)

For homogeneous waves, this equation implies the relation

$$\alpha = \frac{\pi f}{Qv_p}, \quad (3.131)$$

which holds without requiring the condition $Q \gg 1$.

113

3.4.2 S waves

The results for the type-I S wave have the same form as those for the P wave, while the results for the type-II S wave are similar, but differ by a factor of 2 in the inhomogeneous term. We have

$$\begin{aligned} \langle \mathbf{p} \rangle &= \frac{1}{4} \omega |\Xi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) [\rho \omega^2 \boldsymbol{\kappa} + a(\boldsymbol{\kappa} \times \boldsymbol{\alpha}) \times (\mu_I \boldsymbol{\kappa} - \mu_R \boldsymbol{\alpha})], \\ \langle T \rangle &= \frac{1}{4} \rho \omega^2 |\Xi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) (|\boldsymbol{\kappa}|^2 + |\boldsymbol{\alpha}|^2), \\ \langle V \rangle &= \frac{1}{4} |\Xi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) [\rho \omega^2 (|\boldsymbol{\kappa}|^2 - |\boldsymbol{\alpha}|^2) + 2a\mu_R |\boldsymbol{\kappa} \times \boldsymbol{\alpha}|^2], \\ \langle E \rangle &= \frac{1}{2} |\Xi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) [\rho \omega^2 |\boldsymbol{\kappa}|^2 + a\mu_R |\boldsymbol{\kappa} \times \boldsymbol{\alpha}|^2], \\ \langle \dot{D} \rangle &= \omega |\Xi_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}) [\rho \omega^2 (\boldsymbol{\kappa} \cdot \boldsymbol{\alpha}) + a\mu_I |\boldsymbol{\kappa} \times \boldsymbol{\alpha}|^2], \end{aligned}$$
(3.132)

where $\Xi_0 = \Phi_0$ and a = 4 for P waves, $\Xi_0 = \Theta_0$ and a = 4 for type-I S waves, and $\Xi_0 = \Gamma_0/(\mathbf{k} \cdot \mathbf{k}^*)$ and a = 2 for type-II S waves (see equations (3.25), (3.57) (3.59)). Some details of the preceding equations can be found in Buchen (1971a) and Krebes (1983a), and in Caviglia and Morro (1992, pp. 57-60), where an additional correction term is added to the time-averaged energy-flow vector of the type-I S wave. Moreover, relations (3.120), (3.123) and (3.118) are valid in general, as are the expressions for the energy velocity and quality factor (3.122) and (3.128), respectively. The extension to isotropic poro-viscoelastic media is given by Rasolofosaon (1991).

3.5 Boundary conditions and Snell's law

A picture illustrating the reflection-transmission phenomenon is shown in Figure 3.5, where the vectors κ , α and \hat{n} need not be coplanar.



Figure 3.5: The reflection-transmission problem in viscoelastic media.

Depending on the nature of the incident waves, four or two waves are generated at the interface, corresponding to the P-SV (P-(S-I)) and SH/SH ((S-II)-(S-II)) scattering problems. If the two media are in welded contact, the boundary conditions are the continuity of the displacements (or particle velocities) and normal stresses across the interface, that is continuity of

$$u_i$$
 and $\sigma_{ij}n_j$ (3.133)

(Auld, 1990a, p. 124). This implies that the complex phase $\mathbf{k} \cdot \mathbf{x}$ at any point of the interface is the same for all the waves involved in the process, that is

$$\mathbf{k}^{I} \cdot \mathbf{x} = \mathbf{k}^{R} \cdot \mathbf{x} = \mathbf{k}^{T} \cdot \mathbf{x}, \quad \mathbf{x} \cdot \hat{\mathbf{n}} = 0,$$
 (3.134)

or

$$\mathbf{k}^{I} \times \hat{\mathbf{n}} = \mathbf{k}^{R} \times \hat{\mathbf{n}} = \mathbf{k}^{T} \times \hat{\mathbf{n}}.$$
 (3.135)

In terms of the complex wavevector \mathbf{k} and slowness vector $\mathbf{s} = \mathbf{k}/\omega$, and identifying the interface with the plane z = 0, we have

$$k_1^{\prime} = k_1^R = k_1^T$$
 and $k_2^{\prime} = k_2^R = k_2^T$, (3.136)

$$s_1^{\prime} = s_1^R = s_1^T$$
 and $s_2^{\prime} = s_2^R = s_2^T$. (3.137)

This general form of Snell's law implies the continuity at the interface of the tangential component of the real and imaginary parts of the complex wavevector (or complexslowness vector) and, therefore, the continuity of the tangential components of κ and α . This condition can be written as

$$\kappa^{I} \sin \theta^{I} = \kappa^{R} \sin \theta^{R} = \kappa^{T} \sin \theta^{T}$$

 $\alpha^{I} \sin(\theta^{I} - \gamma^{I}) = \alpha^{R} \sin(\theta^{R} - \gamma^{R}) = \alpha^{T} \sin(\theta^{T} - \gamma^{T}).$
(3.138)

It is not evident from this equation that the reflection angle is equal to the incidence angle for waves of the same type. Because k^2 is a material property independent of the inhomogeneity angle (see equation (3.24)), the relation

$$k_3^2 = k^2 - (k_1^2 + k_2^2) \qquad (3.139)$$

and equation (3.136) imply $k_3^{I^2} = k_3^{R^2}$. Since the z-components of the incident and reflected waves should have opposite signs, we have

$$k_3^R = -k_3^R$$
. (3.140)

This relation and equation (3.136) imply $\kappa^{I} = \kappa^{R}$ and $\alpha^{I} = \alpha^{R}$, and from (3.138)

$$\theta^R = \theta^I$$
 and $\gamma^R = \gamma^I$. (3.141)

Therefore, the reflected wave is homogeneous only if the incident wave is homogeneous. More consequences from the viscoelastic nature of Snell's law are discussed in Section 3.8, where we solve the problem of reflection and transmission of SH waves¹.

¹Note that to obtain Snell's law we have not used the assumption of isotropy. Thus, equations (3.138) are also valid for anisotropic anelastic media.

3.6 The correspondence principle

The correspondence principle allows us to obtain viscoelastic solutions from the corresponding elastic (lossless) solutions. The stress-strain relation (3.2) can be rewritten as

$$\sigma_{ij} = \psi_{\kappa} * \partial_t \epsilon_{kk} \delta_{ij} + 2\psi_{\mu} * \partial_t d_{ij},$$
 (3.142)

where d_{ij} is defined in equation (3.6).

Note that the Fourier transform of the stress-strain relations (3.142) is

$$\sigma_{ij}(\omega) = \mathcal{K}(\omega)\epsilon_{kk}(\omega)\delta_{ij} + 2\mu(\omega)d_{ij}(\omega),$$
 (3.143)

where

$$\mathcal{K}(\omega) = \mathcal{F}[\partial_t \psi_{\mathcal{K}}(t)]$$
 and $\mu(\omega) = \mathcal{F}[\partial_t \psi_{\mu}(t)]$ (3.144)

are the corresponding complex moduli. The form (3.143) is similar to the stress-strain relation of linear elasticity theory, except that the moduli are complex and frequency dependent. Note also that Euler's differential equations (1.23) are the same for lossy and lossless media. Therefore, if the elastic solution is available, the viscoelastic solution is obtained by replacing the elastic moduli with the corresponding viscoelastic moduli. This is known as the correspondence principle². We show specific examples of this principle in Section 3.10. Extensions of the correspondence principle are given in Golden and Graham (1988, p. 68).

3.7 Rayleigh waves

The importance of Rayleigh waves can be noted in several fields, from carthquake seismology to material science (Parker and Maugin, 1988; Chadwick, 1989). The first theoretical investigations carried out by Lord Rayleigh (1885) in isotropic elastic media showed that these waves are confined to the surface and, therefore, they do not scatter in depth as do seismic body waves.

Hardtwig (1943) was the first to study viscoelastic Rayleigh waves, though he erroneously restricts their existence to a particular choice of the complex Lamé parameters. Scholte (1947) rectifies this mistake and verifies that the waves always exist in viscoelastic solids. He also predicts the existence of a second surface wave, mainly periodic with depth, whose exponential damping is due to anelasticity and not to the Rayleigh character – referred to later as v.e. mode. Caloi (1948) and Horton (1953) analyze the anelastic characteristics and displacements of the waves considering a Voigt-type dissipation mechanism with small viscous damping, and a Poisson solid. Borcherdt (1973) analyzes the particle motion at the free surface and concludes that the differences between elastic and viscoelastic Rayleigh waves arise from differences in their components: the usual inhomogeneous plane waves in the elastic case, and viscoelastic inhomogeneous plane waves in the anelastic case, which allow any angle between the propagation and attenuation vectors.

²Although the principle has been illustrated for isotropic media, its extension to the anisotropic case can be obtained by taking the Fourier transform of the stress-strain relation (2.22), which leads to equation (4.4).

3.7 Rayleigh waves

A complete analysis is carried out by Currie, Hayes and O'Leary (1977), Currie and O'Leary (1978) and Currie (1979). They show that for viscoelastic Rayleigh waves: (i) more than one wave is possible, (ii) the particle motion may be either direct or retrograde at the surface, (iii) the motion may change sense at many or no levels with depth, (iv) the wave energy velocity may be greater than the body waves energy velocities. They refer to the wave that corresponds to the usual elastic surface wave as quasi-elastic (q.e.), and to the wave that only exists in the viscoelastic medium as viscoelastic (v.e.). This mode is possible only for certain combinations of the complex Lamé constants and for a given range of frequencies. Using the method of generalized rays, Borejko and Ziegler (1988) study the characteristics of the v.e. surface waves for the Maxwell and Kelvin-Voigt solids.

3.7.1 Dispersion relation

Since the medium is isotropic, we assume without loss of generality that the wave propagation is in the (x, z)-plane with z = 0 being the free surface. Let a plane-wave solution to equation (1.23) be of the form

$$\mathbf{u} = \mathbf{U} \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})]. \quad (3.145)$$

For convenience, let m = 1 denote the compressional wave and m = 2 the shear wave. We rewrite the dispersion relations (3.24) as

$$k^{(m)^2} = \frac{\omega^2}{v_m^2}, \quad v_1^2 = \frac{\mathcal{E}}{\rho}, \quad v_2^2 = \frac{\mu}{\rho},$$
 (3.146)

where $\mathcal{E}(\omega) = \lambda(\omega) + 2\mu(\omega)$.

A general solution is given by the superposition of the compressional and shear modes,

$$\mathbf{u} = \mathbf{U}^{(m)} \exp[\mathbf{i}(\omega t - \mathbf{k}^{(m)} \cdot \mathbf{x})], \qquad (3.147)$$

where

$$U^{(1)} = U_0 \mathbf{k}^{(1)}, \quad U^{(2)} \cdot \mathbf{k}^{(2)} = 0.$$
 (3.148)

At the free surface (z = 0), the boundary conditions are

$$\sigma_{33} = \lambda \partial_1 u_1 + (\lambda + 2\mu) \partial_3 u_3 = 0$$
, and $\sigma_{13} = \mu (\partial_1 u_3 + \partial_3 u_1) = 0.$ (3.149)

These boundary conditions imply that the horizontal wavenumber is the same for each mode,

$$k_1^{(1)} = k_1^{(2)} \equiv k_1 = \kappa_1 - i\alpha_1.$$
 (3.150)

From equations (3.147) and (3.150), the displacement components are

$$u_1 = F(z) \exp[i(\omega t - k_1 x)],$$
 $F(z) = U_1^{(m)} \exp(-ik_3^{(m)} z),$
 $u_3 = G(z) \exp[i(\omega t - k_1 x)],$ $G(z) = U_3^{(m)} \exp(-ik_3^{(m)} z),$
(3.151)

where the vertical wavenumbers are

$$k_3^{(m)} = \kappa_3^{(m)} - i\alpha_3^{(m)}$$
. (3.152)

Chapter 3. Isotropic anelastic media

From equations (3.146), (3.148) and (3.150),

$$k_3^{(m)^2} = \frac{\omega^2}{v_m^2} - k_1^2 = \omega^2 \left(\frac{1}{v_m^2} - \frac{1}{v_e^2}\right), \qquad (3.153)$$

and

$$\frac{U_3^{(1)}}{U_1^{(1)}} = \frac{k_3^{(1)}}{k_1} = \sqrt{\frac{v_c^2}{v_1^2} - 1}, \quad \frac{U_3^{(2)}}{U_1^{(2)}} = -\frac{k_1}{k_3(2)} = -\left(\frac{v_c}{v_2^2} - 1\right)^{-1/2}, \quad (3.154)$$

where

$$v_e = \frac{\omega}{k_1}$$
 (3.155)

is the Rayleigh-wave complex velocity. The boundary conditions (3.149) and equations (3.154) imply

$$\frac{U_1^{(2)}}{U_1^{(1)}} = \frac{v_c^2}{2v_2^2} - 1 \equiv A \qquad (3.156)$$

and

$$A^2 + \frac{k_3^{(1)}k_3^{(2)}}{k_1^2} = 0.$$
 (3.157)

The squaring of (3.157) and reordering of terms gives a cubic equation for the complex velocity,

$$q^3 - 8q^2 + \left(24 - 16\frac{v_2^2}{v_1^2}\right)q - 16\left(1 - \frac{v_2^2}{v_1^2}\right) = 0, \quad q = \frac{v_c^2}{v_2^2},$$
 (3.158)

which could, alternatively, be obtained by using the correspondence principle (see Section 3.6) and the elastic Rayleigh-wave dispersion relation. The dispersion relation (3.158), together with equation (3.157), may determine one or more wave solutions. The solution of the q.e. surface wave is always possible, since it is the equivalent of the elastic Rayleigh wave. The other surface waves, called v.e. modes, are possible depending on the frequency and the material properties (Currie, 1979).

3.7.2 Displacement field

The amplitude coefficients may be referred to $U_1^{(1)} = 1$ without loss of generality. Thus, from equations (3.154) and (3.156),

$$U_1^{(1)} = 1, \quad U_1^{(2)} = A, \quad U_3^{(1)} = \frac{k_3^{(1)}}{k_1}, \quad U_3^{(2)} = \frac{k_3^{(1)}}{k_1 A}.$$
 (3.159)

From these equations, the displacements (3.151) become

$$u_1 = [\exp(-ik_3^{(1)}z) + A \exp(-ik_3^{(2)}z)] \exp[i(\omega t - k_1x)],$$

$$u_3 = (k_3^{(1)}/k_1) \left[\exp(-ik_3^{(1)}z) + A^{-1} \exp(-ik_3^{(2)}z)\right] \exp[i(\omega t - k_1x)].$$
(3.160)

These displacements are a combination of compressional and shear modes, with the phase factors

$$\exp\{i[\omega t - (\kappa_1 x + \kappa_3^{(m)} z)]\} \exp[-(\alpha_1 x + \alpha_3^{(m)} z)], \quad m = 1, 2, \quad (3.161)$$

118

3.7 Rayleigh waves

given by virtue of equations (3.150) and (3.152). It is clear, from the last equation, that to have attenuating waves, a physical solution of equation (3.158) must satisfy the following conditions:

$$\alpha_1 > 0$$
, $\alpha_3^{(m)} > 0$, $\kappa_1 > 0$. (3.162)

The last condition imposes wave propagation along the positive x-direction. In terms of the complex velocities, these conditions read

$$-\omega \operatorname{Im}\left(\frac{1}{v_{c}}\right) > 0, \quad -\operatorname{Im}\left(\sqrt{\frac{1}{v_{m}^{2}} - \frac{1}{v_{c}^{2}}}\right) > 0, \quad \omega \operatorname{Re}\left(\frac{1}{v_{c}}\right) > 0.$$
(3.163)

Also, equation (3.157) must be satisfied in order to avoid spurious roots.

3.7.3 Phase velocity and attenuation factor

The phase velocity in the x-direction is defined as the frequency divided by the xcomponent of the real wavenumber κ_1 ,

$$v_p \equiv \frac{\omega}{\kappa_1} = \frac{\omega}{\operatorname{Re}(k_1)} = \left[\operatorname{Re}\left(\frac{1}{v_c}\right)\right]^{-1}.$$
 (3.164)

From equation (3.161), the phase velocities associated with each component wave mode are

$$\mathbf{v}_{pm} = \omega \left(\frac{\kappa_1 \hat{\mathbf{e}}_1 + \kappa_3^{(m)} \hat{\mathbf{e}}_3}{\kappa_1^2 + \kappa_3^{(m)^2}} \right), \quad \text{and} \quad \mathbf{v}_{pm} = \left(\frac{\omega}{\kappa_1} \right) \hat{\mathbf{e}}_1, \quad \text{(lossless case)}. \tag{3.165}$$

In the elastic (lossless) case, there is only a single and physical solution to equation (3.158). Moreover, because the velocities are real and $v_c < v_2 < v_1$, $k_3^{(1)}$ and $k_3^{(2)}$ are purely imaginary and $\kappa_3^{(m)} = 0$. Hence, $v_{pm} = v_p$, and equation (3.161) reduces to

$$\exp[i(\omega t - \kappa_1 x)] \exp(-\alpha_3^{(m)} z),$$
 (3.166)

with $\kappa_1 = k_1$ and $\alpha_3^{(m)} = ik_3^{(m)}$. In this case, the propagation vector points along the surface and the attenuation vector is normal to the surface. However, in a viscoelastic medium, according to equation (3.165)₁, these vectors are inclined with respect to those directions.

The attenuation factor in the x-direction is given by

$$\alpha = -\omega \operatorname{Im}\left(\frac{1}{v_c}\right) > 0. \quad (3.167)$$

Each wave mode has an attenuation vector given by

$$\alpha_1 \hat{\mathbf{e}}_1 + \alpha_3^{(m)} \hat{\mathbf{e}}_3$$
, and $\alpha_3^{(m)} \hat{\mathbf{e}}_3$ (lossless case). (3.168)

Carcione (1992b) calculates the energy-balance equation and shows that, in contrast to elastic materials, the energy flow is not directed along the surface and the energy velocity is not equal to the phase velocity.

3.7.4 Special viscoelastic solids

Incompressible solid

Incompressibility implies $\lambda \rightarrow \infty$, or, equivalently, $v_1 \rightarrow \infty$. Hence, from equation (3.158), the dispersion relation becomes

$$q^3 - 8q^2 + 24q - 16 = 0.$$
 (3.169)

The roots are $q_1 = 3.5437 + i 2.2303$, $q_2 = 3.5437 - i 2.2303$ and $q_3 = 0.9126$. As shown by Currie, Hayes and O'Leary (1977), two Rayleigh waves exist, the quasi-elastic mode, represented by q_3 , and the viscoelastic mode, represented by q_1 , which is admissible if $Im(v_2^2)/Re(v_2^2) > 0.159$, in order to fulfill conditions (3.162). In Currie, Hayes and O'Leary (1977), the viscoelastic root is given by q_2 , since they use the opposite sign convention to compute the time-Fourier transform (see also Currie, 1979). Carcione (1992b) shows that at the surface, the energy velocity is equal to the phase velocity.

Poisson solid

A Poisson solid has $\lambda = \mu$, so that $v_1 = \sqrt{3}v_2$ and, therefore, equation (3.158) becomes

$$3q^3 - 24q^2 + 56q - 32 = 0.$$
 (3.170)

This equation has three real roots: $q_1 = 4$, $q_2 = 2 + 2 / \sqrt{3}$, and $q_3 = 2 - 2 / \sqrt{3}$. The last root corresponds to the q.e. mode. The other two roots do not satisfy equation (3.157) and, therefore, there are no v.e. modes in a Poisson solid. As with the incompressible solid, the energy velocity is equal to the phase velocity at the surface.

Hardtwig solid

Hardtwig (1943) investigates the properties of a viscoelastic Rayleigh wave for which $\operatorname{Re}(\lambda)/\operatorname{Re}(\mu) = \operatorname{Im}(\lambda)/\operatorname{Im}(\mu)$. In this case, the coefficients of the dispersion relation (3.158) are real, ensuring at least one real root corresponding to the q.e. mode. This implies that the energy velocity coincides with the phase velocity at the surface (Carcione, 1992b). A Poisson medium is a particular type of Hardtwig solid.

3.7.5 Two Rayleigh waves

Carcione (1992b) studies a medium with $\rho = 2 \text{ gr/cm}^3$, and complex Lamé constants

$$\lambda = (-1.15 - i \ 0.197)$$
 GPa, $\mu = (4.91 + i \ 0.508)$ GPa

at a frequency of 20 Hz. The P-wave and S-wave velocities are 2089.11 m/s and 1573 m/s, respectively. Two roots satisfy equations (3.157) and (3.158): $q_1 = 0.711 - i 0.0046$ corresponds to the q.e. mode, and $q_2 = 1.764 - i 0.0156$ corresponds to the v.e. mode.

Figure 3.6 shows the absolute value of the horizontal and vertical displacements, $|u_1|$ and $|u_3|$, as a function of depth, for the q.e. Rayleigh wave (a) and the v.e. Rayleigh wave. Their phase velocities are 1326 m/s and 2089.27 m/s, respectively. The horizontal motion predominates in the v.e. Rayleigh wave and its phase velocity is very close to that of the P wave. For higher frequencies, this wave shows a strong oscillating behavior (Carcione, 1992b).



Figure 3.6: Absolute value of the horizontal and vertical displacements, $|u_1|$ (dashed line) and $|u_3|$ (solid line) versus depth, at a frequency of 20 Hz; (a) corresponds to the quasi-elastic Rayleigh wave, and (b) to the viscoelastic Rayleigh wave.

3.8 Reflection and transmission of cross-plane shear waves

The reflection-transmission problem in isotropic viscoelastic media is addressed by many researchers (for example, Cooper (1967), Buchen (1971b), Schoenberg (1971) and Stovas and Ursin (2001)). Borcherdt, Glassmoyer and Wennerberg (1986) present theoretical and experimental results and cite most of the relevant work carried out by R. Borcherdt on the subject. E. Krebes also contributes to the solution of the problem, mainly in connection with ray tracing in viscoelastic media (for example, Krebes, 1984; Krebes and Slawinski, 1991). A comprehensive review of the problem is given in Caviglia and Morro (1992).

In order to illustrate the main effects due to the presence of viscoelasticity, we analyze in some detail the reflection-transmission problem of SH waves at a plane interface, following Borcherdt (1977). The P-SV problem for transversely isotropic media (the symmetry axes are perpendicular to the interface) is analyzed in detail in Chapter 6.

The reflection and transmission coefficients for SH waves have the same form as the coefficients for lossless isotropic media, but they are not identical because the quantities involved are complex. Consequently, we may apply the correspondence principle (see Section 3.6) to the expressions found for perfect elastic media (equation (1.282)). We set $c_{46} = 0$, replace c_{44} and c_{66} by μ , and c'_{44} and c'_{66} by μ' . We obtain

$$R_{SS} = \frac{Z^I - Z^T}{Z^I + Z^T}, \quad T_{SS} = \frac{2Z^I}{Z^I + Z^T},$$
 (3.171)

where

$$Z^{I} = \mu s_{3}^{I}, \quad s_{3}^{I} = \sqrt{\rho/\mu - s_{1}^{2}},$$
 (3.172)
and

$$Z^T = \pm \mu' pv \sqrt{\rho' / \mu' - s_1^2},$$
 (3.173)

where pv denotes the principal value of the complex square root. (For the principal value, the argument of the square root lies between $-\pi/2$ and $+\pi/2$). As indicated by Krebes (1984), special care is needed when choosing the sign in equation (3.173), since a wrong choice may lead to discontinuities of the vertical wavenumber as a function of the incidence angle. Unlike the elastic case, the amplitude of the scattered waves can grow exponentially with distance from the interface (Richards, 1984). Thus, the condition of an exponentially decaying wave is not sufficient to obtain the reflection and transmission coefficients. Instead, the signs of the real and imaginary parts of s_3^T should be chosen to guarantee a smooth variation of s_3^T versus the incidence angle. Such an analysis is illustrated by Richards (1984).

Let us assume that the incident and transmitted waves are homogeneous. Then, $k = \kappa - i\alpha$ (see equation (3.27)), $\gamma = 0$ and from Snell's law (3.138), we have that

$$\frac{k^{T^2}}{k^2} = \frac{\sin^2 \theta^I}{\sin^2 \theta^T}$$
(3.174)

is a real quantity (we have omitted the superscript I in the wavenumber of the incident wave). This equation also implies the condition

$$\sin^2 \theta' \le \frac{k^{T^2}}{k^2}$$
. (3.175)

Let us denote the quality factor of the homogeneous plane wave by Q_H , as defined in equation (3.32). As for P waves, the quality factor of homogeneous SH waves is given by equation (3.128). In this case, $Q_H = \text{Re}(v_c^2)/\text{Im}(v_c^2) = \text{Re}(\mu)/\text{Im}(\mu) = \mu_R/\mu_I$. We deduce from equation (3.31) that if $Q'_H = Q_H$, then k^{T^2}/k^2 is real and vice versa, and from (3.24)₂, (3.37)₁, and (3.38)₁, we note that

$$\frac{k^{T^2}}{k^2} = \left(\frac{\rho'}{\rho}\right) \left(\frac{\mu_R}{\mu'_R}\right) = \frac{v_{pH}^2}{v_{pH}^{T^2}}.$$
(3.176)

Then, we may state a theorem attributed to Borcherdt (1977):

Theorem 1: If the incident SH wave is homogeneous and not normally incident, then the transmitted SH wave is homogeneous if and only if

$$Q'_{H} = Q_{H}, \quad \sin^{2} \theta' \le \frac{k^{T^{2}}}{k^{2}} = \left(\frac{\rho'}{\rho}\right) \left(\frac{\mu_{R}}{\mu'_{R}}\right) = \frac{v_{pH}^{2}}{v_{pH}^{T-2}}.$$
 (3.177)

Let us analyze now the reflection coefficient when $Q'_H = Q_H$. We can write

$$\mu = \mu_R (1 + iQ_H^{-1}) \equiv \mu_R W, \quad \mu' = \mu'_R W.$$
 (3.178)

Let us evaluate the numerator and denominator of R in equation (3.171) for precritical incidence angles $(\sin \theta^I \le v_{pH}/v_{pH}^T)$. Using (3.172) and (3.173), we have

$$Z' \pm Z^T = \mu \sqrt{\frac{\rho}{\mu} - s_1^2} \pm \mu' \sqrt{\frac{\rho'}{\mu'} - s_1^2}.$$
 (3.179)

122

For a homogeneous wave $s_1 = \sin \theta^I / v_c$, where $v_c = \sqrt{\mu/\rho}$ is the complex shear-wave velocity. Using this relation and (3.178), equation (3.179) becomes

$$Z^{I} \pm Z^{T} = \sqrt{W} \left\{ \sqrt{\rho \mu_{R}} \cos \theta^{I} \pm \sqrt{\rho' \mu_{R}'} \sqrt{1 - \left(\frac{\rho \mu_{R}'}{\rho' \mu_{R}}\right) \sin^{2} \theta^{I}} \right\}.$$
 (3.180)

Because W, which is the only complex quantity, appears as a multiplying factor in both the numerator and the denominator of R (see equation (3.171)), we obtain the expression of the elastic reflection coefficient (as if $Q_H^{-1} = Q'_H^{-1} = 0$). It can also be proved that for supercritical angles, the transmission coefficient is that of the lossless case. (See Krebes (1983b), or the reader can check these statements as an exercise). Note that there is no low-loss approximation, only the condition $Q'_H = Q_H$.

For lossless materials $Q_{II} = Q'_{II} = \infty$, and if $v_{pH}/v^T_{pH} < 1$, we have the well-known result that the transmitted wave is homogeneous if and only if $\sin \theta^I \leq v_{pH}/v^T_{pH} < 1$, with the equal sign corresponding to the critical angle (see equation (3.177)). Another consequence of Theorem 1 is that a normally incident homogeneous wave generates a homogeneous transmitted wave perpendicular to the interface. The most important consequence of Theorem 1 is that the transmitted wave will be, in general, inhomogeneous since in most cases $Q_{II} \neq Q'_{II}$. This implies that the velocity and the attenuation of the transmitted wave will be less than and greater than that of the corresponding homogeneous wave in the same medium. Moreover, the direction of energy flow will not coincide with the direction of phase propagation, and the velocity of the energy will not be equal to the phase velocity (see equation (3.123)).

The phase velocity of the transmitted wave is

$$v_p^{T^2} = \frac{\omega^2}{\kappa^{T^2}} = \frac{\omega^2}{\kappa_1^{T^2} + \kappa_3^{T^2}},$$
 (3.181)

where, from Snell's law

$$\kappa_1^T = \kappa_1$$
, (3.182)

$$\kappa_3^T = \pm \operatorname{Re}\left(\operatorname{pv}\sqrt{k^{T^2} - k_1^{T^2}}\right) = \pm \operatorname{Re}\left(\operatorname{pv}\sqrt{k^{T^2} - k_1^2}\right).$$
 (3.183)

For equation (3.183), we have assumed propagation in the x-direction, without loss of generality. Hence, unlike the lossless case, the phase velocity of the transmitted wave depends on the angles of incidence and on the inhomogeneity of the incident wave. From (3.138), the angle of refraction of the transmitted wave is

$$\frac{\sin^2 \theta^T}{\sin^2 \theta^I} = \frac{\kappa^2}{\kappa^{T^2}} = \frac{\kappa_1^2 + \kappa_3^2}{\kappa_1^2 + \kappa_1^{T^2}},$$
(3.184)

which depends on the angles of incidence. Moreover, the dependence of the frequency of all these quantities through k^2 and k^{T^2} implies that an incident wave composed of different frequencies will transmit a fan of inhomogeneous waves at different angles. In the lossless case, each wave of different frequency is transmitted at the same angle.

Another important result, given below, is related to the existence of critical angles (Borcherdt, 1977). Theorem 2: If the incidence medium is lossless and the transmission medium is anelastic, then there are no critical angles.

If θ^I is a critical angle, then $\theta^T = \pi/2$. Because the incidence medium is elastic, by Snell's law, the attenuation vector in the transmission medium is perpendicular to the interface and, hence, to the direction of propagation. However, since the transmission medium is anelastic, such a wave cannot exist (see condition (3.36) and equation (3.118)).

The analysis about the existence of critical angles and the energy flow and dissipation of the different waves is given in detail in Chapter 6, where the reflection-transmission problem of SH waves in the symmetry planes of monoclinic media is discussed. The main results are that critical angles in anelastic media exist only under very particular conditions, and that interference fluxes are not present in the lossless case (see Section 6.1.7). Some researchers define the critical angle as the angle of incidence for which the propagation angle of the transmitted wave is $\pi/2$, i.e., when the wavenumber vector κ is parallel to the interface (e.g., Borcherdt, 1977; Wennerberg, 1985, Caviglia, Morro and Pagani, 1989). This is not correct from a physical point of view. In Chapter 6, we adopt the criterion that the Umov-Poynting vector or energy-flow direction is parallel to the interface, which is the criterion used in anisotropic media. The two definitions coincide only in particular cases, because, in general, the phase-velocity and energy-velocity directions do not coincide. Theorem 2 is still valid when using the second criterion since the attenuation and Umov-Poynting vectors can never be perpendicular in an anelastic medium (see equation (3.118)).

3.9 Memory variables and equation of motion

The memory-variable approach introduced in Section 2.7 is essential to avoid numerical calculations of time convolutions when modeling wave propagation in the time domain. With this approach, we obtain a complete differential formulation. The relaxation functions in the stress-strain relation (3.142) for isotropic media have the form (2.198). We set

$$\psi_{\mathcal{K}}(t) = \mathcal{K}_{\infty} \left[1 - \frac{1}{L_1} \sum_{l=1}^{L_1} \left(1 - \frac{\tau_{cl}^{(1)}}{\tau_{\sigma l}^{(1)}} \right) \exp(-t/\tau_{\sigma l}^{(1)}) \right] H(t), \quad (3.185)$$

$$\psi_{\mu}(t) = \mu_{\infty} \left[1 - \frac{1}{L_2} \sum_{l=1}^{L_2} \left(1 - \frac{\tau_{el}^{(2)}}{\tau_{\sigma l}^{(2)}} \right) \exp(-t/\tau_{\sigma l}^{(2)}) \right] H(t), \quad (3.186)$$

where $\tau_{el}^{(\nu)}$ and $\tau_{\sigma l}^{(\nu)}$ are relaxation times corresponding to dilatational ($\nu = 1$) and shear ($\nu = 2$) attenuation mechanisms. They satisfy the condition (2.169), $\tau_{el}^{(\nu)} \ge \tau_{\sigma l}^{(\nu)}$, with the equal sign corresponding to the elastic case.

In terms of the Boltzmann operation (2.6), equation (3.142) reads

$$\sigma_{ij} = \psi_K \odot \epsilon_{kk} \delta_{ij} + 2\psi_\mu \odot d_{ij},$$
 (3.187)

or,

$$\sigma_{ij} = \mathcal{K}_U \left(\epsilon_{kk} + \sum_{l=1}^{L_1} e_l^{(1)} \right) \delta_{ij} + 2\mu_U \left(d_{ij} + \sum_{l=1}^{L_2} e_{ijl}^{(2)} \right), \tag{3.188}$$

3.9 Memory variables and equation of motion

where

$$\mathcal{K}_{U} = \frac{\mathcal{K}_{\infty}}{L_{1}} \sum_{l=1}^{L_{1}} \frac{\tau_{d}^{(1)}}{\tau_{\sigma l}^{(1)}}, \quad \mu_{U} = \frac{\mu_{\infty}}{L_{2}} \sum_{l=1}^{L_{2}} \frac{\tau_{d}^{(2)}}{\tau_{\sigma l}^{(2)}}, \quad (3.189)$$

and

$$e_l^{(1)} = \varphi_{1l} * \epsilon_{kk}, \quad l = 1, \dots, L_1$$
 (3.190)

and

$$e_{ijl}^{(2)} = \varphi_{2l} * d_{ij}, \quad l = 1, \dots, L_2$$
 (3.191)

are sets of memory variables for dilatation and shear mechanisms, with

$$\tilde{\varphi}_{\nu l} = \frac{1}{\tau_{\sigma l}^{(\nu)}} \left(\sum_{l=1}^{L_{\nu}} \frac{\tau_{el}^{(\nu)}}{\tau_{\sigma l}^{(\nu)}} \right)^{-1} \left(1 - \frac{\tau_{el}^{(\nu)}}{\tau_{\sigma l}^{(\nu)}} \right) \exp(-t/\tau_{\sigma l}^{(\nu)}).$$
(3.192)

As in the 1-D case (see equation (2.292)), the memory variables satisfy

$$e_l^{(1)} = \varphi_{1l}(0)\epsilon_{kk} - \frac{e_l^{(1)}}{\tau_{ol}^{(1)}}, \qquad e_{ijl}^{(2)} = \varphi_{2l}(0)d_{ij} - \frac{e_{ijl}^{(2)}}{\tau_{ol}^{(2)}}.$$
 (3.193)

For n = 2 and say, the (x, z)-plane, we have three independent sets of memory variables. In fact, since $d_{11} = -d_{33} = (\epsilon_{11} - \epsilon_{33})/2$, then $e_{11l}^{(2)} = \varphi_{2l} * d_{11} = -\varphi_{2l} * d_{33}$. The other two sets are $e_l^{(1)} = \varphi_{1l} * \epsilon_{kk}$ and $e_{13l}^{(2)} = \varphi_{2l} * \epsilon_{13}$. In 3-D space (n = 3), there are six sets of memory variables, since $d_{11} + d_{22} + d_{33} = 0$ implies $e_{11l}^{(2)} + e_{22l}^{(2)} + e_{33l}^{(2)} = 0$, and two of these sets are independent. The other four sets are $e_l^{(1)} = \varphi_{1l} * \epsilon_{kk}$, $e_{23l}^{(2)} = \varphi_{2l} * \epsilon_{23}$, $e_{13l}^{(2)} = \varphi_{2l} * \epsilon_{13}$ and $e_{12l}^{(2)} = \varphi_{2l} * \epsilon_{12}$.

The equation of motion in 3-D space is obtained by substituting the stress-strain relation (3.188) into Euler's differential equations (1.23),

$$\partial_{tt}^2 u_1 = \rho^{-1} (\partial_1 \sigma_{11} + \partial_2 \sigma_{12} + \partial_3 \sigma_{13} + f_1)$$

 $\partial_{tt}^2 u_2 = \rho^{-1} (\partial_1 \sigma_{12} + \partial_2 \sigma_{22} + \partial_3 \sigma_{23} + f_2)$
 $\partial_{tt}^2 u_3 = \rho^{-1} (\partial_1 \sigma_{13} + \partial_2 \sigma_{23} + \partial_3 \sigma_{33} + f_3),$
(3.194)

and making use of the strain-displacement relations (1.2)

$$\epsilon_{ij} = \frac{1}{2} (\partial_i u_j + \partial_j u_i). \qquad (3.195)$$

In 2-D space and in the (x, z)-plane, all the derivatives ∂_2 vanish, u_2 is constant, and we should consider the first and third equations in (3.194). Applications of this modeling algorithm to compute the seismic response of reservoir models can be found in Kang and McMechan (1993), where Q effects are shown to be significant in both surface and offset vertical seismic profile data.

Assuming $L_1 = L_2$ and grouping the memory variables in the equation for each displacement component, the number of memory variables can be reduced to 2 in 2-D space and 3 in 3-D space (Xu and McMechan, 1995). Additional memory-storage savings can be achieved by setting $\tau_{ol}^{(1)} = \tau_{ol}^{(2)}$ (Emmerich and Korn, 1987). To further reduce storage, only a single relaxation time can be assigned to each grid point if a direct method is used to solve the viscoacoustic equation of motion (Day, 1998). A suitable spatial distribution of these relaxation times simulates the effects of the full relaxation spectrum.

125

3.10 Analytical solutions

Analytical solutions are useful to study the physics of wave propagation and test numerical modeling algorithms. They are essential in anelastic wave simulation to distinguish between numerical dispersion – due to the time and space discretization – and physical velocity dispersion. As stated in Section 3.6, if the elastic solution is available in explicit form in the frequency domain, the viscoelastic solution can be obtained by using the correspondence principle, that is, replacing the elastic moduli or the wave velocities by the corresponding complex viscoelastic moduli and velocities. The time-domain solution is generally obtained by an inverse Fourier transform and, therefore, is a semi-analytical solution. In very simple cases, such as the case of wave propagation in a semi-infinite rod represented by a Maxwell model, a closed-form time-domain solution can be obtained (Christensen, 1982, p. 190).

3.10.1 Viscoacoustic media

We start with the frequency-domain Green's function for acoustic (dilatational) media and apply the correspondence principle. To obtain the Green function $G(x, z, x_0, z_0, t)$ for a 2-D acoustic medium, we need to solve the inhomogeneous scalar wave equation

$$\Delta G - \frac{1}{c_a^2} \partial_{tt}^2 G = -4\pi \delta(x - x_0) \delta(z - z_0) \delta(t), \qquad (3.196)$$

where x and z are the receiver coordinates, x_0 and z_0 are the source coordinates, and c_a is the acoustic-wave velocity. The solution to equation (3.196) is given by

$$G(x, z, x_0, z_0, t) = 2H\left(t - \frac{r}{c_a}\right)\left(t^2 - \frac{r^2}{c_a^2}\right)^{-1/2},$$
 (3.197)

where

$$r = \sqrt{(x - x_0)^2 + (z - z_0)^2},$$
 (3.198)

and H is Heaviside's function (Morse and Feshbach, 1953, p. 1363; Bleistein, 1984, p. 65). Taking a Fourier transform with respect to time, equation (3.197) gives

$$G(x, z, x_0, z_0, \omega) = 2 \int_{r/c_a}^{\infty} \left(t^2 - \frac{r^2}{c_a^2}\right)^{-1/2} \exp(-i\omega t) dt.$$
 (3.199)

By making a change of variable $\tau = c_a(t/r)$, equation (3.199) becomes

$$G(x, z, x_0, z_0, \omega) = 2 \int_1^\infty (\tau^2 - 1)^{-1/2} \exp\left(-\frac{i\omega}{c_0}\tau\right) d\tau.$$
 (3.200)

This expression is the integral representation of the zero-order Hankel function of the second kind (Morse and Feshbach, 1953, p. 1362):

$$G(x, z, x_0, z_0, \omega) = -i\pi H_0^{(2)} \left(\frac{\omega r}{c_a}\right).$$
 (3.201)

Using the correspondence principle, we replace the acoustic-wave velocity c_a by the complex velocity $v_c(\omega)$, which is equivalent to replacing the acoustic bulk modulus ρc_a^2 by the complex modulus $M(\omega) = \rho v_c^2(\omega)$. Then, the viscoacoustic Green's function is

 $G(x, z, x_0, z_0, \omega) = -i\pi H_0^{(2)} \left[\frac{\omega r}{v_e(\omega)} \right].$ (3.202)

We set

$$G(-\omega) = G^{*}(\omega).$$
 (3.203)

This equation ensures that the inverse Fourier transform of the Green function is real.

For the dilatational field, for instance (see Section 2.7.4), the frequency-domain solution is given by

$$\epsilon(\omega) = G(\omega)F(\omega),$$
 (3.204)

where $F(\omega)$ is the time Fourier transform of the source wavelet.

A wavelet representative of typical seismic pulses is given by equations (2.233) and (2.234). Because the Hankel function has a singularity at $\omega = 0$, we assume G = 0 for $\omega = 0$, an approximation that has no significant effect on the solution. (Note, moreover, that F(0) is small). The time-domain solution $\epsilon(t)$ is obtained by a discrete inverse Fourier transform. We have tacitly assumed that ϵ and $\partial_t \epsilon$ are zero at time t = 0.

3.10.2 Constant-Q viscoacoustic media

Let us consider the Green function problem in anelastic viscoacoustic media, based on the constant-Q model (Section 2.5). Equation (2.220) can be solved in terms of the Green function, which is obtained from

$$\Delta G - \frac{(i\omega)^{\beta}}{b}G = -4\pi\delta(x - x_0)\delta(z - z_0). \qquad (3.205)$$

Let us define the quantity

$$Ω = -i(iω)^{β/2}.$$
(3.206)

Expressing equation (3.205) in terms of this quantity gives the Helmholtz equation

$$\Delta G + \left(\frac{\Omega}{\sqrt{b}}\right)^2 G = -4\pi\delta(x - x_0)\delta(z - z_0). \qquad (3.207)$$

The solution to this equation is the zero-order Hankel function of the second kind (Morse and Feshbach, 1953, p. 1362),

$$G(x, z, x_0, z_0, \omega) = -i\pi H_0^{(2)} \left(\frac{\Omega r}{\sqrt{b}}\right),$$
 (3.208)

where r is given in equation (3.198). An alternative approach is to use the correspondence principle and replace the elastic wave velocity c_a in equation (3.201) by the complex velocity (2.213). When $\beta = 2$, we obtain the classical solution for the Green function in an acoustic medium (equation (3.201)). We require the condition (3.203) that ensures a real Green's function. The frequency-domain solution is given by

$$w(\omega) = G(\omega)F(\omega),$$
 (3.209)

where F is the Fourier transform of the source. As before, we assume G = 0 for $\omega = 0$ in order to avoid the singularity. The time-domain solution w(t) is obtained by a discrete inverse Fourier transform.

We have seen in Section 2.5.2 that constant-Q propagation is governed by an evolution equation based on fractional derivates. Mainardi and Tomirotti (1997) obtained the fundamental solutions for the 1-D version of equation (2.220) in terms of entire functions of the Wright type. Let us consider this equation and define $\beta = 2\eta$. Mainardi and Tomirotti (1997) define the signalling problem as

$$\frac{\partial^{2\eta}w}{\partial t^{2\eta}} = b\partial_1^2 w, \quad w(x,0^+) = 0, (x > 0); \quad w(0^+,t) = \delta(t), \quad w(+\infty,t) = 0, (t > 0). \quad (3.210)$$

The corresponding Green's function can be written as

$$G(x,t) = \frac{\eta x}{\sqrt{b}t^{1+\eta}} W_{-\eta,1-\eta}(-\bar{x}), \qquad \bar{x} = \frac{x}{\sqrt{b}t^{\eta}}, \qquad (3.211)$$

where

$$W_{q,r}(\bar{x}) = \sum_{k=0}^{\infty} \frac{\bar{x}^k}{k! \, \Gamma(qk+r)}, \quad q > -1, \quad r > 0$$
 (3.212)

is the Wright function (Podlubny, 1999). The exponential and Bessel functions are particular cases of the Wright function (e.g., Podlubny, 1999). For instance, $W_{0,1}(\bar{x}) = \exp(\bar{x})$.

3.10.3 Viscoelastic media

The solution of the wave field generated by an impulsive point force in a 2-D elastic medium is given by Eason, Fulton and Sneddon (1956) (see also Pilant, 1979, p. 59). For a force acting in the positive z-direction, this solution can be expressed as

$$u_1(r,t) = \left(\frac{F_0}{2\pi\rho}\right) \frac{xz}{r^2} [G_1(r,t) + G_3(r,t)], \qquad (3.213)$$

$$u_3(r,t) = \left(\frac{F_0}{2\pi\rho}\right) \frac{1}{r^2} [z^2 G_1(r,t) - x^2 G_3(r,t)], \qquad (3.214)$$

where F_0 is a constant that gives the magnitude of the force, $r = \sqrt{x^2 + z^2}$,

$$G_1(r,t) = \frac{1}{c_P^2} (t^2 - \tau_P^2)^{-1/2} H(t - \tau_P) + \frac{1}{r^2} \sqrt{t^2 - \tau_P^2} H(t - \tau_P) - \frac{1}{r^2} \sqrt{t^2 - \tau_S^2} H(t - \tau_S) \quad (3.215)$$

and

$$G_3(r,t) = -\frac{1}{c_S^2} (t^2 - \tau_S^2)^{-1/2} H(t - \tau_S) + \frac{1}{r^2} \sqrt{t^2 - \tau_P^2} H(t - \tau_P) - \frac{1}{r^2} \sqrt{t^2 - \tau_S^2} H(t - \tau_S),$$
(3.216)

where

$$\tau_P = \frac{r}{c_P}, \quad \tau_S = \frac{r}{c_S} \quad (3.217)$$

and c_P and c_S are the compressional and shear phase velocities. To apply the correspondence principle, we need the frequency-domain solution. Using the transform pairs of the zero- and first-order Hankel functions of the second kind,

$$\int_{-\infty}^{\infty} \frac{1}{\tau^2} \sqrt{t^2 - \tau^2} H(t - \tau) \exp(-i\omega t) dt = \frac{i\pi}{2\omega\tau} H_1^{(2)}(\omega\tau), \quad (3.218)$$

$$\int_{-\infty}^{\infty} (t^2 - \tau^2)^{-1/2} H(t - \tau) \exp(-i\omega t) dt = -\frac{i\pi}{2} H_0^{(2)}(\omega \tau), \qquad (3.219)$$

we obtain

$$u_1(r, \omega, c_P, c_S) = \left(\frac{F_0}{2\pi\rho}\right) \frac{xz}{r^2} [G_1(r, \omega, c_P, c_S) + G_3(r, \omega, c_P, c_S)], \quad (3.220)$$

$$u_3(r,\omega,c_P,c_S) = \left(\frac{F_0}{2\pi\rho}\right) \frac{1}{r^2} [z^2 G_1(r,\omega,c_P,c_S) - x^2 G_3(r,\omega,c_P,c_S)], \quad (3.221)$$

where

$$G_1(r,\omega,c_P,c_S) = -\frac{i\pi}{2} \left[\frac{1}{c_P^2} H_0^{(2)} \left(\frac{\omega r}{c_P} \right) + \frac{1}{\omega r c_S} H_1^{(2)} \left(\frac{\omega r}{c_S} \right) - \frac{1}{\omega r c_P} H_1^{(2)} \left(\frac{\omega r}{c_P} \right) \right], \quad (3.222)$$

$$G_{3}(r,\omega,c_{P},c_{S}) = \frac{\mathrm{i}\pi}{2} \left[\frac{1}{c_{S}^{2}} H_{0}^{(2)} \left(\frac{\omega r}{c_{S}} \right) - \frac{1}{\omega r c_{S}} H_{1}^{(2)} \left(\frac{\omega r}{c_{S}} \right) + \frac{1}{\omega r c_{P}} H_{1}^{(2)} \left(\frac{\omega r}{c_{P}} \right) \right].$$
(3.223)

Using the correspondence principle, we replace the elastic wave velocities in (3.220) and (3.221) by the viscoelastic wave velocities v_P and v_S defined in (3.18). The 2-D viscoelastic Green's function can then be expressed as

$$u_1(r,\omega) = \begin{cases} u_1(r,\omega, v_P, v_S), & \omega \ge 0, \\ u_1^*(r, -\omega, v_P, v_S), & \omega < 0, \end{cases}$$
(3.224)

and

$$u_3(r, \omega) = \begin{cases} u_3(r, \omega, v_P, v_S), & \omega \ge 0, \\ u_3^*(r, -\omega, v_P, v_S), & \omega < 0. \end{cases}$$
(3.225)

Multiplication with the source time function and a numerical inversion by the discrete Fourier transform yield the desired time-domain solution (G_1 and G_3 are assumed to be zero at $\omega = 0$).

3.11 The elastodynamic of a non-ideal interface

In seismology, exploration geophysics and several branches of mechanics (for example, metallurgical defects, adhesive joints, frictional contacts and composite materials), the problem of imperfect contact between two media is of particular interest. Seismological applications include wave propagation through dry and partially saturated cracks and fractures present in the Earth's crust, which may constitute possible earthquake sources. Similarly, in oil exploration, the problem finds applications in hydraulic fracturing, where a fluid is injected through a borehole to open a fracture in the direction of the least principal stress. Active and passive seismic waves are used to monitor the position and geometry of the fracture. In addition, in material science, a suitable model of an imperfect interface is necessary, since strength and fatigue resistance can be degraded by subtle differences between microstructures of the interface region and the bulk material.

Theories that consider imperfect bonding are mainly based on the displacement discontinuity model at the interface. Pyrak-Nolte, Myer and Cook (1990) propose a non-welded interface model based on the discontinuity of the displacement and the particle velocity across the interface. The stress components are proportional to the displacement discontinuity through the specific stiffnesses, and to the particle-velocity discontinuity through the specific viscosity. Displacement discontinuities conserve energy and yield frequency dependent reflection and transmission coefficients. On the other hand, particle-velocity discontinuities imply an energy loss at the interface and frequency-independent reflection and transmission coefficients. The specific viscosity accounts for the presence of a liquid under saturated conditions. The liquid introduces a viscous coupling between the two surfaces of the fracture (Schoenberg, 1980) and enhances energy transmission. However, at the same time, energy transmission is reduced by viscous losses.

3.11.1 The interface model

Consider a planar interface in an elastic and isotropic homogeneous medium; that is, the material on both sides of the interface is the same. The non-ideal characteristics of the interface are modeled through the boundary conditions between the two half-spaces. If the displacement and the stress field are continuous across the interface (ideal or welded contact), the reflection coefficient is zero and the interface cannot be detected. However, if the half-spaces are in non-ideal contact, reflected waves with appreciable amplitude can exist. The model is based on the discontinuity of the displacement and particle velocity fields across the interface.

Let us assume in this section the two-dimensional P-SV case in the (x, z)-plane, and refer to the upper and lower half-spaces with the labels I and II, respectively. Then, the boundary conditions for a wave impinging on the interface (z = 0) are

$$[v_1] \equiv (v_1)_{II} - (v_1)_I = \psi_1 * \partial_t \sigma_{13},$$
 (3.226)

$$[v_3] \equiv (v_3)_{II} - (v_3)_I = \psi_3 * \partial_t \sigma_{33},$$
 (3.227)

$$(\sigma_{13})_I = (\sigma_{13})_{II},$$
 (3.228)

$$(\sigma_{33})_I = (\sigma_{33})_{II},$$
 (3.229)

where v_1 and v_3 are the particle-velocity components, σ_{13} and σ_{33} are the stress components, and ψ_1 and ψ_3 are relaxation-like functions of the Maxwell type governing the tangential and normal coupling properties of the interface. The relaxation functions can be expressed as

$$\psi_i(t) = \frac{1}{\eta_i} \exp(-t/\tau_i) H(t), \quad \tau_i = \frac{\eta_i}{p_i}, \quad i = 1, 3,$$
 (3.230)

where H(t) is Heaviside's function, $p_1(x)$ and $p_3(x)$ are specific stiffnesses, and $\eta_1(x)$ and $\eta_3(x)$ are specific viscosities. They have dimensions of stiffness and viscosity per unit length, respectively. In the frequency domain, equations (3.226) and (3.227) can be compactly rewritten as

$$[v_i] = M_i \sigma_{i3}, \quad i = 1, 3,$$
 (3.231)

where

$$M_i(\omega) = \mathcal{F}(\partial_t \psi_i) = \frac{i\omega}{p_i + i\omega\eta_i}$$

(3.232)

(see equation (2.147)) is a specific complex modulus having dimensions of admittance (reciprocal of impedance).

The characteristics of the medium are completed with the stress-strain relations. In isotropic media, stresses and particle velocities are related by the following equations:

$$\rho \partial_t \sigma_{11} = I_P^2 \partial_1 v_1 + (I_P^2 - 2I_S^2) \partial_3 v_3, \qquad (3.233)$$

$$\rho \partial_t \sigma_{33} = (I_P^2 - 2I_S^2) \partial_1 v_1 + I_P^2 \partial_3 v_3,$$
 (3.234)

$$\rho \partial_t \sigma_{13} = I_S^2 (\partial_1 v_3 + \partial_3 v_1),$$
 (3.235)

where $I_P = \rho c_P$ and $I_S = \rho c_S$ are the compressional and shear impedances, with c_P and c_S denoting the elastic wave velocities, respectively.

Boundary conditions in differential form

The boundary equations (3.226) and (3.227) could be implemented in a numerical solution algorithm. However, the evaluation of the convolution integrals is prohibitive when solving the differential equations with grid methods. In order to circumvent the convolutions, we recast the boundary conditions in differential form. From equations (3.226) and (3.227), and using convolution properties, we have

$$[v_i] = \partial_t \psi_i * \sigma_{i3}.$$
 (3.236)

Using equation (3.230) and after some calculations, we note that

$$[v_i] = \psi_i(0)\sigma_{i3} - \frac{1}{\tau_i}\psi_i * \sigma_{i3}.$$
 (3.237)

Since $[v_i] = \partial_t[u_i]$, where u_i is the displacement field, we can infer from equation (3.236) that

$$[u_i] = \psi_i * \sigma_{i3}.$$
 (3.238)

Then, equation (3.237) becomes

$$\partial_t[u_i] = \frac{1}{\eta_i}(\sigma_{i3} - p_i[u_i]).$$
 (3.239)

Alternatively, this equation can be written as

$$p_i[u_i] + \eta_i[v_i] = \sigma_{i3}.$$
 (3.240)

Note that $p_i = 0$ gives the displacement discontinuity model, and $\eta_i = 0$ gives the particlevelocity discontinuity model. On the other hand, if $\eta_i \rightarrow \infty$ (see equation (3.239)), the model gives the ideal (welded) interface.

3.11.2 Reflection and transmission coefficients of SH waves

The simplicity of the SH case permits a detailed treatment of the reflection and transmission coefficients, and provides some insight into the nature of energy loss in the more cumbersome P-SV problem. We assume an interface separating two dissimilar materials of shear impedances I_S^I and I_S^{II} . The theory, corresponding to a specific stiffness p_2 and a specific viscosity η_2 , satisfies the following boundary conditions:

$$(v_2)_{II} - (v_2)_I = \psi_2 * \partial_t \sigma_{23},$$
 (3.241)

$$(\sigma_{23})_I = (\sigma_{23})_{II},$$
 (3.242)

where

$$\rho \sigma_{23} = I_S^2 \partial_3 u_2,$$
 (3.243)

and u_2 is the displacement field. The relaxation function ψ_2 has the same form (3.230), where i = 2.

In half-space I, the displacement field is

$$u_2)_I = \exp[i\kappa^I(x\sin\theta + z\cos\theta)] + R_{SS}\exp[i\kappa^I(x\sin\theta - z\cos\theta)],$$
 (3.244)

where κ^{I} is the real wavenumber and R_{SS} is the reflection coefficient. In half-space II, the displacement field is

$$u_2)_{II} = T_{SS} \exp[i\kappa^{II}(x \sin \delta + z \cos \delta)],$$
 (3.245)

where T_{SS} is the transmission coefficient and

$$\delta = \arcsin\left(\frac{\kappa^{I}}{\kappa^{II}}\right)\sin\theta,$$

according to Snell's law. For clarity, the factor $exp(-i\omega t)$ has been omitted in equations (3.244) and (3.245).

Considering that $v_2 = -i\omega u_2$, the reflection and transmission coefficients are obtained by substituting the displacements into the boundary conditions. This gives

$$R_{\rm SS} = \frac{Y_I - Y_{II} + Z}{Y_I + Y_{II} + Z}, \qquad T_{\rm SS} = \frac{2Y_I}{Y_I + Y_{II} + Z},$$
 (3.246)

where

$$Y_I = I_S^I \cos \theta$$
 and $Y_{II} = I_S^{II} \cos \delta$, (3.247)

$$Z(\omega) = Y_I Y_{II} M_2(-\omega),$$
 (3.248)

and the relation $\kappa^{I(II)}I_S^{I(II)} = \rho\omega$ has been used.

Since

$$M_2(\omega) = \frac{i\omega}{p_2 + i\omega\eta_2}, \quad (3.249)$$

the reflection and transmission coefficients are frequency independent for $p_2 = 0$ and, moreover, there are no phase changes. In this case, when $\eta_2 \rightarrow 0$, $R_{SS} \rightarrow 1$ and $T_{SS} \rightarrow 0$, and the free-surface condition is obtained; when $\eta_2 \rightarrow \infty$, $R_{SS} \rightarrow 0$ and $T_{SS} \rightarrow 1$, the ideal (welded) interface is obtained

Energy loss

In a completely welded interface, the normal component of the time-averaged energy flux is continuous across the plane separating the two media. This is a consequence of the boundary conditions that impose continuity of normal stress and particle velocity. The normal component of the time-averaged energy flux is proportional to the real part of $\sigma_{23}v_2^*$. Since the media are elastic, the interference terms between different waves (see Section 6.1.7) vanish and only the fluxes corresponding to each single beam need be considered. After normalizing with respect to the incident wave, the energy fluxes of the reflected and transmitted waves are

reflected wave
$$\rightarrow |R_{SS}|^2$$
, (3.250)

transmitted wave
$$\rightarrow \frac{Y_{II}}{Y_I} |T_{SS}|^2$$
. (3.251)

The energy loss at the interface is obtained by subtracting the energies of the reflected and transmitted waves from the energy of the incident wave. The normalized dissipated energy is

$$D = 1 - |R_{SS}|^2 - \frac{Y_{II}}{Y_I} |T_{SS}|^2.$$
 (3.252)

Substituting the reflection and transmission coefficients, we note that the energy loss becomes

$$D = \frac{4Y_{II}Z_R}{(Y_I + Y_{II} + Z_R)^2 + Z_I^2}, \quad (3.253)$$

where Z_R and Z_I are the real and imaginary parts of Z, given by

$$Z_R = \frac{\omega^2 \eta_2 Y_I Y_{II}}{p_2^2 + \omega^2 \eta_2^2}, \quad \text{and} \quad Z_I = \frac{\omega p_2 Y_I Y_{II}}{p_2^2 + \omega^2 \eta_2^2}.$$
 (3.254)

If $p_2 = 0$, then $Z_I = 0$, $Z_R = Y_I Y_{II}/\eta_2$, and the energy loss is frequency independent. When $\eta_2 \rightarrow 0$ (complete decoupling) and $\eta_2 \rightarrow \infty$ (welded contact), there is no energy dissipation. If $p_2 = 0$, the maximum loss is obtained for

$$\eta_2 = \frac{Y_I Y_{II}}{Y_I + Y_{II}}.$$
 (3.255)

At normal incidence and in equal lower and upper media, this gives $\eta_2 = I_S/2$, and a (normalized) energy loss D = 0.5, i.e., half of the energy of the normally incident wave is dissipated at the interface.

3.11.3 Reflection and transmission coefficients of P-SV waves

Consider an interface separating two half-spaces with equal material properties, where the boundary conditions are given by equations (3.226)-(3.229). Application of Snell's law indicates that the angle of the transmitted wave is equal to the angle of the incident wave, and that

$$\kappa_P \sin \theta = \kappa_S \sin \alpha$$
,

where κ_P and κ_S are the real compressional and shear wavenumbers, and θ and α are the respective associated angles. The boundary conditions do not influence the emergence angles of the transmitted and reflected waves.

In terms of the dilatational and shear potentials ϕ and ψ , the displacements are given by

$$u_1 = \partial_1 \phi - \partial_3 \psi$$
, and $u_3 = \partial_3 \phi + \partial_1 \psi$, (3.256)

(Pilant, 1979, p. 45) and the stress components by

$$\sigma_{13} = \frac{I_S^2}{\rho} \left(2\partial_1 \partial_3 \phi + \partial_1 \partial_1 \psi - \partial_3 \partial_3 \psi \right), \qquad (3.257)$$

and

$$\sigma_{33} = \frac{I_P^2}{\rho} \left(\partial_1 \partial_1 \phi + \partial_3 \partial_3 \phi \right) - \frac{2I_S^2}{\rho} \left(\partial_1 \partial_1 \phi - \partial_1 \partial_3 \psi \right). \tag{3.258}$$

Consider a compressional wave incident from half-space I. Then, the potentials of the incident and reflected waves are

$$\phi^{I} = \exp[i\kappa_{P}(x \sin \theta + z \cos \theta)],$$
 (3.259)

$$\phi^R = R_{PP} \exp[i\kappa_P(x \sin \theta - z \cos \theta)],$$
 (3.260)

and

$$ψ^R = R_{PS} \exp[iκ_S(x \sin α - z \cos α)].$$
 (3.261)

In half-space II, the potentials of the transmitted wave are

$$\phi^T = T_{PP} \exp[i\kappa_P(x \sin \theta + z \cos \theta)],$$
 (3.262)

and

$$ψ^T = T_{PS} \exp[iκ_S(x \sin α + z \cos α)].$$
 (3.263)

Considering that $v_1 = -i\omega u_1$ and $v_3 = -i\omega u_3$, the solution for an incident P wave is

$$\begin{pmatrix} \sin \alpha \left(1 + 2\gamma_{1}I_{\rm SP}\cos\theta\right) & \cos \alpha + \gamma_{1}\cos 2\alpha & -\sin \alpha & \cos \alpha \\ -\gamma_{3}\cos 2\alpha - \cos \theta & \sin \theta + \gamma_{3}\sin 2\alpha & -\cos \theta & -\sin \theta \\ 2I_{\rm SP}\sin \alpha\cos\theta & \cos 2\alpha & 2I_{\rm SP}\sin\alpha\cos\theta & -\cos 2\alpha \\ -\cos 2\alpha & \sin 2\alpha & \cos 2\alpha & \sin 2\alpha \end{pmatrix}$$
$$\cdot \begin{pmatrix} R_{\rm PP} \\ R_{\rm PS} \\ T_{\rm PP} \\ T_{\rm PS} \end{pmatrix} = \begin{pmatrix} -\sin \alpha \left(1 - 2\gamma_{1}I_{\rm SP}\cos\theta\right) \\ \gamma_{3}\cos 2\alpha - \cos \theta \\ 2I_{\rm SP}\sin\alpha\cos\theta \\ \cos 2\alpha \end{pmatrix}, \qquad (3.264)$$

where $I_{SP} = I_S / I_P$,

$$\gamma_1 = I_S M_1(-\omega) = \frac{i\omega I_S}{i\omega \eta_1 - p_1}$$
 and $\gamma_3 = I_P M_3(-\omega) = \frac{i\omega I_P}{i\omega \eta_3 - p_3}$, (3.265)

and the following relations have been used:

$$I_S \kappa_S = \rho \omega$$
, $I_P \kappa_P = \rho \omega$, (3.266)

3.11 The elastodynamic of a non-ideal interface

and

$$\rho \mu = I_S^2$$
, $\rho \lambda = I_P^2 - 2I_S^2$. (3.267)

Equations (3.264), which yield the potential amplitude coefficients, were obtained by Carcione (1996a) to investigate the scattering of cracks and fractures. Chiasri and Krebes (2000) obtain similar expressions for the displacement amplitude coefficients. The multiplying conversion factor from one type of coefficient to the other is 1 for PP coefficients and I_S/I_P for PS coefficients (Aki and Richards, 1980, p. 139).

The reflection and transmission coefficients for a P wave at normal incidence are

$$R_{\rm PP} = -\left(1 + \frac{2}{\gamma_3}\right)^{-1} \qquad (3.268)$$

and

$$T_{PP} = \left(1 + \frac{\gamma_3}{2}\right)^{-1}$$
, (3.269)

respectively. If $\eta_3 = 0$, the coefficients given in Pyrak-Nolte, Myer and Cook (1990) are obtained. If, moreover, $p_3 \rightarrow 0$, $R_{\rm PP} \rightarrow -1$ and $T_{\rm PP} \rightarrow 0$, the free-surface condition is obtained; when $\eta_3 \rightarrow \infty$, $R_{\rm PP} \rightarrow 0$ and $T_{\rm PP} \rightarrow 1$, we get the solution for a welded contact. On the other hand, it can be seen that $\eta_3 = 0$ and $p_3 = \omega I_P/2$ gives $|R_{\rm PP}|^2 = 1/2$. The characteristic frequency $\omega_P \equiv 2p_3/I_P$ defines the transition from the apparently perfect interface to the apparently delaminated one.

The reflection and transmission coefficients corresponding to an incident SV wave can be obtained in the same way as for the incident P wave. In particular, the coefficients of the normally incident wave, R_{SS} and T_{SS} , have the same form as in equations (3.268) and (3.269), but γ_1 is substituted for γ_3 .

Energy loss

Following the procedure used to obtain the energy flow in the SH case, we get the following normalized energies for an incident P wave:

reflected P wave
$$\rightarrow |R_{PP}|^2$$
,
reflected S wave $\rightarrow \frac{\tan \theta}{\tan \alpha} |R_{PS}|^2$,
transmitted P wave $\rightarrow |T_{PP}|^2$,
transmitted S wave $\rightarrow \frac{\tan \theta}{\tan \alpha} |T_{PS}|^2$.
(3.270)

Hence, the normalized energy loss is

$$\mathcal{D} = 1 - |R_{\rm PP}|^2 - |T_{\rm PP}|^2 - \frac{\tan\theta}{\tan\alpha} (|R_{\rm PS}|^2 + |T_{\rm PS}|^2). \tag{3.271}$$

It can be easily shown that the amount of dissipated energy at normal incidence is

$$D = \frac{4\gamma_{3R}}{(2 + \gamma_{3R})^2 + \gamma_{3I}^2}, \quad (3.272)$$

where the subindices R and I denote real and imaginary parts, respectively. If $p_3 = 0$, the maximum loss is obtained for $\eta_3 = I_P/2$. Similarly, if $p_1 = 0$, the maximum loss for an incident SV wave occurs when $\eta_1 = I_S/2$.



Figure 3.7: Non-ideal interface in a homogeneous medium. Normal incidence reflection coefficient R_{PP} (a) and normalized energy loss D (b) at $\theta = 0$ versus normalized specific viscosity η_2/I_P . Only the particle-velocity discontinuity ($p_3 = 0$) has been considered. As $\eta_3 \rightarrow 0$, complete decoupling (freesurface condition) is obtained. As $\eta_2 \rightarrow \infty$, the contact is welded. The maximum dissipation occurs for $\eta_3 = I_P/2$.

Examples

The following example considers a crack in a homogeneous medium bounded by a free surface. The medium is a Poisson solid with compressional and shear velocities $c_P = I_P/\rho$ = 2000 m/s and $c_S = I_S/\rho = 1155$ m/s, respectively, and density $\rho = 2$ g/cm³. Figure 3.7 represents the normal incidence reflection coefficient R_{PP} (a) and the normalized energy loss (b) versus the normalized specific viscosity η_3/I_P , with $p_3 = 0$. As can be seen, the limit $\eta_3 \rightarrow 0$ gives the complete decoupled case, and the limit $\eta_3 \rightarrow \infty$ gives the welded interface, since $R_{PP} \rightarrow 0$. The maximum dissipation occurs for $\eta_3 = I_P/2$. Similar plots and conclusions are obtained for an incident SV wave, for which the maximum loss occurs when $\eta_1 = I_S/2$. It can be shown that, for any incidence angle and values of the specific stiffnesses, there is no energy loss when $\eta_3 \rightarrow 0$ and $\eta_3 \rightarrow \infty$.

In the second example, we consider two different cases. The first case has the parameters $\eta_1 = I_S/2$ and $\eta_3 = I_P/2$ and zero specific stiffnesses. Figure 3.8 represents the reflection and transmission coefficients for an incident compressional wave (a) and the energy loss (b) versus the incidence angle (equation (3.264)). As can be seen, the dissipated energy is nearly 50 % up to 80°.

The second case has the following parameters: $p_1 = \pi f_0 I_S$, $p_3 = \pi f_0 I_P$, $\eta_1 = I_S/100$ and $\eta_3 = I_P/100$, where $f_0 = 11$ Hz. The model is practically based on the discontinuity of the displacement field. Figure 3.9 represents the reflection and transmission coefficients for an incident compressional wave versus the incidence angle. In this case, the energy loss is nearly 2 % of the energy of the incident wave.

Figure 3.10 shows a snapshot of the vertical particle velocity v_3 when the crack surface satisfies stress-free boundary conditions. Energy is conserved and there is no transmission through the crack. Two Rayleigh waves, traveling along the crack plane, can be appreciated.



Figure 3.8: Non-ideal interface in a homogeneous medium. Reflection and transmission coefficients (a) and normalized energy loss D (b) versus incidence angle θ for a fracture defined by the following specific stiffnesses and viscosities: $p_1 = p_3 = 0$, $\eta_1 = I_S/2$ and $\eta_3 = I_P/2$.



Figure 3.9: Reflection and transmission coefficients versus incidence angle θ for a non-ideal interface defined by the following specific stiffnesses and viscosities: $p_1 = \pi f_0 I_S$ and $p_3 = \pi f_0 I_P$, and $\eta_1 = I_S/100$ and $\eta_3 = I_P/100$, where $f_0 = 11$ Hz.



Figure 3.10: Vertical surface load radiation and crack scattering. The snapshot shows the v_3 -component at 1.4 ms. "R" denotes the Rayleigh wave, "P" the compressional wave, "S" the shear wave, and "dP" and "dS" the compressional and shear waves diffracted by the crack tips, respectively. The size of the model is 75 × 30 cm, and the source central frequency is 110 kHz. The crack is at 14.6 cm from the surface and is 14.4 cm in length. The specific stiffnesses and viscosities of the crack are zero, implying a complete decoupling of the crack surfaces (Carcione, 1996a).

Chapter 4

Anisotropic anelastic media

... a single system of six mutually orthogonal types [strains] may be determined for any homogeneous elastic solid, so that its potential energy when homogeneously strained in any way is expressed by the sum of the products of the squares of the components of the strain, according to those types, respectively multiplied by six determinate coefficients [eigenstiffnesses]. The six strain-types thus determined are called the Six Principal Strain-types of the body. The coefficients ... are called the six Principal Elasticities of the body. If a body be strained to any of its six Principal Types, the stress required to hold it so is directly concurrent with [proportional to] the strain.

Lord Kelvin (Kelvin, 1856)

The so-called Neumann's principle (Neumann, 1885; Nye, 1987, p. 20) states, roughly speaking, that the symmetry of the consequences is at least as high as that of the causes. This implies that any kind of symmetry possessed by wave attenuation must be present within the crystallographic class of the material. This symmetry principle was clearly stated in 1884 by Pierre Curie in an article published in the *Bulletin de la Société Minéralogique de France*.

The quality factor or the related attenuation factor, which can be measured experimentally by various techniques (Toksöz and Johnston, 1981), quantifies dissipation in a given direction. Most experimental data about anisotropic attenuation are obtained in the laboratory at ultrasonic frequencies, but are not usually collected during seismic surveys. This lack of actual seismic data constitutes a serious problem because, unlike the slownesses, the attenuation behavior observed at ultrasonic frequency ranges cannot be extrapolated to the sonic and seismic ranges, since the mechanisms of dissipation can differ substantially in different frequency ranges.

Hosten, Deschamps and Tittmann (1987) measure the dependence of attenuation with propagation direction in a carbon-epoxy composite. They find that, in a sense, attenuation is more anisotropic than slowness, and while shear-wave dissipation is larger than longitudinal dissipation in the isotropy planes, the opposite behavior occurs in planes containing the axis of rotational symmetry. Arts, Rasolofosaon and Zinszner (1992) obtain the viscoelastic tensor of dry and saturated rock samples (sandstone and limestone). Their results indicate that attenuation in dry rocks is one order of magnitude lower than attenuation in saturated samples. Moreover, the attenuation is again more anisotropic than the slowness, a fact that Arts, Rasolofosaon and Zinszner interpret as attenuation having lower symmetry than the slowness, or, alternatively, a consequence of experimental error. According to Baste and Audoin (1991), the elastic stiffnesses are quite adequate to describe the closing of cracks – provided that the proper experimental techniques are employed. On the other hand, laboratory data obtained by Yin (1993) on prestressed rocks suggest that attenuation may be more sensitive to the closing of cracks than the elastic stiffnesses, and that its symmetry is closely related to the type of loading. Yin finds a simple relation between wave amplitude and loading stress, and concludes that accurate estimates of wave attenuation can be used to quantify stress-induced anisotropy.

Since attenuation can be explained by many different mechanisms, it is difficult, if not impossible, to build a general microstructural theory. A phenomenological theory, such as viscoelasticity, leads to a convenient model. Although such a model does not allow us to predict attenuation levels, it can be used to estimate the anisotropy of attenuation. The problem is the determination of the time (or frequency) dependence of the relaxation tensor – 21 components in triclinic media. Most applications use the Kelvin-Voigt constitutive law, based on 21 independent viscosity functions (Lamb and Richter, 1966; Auld, 1990a, p. 101), corresponding to complex constants in the frequency domain. Occasionally, it has been possible to estimate all these constants satisfactorily (Hosten, Deschamps and Tittmann, 1987). This chapter presents alternative models based on fewer parameters, which are not the imaginary elasticity constants in themselves, but real quality factors – often more readily available in seismic practice. Moreover, we give a detailed description of the physical properties and energy associated with wave propagation in anisotropic anelastic media.

4.1 Stress-strain relations

Attenuation is a characteristic associated with a deformation state of the medium (e.g., a wave mode) and, therefore, a small number of parameters should suffice to obtain the relaxation components. In isotropic media, two – dilatational and shear – relaxation functions completely define the anelastic properties. For finely layered media, Backus averaging is a physically sound approach for obtaining the relaxation components of a transversely isotropic medium (referred to below as model 1; Carcione (1992c)). Two alternative constitutive laws (Carcione and Cavallini, 1994b, 1995d), not restricted to layered media, as is the Backus approach, relate waves and deformation modes to anelastic processes, using at most six relaxation functions. These laws are referred to as models 2 and 3.

We have seen in Section 2.1 (see equation (2.9)) that the stress-strain relation for an isothermal, anisotropic viscoelastic medium can be written as

$$\sigma_{ij}(\mathbf{x}, t) = \psi_{ijkl}(\mathbf{x}, t) * \partial_t \epsilon_{kl}(\mathbf{x}, t).$$
 (4.1)

Using the shortened Voigt's notation, we note that

$$\sigma = \Psi * \partial_t e$$
 (4.2)

((equation (2.22)). Time-harmonic fields are represented by the real part of

$$[\cdot] \exp(i\omega t)$$
, (4.3)

4.1 Stress-strain relations

where [·] represents a complex vector that depends only on the spatial coordinates. Substituting the time dependence (4.3) into the stress-strain relations (4.2), we obtain

$$\sigma = \mathbf{P} \cdot \mathbf{e}, \quad (\sigma_I = p_{IJ} e_J),$$
 (4.4)

where

$$p_{IJ} = \int_{-\infty}^{\infty} \partial_t \psi_{IJ}(t) \exp(-i\omega t) dt \qquad (4.5)$$

are the components of the stiffness matrix $P(\mathbf{x}, \omega)$. For anelastic media, the components of \mathbf{P} are complex and frequency dependent. Note that the anelastic stress-strain relation discussed by Auld (1990a, p. 87) is a particular case of (4.4). Auld introduces a viscosity matrix $\boldsymbol{\eta}$ such that $\mathbf{P}(\omega) = \mathbf{C} + i\omega\boldsymbol{\eta}$, with \mathbf{C} being the low-frequency limit elasticity matrix. This equation corresponds to a Kelvin-Voigt stress-strain relation (see equation (2.161)).

We can use any complex moduli, satisfying the conditions listed in Section 2.2.5, to describe the anelastic properties of the medium. The simplest realistic model is a single Zener element (see Section 2.4.3) describing each anelastic deformation mode (identified by the index ν), whose (dimensionless) complex moduli can be expressed as

$$M_{\nu}(\omega) = \frac{\sqrt{Q_{0\nu}^2 + 1} - 1 + i\omega Q_{0\nu}\tau_0}{\sqrt{Q_{0\nu}^2 + 1} + 1 + i\omega Q_{0\nu}\tau_0}, \quad (4.6)$$

where the parameterization (2.200) and (2.202) is used. We shall see that depending on the symmetry class, the subscript ν goes from 1 to 6 at most. The quality factor Q_{ν} , associated with each modulus, is equal to the real part of M_{ν} divided by its imaginary part (see equation (2.120)). At $\omega_0 = 1/\tau_0$, the curve $Q_{\nu}(\omega)$ has its lowest value: $Q_{\nu}(\omega_0)$ $= Q_{0\nu}$. The high-frequency limit corresponds to the elastic case, with $M_{\nu} \rightarrow 1$. Other complex moduli, other than (4.6), may also be appropriate, depending on the desired frequency dependence of attenuation¹.

Let us denote by c_{IJ} the elastic (or unrelaxed) stiffness constants. Then, $p_{IJ}(\omega \rightarrow \infty) = c_{IJ}$. Hooke's Law can be written either in the Voigt's notation as

$$\begin{pmatrix} \sigma_{11} \\ \sigma_{22} \\ \sigma_{33} \\ \sigma_{23} \\ \sigma_{13} \\ \sigma_{12} \end{pmatrix} = \begin{pmatrix} p_{11} & p_{12} & p_{13} & p_{14} & p_{15} & p_{16} \\ p_{12} & p_{22} & p_{23} & p_{24} & p_{25} & p_{26} \\ p_{13} & p_{23} & p_{33} & p_{34} & p_{35} & p_{36} \\ p_{14} & p_{24} & p_{34} & p_{44} & p_{45} & p_{46} \\ p_{15} & p_{25} & p_{35} & p_{45} & p_{55} & p_{56} \\ p_{16} & p_{26} & p_{36} & p_{46} & p_{56} & p_{66} \end{pmatrix} \cdot \begin{pmatrix} \epsilon_{11} \\ \epsilon_{22} \\ \epsilon_{33} \\ 2\epsilon_{23} \\ 2\epsilon_{13} \\ 2\epsilon_{12} \end{pmatrix}$$
(4.7)

or in "Kelvin's notation" - required by model 2 below - as

$$\begin{pmatrix} \sigma_{11} \\ \sigma_{22} \\ \sigma_{33} \\ \sqrt{2}\sigma_{23} \\ \sqrt{2}\sigma_{13} \\ \sqrt{2}\sigma_{12} \end{pmatrix} = \begin{pmatrix} p_{11} & p_{12} & p_{13} & \sqrt{2}p_{14} & \sqrt{2}p_{15} & \sqrt{2}p_{16} \\ p_{12} & p_{22} & p_{23} & \sqrt{2}p_{24} & \sqrt{2}p_{25} & \sqrt{2}p_{26} \\ p_{13} & p_{23} & p_{33} & \sqrt{2}p_{34} & \sqrt{2}p_{35} & \sqrt{2}p_{36} \\ \sqrt{2}p_{14} & \sqrt{2}p_{24} & \sqrt{2}p_{34} & 2p_{44} & 2p_{45} & 2p_{46} \\ \sqrt{2}p_{15} & \sqrt{2}p_{25} & \sqrt{2}p_{35} & 2p_{45} & 2p_{55} & 2p_{56} \\ \sqrt{2}p_{16} & \sqrt{2}p_{26} & \sqrt{2}p_{36} & 2p_{46} & 2p_{56} & 2p_{66} \end{pmatrix} \cdot \begin{pmatrix} \epsilon_{11} \\ \epsilon_{22} \\ \epsilon_{33} \\ \sqrt{2}\epsilon_{23} \\ \sqrt{2}\epsilon_{13} \\ \sqrt{2}\epsilon_{12} \end{pmatrix},$$
(4.8)

¹Use of the Kelvin-Voigt and constant-Q models require us to define the elastic case at a reference frequency, since the corresponding phase velocities tend to infinite at infinite frequency. (Mehrabadi and Cowin, 1990; Helbig, 1994, p. 406), where the p_{IJ} are functions of c_{IJ} and M_{ν} . The three arrays in equation (4.8) are true tensors in 6-D space, while in equation (4.7) they are just arrays (Helbig, 1994, p. 406).

4.1.1 Model 1: Effective anisotropy

In Section 1.5, we showed that fine layering on a scale much finer than the dominant wavelength of the signal yields effective anisotropy (Backus, 1962). Carcione (1992c) uses this approach and the correspondence principle (see Section 3.6) to study the anisotropic characteristics of attenuation in viscoelastic finely layered media. In agreement with the theory developed in Sections 3.1 and 3.2, let each medium be isotropic and anelastic with complex Lamé parameters given by

$$\lambda(\omega) = \rho \left(c_P^2 - \frac{4}{3}c_S^2\right) M_1(\omega) - \frac{2}{3}\rho c_S^2 M_2(\omega) \quad \text{and} \quad \mu(\omega) = \rho V_S^2 M_2(\omega), \tag{4.9}$$

(see Section 1.5), or

$$\mathcal{K} = \lambda + \frac{2}{3}\mu$$
, and $\mathcal{E} = \mathcal{K} + \frac{4}{3}\mu$, (4.10)

where M_1 and M_2 are the dilatational and shear complex moduli, respectively, c_P and c_S are the elastic high-frequency limit compressional and shear velocities, and ρ is the density. (In the work of Carcione (1992c), the relaxed moduli correspond to the elastic limit.) According to equation (1.188), the equivalent transversely isotropic medium is defined by the following complex stiffnesses:

$$p_{11} = \langle \mathcal{E} - \lambda^2 \mathcal{E}^{-1} \rangle + \langle \mathcal{E}^{-1} \rangle^{-1} \langle \mathcal{E}^{-1} \lambda \rangle^2$$

$$p_{33} = \langle \mathcal{E}^{-1} \rangle^{-1}$$

$$p_{13} = \langle \mathcal{E}^{-1} \rangle^{-1} \langle \mathcal{E}^{-1} \lambda \rangle$$

$$p_{55} = \langle \mu^{-1} \rangle^{-1}$$

$$p_{66} = \langle \mu \rangle,$$
(4.11)

where $\langle \cdot \rangle$ denotes the thickness weighted average. In the case of a periodic sequence of two alternating layers, equations (4.11) are similar to those of Postma (1955).

4.1.2 Model 2: Attenuation via eigenstrains

We introduce now a stress-strain relation based on the fact that each eigenvector (called eigenstrain) of the stiffness tensor defines a fundamental deformation state of the medium (Kelvin, 1856; Helbig, 1994, p. 399). The six eigenvalues – called eigenstiffnesses – represent the genuine elastic parameters. For example, in the elastic case, the strain energy is uniquely parameterized by the six eigenstiffnesses. From this fact and the correspondence principle (see Section 3.6), we infer that in a real medium the rheological properties depend essentially on six relaxation functions, which are the generalization of the eigenstiffnesses to the viscoelastic case. The existence of six or less complex moduli depends on the symmetry class of the medium. This theory is developed in the work of Carcione and Cavallini (1994b). According to this approach, the principal steps in the construction of a viscoelastic rheology from a given elasticity tensor C are the following:

4.1 Stress-strain relations

1. Decompose the elasticity tensor, i.e., expressed in Kelvin's notation, as

$$\mathbf{C} = \sum_{I=1}^{6} \Lambda_{I} \mathbf{e}_{I} \otimes \mathbf{e}_{I}, \qquad (4.12)$$

where Λ_I and \mathbf{e}_I are the eigenvalues and normalized eigenvectors of \mathbf{C} , respectively; Λ_I and \mathbf{e}_I are real, because \mathbf{C} is a symmetric matrix.

 Invoke the correspondence principle to obtain a straightforward viscoelastic generalization of the above equation for time-harmonic motions of angular frequency ω,

$$\mathbf{P} = \sum_{I=1}^{6} \Lambda_{I}^{(v)} \mathbf{e}_{I} \otimes \mathbf{e}_{I}, \qquad \Lambda_{I}^{(v)} = \Lambda_{I} M_{I}(\omega), \qquad (4.13)$$

where $M_I(\omega)$ are complex moduli, for instance, of the form (4.6). By construction, the eigenstiffnesses of **P** are complex, but the eigenstrains are the same as those of **C** and, hence, real.

The eigenstiffness and eigenstrains of materials of lower symmetry are given by Mehrabadi and Cowin (1990). The eigentensors may be represented as 3×3 symmetric matrices in 3-D space; therein, their eigenvalues are invariant under rotations and describe the magnitude of the deformation. Furthermore, their eigenvectors describe the orientation of the eigentensor in a given coordinate system. For instance, pure volume dilatations correspond to eigenstrains with three equal eigenvalues, and the trace of an isochoric eigenstrain is zero. Isochoric strains with two equal eigenvalues but opposite signs and a third eigenvalue of zero are plane shear tensors. To summarize, the eigentensors identify preferred modes of deformation associated with the particular symmetry of the material. An illustrative pictorial representation of these modes or eigenstrains has been designed by Helbig (1994, p. 451).

A given wave mode is characterized by its proper complex effective stiffness. This can be expressed and, hence, defined in terms of the complex eigenstiffnesses. For example, let us consider an isotropic viscoelastic solid. We have seen in section 1.1 that the total strain can be decomposed into the dilatational and deviatoric eigenstrains, whose eigenstiffnesses are related to the compressibility and shear moduli, respectively, the last with multiplicity five. Therefore, there are only two relaxation functions (or two complex eigenstiffnesses) in an isotropic medium: one describing pure dilatational anelastic behavior and the other describing pure shear anelastic behavior. Every eigenstress is directly proportional to its eigenstrain of identical form, the proportionality constant being the complex eigenstiffness.

For orthorhombic symmetry, the characteristic polynomial of the elasticity matrix, when in Kelvin's form, factors into the product of three linear factors and a cubic one. Therefore, eigenstiffnesses are found by resorting to Cardano's formulae. For a transversely isotropic medium, the situation is even simpler, as the characteristic polynomial factors into the product of two squared linear factors and a quadratic one. A straightforward computation then yields the independent entries of the complex stiffness matrix, in Voigt's notation, namely,

$$p_{11} = \Lambda_1^{(v)} (2 + a^2)^{-1} + \Lambda_2^{(v)} (2 + b^2)^{-1} + \Lambda_4^{(v)} / 2$$

$$p_{12} = p_{11} - \Lambda_4^{(v)}$$

$$p_{33} = a^2 \Lambda_1^{(v)} (2 + a^2)^{-1} + b^2 \Lambda_2^{(v)} (2 + b^2)^{-1}$$

$$p_{13} = a \Lambda_4^{(v)} (2 + a^2)^{-1} + b \Lambda_2^{(v)} (2 + b^2)^{-1}$$

$$p_{55} = \Lambda_3^{(v)} / 2$$

$$p_{66} = \Lambda_4^{(v)} / 2,$$
(4.14)

where

$$a = \frac{4c_{13}}{c_{11} + c_{12} - c_{33} - \sqrt{c}}, \quad b = \frac{4c_{13}}{c_{11} + c_{12} - c_{33} + \sqrt{c}}, \tag{4.15}$$

and $\Lambda_I^{(\pi)}(\omega)$, I = 1, ..., 4 are the complex and frequency-dependent eigenstiffnesses, given by

$$\Lambda_1^{(v)} = \frac{1}{2}(c_{11} + c_{12} + c_{33} + \sqrt{c})M_1$$

$$\Lambda_2^{(v)} = \frac{1}{2}(c_{11} + c_{12} + c_{33} - \sqrt{c})M_2$$

$$\Lambda_3^{(v)} = 2c_{55}M_3$$

$$\Lambda_4^{(v)} = (c_{11} - c_{12})M_4,$$
(4.16)

with

$$c = 8c_{13}^2 + (c_{11} + c_{12} - c_{33})^2.$$
 (4.17)

The two-fold eigenstiffnesses Λ_3 and Λ_4 are related to pure "isochoric" eigenstrains, i.e., to volume-preserving changes of shape only, while the single eigenstiffnesses Λ_1 and Λ_2 are related to eigenstrains that consist of simultaneous changes in volume and shape. For relatively weak anisotropy, Λ_1 corresponds to a quasi-dilatational deformation and Λ_2 to a quasi-shear deformation. Moreover, Λ_3 and Λ_4 determine the Q values of the shear waves along the principal axes. This stress-strain relation can be implemented in a time-domain modeling algorithm with the use of Zener relaxation functions and the introduction of memory variables (Robertsson and Coates, 1997). At each time step, stresses and strain must be projected on the bases of the eigenstrains. These transformations increase the required number of computations compared to the approach presented in the next section.

4.1.3 Model 3: Attenuation via mean and deviatoric stresses

We design the constitutive law in such a way that M_1 is the dilatational modulus and M_2 , M_3 and M_4 are associated with shear deformations. In this stress-strain relation (Carcione, 1990; Carcione and Cavallini, 1995d), the mean stress (i.e., the trace of the stress tensor) is only affected by the dilatational complex modulus M_1 . Moreover, the deviatoric-stress components solely depend on the shear complex moduli, denoted by M_2 , M_3 and M_4 . The trace of the stress tensor is invariant under transformations of the coordinate system. This fact assures that the mean stress depends only on M_1 in any system.

The complex stiffnesses for an orthorhombic medium are given by

$$p_{I(I)} = c_{I(I)} - \tilde{\mathcal{E}} + \tilde{\mathcal{K}}M_1 + \frac{4}{3}\bar{\mu}M_{\delta}, \quad I = 1, 2, 3,$$
 (4.18)

4.2 Wave velocities, slowness and attenuation vector

$$p_{IJ} = c_{IJ} - \bar{\mathcal{E}} + \bar{\mathcal{K}}M_1 + 2\bar{\mu}\left(1 - \frac{1}{3}M_\delta\right), \quad I, J = 1, 2, 3; \ I \neq J, \tag{4.19}$$

$$p_{44} = c_{44}M_2$$
, $p_{55} = c_{55}M_3$, $p_{66} = c_{66}M_4$, (4.20)

where

$$\bar{\mathcal{K}} = \bar{\mathcal{E}} - \frac{4}{3}\bar{\mu} \qquad (4.21)$$

and

$$\bar{\mathcal{E}} = \frac{1}{3} \sum_{I=1}^{3} c_{II}, \quad \bar{\mu} = \frac{1}{3} \sum_{I=4}^{6} c_{II}.$$
 (4.22)

The index δ can be chosen to be 2, 3 or 4. Transverse isotropy requires $M_4 = M_3 = M_2$ and $p_{66} = c_{66} + \bar{\mu}(M_2 - 1)$.

The mean stress $\bar{\sigma} = \sigma_{ii}/3$ can be expressed in terms of the mean strain $\bar{\epsilon} = \epsilon_{ii}/3$ and strain components (1.2) as

$$\bar{\sigma} = \frac{1}{3}(c_{J1} + c_{J2} + c_{J3})e_J + 3\bar{\mathcal{K}}(M_1 - 1)\bar{\epsilon}, \qquad (4.23)$$

which only depends on the dilatational complex modulus, as required above. Moreover, the deviatoric stresses are

$$\sigma_I - \bar{\sigma} = \sum_{K=1}^{3} \left(\delta_{IK} - \frac{1}{3} \right) c_{KJ} e_J + 2\bar{\mu} (M_{\delta} - 1) (e_I - \bar{\epsilon}), \quad I \le 3, \quad (4.24)$$

and

$$\sigma_I = \sum_{J=1}^{3} c_{IJ} e_J + \sum_{J=4}^{6} c_{IJ} M_{J-2} e_J, \quad I > 3, \quad (4.25)$$

which depend on the complex moduli associated with the quasi-shear mechanisms. This stress-strain relation has the advantage that the stiffnesses have a simple time-domain analytical form when using the Zener model. This permits the numerical solution of the visco-elastodynamic equations in the space-time domain (see Section 4.5). Examples illustrating the use of the three stress-strain relations are given in Carcione, Cavallini and Helbig (1998).

4.2 Wave velocities, slowness and attenuation vector

The dispersion relation for homogeneous viscoelastic plane waves has the form of the elastic dispersion relation, but the quantities involved are complex and frequency dependent. The generalization of equation (1.68) to the viscoelastic case, by using the correspondence principle (Section 3.6), can be written as

$$k^2 \mathbf{\Gamma} \cdot \mathbf{u} = \rho \omega^2 \mathbf{u}, \quad (k^2 \Gamma_{ij} u_j = \rho \omega^2 u_i),$$
 (4.26)

where

$$\Gamma = \mathbf{L} \cdot \mathbf{P} \cdot \mathbf{L}^{\top}, \quad (\Gamma_{ij} = l_{iI}p_{IJ}l_{Jj}).$$
(4.27)

145

The components of the Kelvin-Christoffel matrix Γ are given in equation (1.73), with the substitution of p_{IJ} for c_{IJ} . As in the isotropic case (see Section 3.3.1), the complex velocity is

$$v_e = \frac{\omega}{k}$$
, (4.28)

and the phase velocity is

$$\mathbf{v}_{p} = \left[\frac{\omega}{\operatorname{Re}(k)}\right] \hat{\boldsymbol{\kappa}} = \frac{\omega}{\kappa} = \left[\operatorname{Re}\left(\frac{1}{v_{c}}\right)\right]^{-1} \hat{\boldsymbol{\kappa}}.$$
(4.29)

Equation (4.26) constitutes an eigenequation

$$(\Gamma - \rho v_c^2 \mathbf{I}_3) \cdot \mathbf{u} = 0 \qquad (4.30)$$

for the eigenvalues $(\rho v_e^2)_m$ and eigenvectors $(\mathbf{u})_m$, m = 1, 2, 3. The dispersion relation is then

$$det(\Gamma - \rho v_c^2 \mathbf{I}_3) = 0, \qquad (4.31)$$

or, using (4.28) and $k_i = kl_i$,

$$F(k_1, k_2, k_3, \omega) = 0.$$
 (4.32)

The form (4.32) holds also for inhomogeneous plane waves.

The slowness, defined as the inverse of the phase velocity, is

$$\mathbf{s}_{R} = \left(\frac{1}{v_{p}}\right)\hat{\boldsymbol{\kappa}} = \operatorname{Re}\left(\frac{1}{v_{c}}\right)\hat{\boldsymbol{\kappa}},$$
(4.33)

i.e., its magnitude is the real part of the complex slowness $1/v_c$.

According to the definition (3.26), the attenuation vector for homogeneous plane waves is

$$\alpha = -\text{Im}(\mathbf{k}) = -\omega \text{ Im}\left(\frac{1}{v_c}\right)\hat{\kappa}.$$
 (4.34)

The group-velocity vector is given by (1.126). Because an explicit real equation of the form $\omega = \omega(\kappa_1, \kappa_2, \kappa_3)$ is not available in general, we need to use implicit differentiation of the dispersion relation (4.32). For instance, for the x-component,

$$\frac{\partial \omega}{\partial \kappa_1} = \left(\frac{\partial \kappa_1}{\partial \omega}\right)^{-1}$$
, (4.35)

or, because $\kappa_1 = \operatorname{Re}(k_1)$,

$$\frac{\partial \omega}{\partial \kappa_1} = \left[\operatorname{Re} \left(\frac{\partial k_1}{\partial \omega} \right) \right]^{-1}$$
(4.36)

Implicit differentiation of the complex dispersion relation (4.32) gives

$$\left(\frac{\partial F}{\partial \omega}\delta\omega + \frac{\partial F}{\partial k_1}\delta k_1\right)_{k_2,k_3} = 0.$$
 (4.37)

Then,

$$\left(\frac{\partial k_1}{\partial \omega}\right) = -\frac{\partial F/\partial \omega}{\partial F/\partial k_1},$$
 (4.38)

and similar results are obtained for the k_2 and k_3 components. Substituting the partial derivatives in equation (1.126), we can evaluate the group velocity as

$$\mathbf{v}_{g} = -\left[\operatorname{Re}\left(\frac{\partial F/\partial\omega}{\partial F/\partial k_{1}}\right)\right]^{-1}\hat{\mathbf{e}}_{1} - \left[\operatorname{Re}\left(\frac{\partial F/\partial\omega}{\partial F/\partial k_{2}}\right)\right]^{-1}\hat{\mathbf{e}}_{2} - \left[\operatorname{Re}\left(\frac{\partial F/\partial\omega}{\partial F/\partial k_{3}}\right)\right]^{-1}\hat{\mathbf{e}}_{3}, \quad (4.39)$$

which is a generalization of equation (1.130).

Finally, the velocity of the envelope of homogeneous plane waves has the same form (1.146) obtained for the anisotropic elastic case, where θ is the propagation – and attenuation – angle.

4.3 Energy balance and fundamental relations

The derivation of the energy-balance equation or Umov-Poynting theorem is straightforward when using complex notation. The basic equations for the time average of the different quantities involved in the energy-balance equation are (1.105) and (1.106). We also need to calculate the peak or maximum values of the physical quantities. We use the following property

$$[\operatorname{Re}(\mathbf{a}^{\top}) \cdot \operatorname{Re}(\mathbf{b})]_{\text{peak}} = \frac{1}{2}[|a_k||b_k|\cos(\arg(\mathbf{a}_k) - \arg(\mathbf{b}_k)) + \sqrt{|a_k||b_k||a_j||b_j|\cos(\arg(\mathbf{a}_k) + \arg(\mathbf{b}_k) - \arg(\mathbf{a}_j) + \arg(\mathbf{b}_j))]}, \quad (4.40)$$

where $|a_k|$ is the magnitude of the k-component of the field variable **a**, and implicit summation over repeated indices is assumed (Carcione and Cavallini, 1993). When, for every k, $\arg(\mathbf{a}_k) = \phi_a$ and $\arg(\mathbf{b}_k) = \phi_b$, i.e., all the components of each variable are in phase, equation (4.40) reduces to

$$[\operatorname{Re}(\mathbf{a}^{\top}) \cdot \operatorname{Re}(\mathbf{b})]_{\operatorname{peak}} = |a_k| |b_k| \cos\left(\frac{\phi_a - \phi_b}{2}\right), \qquad (4.41)$$

and if, moreover, $\mathbf{a} = \mathbf{b}$, then

$$[\operatorname{Re}(\mathbf{a}^{\top}) \cdot \operatorname{Re}(\mathbf{b})]_{\operatorname{peak}} = 2\langle \operatorname{Re}(\mathbf{a}^{\top}) \cdot \operatorname{Re}(\mathbf{a}) \rangle.$$
 (4.42)

When all the components of a are in phase,

$$(\operatorname{Re}(\mathbf{a}^{\top}) \cdot \operatorname{Re}(\mathbf{D}) \cdot \operatorname{Re}(\mathbf{a}))_{\operatorname{peak}} = 2(\operatorname{Re}(\mathbf{a}^{\top}) \cdot \operatorname{Re}(\mathbf{D}) \cdot \operatorname{Re}(\mathbf{a}))$$

$$(4.43)$$

(Carcione and Cavallini, 1993, 1995a).

For time-harmonic fields of angular frequency ω , the strain/particle-velocity relation (1.26) and the equation of momentum conservation (1.28) can be expressed as

$$i\omega \mathbf{e} = \nabla^{\top} \cdot \mathbf{v}$$
 (4.44)

and

$$\nabla \cdot \boldsymbol{\sigma} = i \omega \rho \mathbf{v} - \mathbf{f},$$
 (4.45)

where v is the particle-velocity vector.

To derive the balance equation, the dot product of the equation of motion (4.45) is first taken with $-v^*$ to give

$$-\mathbf{v}^* \cdot \nabla \cdot \boldsymbol{\sigma} = -i\omega\rho \mathbf{v}^* \cdot \mathbf{v} + \mathbf{v}^* \cdot \mathbf{f}. \qquad (4.46)$$

On the other hand, the dot product of $-\sigma^{\top}$ with the complex conjugate of (4.44) is

$$-\boldsymbol{\sigma}^{\top} \cdot \nabla^{\top} \cdot \mathbf{v}^{*} = i\omega \boldsymbol{\sigma}^{\top} \cdot \mathbf{e}^{*}.$$
 (4.47)

Adding equations (4.46) and (4.47), we get

$$-\mathbf{v}^* \cdot \nabla \cdot \boldsymbol{\sigma} - \boldsymbol{\sigma}^\top \cdot \nabla^\top \cdot \mathbf{v}^* = -\mathrm{i}\omega\rho\mathbf{v}^* \cdot \mathbf{v} + \mathrm{i}\omega\boldsymbol{\sigma}^\top \cdot \mathbf{e}^* + \mathbf{v}^* \cdot \mathbf{f}. \tag{4.48}$$

The left-hand side of (4.48) is simply

$$-\mathbf{v}^* \cdot \nabla \cdot \boldsymbol{\sigma} - \boldsymbol{\sigma}^\top \cdot \nabla^\top \cdot \mathbf{v}^* = -\operatorname{div}(\boldsymbol{\Sigma} \cdot \mathbf{v}^*),$$
 (4.49)

where Σ is the 3 × 3 stress tensor defined in equation (1.108). Then, equation (4.48) can be expressed as

$$-\frac{1}{2}\operatorname{div}(\boldsymbol{\Sigma}\cdot\mathbf{v}^{*}) = -\mathrm{i}\omega\frac{1}{2}\rho\mathbf{v}^{*}\cdot\mathbf{v} + \mathrm{i}\omega\frac{1}{2}\boldsymbol{\sigma}^{\top}\cdot\mathbf{e}^{*} + \frac{1}{2}\mathbf{v}^{*}\cdot\mathbf{f}.$$
(4.50)

After substitution of the stress-strain relation (4.4), equation (4.50) gives

$$-\frac{1}{2}\operatorname{div}(\boldsymbol{\Sigma}\cdot\mathbf{v}^*) = 2\mathrm{i}\omega\left[-\frac{1}{4}\rho\mathbf{v}^*\cdot\mathbf{v} + \frac{1}{4}\operatorname{Re}(\mathbf{e}^{\top}\cdot\mathbf{P}\cdot\mathbf{e}^*)\right] - \frac{\omega}{2}\operatorname{Im}(\mathbf{e}^{\top}\cdot\mathbf{P}\cdot\mathbf{e}^*) + \frac{1}{2}\mathbf{v}^*\cdot\mathbf{f}.$$
(4.51)

The significance of this equation becomes clear when we recognize that each of its terms has a precise physical meaning on a time-average basis. For instance, from equation (1.105),

$$\frac{1}{4}\rho \mathbf{v}^* \cdot \mathbf{v} = \frac{1}{2}\rho \langle \operatorname{Re}(\mathbf{v}) \cdot \operatorname{Re}(\mathbf{v}) \rangle = \langle T \rangle \qquad (4.52)$$

is the time-averaged kinetic-energy density; from (1.106)

$$\frac{1}{4}\operatorname{Re}(\mathbf{e}^{\top} \cdot \mathbf{P} \cdot \mathbf{e}^{*}) = \frac{1}{2} \langle \operatorname{Re}(\mathbf{e}^{\top}) \cdot \operatorname{Re}(\mathbf{P}) \cdot \operatorname{Re}(\mathbf{e}) \rangle = \langle V \rangle \qquad (4.53)$$

is the time-averaged strain-energy density, and

$$\frac{\omega}{2} \operatorname{Im}(\mathbf{e}^{\top} \cdot \mathbf{P} \cdot \mathbf{e}^{*}) = \frac{\omega}{2} \langle \operatorname{Re}(\mathbf{e}^{\top}) \cdot \operatorname{Im}(\mathbf{P}) \cdot \operatorname{Re}(\mathbf{e}) \rangle = \langle \dot{D} \rangle$$
(4.54)

is the time-averaged rate of dissipated-energy density. Because the strain energy and the rate of dissipated energies should always be positive, $\text{Re}(\mathbf{P})$ and $\text{Im}(\mathbf{P})$ must be positive definite matrices (see Holland, 1967). These conditions are the generalization of the condition of stability discussed in Section 1.2. If expressed in terms of the eigenvalues of matrix \mathbf{P} (see Section 4.1.2)), the real and imaginary parts of these eigenvalues must be positive. It can be shown that the three models introduced in Section 4.1 satisfy the stability conditions (see Carcione (1990) for a discussion of model 3).

The complex power-flow vector or Umov-Poynting vector is defined as

$$\mathbf{p} = -\frac{1}{2}\boldsymbol{\Sigma} \cdot \mathbf{v}^* \qquad (4.55)$$

and

$$p_s = \frac{1}{2} \mathbf{v}^* \cdot \mathbf{f} \qquad (4.56)$$

is the complex power per unit volume supplied by the body forces. Substituting the preceding expressions into equation (4.51), we obtain the energy-balance equation

div
$$\mathbf{p} - 2i\omega(\langle V \rangle - \langle T \rangle) + \langle D \rangle = p_s.$$
 (4.57)

The time-averaged energy density is

$$\langle E \rangle = \langle T \rangle + \langle V \rangle = \frac{1}{4} [\rho \mathbf{v}^* \cdot \mathbf{v} + \operatorname{Re}(\mathbf{e}^\top \cdot \mathbf{P} \cdot \mathbf{e}^*)].$$
 (4.58)

In lossless media, $\langle \dot{D} \rangle = 0$, and because in the absence of sources the net energy flow into or out of a given closed surface must vanish, div $\mathbf{p} = 0$. Thus, the average kinetic energy equals the average strain energy. As a consequence, the average stored energy is twice the average strain energy.

By separating the real and imaginary parts of equation (4.57), two independent and separately meaningful physical relations are obtained:

$$-\operatorname{Re}(\operatorname{div} \mathbf{p}) + \operatorname{Re}(p_s) = \langle D \rangle$$
 (4.59)

and

$$-\text{Im}(\text{div } \mathbf{p}) + \text{Im}(p_s) = 2\omega(\langle T \rangle - \langle V \rangle). \quad (4.60)$$

For linearly polarized fields, the components of the particle-velocity vector \mathbf{v} are in phase, and the average kinetic energy is half the peak kinetic energy by virtue of equation (4.42). The same property holds for the strain energy if the components of the strain array \mathbf{e} are in phase (see equation (4.43)). In this case, the energy-balance equation reads

div
$$\mathbf{p} - i\omega(\langle V \rangle_{\text{peak}} - \langle T \rangle_{\text{peak}}) + \langle \dot{D} \rangle = p_s,$$
 (4.61)

in agreement with Auld (1990a, p. 154). Equation (4.61) is found to be valid only for homogeneous viscoelastic plane waves, i.e., when the propagation direction coincides with the attenuation direction, although Auld (1990a, eq. 5.76) seems to attribute a general validity to that equation. Notably, it should be pointed out that for inhomogeneous viscoelastic plane waves, the peak value is not twice the average value. The same remark applies to Ben-Menahem and Singh (1981, p. 883).

4.3.1 Plane waves. Energy velocity and quality factor

A general solution representing inhomogeneous viscoelastic plane waves is of the form

$$[\cdot] \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})],$$
 (4.62)

where $[\cdot]$ is a constant complex vector. The wavevector is complex and can be written as in equation (3.26), with $\kappa \cdot \alpha$ strictly different from zero, unlike the interface waves in elastic media. For these plane waves, the operator (1.25) takes the form

$$\nabla \rightarrow -i\mathbf{K}$$
, (4.63)

where

$$\mathbf{K} = \begin{pmatrix} k_1 & 0 & 0 & k_3 & k_2 \\ 0 & k_2 & 0 & k_3 & 0 & k_1 \\ 0 & 0 & k_3 & k_2 & k_1 & 0 \end{pmatrix},$$
(4.64)

with k_1 , k_2 and k_3 being the components of the complex wavevector k. Note that for the corresponding conjugated fields, the operator should be replaced by iK^{*}.

Substituting the differential operator ∇ into equations (4.46) and (4.47) and assuming $\mathbf{f} = 0$, we obtain

$$\mathbf{v}^* \cdot \mathbf{K} \cdot \boldsymbol{\sigma} = -\omega \rho \mathbf{v}^* \cdot \mathbf{v}^*$$
(4.65)

and

$$\sigma^{\top} \cdot \mathbf{K}^{*\top} \cdot \mathbf{v}^{*} = -\omega \sigma^{\top} \cdot \mathbf{e}^{*},$$
 (4.66)

respectively. From equation (4.4), the right-hand side of (4.66) gives

$$-\omega \sigma^{\top} \cdot \mathbf{e}^* = -\omega \mathbf{e}^{\top} \cdot \mathbf{P} \cdot \mathbf{e}^*,$$
 (4.67)

since **P** is symmetric. The left-hand sides of (4.65) and (4.66) contain the Umov-Poynting vector (4.55) because $\mathbf{K} \cdot \boldsymbol{\sigma} = \boldsymbol{\Sigma} \cdot \mathbf{k}$ and $\mathbf{K}^* \cdot \boldsymbol{\sigma} = \boldsymbol{\Sigma} \cdot \mathbf{k}^*$; thus

$$2\mathbf{k} \cdot \mathbf{p} = \omega \rho \mathbf{v}^* \cdot \mathbf{v}$$
 (4.68)

and

$$2\mathbf{k}^* \cdot \mathbf{p} = \omega \mathbf{e}^\top \cdot \mathbf{P} \cdot \mathbf{e}^*.$$
 (4.69)

In terms of the energy densities (4.52), (4.53) and (4.54),

$$\mathbf{k} \cdot \mathbf{p} = 2\omega \langle T \rangle$$
 (4.70)

and

$$\mathbf{k}^* \cdot \mathbf{p} = 2\omega \langle V \rangle + i \langle \dot{D} \rangle.$$
 (4.71)

Because the right-hand side of (4.70) is real, the product $\mathbf{k} \cdot \mathbf{p}$ is also real. For elastic (lossless) media, \mathbf{k} and the Umov-Poynting vectors are both real quantities.

Adding equations (4.70) and (4.71) and using $\mathbf{k}^* + \mathbf{k} = 2\kappa$, with κ being the real wavevector (see equation (3.26)), we obtain

$$\kappa \cdot \mathbf{p} = \omega \langle E \rangle + \frac{i}{2} \langle \dot{D} \rangle,$$
 (4.72)

where the time-averaged energy density (4.58) has been used to obtain (4.72). Splitting equation (4.72) into real and imaginary parts, we have

$$\kappa \cdot \langle \mathbf{p} \rangle = \omega \langle E \rangle$$
(4.73)

and

$$\kappa \cdot \text{Im}(\mathbf{p}) = \frac{1}{2} \langle \dot{D} \rangle,$$
 (4.74)

where

$$\langle \mathbf{p} \rangle = \text{Re}(\mathbf{p})$$
 (4.75)

150

is the average power-flow density. The energy-velocity vector is defined as

$$\mathbf{v}_e = \frac{\langle \mathbf{p} \rangle}{\langle E \rangle} = \frac{\langle \mathbf{p} \rangle}{\langle T + V \rangle},$$
 (4.76)

which defines the location of the wave surface associated with each Fourier component, i.e., with each frequency ω . In lossy media, we define the wave front as the wave surface corresponding to infinite frequency, since the unrelaxed energy velocity is greater than the relaxed energy velocity.

Since the phase velocity is

$$\mathbf{v}_{p} = \left(\frac{\omega}{\kappa}\right) \hat{\kappa},$$
 (4.77)

where $\bar{\kappa}$ defines the propagation direction, the following relation is obtained from (4.73):

$$\hat{\kappa} \cdot v_e = v_p.$$
 (4.78)

This relation, as in the lossless case (equation (1.114)) and the isotropic viscoelastic case (equation (3.123)), means that the phase velocity is the projection of the energy velocity onto the propagation direction. Note also that equation (4.74) can be written as

$$\hat{\kappa} \cdot \mathbf{v}_d = v_p,$$
 (4.79)

where v_d is a velocity defined as

$$\mathbf{v}_{d} = \frac{2\omega \text{Im}(\mathbf{p})}{\langle \dot{D} \rangle}, \quad (4.80)$$

and associated with the rate of dissipated-energy density. Relations (4.78) and (4.79) are illustrated in Figure 4.1.



Figure 4.1: Graphical representation of equations (4.78) and (4.79). The projection of the energyvelocity vector onto the propagation direction gives the phase velocity. The same result is obtained by the projection of a pseudo-velocity vector related to the dissipated energy.

Another important relation obtained from equation (4.73) is

$$\langle E \rangle = \frac{1}{\omega} \kappa \cdot \langle \mathbf{p} \rangle,$$
 (4.81)

which means that the time-averaged energy density can be computed from the component of the average power-flow vector along the propagation direction.

Subtracting (4.71) from (4.70), we get

$$-2\alpha \cdot \mathbf{p} = 2i\omega(\langle V \rangle - \langle T \rangle) - \langle D \rangle,$$
 (4.82)

which can also be deduced from the energy-balance equation (4.57), since for plane waves of the form (4.62), div $\mathbf{p} = -2\boldsymbol{\alpha} \cdot \mathbf{p}$. Taking the real part of (4.82), we have

$$\langle \hat{D} \rangle = 2 \alpha \cdot \langle \mathbf{p} \rangle,$$
 (4.83)

which states that the time average of the rate of dissipated-energy density can be obtained from the projection of the average power-flow vector onto the attenuation direction. Relations (4.81) and (4.83) are illustrated in Figure 4.2.



Figure 4.2: Graphical representation of equations (4.81) and (4.83). The time-averaged energy density can be calculated as the component of the average power-flow vector onto the propagation direction, while the time average of the rate of dissipated-energy density depends on the projection of the average power-flow vector onto the attenuation direction.

We define the quality factor as in the 1-D and isotropic cases (equations (2.119) and (3.126), respectively); that is

$$Q = \frac{2\langle V \rangle}{\langle D \rangle}, \qquad (4.84)$$

152

where

$$\langle D \rangle \equiv \omega^{-1} \langle \dot{D} \rangle$$
 ($\omega > 0$) (4.85)

is the time-averaged dissipated-energy density. Substituting the time-averaged strainenergy density (4.53) and the time-averaged dissipated energy (4.85) into equation (4.84) and using (4.54), we obtain

$$Q = \frac{\text{Re}(\mathbf{e}^{\top} \cdot \mathbf{P} \cdot \mathbf{e}^{*})}{\text{Im}(\mathbf{e}^{\top} \cdot \mathbf{P} \cdot \mathbf{e}^{*})}. \quad (4.86)$$

This equation requires the calculation of $\mathbf{e}^{\top} \cdot \mathbf{P} \cdot \mathbf{e}^*$.

For homogeneous plane waves and using equation (1.26), we obtain

$$\mathbf{e} = -\mathbf{i}k\mathbf{L}^{\top} \cdot \mathbf{u}$$
 (4.87)

and

$$e^* = ik^* L^\top \cdot u^*$$
, (4.88)

where L is defined in equation (1.67). Replacing these expressions in $e^{\top} \cdot P \cdot e^*$, we get

$$\mathbf{e}^{T} \cdot \mathbf{P} \cdot \mathbf{e}^{*} = |k|^{2} \mathbf{u} \cdot \mathbf{\Gamma} \cdot \mathbf{u}^{*},$$
 (4.89)

where Γ is the Kelvin-Christoffel matrix (4.27). But from the transpose of (4.30),

$$\mathbf{u} \cdot \mathbf{\Gamma} = \rho v_e^2 \mathbf{u}.$$
 (4.90)

Therefore, the substitution of this expression into (4.89) gives

$$\mathbf{e}^{\top} \cdot \mathbf{P} \cdot \mathbf{e}^{*} = \rho |k|^{2} v_{e}^{2} \mathbf{u} \cdot \mathbf{u}^{*} = \rho |k|^{2} v_{e}^{2} |\mathbf{u}|^{2}.$$
(4.91)

Consequently, substituting this expression into equation (4.86), the quality factor for homogeneous plane waves in anisotropic viscoelastic media takes the following simple form as a function of the complex velocity:

$$Q = \frac{\text{Re}(v_c^2)}{\text{Im}(v_c^2)}.$$
 (4.92)

The relation (3.128) and the approximation (3.129), obtained for isotropic media, are also valid in this case. Similarly, because for a homogeneous wave $k^2 = \kappa^2 - \alpha^2 - 2i\kappa\alpha$, it follows from (4.34) and (3.126) that the quality factor relates to the wavenumber and attenuation vectors as

$$\boldsymbol{\alpha} = \left(\sqrt{Q^2 + 1} - Q\right) \boldsymbol{\kappa}. \tag{4.93}$$

For low-loss solids, the quality factor is $Q \gg 1$, and a Taylor expansion yields

$$\alpha = \frac{1}{2Q}\kappa$$
, (4.94)

which is equivalent to equation (3.129).

4.3.2 Polarizations

We have shown in Section 3.3.4, that in isotropic media the polarizations of P and S-I homogeneous planes waves can be orthogonal under certain conditions. In anisotropic anelastic media, the symmetry of the Kelvin-Christoffel matrix Γ (equation (4.27)) implies the orthogonality – in the complex sense – of the eigenvectors associated with the three homogeneous plane-wave modes. (This can be shown by using the same steps followed in Section 1.3.3). Let as assume that Γ has three distinct eigenvalues and denote two of the corresponding eigenvectors by \mathbf{u}_{s} and \mathbf{u}_{b} . Orthogonality implies

$$u_a \cdot u_b = 0$$
, (4.95)

or

$$\operatorname{Re}(\mathbf{u}_{a}) \cdot \operatorname{Re}(\mathbf{u}_{b}) - \operatorname{Im}(\mathbf{u}_{a}) \cdot \operatorname{Im}(\mathbf{u}_{b}) = 0.$$
 (4.96)

This condition does not imply orthogonality of the polarizations, i.e., $Re(\mathbf{u}_a) \cdot Re(\mathbf{u}_b) \neq 0$.

The real displacement vector of an inhomogeneous plane wave can be expressed as

$$\operatorname{Re}(\mathbf{u}) = U_0 \operatorname{Re}\{\mathbf{U} \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})]\},$$
 (4.97)

where U_0 is a real quantity, and U can be normalized in the Hermitian sense; that is

$$U \cdot U^* = 1.$$
 (4.98)

Decomposing the complex vectors into their real and imaginary parts and using $\mathbf{k} = \kappa - i\alpha$, we obtain:

$$\operatorname{Re}(\mathbf{u}) = U_0 \exp(-\boldsymbol{\alpha} \cdot \mathbf{x})[\operatorname{Re}(\mathbf{U}) \cos \varsigma - \operatorname{Im}(\mathbf{U}) \sin \varsigma],$$
 (4.99)

where

$$\varsigma = \omega t - \kappa \cdot \mathbf{x}.$$
 (4.100)

The displacement vector describes an ellipse homothetic² to the ellipse defined by

$$\mathbf{w} = \operatorname{Re}(\mathbf{U}) \cos \varsigma - \operatorname{Im}(\mathbf{U}) \sin \varsigma.$$
 (4.101)

Let us consider two displacement vectors \mathbf{w}_a and \mathbf{w}_b associated with two different wave modes at the same time and at the same frequency. The scalar product between those displacements is

$$\mathbf{w}_{a} \cdot \mathbf{w}_{b} = \operatorname{Re}(\mathbf{U}_{a}) \cdot \operatorname{Re}(\mathbf{U}_{b}) \cos(\varsigma_{a}) \cos(\varsigma_{b}) + \operatorname{Im}(\mathbf{U}_{a}) \cdot \operatorname{Im}(\mathbf{U}_{b}) \sin(\varsigma_{a}) \sin(\varsigma_{b})$$

 $-\operatorname{Im}(\mathbf{U}_{a}) \cdot \operatorname{Re}(\mathbf{U}_{b}) \sin(\varsigma_{a}) \cos(\varsigma_{b}) - \operatorname{Re}(\mathbf{U}_{a}) \cdot \operatorname{Im}(\mathbf{U}_{b}) \cos(\varsigma_{a}) \sin(\varsigma_{b}).$ (4.102)

Using the condition (4.95), which holds for homogeneous waves, equation (4.102) simplifies to

$$\mathbf{w}_{a} \cdot \mathbf{w}_{b} = \operatorname{Re}(\mathbf{U}_{a}) \cdot \operatorname{Re}(\mathbf{U}_{b}) \cos(\varsigma_{a} - \varsigma_{b}) + \operatorname{Re}(\mathbf{U}_{a}) \cdot \operatorname{Im}(\mathbf{U}_{b}) \sin(\varsigma_{a} - \varsigma_{b}),$$
 (4.103)

but it is not equal to zero. In general, the planes of the three elliptical polarizations are not mutually perpendicular. See Arts (1993) for an analysis of the characteristics of the elliptical motion associated with (4.101).

²Two figures are homothetic if they are related by an expansion or a geometric contraction.

4.4 The physics of wave propagation for viscoelastic SH waves

We have seen in Chapter 1 that in anisotropic lossless media, the energy, group and envelope velocities coincide, but the energy velocity is not equal to the phase velocity. On the other hand, in dissipative isotropic media, the group velocity loses its physical meaning, and the energy velocity equals the phase velocity only for homogeneous viscoelastic plane waves. In this section, we investigate the relations between the different velocities for SH homogeneous viscoelastic plane waves. Moreover, we study the perpendicularity properties – shown to hold for elastic media (see Section 1.4.6) – between slowness surface and energy-velocity vector, and between wave or ray surface and slowness vector.

4.4.1 Energy velocity

Let us first obtain the relation between the energy velocity and the envelope velocity, as defined in equation (1.146) for the (x, z)-plane. Differentiating equation (4.78) with respect to the propagation – or attenuation – angle θ , squaring it and adding the result to the square of equation (4.78), we obtain

$$v_e^2 = v_{env}^2 - \frac{d\mathbf{v}_e}{d\theta} \cdot \hat{\mathbf{\kappa}} \left(\frac{d\mathbf{v}_e}{d\theta} \cdot \hat{\mathbf{\kappa}} + 2\frac{d\hat{\mathbf{\kappa}}}{d\theta} \cdot \mathbf{v}_e \right),$$
 (4.104)

where we have used the relations $d\tilde{\kappa}/d\theta = (l_3, -l_1)$, $l_1^2 + l_3^2 = 1$, and $l_1 = \sin \theta$ and $l_3 = \cos \theta$ are the direction cosines.

The dispersion relation for SH propagation in the symmetry plane of a monoclinic medium can be expressed as

$$p_{66}l_1^2 + p_{44}l_3^2 - \rho v_c^2 = 0,$$
 (4.105)

where v_e is the corresponding complex velocity. Since the complex-slowness vector for homogeneous plane waves is $\mathbf{s} = \mathbf{k}/\omega = (s_1, s_3)\hat{\mathbf{k}}$, equation (4.105) generalizes equations (1.261) to the lossy case – an appropriate rotation of coordinates eliminates the stiffness p_{46} . The solution of equation (4.105) is

$$v_e = \sqrt{\frac{p_{66}l_1^2 + p_{44}l_3^2}{\rho}}.$$
 (4.106)

The displacement field has the following form

$$\mathbf{u} = \dot{\mathbf{e}}_2 U_0 \exp[i(\omega t - k_1 x - k_3 z)],$$
 (4.107)

or

$$\mathbf{u} = \hat{\mathbf{e}}_2 U_0 \exp(-\boldsymbol{\alpha} \cdot \mathbf{x}) \exp[i\omega(t - \mathbf{s}_R \cdot \mathbf{x})],$$
 (4.108)

where U_0 is a complex quantity, k_1 and k_3 are the components of the complex wavevector **k**, and $\mathbf{s}_R = \boldsymbol{\kappa}/\omega$ is the slowness vector.

The associated strain components are

$$e_4 = \partial_3 u = -ik_3 U_0 \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})],$$

 $e_6 = \partial_1 u = -ik_1 U_0 \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})],$

$$(4.109)$$

and the stress components are

$$\sigma_4 = p_{44}e_4 = -ip_{44}k_3U_0 \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})],$$

$$\sigma_6 = p_{66}e_6 = -ip_{66}k_1U_0 \exp[i(\omega t - \mathbf{k} \cdot \mathbf{x})].$$
(4.110)

From equation (4.55), the Umov-Poynting vector is

$$\mathbf{p} = -\frac{1}{2}v^*(\sigma_4 \hat{\mathbf{e}}_3 + \sigma_6 \hat{\mathbf{e}}_1) = \frac{1}{2v_c}\omega^2 |U_0|^2 (l_3 p_{44} \hat{\mathbf{e}}_3 + l_1 p_{66} \hat{\mathbf{e}}_1) \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}), \quad (4.111)$$

where $v = i\omega u$ is the particle velocity. Note that for elastic media, the Umov-Poynting vector is real because v_c , p_{44} and p_{66} become real valued.

The time-averaged kinetic-energy density is, from equation (4.52),

$$\langle T \rangle = \frac{1}{4} \rho \mathbf{v}^* \cdot \mathbf{v} = \frac{1}{4} \rho \omega^2 |U_0|^2 \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}),$$
 (4.112)

and the time-averaged strain-energy density is, from equation (4.53),

$$\langle V \rangle = \frac{1}{4} \operatorname{Re}(p_{44}|e_4|^2 + p_{66}|e_6|^2) = \frac{1}{4} \rho \omega^2 |U_0|^2 \frac{\operatorname{Re}(v_c^2)}{|v_c|^2} \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}),$$
 (4.113)

where equations (4.28) and (4.106) have been used. Similarly, the time average of the rate of dissipated-energy density (4.54) is

$$\langle \dot{D} \rangle = \frac{1}{2} \rho \omega^3 |U_0|^2 \frac{\text{Im}(v_c^2)}{|v_c|^2} \exp(-2\alpha \cdot \mathbf{x}).$$
 (4.114)

As can be seen from equations (4.112) and (4.113), the two time-averaged energy densities are identical for elastic media, because v_c is real. For anelastic media, the difference between them is given by the factor $\operatorname{Re}(v_c^2)/|v_c|^2$ in the strain-energy density. Since the property (4.42) can be applied to homogeneous plane waves, we have $T_{\text{peak}} = 2\langle T \rangle$ and $V_{\text{peak}} = 2\langle V \rangle$.

Substitution of the Umov-Poynting vector and energy densities into equation (4.76) gives the energy velocity for SH waves, namely,

$$\mathbf{v}_{e} = \frac{v_{p}}{\operatorname{Re}(v_{e})} \left[l_{1}\operatorname{Re}\left(\frac{p_{66}}{\rho v_{e}}\right) \hat{\mathbf{e}}_{1} + l_{3}\operatorname{Re}\left(\frac{p_{44}}{\rho v_{e}}\right) \hat{\mathbf{e}}_{3} \right]. \quad (4.115)$$

Note the difference from the energy velocity (1.154) in the elastic case, for which $v_c = v_p$.

4.4.2 Group velocity

Using equation (4.28) and noting that $k_1 = kl_1$ and $k_3 = kl_3$ for homogeneous plane waves, we can rewrite the complex dispersion relation (4.105) as

$$F(k_1, k_3, \omega) = p_{66}k_1^2 + p_{44}k_3^2 - \rho\omega^2 = 0.$$
 (4.116)

The group velocity (4.39) can be computed using this implicit relation between ω and the real wavenumber components κ_1 and κ_3 . The partial derivatives are given by

$$\frac{\partial F}{\partial k_1} = 2p_{66}k_1, \quad \frac{\partial F}{\partial k_3} = 2p_{44}k_3,$$
 (4.117)

and

$$\frac{\partial F}{\partial \omega} = p_{66,\omega}k_1^2 + p_{44,\omega}k_3^2 - 2\rho\omega, \qquad (4.118)$$

where the subscript ω denotes the derivative with respect to ω . Consequently, substituting these expressions into equation (4.39), we get

$$\mathbf{v}_{g} = -2l_{1} \left[\operatorname{Re} \left(\frac{d}{v_{c} p_{66}} \right) \right]^{-1} \hat{\mathbf{e}}_{1} - 2l_{3} \left[\operatorname{Re} \left(\frac{d}{v_{c} p_{44}} \right) \right]^{-1} \hat{\mathbf{e}}_{3}, \quad (4.119)$$

where

$$d = \omega (p_{66,\omega} l_1^2 + p_{44,\omega} l_3^2) - 2\rho v_e^2. \qquad (4.120)$$

Comparison of equations (4.115) and (4.119) indicates that the energy velocity is not equal to the group velocity for all frequencies. The group velocity has physical meaning only for low-loss media as an approximation to the energy velocity. It is easy to verify that the two velocities coincide for lossless media.

4.4.3 Envelope velocity

Differentiating the phase velocity (4.29) (by using equation (4.106)), and substituting the result into equation (1.146), we obtain the magnitude of the envelope velocity:

$$v_{env} = v_p \sqrt{1 + l_1^2 l_3^2 v_p^2} \left[\operatorname{Re}\left(\frac{p_{66} - p_{44}}{\rho v_e^3}\right) \right]^2.$$
 (4.121)

If the medium is isotropic, $p_{66} = p_{44}$, and the envelope velocity equals the phase velocity and the energy velocity (4.115). For lossless media $p_{IJ} = c_{IJ}$ (the elasticity constants) are real quantities, $v_p = v_c$, and

$$v_{env} = v_e = v_g = \frac{1}{\rho v_p} \sqrt{c_{66}^2 l_1^2 + c_{44}^2 l_3^2}$$
 (4.122)

(see equation (1.149)).

4.4.4 Perpendicularity properties

In anisotropic elastic media, the energy velocity is perpendicular to the slowness surface and the wavevector is perpendicular to the energy-velocity surface or wave surface (see Section 1.4.6). These properties do not apply, in general, to anisotropic anelastic media as will be seen in the following derivations. The equation of the slowness curve can be obtained by using the dispersion relation (4.105). Dividing the slowness $s_R = 1/\text{Re}(v_c)$ by s_R and using $s_{R1} = s_R l_1$ and $s_{R3} = s_R l_3$, we obtain the equation for the slowness curve, namely,

$$\Omega(s_{R1}, s_{R3}) = \operatorname{Re}\left[\left(\frac{s_{R1}^2}{\rho/p_{66}} + \frac{s_{R3}^2}{\rho/p_{44}}\right)^{-1/2}\right] - 1 = 0.$$
(4.123)

A vector perpendicular to this curve is given by

$$\nabla_{s_R}\Omega = \left(\frac{\partial\Omega}{\partial s_{R1}}, \frac{\partial\Omega}{\partial s_{R3}}\right) = -v_p^2 \left[l_1 \operatorname{Re}\left(\frac{p_{66}}{\rho v_c^3}\right) \hat{\mathbf{e}}_1 + l_3 \operatorname{Re}\left(\frac{p_{44}}{\rho v_c^3}\right) \hat{\mathbf{e}}_3 \right].$$
(4.124)
It is clear from equation (4.115) that \mathbf{v}_e and $\nabla_{s_R}\Omega$ are not collinear vectors; thus, the energy velocity is not perpendicular to the slowness surface. However, if we consider the limit $\omega \to \infty$ – the elastic, lossless limit by convention – for which $p_{IJ} \to c_{IJ}$, we may state that in this limit, the energy-velocity vector is perpendicular to the unrelaxed slowness surface. The same perpendicularity properties hold for the static limit ($\omega \to 0$).

Similarly, the other perpendicularity property of elastic media, i.e., that the slowness vector must be perpendicular to the energy-velocity surface, is not valid for anelastic media at all frequencies. By using equation (4.29), and differentiating equation (4.78) with respect to θ , we obtain

$$\frac{d\kappa}{d\theta} \cdot \mathbf{v}_e + \kappa \cdot \frac{d\mathbf{v}_e}{d\theta} = \frac{d\kappa}{d\theta} \cdot \mathbf{v}_e + \varrho \kappa \cdot \frac{d\mathbf{v}_e}{d\phi} = 0,$$
 (4.125)

where $\rho = d\phi/d\theta$, with

$$\tan \phi = \frac{v_{e1}}{v_{e3}} = \frac{\text{Re}[p_{66}/v_e(\theta)]}{\text{Re}[p_{44}/v_e(\theta)]} \tan \theta$$
(4.126)

(from equation (4.115)).



Figure 4.3: Relation between the energy velocity and the phase velocity in terms of the propagation and energy angles.

Figure 4.3 shows the relation between the propagation and energy angles. It can be shown that ρ is always different from zero, in particular $\rho = 1$ for isotropic media. Since $d\kappa/d\theta$ is tangent to the slowness surface – recall that $\kappa = \omega s_R$ – and v_e is not perpendicular to it, the first term in equation (4.125) is different from zero. Since $dv_e/d\phi$ is tangent to the wave surface, equation (4.125) implies that the real wavevector κ is not perpendicular to that surface. In fact, taking into account that $\kappa(\theta) = [\omega/v_p(\theta)](\sin(\theta)\hat{e}_1 +$ $\cos(\theta)\hat{\mathbf{e}}_3$, and after a lengthy but straightforward calculation of the first term of equation (4.125), we have

$$\frac{d\boldsymbol{\kappa}}{d\theta} \cdot \mathbf{v}_e = \omega v_p l_1 l_3 \operatorname{Re} \left[\frac{(p_{66} - p_{44})}{\rho v_e} \left(\frac{1}{|v_e|^2} - \frac{1}{v_e^2} \right) \right].$$
(4.127)

For lossless media, v_c is real and equation (4.127) is identically zero; in this case, the perpendicularity properties are verified:

$$\frac{d\mathbf{v}_e}{d\theta} \cdot \boldsymbol{\kappa} = \frac{d\boldsymbol{\kappa}}{d\theta} \cdot \mathbf{v}_e = 0,$$
 (4.128)

and from equation (4.104) the envelope velocity equals the energy velocity. In lossless media or at the unrelaxed and static limits in lossy media, the wavevector is perpendicular to the wave surface – the wave front in the unrelaxed case.

Perpendicularity for all frequencies in anelastic media holds between the slowness surface and the envelope-velocity vector, as well as the surface determined by the envelopevelocity vector and the slowness vector. Using equations (1.142), (4.29) and (4.106), we obtain the components of the envelope velocity,

$$v_{env})_1 = v_p^2 l_1 \operatorname{Re}\left(\frac{p_{66}}{\rho v_c^3}\right)$$
, and $v_{env})_3 = v_p^2 l_3 \operatorname{Re}\left(\frac{p_{44}}{\rho v_c^3}\right)$. (4.129)

The associated vector is collinear to $\nabla_{s_R}\Omega$ for all frequencies (see equation (4.124)). Moreover, since the expression of the envelope of plane waves has the same form as in the elastic case (equation (1.165)), the same reasoning used in Section 1.4.6 implies that the real wavenumber vector and the slowness vector are perpendicular to the surface defined by the envelope velocity.

4.4.5 Numerical evaluation of the energy velocity

Let us compare the different physical velocities. Figure 4.4 compares a numerical evaluation of the location of the energy (white dots) and the theoretical energy velocity (solid curve). The energy velocity is computed from the snapshot by finding the baricenter of $|\mathbf{u}|^2$ along the radial direction. Attenuation is modeled by Zener elements for which the characteristic frequency coincides with the source dominant frequency.

This comparison is represented in a linear plot in Figure 4.5, where the envelope and group velocities are also represented. While the envelope and energy velocities practically coincide, the group velocity gives a wrong prediction of the energy location. More details about this comparison are given in Carcione, Quiroga-Goode and Cavallini (1996). Carcione (1994a) also considers the qP-qS case.

It is important to note here that there exist conditions under which the group velocity has a clear physical meaning. The concept of signal velocity introduced by Sommerfeld and Brillouin (Brillouin, 1960; Mainardi, 1983) describes the velocity of energy transport for the Lorentz model. It is equal to the group velocity in regions of dispersion without attenuation (Felsen and Marcuvitz, 1973; Mainardi, 1987; Oughstun and Sherman, 1994). This happens, under certain conditions, in the process of resonance attenuation in solid, liquid and gaseous media. The Lorentz model describes dielectric-type media as a set of



Figure 4.4: Comparison between a numerical evaluation of the energy location (white dots) and the theoretical energy-velocity curve (solid line). The former is computed by finding the center of gravity of the energy-like quantity $|\mathbf{u}|^2$ along the radial direction.



Figure 4.5: Same comparison as in Figure 4.4, but here the envelope and group velocities are also represented. The dotted line corresponds to the numerical evaluation of the energy velocity and θ is the propagation angle.

neutral atoms with "elastically" bound electrons to the nucleus, where each electron is bound by a Hooke's law restoring force (Nussenzveig, 1972; Oughstun and Sherman, 1994). The atoms vibrate at a resonance frequency under the action of an electromagnetic field. This process implies attenuation and dispersion, since the electrons emit electromagnetic waves which carry away energy.

Garret and McCumber (1970) and Steinberg and Chiao (1994) show that the group velocity describes the velocity of the pulse for electromagnetic media such as, for example, gain-assisted linear anomalous dispersion in cesium gas. Basically, the conditions imply that the group velocity remains constant over the pulse bandwidth so that the light pulse maintains its shape during the propagation. These theoretical results are confirmed by Wang, Kuzmich and Dogariu (2000), who report a very large superluminal effect for laser pulses of visible light, in which a pulse propagates with a negative group velocity without violating causality.

However, the classical concepts of phase, energy and group velocities generally break down for the Lorentz model, depending on the value of the source dominant frequency and source bandwidth compared to the width of the spectral line. Loudon (1970) has derived an expression of the energy velocity which does not exceed the velocity of light. It is based on the fact that when the frequency of the wave is close to the oscillator frequency, part of the energy resides in the excited oscillators. This part of the energy must be added to the electromagnetic field energy.

4.4.6 Forbidden directions of propagation

There is a singular phenomenon when inhomogeneous plane waves propagate in a medium with anisotropy and attenuation. The theory predicts, beyond a given degree of inhomogeneity, the existence of forbidden directions (forbidden solutions) or "stop bands" where there is no wave propagation (not to be confused with the frequency stop bands of periodic structures (e.g., Silva, 1991; Carcione and Poletto, 2000)). This phenomenon does not occur in dissipative isotropic and anisotropic elastic media. The combination of anelasticity and anisotropy activates the bands. These solutions are found even in very weakly anisotropic and quasi-elastic materials; only a finite value of Q is required. Weaker anisotropy does not affect the width of the bands, but increases the threshold of inhomogeneity above which they appear; moreover, near the threshold, lower attenuation implies narrower bands.

This phenomenon was discovered by Krebes and Le (1994) and Carcione and Cavallini (1995a) for wave propagation of pure shear inhomogeneous viscoelastic plane waves in the symmetry plane of a monoclinic medium. Carcione and Cavallini (1997) predict the same phenomenon in electromagnetic media on the basis of the acoustic-electromagnetic analogy (Carcione and Cavallini, 1995b). Figure 4.6a-b represents the square of the phase velocity as a function of the propagation angle, where the dashed line corresponds to the homogeneous wave ($\gamma = 0$); (a) and (b) correspond to strong and weak attenuation, respectively. Observe that in the transition from $\gamma = 60^{\circ}$ to $\gamma = 68^{\circ}$, two "stop bands" develop (for $\gamma > \gamma_0 \approx 64^{\circ}$) where the wave does not propagate (Figure 4.6a). Note that the stop bands exist even for high values of Q, as is the case in Figure 4.6b. The behavior is such that these stop bands exist for any finite value of Q, with their width decreasing with increasing Q. Červený and Pšenčík (2005a,b) have used a form of the sextic Stroh formalism (Ting, 1996; Caviglia and Morro, 1999) to re-interpret the forbidden-directions phenomenon by using a different inhomogeneity parameter, instead of angle γ . The new approach involves the solution of a 6 × 6 complex-valued eigensystem and the parameterization excludes the forbidden solutions.



Figure 4.6: "Stopbands" for propagation of inhomogeneous viscoelastic plane waves in anisotropic anelastic media. The figure shows the square of the phase velocity as a function of the propagation angle for different values of the inhomogeneity angle γ . In (a) the medium has strong dissipation and in (b) the dissipation is weak.

4.5 Memory variables and equation of motion in the time domain

As in the isotropic viscoelastic case (Section 3.9), we obtain, in this section, the memoryvariable differential equations, which allow us to avoid numerical calculations of time convolutions when modeling wave propagation.

We define the reference elastic limit in the unrelaxed regime ($\omega \rightarrow \infty$ or $t \rightarrow 0$), and denote the unrelaxed stiffnesses by c_{IJ} . The following equations correspond to the 3-D case, but the space dimension is indicated by n instead of 3 to facilitate the particularization to the 2-D case (n=2). Using model 3 of Section 4.1.3, the time-domain relaxation matrix for a medium with general anisotropic properties (a triclinic medium) has the following symmetric form

$$\Psi(t) = \begin{pmatrix} \psi_{11} & \psi_{12} & \psi_{13} & c_{14} & c_{15} & c_{16} \\ & \psi_{22} & \psi_{23} & c_{24} & c_{25} & c_{26} \\ & & \psi_{33} & c_{34} & c_{35} & c_{36} \\ & & & & \psi_{44} & c_{45} & c_{46} \\ & & & & & & \psi_{55} & c_{56} \\ & & & & & & & \psi_{66} \end{pmatrix} H(t). \quad (4.130)$$

We may express the components as

$$\check{\psi}_{I(I)} = c_{I(I)} - \check{\mathcal{E}} + \check{\mathcal{K}}\chi_1 + 2\left(1 - \frac{1}{n}\right)\check{\mu}\chi_{\delta}, \quad I = 1, 2, 3,$$
(4.131)

$$\check{\psi}_{IJ} = c_{IJ} - \vec{\mathcal{E}} + \bar{\mathcal{K}}\chi_1 + 2\bar{\mu}\left(1 - \frac{1}{n}\chi_\delta\right), \quad I, J = 1, 2, 3; \ I \neq J,$$
(4.132)

$$\tilde{\psi}_{44} = c_{44}\chi_2, \quad \tilde{\psi}_{55} = c_{55}\chi_3, \quad \tilde{\psi}_{66} = c_{66}\chi_4,$$
(4.133)

where

$$\bar{\mathcal{E}} = \bar{\mathcal{K}} + 2\left(1 - \frac{1}{n}\right)\bar{\mu}, \qquad (4.134)$$

$$\chi_{\nu}(t) = L_{\nu} \left(\sum_{l=1}^{L_{\nu}} \frac{\tau_{el}^{(\nu)}}{\tau_{\sigma l}^{(\nu)}} \right)^{-1} \left[1 - \frac{1}{L_{\nu}} \sum_{l=1}^{L_{\nu}} \left(1 - \frac{\tau_{el}^{(\nu)}}{\tau_{\sigma l}^{(\nu)}} \right) \exp(-t/\tau_{\sigma l}^{(\nu)}) \right], \quad \nu = 1, \dots, 4,$$
(4.135)

with $\tau_{cl}^{(\nu)}$ and $\tau_{\sigma l}^{(\nu)}$ being relaxation times satisfying $\tau_{cl}^{(\nu)} \ge \tau_{\sigma l}^{(\nu)}$. Moreover, $\chi = 1$ for t = 0 and $\tau_{cl}^{(\nu)} = \tau_{\sigma l}^{(\nu)}$. The index δ can be chosen to be 2, 3 or 4 (see Section 4.1.3).

The complex modulus is the time Fourier transform of $d(\chi_{\nu}H)/dt$. It yields

$$M_{\nu}(\omega) = \left(\sum_{l=1}^{L_{\nu}} \frac{\tau_{el}^{(\nu)}}{\tau_{\sigma l}^{(\nu)}}\right)^{-1} \sum_{l=1}^{L_{\nu}} \frac{1 + i\omega \tau_{el}^{(\nu)}}{1 + i\omega \tau_{\sigma l}^{(\nu)}}$$
(4.136)

(see equation (2.196)), which has the property $M_{\nu} \rightarrow 1$ for $\omega \rightarrow \infty$.

The relaxation functions (4.135) are sufficiently general to describe any type of frequency behavior of attenuation and velocity dispersion.

4.5.1 Strain memory variables

The time-domain stress-strain relation can be expressed as

$$\sigma_I = \psi_{IJ} * \partial_t e_J \qquad (4.137)$$

(see equation (2.22)), where σ_I and e_J are the components of the stress and strain 6 × 1 arrays – equations (1.20) and (1.27), respectively.

163

Applying the Boltzmann operation (2.6) to equation (4.137), we obtain

$$\sigma_I = A_{IJ}^{(\nu)} e_J + B_{IJ}^{(\nu)} \sum_{l=1}^{L_{\nu}} e_{Jl}^{(\nu)},$$
 (4.138)

where the A's and the B's are combinations of the elasticity constants c_{IJ} , and

$$e_{Jl}^{(\nu)} = \varphi_{\nu l} * e_J, \quad J = 1, \dots, 6, \quad l = 1, \dots, L_{\nu}, \quad \nu = 1, \dots, 4$$
 (4.139)

are the components of the 6×1 strain memory array $e_i^{(\nu)}$, with

$$\tilde{\varphi}_{\nu l}(t) = \frac{1}{\tau_{\sigma l}^{(\nu)}} \left(\sum_{l=1}^{L_{\nu}} \frac{\tau_{cl}^{(\nu)}}{\tau_{\sigma l}^{(\nu)}} \right)^{-1} \left(1 - \frac{\tau_{cl}^{(\nu)}}{\tau_{\sigma l}^{(\nu)}} \right) \exp(-t/\tau_{\sigma l}^{(\nu)})$$
(4.140)

being the response function corresponding to the *l*-th dissipation mechanism. In 3-D space, the strain memory array is a symmetric tensor given by

$$\mathbf{e}_{l}^{(\nu)} = \begin{pmatrix} e_{11l}^{(\nu)} & e_{12l}^{(\nu)} & e_{13l}^{(\nu)} \\ e_{22l}^{(\nu)} & e_{23l}^{(\nu)} \\ e_{33l}^{(\nu)} \end{pmatrix} = \varphi_{\nu l} * \mathbf{e}, \qquad (4.141)$$

corresponding to the *l*-th dissipation mechanism of the relaxation function χ_{ν} . This tensor contains the past history of the material due to that mechanism. In the elastic case $(\tau_{\sigma l}^{(\nu)} \rightarrow \tau_{el}^{(\nu)})$, $\tilde{\varphi}_{\nu l} \rightarrow 0$ and the strain memory tensor vanishes. As the strain tensor, the memory tensor possesses the unique decomposition

$$\mathbf{e}_{l}^{(\nu)} = \mathbf{d}_{l}^{(\nu)} + \frac{1}{n} \operatorname{tr}(\mathbf{e}_{l}^{(\nu)}) \mathbf{I}_{n}, \quad \operatorname{tr}(\mathbf{d}_{l}^{(\nu)}) = 0,$$
 (4.142)

where the traceless symmetric tensor $\mathbf{d}_l^{(\nu)}$ is the deviatoric strain memory tensor. Then, the dilatational and shear memory variables can be defined as

$$e_{1l} = \operatorname{tr}(\mathbf{e}_l^{(1)}) \quad \text{and} \quad e_{ijl}^{(\nu)} = \left(\mathbf{d}_l^{(\nu)}\right)_{ij},$$

$$(4.143)$$

respectively, where $\nu = \delta$ for i = j, $\nu = 2$ for ij = 23, $\nu = 3$ for ij = 13, and $\nu = 4$ for ij = 12.

In explicit form, the stress-strain relations in terms of the strain components and memory variables are

$$\sigma_{1} = c_{1J}e_{J} + \tilde{\mathcal{K}} \sum_{l=1}^{L_{1}} e_{1l} + 2\bar{\mu} \sum_{l=1}^{L_{2}} e_{1ll}^{(d)}$$

$$\sigma_{2} = c_{2J}e_{J} + \tilde{\mathcal{K}} \sum_{l=1}^{L_{1}} e_{1l} + 2\bar{\mu} \sum_{l=1}^{L_{4}} e_{22l}^{(d)}$$

$$\sigma_{3} = c_{3J}e_{J} + \tilde{\mathcal{K}} \sum_{l=1}^{L_{1}} e_{1l} - 2\bar{\mu} \sum_{l=1}^{L_{4}} (e_{11l}^{(d)} + e_{22l}^{(d)})$$

$$\sigma_{4} = c_{4J}e_{J} + c_{44} \sum_{l=1}^{L_{2}} e_{22l}^{(2)}$$

$$\sigma_{5} = c_{5J}e_{J} + c_{55} \sum_{l=1}^{L_{3}} e_{13l}^{(d)}$$

$$\sigma_{6} = c_{6J}e_{J} + c_{66} \sum_{l=1}^{L_{4}} e_{12l}^{(d)},$$
(4.144)

where, as stated before,

$$c_{IJ} = \psi_{IJ}(t = 0^+)$$
 (4.145)

are the unrelaxed elasticity constants. In the work of Carcione (1995), the memory variables are multiplied by relaxed elasticity constants. This is due to a different definition of the response function (4.140). For instance, in the 1-D case with one dissipation mechanism (see equations (2.283) and (2.285)), the difference is the factor τ_e/τ_σ .

The terms containing the stress components describe the instantaneous (unrelaxed) response of the medium, and the terms involving the memory variables describe the previous states of deformation. Note that because $\mathbf{d}_{l}^{(\nu)}$ is traceless, $e_{11l}^{(\delta)} + e_{22l}^{(\delta)} + e_{33l}^{(\delta)} = 0$, and the number of independent variables is six, i.e., the number of strain components. The nature of the terms can be easily identified: in the diagonal stress components, the dilatational memory variables are multiplied by a generalized bulk modulus $\bar{\mathcal{K}}$, and the shear memory variables are multiplied by a generalized rigidity modulus $\bar{\mu}$.

4.5.2 Memory-variable equations

Application of the Boltzmann operation (2.6) to the deviatoric part of equation (4.141) gives

$$\partial_t \mathbf{d}_l^{(\nu)} = \varphi_{\nu \ell}(0)\mathbf{d} + (\partial_t \tilde{\varphi}_{\nu \ell} H) * \mathbf{d},$$
 (4.146)

where **d** is the deviatoric strain tensor whose components are given in equation (1.15). Because $\partial_t \tilde{\varphi}_{vl} = -\tilde{\varphi}_{vl} / \tau_{vl}^{(\nu)}$, equation (4.146) becomes

$$\partial_t \mathbf{d}_l^{(\nu)} = \varphi_{\nu l}(0) \mathbf{d} - \frac{1}{\tau_{\sigma l}^{(\nu)}} \mathbf{d}_l^{(\nu)}, \quad \mathbf{d}_l^{(\nu)} = \varphi_{\nu l} * \mathbf{d}, \quad \nu = 2, 3, 4.$$
 (4.147)

Similarly, applying the Boltzmann operation to $tr(\mathbf{e}_{l}^{(1)})$, we obtain

$$\partial_t \operatorname{tr}(\mathbf{e}_i^{(1)}) = \varphi_{1l}(0)\operatorname{tr}(\mathbf{e}) - \frac{1}{\tau_{\sigma l}^{(1)}}\operatorname{tr}(\mathbf{e}_l^{(1)}).$$
 (4.148)

The explicit equations in terms of the memory variables are

$$\partial_t e_{1l} = n\varphi_{1l}(0)\bar{\epsilon} - \frac{1}{\tau_{st}^{(1)}}e_{1l}, \quad l = 1, ..., L_1$$

$$\partial_t e_{11l}^{(\delta)} = \varphi_{\delta l}(0)(e_{11}^{(\delta)} - \bar{\epsilon}) - \frac{1}{\tau_{st}^{(\delta)}}e_{11l}^{(\delta)}, \quad l = 1, ..., L_{\delta}$$

$$\partial_t e_{22l}^{(\delta)} = \varphi_{\delta l}(0)(e_{22}^{(\delta)} - \bar{\epsilon}) - \frac{1}{\tau_{st}^{(\delta)}}e_{22l}^{(\delta)}, \quad l = 1, ..., L_{\delta}$$

$$\partial_t e_{23l} = \varphi_{2l}(0)e_{23} - \frac{1}{\tau_{st}^{(2)}}e_{23l}, \quad l = 1, ..., L_2$$

$$\partial_t e_{13l} = \varphi_{3l}(0)e_{13} - \frac{1}{\tau_{st}^{(\delta)}}e_{13l}, \quad l = 1, ..., L_3$$

$$\partial_t e_{12l} = \varphi_{4l}(0)e_{12} - \frac{1}{\tau_{st}^{(\delta)}}e_{12l}, \quad l = 1, ..., L_4,$$
(4.149)

where $\bar{\epsilon} = e_{II}/n$. The index δ can be chosen to be 2, 3 or 4 (see Section 4.1.3).

Two different formulations of the anisotropic viscoelastic equation of motion follow. In the displacement formulation, the unknown variables are the displacement field and the memory variables. In this case, the equation of motion is formulated using the straindisplacement relations (1.2), the stress-strain relations (4.144), the equations of momentum conservation (1.23) and the memory-variable equations (4.149). In the particlevelocity/stress formulation, the field variables are the particle velocities, the stress components and the time derivative of the memory variables, because the first time derivative of the stress-strain relations are required. The first formulation is second-order in the time derivatives, while the second is first-order. In the particle-velocity/stress formulation case, the material properties are not differentiated explicitly, as they are in the displacement formulation. A practical example of 3-D viscoelastic anisotropic modeling applied to an exploration-geophysics problem is given by Dong and McMechan (1995).

2-D equations of motion – referred to as SH and qP-qSV equations of motion – can be obtained if the material properties are uniform in the direction perpendicular to the plane of wave propagation. Alternatively, the decoupling occurs in three dimensions in a symmetry plane. This situation can be generalized up to monoclinic media provided that the plane of propagation is the plane of symmetry of the medium. In fact, propagation in the plane of mirror symmetry of a monoclinic medium is the most general situation for which pure shear waves exist at all propagation angles.

Alternative methods for simulating wave propagation in anisotropic media – including attenuation effects – are based on ray-tracing algorithms. Gajewski, and Pšenčík (1992) use the ray method for weakly anisotropic media, and Le, Krebes and Quiroga-Goode (1994) simulate SH-wave propagation by complex ray tracing.

4.5.3 SH equation of motion

Let us assume that the (x, z)-plane is the symmetry plane of a monoclinic medium and $\partial_2 = 0$. The cross-plane assumption implies that the only non-zero stress components are σ_{12} and σ_{23} . Following the same steps to obtain the 3-D equation of motion, the displacement formulation of the SH-equation of motion is given by

Euler's equation (1.46)₁.

ii) The stress-strain relations

$$\sigma_4 = c_{44}e_4 + c_{46}e_6 + c_{44}\sum_{l=1}^{L_2} e_{23l},$$

$$\sigma_6 = c_{46}e_4 + c_{66}e_6 + c_{66}\sum_{l=1}^{L_4} e_{12l}.$$
(4.150)

iii) Equations (4.149)₄ and (4.149)₆.

See Carcione and Cavallini (1995c) for more details about this wave equation.

4.5.4 qP-qSV equation of motion

Let us consider the two-dimensional particle-velocity/stress equations for propagation in the (x, z)-plane of a transversely isotropic medium. In this case, we explicitly consider a two-dimensional world, i.e., n = 2. We assign one relaxation mechanism to dilatational anelastic deformations ($\nu = 1$) and one relaxation mechanism to shear anelastic deformations ($\nu = 2$). The equations governing wave propagation can be expressed by

Euler's equations (1.45)₁ and (1.45)₂:

$$\partial_1 \sigma_{11} + \partial_3 \sigma_{13} + f_1 = \rho \partial_t v_1, \tag{4.151}$$

$$\partial_1 \sigma_{13} + \partial_3 \sigma_{33} + f_3 = \rho \partial_t v_3,$$
 (4.152)

where f_1 and f_3 are the body-force components.

ii) Stress-strain relations:

$$\partial_t \sigma_{11} = c_{11} \partial_1 v_1 + c_{13} \partial_3 v_3 + \hat{K} \epsilon_1 + 2c_{55} \epsilon_2,$$
 (4.153)

$$\partial_t \sigma_{33} = c_{13} \partial_1 v_1 + c_{33} \partial_3 v_3 + \hat{\kappa} \epsilon_1 - 2c_{55} \epsilon_2,$$
 (4.154)

$$\partial_t \sigma_{13} = c_{55}[(\partial_3 v_1 + \partial_1 v_3) + \epsilon_3],$$
 (4.155)

where ϵ_1 , ϵ_2 and ϵ_3 are first time derivatives of the memory variables ($\partial_t e_{11}$, $\partial_t e_{22}$ and $\partial_t e_{13}$, respectively), and

$$\bar{\mathcal{K}} = \bar{\mathcal{E}} - c_{55}, \quad \bar{\mathcal{E}} = \frac{1}{2}(c_{11} + c_{33}).$$
 (4.156)

As in the 3-D case, the stress-strain relations satisfy the condition that the mean stress depends only on the dilatational relaxation function in any coordinate system – the trace of the stress tensor should be invariant under coordinate transformations. Moreover, the deviatoric stresses solely depend on the shear relaxation function.

iii) Memory-variable equations:

$$\partial_t \epsilon_1 = \frac{1}{\tau_{\sigma}^{(1)}} \left[\left(\frac{\tau_{\sigma}^{(1)}}{\tau_{\epsilon}^{(1)}} - 1 \right) \left(\partial_1 v_1 + \partial_3 v_3 \right) - \epsilon_1 \right], \tag{4.157}$$

$$\partial_t \epsilon_2 = \frac{1}{2\tau_{\sigma}^{(2)}} \left[\left(\frac{\tau_{\sigma}^{(2)}}{\tau_{\epsilon}^{(2)}} - 1 \right) \left(\partial_1 v_1 - \partial_3 v_3 \right) - 2\epsilon_2 \right], \tag{4.158}$$

$$\partial_t \epsilon_3 = \frac{1}{\tau_{\sigma}^{(2)}} \left[\left(\frac{\tau_{\sigma}^{(2)}}{\tau_{\epsilon}^{(2)}} - 1 \right) (\partial_3 v_1 + \partial_1 v_3) - \epsilon_3 \right]. \tag{4.159}$$

Transforming the memory-variable equations (4.157), (4.158) and (4.159) to the ω domain (e.g., $\partial_t \epsilon_1 \rightarrow i\omega \epsilon_1$), and substituting the memory variables into equations (4.153), (4.154) and (4.155), we obtain the frequency-domain stress-strain relation:

$$i\omega \begin{pmatrix} \sigma_{11} \\ \sigma_{33} \\ \sigma_{13} \end{pmatrix} = \begin{pmatrix} p_{11} & p_{13} & 0 \\ p_{13} & p_{33} & 0 \\ 0 & 0 & p_{55} \end{pmatrix} \cdot \begin{pmatrix} \partial_1 v_1 \\ \partial_3 v_3 \\ \partial_3 v_1 + \partial_1 v_3 \end{pmatrix},$$
(4.160)

where

$$p_{11} = c_{11} - \mathcal{E} + \mathcal{K}M_1 + c_{55}M_2$$

$$p_{33} = c_{33} - \tilde{\mathcal{E}} + \mathcal{K}M_1 + c_{55}M_2$$

$$p_{13} = c_{13} - \tilde{\mathcal{E}} + \mathcal{K}M_1 + c_{55}(2 - M_2)$$

$$p_{55} = c_{55}M_2$$
(4.161)

are the complex stiffnesses, and

$$M_{\nu} = \frac{\tau_{\sigma}^{(\nu)}}{\tau_{\epsilon}^{(\nu)}} \left(\frac{1 + i\omega \tau_{\epsilon}^{(\nu)}}{1 + i\omega \tau_{\sigma}^{(\nu)}} \right), \quad \nu = 1, 2$$

$$(4.162)$$

are the Zener complex moduli. Note that when $\omega \rightarrow \infty$, $p_{IJ} \rightarrow c_{IJ}$.

The relaxation times can be expressed as (see Section 2.4.5)

$$\tau_{\epsilon}^{(\nu)} = \frac{\tau_0}{Q_{0\nu}} \left(\sqrt{Q_{0\nu}^2 + 1} + 1 \right), \quad \text{and} \quad \tau_{\sigma}^{(\nu)} = \frac{\tau_0}{Q_{0\nu}} \left(\sqrt{Q_{0\nu}^2 + 1} - 1 \right), \tag{4.163}$$

where τ_0 is a relaxation time such that $1/\tau_0$ is the center frequency of the relaxation peak and $Q_{0\nu}$ are the minimum quality factors.

4.6 Analytical solution for SH waves in monoclinic media

The following is an example of the use of the correspondence principle to obtain a transient solution in anisotropic anelastic media, where an analytical solution is available in the frequency domain.

In the plane of mirror symmetry of a lossless monoclinic medium, say, the (x, z)-plane, the relevant stiffness matrix describing wave propagation of the cross-plane shear wave is

$$\mathbf{C} = \begin{pmatrix} c_{44} & c_{46} \\ c_{46} & c_{66} \end{pmatrix}. \tag{4.164}$$

Substitution of the stress-strain relation based on (4.164) into Euler's equation $(1.46)_1$ gives

$$\nabla \cdot \mathbf{C} \cdot \nabla u - \rho \partial_{tt}^2 u = f_u,$$
 (4.165)

where u is the displacement field, $f_u = -f_2$ is the body force, and, here,

$$\nabla = \begin{pmatrix} \partial_3 \\ \partial_1 \end{pmatrix}. \quad (4.166)$$

For a homogeneous medium, equation (4.165) becomes

$$(c_{44}\partial_3\partial_3 + c_{46}\partial_1\partial_3 + c_{66}\partial_1\partial_1)u - \rho\partial_{tt}^2 u = f_u.$$
 (4.167)

We show below that it is possible to transform the spatial differential operator on the left-hand side of equation (4.167) to a pure Laplacian differential operator. In that case, equation (4.165) becomes

$$(\partial_{3'}\partial_{3'} + \partial_{1'}\partial_{1'})u - \rho \partial_{tt}^2 u = f,$$
 (4.168)

where x' and z' are the new coordinates. Considering the solution for the Green function – the right-hand side of (4.168) is Dirac's function in time and space at the origin – and transforming the wave equation to the frequency domain, we obtain

$$(\partial_{3'}\partial_{3'} + \partial_{1'}\partial_{1'})\tilde{g} + \rho\omega^2 \tilde{g} = -4\pi\delta(x')\delta(z'),$$
 (4.169)

where \tilde{g} is the Fourier transform of the Green function, and the constant -4π is introduced for convenience. The solution of (4.169) is

$$\tilde{g}(x', z', \omega) = -i\pi H_0^{(2)}(\sqrt{\rho}\omega r'),$$
 (4.170)

(see Section 3.10.1), where $H_0^{(2)}$ is the Hankel function of the second kind, and

$$r' = \sqrt{x'^2 + z'^2} = \sqrt{\mathbf{x'}^{\top} \cdot \mathbf{x'}},$$
 (4.171)

with $\mathbf{x}' = (z', x')$. We need to compute (4.170) in terms of the original position vector $\mathbf{x} = (z, x)$. Matrix C may be decomposed as $\mathbf{C} = \mathbf{A} \cdot \mathbf{\Lambda} \cdot \mathbf{A}^{\top}$, where $\mathbf{\Lambda}$ is the diagonal matrix of the eigenvalues, and A is the matrix of the normalized eigenvectors. Thus, the Laplacian operator in (4.165) becomes

$$\nabla \cdot \mathbf{C} \cdot \nabla = \nabla \cdot \mathbf{A} \cdot \mathbf{\Lambda} \cdot \mathbf{A}^{\top} \cdot \nabla = \nabla \cdot \mathbf{A} \cdot \mathbf{\Omega} \cdot \mathbf{\Omega} \cdot \mathbf{A}^{\top} \cdot \nabla = \nabla' \cdot \nabla', \tag{4.172}$$

where $\Lambda = \Omega^2$, and

$$\nabla' = \Omega \cdot \mathbf{A}^{\top} \cdot \nabla.$$
 (4.173)

Recalling that Ω is diagonal and $A^{\top} = A^{-1}$, we get

$$\mathbf{x}' = \mathbf{\Omega}^{-1} \cdot \mathbf{A}^{\top} \cdot \mathbf{x}. \tag{4.174}$$

The substitution of (4.174) into equation (4.171) squared gives

$$r^{\prime 2} = \mathbf{x} \cdot \mathbf{A} \cdot \mathbf{\Omega}^{-1} \cdot \mathbf{\Omega}^{-1} \cdot \mathbf{A}^{\top} \cdot \mathbf{x} = \mathbf{x} \cdot \mathbf{A} \cdot \mathbf{\Lambda}^{-1} \cdot \mathbf{A}^{\top} \cdot \mathbf{x}.$$
(4.175)

Since $\mathbf{A} \cdot \mathbf{\Lambda}^{-1} \cdot \mathbf{A}^{\top} = \mathbf{C}^{-1}$, we finally have

$$r^{2} = \mathbf{x} \cdot \mathbf{C}^{-1} \cdot \mathbf{x}^{-1} = (c_{66}z^{2} + c_{44}x^{2} - 2c_{46}xz)/c, \qquad (4.176)$$

where c is the determinant of C.

Then, substituting (4.176) into equation (4.170), we note that the elastic Green's function becomes

$$\tilde{g}(x, z, \omega) = -i\pi H_0^{(2)} \left(\omega \sqrt{\mathbf{x} \cdot \rho \mathbf{C}^{-1} \cdot \mathbf{x}} \right).$$
(4.177)

Application of the correspondence principle (see Section 3.6) gives the viscoelastic Green's function

$$\tilde{g}(x, z, \omega) = -i\pi H_0^{(2)} \left(\omega \sqrt{\mathbf{x} \cdot \rho \mathbf{P}^{-1} \cdot \mathbf{x}} \right), \qquad (4.178)$$

where **P** is the complex and frequency-dependent stiffness matrix. When solving the problem with a band-limited wavelet f(t), the solution is

$$\tilde{u}(\mathbf{x}, \omega) = -i\pi \tilde{f} H_0^{(2)} \left(\omega \sqrt{\mathbf{x} \cdot \rho \mathbf{P}^{-1} \cdot \mathbf{x}} \right),$$
(4.179)

where \tilde{f} is the Fourier transform of f. To ensure a time-domain real solution, when $\omega > 0$ we take

$$\tilde{u}(\mathbf{x}, \omega) = \tilde{u}^{*}(\mathbf{x}, -\omega),$$
 (4.180)

where the superscript * denotes the complex conjugate. Finally, the time-domain solution is obtained by an inverse transform based on the discrete fast Fourier transform. An example where dissipation is modeled with Zener models can be found in Carcione and Cavallini (1994a). Other investigations about anisotropy and loss of SH waves are published by Le (1993) and Le, Krebes and Quiroga-Goode (1994).

169

This page intentionally left blank

Chapter 5

The reciprocity principle

The reciprocal property is capable of generalization sicus to apply to all invensionlosistems inhibiener expedite of exhapting about a configuration of equilibrium, as I proved in the Directedings of the Mathematical Society for June 1874 [Act. XXI], and is not lost even when the systems are subject to damping

John William Strutt (Lord Raylogh) (Rayloigh, 1890a)

For reciprocity principle relates two solutions of a medium where the sources and the field receivers are interchanged. The principle for static displacements is received to Bétti (1872). Bayleigh (1873) eventhal the principle to vibrating backs and method the action of dissipative forces (see Bayleigh, 1945, vol. 1, p. 1577). Lam'e (1888) showed how the reciprocal theorems of Helmholtz — in the theory of least action in acoustics and optics — and of Lord Rayleigh — in acoustics — can be derived from a formula established by Lagrangemethe *Welchangue* Arcily(up), (1809), thereby anticipation Lagrange's theory of the variation of arbitrary constants (Funy, 1965, p. 129).

In this criatury, the work of Graffie (1939, 1954, 1963) is initiable. Graffi derived the list convolutional avejancity theorem for an isotropic, homogeneous, elastic solid. Extension to inhomogeneous clustic anisotropic media was achieved by Knopoff and Gaugi (1959). Gaugi (1970) developed a volume integral, time-convolution formulation of the reciprocity principle for inhomogeneous anisotropic linearly elastic media. This formulation permits the use of distributed sources as well as implicit outponent sources (i.e., maples with and without moment). Gaugi also derived a representation of particle displatement in terms of Greek's theorem.

de Huopi 1966) generalized the principle to the anisotropic viscoelastic ran. It is worth mentioning the work of Hedrarski (1983), who distinguished between convolution type and correlation-type reciprocity relations. Recently, de Hoop and Stam (1988) derived a general reciprocity theorem valid for solids with relaxation, including reciprocity for stress, as well as for parallely decided with relaxation, including reciprocity for stress, as well as for parallely decided by use above Hoop, 1995). Laboratory experiments of the enciprocity principle were performed by Gaugi (1980)(), who used a gravite block containing a cylindrical mass obstach that as a scatterer, and piezuelectric transforces to act as vertical source and vertical receiver. A direct numerical test of the principle in the information of satisfortopic clastic case was performed by Carcina and Gaugi (1998).

A reciprocity relation when the source is a dipole rather a manopole has been derived by field Rayley, in 1876. Useful applications of the reciprocity principle can be found in Fokkema and van den Berg (1993)

5.1 Sources, receivers and reciprocity

Reciprocity is usually applied to concentrated point forces and point receivers. However, reciprocity has a much wider application potential: in many cases, it is not used at its full potential, either because a variety of source and receiver types are not considered or their nephrmentation is not well understood.

heriprority holds for the very general case of an inhomogeneous anisotropic viscuelastic solid, in the presence of homology surfaces satisfying Dirichlet and the Neumann boundary conditions (e.g., Lamb's problem, chamber 1904)) (Fung. 1965, p. 214). However, it is not clear new the principle is applied when the sources are complex (Fenational Rocca, 1984). For restance, Mitter and Bokstad (1995) use reciprocity to transform walk-away VSP data into reverse VSP data, for offshore acquisition – Neutral, Hron and Roccay (1996) claim that the analytical solution to Lamb's problem – expressed in terms of particle displacement – for a dilatational point source does ant exhibit reciprocity when the source and receiver locations are interchanged. Hence, the following question arises: what, if any, source-terceiver configuration is reciprocity principle to the case of sources of couples and demonstrate that for any particular source, there is a corresponding receivercontigration that makes the source-receiver particular source.

We obtain reciprocity relations for informageneous anisotropic viscoelastic solids, and for distributed somers and receivers. We show that, in addition to the datal relations involving directional forces, the following results exists i) the diagonal comparatus of the strain tensor are reciprocal for dipole sources (single couple without noticent), in the offdiagonal components of the stress tensor are reciprocal for double couples with moments in the dilatation due to a directional force is reciprocal to the particle velocity due to a dilatation source, and (v) some combinations of the off-diagonal strains are reciprocal for single couples with moments.

5.2 The reciprocity principle

Let us consider a volume Ω_{c} enclosed by a surface S_{c} in a viscoelastic solid of density $\rho(\mathbf{x})$ and relaxation tensor $\phi_{0,0}(\mathbf{x}, 0)$, where $\mathbf{x} = [x, y, z]$ denotes the position vector. In full explicit form, the opportion of motion (1.23) and the stress-strain relation (2.9) can be written as

$$\rho(\mathbf{x})\partial_0^T a_i(\mathbf{x}, l) = -\partial_l \delta_0(\mathbf{x}, l) + f_i(\mathbf{x}, l),$$
 (5.1)

$$-\sigma_{ij}(\mathbf{x}, t) = -e_{ij}a_i(\mathbf{x}, t) * \partial \sigma_{ij}(\mathbf{x}, t)$$
 (5.2)

A reciprocity theorem valid for a general anisotropic viscorlastic medium can be derived from the equation of motion (5.1) and the stress-strain relation (5.2), and can be written in the form

$$\int_{\Omega} [u_i(\mathbf{x}, t) * f_i'(\mathbf{x}, t) + f_i(\mathbf{x}, t) * u_i'(\mathbf{x}, t)] d\Omega = 0$$
(5.3)

(Knopolf and Gaugi, 1959) de Hoop, 1975). Here σ_i is the *i*-th component of the displacement due to the source **f**, while u_i^i is the *i*-th component of the displacement due to the source **f**. The derivation of equation (5.3) assumes that the displacements and stresses are zero on the bingulary S_i . Zero initial conditions for the displacements are also assumed. Equation (5.3) is well known and can convidently be used for deriving representations of the displacement in terms of Green's tensor (representation theorem, see Green 1970).

Assuming that the time Fourier transform of displacements and sources exist, equation (5.3) can be transformed into the frequency domain and written as

$$\int_{\Omega} [\hat{u}_{i}(\mathbf{x},\omega)\hat{f}_{i}^{i}(\mathbf{x},\omega) - \hat{f}_{i}(\mathbf{x},\omega)\hat{u}_{i}^{i}(\mathbf{x},\omega)]d\Omega = 0.$$
(5.4)

Equation 15.11 can also be expressed in terms of the particle velocity $\hat{c}_i(\mathbf{x}, \pm) = i\pm \hat{c}_i(\mathbf{x}, \pm)$ by multiplying both sides with $i\omega$.

$$\int_{\Omega} [e_i(\mathbf{x}, \omega) \int_{\Omega} [\mathbf{x}, \omega) - f_i(\mathbf{x}, \omega) i f_i(\mathbf{x}, \omega)] \partial \Omega = 0, \qquad (5.5)$$

In the time domain, equation (5.5) reads

$$\int_{\Omega} [v_i(\mathbf{x},t) + f_i'(\mathbf{x},t) + [f_i(\mathbf{x},t) + v_i'(\mathbf{x},t)] d\Theta = 0, \qquad (\alpha(6))$$

In the special case that the sources f_i and f'_i have the same time dependence and can be written as

$$f_i(\mathbf{x}, t) = h(Og_i(\mathbf{x}), \quad f_i^{i_1} \mathbf{x}, t) = h(t)g_i^{i_1}(\mathbf{x})$$
(5.7)

equation (5.5) reads

$$\int_{\Omega} [i_{i}(\mathbf{x},\omega)g_{i}'(\mathbf{x}) - g_{i}(\mathbf{x})i_{i}''(\mathbf{x},\omega)]d\Theta = 0.$$
(5.8)

In the time domain equation 55.8) reads

$$\int_{\Omega} [r_{i}(\mathbf{x}, t)g_{i}'(\mathbf{x}) - g_{i}(\mathbf{x})r_{i}'(\mathbf{x}, t)]d\Omega = 0.$$
(5.9)

5.3 Reciprocity of particle velocity. Monopoles

In the following discussion, the indices *m* and *p* indicate either *x*, *y* or z. The spatial part g_i of the source f_i is referred to as the body force. To indicate the direction of the body force, a superscript is used so that the *i*-th component g_i^m of a body force atting at $\mathbf{x} = \mathbf{x}$, in the *m*-direction is specified by

$$g_{0}^{m}(\mathbf{x}; \mathbf{x}_{0}) = 2(\mathbf{x} - \mathbf{x}_{0})\delta_{m},$$
 (5.10)

where $d(\mathbf{x})$ and δ_{pn} are Dirac's and Kronecker's delta functions, respectively. The rath component g_{i}^{n} of a loody force acting at $\mathbf{x} \in \mathbf{x}_{0}^{n}$ in the *p*-direction is, similarly, given by

$$u_i^{0}(\mathbf{x}_i|\mathbf{x}_i) = \delta(\mathbf{x}_i - \mathbf{x}_i^2)\delta_{ij}$$
 (5.11)

We refer to body forces of the type given by equations (5.10) and (5.11) as promotes in the following formulation, we use a superscript on the particle velocity to calicate the direction of the corresponding holy force. Then, $c_{i}^{(0)}$ indicates the *i*-th component of the particle velocity due to a body force acting in the *m*-direction, while $c_{i}^{(0)}$ indicates the *i*-th component of the particle velocity due to a body force acting in the particle velocity. A addition, we indicate the position of the source in the argument of the particle velocity. A complete specification of the *i*-th component of the particle velocity, Δ acting at \mathbf{x}_{0} in the value to a body force as $t_{i}^{(0)}(\mathbf{x}, t; \mathbf{x}_{0})$. Similarly, we have $t_{i}^{(0)}(\mathbf{x}, t; \mathbf{x}'_{0})$ for the primed system.

Using the above notation, we can write the reciprocity relation (5.9) as

$$\int_{\Omega} [v_i^m(\mathbf{x}, t; \mathbf{x}_i) g_i^r(\mathbf{x}; \mathbf{x}_i^r) + g_i^m(\mathbf{x}, \mathbf{x}_i))]^r(\mathbf{x}, t; \mathbf{x}_i^r) [d\Omega \Rightarrow 0.$$
(5.12)

Substituting expansions (5.10) and (5.11) into expansion (5.12), we obtain

$$\int_{\Omega} e_{\mu}^{\mu}(\mathbf{x}|l) \mathbf{x}_{\nu} \left[\delta(\mathbf{x} - \mathbf{x}_{\nu}^{\prime}) \delta_{\mu} - \delta^{2} \mathbf{x} - \mathbf{x}_{\nu}^{\prime} \delta_{\mu}, e_{\nu}^{\mu}(\mathbf{x},l) \mathbf{x}_{\nu}^{\prime}\right] d\Omega = 0$$
(5.13)

Recalling the properties of Dirac's and Kronocket's functions, we note that equation (5.13) implies

$$v_{\mu}^{(0)}(\mathbf{x}_{\mu}^{\prime}, t^{\prime}|\mathbf{x}_{\mu}) = v_{\mu\nu}^{\prime}(\mathbf{x}_{\nu}, t(\mathbf{x}_{\mu}^{\prime}))$$
(5.14)

This equation reveals a fundamental seminetry of the wave held. In any given experiment, the somer and receiver positions may be interchanged provided that the particle velocity component indices and the lorse component indices are interchanged. Note that this equation only applies to the situation where the source consists of a simple body force. In order to albestrate the interpretation of equation (5.14). Figure 5.1 shows three possible 2-D reciprocal experiments

5.4 Reciprocity of strain

For more complex sources than a body force oriented along one of the coordinate axes, the reciprocity relation will differ from explanen (5.14). Equation (5.9) is, however, valid for an arbitrary spatially distributed source and can be used to derive reciprocity relations for couples of forces . A review of the use of cooples for modeling earthquake sources can be found in Ake and Richards (1980, μ , 50) and Pilant (1979, μ , 366)

5.4.1 Single couples

We consider sources consisting of force couples where the *i*-th component of the body force takes the particular form

$$g_{i}^{\ell,m}(\mathbf{x}, \mathbf{x}_{0}) \approx \partial_{j} \sigma(\mathbf{x} - \mathbf{x}_{0}) \delta_{im} \delta_{im}$$
(5.15)

Here the double superscript out indicates that the force couple depends on the tos and realize tions. Similarly, in the primed system, the source components are specified by

$$g_{i}^{pq}(\mathbf{x}; \mathbf{x}_{0}^{\prime}) = \partial_{i} \delta(\mathbf{x} - \mathbf{x}_{0}^{\prime}) \delta_{0} \delta_{ij}$$
(5.16)



Figure 5.11 240 respectively spectrum is for single forces

The corresponding particle velocities are expressed as $v_1^{(n)}(\mathbf{x}|t; \mathbf{x}_n)$ and $v_1^{(n)}(\mathbf{x},t; \mathbf{x}_n')$ respectively. Following Aki and Richards (1980, $p_1(500)$ the forces in equations (5.15) and (5.16) may be thought of as composed of a simple (paint) force in the positive *m* direction and another force of equal magnitude in the negative realization. These two forces are separated by a small distance in the volice tion. The magnitude of the forces must be thosen such that the product of the distance between the forces and the magnitude is unity. This is this rated by the examples in figures 5.2 and 5.3. The source in the top hand m = n = 1. Then, $q_1^{(1)} = q_1^{(1)} = 0$ and

$$q^{(i)}(\mathbf{x}; 0) = \partial_i \delta(\mathbf{x}),$$
 (5.17)

Consider now the source in the top left experiment of Figure 5.3. Using equation (5.15) and assuming m = 1 and n = 3, we have $y_{i}^{(1)} = y_{i}^{(3)} = 0$ and

$$q_{i}^{(4)}(\mathbf{x};0) = d_{i}\sigma(\mathbf{x})$$
 (5.18)

This body force possesses a nonneut around the q-axis, in contrast to the source considered in Figure 5.2, which has zero moment around the q-axis. Whenever m = n, the body force is referred to as a couple without moment, whereas when $m \neq n$ the corresponding body force is referred to as a couple with moment.



Figure 4.25 2-D examinates permetes for couples without tanamat



Figure 5.3: Some of the 2-D composed experiments for single couples with respond

Substituting equations (5.15) and (5.16) into equation (5.9), we obtain the recuprocity relation for couple forces

$$\partial_{a}[r_{i}^{aaa}(\mathbf{x}_{i}^{\prime}, t; \mathbf{x}_{i})] = \partial_{a}[r_{i}^{aa}(\mathbf{x}_{i}, t; \mathbf{x}_{i}^{\prime})], \qquad (5.19)$$

The interpretation of equation (5.19) is similar to that of equation (5.14), except that the spatial derivatives of the particle velocity are respondired) or the particle velocities themself. The following cases are most relevant.

Single couples without moment

When m = n and p = q in equation (5.19), the derivatives are calculated along the force directions. The resulting couples have orientations depending on those directions. This is illustrated in Figure 5.2 for three different experiments.

Single couples with moment

This situation corresponds to the case $m \neq n$ and $p \neq q$ in equation (5.19). The treating couples have noments. These cases are illustrated in Figure 5.3.

5.4.2 Double couples

Double couple without moment. Dilatation.

Two perpendicular couples without moments constitute a dilatational source. Such couples have the form

$$q_i(\mathbf{x}; \mathbf{x}_0) = \partial_i \delta^i \mathbf{x} - \mathbf{x}_i$$
 (5.20)

aad

$$\eta_i(\mathbf{x}^* \mathbf{x}_{i}^*) = \partial_i \delta(\mathbf{x} - \mathbf{x}_{i}^*), \qquad (5.21)$$

The respective particle-velocity components are $c_i(\mathbf{x},t|\mathbf{x}_0)$ and $c_i(\mathbf{x},t|\mathbf{x}_0')$

Substituting equation (5.20) and (5.21) into (5.2), we obtain:

$$\int_{\partial S} [i_{0}(\mathbf{x},t;\mathbf{x}_{0})\partial_{t}\theta(\mathbf{x}-\mathbf{x}_{0}') + \partial_{t}\theta(\mathbf{x}-\mathbf{x}_{0})i_{0}(\mathbf{x},t;\mathbf{x}_{0}')]d\Theta = 0$$
(5.22)

111

$$\hat{\vartheta}_{i} \mathbf{x}_{i}^{\ell} | \vartheta(\mathbf{x}_{0}) = \hat{\vartheta}(\mathbf{x}_{0}, t; \mathbf{x}_{0}^{\ell}),$$
 (5.23)

where

$$\dot{v} = \partial_{t} v_{t}$$
 (5.24)

(see equation (1411). Equation (5.23) indicates that for a chlatation point source revphysical, the time derivative of the chlatation helds are comproval when the source and receiver are interchanged.

Double couple without moment and monopole force

Let us consider a double couple webinit mannent (dilatation source) at $\mathbf{x} = \mathbf{x}_{0}$

$$g(\mathbf{x}; \mathbf{x}_0) = \partial [\delta(\mathbf{x} - \mathbf{x}_0)]$$
 (5.25)

and a usoropole force at $\mathbf{x} = \mathbf{x}'_{i}$

$$g_{i}^{m}(\mathbf{x}; \mathbf{x}_{n}^{t}) = g_{i} \psi(\mathbf{x} - \mathbf{x}_{i}^{t})\psi_{mi}, \qquad (5.26)$$

where g_0 is a constant with dimensions of 1/Beight. The respective particle-velocity components are $(1 \mathbf{x}, t, \mathbf{x}, t)$ and $(["(\mathbf{x}, t; \mathbf{x}'_t)]$ Substituting equation (5.25) and (5.26) with (5.9), we have

$$\int_{\partial \Omega} v_t(\mathbf{x}, t; \mathbf{x}_0) g_0 \delta(\mathbf{x} - \mathbf{x}_0) \delta_{tot} + \partial_t \delta(\mathbf{x} - \mathbf{x}_0) v_t^{(t)}(\mathbf{x}, t, \mathbf{x}_0^{(t)}) d\Omega = 0.$$
(5.27)

Integration of (5.27) implies

$$g_0(c(\mathbf{x}_0^*, t)\mathbf{x}_0|0)_{on} = \partial_{ab}[c(\mathbf{x}_0, t, \mathbf{x}_0^*) = 0],$$
 (5.28)

which can be written as

$$g_{i\ell} \epsilon_{i\ell}(\mathbf{x}_{i\ell}^{\ell}, t; \mathbf{x}_{\ell}) = \partial_{i\ell}(\mathbf{x}_{\ell}, \ell, \mathbf{x}_{\ell}^{\ell}),$$
 (5.29)

where

$$\sigma_{m} = d_{0} c_{0}^{m}$$
, (5.30)

Equation (5.29) indicates that the particle velocity and time derivative of the dilatation field must be substituted when the source and receiver are interchanged. The case $g_{0,0}$ w_1 is illustrated in Figure 5.4 (top)

The question posed by Nyitrai. Hence and Razavy (1996) regarding reciprocity in Land's problem (see Section 5.1) has then the following answer: the horizontal (vertical) particle velocity due to a dilatation source is reciprocal with the three derivative of the dilatation due to a horizontal (vertical) force.

Double couple without moment and single couple.

for this consider a double couple without mannent at $\mathbf{x} = \mathbf{x}$,

$$g(\mathbf{x}; \mathbf{x}_0) = \partial \delta(\mathbf{x} - \mathbf{x}_0).$$
 (5.31)

and a single couple at $\mathbf{x} = \mathbf{x}_{11}^{\prime}$

$$g_{i}^{\mu\nu}(\mathbf{x}, \mathbf{x}'_{i}) = \partial_{j}\vartheta(\mathbf{x} - \mathbf{x}'_{i})\delta_{\mu\nu}\delta_{\mu\nu}$$
(5.32)

The particle-velocity components are $c_i(\mathbf{x}, t)$ and $c_i^{(n)}(\mathbf{x}, t) \mathbf{x}_n^t$, respectively

Substituting equation (5.31) and (5.32) into (5.9), we obtain

$$\int_{\partial \Omega} [e_i(\mathbf{x},t) \,\mathbf{x}_a] \partial_i \delta(\mathbf{x} - \mathbf{x}_a^i) \partial_{im} \sigma_{in} + \partial_i \phi(\mathbf{x} - \mathbf{x}_a) e^{int}(\mathbf{x},t) \,\mathbf{x}_a^i] d\Omega = 0.$$
 (5.33)

Integration of (5.33) implies

$$\partial_{t} e_{t} (\mathbf{x}_{t}^{t}, t, \mathbf{x}_{0}) \delta_{tm} v_{tm} < \partial_{t} e_{t}^{m} (\mathbf{x}_{tm} t, \mathbf{x}_{0}^{t}) = 0.$$
 (5.34)

which can be written as

$$d_i \left(\boldsymbol{x}_{i}^{\dagger}(\mathbf{x}_{i}^{\dagger}(t;\mathbf{x}_{i})) - d_{ini}\left(\mathbf{x}_{i}(t;\mathbf{x}_{i}^{\dagger})\right) \right)$$

$$(5.35)$$

where

$$\vartheta_m = \vartheta_i v_i^{m_i}$$
, (5.36)

In this case, the time derivative of the dilatation is reciproral with the derivatives of the particle velocity. Two examples are illustrated in Figure 5.1 (mildle and bottom pictures).



Figure 5.1: 200 respectsd experiments for double couples without moder thand single cligdes

5.5 Reciprocity of stress

 λ proper closer of the body forces f_i and f'_i leads to recipracity relations for stress. This occurs for the following forces:

$$J_{1}^{(n)}(\mathbf{x}, t; \mathbf{x}_{0}) = [(\gamma_{1}\gamma_{1})]\mathbf{x}_{0}, t'o_{1}\delta(\mathbf{x} - \mathbf{x}_{0})\delta_{0}] \star h(t)$$

$$(5.37)$$

and

$$f_{1}^{p_{0}}(\mathbf{x}, t; \mathbf{x}_{0}') \simeq [e_{ijk}(\mathbf{x}_{0}', t)\partial_{j})(\mathbf{x} - \mathbf{x}_{0}')e_{kj}\partial_{kj} + h(t)$$
(5.38)

The associated particle-velocity components are $v_i(\mathbf{x}, t, \mathbf{x}_i)$ and $v_i(\mathbf{x}, t, \mathbf{x}'_i)$, respectively, where we have omitted the superscripts for simplicity. The corresponding components of the stress tensor are denoted by $\sigma_{i_1}^{(n)}(\mathbf{x}, t; \mathbf{x}_i)$ and $\sigma_{i_2}^{(n)}(\mathbf{x}, t; \mathbf{x}'_i)$. Substituting equation (5.37) and (5.38) into (5.6), we obtain

$$\int_{\partial A} \{ \psi_1(\mathbf{x}, t; \mathbf{x}_0) + [\psi_{0,S1}(\mathbf{x}_0, t)\partial_1 \delta(\mathbf{x} - \mathbf{x}_0)\partial_{tp} \delta_{tp} + h(t) \}$$

$$v_0(\mathbf{x}, t; \mathbf{x}_0') + [\psi_{0,0})(\mathbf{x}_0, t)\partial_1 \delta(\mathbf{x} - \mathbf{x}_0) \delta_{tp} \delta_{tp} + h(t) \} d\Omega = 0.$$
(5.39)

hitegrating this equation, we obtain

$$\partial_{\mu\nu\rho}(\mathbf{x}_{1i}^{\prime}t)\mathbf{x}_{1}(\mathbf{x}_{1i}) + \partial_{\mu\rho}(\mathbf{x}_{1i}^{\prime}t)\partial_{\mu\rho}\delta_{0\rho}] + \hbar 0$$

 $\partial_{\mu}c_{\mu}(\mathbf{x}_{1i},t)\mathbf{x}_{2i}(\mathbf{x}_{2i},t)\partial_{\mu}(\mathbf{x}_{2i},t)\delta_{\mu\nu}\delta_{1\nu} + \hbar (t_{\mu} = 0).$ (5.40)

We now use the symmetry properties (2.23), to rewrite equation (5.40) as

$$\left[(\gamma_{AI}(\mathbf{x}_{t}^{\prime}, t, \mathbf{x}_{n}) * \partial_{t} c_{t}(\mathbf{x}_{t}^{\prime}, t, \mathbf{x}_{n}) \delta_{0i} \delta_{ij} \right] * h(t)$$

+ $c_{ij1} \cdot (\mathbf{x}_{n}, t; \mathbf{x}_{t}^{\prime}) * \partial_{t} c_{i}(\mathbf{x}_{n}, 0, t; \mathbf{x}_{t}^{\prime}) \delta_{im} \delta_{in} \right] * h(t) = 0.$ (5.11)

ы

$$\sigma_{ij}^{ab}(\mathbf{x}_{i}^{t},t,\mathbf{x}_{0})\delta_{ij} \delta_{ij}^{t} + h(t) = \sigma_{ij}^{aa}(\mathbf{x}_{0},t) \mathbf{x}_{0}^{t}[\delta_{ij},\delta_{ij}] + h(t) = 0, \qquad (5.12)$$

where the stress-strain relation (5.2) and the relation $(\gamma_{pr} * (\partial_t v) + \partial_t v_t) = 2\gamma_{p1} * \partial_t v_t$ have been used. Contraction of indices implies

$$\sigma_{\mu\nu}^{een}(\mathbf{x}_{0}^{i}, t, \mathbf{x}_{0}) * h(t) < \sigma_{\mu\nu\nu}^{ee}(\mathbf{x}_{0}, t, \mathbf{x}_{0}^{i}) * h(t) = 0.$$
 (5.13)

If h is such that h satisfying $h + h = \delta$ exists, it is easy to show that equation (5.13) is equivalent to

$$q_{pq}^{r,n}(\mathbf{x}_{n}^{r}, t; \mathbf{x}_{n}) = a_{nm}^{r,q}(\mathbf{x}_{n}, t; \mathbf{x}_{n}^{r}).$$
(5.11)

The interpretation of equation (5.11) follows. The p_d stress component at \mathbf{x}'_a due to a body force with e-th component given by $f_i^{(n)}$ at \mathbf{x}_a equals the *num* stress component at \mathbf{x}_a due to a body force with e-th component given by $f_i^{(n)}$ and applied at \mathbf{x}'_a .

Figure 5.5 illustrates the source and receiver configuration for an experiment corrasponding to originarity of stress. The sources of the experiments are

$$f_{i}^{-s}(\mathbf{x},t,\mathbf{x}_{0}) = (\phi_{i} + h(t)\partial_{i}\delta(\mathbf{x} + \mathbf{x}_{0})) = f_{i}^{-s}(\mathbf{x},t,\mathbf{x}_{0}) = (\phi_{i} + h(t)\partial_{i}\delta(\mathbf{x} + \mathbf{x}_{0}))$$

and

$$f_1^{35}(\mathbf{x}, t^* \mathbf{x}'_0) = \psi_{11} \star h(t(\partial_t \delta(\mathbf{x} - \mathbf{x}'_0)) - f_3^{35}(\mathbf{x}, t^* \mathbf{x}'_0) - \psi_{31} \star h(t(\partial_t \delta(\mathbf{x} - \mathbf{x}'_0)))$$

1511



Figure 5.5. Source and receiver configuration for comproved stress experiments.

where r_{cr} are the relaxation components in the Voigt's antation. In this case, $\sigma_{cs}^{(1)}$ is equal to $\sigma_{cs}^{(1)}$ when the source and receiver positions are interchanged.

We then conclude that for nany types of sources, such as, for example, dipoles or explosions (chlatations), there is a field that satisfies the reciprocity principle. An example of the application of dur temptority relations can be Saurd, for notation, in oll-dure seismic experiments, since the sources are of dilatational type and the hydrophanes record the pressure field, i.e., the dilatation multiplied by the water bulk modulus. In hand seismic acquisition, an example is the differentiation of the radiation pattern for a point source on a homogeneous hall-space (Lamb's problem). The radiation pattern can be obtained by using reciprocity and the displacements on the half-space surface due to incident plane waves (White, 1960). The reciprocity relations can be useful in borchole seismic exterinjects, where couples and pressure sources and preparers are employed. An example of how not to use the reciprocity principle is given by Gaugi (1980a). It is the case of an explosive source in a bare that is rapped so that the explosion is a pressure source, and displacements are measured on the station using a vertical grophone. An explosion at the surface and a vertical graphone in the brachole will not necessarily provide the configuration that is reciprocal to the htst experiment. The correct reciprocal configuration involves a liveliciplicate in the well and a directional vertical source as the surface. A set of menutical experiments contineing for esciptocity relations obtained in this chapter conb) found in Avutseis and Carciono (2000).

This page intentionally left blank

Chapter 6

Reflection and transmission of plane waves

If a pencil of WHITE light polarised by reflexion is incident at the polarising angle upon any transparent surface, so that the plane of the second reflexion is at right angles to the plane of its primitive polarisation, a portion of the pencil consisting of the mean refrangible rays will lose its reflexibility, and will entirely penetrate the second surface, while another portion of the beam, composed of the blue and red rays, will not lose its reflexibility, but will suffer reflexion and refraction like ordinary light.

David Brewster (Brewster, 1815)

The reflection and transmission problem at an interface between anisotropic anelastic media is a complex phenomenon. A general approach has, in this case, the disadvantage of limiting the depth to which we can study the problem analytically and precludes us from gaining further physical insight into the nature of the problem. In this chapter, the main physical results are illustrated by considering relatively simple cases, that is, propagation of SH waves in the plane of symmetry of monoclinic media, and propagation of qP-qSV waves in a plane containing the axes of symmetry of transversely isotropic media. In both cases, the refracting boundary is plane and perpendicular to the symmetry planes.

The problem of reflection and transmission at an interface between two anelastic transversely isotropic media whose symmetry axes are perpendicular to the interface has a practical application in the exploration for hydrocarbon reservoirs using seismic waves. The interface may separate two finely layered formations whose contact plane is parallel to the stratification. Anelastic rheology models the different attenuation mechanisms resulting from the presence of cracks and fluid saturation.

We have seen in Chapters 3 and 4 that the most relevant difference from the elastic case is the presence of inhomogeneous waves which have a body-wave character, in contrast to the inhomogeneous waves of the elastic type, which propagate along interfaces. For viscoelastic inhomogeneous waves, the angle between the propagation and attenuation vectors is strictly less than 90°, unlike inhomogeneous waves in elastic media. In addition, depending on the inhomogeneity of the wave, its behavior (e.g., phase velocity, attenuation, particle motion) may also differ substantially. Moreover, as we have seen in Chapter 1, in the anisotropic case, the energy-flow direction, in general, does not coincide with the propagation (wavevector) direction, and critical angles occur when the ray (energyflow) direction is parallel to the interface. The theoretical developments presented in this chapter follow from Carcione (1997a,b).

6.1 Reflection and transmission of SH waves

The cross-plane shear problem is one of relative mathematical simplicity and includes the essential physics common to more complicated cases, where multiple and coupled deformations occur (Horgan, 1995). In this sense, analysis of the reflection and transmission of cross-plane shear waves serves as a pilot problem for investigating the influence of anisotropy and/or anelasticity on solution behavior. As is well known, propagation in the plane of mirror symmetry of a monoclinic medium is the most general situation for which cross-plane strain motion exists in all directions – the corresponding waves are also termed type-II S and SH in the geophysical literature (Borcherdt, 1977; Helbig, 1994).

Besides the work of Hayes and Rivlin (1974), who considered a low-loss approximation, the study of wave propagation in anisotropic viscoelastic media is a relatively recent topic. In the following discussion, we consider two monoclinic media with a common mirror plane of symmetry in contact along a plane perpendicular to the symmetry plane. The incidence and refraction planes are assumed to be coincident with this plane of symmetry. Then, an incident cross-plane shear wave will generate reflected and transmitted shear waves without conversion to the coupled quasi-compressional and quasi-shear modes.

The physics of the problem may differ depending on the values of the elasticity constants and the anisotropic dissipation of the upper and lower media. For this reason, we follow a general treatment and, simultaneously, consider a numerical example including the essential physical aspects. In this way, the analysis provides further insight into the nature of the reflection and transmission problem.

6.1.1 Symmetry plane of a homogeneous monoclinic medium

Assume a homogeneous and viscoelastic monoclinic medium with the vertical (x, z)-plane as its single symmetry plane. Then, cross-plane shear waves with particle velocity $\mathbf{v} = v(x, z)\hat{\mathbf{e}}_2$ propagate such that

$$v = i\omega u_0 \exp[i\omega(t - s_1x - s_3z)],$$
 (6.1)

where s_1 and s_3 are the components of the complex-slowness vector, ω is the angular frequency satisfying $\omega \ge 0$, and u_0 is a complex quantity. The slowness and attenuation vectors are given by

$$s_R = (\text{Re}(s_1), \text{Re}(s_3))$$
 (6.2)

and

$$\alpha = -\omega(Im(s_1), Im(s_3)),$$
 (6.3)

respectively, such that the complex-slowness vector is $s = s_R - i(\alpha/\omega)$.

The cross-plane assumption implies that σ_{12} and σ_{32} are the only non-zero components of stress that satisfy the stress-strain relations

$$i\omega\sigma_{12} = p_{46}\partial_3 v + p_{66}\partial_1 v$$
, and $i\omega\sigma_{32} = p_{44}\partial_3 v + p_{46}\partial_1 v$, (6.4)

where p_{IJ} are the complex stiffnesses (see Sections 4.1 and 4.6). These complex stiffnesses equal the real high-frequency limit elasticity constants c_{IJ} in the elastic case.

The complex-slowness relation has the following simple form:

$$F(s_1, s_3) \equiv p_{44}s_3^2 + p_{66}s_1^2 + 2p_{46}s_1s_3 - \rho = 0.$$
 (6.5)

(See the elastic version in equation (1.261).)

Let us assume that the positive z-axis points downwards. In order to distinguish between down and up propagating waves, the slowness relation is solved for s_3 , given the horizontal slowness s_1 . It yields

$$s_{3\pm} = \frac{1}{p_{44}} \left(-p_{46}s_1 \pm pv \sqrt{\rho p_{44} - p^2 s_1^2} \right),$$
 (6.6)

where

$$p^2 = p_{44}p_{66} - p_{46}^2$$
(6.7)

and $pv\sqrt{w}$ denotes the principal value of the square root of the complex number w. In principle, the + sign corresponds to downward or +z propagating waves, while the - sign corresponds to upward or -z propagating waves.

We recall that, as shown in Section 1.4.2, the group velocity equals the energy velocity only when there is no attenuation. Therefore, analysis of the physics requires explicit calculation of the energy velocity, since the concept of group velocity loses its physical meaning in anelastic media (see Section 4.4.5). The mean energy flux or time-averaged Umov-Poynting vector $\langle \mathbf{p} \rangle$ is the real part of the corresponding complex vector

$$\mathbf{p} = -\frac{1}{2}(\sigma_{12}\hat{\mathbf{e}}_1 + \sigma_{32}\hat{\mathbf{e}}_3)v^* \qquad (6.8)$$

(equation (4.111)). Substituting the plane wave (6.1) and the stress-strain relations (6.4) into equation (6.8), we obtain

$$\mathbf{p} = \frac{1}{2}\omega^2 |u_0|^2 \exp\{2\omega [\operatorname{Im}(s_1)x + \operatorname{Im}(s_3)z]\}(X\hat{\mathbf{e}}_1 + Z\hat{\mathbf{e}}_3),$$
(6.9)

where

$$X = p_{66}s_1 + p_{46}s_3$$
, and $Z = p_{46}s_1 + p_{44}s_3$. (6.10)

For time harmonic fields, the time-averaged strain- and dissipated-energy densities, $\langle V \rangle$ and $\langle D \rangle$, can be obtained from a complex strain-energy density Φ . This can be deduced from equations (4.53), (4.54), (4.85) and (6.4). Hence, we have

$$\Phi = \frac{1}{2} \mathbf{e}^{\top} \cdot \mathbf{P} \cdot \mathbf{e}^{*}, \tag{6.11}$$

which for SH waves propagating in a monoclinic medium is given by

$$\Phi = \frac{1}{2} \left\{ p_{44} \left| \frac{\partial_3 v}{i\omega} \right|^2 + p_{66} \left| \frac{\partial_1 v}{i\omega} \right|^2 + 2p_{46} \operatorname{Re} \left[\frac{\partial_3 v}{i\omega} \left(\frac{\partial_1 v}{i\omega} \right)^* \right] \right\}.$$
(6.12)

(The demonstration is left to the reader.) Then,

$$\langle V \rangle = \frac{1}{2} \operatorname{Re}(\Phi), \quad \langle D \rangle = \operatorname{Im}(\Phi).$$
 (6.13)

Substituting the plane wave (6.1) into (6.12), we find that the energy densities become

$$\langle V \rangle = \frac{1}{4} \omega^2 |u_0|^2 \exp\{2\omega [\operatorname{Im}(s_1)x + \operatorname{Im}(s_3)z]\} \operatorname{Re}(\varrho)$$
 (6.14)

and

$$\langle D \rangle = \frac{1}{2} \omega^2 |u_0|^2 \exp\{2\omega [\operatorname{Im}(s_1)x + \operatorname{Im}(s_3)z]\} \operatorname{Im}(\varrho),$$
 (6.15)

where

$$\varrho = p_{44}|s_3|^2 + p_{66}|s_1|^2 + 2p_{46}\operatorname{Re}(s_1^*s_3).$$
(6.16)

From equation (4.52), we obtain the time-averaged kinetic-energy density, namely,

$$\langle T \rangle = \frac{1}{4}\rho |v^2| = \frac{1}{4}\rho \omega^2 |u_0|^2 \exp\{2\omega[\operatorname{Im}(s_1)x + \operatorname{Im}(s_3)z]\}.$$
 (6.17)

6.1.2 Complex stiffnesses of the incidence and transmission media

A realistic viscoelastic model is the Zener model (see Section 2.4.3). It satisfies causality and gives relaxation and creep functions in agreement with experimental results (e.g., aluminum (Zener, 1948) and shale (Johnston, 1987)).

We assign different Zener elements to p_{44} and p_{66} in order to define the attenuation (or quality factor) along the horizontal and vertical directions (x- and z-axes), respectively. Hence, the stiffnesses are

$$p_{44} = c_{44}M_1$$
, $p_{66} = c_{66}M_2$, $p_{46} = c_{46}$, (6.18)

where

$$M_{\nu} = \frac{\tau_{\sigma\nu}}{\tau_{e\nu}} \left(\frac{1 + i\omega\tau_{e\nu}}{1 + i\omega\tau_{\sigma\nu}} \right), \quad \nu = 1, 2 \quad (6.19)$$

are the complex moduli (see Section 2.4.3). The relaxation times are given by

$$\tau_{e\nu} = \frac{\tau_0}{Q_{0\nu}} \left(\sqrt{Q_{0\nu}^2 + 1} + 1 \right), \quad \tau_{\sigma\nu} = \frac{\tau_0}{Q_{0\nu}} \left(\sqrt{Q_{0\nu}^2 + 1} - 1 \right), \quad (6.20)$$

where τ_0 is a characteristic relaxation time and $Q_{0\nu}$ is a characteristic quality factor. An alternative form of the complex modulus is given by equation (4.6). It can be shown from equations (2.201), (4.92) and (4.106), that the quality factors for homogeneous waves along the axes are

$$Q_{\nu} = Q_{0\nu} \left(\frac{1 + \omega^2 \tau_0^2}{2\omega \tau_0} \right). \tag{6.21}$$

Then, $1/\tau_0$ is the angular frequency for which the quality factor has the minimum value $Q_{0\nu}$. The choice $\tau_0 = \sqrt{\tau_{e1}\tau_{\sigma1}} = \sqrt{\tau_{e2}\tau_{\sigma2}}$ implies that the maximum dissipation for both mechanisms occurs at the same frequency. As $\omega \to \infty$, $M_\nu \to 1$ and the complex stiffnesses p_{IJ} approach the unrelaxed elasticity constants c_{IJ} .

186

6.1 Reflection and transmission of SH waves

In the reflection-transmission problem, the upper medium is defined by the properties c_{IJ} , $Q_{0\nu}$ and τ_0 , and the lower medium is defined by the corresponding primed properties c'_{IJ} , $Q'_{0\nu}$ and τ'_0 . The numerical example assumes

$$c_{44} = 9.68 \text{ GPa}, \quad Q_{01} = 10,$$

 $c_{66} = 12.5 \text{ GPa}, \quad Q_{02} = 20,$
 $c'_{44} = 19.6 \text{ GPa}, \quad Q'_{01} = 20,$
 $c'_{66} = 25.6 \text{ GPa}, \quad Q'_{02} = 30.$
(6.22)

Moreover,

$$c_{46} = -\frac{1}{2}\sqrt{c_{44}c_{66}}, \quad c'_{46} = \frac{1}{2}\sqrt{c'_{44}c'_{66}},$$
 (6.23)

and

$$\rho = 2 \text{ gr/cm}^3$$
, $\rho' = 2.5 \text{ gr/cm}^3$. (6.24)

The characteristic relaxation time is taken as $\tau_0 = \tau'_0 = (2\pi f_0)^{-1}$, i.e., the maximum attenuation occurs at a frequency f_0 . The above parameters give horizontal and vertical (elastic or unrelaxed) phase velocities of 2500 m/s and 2200 m/s, respectively, for the upper medium, and 3200 m/s and 2800 m/s, respectively, for the lower medium.

Several subcases treated in the analysis make use of the following limiting situations:

elastic :
$$Q_{0\nu} = Q'_{0\nu} = \infty$$
 $(\tau_{e\nu} = \tau_{\sigma\nu}, \tau'_{e\nu} = \tau'_{\sigma\nu})$ or $M_{\nu} = M'_{\nu} = 1$,
isotropic : $p_{44} = p_{66} = \mu$, $p'_{44} = p'_{66} = \mu'$, $p_{46} = p'_{46} = 0$, (6.25)
transversely isotropic : $p_{46} = p'_{46} = 0$.

Note, however, that the condition $p_{46} = p'_{46} = 0$ does not necessarily mean that the media are transversely isotropic (see Section 1.2.1).

The analysis of the problem is carried out at the frequency f_0 and, therefore, its value is immaterial, because $\omega \tau_0 = 1$ Moreover, at a fixed frequency, the analysis does not depend on the viscoelastic model.

6.1.3 Reflection and transmission coefficients

Let us assume that the incident, reflected and transmitted waves are identified by the superscripts I, R and T. The solution to the problem parallels those of the anisotropic elastic case (Section 1.9.1) and isotropic viscoelastic case (Section 3.8).

The particle velocity of the incident wave can be written as

$$v' = i\omega \exp[i\omega(t - s_1x - s'_3z)],$$
 (6.26)

where, for simplicity, the superscript I in the horizontal slowness has been omitted here and in all the subsequent analysis.

Inhomogeneous viscoelastic plane waves have the property that equiphase planes – planes normal to the slowness vector – do not coincide with equiamplitude planes – planes normal to the attenuation vector. When the directions of propagation and attenuation coincide, the wave is called homogeneous. For a homogeneous wave (see Section 4.2),

$$s_1 = \sin \theta^I / v_c(\theta^I), \quad s_3^I = \cos \theta^I / v_c(\theta^I), \quad (6.27)$$

where θ^{I} is the incidence propagation – or attenuation – angle (see Figure 6.1), and

$$v_c(\theta) = \sqrt{(p_{44}\cos^2\theta + p_{66}\sin^2\theta + p_{46}\sin 2\theta)/\rho}$$
(6.28)

is the complex velocity, according to the dispersion relation (6.5) and equations (4.28) and (4.33).

As in the isotropic viscoelastic case (Section 3.8), the boundary conditions – continuity of v and σ_{32} – give the reflection and transmission coefficients. Snell's law, i.e., the continuity of the horizontal complex slowness,

$$s_1^R = s_1^T = s_1$$
, (6.29)

(see Section 3.5) is a necessary condition for the existence of the boundary conditions.

Denoting the reflection and transmission coefficients by R_{SS} and T_{SS} , we express the particle velocities of the reflected and transmitted waves as

$$v^{R} = i\omega R_{SS} \exp[i\omega(t - s_{1}x - s_{3}^{R}z)] \qquad (6.30)$$

and

$$v^T = i\omega T_{SS} \exp[i\omega(t - s_1x - s_3^T z)],$$
 (6.31)

respectively.

Then, continuity of v and σ_{32} at z = 0 gives

$$T_{SS} = 1 + R_{SS}$$
 (6.32)

and

$$Z_I + R_{SS}Z_T = T_{SS}Z_T$$
, (6.33)

which have the following solution:

$$R_{SS} = \frac{Z^I - Z^T}{Z^T - Z^R}, \quad T_{SS} = \frac{Z^I - Z^R}{Z^T - Z^R}.$$
 (6.34)

Since both the incident and reflected waves satisfy the slowness relation (6.5), the vertical slowness s_3^R can be obtained by subtracting $F(s_1, s_3^I)$ from $F(s_1, s_3^R)$ and assuming $s_3^R \neq s_3^I$. This yields

$$s_3^R = -\left(s_3' + \frac{2p_{46}}{p_{44}}s_1\right).$$
 (6.35)

Then, using equation (6.10), we obtain

$$Z^{R} = -Z^{I}$$
, (6.36)

and the reflection and transmission coefficients (6.34) become

$$R_{\rm SS} = \frac{Z^I - Z^T}{Z^I + Z^T}, \qquad T_{\rm SS} = \frac{2Z^I}{Z^I + Z^T},$$
 (6.37)

where

$$Z^{I} = p_{46}s_{1} + p_{44}s_{3}^{I} \quad \text{and} \quad Z^{T} = p_{46}'s_{1} + p_{44}'s_{3}^{T}.$$
(6.38)

6.1 Reflection and transmission of SH waves

The slowness relation (6.5) of the transmission medium gives s_3^T in terms of s_1 :

$$s_3^T = \frac{1}{p_{44}'} \left(-p_{46}' s_1 + pv \sqrt{\rho' p_{44}' - {p'}^2 s_1^2} \right), \tag{6.39}$$

with

$$p'^2 = p'_{44} p'_{66} - p'_{46}^2. \tag{6.40}$$

Alternatively, from equation (6.10),

$$s_3^T = \frac{1}{p'_{44}} \left(Z^T - p'_{46} s_1 \right).$$
 (6.41)

Figure 6.1 represents the incident (I), reflected (R) and transmitted (T) waves at a boundary between two linear viscoelastic and monoclinic media. The angles θ , δ and ψ denote the propagation, attenuation and Umov-Poynting vector (energy) directions. Note that the propagation and energy directions do not necessarily coincide. Moreover, $|\theta - \delta|$ may exceed 90° in anisotropic viscoelastic media, while $|\theta - \delta|$ is strictly less than 90° in isotropic media (see equation (3.36)).



Figure 6.1: Incident (I), reflected (R) and transmitted (T) waves at a boundary between two linear viscoelastic and monoclinic media. The angles θ , δ and ψ denote the propagation, attenuation and Umov-Poynting vector (energy) directions. The reflection angle is negative as shown.

189



Figure 6.2: Limiting rays for the fan of incidence angles. (a) $\theta^I = 24.76^{\circ}$ and (b) $\theta^I = 58.15^{\circ}$ (23.75° and 60.39°, respectively, in the elastic case). They are determined by the condition that the energy propagation direction is downwards (+z) and to the right (+x), i.e., $0 \le \psi^I \le 90^{\circ}$. The larger curve is the slowness for homogeneous waves in the incidence medium and the other curve is the slowness for homogeneous waves in the transmission medium.

6.1.4 Propagation, attenuation and energy directions

The fan of incident rays is determined by the condition that the energy propagation direction is downwards (+z) and to the right (+x). The limiting rays for the numerical example are represented in Figures 6.2a ($\theta^I = 24.76^\circ$) and 6.2b ($\theta^I = 58.15^\circ$) (23.75° and 60.39°, respectively, in the elastic case). The larger curve is the slowness for homogeneous waves in the incidence medium, and the other curve is the slowness for homogeneous waves in the transmission medium. As we have seen in Section 4.4.4, the energy direction is not perpendicular to the corresponding slowness curve for all frequencies. The perpendicularity property is only verified in the low- and high-frequency limits.

Given the components of the complex-slowness vector, the propagation and attenuation angles θ and δ for all the waves are

$$\tan \theta = \frac{\operatorname{Re}(s_1)}{\operatorname{Re}(s_3)} \tag{6.42}$$

and

$$\tan \delta = \frac{\text{Im}(s_1)}{\text{Im}(s_3)}.$$
 (6.43)

These equations can be easily verified for the incident wave (6.26), for which $\delta^{I} = \theta^{I}$, by virtue of equation (6.27).

Moreover, from equations (6.30) and (6.35), the reflection propagation and attenuation angles are

$$\tan \theta^{R} = -\frac{\text{Re}(s_{1})}{\text{Re}(s_{3}^{I} + 2p_{46}p_{44}^{-1}s_{1})} \qquad (6.44)$$

6.1 Reflection and transmission of SH waves

and

$$\tan \delta^{R} = -\frac{\text{Im}(s_{1})}{\text{Im}(s_{3}^{I} + 2p_{46}p_{44}^{-1}s_{1})}, \quad (6.45)$$

respectively. Unlike the isotropic case, the reflected wave is, in general, inhomogeneous.

Theorem 1: If the incident wave is homogeneous and not normally incident, the reflected wave is homogeneous if and only if $Im(p_{46}/p_{44}) = 0$.

Proof: Assume that the reflected wave is homogeneous. Then, from equations (6.44) and (6.45), $\tan \theta^R = \tan \delta^R$ implies that $\operatorname{Im}[s_1^*(s_3^I + 2p_{46}p_{44}^{-1}s_1)] = 0$. Assuming $\theta^I \neq 0$ and using equation (6.27), we obtain $\operatorname{Im}(p_{46}/p_{44}) = 0$. The same reasoning shows that this condition implies a homogeneous reflected wave.

A corollary of Theorem 1 is

Corollary 1.1: If the upper medium has $p_{46} = 0$, the reflected wave is homogeneous. This follows immediately from Theorem 1.

In the elastic case, all the quantities in equation (6.44) are real, and the incidence and reflection angles are related by

$$\cot \theta^R = -\left(\cot \theta^I + 2\frac{c_{46}}{c_{44}}\right). \quad (6.46)$$

From equation (6.31), the transmission propagation and attenuation angles are

$$\tan \theta^T = \frac{\operatorname{Re}(s_1)}{\operatorname{Re}(s_3^T)} \qquad (6.47)$$

and

$$\tan \delta^T = \frac{\text{Im}(s_1)}{\text{Im}(s_3^T)}, \quad (6.48)$$

respectively. In general, the transmitted wave is inhomogeneous.

Theorem 2: If the transmission medium is elastic and the incidence is non-normal, the attenuation and Umov-Poynting vectors of the transmitted wave are perpendicular, i.e., $|\psi^T - \delta^T| = 90^{\circ}$.

Proof: In the first place, α^T must be different from zero at non-normal incidence, because the incident wave is homogeneous, and, therefore, Snell's law requires a nonzero component of the attenuation vector. The time-averaged dissipated-energy density for cross-plane inhomogeneous waves in the plane of symmetry of a monoclinic medium is given by equation (6.15) (see also Krebes and Le, 1994; and Carcione and Cavallini, 1995a). For the transmitted wave, it is

$$\langle D^T \rangle = \frac{1}{2} \omega^2 |T_{\rm SS}|^2 \exp\{2\omega [\operatorname{Im}(s_1)x + \operatorname{Im}(s_3^T)z]\} \operatorname{Im}(\varrho^T),$$
 (6.49)

where

$$\varrho^{T} = p_{44}' |s_{3}^{T}|^{2} + p_{66}' |s_{1}|^{2} + 2p_{46}' \operatorname{Re}(s_{1}^{*} s_{3}^{T}).$$
(6.50)

Since the medium is elastic $(p'_{IJ} \rightarrow c'_{IJ})$, ϱ^T is real and $\langle D^T \rangle = 0$. On the other hand, equations (4.83) and (4.85) imply that an inhomogeneous wave satisfies

$$\langle D^T \rangle = \frac{2}{\omega} \boldsymbol{\alpha}^T \cdot \langle \mathbf{p}^T \rangle.$$
 (6.51)

Since the energy loss is zero, it is clear from equation (6.51) that α^T is perpendicular to the average Umov-Poynting vector (\mathbf{p}^T) .

The existence of an inhomogeneous plane wave propagating away from the interface in elastic media, is not intuitively obvious, since it is not the usual interface wave with its attenuation vector perpendicular to the boundary. Such body waves appear, for instance, in the expansion of a spherical wave (Brekhovskikh, 1960, p. 240).

Corollary 2.1: Theorem 2 implies that, in general, the attenuation direction of the transmitted wave is not perpendicular to the propagation direction. That is, $\alpha^T \cdot \mathbf{s}_R^T \neq 0$, or

$$\text{Re}(s_1)\text{Im}(s_1) + \text{Re}(s_3^T)\text{Im}(s_3^T) \neq 0,$$
 (6.52)

which implies $\text{Im}(s_1^2 + s_3^{T^2}) = 0$. The orthogonality property only applies in the isotropic case (Romeo, 1994). Assume, for simplicity, transverse isotropy. Using (6.52) and the slowness relation (6.5), we obtain

$$\text{Im}(s_1^2)(c_{66}' - c_{44}') = 0,$$
 (6.53)

which gives $c'_{66} = c'_{44}$, that is, the isotropic case.

Proposition 1: If the incidence medium is elastic, the attenuation of the transmitted wave is perpendicular to the interface.

This result follows immediately from equation (6.48), since s_1 real (see equation (6.27)) implies $\delta^T = 0$.

The expressions for the time-averaged Umov-Poynting vectors of the reflected and transmitted waves, are obtained from equation (6.9), with $u_0 = R_{SS}$ and $u_0 = T_{SS}$, respectively. Then, the angles of the reflection and transmission energy vectors are obtained from

$$\tan \psi' = \frac{\operatorname{Re}(X^I)}{\operatorname{Re}(Z^I)},\tag{6.54}$$

$$\tan \psi^{R} = \frac{\operatorname{Re}(X^{R})}{\operatorname{Re}(Z^{R})} \qquad (6.55)$$

and

$$\tan \psi^T = \frac{\text{Re}(X^T)}{\text{Re}(Z^T)}, \quad (6.56)$$

respectively. From equations (6.10) and (6.35) $Z^R = -Z^I$ and $X^R = X^I - 2p_{46}p_{44}^{-1}Z^I$; therefore,

$$\tan \psi^{R} = \frac{2 \operatorname{Re}(p_{46} p_{44}^{-1} Z^{I})}{\operatorname{Re}(Z^{I})} - \tan \psi^{I}. \qquad (6.57)$$

In the elastic case p_{46} and p_{44} are real and

$$\tan \psi^R = 2 \frac{c_{46}}{c_{44}} - \tan \psi^I.$$
 (6.58)

In the evaluation of each angle, particular attention should be given to the choice of the branch of the arctangent.

Figure 6.3 represents the propagation, attenuation and energy angles for the fan of incident rays. Note that the energy angle of the incident wave satisfies $0^o \le \psi^I \le 90^o$ and that the inhomogeneity angles of the reflected and transmitted waves $-|\theta^R - \delta^R|$ and



Figure 6.3: Propagation, attenuation and energy angles for the incident, reflected and transmitted waves versus the incidence angle θ^{I} .

 $|\theta^T - \delta^T|$, respectively – never exceed 90°. However, consider a transmission medium with stronger dissipation, for instance, $Q'_{01} = 2$ and $Q'_{02} = 3$. In this case, $|\theta^T - \delta^T| > 90°$ for $\theta^I \ge 50.46°$, meaning that the amplitude of the transmitted wave grows in the direction of phase propagation. A physical interpretation of this phenomenon is given by Krebes and Le (1994) who show that the amplitude of an inhomogeneous wave decays in the direction of energy propagation, i.e., in our case, $|\psi^T - \delta^T|$ is always less than 90°. Indeed, since the energy loss is always positive, equation (6.51) implies that the magnitude of the angle between α^T and $\langle \mathbf{p}^T \rangle$ is always strictly less than 90°.

Proposition 2: There is an incidence angle θ_0^I such that the incidence and reflection propagation directions coincide, i.e., $\theta_0^I - \theta^R = 180^o$.

The angle can be found by equating (6.42) (for the incident wave) with (6.44) and using equation (6.10). This yields

$$\text{Re}(Z^{I}/p_{44}) = 0,$$
 (6.59)

whose solution is $\theta_0^I = 58.15^{\circ}$, which corresponds to Figure 6.2b. In the elastic case, we obtain

$$\theta_0^I = -\arctan(c_{44}/c_{46}),$$
 (6.60)

whose solution is $\theta_0^I = 60.39^{\circ}$. The angle is 90° in the isotropic case.

Proposition 3: There is an incidence angle θ_1^I such that the reflection and transmission propagation directions coincide, i.e., $\theta^T - \theta^R = 180^{\circ}$.

The angle is obtained from equations (6.44) and (6.47) and the solution is $\theta_1^I = 33.40^\circ$, with $\theta^R = -74.46^\circ$. There is an explicit expression in the elastic case that can be obtained from equations (6.27), (6.28), (6.39), (6.44) and (6.47). It is

$$\tan \theta_1^I = (-b - \sqrt{b^2 - 4ac})/(2a),$$
 (6.61)
where

$$a = \rho' c_{66} - \rho c'_{66} + 4\rho c_{46} (c'_{66}c_{44} - c_{46}c'_{44})/c^2_{44},$$

$$b = 2(\rho' c_{46} + \rho c'_{46} - 2\rho c_{46}c'_{44}/c_{44}),$$

$$c = \rho' c_{44} - \rho c'_{44}.$$
(6.62)

The solutions are $\theta_1^I = 34.96^o$ and $\theta^R = -73.63^o$. In the isotropic case, a = c, b = 0 and there is no solution.



Figure 6.4: At the incidence angle $\theta_1^I = 33.40^\circ$, the reflection and transmission propagation directions coincide. However, note that the Umov-Poynting vector of the transmitted wave (empty arrow) points downward.

This situation is shown in Figure 6.4, where the Umov-Poynting and attenuation vectors of the reflected wave point upward and downward, respectively, while the Umov-Poynting and attenuation vectors of the transmitted wave point downward and upward, respectively. Thus, there is no contradiction since the energy of the transmitted wave is actually pointing to the lower medium.

Proposition 4: There is an incidence angle θ_2^I such that the propagation direction of the incident wave coincides with the corresponding Umov-Poynting vector direction, i.e., $\theta^I = \psi^I = \theta_2^I$. This angle is associated with the symmetry axis of the incidence medium, which is a pure mode direction where the waves behave as in isotropic media.

From equations (6.42) and (6.54), we note that this proposition is verified when

$$\frac{\operatorname{Re}(s_1)}{\operatorname{Re}(s_3^I)} = \frac{\operatorname{Re}(X^I)}{\operatorname{Re}(Z^I)}.$$
(6.63)

Using equations (6.10) and (6.27), and after some algebra, we obtain an approximation for $Q_{0\nu} \gg 1$ (Im $(v_c) \ll \text{Re}(v_c)$):

$$\tan \theta_2^I = \left\{ \operatorname{Re}(p_{66} - p_{44}) - \sqrt{[\operatorname{Re}(p_{66} - p_{44})]^2 + 4[\operatorname{Re}(p_{46})]^2} \right\} / [2\operatorname{Re}(p_{46})].$$
(6.64)

The solution is $\theta_2^I = 36.99^o$ (the exact solution is 37.04°). In the isotropic case, $\psi^I = \theta^I$ for all incident rays.

Proposition 5: There is an incidence angle θ_3^I such that the propagation direction of the reflected wave coincides with the corresponding Umov-Poynting vector direction, i.e., $\theta^R = \psi^R$. From equations (6.44) and (6.55), we note that this proposition is verified when

$$\frac{\operatorname{Re}(s_1)}{\operatorname{Re}(s_3^R)} = \frac{\operatorname{Re}(X^R)}{\operatorname{Re}(Z^R)}.$$
(6.65)

The solutions are $\theta_3^I = 26.74^o$ and $\theta^R = -53.30^o$.

In the elastic case,

$$\tan \theta_3^I = (-b - \sqrt{b^2 - 4ac})/(2a), \tag{6.66}$$

where

$$a = c_{46} \left(\frac{2d}{c_{44}^2} - 1 \right), \quad b = \frac{c_{44}}{c_{46}} a - c_{66}, \quad c = -c_{46},$$
 (6.67)

with

$$d = c_{44}c_{66} - 2c_{46}^2$$
. (6.68)

The corresponding reflection angle is obtained from equations (6.44) and (6.55), and given by

$$\tan \theta_3^R = \left[c_{66} - c_{44} + \sqrt{(c_{66} - c_{44})^2 + 4c_{46}^2} \right] / (2c_{46}). \tag{6.69}$$

The solutions are $\theta_3^I = 27.61^o$ and $\theta_3^R = -52.19^o$. In the isotropic case, $\psi^R = \theta^R$ for all incident rays.

Proposition 6: An incident wave whose energy-flux vector is parallel to the interface ($\operatorname{Re}(Z^{I}) = 0$, see (6.9)) generates a reflected wave whose energy-flux vector is parallel to the interface ($\operatorname{Re}(Z^{R}) = 0$). Moreover, in the lossless case and beyond the critical angle, the energy-flux vector of the transmitted wave is parallel to the interface, i.e., $\operatorname{Re}(Z^{T}) = 0$.

This first statement can be deduced from equation (6.36). Moreover, from equations (6.39) and (6.41),

$$Z^T = pv \sqrt{\rho' c'_{44} - c'^2 s_1^2}.$$
 (6.70)

Beyond the critical angle, the horizontal slowness s_1 is greater than $\sqrt{\rho' c'_{44}}/c'$, where $c' = c'_{44}c'_{66} - c'_{46}$. Therefore, the quantity inside the square root becomes negative and $\operatorname{Re}(Z^T) = 0$.

6.1.5 Brewster and critical angles

In 1815, David Brewster, basing his observations on an experiment by Malus, noted the existence of an angle (θ_B) such that: if light is incident under this angle, the electric vector of the reflected light has no component in the plane of incidence (Born and Wolf, 1964, p. 43). When this happens, $\theta_B + \theta^T = 90^\circ$ and the reflection coefficient of the wave with the electric vector in the plane of incidence vanishes. Here, we define the Brewster angle as the incidence angle for which $R_{SS} = 0$ (in elastodynamics, $\theta_B + \theta^T \neq 90^\circ$ in general).

From equation (6.34), this occurs when $Z^{I} = Z^{T}$, or from (6.10), when

$$p_{46}s_1 + p_{44}s_3' = p_{46}'s_1 + p_{44}'s_3^T, ag{6.71}$$

Using (6.27), (6.28) and (6.39), we see that equation (6.71) yields the following solution

$$\cot \theta_B = \left(-b \pm pv\sqrt{b^2 - 4ac}\right)/(2a), \qquad (6.72)$$

where

$$a = p_{44}(\rho p_{44} - \rho' p'_{44})/\rho, \quad b = 2p_{46}a/p_{44},$$
 (6.73)

and

$$c = p_{46}^2 - p_{46}'^2 - p_{44}'(\rho' p_{66} - \rho p_{66}')/\rho.$$
 (6.74)

In general, $\cot \theta_B$ is complex and there is no Brewster angle. In the elastic limit of the example, the Brewster angle is $\theta_B = 32.34^o$ (see Figure 6.5).



Figure 6.5: Absolute values of the reflection and transmission coefficients versus the incidence angle for the elastic (dotted line) and viscoelastic (solid line) cases ($\theta_P = 31.38^\circ$, $\theta_B = 32.34^\circ$ and $\theta_C = 36.44^\circ$).

In the isotropic viscoelastic case, the solution is

ć

$$\cot \theta_B = \pm pv \sqrt{\frac{\rho' - \rho \mu'/\mu}{\rho \mu/\mu' - \rho'}}, \qquad (6.75)$$

which is generally complex. The Brewster angle exists only in rare instances. For example, $\cot \theta_B$ is real for $\operatorname{Im}(\mu/\mu')=0$. In isotropic media, the complex velocity (6.28) is simply $v_c = \sqrt{\mu/\rho}$. Thus, the quality factor (4.92) for homogeneous waves in isotropic media is $Q = Q_H = \operatorname{Re}(\mu)/\operatorname{Im}(\mu)$. The condition $\operatorname{Im}(\mu/\mu') = 0$ implies that the Brewster angle exists when $Q_H = Q'_H$, where $Q'_H = \operatorname{Re}(\mu')/\operatorname{Im}(\mu')$.

In the lossless case and when $\rho = \rho'$, the reflected and transmitted rays are perpendicular to each other at the Brewster angle, i.e., $\theta_B + \theta^T = 90^\circ$. This property can be proved by using Snell's law and equation (6.75) (this exercise is left to the reader). On

the basis of the acoustic-electromagnetic mathematical analogy (Carcione and Cavallini, 1995b), the magnetic permeability is equivalent to the material density and the dielectric permittivity is equivalent to the reciprocal of the shear modulus (see Chapter 8). There is then a complete analogy between the reflection-transmission problem for isotropic, lossless acoustic media of equal density and the same problem in electromagnetism, where the media have zero conductivity and their magnetic permeability are similar (perfectly transparent media, see Born and Wolf (1964, p. 38)).

In anisotropic media, two singular angles can be defined depending on the orientation of both the propagation and the Umov-Poynting vectors with respect to the interface. The pseudocritical angle θ_P is defined as the angle of incidence for which the transmitted slowness vector is parallel to the interface. In Auld (1990b, p. 9), the critical angle phenomenon is related to the condition $s_3^T = 0$, but, as we shall see below, this is only valid when the lower medium has $p'_{46} = 0$ (e.g., transversely isotropic). The correct interpretation was given by Henneke II (1971), who defined the critical angle θ_C as the angle(s) of incidence beyond which the Umov-Poynting vector of the transmitted wave is parallel to the interface (see also Rokhlin, Bolland and Adler (1986)). From equations (2.113), (6.9) and (6.10), this is equivalent to $\text{Re}(Z^T) = 0$. We keep the same interpretation for viscoelastic media. Actually, the pseudocritical angle does not play any important physical role in the anisotropic case.

The condition $\text{Re}(Z^T)=0$ in equation (6.56) yields the critical angle θ_C , because $\psi_T = \pi/2$. Using equation (6.10), this gives

$$\operatorname{Re}(p_{46}'s_1 + p_{44}'s_3^T) = 0,$$
 (6.76)

or, from (6.39) and (6.41),

$$\operatorname{Re}\left(\operatorname{pv}\sqrt{\rho' p'_{44} - p'^2 s_1^2}\right) = 0. \quad (6.77)$$

Since for a complex number q, it is $[\operatorname{Re}(\sqrt{q})]^2 = [|q| + \operatorname{Re}(q)]/2$, equation (6.77) is equivalent to

$$Im(\rho' p'_{44} - p'^2 s_1^2) = 0. \qquad (6.78)$$

For the particular case when $p'p'_{44} - p'^2s_1^2 = 0$ and using (6.28), the following explicit solution is obtained:

$$\cot \theta_C = \frac{1}{p_{44}} \left(-p_{46} + \text{pv} \sqrt{\frac{\rho p_{44}}{\rho' p'_{44}} p'^2 - p^2} \right). \tag{6.79}$$

There is a solution if the right-hand side of equation (6.79) is real. This occurs only in very particular situations.

In the isotropic case (see (6.25)), a critical angle exists if

$$\cot \theta_C = \sqrt{\frac{\rho}{\rho'} \frac{\mu'}{\mu} - 1} \tag{6.80}$$

is a real quantity. This is verified for μ'/μ real or $Q_H = Q'_H$ and $\rho\mu' > \rho'\mu$. Then, $\mu'/\mu = \text{Re}(\mu')/\text{Re}(\mu)$ and

$$\sin \theta_P = \sqrt{\frac{\rho' \operatorname{Re}(\mu)}{\rho \operatorname{Re}(\mu')}} = \sin \theta_C \tag{6.81}$$

(Borcherdt, 1977). The last equality holds since $p'_{46} = 0$ implies $\text{Re}(s_3^T) = 0$ from equation (6.76).

Figure 6.5 shows the absolute values of the reflection and transmission coefficients versus the incidence angle for the elastic (dotted line) and viscoelastic (solid line) cases, respectively, with $\theta_P = 31.38^\circ$, $\theta_B = 32.34^\circ$ and $\theta_C = 36.44^\circ$. The directions of the slowness and Umov-Poynting vectors, corresponding to the critical angle θ_C , can be appreciated in Figure 6.6, which illustrates the elastic case.



Figure 6.6: Directions of the slowness and Umov-Poynting vectors, corresponding to the critical angle $\theta_C = 36.44^\circ$ for the elastic case. At the critical angle and beyond, the Umov-Poynting vector of the transmitted wave is parallel to the interface. Moreover, the transmitted wave becomes evanescent.

According to Proposition 6, at the critical angle and beyond, the Umov-Poynting vector of the transmitted wave is parallel to the interface and the wave becomes evanescent. A geometrical interpretation is that, in the elastic case, critical angles are associated with tangent planes to the slowness surface that are normal to the interface (see Figure 6.6). Snell's law requires that the end points of all the slowness vectors lie in a common normal line to the interface. We get the critical angle when this line is tangent to the slowness curve of the transmission medium. Beyond the critical angle, there is no intersection between that line and the slowness curve, and the wave becomes evanescent (Henneke II, 1971; Rokhlin, Bolland and Adler, 1986; Helbig 1994, p. 241).

In the lossless case, the Umov-Poynting vector is parallel to the boundary beyond the critical angle. Moreover, since Z^T is purely imaginary, equations (6.39) and (6.70) imply that $\text{Re}(s_3^T) = -c'_{46}s_1/c'_{44}$. Finally, using equation (6.47), we obtain the propagation angle of the transmitted wave, namely,

$$\theta^{T} = -\arctan(c'_{44}/c'_{46}).$$
 (6.82)

6.1 Reflection and transmission of SH waves

This angle takes the value $\theta^T = 119.75^\circ$ ($\psi^T = 90^\circ$) and remains constant for $\theta^I \ge \theta_C$. This phenomenon does not occur in the anelastic case.

As can be seen in Figure 6.5, there is no critical angle in the viscoelastic case and the reflection coefficient is always greater than zero (no Brewster angle). As in the isotropic case (Borcherdt, 1977), critical angles exist under very particular conditions.

Theorem 3: If one of the media is elastic and the other is anelastic, then there are no critical angles.

Proof: Suppose there exists a critical angle; that is, the Umov-Poynting vector of the transmitted wave is parallel to the interface. Assume first that the incidence medium is elastic. Proposition 1 implies that the attenuation of the transmitted wave is normal to the interface. However, since the transmission medium is anelastic, such an inhomogeneous wave – associated with elastic media – cannot propagate, otherwise $\langle D^T \rangle = 0$ (see equation (6.51)).

Conversely, assume a homogeneous plane wave, non-normal incidence and that the transmission medium is elastic. Since the incidence medium is anelastic, Snell's law requires a transmitted inhomogeneous wave of the viscoelastic type ($\alpha \cdot \langle \mathbf{p} \rangle \neq 0$) in the transmission medium. However, this wave cannot propagate in an elastic medium (see equation (6.51)).

A special case: Let us consider that both media are transversely isotropic and that M_2 = $M'_1 = M'_2 = M_1$. This case is similar to the one studied by Krebes (1983b) in isotropic media. Equation (6.78) gives the solution

$$\cot \theta_C = \sqrt{\frac{\rho c_{66}^*}{\rho' c_{44}} - \frac{c_{66}}{c_{44}}}$$
(6.83)

and $s_1 = \sqrt{\rho'/p'_{66}}$, which implies $s_3^T = 0$. The critical angle for this case is $\theta_C = 47.76^\circ$. It can be shown from equations (6.27), (6.32), (6.34), (6.35) and (6.39) that the reflection and transmission coefficients are identical to those for perfect elasticity. However, beyond the critical angle, there is a normal interference flux (see Section 6.1.7) towards the boundary, complemented by a small energy flow away from the boundary in the transmission medium. This means that θ_C is a "discrete critical angle", i.e., the Umov-Poynting vector of the transmitted wave is parallel to the boundary only for the incidence angle θ_C . (In the elastic case this happens for $\theta^I \ge \theta_C$.) Since $s_3^T = 0$ at the critical angle, this occurs when the normal to the interface with abscissa $\operatorname{Re}(s_1)$ is tangent to the slowness curve of the transmitted wave and, simultaneously, the normal to the interface with abscissa $\operatorname{Im}(s_1)$ is tangent to the attenuation curve of the same wave.

6.1.6 Phase velocities and attenuations

The magnitude of the phase velocities can be obtained as the reciprocal of the slownesses. From equations (6.2) and (6.27), the phase velocity of the incident wave is simply

$$v_p^I = \{ [\operatorname{Re}(s_1)]^2 + [\operatorname{Re}(s_3^I)]^2 \}^{-1/2} = [\operatorname{Re}(v_e^{-1})]^{-1}.$$
 (6.84)

The phase velocity of the reflected wave is obtained from equation (6.30) and written as

$$v_p^R = \{ [\operatorname{Re}(s_1)]^2 + [\operatorname{Re}(s_3^R)]^2 \}^{-1/2},$$
 (6.85)

or, using equations (6.10), (6.27) and (6.35), as

$$v_p^R = \left\{ (v_p^I)^{-2} + 4\sin(\theta^I) \operatorname{Re}\left(p_{46} p_{44}^{-1} v_e^{-1}\right) \operatorname{Re}\left(p_{44}^{-1} Z^I\right) \right\}^{-1/2}.$$
(6.86)

When the Umov-Poynting vector of the incident wave is parallel to the interface ($Z^{I} = 0$), or when the upper medium is transversely isotropic ($p_{46} = 0$), v_{p}^{R} equals v_{p}^{I} . In the elastic case, equation (6.86) reduces to

$$v_p^R = v_p^I \{1 + 4\sin(\theta^I)c_{46}c_{44}^{-1}[c_{46}c_{44}^{-1}\sin\theta^I + \cos\theta^I]\}^{-1/2}.$$
 (6.87)

Similarly, the phase velocity of the transmitted wave is obtained from equation (6.31) and written as

$$v_p^T = \{ [\operatorname{Re}(s_1)]^2 + [\operatorname{Re}(s_3^T)]^2 \}^{-1/2}.$$
 (6.88)

The phase velocities of the incident, reflected and transmitted waves, versus the incidence angle, are represented in Figure 6.7, where the dotted line corresponds to the elastic case. The velocity in the elastic case is always higher than the velocity in the viscoelastic case, since the former case is taken at the high-frequency limit.





By virtue of equations (6.3), (6.27) and (6.30), the magnitudes of the attenuation vectors of the incident and reflected waves are given by

$$\alpha^{I} = \omega \sqrt{[\mathrm{Im}(s_{1})]^{2} + [\mathrm{Im}(s_{3}^{I})]^{2}} = -\omega \mathrm{Im}(v_{c}^{-1})$$
(6.89)

and

$$\alpha^R = \omega \sqrt{[\text{Im}(s_1)]^2 + [\text{Im}(s_3^R)]^2}$$

(6.90)

or, using equations (6.10), (6.27) and (6.35),

$$\alpha^{R} = \sqrt{(\alpha^{I})^{2} + 4\omega^{2}\sin(\theta)} \operatorname{Im}\left(p_{46}p_{44}^{-1}v_{c}^{-1}\right) \operatorname{Im}\left(p_{44}^{-1}Z^{I}\right).$$
(6.91)

In the transversely isotropic case $(p_{46} = 0)$, $\alpha^R = \alpha^I$. The magnitude of the attenuation vector of the transmitted wave is obtained from equation (6.31), and written as



$$\alpha^T = \omega \sqrt{[\text{Im}(s_1)]^2 + [\text{Im}(s_3^T)]^2}.$$
 (6.92)

Figure 6.8: Attenuations of the incident, reflected and transmitted waves versus the incidence angle.

The attenuations are represented in Figure 6.8. The high attenuation value of the transmitted wave can be explained as follows. Figure 6.3 indicates that, at approximately the elastic-case critical angle and beyond, the energy angle of the transmitted wave ψ^T is close to $\pi/2$ and that the attenuation vector is almost perpendicular to the interface. In practice, this implies that the transmitted wave resembles an evanescent wave of the elastic type. This effect tends to disappear when the intrinsic quality factors of the lower and/or upper media are lower than the values given in Section 6.1.2.

6.1.7 Energy-flux balance

In order to balance energy flux at an interface between two isotropic single-phase media, it is necessary to consider the interaction energy fluxes when the media are viscoelastic (Borcherdt, 1977; Krebes, 1983b). In the incidence medium, for instance, the interaction energy fluxes arise from the interaction of the stress and velocity fields of the incident and reflected waves. A similar phenomenon takes place at an interface separating two porous media when the fluid viscosity is different from zero. Dutta and Odé (1983) call these fluxes interference fluxes and show that they vanish for zero viscosity.

In a welded interface, the normal component of the average Umov-Poynting $\hat{\mathbf{e}}_3 \cdot \langle \mathbf{p} \rangle$ is continuous across the interface. This is a consequence of the boundary conditions that impose continuity of normal stress σ_{32} , and particle velocity. Then, according to equation (4.111), the balance of power flow at the interface can be expressed as

$$-\frac{1}{2}\operatorname{Re}[(\sigma_{32}^{I} + \sigma_{32}^{R})(v^{I} + v^{R})^{*}] = -\frac{1}{2}\operatorname{Re}(\sigma_{32}^{T}v^{T^{*}}). \quad (6.93)$$

Using equations (6.9) and (6.36), equation (6.93) is of the form

$$\langle p_3^I \rangle + \langle p_3^R \rangle + \langle p_3^{IR} \rangle = \langle p_3^T \rangle,$$
 (6.94)

where

$$\langle p_3^I \rangle = -\frac{1}{2} \operatorname{Re}(\sigma_{32}^I v^{I^*}) = \frac{1}{2} \omega^2 \operatorname{Re}(Z^I) \exp[2\omega \operatorname{Im}(s_1)x]$$
 (6.95)

is the flux of the incident wave,

$$\langle p_3^R \rangle = -\frac{1}{2} \operatorname{Re}(\sigma_{32}^R v^{R^*}) = \frac{1}{2} \omega^2 |R_{SS}|^2 \operatorname{Re}(Z^R) \exp[2\omega \operatorname{Im}(s_1)x]$$
 (6.96)

is the reflected flux,

$$\langle p_3^{IR} \rangle = -\frac{1}{2} \operatorname{Re}(\sigma_{32}^{I} v^{R^*} + \sigma_{32}^{R} v^{I^*}) = \frac{1}{2} \omega^2 (Z^I R_{SS}^* + Z^R R_{SS}) \exp[2\omega \operatorname{Im}(s_1)x]$$

= $\omega^2 \operatorname{Im}(R_{SS}) \operatorname{Im}(Z^I) \exp[2\omega \operatorname{Im}(s_1)x]$ (6.97)

is the interference between the normal fluxes of the incident and reflected waves, and

$$\langle p_3^T \rangle = -\frac{1}{2} \operatorname{Re}(\sigma_{32}^T v^{T^*}) = \frac{1}{2} \omega^2 |T_{SS}|^2 \operatorname{Re}(Z^T) \exp[2\omega \operatorname{Im}(s_1)x]$$
 (6.98)

is the flux of the transmitted wave. In the elastic case, Z^{I} is real and the interference flux vanishes.



Figure 6.9: Normalized fluxes (energy coefficients) versus the incidence angle. The fluxes are normalized with respect to the flux of the incident wave. The elastic case is represented by a dotted line.

The normalized normal fluxes (energy coefficients) versus the incidence angle are shown in Figures 6.9, with the dotted line representing the elastic case. Beyond the

critical angle, the normal component of the Umov-Poynting vector of the transmitted wave vanishes and there is no transmission to the lower medium. The energy travels along the interface and, as stated before, the plane wave is evanescent. In the viscoelastic case, these effects disappear and the fluxes of the reflected and transmitted waves have to balance with a non-zero interference flux. Since the flux of the transmitted wave is always greater than zero, there is transmission for all the incidence angles.

6.1.8 Energy velocities and quality factors

The energy velocity \mathbf{v}_e is the ratio of the average power-flow density $\langle \mathbf{p} \rangle = \text{Re}(\mathbf{p})$ to the mean energy density $\langle T + V \rangle$ (see equation (4.76)). For the incident homogeneous wave, substitution of (6.27) into (6.16) and use of (6.28) gives $\rho^I = \rho v_c^2 / |v_c|^2$. Then, equations (6.14) and (6.17) imply

$$\langle T + V \rangle = \frac{1}{4} \rho \omega^2 \left[\frac{\text{Re}(v_c^2)}{|v_c|^2} + 1 \right] \exp\{2\omega[\text{Im}(s_1)x + \text{Im}(s_3^I)z]\}.$$
 (6.99)

Using the relation $[\operatorname{Re}(v_c)]^2 = [|v_c|^2 + \operatorname{Re}(v_c^2)]/2$, equation (6.99) becomes

$$\langle T+V \rangle = \frac{1}{2} \rho \omega^2 (v_p^I)^{-1} \exp\{2\omega [\operatorname{Im}(s_1)x + \operatorname{Im}(s_3^I)z]\} \operatorname{Re}(v_c),$$
 (6.100)

where v_p^I is the phase velocity (6.84). Finally, combining (6.9) and (6.100), we obtain

$$\mathbf{v}_{e}^{I} = \frac{v_{p}^{I}}{\rho \operatorname{Re}(v_{e})} \operatorname{Re}(X^{I} \hat{\mathbf{e}}_{1} + Z^{I} \hat{\mathbf{e}}_{3}).$$
(6.101)

The energy velocity of the reflected wave is obtained from equations (6.9), (6.14), (6.16)and (6.17), and written as

$$\mathbf{v}_{e}^{R} = \frac{2\text{Re}(X^{R}\hat{\mathbf{e}}_{1} + Z^{R}\hat{\mathbf{e}}_{3})}{\rho + \text{Re}(\varrho^{R})}, \quad (6.102)$$

where $\rho^R = \rho(s_3^R)$. If the upper medium has $p_{46} = 0$ (e.g., transverse isotropy), $X^R = X^I$ and $\rho^R = \rho v_e^2 / |v_e|^2$. After some algebra, it can be shown, using (6.35) and (6.36), that $v_e^R = v_e^I$.

Similarly, the energy velocity of the transmitted wave is

$$\mathbf{v}_{e}^{T} = \frac{2\text{Re}(X^{T}\hat{\mathbf{e}}_{1} + Z^{T}\hat{\mathbf{e}}_{3})}{\rho' + \text{Re}(\varrho^{T})}, \quad (6.103)$$

where $\rho^T = \rho(s_3^T)$.

An alternative expression for the energy velocity is obtained from the fact that, as in the elastic case, the phase velocity is the projection of the energy-velocity vector onto the propagation direction. This relation is demonstrated in Section 4.3.1 (equation (4.78)) for inhomogeneous waves propagating in a general anisotropic viscoelastic medium. For cross-plane shear waves, we have

$$v_e = v_p / \cos(\psi - \theta)$$
. (6.104)

In terms of the tangents defined in Section 6.1.4,

$$v_e = \left[\frac{\sqrt{(1 + \tan^2 \psi)(1 + \tan^2 \theta)}}{(1 + \tan \psi \tan \theta)}\right] v_p. \quad (6.105)$$

The energy velocities of the incident, reflected and transmitted waves, versus the incidence angle, are represented in Figures 6.10, with the dotted line corresponding to the elastic case.



Figure 6.10: Energy velocities of the incident, reflected and transmitted waves, versus the incidence angle. The elastic case is represented by a dotted line.

Comparison of Figures 6.7 and 6.10 indicates that the energy velocity in anisotropic viscoelastic media is greater or equal than the phase velocity – as predicted by equation (6.104).

The quality factor is the ratio of twice the average strain-energy density (6.14) to the dissipated-energy density (6.15). For the incident homogeneous wave it is simply

$$Q'_{H} = \frac{\operatorname{Re}(\varrho')}{\operatorname{Im}(\varrho')} = \frac{\operatorname{Re}(v_{c}^{2})}{\operatorname{Im}(v_{c}^{2})}, \quad (6.106)$$

while for the reflected and transmitted waves,

$$Q^{R} = \frac{\operatorname{Re}(\varrho^{R})}{\operatorname{Im}(\varrho^{R})}$$
(6.107)

and

$$Q^{T} = \frac{\operatorname{Re}(\varrho^{T})}{\operatorname{Im}(\varrho^{T})}, \quad (6.108)$$

respectively. When $p_{46} = 0$ and using (6.35), $\rho^R = \rho v_c^2 / |v_c|^2$, and Q^R equals Q^I .

Let us consider the incident homogeneous wave. From equation (6.28), $\rho v_c^2 = p_{44}$ along the z-axis. Substitution of (6.18) into (6.106) and the use of (6.19) gives equation (6.21). Then, the quality factor along the vertical direction is Q_{01} at the reference frequency f_0 . Similarly, it can be shown that Q_{02} is the quality factor along the horizontal direction.

In Chapter 8, we demonstrate that the equations describing propagation of the TM (transverse magnetic) mode in a conducting anisotropic medium are completely analogous – from the mathematical point of view – to the propagation of viscoelastic cross-plane shear waves in the plane of symmetry of a monoclinic medium. This equivalence identifies the magnetic field with the particle velocity, the electric field with the stress components, and the compliance components p_{IJ}^{-1} with the complex dielectric-permittivity components. Therefore, the present reflection-transmission analysis can be applied to the electromagnetic case with minor modifications.

6.2 Reflection and transmission of qP-qSV waves

A review of the literature pertaining to the reflection-transmission problem in anisotropic elastic media and isotropic viscoelastic media is given in Sections 1.9 and 3.8, respectively. The time-domain equations for propagation in a heterogeneous viscoelastic transversely isotropic medium are given in Chapter 4, Section 4.5.

6.2.1 Propagation characteristics

A general plane-wave solution for the particle-velocity field $\mathbf{v} = (v_1, v_3)$ is

$$\mathbf{v} = i\omega \mathbf{U} \exp [i\omega(t - s_1x - s_3z)], \quad (6.109)$$

where s_1 and s_3 are the components of the complex-slowness vector, and U is a complex vector. The real-valued slowness and attenuation vectors are given by

$$s_R = (\text{Re}(s_1), \text{Re}(s_3))$$
 (6.110)

and

$$\alpha = -\omega(Im(s_1), Im(s_3)),$$
 (6.111)

respectively. The complex-slowness vector is then

$$\mathbf{s} = \mathbf{s}_R - \frac{\mathbf{i}\alpha}{\omega}, \quad s^2 = s_1^2 + s_3^2.$$
 (6.112)

The dispersion relation can be obtained from equation $(1.78)_2$, by using (1.74), $s_1 = sl_1$, $s_3 = sl_3$, and the correspondence principle ((Section 3.6) $(c_{IJ} \rightarrow p_{IJ})$). Hence, we have

$$F(s_1, s_3) = (p_{11}s_1^2 + p_{55}s_3^2 - \rho)(p_{33}s_3^2 + p_{55}s_1^2 - \rho) - (p_{13} + p_{55})^2 s_1^2 s_3^2 = 0,$$
(6.113)

which has two solutions corresponding to the quasi-compressional (qP) and quasi-shear (qS) waves. The form (6.113) holds for inhomogeneous plane waves in viscoelastic media. Let us assume that the positive z-axis points downwards. In order to distinguish between down and up propagating waves, the slowness relation equation (6.113) is solved for s_3 , given the horizontal slowness s_1 . This yields

$$s_3 = \pm \frac{1}{\sqrt{2}} \sqrt{K_1 \mp pv} \sqrt{K_1^2 - 4K_2K_3},$$
 (6.114)

where

$$\begin{split} K_1 &= \rho \left(\frac{1}{p_{55}} + \frac{1}{p_{33}} \right) + \frac{1}{p_{55}} \left[\frac{p_{13}}{p_{33}} (p_{13} + 2p_{55}) - p_{11} \right] s_1^2, \\ K_2 &= \frac{1}{p_{33}} (p_{11} s_1^2 - \rho), \quad K_3 = s_1^2 - \frac{\rho}{p_{55}}. \end{split}$$

The signs in s_3 correspond to

(+,-) downward propagating qP wave
 (+,+) downward propagating qS wave
 (-,-) upward propagating qP wave
 (-,+) upward propagating qS wave.

The plane-wave eigenvectors (polarizations) belonging to a particular eigenvalue can be obtained from the qP-qS Kelvin-Christoffel equation by using equation (1.81) and the correspondence principle. We obtain

$$\mathbf{U} = U_0 \begin{pmatrix} \beta \\ \xi \end{pmatrix}, \quad (6.115)$$

where U_0 is the plane-wave amplitude and

$$\beta = pv \sqrt{\frac{p_{55}s_1^2 + p_{33}s_3^2 - \rho}{p_{11}s_1^2 + p_{33}s_3^2 + p_{55}(s_1^2 + s_3^2) - 2\rho}}$$
(6.116)

and

$$\xi = \pm pv \sqrt{\frac{p_{11}s_1^2 + p_{55}s_3^2 - \rho}{p_{11}s_1^2 + p_{33}s_3^2 + p_{55}(s_1^2 + s_3^2) - 2\rho}}.$$
(6.117)

In general, the + and - signs correspond to the qP and qS waves, respectively. However one must choose the signs such that ξ varies smoothly with the propagation angle. In the elastic case, the qP eigenvectors are orthogonal to the qS eigenvectors only when the respective slownesses are parallel. In the viscoelastic case, this property is not satisfied. From equations (6.109), (6.116) and (6.117), and using (6.110) and (6.111), the particlevelocity field can be written as

$$\mathbf{v} = \mathrm{i}\omega U_0 \begin{pmatrix} \beta \\ \xi \end{pmatrix} \exp\{\omega[\mathrm{Im}(s_1)x + \mathrm{Im}(s_3)z]\} \exp\{\mathrm{i}\omega[t - \mathrm{Re}(s_1)x - \mathrm{Re}(s_3)z]\}.$$
 (6.118)

The mean flux or time-averaged Umov-Poynting vector $\langle \mathbf{p} \rangle$ is the real part of the corresponding complex vector (see equation (4.55)),

$$\mathbf{p} = -\frac{1}{2} [(\sigma_{11}v_1^* + \sigma_{13}v_3^*)\hat{\mathbf{e}}_1 + (\sigma_{13}v_1^* + \sigma_{33}v_3^*)\hat{\mathbf{e}}_3].$$
(6.119)

Substituting the plane wave (6.118) and the stress-strain relation (4.160) into equation (6.119), we obtain

$$\mathbf{p} = \frac{1}{2}\omega^2 |U_0|^2 \left(\begin{array}{c} \beta^* X + \xi^* W\\ \beta^* W + \xi^* Z \end{array} \right) \exp\{2\omega [\operatorname{Im}(s_1)x + \operatorname{Im}(s_3)z]\},$$
(6.120)

where

$$W = p_{55}(\xi s_1 + \beta s_3),$$
 (6.121)

$$X = \beta p_{11}s_1 + \xi p_{13}s_3, \qquad (6.122)$$

$$Z = \beta p_{13}s_1 + \xi p_{33}s_3 \qquad (6.123)$$

and the strain-displacement relations (1.2) have been used.

6.2.2 Properties of the homogeneous wave

For homogeneous waves, the directions of propagation and attenuation coincide and

$$s_1 = \sin \theta / v_c(\theta), \quad s_3 = \cos \theta / v_c(\theta), \quad (6.124)$$

where θ is the propagation angle, measured with respect to the z-axis, and $v_c = 1/s$ is the complex velocity that can be obtained from the slowness relation (6.113). Hence, we have

$$\rho v_c^2 = \frac{1}{2} (p_{55} + p_{11} \sin^2 \theta + p_{33} \cos^2 \theta \pm C), \qquad (6.125)$$

with

$$C = \sqrt{[(p_{33} - p_{55})\cos^2\theta - (p_{11} - p_{55})\sin^2\theta]^2 + (p_{13} + p_{55})^2\sin^22\theta}.$$
 (6.126)

The + sign corresponds to the qP wave, and the - sign to the qS wave.

Combining (6.110), (6.111) and (6.124) yields

$$\mathbf{s}_R = \operatorname{Re}\left(\frac{1}{v_c}\right)(\sin\theta,\cos\theta),$$
(6.127)

and

$$\boldsymbol{\alpha} = -\omega \operatorname{Im}\left(\frac{1}{v_c}\right) (\sin \theta, \cos \theta). \tag{6.128}$$

The quality factor is

$$Q = \frac{\text{Re}(v_c^2)}{\text{Im}(v_c^2)}$$
(6.129)

(see equation (4.92)). For instance, for model 3 (for the 2-D case, see Sections 4.1.3 and 4.5.4, equation (4.161)) we point out the following properties. At the symmetry axis ($\theta = 0$), for qP waves, $v_c^2 = \rho p_{33}$, and at the isotropy plane ($\theta = \pi/2$), $v_c^2 = \rho p_{11}$. Then, the relation between Q factors is

$$\frac{Q(\text{symmetry axis})}{Q(\text{isotropy plane})} = \frac{c_{33} - a}{c_{11} - a}, \quad a = \tilde{\mathcal{E}} - \tilde{\mathcal{K}} \operatorname{Re}(M_1) - c_{55} \operatorname{Re}(M_2). \quad (6.130)$$

We can verify that a > 0, $a < c_{11}$ and $a < c_{33}$, for most realistic materials – in the elastic case, a = 0 ($M_1 = M_2 = 0$, see equation (4.156)). This implies that, whatever the ratio c_{33}/c_{11} , the ratio between Q factors is farther from unity than the elastic-velocity ratio $\sqrt{c_{33}/c_{11}}$. It follows that attenuation is a better indicator of anisotropy than elastic velocity. Similarly, it can be shown that the ratio between the viscoelastic phase velocities $\operatorname{Re}(1/\sqrt{p_{11}})/\operatorname{Re}(1/\sqrt{p_{33}})$ is closer to one than the Q ratio.

Another important property is that, when $c_{11} > c_{33}$ (e.g., finely layered media, see Section 1.5), the qP wave attenuates more along the symmetry axis than in the plane of isotropy. Note that we do not use an additional relaxation function to model Q anisotropy of the qP wave. It is the structure of the medium – described by the stiffnesses – that dictates the Q ratio between different propagation directions.

On the other hand, the quality factor of the shear wave at the symmetry axis is equal to the quality factor in the plane of isotropy, since $v_c^2 = \rho p_{55}$ in both cases. This is so, since any kind of symmetry possessed by the attenuation should follow the symmetry of the crystallographic form of the material (Neumann's principle, see Nye, 1987, p. 20). A qS wave anisotropy factor can be defined as the ratio of the vertical phase velocity to the phase velocity at an angle of 45° to the axis of symmetry. Again, it can be shown that, for most realistic materials, this factor is closer to one than the ratio between the respective quality factors.

6.2.3 Reflection and transmission coefficients

The upper layer is denoted by the subscript 1 and the lower layer by the subscript 2. For clarity, the material properties of the lower medium are primed and the symbols P and S indicate the qP and qS waves, respectively. Moreover, the subscripts I, R and T denote the incident, reflected and transmitted waves. Using symmetry properties to define the polarization of the reflected waves, the particle velocities for a qP wave incident from above the interface are given by

$$v_1 = v_{P_f} + v_{P_R} + v_{S_R}$$
, (6.131)

$$v_2 = v_{P_T} + v_{S_T}$$
, (6.132)

where

$$\mathbf{v}_{P_{I}} = i\omega(\beta_{P_{1}}, \xi_{P_{1}}) \exp [i\omega(t - s_{1}x - s_{3P_{1}}z)],$$
 (6.133)

$$\mathbf{v}_{P_R} = i\omega R_{PP}(\beta_{P_1}, -\xi_{P_1}) \exp[i\omega(t - s_1x + s_{3P_1}z)],$$
 (6.134)

$$\mathbf{v}_{S_R} = i\omega R_{PS}(\beta_{S_1}, -\xi_{S_1}) \exp[i\omega(t - s_1x + s_{3S_1}z)],$$
 (6.135)

$$\mathbf{v}_{P_T} = i\omega T_{PP}(\beta_{P_2}, \xi_{P_2}) \exp[i\omega(t - s_1x - s_{3P_2}z)],$$
 (6.136)

and

$$\mathbf{v}_{S_T} = i\omega T_{PS}(\beta_{S_2}, \xi_{S_2}) \exp[i\omega(t - s_1x - s_{3S_2}z)].$$
 (6.137)

The boundary conditions (continuity of the particle velocity and normal stress components) give Snell's law, i.e., the continuity of the horizontal complex slowness s_1 . The vertical slownesses s_{3P} and s_{3S} , as well as β_P , β_S , ξ_P and ξ_S , follow respectively the (+, -) and (+, +) sign sets given in equation (6.114). The choice $U_0 = 1$ implies no loss of generality. The boundary conditions require continuity of

$$v_1$$
, v_3 , σ_{33} , and σ_{13} . (6.138)

The stresses are obtained by the substitution of equations (6.131) and (6.132) into the stress-strain relation (4.160). The boundary conditions generate the following matrix equation for the reflection and transmission coefficients:

$$\begin{pmatrix} \beta_{P_1} & \beta_{S_1} & -\beta_{P_2} & -\beta_{S_2} \\ \xi_{P_1} & \xi_{S_1} & \xi_{P_2} & \xi_{S_2} \\ Z_{P_1} & Z_{S_1} & -Z_{P_2} & -Z_{S_2} \\ W_{P_1} & W_{S_1} & W_{P_2} & W_{S_2} \end{pmatrix} \cdot \begin{pmatrix} R_{PP} \\ R_{PS} \\ T_{PP} \\ T_{PS} \end{pmatrix} = \begin{pmatrix} -\beta_{P_1} \\ \xi_{P_1} \\ -Z_{P_1} \\ W_{P_1} \end{pmatrix}, \quad (6.139)$$

where W and Z are given by equations (6.121) and (6.123), respectively.

The steps to compute the reflection and transmission coefficients are the following:

- The horizontal slowness s₁ is the independent parameter. It is the same for all the waves (Snell's law for viscoelastic media). For an incident homogeneous wave, the independent variable becomes the incidence angle θ, and s₁ is obtained from equation (6.124).
- Compute s_{3P1}, s_{3S1}, s_{3P2} and s_{3S2} from equation (6.114), where the first sign is positive. For an incident homogeneous wave, s_{3P1} can be calculated either from equation (6.114) or from equation (6.124).
- Compute β_{P1}, β_{S1}, β_{P2}, β_{S2}, ξ_{P1}, ξ_{S1}, ξ_{P2} and ξ_{S2} from equations (6.116) and (6.117).
- Compute W_{P1}, W_{S1}, W_{P2} and W_{S2} from equation (6.121), and Z_{P1}, Z_{S1}, Z_{P2} and Z_{S2} from equation (6.123).
- Compute the reflection and transmission coefficients by numerically solving equation (6.139).

The reflection and transmission coefficients R_{SP} , R_{SS} , T_{SP} and T_{SS} for an incident qS wave have the same 4 × 4 scattering matrix as the qP incident wave, but the vector in the right-hand side of (6.139) is

$$(-\beta_{S_1}, \xi_{S_1}, -Z_{S_1}, W_{S_1})^\top$$
. (6.140)

6.2.4 Propagation, attenuation and energy directions

Figure 6.1 illustrates the convention used to define the propagation, attenuation and energy angles. The propagation direction is perpendicular to the plane-wave front. Given the components of the complex-slowness vector, the propagation and attenuation angles for all the waves can be obtained and expressed as

$$\tan \theta = \frac{\operatorname{Re}(s_1)}{\operatorname{Re}(s_3)}, \quad \operatorname{and} \quad \tan \delta = \frac{\operatorname{Im}(s_1)}{\operatorname{Im}(s_3)}.$$
(6.141)

By hypothesis (see equation (6.124)), $\delta_{P_I} = \theta_{P_I}$, and by symmetry, $\theta_{P_R} = -\theta_{P_I}$ and $\delta_{P_R} = \theta_{P_R}$. Hence, the reflected qP wave is homogeneous. The complex vertical slowness component of the reflected qS wave is $-s_{3S_1}$, following the (-, +) sign in equation (6.114). Then, the propagation and attenuation angles θ_{S_R} and δ_{S_R} are obtained from (6.141) with the substitution $s_3 = -s_{3S_1}$. In general $\theta_{S_R} \neq \delta_{S_R}$ and the wave is inhomogeneous. Analogously, the angles of the transmitted qP wave $(\theta_{P_T}$ and $\delta_{P_T})$ and the transmitted qS wave $(\theta_{S_T} \text{ and } \delta_{S_T})$ are given by (6.141) when $s_3 = s_{3P_2}$ and $s_3 = s_{3S_2}$, respectively. The transmitted waves are, in general, inhomogeneous.

The expressions of the time-averaged Umov-Poynting vectors of the reflected and transmitted waves are given by equation (6.120). Then, the angles of the energy vectors of the incident, reflected and transmitted waves are obtained from

$$\tan \psi = \frac{\operatorname{Re}(\beta^* X + \xi^* W)}{\operatorname{Re}(\beta^* W + \xi^* Z)}. \quad (6.142)$$

By symmetry, we have $\psi_{P_R} = -\psi_{P_I}$.

6.2.5 Phase velocities and attenuations

The magnitude of the phase velocities can be obtained as the reciprocal of the slownesses. From equation (6.110), the phase velocity of the incident and reflected waves is simply

$$v_p = \{ [\operatorname{Re}(s_1)]^2 + [\operatorname{Re}(s_3)]^2 \}^{-1/2}.$$
 (6.143)

Since the incident wave is homogeneous, the use of equation (6.124) yields

$$v_{p_{P_l}} = [\operatorname{Re}(v_{c1}^{-1})]^{-1},$$
 (6.144)

where v_{c1} is the complex velocity for homogeneous waves in the incidence medium (equation (6.125)). By symmetry (see also Section 3.5), the phase velocity of the reflected qP wave $v_{p_{P_B}}$ equals $v_{p_{P_r}}$.

The velocities $v_{p_{S_R}}$, $v_{p_{P_T}}$ and $v_{p_{S_T}}$ are obtained from (6.143) by replacing s_3 by $-s_{3S_1}$, s_{3P_2} and s_{3S_2} , respectively.

The magnitude of the attenuation vectors is given by

$$\alpha = \omega \{ [Im(s_1)]^2 + [Im(s_3)]^2 \}^{-1/2}. \quad (6.145)$$

The incident and qP reflected waves have the same value:

$$\alpha_{P_{\ell}} = -\omega \operatorname{Im}(v_{el}^{-1}),$$
 (6.146)

while the attenuations α_{S_R} , α_{P_T} and α_{S_T} are obtained from (6.145) by replacing s_3 by - s_{3S_1} , s_{sP_2} and s_{3S_2} , respectively.

6.2.6 Energy-flow balance

We have seen in Section 6.1.7 that to balance energy flux at an interface between two isotropic single-phase media, it is necessary to consider the interaction energy fluxes when the media are viscoelastic. In the incidence medium, for instance, these fluxes arise from the interaction of the stress and particle-velocity fields of the incident and reflected waves.

6.2 Reflection and transmission of qP-qSV waves 211

In a welded interface, the normal component of the average Umov-Poynting $\hat{\mathbf{e}}_3 \cdot \langle \mathbf{p} \rangle$ is continuous across the interface. This is a consequence of the boundary conditions that impose continuity of normal stresses and particle velocities. Then, using equation (6.119), the balance of power flow implies the continuity of

$$-\frac{1}{2}\operatorname{Re}(\sigma_{13}v_1^* + \sigma_{33}v_3^*), \qquad (6.147)$$

where each component is the sum of the components of the respective waves, e.g., $v_1 = v_{1P_I} + v_{1P_R} + v_{1S_R}$ in the incidence medium and $\sigma_{33} = \sigma_{33P_T} + \sigma_{33S_T}$ in the transmission medium. Denoting by F the vertical component of the energy flux (equation (6.147)), we obtain

$$F_{P_I} + F_{P_R} + F_{S_R} + F_{P_I P_R} + F_{P_I S_R} + F_{P_R S_R} = F_{P_T} + F_{S_T} + F_{P_T S_T},$$
 (6.148)

where

$$\begin{aligned} -2F_{P_{l}} &= \operatorname{Re}(\sigma_{13P_{l}}v_{1P_{l}}^{*} + \sigma_{33P_{l}}v_{3P_{l}}^{*}) \\ -2F_{P_{R}} &= \operatorname{Re}(\sigma_{13P_{R}}v_{1P_{R}}^{*} + \sigma_{33P_{R}}v_{3P_{R}}^{*}) \\ -2F_{S_{R}} &= \operatorname{Re}(\sigma_{13P_{R}}v_{1S_{R}}^{*} + \sigma_{33S_{R}}v_{3S_{R}}^{*}) \\ -2F_{P_{l}P_{R}} &= \operatorname{Re}(\sigma_{13P_{l}}v_{1P_{R}}^{*} + \sigma_{13P_{R}}v_{1P_{l}}^{*} + \sigma_{33P_{l}}v_{3S_{R}}^{*} + \sigma_{33P_{R}}v_{3P_{l}}^{*}) \\ -2F_{P_{l}S_{R}} &= \operatorname{Re}(\sigma_{13P_{l}}v_{1S_{R}}^{*} + \sigma_{13S_{R}}v_{1P_{l}}^{*} + \sigma_{33P_{l}}v_{3S_{R}}^{*} + \sigma_{33S_{R}}v_{3P_{l}}^{*}) \\ -2F_{P_{R}S_{R}} &= \operatorname{Re}(\sigma_{13P_{R}}v_{1S_{R}}^{*} + \sigma_{13S_{R}}v_{1P_{R}}^{*} + \sigma_{33P_{R}}v_{3S_{R}}^{*} + \sigma_{33S_{R}}v_{3P_{R}}^{*}) \\ -2F_{P_{R}}S_{R} &= \operatorname{Re}(\sigma_{13P_{R}}v_{1S_{R}}^{*} + \sigma_{13S_{R}}v_{1P_{R}}^{*} + \sigma_{33P_{R}}v_{3S_{R}}^{*} + \sigma_{33S_{R}}v_{3P_{R}}^{*}) \\ -2F_{P_{T}} &= \operatorname{Re}(\sigma_{13P_{T}}v_{1P_{T}}^{*} + \sigma_{33P_{T}}v_{3P_{T}}^{*}) \\ -2F_{S_{T}} &= \operatorname{Re}(\sigma_{13S_{T}}v_{1S_{T}}^{*} + \sigma_{33S_{T}}v_{3S_{T}}^{*}) \\ -2F_{P_{R}S_{R}} &= \operatorname{Re}(\sigma_{13P_{T}}v_{1S_{T}}^{*} + \sigma_{3S}v_{T}v_{3S_{T}}^{*}) \\ -2F_{P_{R}S_{R}} &= \operatorname{Re}(\sigma_{13P_{T}}v_{1S_{T}}^{*} + \sigma_{3S}v_{T}v_{S_{T}}^{*}) \\ -2F_{P_{R}S_{R}} &= \operatorname{Re}(\sigma_{13P_{T}}v_{1S_{T}}^{*} + \sigma_{1S}v_{T}v_{1P_{T}}^{*} + \sigma_{3S}v_{T}v_{S_{T}}^{*}) \\ -2F_{P_{R}S_{R}} &= \operatorname{Re}(\sigma_{13P_{T}}v_{1S_{T}}^{*} + \sigma_{1S}v_{T}v_{T}^{*}) \\ -2F_{P_{R}S_{R}} &= \operatorname{Re}(\sigma_{13P_{T}}v_{1S_{T}}^{*} + \sigma_{1S}v_{T}v_{T}^{*}) \\ -2F_{P_{R}S_{R}} &= \operatorname{Re}(\sigma_{1S}v_{T}v_{1S_{T}}^{*} + \sigma_{1S}v_{T}v_{T}^{*}) \\ -2F_{P_{R}S_{R}} &= \operatorname{Re}(\sigma_{1S}v_{T}v_{1S_{T}}^{*} + \sigma_{1S}v_{T}v_{T}^{*} + \sigma_{1S}v_{T}v_{T}^{*} + \sigma_{1S}v_{T}v_{T}^{*} + \sigma_{1S}v_{T}v_{T}^{*} + \sigma_{1S}v_{T}v_{T}^{*} + \sigma_{$$

For instance, F_{P_I} is the energy flux of the incident qP wave and $F_{P_IP_R}$ is the interference flux between the incident and reflected qP waves. In the elastic limit, it can be shown that the interference fluxes vanish. Further algebra implies that the fluxes given in the preceding equations are proportional to the real parts of

$$F_{P_{I}} \propto \beta_{P_{I}}^{*} W_{P_{I}} + \xi_{P_{I}}^{*} Z_{P_{I}}$$

$$F_{P_{R}} \propto -(\beta_{P_{I}}^{*} W_{P_{I}} + \xi_{P_{I}}^{*} Z_{P_{I}})|R_{PP}|^{2}$$

$$F_{S_{R}} \propto -(\beta_{S_{I}}^{*} W_{S_{I}} + \xi_{S_{I}}^{*} Z_{S_{I}})|R_{PS}|^{2}$$

$$F_{P_{I}P_{R}} \propto -2i(\beta_{P_{I}}^{*} W_{P_{I}} - \xi_{P_{I}}^{*} Z_{P_{I}})\text{Im}(R_{PP})$$

$$F_{P_{I}S_{R}} \propto (\beta_{S_{I}}^{*} W_{P_{I}} - \xi_{S_{I}}^{*} Z_{P_{I}})R_{PS}^{*} - (\beta_{P_{I}}^{*} W_{S_{I}} - \xi_{P_{I}}^{*} Z_{S_{I}})R_{PS}$$

$$F_{P_{R}S_{R}} \propto -[(\beta_{S_{I}}^{*} W_{P_{I}} + \xi_{S_{I}}^{*} Z_{P_{I}})R_{PP}R_{PS}^{*} + (\beta_{P_{I}}^{*} W_{S_{I}} + \xi_{P_{I}}^{*} Z_{S_{I}})R_{PP}^{*}R_{PS}]$$

$$F_{P_{T}} \propto (\beta_{P_{J}}^{*} W_{P_{J}} + \xi_{P_{J}}^{*} Z_{P_{J}})|T_{PP}|^{2}$$

$$F_{S_{T}} \propto (\beta_{S_{2}}^{*} W_{S_{J}} + \xi_{S_{2}}^{*} Z_{S_{J}})|T_{PS}|^{2}$$

$$F_{P_{T}S_{T}} \propto (\beta_{S_{S}}^{*} W_{P_{J}} + \xi_{S_{S}}^{*} Z_{P_{J}})T_{PP}T_{PS}^{*} + (\beta_{P_{S}}^{*} W_{S_{S}} + \xi_{P_{S}}^{*} Z_{S_{J}})T_{PP}^{*}T_{PS},$$
(6.150)

where the proportionality factor is $\frac{1}{2}\omega^2$.

We define the energy reflection and transmission coefficients as

$$ER_{\rm PP} = \sqrt{\frac{F_{P_R}}{F_{P_l}}}, \quad ER_{\rm PS} = \sqrt{\frac{F_{S_R}}{F_{P_l}}},$$

Chapter 6. Reflection and transmission of plane waves

$$ET_{\rm PP} = \sqrt{\frac{F_{P_T}}{F_{P_I}}}, \quad ET_{\rm PS} = \sqrt{\frac{F_{S_T}}{F_{P_I}}},$$
 (6.151)

and the interference coefficients as

$$I_{P_{I}P_{R}} = \frac{F_{P_{I}P_{R}}}{F_{P_{I}}}, \quad I_{P_{I}S_{R}} = \frac{F_{P_{I}S_{R}}}{F_{P_{I}}}, \quad I_{P_{R}S_{R}} = \frac{F_{P_{R}S_{R}}}{F_{P_{I}}}, \quad I_{P_{T}S_{T}} = \frac{F_{P_{T}S_{T}}}{F_{P_{I}}}, \quad (6.152)$$

to obtain the following energy-balance equation:

$$1 + ER_{PP}^2 + ER_{PS}^2 + I_{P_IP_R} + I_{P_IS_R} + I_{P_RS_R} = ET_{PP}^2 + ET_{PS}^2 + I_{P_TS_T}.$$
(6.153)

We have chosen the square root of the energy ratio (Gutenberg, 1944) since it is more nearly related to the response, in terms of particle velocities and displacements.

6.2.7 Umov-Poynting theorem, energy velocity and quality factor

The energy-balance equation or Umov-Poynting theorem for the propagation of time harmonic fields in anisotropic viscoelastic media is given in Section 4.3.1, equation (4.82). For inhomogeneous viscoelastic plane waves, it is

$$-2\alpha \cdot \mathbf{p} = 2i\omega[\langle V \rangle - \langle T \rangle] - \omega \langle D \rangle,$$
 (6.154)

where $\langle V \rangle$ and $\langle T \rangle$ are the time-averaged strain- and kinetic-energy densities, respectively, and $\langle D \rangle = \langle \dot{D} \rangle / \omega$ is the time-averaged dissipated-energy density.

The energy velocity \mathbf{v}_e is defined as the ratio of the average power-flow density $\langle \mathbf{p} \rangle$ to the mean energy density $\langle E \rangle = \langle V + T \rangle$ (equation (4.76)). Fortunately, it is not necessary to calculate the strain and kinetic energies explicitly, since, using equation (4.73) and $\omega \mathbf{s}_R = \boldsymbol{\kappa}$,

$$\langle E \rangle = \mathbf{s}_R \cdot \langle \mathbf{p} \rangle.$$
 (6.155)

Then, the energy velocity can be calculated as

$$\mathbf{v}_{e} = \frac{\langle \mathbf{p} \rangle}{\mathbf{s}_{R} \cdot \langle \mathbf{p} \rangle}.$$
 (6.156)

Using equations (6.110) and (6.120), we find that the energy velocity is

$$\mathbf{v}_{e} = \frac{\operatorname{Re}(\beta^{*}X + \xi^{*}W)\dot{\mathbf{e}}_{1} + \operatorname{Re}(\beta^{*}W + \xi^{*}Z)\dot{\mathbf{e}}_{3}}{\operatorname{Re}(s_{1})\operatorname{Re}(\beta^{*}X + \xi^{*}W) + \operatorname{Re}(s_{3})\operatorname{Re}(\beta^{*}W + \xi^{*}Z)},$$
(6.157)

which, by (6.142) becomes

$$\mathbf{v}_e = [\operatorname{Re}(s_1 + s_3 \cot \psi)]^{-1} \hat{\mathbf{e}}_1 + [\operatorname{Re}(s_1 \tan \psi + s_3)]^{-1} \hat{\mathbf{e}}_3.$$
 (6.158)

An alternative expression for the energy velocity is obtained from the fact that, as in the elastic case, the phase velocity is the projection of the energy velocity onto the propagation direction (equation (4.78)). Thus, we have

$$v_e = v_p / \cos(\psi - \theta).$$
 (6.159)

In terms of the tangents defined in equations (6.141) and (6.142), the magnitude of the energy velocity is

$$v_e = \left[\frac{\sqrt{(1 + \tan^2 \psi)(1 + \tan^2 \theta)}}{(1 + \tan \psi \tan \theta)}\right] v_p.$$
 (6.160)

The quality factor, as defined in equation (4.84), is twice the ratio of the average strainenergy density and the average dissipated-energy density, and is written as

$$Q = \frac{2\langle V \rangle}{\langle D \rangle}.$$
 (6.161)

From equation (4.92), the quality factor of the incident homogeneous wave is simply

$$Q_{P_{f}} = \frac{\text{Re}(v_{c1}^{2})}{\text{Im}(v_{c1}^{2})}.$$
 (6.162)

For the reflected and transmitted waves, we make use of the following fundamental relations derived in Section 4.3.1 (equations (4.83) and (4.71), respectively):

$$\langle D \rangle = \frac{2}{\omega} \boldsymbol{\alpha} \cdot \langle \mathbf{p} \rangle$$
 (6.163)

and

$$\langle V \rangle = \frac{1}{2} \operatorname{Re}(\mathbf{s}^* \cdot \mathbf{p}),$$
 (6.164)

where we have used $\omega s_R = \kappa$ and $\langle D \rangle = \langle D \rangle / \omega$. Substitution of these relations into equation (6.161), and the use of equation (6.110) yields

$$Q = -\frac{\text{Re}(\mathbf{s}^* \cdot \mathbf{p})}{2 \text{ Im}(\mathbf{s}) \cdot \langle \mathbf{p} \rangle}, \quad (6.165)$$

or, using equation (6.120),

$$Q = -\frac{\text{Re}[(\beta^* X + \xi^* W)s_1^* + (\beta^* W + \xi^* Z)s_3^*)]}{2[\text{Re}(\beta^* X + \xi^* W)\text{Im}(s_1) + \text{Re}(\beta^* W + \xi^* Z)\text{Im}(s_3)]}.$$
(6.166)

Thus, we have an expression for the quality factor in terms of the complex slowness and Umov-Poynting vector, which, unlike equation (6.162), holds for inhomogeneous plane waves.

6.2.8 Reflection of seismic waves

We consider the reflection and transmission of seismic waves and compare the results with the elastic case, i.e., the case where both media are elastic. To begin, we briefly consider the following two special cases and the implications of the theory. Firstly, if the incidence medium is elastic and the transmission medium anelastic, the theory imposes that the attenuation vectors of the transmitted waves are perpendicular to the interface. Secondly, if the incidence medium is anelastic, the incident wave is homogeneous, and the transmission medium is elastic, then the transmitted waves are inhomogeneous of the elastic type, i.e., the angle between the Umov-Poynting vector and the attenuation vector is $\pi/2$ (Theorem 2 of Section 6.1.4 considers the cross-plane wave case). The interpretation for the isotropic case is given by Krebes and Slawinski (1991).

The elastic or unrelaxed stiffnesses of the incidence medium are given by

$$c_{11} = \rho c_P^2(\pi/2), \quad c_{33} = \rho c_P^2(0), \quad c_{55} = \rho c_S^2, \quad c_{13} = 3.906 \text{ GPa},$$

where

$$c_P(\pi/2) = 2.79 \text{ km/s}, \quad c_P(0) = 2.24 \text{ km/s}, \quad c_S = 1.01 \text{ km/s}, \quad \rho = 2700 \text{ kg/m}^3.$$

It is assumed that the medium has two relaxation peaks centered at $f_0 = 12.625$ Hz ($\tau_0 = 1/2\pi f_0$), with minimum quality factors of $Q_{01} = 20$ and $Q_{02} = 15$, corresponding to dilatation and shear deformations, respectively.

On the other hand, the unrelaxed properties of the transmission medium are

$$c'_{11} = \rho' c'_P{}^2(\pi/2), \quad c'_{33} = \rho' c'_P{}^2(0), \quad c'_{55} = \rho' c'_S{}^2, \quad c'_{13} = 28.72 \text{ GPa},$$

where

$$c'_P(\pi/2) = 4.6 \text{ km/s}, \quad c'_P(0) = 4.1 \text{ km/s}, \quad c'_S = 2.4 \text{ km/s}, \quad \rho' = 3200 \text{ kg/m}^3.$$

As before, there are two relaxation peaks centered at the same frequency, with $Q_{01} = 60$ and $Q_{02} = 35$.



Figure 6.11: Reflected (qP_R and qS_R) and transmitted (qP_T and qS_T) plane waves for an incident qP wave with $\theta_{P_T} = 25^{\circ}$. The slowness curves for homogeneous waves of the upper and lower medium are represented, with the inner curves corresponding to the quasi-compressional waves. The lines coincide with the propagation direction and the convention for the attenuation and energy vectors is that indicated in Figure 6.1.

The slowness curves for homogeneous waves are represented in Figure 6.11, where the inner curve corresponds to the qP wave. The figure also shows the attributes of the

incident, reflected and transmitted waves for an incidence angle $\theta_{P_T} = 25^{\circ}$. In the anelastic case, the Umov-Poynting vectors (empty arrows) of the incident and reflected qP waves are almost perpendicular to the slowness surface. The perpendicularity property is only verified in the elastic case. The transmitted waves show a high degree of inhomogeneity – i.e., the propagation and attenuation vectors do not have the same direction. This is true, in particular, for the qP wave, whose attenuation vector is almost perpendicular to the direction of the energy vector.

Figure 6.12 represents the absolute value of the amplitude coefficients versus the incidence propagation angle for the elastic (a) and viscoelastic (b) cases. If the two media are elastic, there is a critical angle at approximately 27°, which occurs when the Umov-Poynting vector of the transmitted qP wave becomes parallel to the interface. If the lower medium is anelastic or both media are anelastic, the energy vector of the transmitted qP wave points downwards for all the incidence propagation angles. Thus, there is no critical angle. This can be seen in Figure 6.13, where the absolute values of the energy coefficients are displayed as a function of θ_{P_1} (a) and ψ_{P_1} (b). Since ET_{PP} is always strictly greater than zero, the P_T Umov-Poynting vector is never parallel to the interface.

The propagation, energy and attenuation angles, as a function of the incidence angle, are represented in Figure 6.14. By symmetry, the propagation and energy angles of the reflected P_R wave are equal to θ_{P_l} and ψ_{P_l} , respectively. For viscoelastic plane waves traveling in an anisotropic medium, $|\theta - \delta|$ may exceed 90°. However, the difference $|\psi - \delta|$ must be less than 90°. Indeed, since the energy loss is always positive, equation (6.163) implies that the magnitude of the angle between α and $\langle \mathbf{p} \rangle$ is always strictly less than $\pi/2$. This property is verified in Figure 6.14. Moreover, this figure shows that, at approximately the elastic critical angle and beyond, the P_T energy angle is close to $\pi/2$ and that the attenuation vector is almost perpendicular to the interface. This indicates that, practically, the transmitted qP wave behaves as an evanescent wave of the elastic type beyond the (elastic) critical angle.

Figure 6.15 displays the phase shifts versus incidence propagation angle, indicating that there are substantial differences between the elastic (a) and the anelastic (b) cases. The phase velocities are represented in Figure 6.16. They depend on the propagation direction, mostly because the media are anisotropic, but, to a lesser extent, also because of their viscoelastic inhomogeneous character. Despite the fact that there is no critical angle, the phase velocity of the transmitted qP wave shows a similar behavior – in qualitative terms – to the elastic phase velocity. Beyond the elastic critical angle, the velocity is mainly governed by the value of the horizontal slowness, and finally approaches the phase velocity of the incidence wave. The attenuation curves (see Figure 6.17) show that dissipation of the S_R and P_T waves is very anisotropic. In particular the P_T attenuation is very high after the elastic critical angle, due to the evanescent character of the wave.

Figure 6.18 shows the energy velocity of the different waves. The difference between energy and phase velocities is due solely to the anisotropy, since they coincide in isotropic media. The quality factors are represented in Figure 6.19. Below the critical angle, the higher quality factor is that of the P_T wave, in agreement with its attenuation curve displayed in Figure 6.17. However, beyond that angle, the quality factor seems to contradict the attenuation curve of the other waves: the very strong attenuation is not reflected in the quality factor. This apparent paradox means that the usual relation $\alpha \approx \omega s_R/2Q$ (equation (4.94)) is not valid for evanescent-type waves traveling closer to interfaces, even



Figure 6.12: Absolute values of the reflection and transmission amplitude coefficients versus incidence propagation angle corresponding to the elastic (a) and viscoelastic (b) cases.



Figure 6.13: Absolute values of the reflection and transmission energy coefficients versus incidence propagation angle (a) and ray (energy) angle (b) corresponding to the viscoelastic case.



Figure 6.14: Energy (a) and attenuation (b) angles versus incidence angle for the incident, reflected and transmitted waves. The propagation angle is also represented in both cases.



Figure 6.15: Phase angles versus incidence propagation angle for the incident, reflected and transmitted waves corresponding to the elastic (a) and viscoelastic (b) cases.



Figure 6.16: Phase velocities of the incident, reflected and transmitted waves versus the incidence propagation angle for the viscoelastic case.

if $Q \gg 1$. Finally, Figure 6.20 shows the square root of the interference coefficients versus the incidence propagation angle. It indicates that much of the energy is lost due to interference between the different waves beyond the elastic critical angle. The interference between the P_T and S_T waves is particularly high and is comparable to ET_{PP} around 30° incidence. Note that these coefficients vanish in the elastic case.

The reflection-transmission problem can be solved for transient fields by using the equations given in Section 4.5.4. A wave modeling algorithm based on the Fourier pseudospectral method is used to compute the spatial derivatives, and a fourth-order Runge-Kutta technique to compute the wave field recursively in time (see Chapter 9). The numerical mesh has 231×231 points with a grid spacing of 20 m. The source is a Rickertype wavelet located at 600 m above the interface, and has a dominant frequency of 12.625 Hz, i.e., the central frequency of the relaxation peaks. In order to generate mainly qP energy, the source is a discrete delta function, equally distributed in the stress components σ_{11} and σ_{33} – a mean stress perturbation. Figure 6.21 shows a snapshot at 800 ms, which covers the incidence ray angles from 0° to approximately 62°. In the upper medium, the primary waves are the qP wave followed by the qS wave, which shows high amplitude cuspidal triangles despite the dilatational nature of the source. Moreover, the P_R and S_R are traveling upwards, away from the interface. Near the center of the mesh, the events are mainly related to the reflection of the cuspidal triangles. In the lower medium, the P_T wave is followed by the S_T wave, which resembles a continuation of the incident qP wave, since both events have similar velocities (see Figure 6.11). In principle, Figure 6.21 should be interpreted by comparison with Figure 6.13. However, Figure 6.21 displays the vertical particle velocity v_3 , and Figure 6.13b the square root of the normal power flow. Moreover, the interpretation must take into account that the source has a non-isotropic radiation pattern, and that the incidence wave is also affected by anisotropic attenuation effects. Despite these considerations, a qualitative interpretation can be attempted. First,



Figure 6.17: Attenuations of the incident, reflected and transmitted waves versus the incidence propagation angle. Figure 6.17b corresponds to the transmitted quasi-compressional wave.



Figure 6.18: Energy velocities of the incident, reflected and transmitted waves versus the incidence propagation angle for the viscoelastic case.



Figure 6.19: Quality factors of the incident, reflected and transmitted waves versus the incidence propagation angle for the viscoelastic case.



Figure 6.20: Square root of the interference coefficients versus the incidence propagation angle.



Figure 6.21: Snapshot of the vertical particle-velocity component v_3 , corresponding to the viscoelastic reflection-transmission problem at 800 ms.

the amplitudes of the S_R and S_T waves are very low at normal incidence, as predicted by the ER_{PS} and ET_{PS} curves, respectively. In particular, the amplitude of the S_T wave increases for increasing ray angle, in agreement with ET_{PS} . In good agreement also, is the amplitude variation of the P_T wave compared to the ET_{PP} curve. Another event is the planar wave front connecting the reflected and transmitted qP waves. This is a conical or head wave that cannot be entirely explained by the plane-wave analysis. Despite the fact that a critical angle does not exist, since ψ_{PT} never reaches $\pi/2$ (see Figure 6.14a), some of the P_T energy disturbs the interface, giving rise to the conical wave.

6.2.9 Incident inhomogeneous waves

In the previous section, we assumed incident homogeneous waves. Here, we consider the more realistic case of inhomogeneous plane waves, illustrated with a geophysical example. In offshore seismic exploration, the waves transmitted at the ocean bottom have a particular characteristic. Assuming that water is lossless and using Snell's law (Section 3.5), their attenuation vectors are perpendicular to the ocean-bottom interface. This fact affects the amplitude variations with offset (AVO) of reflection events generated at the lower layers.

Winterstein (1987) investigates the problem from a "kinematic" point of view. He analyzes how the angle between propagation and maximum attenuation varies in an anelastic layered medium, and shows that departures from elastic wave ray paths can be large. In addition, compressional-wave reflection coefficients for different incidence inhomogeneity angles are compared by Krebes (1984). He shows that the deviations from the elastic case can be important at supercritical angles.

Here, we study the AVO response for an inhomogeneous wave generated at the ocean bottom and incident at a lower interface separating two viscoelastic transversely isotropic media. Unlike the analysis performed by Krebes (1984), the inhomogeneity angle is not constant with offset, but is equal to the incidence angle, since the interface is assumed to be parallel to the ocean bottom (see Figure 6.22). The interface may separate two finely layered formations whose contact plane is parallel to the stratification, or two media with intrinsic anisotropic properties, such as shale and limestone.

The consistent 2-D stress-strain relation for qP-qS propagation is given in Section 4.5.4, based on model 3 (see Section 4.1.3). The convention is to denote the quasidilatational and quasi-shear deformations with $\nu = 1$ and $\nu = 2$, respectively. The complex stiffnesses relating stress and strain for a 2-D transversely isotropic medium (4.161) can be expressed as

$$p_{11} = c_{11} - \frac{1}{2}(c_{11} + c_{33}) + \left[\frac{1}{2}(c_{11} + c_{33}) - c_{55}\right] M_1 + c_{55}M_2$$

$$p_{33} = c_{33} - \frac{1}{2}(c_{11} + c_{33}) + \left[\frac{1}{2}(c_{11} + c_{33}) - c_{55}\right] M_1 + c_{55}M_2$$

$$p_{13} = c_{13} - \frac{1}{2}(c_{11} + c_{33}) + \left[\frac{1}{2}(c_{11} + c_{33}) - c_{55}\right] M_1 + c_{55}(2 - M_2)$$

$$p_{55} = c_{55}M_2.$$
(6.167)

The elasticity constants c_{IJ} , I, J = 1, ..., 6 are the unrelaxed or high-frequency limit stiffnesses, and $M_{\nu}(\omega)$ are dimensionless complex moduli. For one Zener mechanism, M_{ν} is given in equation (4.6). The form of $M_{\nu}(\omega)$ for L Zener models connected in parallel is given in equation (2.196). In the lossless case ($\omega \rightarrow \infty$), $M_{\nu} \rightarrow 1$.



Figure 6.22: Snell's law for a plane wave incident on the ocean-bottom interface. The diagram shows the continuity of the horizontal component of the complex-slowness vector. In the ocean, this vector is real, since water is assumed to be lossless. In the shale layer the attenuation vector is perpendicular to the ocean bottom.

Generation of inhomogeneous waves

Let us assume that the positive z-axis points downwards. A general solution for the particle-velocity field $\mathbf{v} = (v_1, v_3)$ is

$$\mathbf{v} = i\omega \mathbf{U} \exp [i\omega (t - s_1 x - s_3 z)], \quad (6.168)$$

where s_1 and s_3 are the components of the complex-slowness vector and U is a complex vector. The slowness vector

$$\mathbf{s}_R = (\text{Re}(s_1), \text{Re}(s_3))$$
 (6.169)

and the attenuation vector

$$\alpha = -\omega(Im(s_1), Im(s_3)),$$
 (6.170)

in general, will not point in the same direction. Figure 6.22 depicts a transmitted inhomogeneous wave generated at the ocean bottom. As mentioned before, since the attenuation vector of waves propagating in the water layer is zero, the viscoelastic Snell's law implies that the attenuation vector of the transmitted wave is perpendicular to the ocean bottom. Note that the inhomogeneity angle is equal to the propagation angle θ .

The complex-slowness components below the ocean bottom are

$$s_1 = s_R \sin \theta, \quad s_3 = s_R \cos \theta - \frac{i\alpha}{\omega},$$
 (6.171)

where s_R and α are the magnitudes of \mathbf{s}_R and $\boldsymbol{\alpha}$, respectively. For a given angle θ , s_R and α can be computed from the dispersion relation (6.113). Then, the substitution of these quantities into equation (6.171) yields the slowness components of the incident inhomogeneous wave. However, this method requires the numerical solution of two fourthdegree polynomials. A simpler approach is the following: Assume a given propagation angle θ_H for a hypothetical transmitted homogeneous wave below the ocean bottom. Then, according to equation (6.125), the complex slowness is

$$s = \sqrt{2\rho} (p_{55} + p_{11} \sin^2 \theta_H + p_{33} \cos^2 \theta_H \pm C)^{-1/2},$$
 (6.172)

where ρ is the density and C is given by equation (6.126) with $\theta = \theta_H$.

- Choose s₁ for the inhomogeneous wave equal to Re(s) sin θ_H, a real quantity according to Snell's law – since the projection of α on the interface is zero.
- Compute s₃ from the dispersion relation (6.113).
- Compute the incidence propagation angle θ for the inhomogeneous wave from sin θ = s₁/|s| as

$$\theta = \arcsin\left(\frac{s_1}{\sqrt{s_1^2 + [\text{Re}(s_3)]^2}}\right).$$
 (6.173)

In this way, a vector (s_1, s_3) , satisfying equation (6.113) and providing input to the reflection-transmission problem, can be obtained for each incidence angle θ . The price we pay for this simplicity is that the ray angle does not reach 90°, but this is not relevant since the offsets of interest in exploration geophysics are sufficiently covered.

Ocean bottom

The material properties of the incidence and transmission media – shale and chalk, respectively – are given in Table 6.1, where $v_{IJ} = \sqrt{c_{IJ}/\rho}$.

ROCK	v ₁₁ (m/s)	v ₃₃ (m/s)	v35 (m/s)	v ₁₃ (m/s)	Q_{01}	Q_{02}	$\rho (g/cm^3)$
shale	3810	3048	1402	1828	10	5	2.3
chalk	5029	5029	2621	3414	100	70	2.7

Table 6.1. Material properties

The unrelaxed velocities are indicated in the table, and attenuation is quantified by the parameters $Q_{0\nu} = \text{Re}(M_{\nu})/\text{Im}(M_{\nu})$ at the reference frequency. Wright (1987) calculates the reflection coefficients for the elastic case, which is obtained in the unrelaxed limit.

The comparison between the absolute values of the qP wave reflection coefficients, together with the corresponding phase angles, is shown in Figure 6.23. In the figure, "E" corresponds to the elastic case (i.e., elastic shale), "H" to an incident viscoelastic homogeneous wave, and "I" to an incident inhomogeneous wave with the characteristics indicated in Figure 6.22 – the chalk is assumed anelastic in the three cases. In the elastic case, i.e., shale and chalk both elastic (Wright, 1987), there is a critical angle between 40° and 50°. It can be shown that the energy vector of the transmitted qP wave points downwards for all incidence angles. Thus, there is no critical angle in the strict sense. However, the shape of the E and I curves indicates that a quasi-evanescent wave propagates through the interface. This character is lost in the H curve. In the near-offsets



Figure 6.23: Comparison between the absolute values of the R_{PP} reflection coefficients together with the corresponding phase angles, where "E" corresponds to the elastic case (i.e., elastic shale), "H" to an incident viscoelastic homogeneous wave, and "I" to an incident inhomogeneous wave exhibiting the characteristics indicated in Figure 6.22.

- up to 20° - the three coefficients follow the same trend and are very similar to each other. The difference between this case and the elastic case (E) is due to the anelastic properties of the shale. Beyond 30°, the differences are important, mainly for the incident homogeneous wave. These can also be observed in the phase where the H curve has the opposite sign with respect to the other curves. A similar effect is reported by Krebes (1984). More details and results about this problem are given by Carcione (1999b).

6.3 Reflection and transmission at fluid/solid interfaces.

Fluid/solid interfaces are important in seismology and exploration geophysics, particularly in offshore seismic prospecting, where the ocean bottom is one of the main reflection events. We consider this problem by assuming an incident homogeneous P wave.

6.3.1 Solid/fluid interface

A general plane-wave solution for the particle-velocity field $\mathbf{v} = (v_1, v_3)$ is

$$\mathbf{v} = i\omega \mathbf{U} \exp [i\omega(t - s_1 x - s_3 z)], \qquad (6.174)$$

where U is a complex vector of magnitude U. For homogeneous waves,

$$s_1 = \sin \theta / v_{P_1}$$
, (6.175)

where θ is the propagation angle measured with respect to the z-axis, and v_{P_1} is the complex velocity, in this case, the complex P-wave velocity of the solid. Let us denote the complex S-wave velocity of the solid by v_{S_1} and the complex P-wave velocity of the fluid by v_{P_2} .

The upper viscoelastic medium (the solid) is denoted by subscript 1 and the viscoacoustic medium (the fluid) by subscript 2. The symbol P indicates the compressional wave in the fluid or the P wave in the upper layer, and S denotes the S wave in this medium. As before, the subscripts I, R and T denote the incident, reflected and transmitted waves. Using symmetry properties to define the polarization of the reflected waves and using the fact that Snell's law implies the continuity of the horizontal slowness s_1 (see Section 3.5), we note that the particle velocities for a P wave incident from the upper medium are given by

$$v_1 = v_{P_\ell} + v_{P_R} + v_{S_R}$$
, (6.176)

$$v_2 = v_{P_T}$$
, (6.177)

where

$$\mathbf{v}_{P_{I}} = i\omega(\beta_{P_{1}}, \xi_{P_{1}}) \exp[i\omega(t - s_{1}x - s_{3P_{1}}z)],$$
 (6.178)

$$\mathbf{v}_{P_R} = i\omega R_{PP}(\beta_{P_1}, -\xi_{P_1}) \exp[i\omega(t - s_1x + s_{3P_1}z)],$$
 (6.179)

$$\mathbf{v}_{S_R} = i\omega R_{PS}(\beta_{S_1}, -\xi_{S_3}) \exp[i\omega(t - s_1x + s_{3S_1}z)].$$
 (6.180)

$$\mathbf{v}_{P_T} = i\omega T_{PP}(\beta_{P_2}, \xi_{P_2}) \exp[i\omega(t - s_1x - s_{3P_2}z)],$$
 (6.181)

and the choice U = 1 implies no loss of generality. If we assume an isotropic solid, the slownesses and vertical slowness components are

$$\begin{split} s_{P_1} &= 1/v_{P_1}, \quad s_{3P_1} = \text{pv}\sqrt{s_{P_1}^2 - s_1^2} \\ s_{S_1} &= 1/v_{S_1}, \quad s_{3S_1} = \text{pv}\sqrt{s_{S_1}^2 - s_1^2} \\ s_{P_2} &= 1/v_{P_2}, \quad s_{3P_2} = \text{pv}\sqrt{s_{P_2}^2 - s_1^2}, \end{split}$$
(6.182)

and the polarizations are

$$\beta_{P_m} = \frac{s_1}{s_{P_m}}, \quad \xi_{P_m} = \frac{s_{3P_m}}{s_{P_m}}, \quad \beta_{S_1} = \frac{s_{3S_1}}{s_{S_1}}, \quad \xi_{S_1} = -\frac{s_1}{s_{S_1}}, \quad m = 1, 2.$$
 (6.183)

The boundary conditions require continuity of

$$v_3$$
, σ_{33} , and $\sigma_{13}(=0)$. (6.184)

These conditions generate the following matrix equation for the reflection and transmission coefficients:

$$\begin{pmatrix} \xi_{P_1} & \xi_{S_1} & \xi_{P_2} \\ Z_{P_1} & Z_{S_1} & -Z_{P_2} \\ W_{P_1} & W_{S_1} & 0 \end{pmatrix} \cdot \begin{pmatrix} R_{\rm PP} \\ R_{\rm PS} \\ T_{\rm PP} \end{pmatrix} = \begin{pmatrix} \xi_{P_1} \\ -Z_{P_1} \\ W_{P_1} \end{pmatrix},$$
(6.185)

where, for P_1 or S_1

$$Z = \rho_1 v_{P_1}^2 \xi s_3 + \rho_1 (v_{P_1}^2 - 2v_{S_1}^2) \beta s_1, \quad W = \rho_1 v_{S_1}^2 (\beta s_3 + \xi s_1)$$
(6.186)

for the upper medium – depending on the wave type, the subindex of $\xi,\,\beta$ and s_3 is P_1 or S_1 – and

$$Z_{P_2} = \rho_2 v_{P_2}^2 (\xi_{P_2} s_{3P_2} + \beta_{P_2} s_1), \quad W_{P_2} = 0$$
 (6.187)

for the fluid.

6.3.2 Fluid/solid interface

In this case, the fluid is denoted by the subscript 1 and the lower layer by the subscript 2. The particle velocities for a P wave incident from the fluid are given by

$$v_1 = v_{P_I} + v_{P_B}$$
, (6.188)

$$v_2 = v_{P_T} + v_{S_T}$$
, (6.189)

where

$$\mathbf{v}_{P_{f}} = i\omega(\beta_{P_{1}}, \xi_{P_{1}}) \exp[i\omega(t - s_{1}x - s_{3P_{1}}z)],$$
 (6.190)

$$\mathbf{v}_{P_R} = i\omega R_{PP}(\beta_{P_1}, -\xi_{P_1}) \exp[i\omega(t - s_1x + s_{3P_1}z)],$$
 (6.191)

$$\mathbf{v}_{P_{T}} = i\omega T_{PP}(\beta_{P_{2}}, \xi_{P_{2}}) \exp [i\omega(t - s_{1}x - s_{3}P_{2}z)],$$
 (6.192)

$$\mathbf{v}_{S_T} = i\omega T_{PS}(\beta_{S_2}, \xi_{S_2}) \exp[i\omega(t - s_1x - s_{3S_2}z)],$$
 (6.193)
The boundary conditions (6.184) generate the following matrix equation for the reflection and transmission coefficients:

$$\begin{pmatrix} \xi_{P_1} & \xi_{P_2} & \xi_{S_2} \\ Z_{P_1} & -Z_{P_2} & -Z_{S_2} \\ 0 & W_{P_2} & W_{S_2} \end{pmatrix} \cdot \begin{pmatrix} R_{PP} \\ T_{PP} \\ T_{PS} \end{pmatrix} = \begin{pmatrix} \xi_{P_1} \\ -Z_{P_1} \\ 0 \end{pmatrix}, \quad (6.194)$$

where β_{P_1} , β_{P_2} , β_{S_2} , ξ_{P_1} , ξ_{P_2} , ξ_{S_2} , Z_{P_1} , Z_{P_2} , Z_{S_2} , W_{P_2} and W_{S_2} are obtained from equations (6.183), (6.186) and (6.187), with the material indices interchanged.

The reflection and transmission equations for an anisotropic viscoelastic solid are similar to equations (6.185) and (6.194), but use the appropriate expressions for the β 's, ξ 's, Z's and the W's. (This exercise is left to the reader.)

6.3.3 The Rayleigh window

In this section, we use the reflection-transmission theory to explain a phenomenon that cannot be modeled with a lossless stress-strain relation. Brekhovskikh (1960, p. 34) observed that the amplitude reflection coefficient measured for a water-steel interface was not consistent with that predicted by the elastic theory. Beyond the elastic S critical angle, there is reduction in amplitude of the reflected P wave in a narrow window. Because this occurs for an angle where the apparent phase velocity of the incident wave is near that of the Rayleigh surface wave, the phenomenon is called the "Rayleigh window". The corresponding reflection coefficient was measured experimentally by F. Becker and R. Richardson, and their ultrasonic experiments were verified with an anelastic model in a later paper (Becker and Richardson, 1972). Borcherdt, Glassmoyer and Wennerberg (1986) compared theoretical and experimental results corresponding to the same experiment, and show that the same phenomenon takes place at ocean-bottom interfaces. They find that the anelastic Rayleigh window should be observable in appropriate sets of wideangle reflection data and can be useful in estimating attenuation for various ocean-bottom reflectors. The presence of inhomogeneous viscoelastic waves accounts for the existence of the anelastic Rayleigh window.

The scattering equations involved in this problem are given in Section 6.3.2. The complex velocity of the fluid – a viscoacoustic medium – is

$$v_{P_1} = c_{P_1}\sqrt{M}, \quad M(\omega) = \frac{\sqrt{Q_0^2 + 1} - 1 + i\omega Q_0 \tau_0}{\sqrt{Q_0^2 + 1} + 1 + i\omega Q_0 \tau_0},$$
 (6.195)

where c_{P_1} is the unrelaxed wave velocity of water, and M is a dimensionless complex modulus. At $\omega_0 = 1/\tau_0$, the quality factor of water has the lowest value Q_0 (see Section 4.1 and equation (4.6)).

The complex Lamé constants for steel are given by

$$\mathcal{E}_2 = \rho_2 \left[\left(c_{P_2}^2 - \frac{4}{3} c_{S_2}^2 \right) M_1 + \frac{4}{3} c_{S_2}^2 M_2 \right] \quad \text{and} \quad \mu_2 = \rho_2 c_{S_2}^2 M_2, \tag{6.196}$$

where c_{P_2} and c_{S_2} are the unrelaxed P- and S-wave velocities of steel, and M_1 and M_2 are dimensionless complex moduli, defined in equation (4.6).

The properties of water are $c_{P_1} = 1490$ m/s, $\rho_1 = 1000$ kg/m³, and $Q_0^{-1} = 0.00012$ at $f_0 = 10$ MHz ($f_0 = 1/2\pi\tau_0$). The unrelaxed velocities of steel are $c_{P_2} = 5761$ m/s, and c_{S_2} = 3162 m/s, respectively, the density is $\rho_2 = 7932 \text{ kg/m}^3$ and the dissipation factors at 10 MHz are $Q_{01}^{-1} = 0.0037$ and $Q_{02}^{-1} = 0.0127$. We recall that Q_{01} is a quality factor associated with dilatational deformations and not with the compressional wave. These properties give the homogeneous P- and S-wave dissipation factors and phase velocities indicated in Table 1 of Borcherdt, Glassmoyer and Wennerberg (1986), for a frequency of 10 MHz. Figure 6.24 represents the reflection coefficient (solid line), compared to the experimental values (open circles). The dashed line corresponds to the elastic case. This experiment and its theoretical prediction is a demonstration of the existence of inhomogeneous body waves.



Figure 6.24: Amplitude reflection coefficient predicted for the anelastic Rayleigh window by a viscoelastic model (solid line), compared to the experimental values (open circles) (Becker and Richardson, 1972) for a water-stainless steel interface. The dashed line corresponds to the elastic case.

6.4 Reflection and transmission coefficients of a set of layers

The propagation of waves in solid layers has numerous applications in acoustics and optics. In seismology, for instance, a plane layered system can be a good representation of a stratified Earth. Let a plane wave, with horizontal complex slowness s_1 , be incident on the symmetry plane of an orthorhombic medium, as shown in Figure 6.25. Inside the layer, the particle-velocity field is a superposition of upgoing and downgoing quasicompressional (P) and quasi-shear waves (S) of the form

$$\mathbf{v} = \begin{pmatrix} v_1 \\ v_3 \end{pmatrix} = \mathbf{i}\omega \begin{bmatrix} U_P^- \begin{pmatrix} \beta_P \\ -\xi_P \end{pmatrix} \exp(\mathbf{i}\omega s_{3P}z) + U_S^- \begin{pmatrix} \beta_S \\ -\xi_S \end{bmatrix} \exp(\mathbf{i}\omega s_{3S}z) + U_P^+ \begin{pmatrix} \beta_P \\ \xi_P \end{bmatrix} \exp(-\mathbf{i}\omega s_{3P}z) + U_S^+ \begin{pmatrix} \beta_S \\ \xi_S \end{bmatrix} \exp(-\mathbf{i}\omega s_{3S}z) \end{bmatrix} \exp[\mathbf{i}\omega(t - s_1x)], \quad (6.197)$$

where U^- are upgoing-wave amplitudes, U^+ are downgoing-wave amplitudes, and β and ξ are the polarization components, given in equations (6.116) and (6.117), respectively. The vertical slowness components s_{3P} and s_{3S} are given in equation (6.114). Normal stresses and strains are related by

$$i\omega \sigma_{33} = p_{13}\partial_1 v_1 + p_{33}\partial_3 v_3,$$
 (6.198)

$$i\omega \sigma_{13} = p_{55}(\partial_3 v_1 + \partial_1 v_3)$$
 (6.199)



Figure 6.25: Diagram showing an orthorhombic layer embedded between two isotropic half-spaces.

Using equations (6.197), (6.198) and (6.199), the particle-velocity/stress array, inside the layer at depth z, can be written as

$$\mathbf{t}(z) = \begin{pmatrix} v_1 \\ v_3 \\ \sigma_{33} \\ \sigma_{13} \end{pmatrix} = \mathbf{T}(z) \cdot \begin{pmatrix} U_P^- \\ U_S^- \\ U_P^+ \\ U_S^+ \end{pmatrix}, \quad (6.200)$$

where

$$\mathbf{T}(z) = i\omega \begin{pmatrix} \beta_P & \beta_S & \beta_P & \beta_S \\ -\xi_P & -\xi_S & \xi_P & \xi_S \\ -Z_P & -Z_S & -Z_P & -Z_S \\ W_P & W_S & -W_P & -W_S \end{pmatrix}$$

$$\cdot \begin{pmatrix} \exp(i\omega s_{3P}z) & 0 & 0 & 0 \\ 0 & \exp(i\omega s_{3S}z) & 0 & 0 \\ 0 & 0 & \exp(-i\omega s_{3P}z) & 0 \\ 0 & 0 & 0 & \exp(-i\omega s_{3S}z) \end{pmatrix}, \quad (6.201)$$

6.4 Reflection and transmission coefficients of a set of layers

with W and Z given in equations (6.121) and (6.123), respectively.

Then, the fields at z = 0 and z = h are related by the following equation:

$$t(0) = \mathbf{B} \cdot t(h), \quad \mathbf{B} = \mathbf{T}(0) \cdot \mathbf{T}^{-1}(h),$$
 (6.202)

which plays the role of a boundary condition. Note that when h = 0, **B** is the identity matrix.

Let us denote by the subscript 1 the upper half-space and by the subscript 2 the lower half-space. Moreover, the subscripts I, R and T denote the incident, reflected and transmitted waves. Using symmetry properties to define the polarization of the reflected waves, the particle velocities for a P wave incident from above the layer are given by

$$v_1 = v_{P_1} + v_{P_R} + v_{S_R}$$
, (6.203)

$$v_2 = v_{P_T} + v_{S_T}$$
, (6.204)

where the particle velocities of the right-hand side have the same form as equations (6.133)-(6.137), where, here,

$$\begin{pmatrix} \beta_P \\ \xi_P \end{pmatrix} = \frac{1}{\sqrt{s_1^2 + s_{3P}^2}} \begin{pmatrix} s_1 \\ s_{3P} \end{pmatrix}, \qquad \begin{pmatrix} \beta_S \\ \xi_S \end{pmatrix} = \frac{1}{\sqrt{s_1^2 + s_{3S}^2}} \begin{pmatrix} s_{3S} \\ -s_1 \end{pmatrix}, \tag{6.205}$$

with

$$s_1^2 + s_{3P_i}^2 = \frac{\rho_i}{E_i} \equiv \frac{1}{v_{P_i}^2}, \quad s_1^2 + s_{3S_i}^2 = \frac{\rho_i}{\mu_i} \equiv \frac{1}{v_{S_i}^2}, \quad i = 1, 2,$$
 (6.206)

where v_{P_i} and v_{S_i} are the complex compressional and shear velocities, respectively. On the other hand, the W and Z coefficients for the isotropic half-spaces are

$$W_{P_i} = 2\mu_i s_{3P_i} s_1 v_{P_i}, \qquad W_{S_i} = \mu_i (s_{3S_i}^2 - s_1^2) v_{S_i},$$
 (6.207)

$$Z_{P_i} = (\lambda_i s_1^2 + E_i s_{3P_i}^2) v_{P_i}, \quad Z_{S_i} = -2\mu_i s_{3S_i} s_1 v_{S_i},$$
 (6.208)

where $\lambda_i = E_i - 2\mu_i$ and μ_i are complex Lamé constants. Using equations (6.203) and (6.133)-(6.137), the particle-velocity/stress field at z = 0 can be expressed as

$$t(0) = A_1 \cdot r + i_P$$
, (6.209)

where

$$\mathbf{r} = (R_{PP}, R_{PS}, T_{PP}, T_{PS})^{\top},$$
 (6.210)

$$\mathbf{i}_{P} = \mathbf{i}\omega(\beta_{P_{1}}, \xi_{P_{1}}, -Z_{P_{1}}, -W_{P_{1}})^{\top},$$
 (6.211)

and

$$\mathbf{A}_{1} = i\omega \begin{pmatrix} \beta_{P_{1}} & \beta_{S_{1}} & 0 & 0 \\ -\xi_{P_{1}} & -\xi_{S_{1}} & 0 & 0 \\ -Z_{P_{1}} & -Z_{S_{1}} & 0 & 0 \\ W_{P_{1}} & W_{S_{1}} & 0 & 0 \end{pmatrix}.$$
(6.212)

Using equations (6.204) and (6.133)-(6.137), the particle-velocity/stress field at z = h can be expressed as

$$t(h) = A_2 \cdot r$$
, (6.213)

where

$$\mathbf{A}_{2} = i\omega \begin{pmatrix} 0 & 0 & \beta_{P_{2}} \exp(-i\omega s_{3P_{2}}h) & \beta_{S_{2}} \exp(-i\omega s_{3S_{2}}h) \\ 0 & 0 & \xi_{P_{2}} \exp(-i\omega s_{3P_{2}}h) & \xi_{S_{2}} \exp(-i\omega s_{3S_{2}}h) \\ 0 & 0 & -Z_{P_{2}} \exp(-i\omega s_{3P_{2}}h) & -Z_{S_{2}} \exp(-i\omega s_{3S_{2}}h) \\ 0 & 0 & -W_{P_{2}} \exp(-i\omega s_{3P_{2}}h) & -W_{S_{2}} \exp(-i\omega s_{3S_{2}}h) \end{pmatrix}.$$
(6.214)

Combining equations (6.202), (6.209) and (6.213) yields a matrix equation for the reflectionand transmission-coefficient array r:

$$(\mathbf{A}_1 - \mathbf{B} \cdot \mathbf{A}_2) \cdot \mathbf{r} = -\mathbf{i}_P.$$
 (6.215)

The reflection and transmission coefficients R_{SP} , R_{SS} , T_{SP} and T_{SS} for an incident S wave have the same scattering matrix as the P incident wave, but the array i_P is replaced by

$$\mathbf{i}_{S} = \mathbf{i}\omega(\beta_{S_{1}}, \xi_{S_{1}}, -Z_{S_{1}}, -W_{S_{1}})^{\top}.$$
 (6.216)

In the absence of layer, h = 0, **B** is the identity matrix, and we get the system of equations obtained in Section 6.2.3. When the upper and lower half-spaces are the same medium, it can be shown that the absolute value of the PP-reflection coefficient at normal incidence is given by

$$R_{PP}(0) = \frac{2|R_0 \sin(kh)|}{|R_0^2 \exp(-ikh) - \exp(ikh)|}, \quad (6.217)$$

where

$$k = \frac{\omega}{v_P}, \quad v_P = \sqrt{\frac{p_{33}}{\rho}},$$

and

$$R_0 = \frac{\rho v_P - \rho_1 v_{P_1}}{\rho v_P + \rho_1 v_{P_1}},$$

with index 1 denoting the upper and the lower half-spaces. It is straightforward to generalize this approach for computing the seismic response of a stack of viscoelastic and anisotropic layers. We consider N layers with stiffnesses $p_{IJ\alpha}$, density ρ_{α} , each of them with thickness h_{α} , such that the total thickness is

$$h = \sum_{\alpha=1}^{N} h_{\alpha}.$$
 (6.218)

By matching boundary conditions at the interfaces between layers, it is easy to show that the matrix system giving the reflection and transmission coefficients is

$$\left[\mathbf{A}_{1} - \left(\prod_{\alpha=1}^{N} \mathbf{B}_{\alpha}\right) \cdot \mathbf{A}_{2}\right] \cdot \mathbf{r} = -\mathbf{i}_{P(S)}, \qquad (6.219)$$

where $i_{P(S)}$ is the incidence P(S) array, and

$$B_{\alpha} = T(0) \cdot T^{-1}(h_{\alpha}), \quad \alpha = 1, ..., N.$$
 (6.220)

This recursive approach, which is the basis of most reflectivity methods, dates back to Thomson (1950), and is illustrated by Brekhovskikh (1960, p. 61) for a stack of isotropic and elastic layers. An example of the application of this approach can be found in Carcione (2001b), where amplitude variations with offset (AVO) of pressure-seal reflections are investigated. Ursin and Stovas (2002) derive a second-order approximation for the reflection and transmission coefficients, which is useful for the inversion of seismic reflection data.

234

Chapter 7

Biot's theory for porous media

In Accustics, we have concerned to consider the involution of accual waves upon prioris lookes, in which advistices where such of accual contractly is preserved...

The problem of propagation of sound in a circular labe, having regard to the influence of recessing and heat conduction have been solved analytically by Kirchhoff, on the supportions that the tangential relaxity and the temperature currentian variable at the walks. In discussion, the solution Kirchhoff takes the case in which the dimensions of the table are such that the numerical effects of the dissipative forces are confined to a relatively that students in the resplacement of the walks. In the present apple more interest attractes rather to the appearter extended of the data walks. In the present apple more interest attractes rather to the appearter extends of the data walks to so small that the free total layer prefix will fills the table. Nothing proceeding is last by month example forces and to conclude the totals (following Kirchhoff) = that the extension of propagation of escars and the total effects is negligible to examplify both that of sound

John William Strutt (Lord Rayleigh) (Bayleigh, 1899b)

Biot scheory describes wave propagation in a porous saturated medium, i.e., a medium made of a solid matrix eskeleton or frame), fully saturated with a fluid. Biot (1956a,b) ignores the increase opic level and assumes that continuum mechanics can be applied to measurable management of quantities. He postulates the Lagrangian and uses Hamilton's principle to derive the equations governing wave propagation. Bigmons approaches for obtaining the equations of motion are the homogenization, theory edg. Burridge and Keller, 1985) and volume-averaging methods to g. Pride, Gauge and Morgan, 1991; Pride and Berryman, 1998), both of which relate the uncroscopic and macroscopic worlds. We follow Biot's approach, due to its simplicity.

Sound attenuation in air-fallest porous media was investigated by Zwikker and Kosten (1949). They considered dilutational waves and described the physics of wave propagation by using the energy of impedance. Biot's theory and related theories of deformation and wave pupagation in porous media are discussed in several reviews and honks, notably, Rice and Cleary (1976). Johnson (1984). Bouthié, Coussy and Zinszner (1987). Cristesen (1986). Scoll (1989). Zimmerinam (1991). Allard (1993). Consey (1995). Comprisedured (1986). Mayko, Makery and Dyorkin (1998). Wang (2006). Coderbaum, Li, and Schulgasser (2000). Santamerina, Klem and Lune (2001), and Kong (2005).

Extensions of Biot's theory, from first principles, are given by Brutsaert (1964) and Santos, Douglas and Corbers' (1990) for partial saturation – one solid and two fluids – (see simulations in Carrione – Cavallini – Santos, Basazzah and Gauzellino (2004)): Leclaire. Cohen-Ténondji and Agnirros Fuence (1995) for frozen media: Caroione, Gurevala and Cavallini (2000) for shaley sandstones: Carrione and Seriani (2000) for frozen sediments two subdytind one fluid (precisimulations in Curchure, Santos, Royazzoli and Helle (2000)) and Berryman and Wang (2000) for a double principly dual primeability medium. The extension to non-isotic that conditions, to account for the effects of thermal expansion of both the pore fluid and the matrix, are given, for instance, in Me Figure (1986). In a porous medium saturatest with a fluid electrolyte, acoustic and electromagnetic waves are coupled (see Section 8,15). The extension of Basi's theory to describe the phenomenon is given in Probe (1994) and Pride and Haartsen (1998) (electro-service wave propagation). An important reference are the collected papers of M. A. Bast resulting from the conference held in his memory in Louvain la Neuve (Thimus, Moushrinan, Cheng, Conservant Detourney, 1998).

The manuassimptions of the theory are:

- Infinitesimal transformations occur between the reference and current states of deformation. Displacements, strong and particle velocities are small. Consequently, the Eulerian and Lagrangian formulations coincide up to the fust-order. The constitutive equations, dissipation forces, and kinetic minimum are linear. (The strain energy, dissipation potential and kinetic energy are quadratic forms in the field variables.)
- The principles of continuum mechanics can be applied to measurable macroscopic values. The macroscopic quantities used in Biel's theory are volume averages of the corresponding narroscopic quantities of the constrainents.
- 4. The wavelength is large compared with the chinensions of a macroscopic elementary volume. This volume has well defined properties, such as porosity, permeability and elastic moduli, which are representative of the medium. Scattering effects are thus neglicited.
- 4. The conditions are isothermal.
- The stress distribution in the fluid is hydrostatic. (It may be not completely hydrostatic, since the fluid is viscous.)
- The liquid phase is continuous. The matrix consists of the solid phase and disconnected pores, which do not contribute to the porosity.
- In most cases, the material of the frame is isotropic. Anisotropy is due to a preferential alignment of the pores (or cracks).

Our approach is based on energy considerations. We define the strain, kinetic and desequated energies, and obtain the explanation of motion by solving Lagrange's equations, in the following discussion, the solid matrix is indicated by the index " m_{\odot} , the solid by the index "s" and the fluid phase by the index "f".

7.1 Isotropic media. Strain energy and stress-strain relations

The displacement vectors and strain tensors of the frame and the fluid are marroscopic averages, will defined in the macroscopic elementary vulnum. The stresses are forces acting on the frame or the fluid per unit area of porms material. The stress components for the fluid are

$$\sigma_{ij}^{(I)} \sim -i \eta \eta \delta_{ij}$$
. (7.1)

where p₁ is the fluid pressure and ce is the porosity.

Taking into account equation (1.17), and since the fluid dors ant "support" shear stresses we express the straiteneously of the pursus mediator as

$$V = A\sigma_{m}^{2} + D\sigma_{m}^{2} + C\partial_{m}\partial_{T} + D\sigma_{C}$$
(7.2)

where A, B, C and D are elasticity coefficients to be determined as a function of the solid and fluid properties, as well as by the microstructural properties of the medium. Note the coupling term between the solid and the fluid represented by the coefficient C. The stress components are given by

$$\pi_{b_{\mu}}^{\mu} = \frac{\partial V}{\partial e_{\mu}^{\mu\nu}}, \quad \text{and} \quad \sigma^{(f)} = \frac{\partial V}{\partial e_{\mu}}, \quad (7.3)$$

where $\sigma^{(j)} = -\alpha p_j$. Using these equations, the stress-strain relations are

$$\sigma_{ij}^{(\ell)} = -2B\delta_{ij}^{(\ell)} + (2A\delta_i + C^{ij})\delta_{ij},$$
 (7.4)

and

$$\sigma^{(f)} = C \delta_m + 2D \delta_f,$$
 (7.5)

In order to obtain the elasticity coefficients in terms of known properties, we consider three ideal experiments (under static conditions (Biot and Willis, 1957)). First, the material is subjected to a pure shear differmation $(R_n = R_f = 0)$. In this case, it is clear that R is the shear modulus of the frame, since the fluid does not contribute to the shearing loree. Let us denote

$$B = \rho_{m}$$
, (7.6)

as the dry matrix shear muchlus.

The other two experiments are described in the following sections

7.1.1 Jacketed compressibility test

In the second deal experiment, the material is enclosed in a thin, imperimedule, flexible packet and then subjected to an external hydrostatic pressure μ . The pressure of the fluid inside the packet remains constant, because the interior of the jarket is exposed to the atmosphere by a tube (see Figure 7.1). The pare pressure remains essentially constant and $\sigma I = 0$. From equations (7.1) and (7.5), we obtain

$$p = 2 \log_{10} e C d_{C}$$
(77)



Figure 7.3.1 Paraes material is reclosed in a theorem measure packet and their subjected to accept optimization pressure p. The pressure of the block ensule the packet or mass constant, here is e the mode of the packet is expressive the attacophete by a tilth of stack stresser task.

and

$$0 = U d_m + 2D d_f$$
, (7.8)

In this test, the entire pressure is transmitted to the frame. Therefore,

$$K_{in} = -\mu/n_{in}, \quad (7.9)$$

where K_{ν} is the hulk modulus of the frame, also called the drained modulus. Combining equations (7.7), (7.8) and (7.9), we obtain

$$2A = \frac{C^2}{2D} = K_m$$
 (7.10)

7.1.2 Unjacketed compressibility test

In the third ideal experiment, the sample is numersed in the saturating limit to which a pressure p_f is applied. The pressure acts both on the frame part 1 = ϕ , and the finid part ϕ of the surfaces of the sample (see Figure 7.2).

Therefore, from equations (7.4) and (7.5)

$$(1 - \phi)p_f = 2(1d_m + Cu_f)$$
 (7.11)

and

$$apr = Co_m + 2Dd_G$$
(7.12)

In this experiment, the porosity does not change, since the deformation implies a change of scale. In this case,

$$K_s = p_f \partial_{\mu s}$$
 and $K_f = p_f \partial_{f s}$ (7.13)

where Λ_{∞} is the balk modulus of the closur solid from which the frame is made, and K_f is the balk modulus of the 0_{10} . Since the solid frame is compressed from the mode also

Figure 7.2: Configuration of the unjacketed experiment. The sample is immersed in a saturating fluid to which a pressure p_f is applied. The pressure acts both on the frame part and the fluid part.

in contrast to the jacketed experiment, the involved elastic modulus is that of the solid material.

Combining equations (7.11), (7.12) and (7.13), we get

$$1 - \phi = \frac{2A}{K_s} + \frac{C}{K_f}$$
(7.14)

and

$$\phi = \frac{2D}{K_f} + \frac{C}{K_s}.$$
 (7.15)

Solving equations (7.10), (7.14) and (7.15), we obtain

$$K \equiv 2A = \frac{\left(1 - \phi\right) \left(1 - \phi - \frac{K_m}{K_s}\right) K_s + \phi \frac{K_s}{K_f} K_m}{1 - \phi - K_m/K_s + \phi K_s/K_f} = P - \frac{4}{3}N,$$
(7.16)

$$C = \frac{\left(1 - \phi - \frac{K_m}{K_s}\right)\phi K_s}{1 - \phi - K_m/K_s + \phi K_s/K_f} = Q,$$
 (7.17)

and

$$2D = \frac{\phi^2 K_s}{1 - \phi - K_m/K_s + \phi K_s/K_f} = R, \quad (7.18)$$

where P, N, Q, and R is Biot's notation¹ (Biot, 1956a; Biot and Willis, 1957).

We may recast the stress-strain relations (7.4) and (7.5) as

$$\sigma_{ij}^{(m)} = 2\mu_m d_{ij}^{(m)} + (K\vartheta_m + C\vartheta_f)\delta_{ij},$$
 (7.19)

$$\sigma^{(f)} = C \vartheta_m + R \vartheta_f. \qquad (7.20)$$



¹Readers should not confuse Q with the quality factor defined in previous chapters.

In deriving the preceding experience, we have assumed that the material of the frame is homogeneous. There are cases where the grains are concented with a conterial of different properties, or where two different materials form two interpreterating rack frames. On the basis of Biot's theory, these cases can be treated with different approaches (Brown and Korzhega, 1975, Berryman and Milton, 1991, Gareviel, and Carchate, 2009, Carcion , Gareviel, and Cavallini, 2000;

7.2 The concept of effective stress

The stress-strain relations (7.19) and (7.20) can be interpreted as a relation between incremental fields, where stress and strain are increments with respect to a reference stress and strain – the case of wave propagation – or, as relations between the absolute fields. The last interpretation is used to Castrate the concept of effective stress

Effective stress and effective pressure play at important role in rock physics. The use of this concept is notivated by the fact that pore pressure, p_j , and continue pressure, p_j , read to have opposite effects on the acoustic and transport properties of the rock. Thus, it is convenient to characterize these properties with a single pressure, the effective pressure p_i . Toroughi (1936) proposed $p_i = p_i - op_j$, but his experiments regarding the failure of geological materials, indicated that $p_i = p_i - p_j$. Let us analyze Biot's constitutive expansions to clumin the effective-stress law predicted by this theory.

The total stress is decomposed into an effective stress, which acts on the frame, and into a hydrostatic stress, which acts on the fluid. In order to find this relation, we need to recast the constitutive equation in terms of the total stress.

$$\sigma_{0} = \sigma_{1}^{(m)} + \sigma^{(\ell)} \delta_{0}, \qquad (7.21)$$

and the variation of fluid content

$$\lesssim 1$$
 div $[c_{III}^{(f)} = \mathbf{u}^{(f)})^{\dagger} = \phi(d_f - d_m),$ (7.22)

where $\mathbf{n} \in \operatorname{cod} \mathbf{n}^{-m}$ one the displacement vectors of the fluid and solid matrix, respectively, and we have assumed that α is constant in this derivation. This condition will be removed fater. The variation of fluid content is a measure of the amount of fluid that has flowed in and out of a given volume of porces medium.

First, note that the modulys K defined in equation (7.16) can be written as

$$K = K_m + M(\alpha + \alpha)^2$$
, (7.23)

where

$$M = \frac{K_{f}}{1 - \alpha - K_{m}/K_{f} - \alpha K_{f}/K_{f}}$$
(7.24)

and

$$s = 1 - \frac{K_{co}}{K_{c}}$$
, (7.25)

¹ Readers should not contase M and a with the complex in shifts and atomation factor defined in pressors of optics. Note the relation.

$$\frac{1}{M} = \frac{\alpha - \alpha}{K_c} + \frac{\alpha}{K_c},$$
(7.26)

Substituting equation (7.23) into equation (7.19) and using the expression of the deviatoric strain for the frame user equation (1.15)), yields

$$\begin{split} \sigma_{ij}^{(m)} &\sim 2\mu_m d_{ij}^{(m)} + |k_{im} d_m \delta_{ij}| + |M(\alpha - \alpha)|^2 a_i \rightarrow |C(\ell_j| \delta_{ij}) \\ &= c_{ij,i}^{(m)} c_{ij}^{(m)} + |M(\alpha - \beta)|^2 d_m + |C(\ell_j| \delta_{ij}) \end{split}$$
(7.27)

where

$$c_{\mu\nu}^{(\nu)} = \left(K_{\mu\nu} - \frac{2}{3}\rho_{\mu\nu}\right)\delta_{\mu}\delta_{\nu\nu} + \mu_{\nu}\left(\delta_{\mu}\delta_{\mu} + \delta_{\mu}\delta_{\mu}\right)$$
(7.28)

is the clastic tensor of the frame. The total stress is then obtained by substituting (7.27) into equation (7.21), using (7.1), that is,

$$\pi^2 = o_{11}$$
 (7.20)

equation (7.20), and the relations $\phi = \phi \cup C_c R$ and $M \cup R/\phi^2$. We obtain

$$\sigma_{ij} = \epsilon_{ij}^{(m)} \epsilon_{ij}^{(m)} = \alpha p_f \delta_{ij} + 2\mu_e d_{ij}^{(m)} + K_e (\theta_m \delta_{ij} - \alpha p_f \delta_{ij})$$
(7.30)

Furthermore using equations (7/29) and (7/22) and the relations

$$C \sim \alpha M(\alpha - \alpha), \quad R \sim M(\alpha'),$$
 (7.31)

equation (7.20° can be written as

$$p_f = M(\zeta - \alpha d_m),$$
 (7.32)

An alternative form of the total-stress components is obtained by substituting p_f into equation (7.30)

$$\sigma_{ij} = \beta p_{ij} d_{ij}^{(m)} + (K_m + \alpha^2 M) u_m \delta_{ij} = \delta (M_{\infty} \delta_{ij} - 2p_m d_{ij}^{(m)} + K_{\ell} \delta_m \delta_{\ell} - \alpha M_{\infty} \delta_{\ell j} - \ell_{\ell}^{(n)} 33)$$

where

$$K_{0} = K_{0} + \alpha^{2}M = \frac{K_{0} - K_{m} + \alpha K_{m} (K_{m} K_{f} - 1)}{1 - \alpha - K_{m}/K_{f} + \alpha K_{s}/K_{f}}$$
(7.34)

is a saturation for indicated (modulus, obtained for $\zeta = 0$, (closed system)). Equation (7.34) is known as Gassmann's optation (Gassmann, 1951), which as shown in Section 7.74 (equation (7.2864), gives the low-frequency bulk modulus as a function of the frame and constituent properties. The dry-rock modulus expressed in terms of Gassmann's modulus is

$$K_{\mu\nu} = \frac{(\alpha K_{\nu}/K_{T} + 1 - \alpha)K_{\nu} - K_{\nu}}{(K_{\nu}/K_{V} + K_{U}/K_{N} - 1 - \alpha)}$$
(7.35)

A generalization of Gassmann's equation for two porous frames are given by Berryman and Miltor (1991), Gurevich and Carcione (2000) and Carcione, Gurevich and Cavallini (2000). The case of a manifold is given in Carcione. If the Soutos and Rayazzob (2005). The effective-stress concept means that the response of the saturated porous medium is described by the response of the day proces variance with the applied stress replaced by the effective stress. Thus, we search for a modified stress $\sigma_{1,2}^{2}$ which satisfies

$$\sigma_{ij}^{i} = c_{ij}^{(m)} c_{ij}^{(m)}$$
, (7.36)

Comparison of equations (7.30) and (7.36) allows us to identify the Birt effortive stress.

$$\sigma_{ij}^{\prime} = \sigma_{ij} + \alpha g \eta \delta_{ij} \qquad (7.37)$$

The material constant α is defined in equation (7.25) this referred to be the "Bost effectivestress coefficient". It is the proportion of fluid pressure which will produce the same strains as the total stress.

7.2.1 Effective stress in seismic exploration

Hydrovarbon reservors are generally overpressured. This situation can, in principle, be characterized by seismic waves. To this end, the dependence of the P-wave and S-wave velocities on effective stress plays an important role. It is well known from laboratory experiments that the acoustic and transport properties of rocks generally depend on "effective pressure", a combination of pure and confining pressures.

Pore pressure" — in absolute terms — also known as formation pressure, is the fin situ pressure of the fluids in the pores. The pore pressure is equal to the "hydrostatic pressure" when the pore fluids only support the weight of the overlying pore fluids (mainly lower), he this case, there is communication from the reservoir to the surface — The "lithestate" at "contining pressure" is due to the weight of overlying sediments, including the pore fluids". A rock is said to be overpressured when its pore pressure is significantly greater than the hydrostatic pressure. The difference between contining pressure and pore pressure is called "Efferencial pressure".

Figure 7.3 shows a typical pressures depth relation, where the softment of the transition zone is overpressured. Various physical processes cause anomaliers pressures or an underground third. The must communications of overpressure care discipulibrium empartion and "cracking", i.e., cil to gas emversion.

Let us assume a reservoir at depth 1. The lithostatic pressure for an average sediment density p is equal to p = pg), where g is the acceleration of gravity. The hydrostatic pore pressure is approximately $p_R = p_R g$, where p_R is the density of water. As stated above, $p_R = p_R$ if there are no percondulate barriers between the reservoir and the surface or when the pressure explicit particles fast = high overheiden permeability. Taking the trace is equation (7.30), we get

$$\frac{1}{3}\sigma_{\mu} = K_{\mu\nu}\theta_{\nu} = \alpha\rho_{I}, \quad (7.38)$$

Identifying the left-hand side with norms the continuing pressure and $K_{in}\phi_{in}$ with means the effective pressure ρ_{in} we obtain

$$p_e = p = op_{ij}$$
 (7.39)

A couldy a took in the only office is subjected to a to a bydrowth a state of stress on general, the vectoral stress is greater than the horizontal stress and this struction address an sole type on effective reservoirs and this structure.



Figure 7.3: Typical pressure depth plot, when the different pressure defautness are direction

Terzaglu's equation (Terzaglu, 1925, 1933) is obtained for an incompressible solid material, $K_{\pm} \rightarrow \infty$. Then, from equation (7.25), $\alpha \rightarrow 0$, and the effective pressure, predicted by Baa's theory, is equal to the differential pressure.

Let us consider now an undramed test, that is, $\zeta = 0$. Then, the elimination of θ_m in equations (7.32) and (7.33) gives

$$\mu_f = B \rho_c$$
, (7.40)

where

$$B = \frac{\alpha M}{K_0} = (7.41)$$

is called the Skempton coefficient (Skempton, 1954). In this experiment, the fluid pressure depends linearly on the containing pressure. Measuring the Skempton coefficient allows us to calculate the two percelositecty constants α and M.

$$\phi = \frac{1}{B} \left(1 - \frac{K_{ii}}{K_{ii}} \right), \quad \text{and} \quad M = \frac{B^* K_{ii}^*}{K_{ii} - K_{ii}}$$
(7.42)

Actually, each acoustic or transport property of the medianic such as wave velocity and permeability has a different effective-stress coefficient. For metanice, Googi and Carlson (2006) show that the wave velocities depend on the effective pressure, which can be written as where

$$\mu = \mu = 0, \mu_1 \qquad (7.43)$$

where n_{\pm} the effective-stress coefficient, is a linear function of the differential pressure. This dependence of n_{\pm} versus differential pressure is in good agreement with the experimental values corresponding to the compressional vehicity obtained by Prasad and Manghaani (1997). It is shown in Section 7.2.2 that the effective stress coefficient for the periodity is 1.

Pore-volume balance

The case of disciplibrium compaction is that in which the sedimentation rate is subapid that the poor fatils do not have a chance to "recope". Balancing mass and volume fractions in the pore space yields the pore pressure, the saturations and the polosity versus time and depth of buria? Thermal effects are also taken into account. The pore pressure, together with the confining pressure, determines the effective pressure which, in turn, determines the day-rock moduli

For a constant sediment lurial rate [8, and a constant geothermal gradient] G, the temperature variation of a particular sediment volumer is

$$T = T_0 + G_{-1}$$
, $z = St_0$ (7.49)

where t is here the deposition time and T_{c} the surface temperature). Typical values of G range from 10 to 30° C/km, while S may range between 0.05 and 3 km, may (may) = million years) (Math. and Markenzie, 1990).

Vssuming only liquid hydrocarbox, and water in the porcespace

$$Ωp = Ωp + Ωp, \quad (7.45)$$

where Ω_{μ} is the pore volume and Ω_{μ} and Ω_{ν} are the volumes of Ω_{μ} hydrorarbun and water in the pore space, respectively. We have

$$d\Omega_{i}(p_{i}, T, M_{p}) = d\Omega_{a}(p_{f}, T, M_{a}) + d\Omega_{a}(p_{f}, T, M_{a}),$$
 (7.46)

where M_i and M_i are the masses of the hydrogarbur and water phases and M_p is the total mass in the pure space.

If no mass tof the hydrorathon of the water) leaves the pare space, cand there is no "phase" conversion.), then $dM_{\mu} \sim 0 \rightarrow dM_{\gamma} \sim dM_{\gamma}$ and we have

$$d\Theta_{p} = \left(\frac{\partial\Theta_{p}}{\partial\mu_{r}}\right) d\mu_{r} + \left(\frac{\partial\Theta_{r}}{\partial T}\right) dT$$
$$= \left(\frac{\partial\Theta_{r}}{\partial\mu_{r}} - \frac{\partial\Theta_{r}}{\partial\mu_{r}}\right) d\mu_{f} - \left(\frac{\partial\Theta_{r}}{\partial T} - \frac{\partial\Theta_{r}}{\partial T}\right) dT.$$
(7.17)

We define

$$C_p = -\frac{1}{\Omega_p} \frac{d\Omega_p}{dp_i}, \quad C_s = -\frac{1}{\Omega_s} \frac{d\Omega_s}{dp_f}, \quad C_s = -\frac{1}{\Omega_s} \frac{d\Omega_s}{dp_f}.$$
 (7.48)

the compressibilities for the pore space, hydrocarbon and water, and

$$\frac{\nu_{\mu}}{\Omega_{\mu}} \frac{1}{\partial T} \frac{\partial \Omega_{\mu}}{\partial T} = \frac{1}{\Omega_{\mu}} \frac{\partial \Omega_{\mu$$

[&]quot;Beaches should not confuse 7 with the kenturn seg-

the corresponding thermal-expansion coefficients. Let us assume that the compressibilities of hydrocorbons and water are independent of pressure and temperature. That this is the rase can be seen from the results given by Batzle and Wing (1992), in then Figures 5 and 13, where they show that the density is alreast a linear function of temperature and pressure. This means that the mentioned properties are approximately constant (see also then Figure 7, where the oil compressibility remains almost constant when going from low temperature and low pressure to legb temperature and high pressure). Moreover, let us assume that the next compressibility C_{μ} is independent of temperature but depends on pressure. We consider the following functional form for C_{μ} as a function of effective pressure:

$$C_p \simeq C_p^+ + \beta \exp(-p_0[p^*)).$$
 (7.50)

where $C_p^{(1)}$, if and p^* are coefficients obtained by fitting experimental data. Assume that at time l_p corresponding to depth 1,, the volume of rock behaves as a closed system. That is, if the rock is a shale, its permeability is extremely low, and if the rock is a sandstone, the permeability of the scaling foults and surrounding layers is sufficiently low so that the rate of pressure nurrows greatly exceeds the dissipation of pressure by flow. For spressure excess is measured relative to hydrostatic pressure.

Integration of (7,18) and (7,59) from $p_{f_2}(p_{e_1})$ to $p_2(p_{e_2})$ and T_e to $T_i + \Delta T_i$ where ΔT_i , $i = T_i$, yields

$$|\Omega_{\mu}(p_f, T) - \Omega_{\mu\nu} \exp(-C_{\nu}\Delta p_f + \alpha_{\mu}\Delta T)|,$$
 (7.51)

$$\Omega_{t}(p_{f}, I) = \Omega_{tr} \exp[-C_{t} \Delta p_{f} + \alpha_{r} \Delta I_{r}],$$
 (7.52)

and

$$\Omega_{\mu}(p_{1}|T) = \Omega_{\mu} \{ \exp[E(\Delta p_{1}) + \phi_{1} \Delta F_{1} \}$$
(7.53)

where (see equation (7.501))

$$F(\Delta p_f) = -C_p^{\infty} \Delta p_t + \beta p_t^{\alpha} [\exp(-p_t)p_t^{\alpha}) - \exp(-p_t)(p_t^{\alpha})],$$

 $\Delta p_i = p_i + p_0$ and $\Delta p_l = p_l - p_{fl}$.

Assuming a fracar dependence of the effective-stress coefficient, n_i versus the differential pressure, $p_0 = p_i = p_j$.

$$0 = a_0 - a_0 p_{20}$$
 (7.53)

where n₀ and n₁ are constant coefficients, the effective pressure can be written as

 $p_i = p_i - (n_i - n_i)p_i (p_j - n_i)p_j^2$ (7.55)

Using equation (7.47), the power volume at pure pressure η_{f} and temperature T is given by

$$\langle \Omega_{\mu}(\mu_{f}, T) \rangle = \Omega_{\mu\nu} \{ \exp[iE(\Delta\mu_{f}) + \alpha_{\mu}\Delta T_{\mu}] + \Omega_{\mu\nu} [\exp[i-C_{\mu}\Delta\mu_{f} + \alpha_{\nu}\Delta T_{\mu}] + \Omega_{\mu\nu} [\exp[-C_{\nu}\Delta\mu_{f} + \alpha_{\mu}\Delta T_{\mu}],$$

(7.56)

Since the initial saturations are

$$S_{\mu\nu} = \Omega_{\mu\nu} \langle \Omega_{\mu\nu} - S_{\mu\nu} - \Omega_{\mu\nu} \langle \Omega_{\mu\nu} - 1 - S_{\mu\nu}, \qquad (7.57)$$

equation (7.56) becomes

$$\exp[F(\Delta p_I) + \alpha_r \Delta T] = S_{rel} [\exp[-C_r \Delta p_I + \alpha_r \Delta T)]$$

$$c(1 - \beta_c) \exp[-C_c \Delta_{PI} + \alpha_a \Delta I_c],$$
 (7.58)

The solution of equation (7.58) gives the pore pressure p_{i} as a function of depth and deposition time *i*, with $\Delta T = I - I_{i} - G(z - z_{i}) - GS(t - t_{i})$ for a constant geothermal gradient and a constant solution burnal rate. The excess pure pressure is $p_{i} - p_{R}$

Acoustic properties

In order to obtain the acoustic properties, such as were velocity and attractation factor versus poor and confitting pressures. On day took bulk and rightly mobili K_{α} and μ_{α} should be evaluated as a function of the effective pressure. Then, an appropriate modellike Biot's theory, can be used to obtain the properties of the sammated portors medium. Those moduli can be obtained from informatory measurements in day samples. If σ_{is} and σ_{is} are the experimental dreaveck compressional and shear velocities, the moduli are approximately given by

$$K_{m}(p_{i}) = \left(1 - cop_{i}\left[r_{m,p_{i}}^{i}\right] - \frac{4}{3}\left(\zeta_{i}\left(p_{i}\right)\right], \quad p_{m}(p_{i}) = \left(1 - cop_{i}r_{m}^{i}(p_{i})\right), \quad (7.59)$$

These are rock anothli at almost zero pore pressure, i.e., the case when the balk modulus of the pure fluid is negligible compared with the frame hulk modulus, as, for example, air at more conditions.

The procedure is to fit the experimental data, say K_{m} , by functions of the form

$$K_m = K_m^{(i)} - ap_i - b \exp(a - p_i/p'),$$
 (7.69)

where $K_{n}^{(1)}$, n, b and p^{*} are fitting coefficients. Knowing the effective stress coefficients for K_{m} and p_{m} , it is possible to obtain the wave velocities for different combinations of the pore and confineing pressures, since the property should be constant for a given value of the effective pressure. This is achieved by simply replacing the confining pressure by the effective pressure (7.5) in repeations (7.60), where n corresponds either to K_{m} or to p_{m} . An example of the application of this approach can be finded in Carcina and Gaugi (2000a.h), where the effects of discipations compaction and oil to gas conversion on the seismic properties are investigated.

Use of high-frequency (laboratory) data to make predictions in the scismic – lowfrequency – band should be considered with contion. The fluid effects on wave velocity and attenuation depend on the frequency range. At low frequences, the fluid loss enough time to arbitive pressure republication (relaxed regime) and Gassmann's modulus properly describes the saturated bulk modulus. At high frequencies, the fluid contact relax and this state of matchasation induces pour pressure gradients. Consequently, the bulk and shear moduli are stiffer than at low frequencies (White (1975), Mukerji and Mayko, 1994). Dworkin: Mayko and Nur. 1995). This attentiation mechanism is discussed in Section 7.10

7.2.2 Analysis in terms of compressibilities

The fact that there are two independent volumes and that two independent pressures can be applied to a parage method implies for a different compressibilities. Let us denote Ω_m Ω_{i} and Ω_{i} as the solid, pore and bulk volumes, respectively. The porosity and the vold ratio are, respectively, defined by

$$\alpha = \frac{\Omega_p}{\Omega_r}$$
, and $\epsilon = \frac{\Omega_p}{\Omega_m} = \frac{\Omega_p}{\Omega_p} = \frac{\Omega_r}{1 + \alpha^2}$ (7.61)

The two pressures are the containg and the pore-fluid pressures. Following the work of Zummerman (1995, μ =3), the congressibilities are defined as

$$C_{n} = -\frac{1}{\phi_{l-dy_{n}}} \frac{\partial \Omega_{n}}{\partial r} \Big|_{q_{l}} = \frac{1}{K_{n}}, \qquad (7.62)$$

$$C_{2i} = \frac{1}{\Omega_{i}} \frac{\partial \Omega_{i}}{\partial \rho_{j}} \Big|_{p}$$
(7.63)

$$C_{\mu} = -\frac{1}{\Omega_{\mu}} \frac{\partial \Omega_{\mu}}{\partial \mu_{\nu}} \Big|_{\mu_{\nu}}, \qquad (7.64)$$

acual

$$C_{bp} = \frac{1}{\Omega_p} \frac{\partial \Omega_p}{\partial p_f} \Big|_{p_f} = C_{p_f}$$
(7.65)

The first compressibility can be obtained with the jacketed compressibility test described in Section 7.1.1, and the last compressibility is the pore compressibility (see below). The different signs supply that all the compressibilities will be positive, because positive confiring pressures decrease the volumes Ω_{i} and Ω_{b} , while positive pore pressures increase these volumes. The other intrinsic compressibilities are the solid material and fluid compressibilities.

$$C_{t} = K_{s}^{-1}$$
, and $C_{t} = K_{T}^{-1}$. (7.66)

respectively

In order to obtain the relationships between the compressibilities, we need to perform a series of ideal experiments consisting of different pressure changes (dp_i, dp_j) . The bulk volume changes due to the stress increment (0, dp) are equal to the differences between the volume changes resulting from the stress increments (dp, dp) and (dp, 0). The last of these experiments corresponds to that described by equation (7.13)₄, and the second to the jacketed experiment (Section 7.14). Since, in general, $d\Omega = \pm C\Omega dp$, with C bring the corresponding compressibility, we have

$$C_{sp} \Theta_s dp = -C_s \Theta_s dp = (-C_s \Theta_s dp) = (C_s - C_s) \Theta_s dp, \qquad (7.67)$$

rher

$$C_{0p} = C_0 = C_1$$
 (7.68)

Let us consider now the pote volume changes and the same stress decomposition as before. The stress increment $(d\mu, d\mu)$ generates a change of scale, implying that the change in pore volume is, in this case, given by $-C_s\Omega_\mu d\mu$. This can be interpreted as follows. The stranging produced by $(d\mu, d\mu)$ can be obtained by filling the pores with the solid material and applying a continuing stress $d\mu$. This, is million straining in the solid results in the same straining of the pore space, and the local chlatation is everywhere given by $-C_{\rm e}dp_{\rm e}$. We obtain

$$C_{\mu}\Omega_{\mu}d\rho = -C_{\nu}\Omega_{\mu}d\rho \rightarrow -C_{\mu}\Omega_{\mu}d\rho = (C_{\mu\nu} + C_{\nu})\Omega_{\mu}d\rho,$$
 (7.69)

which implies

$$|C_{ij} = C_{ij} - C_{ij}$$
 (7.70)

A third relation can be obtained by invoking Bétti-Rayleigh's terriptical theorem (Fung. (965), p = 5% in a linear cluster-solid, the work down for a set of forws in traj through the corresponding displacements produced by a second set of forws is equal to the work down by the second set of forws. Hence, if the two forces F_1 and F_2 act on an elastic body, the work down by F_1 acting upon the displacements (for the F_1 is equal to the work down by F_2 acting upon the displacements (for the F_1 is equal to the work down by F_1 acting upon the displacements (for the F_1 is equal to the work down by F_2 acting upon the displacements due to F_1 . Let F_1 and F_2 be the stress increments (dp.0) and (0, dp), respectively. Then, the first work is

$$W_{\ell} = -d\mu G_{\eta} \Omega d\mu \ell = -G_{\eta} \Omega d\mu \ell$$
, (7.74)

when the minus sign is due to the fact that the confining pressure decreases the hulk volume. The second work is

$$W_2 = dp(-C_p \Omega_p dp) = -C_p \Omega_p (dp)^2$$

$$(7.72)$$

Here the sign is positive, since the pole pressure tends to increase the pole volume. Applying BettsBayleigh's themeni and using equation (7.61), we get

$$C_{p} = \phi C_{p}$$
 (7.73)

see also Mayko and Mukerji, 1995). Equations (7.68), (7.70) and (7.73) allow us to express three compressibilities in terms of ϕ , K_{i} and K_{in} .

$$C_{\rm w} = \frac{1}{K_{\rm w}} - \frac{1}{K_{\star}}.$$
(7.74)

$$U_{\rm P} = \frac{1}{\alpha} \left(\frac{1}{K_{\rm m}} - \frac{1}{K_{\rm s}} \right), \qquad (7.75)$$

and

$$C_{fir} = \frac{1}{\alpha} \left(\frac{1}{K_{cr}} - \frac{1 + \alpha}{K_{cr}} \right), \qquad (7.76)$$

Let us now obtain Gassmann's undrained modulus in terms of the above compressibilities, In an undranoxl compression, the fluid is not free to move into or out of the pore space. Such a situation is relevant to round processes such as wave propagation. The bulk and pure strains can be expressed in terms of the compressibilities as

$$dv_{s} = -C_{k}d\mu_{l} + C_{k}d\mu_{l}, \qquad (7.77)$$

and

$$dq = -C_{ii}dp_{ij} + C_{ij}dp_{j} \qquad (7.78)$$

The signs guarantee that decreasing continueg pressure or increasing pore pressure imply positive strain increments. Now note that if the fluid completely fills the pore space and the mass of fluid is constant within the pore, we also have

$$dq_p = -C_p dp_q$$
(7.79)

The ratio $-de_{b}$ to d_{ib} is the undrained compressibility. Combining equations (7.77), (7.78) and (7.79), we get Gasumanti's compressibility.

$$C_{\lambda} = -\frac{d\epsilon_{\lambda}}{d\mu} = C_{\lambda} - \frac{C_{h}C_{\mu}}{C_{\mu\nu} + C_{I}} = \frac{1}{K_{R}}, \quad (7.80)$$

which the virtue of equations (7.6.3), and (7.74), 7.76) gives Gasannianu's compressibility, i.e., the inverse of Gasannianu's undrained manhulus obtained in Section 7.2 (equation (7.34)) by setting the variation of fluid content z_i equal to zero.

Different effective starss coefficients must be used for the variants properties of the medium (Zimmerman, 1991, p. 32-50). Let us consider, for instance, the provsity, Equation (7.61) implies $\ln \phi \sim \ln \Omega_{p} = \ln \Omega_{N}$. Differentiating, we obtain

$$\frac{d\alpha}{\alpha} = \frac{d\Omega_{t}}{\Omega_{p}} - \frac{d\Omega_{b}}{\Omega_{b}} = d\epsilon_{p} - d\alpha, \qquad (5.81)$$

Using oppations (7.77) and (7.78), we have

$$\frac{d\phi}{\phi} = (C_8 - C_{p1}d) = (C_{p2} - C_{p2})dp_1,$$
 (7.82)

and substituting equations (7.75), (7.75) and (7.76), we note that the charace in the pressity becomes

$$d\sigma = -\left(\frac{1-\alpha}{K_m} - \frac{1}{K_s}\right)d(\mu - \mu_l) = -\left(\frac{\alpha - \alpha}{K_m}\right)d(\mu - \mu_l)$$
(7.83)

where α is given by equation (7.2a). The incremental porosity depends on the differential porssing, since its effective stress coefficient is equal to i. The term in parentheses is always positive, and this implies that the porosity is a decreasing function of the differential pressure.

It is found experimentally that the effective-stress coefficients depend on pore and confining pressures. In principle, this may invalidate the whole concept of effective stress. However, if one assures that the differentials de_k and de_k in equations (7.77) and (7.78) are exact differentials, the effective-stress coefficient for bulk deformations can be shown in depend on the differential pressure $p_f = p_f - p_f$. The demonstration follows. If a differential is exact, the Euler condition states that the two mixed partial derivatives are equal, that is

$$\frac{\partial^2 \epsilon_h}{\partial p_I \partial p_i} = \frac{\partial}{\partial p_I} \left(\frac{\partial \epsilon_h}{\partial p_i} \right) = -\frac{\partial C_h}{\partial p_f}.$$
(7.84)

$$\frac{\partial^2 \phi}{\partial p_i \partial p_f} \approx \frac{\partial}{\partial p_i} \left(\frac{\partial \phi}{\partial p_f} \right) \approx \frac{\partial C_{\theta p_i + 1}}{\partial p_i} \frac{\partial^2 C_{\mathbf{k}} - C_{\mathbf{k}}}{\partial p_i} \approx \frac{\partial C_{\mathbf{k}}}{\partial p_i}, \qquad (7.85)$$

where equations (7.62), (7.63) and (7.68) have been used. Then

$$\frac{\partial C_{\kappa}}{\partial \mu} = \frac{\partial C_{\kappa}}{\partial p}$$
, (7.80)

where we assumed that the solid compressibility C_i is independent of pressure. Similarly the application of the Euler condition to ϵ_i yields

$$\frac{\partial C_{\mu}}{\partial \mu_{\ell}} = -\frac{\partial C_{\mu}}{\partial \mu_{\ell}},$$
 (7.87)

The form of the differential equations (7.86) and (7.87) for $C_{n}(p_{1}, p_{2})$ and $C_{n}(p_{2}, p_{3})$ implies that these compressibilities depend on the pressures only through the differential pressure.

 $C_{0} = C_{\infty}(\mu_{0})$ and $C_{\mu} = C_{\mu}(\mu_{0})$. (7.88)

Effective stress coefficients for transport properties are obtained by Berryman (1992), for instance, the permeability effective-stress coefficient is found to be less ther one, in contrast with experimental data for clay-rich sinelstones. This is due to the assumption of metoscopic homogeneity. Using a two-constituent periors medium, the theory predicts, in some cases, a coefficient greater than one.

7.3 Anisotropic media. Strain energy and stress-strain relations

Proofs media are anisotropic due to keelding, compaction and the presence of aligned inferogracks and fractures. In particular, in the exploration of oil and gas reservoirs, it is important to estimate the preferential directions of fluid flow. These are closely related to the permeability of the medium, and consequently to the geometrical characteristics of the skeleton. In other words, an anisotropic skeleton implies that permeability is anisotropic and vice versa. For instance, shales are naturally bredded and possess intrinsic anisotropy at the microscopic level. Similarly, compaction and the presence of inferogracks and fractures make the skeleton anisotropic. Hence, it is reasonable to begin with the theory for the transversely isotropic case, which east be a good approximation, for saturated compacted solutions. The extension to orthorizonlar and lower-symmetry media is straightforward.

We assume that the solid constituent is isotropic and that the anisotropy is solely the to the arrangement of the gravity i.e., the skeleton is anisotropy i. Generalizing the single-phase strain energy (1.8), we can write

$$2V = c_{11}(\epsilon_{11}^2 + \epsilon_{12}^2) + c_{13}\epsilon_{13}^2 + 2(\epsilon_{11} - 2\epsilon_{23})\epsilon_{11} + 2\epsilon_{13}(\epsilon_{11} + \epsilon_{23})\epsilon_{31} + c_{31}(\epsilon_{31}^2 + \epsilon_{32})\epsilon_{31} + c_{32}(\epsilon_{31}^2 + \epsilon_{32})\epsilon_{31} + 2Fd_f^2$$

$$(7.89)$$

(Biot, 1955), where the strains ϕ_{ij} correspond to the frame (The superscript (0)) has been constructed for charity). The last three terms are the coupling and the fluid terms, written in such a way as to explort the invariance under rotations about the (says The stress-strain relations can be derived from equations (7.3). We get

$$\begin{split} \sigma_{22}^{(m)} &= (e_1e_1^{(m)} + 1e_1) - 2e_2(e_2^{(m)} + e_1)e_{13}^{(m)} + C(d_f) \\ \sigma_{22}^{(m)} &= (e_1e_{22}^{(m)} + 1e_1) - 2e_2(e_1^{(m)} + e_1)e_{14}^{(m)} + C(d_f) \\ \sigma_{10}^{(m)} &= (e_1e_{10}^{(m)} + e_1)(e_{11}^{(m)} + e_{12}^{(m)}) + C_1d_f \\ \sigma_{11}^{(m)} &= 2e_1e_{12}^{(m)} \\ \sigma_{12}^{(m)} &= e_{22}^{(m)} + e_{22}^{(m)} + C_1e_{12}^{(m)} + F_1e_f. \end{split}$$

In order to obtain the clasticity coefficients in terms of known properties, we require eight experiments, since there are eight independent coefficients. Let us first recast the stress strain relations in terms of the variation of fluid content, the total stress and the fluid pressure. Use of equation (7.22) explice $n_f = n_{tot} - \sqrt{n}$ and equation (7.90), can be expressed in analogy with (7.32), as

$$p_f = M'(z - \alpha_0 r_0^{(r)}),$$
 (7.91)

where

$$\alpha = \begin{pmatrix} \alpha_1 & 0 & 0 \\ 0 & \alpha & 0 \\ 0 & 0 & \alpha_1 \end{pmatrix}$$
(7.92)

with

$$a_1 = \alpha \left(1 + \frac{C_1}{F}\right), \quad a_1 = \alpha \left(1 + \frac{C_3}{F}\right) \tag{7.93}$$

and

$$M^{0} = T/\phi^{0}$$
, (7.94)

Using the shartened matrix notating, we alternatively write?

$$\alpha = (\alpha_1, \alpha_2, 0, 0, 0)$$
, (7.95)

where the components are denoted by σ_{12}

Let us consider equation (7.90), and compute the total stress according to equation (7.91) to obtain a finer similar to equation (7.30). Using equation (7.29), $d_f = d_m - \zeta/\omega_c$ and equations (7.91) and (7.93), we obtain

$$\sigma_1 = (\epsilon_1 \epsilon_2^{(m)} + 1\epsilon_1 - 2\epsilon_0)(\epsilon_2^{(m)} + \epsilon_1 \epsilon_{d_1}^{(m)} + C)d_m - \frac{C}{\alpha}\alpha_0\epsilon_{d_1}^{(m)} - (\epsilon_1 q_1) = -(7.96)$$

Rearranging terms, we rewrite

$$\sigma_{11} = (c_{11} + C_{12} - C_{10})\phi^{\mu}c_{11}^{\mu} + (c_{12} + C_{12} - C_{10})\phi^{\mu}c_{22}^{\mu}$$

$$= c_{13} + C_{12} - C_{10}(\phi)\phi^{\mu}c_{33}^{\mu} - \phi_{10}p_{11}, \qquad (7.97)$$

Researching Strict configuration with the attendation visition

where $c_{2} = c_{11} - 2c_{22}$. From (7.90), we obtain a using the shortened matrix notation (see equations (1.20) and (1.27)) — on equation of the form

$$a_I = c_{II}^{\infty} e_I^{\infty} - \alpha_I p_I, \qquad (7.98)$$

where

$$\begin{aligned} c_{11}^{(n)} &= c_{22}^{(n)} = c_{1} + C \left(1 - \alpha_{1} / \alpha \right) \\ c_{11}^{(n)} &= c_{11} + C_{1} \left(1 - \alpha_{1} / \alpha \right) \\ c_{11}^{(n)} &= c_{12} + C_{1} \left(1 - \alpha_{1} / \alpha \right) + c_{13} + C_{2} \left(1 - \alpha_{1} / \alpha \right) \\ c_{11}^{(n)} &= c_{13} + C_{1} \left(1 - \alpha_{12} / \alpha \right) \\ c_{12}^{(n)} &= c_{13} + C_{1} \left(1 - \alpha_{12} / \alpha \right) \\ c_{13}^{(n)} &= c_{13} + c_{13} \\ c_{13}^{(n)} &= c_{23} + c_{2$$

For a dramed condition (jacketed test), moduli py is zero, we have

$$\sigma_t = c_{t1}^{(m)} c_{t1}^{(m)}$$
(7.100)

Thus, the coefficients $e_{ij}^{(n)}$ are identified as the components of the dramed frame. Since there are five dry-rock moduli, five experiments are required to measure these moduli.

There other experiments are required to obtain α_1, α_2 and M'. These experiments are injucketed tests, where measurements in the plane of isotropy and in the axis of symmetry are performed. As in the isotropic case, the unjucketed compression test requires $\sigma_{ij}^{(m)}$.

 $(1 - \phi)p_f s_g$, and $\sigma^{(1)} = -\phi_{Pf}$ (see equations (7.11) and (7.12)). Then, from equation (7.21), the total stress is $\sigma_{(1)} = -p_f \delta_{Qr}$. The first three components of equation (7.98) become

$$\begin{aligned} \eta_1(\alpha) &= 1 := r_1^{(\alpha)} e_1^{(\alpha)} + r_2^{(\alpha)} r_2^{(\alpha)} + r_3^{(\alpha)} r_1^{(\alpha)} \\ \eta_1(\alpha) &= 1 := r_2^{(\alpha)} e_1^{(\alpha)} + r_2^{(\alpha)} r_2^{(\alpha)} + r_3^{(\alpha)} r_1^{(\alpha)} \end{aligned}$$

$$\eta_1(\alpha) &= 1 := r_3 (e_1^{(\alpha)} + e_2^{(\alpha)} + r_3 r_3^{(\alpha)} + r_3 r_3^{(\alpha)} \end{aligned}$$
(7.101)

Because the loading corresponds to a change of scale for the porces medium, the resulting strain components are related to the bulk modulus of the solid by

$$e_{0}^{(m)} = -\frac{P_{0}^{(m)}g}{3K_{0}}.$$
(7.102)

assuming that the solid material is isotropic. Then, the effective-stress coefficients are given by

$$\begin{aligned} \alpha_{1} &= 1 - (e_{11}^{(m)} + e_{12}^{(m)} + e_{11}^{(m)})_{\ell}(3K_{\ell}) \\ \alpha_{3} &= 1 - (2e_{11}^{(m)} + e_{11}^{(m)})_{\ell}(3K_{\ell}), \end{aligned}$$
(7.103)

Similar expressions for α_1 and α_2 are given by Thompson and Willis (1991) in terms of Skempton coefficients (Skempton, 1954)

The last unjarketed test involves equation (7.91), and is expressed as

$$p_1 = M_{s,0}^{t} + \alpha_0 p_{11}^{t} + c_0^{t} + \alpha_0 r_{11}^{t}$$
, (7.104)

with

$$\zeta \le \phi(\theta_f - \theta_m) \le \phi\left(\frac{p_f - p_f}{K_s - K_f}\right) \le \phi_f\left(\frac{1 - 1}{K_s - K_f}\right), \qquad (7.105)$$

according to equation (7.13). Substituting equations (7.102) and (7.105) into equation (7.104), we obtain

$$M' \geq K_{\gamma} \left[\left(1 - \frac{K_{\gamma}}{K_{\gamma}} \right) - \alpha \left(1 - \frac{K_{\gamma}}{K_{f}} \right) \right] \quad . \tag{7.106}$$

where

$$K^{*} = c_{n,n}^{(m)} = \frac{1}{6} \left[2c_{n}^{(m)} + 2c_{n}^{(m)} + 3c_{n}^{(m)} + c_{n}^{(m)} \right]$$
(7.107)

is the generalized drained bulk mealulus. Expression (7,106) is similar to (7,24), with K_c replaced by K^* .

The coefficients of the strain-energy density (7.89) can now be derived from explations (7.99). Using opportion (7.93) and (7.94), we obtain

$$\begin{aligned} c &= c_1^{(n)} = (\alpha - \beta)^2 M' \\ c_2 &= c_{12}^{(n)} + (\alpha - \alpha)^2 M' \\ c_3 &= c_{13}^{(n)} + (\alpha - \alpha)(\alpha_s - \beta)M' \\ c_3 &= c_{13}^{(n)} + (\alpha_b - \alpha)^2 M' \\ c_4 &= c_{13}^{(n)} = (c + -c_{12})/2 \\ C &= \alpha(\alpha - \alpha)M' \\ C_5 &= \alpha(\alpha_s - \alpha)M' \\ T &= \alpha^2 M' \end{aligned}$$
(7.108)

Furthermore, from equation (7.91 ($p_f = M^0)_s = \alpha_P \frac{\pi}{t}$), equation (7.98) becomes

$$\sigma_{t} = c_{t,t}^{*} c_{t,t}^{*} = M^{2} c_{t,t}^{*}$$
(7.109)

where

$$c_{II}^{\mu} \rightarrow c_{II}^{\prime \mu} + M^{\prime} \alpha_{I} \alpha_{I}$$
(7.110)

are the components of the indimined-modulus matrix \mathbf{C}^* , obtained for $\zeta = 0$, which is the optivalent of Gassmann's equation (7.34) (for instance, $\delta_1^0 = |\mathbf{b}|^2/4$ is equivalent to K_{01}^0 .

As in equation (7.37), the effective stress can be expressed as

$$\sigma_t^* = \sigma_1 + \alpha_1 p_0 \qquad (7.111)$$

Fulike the isotropic case, an increase in pure pressure also induces shear stresses

Scompton coefficients are obtained in an undraturd test. Let us denote the undratuedcompliance matrix by

$$S^{n} = (C^{n})^{-1}$$
, (7.112)

and its components by s_{11} . When $\zeta = 0$, equation (7.109) gives

$$e_I^{(m)} = s_{II} \sigma_I^{(m)}, \qquad (7.113)$$

Since, from equation (7.91), $p_1 = -M' \alpha_1 e_1^{(m)}$, we obtain

$$p_f = -B_f \sigma_f$$
, (7.114)

where

$$B_0 = M' s_{10}^* \alpha_0$$
 (7.115)

are the components of Skempton's 6 \times 1 array. Unlike the isotropic case (see equation (7.10)), pure pressure can be generated by shear as well as normal stresses. The Skempton array for (racis)etsely isotropic media is

$$\mathbf{b} = (B_0, B_0, R, [0|0|0])$$
 (7.116)

where:

7.3.1 Effective-stress law for anisotropic media

We follow Carroll's demonstration (Carroll, 1979) to obtain the effective-stress law for general (anisotropic porous media). The stress-strain relation for a divergence medium, obtained from equation (7.98) by setting $p_{T} = 0$, is

$$\sigma_{ij} = \epsilon_{ijk}^{00} \epsilon_{kl}^{(i)}, \quad \epsilon_{ij}^{(i)} = s_{ijl}^{(i)} \sigma_{kl}, \qquad 7.1180$$

where s_{int} denotes the compliance tensor satisfying

$$c_{0,\alpha}^{(m)} \gamma_{0,\alpha}^{(m)} \approx \frac{1}{2} (\phi_{\alpha} \phi_{\mu} + c_{\alpha} \phi_{\mu}), \qquad (7.119)$$



Figure 7.4. Surgious protons material, where S_{i} and S_{j} are the outer boundaries of the sample and the put is boundaries, respectively.

Consider may a representative sample of a saturated purious medium (Figure 7.4). It is brended by the matri surface S_{μ} and by the inner surface S_{μ} (pote houndaries). Let us consider the loading

$$t_i = \sigma_{ij} \sigma_{j}$$
 on S_i and $t_i = p_j \sigma_i$ on S_p . (7.120)

254

where α_i are the components of a unit vector perpendicular to the respective bounding surfaces $^{n-1}$. This loading can be treated as a superposition of two separate leadings

$$t_i = -p_f u_i$$
 or, S_i , and $t_i = -p_f u_i$ or S_p (7.121)

aral

$$l_i = a_{ii} n_i + p_I n, \text{ on } S_{ii} \quad \text{and} \quad l_i = 0 \text{ on, } S_i, \qquad (7.122)$$

The first loading gives rise to a hydrostatic pressure p_j in the solid material π it corresponds to a change of scale for the parons medium (see Figure 7.2).

Fur resulting strain is related to the compliance tensor of the solid s_{abc}^{\prime}

$$r_T^3 = -p_T r_{0TS}^3$$
(7.124)

The second loading corresponds to the jacketed experiment (see Figure 7.1). It is related to the compliance tensor of the day took, since $p_f = 0$.

$$e_{0_{1}}^{(2)} = s_{0_{2}0}^{(0)}(\phi_{11} - \mu_{f}\delta_{kl}),$$
 (7.124)

The total strain is then given by

$$\epsilon_{ij}^{(1)} = \epsilon_{ij}^{(1)} + \epsilon_{ij}^{(2)} + \epsilon_{ij}^{(2)} \sigma_{ij} + p_j (\gamma_{ijk}^{(2)} - \gamma_{ijk}^{(2)}).$$
 (7.125)

The offictive-stress law is obtained by substituting equation (7.125) into equation (7.36), and written as

$$\sigma_{i_1}^{x_1} = c_{i_1 i_2}^{x_1} [\gamma_{0 i_2 i_3}^{x_3} \sigma_{m_1} + p_1 (\gamma_{k l m_2 i_3}^{x_3} - \gamma_{0 m_2 i_3}^{x_1})], \qquad (7.126)$$

o. using (7/119).

$$\delta_{\mu} = \sigma_{\mu\nu} + p_f (\delta_{\mu\nu} - \delta_{\mu\rho}^{(\mu)} \delta_{\mu\nu\mu\nu}) = \sigma_{\mu\nu} + p_f \delta_{\mu\nu}$$
(7.127)

This registron provides the effective-stress coefficients of a in the anisotropic case.

$$\alpha_{ij} = \delta_{ij} - \epsilon^{bi}_{ijkl} s^{\dagger}_{ijmal}$$
(7.128)

If the solid material is isotropic.

$$\frac{20}{10} = -\frac{20}{3K_{\odot}}$$
, (7.129)

anal

$$\alpha_{0i} = \delta_{0i} - \epsilon_{ije\ell}^{400} K_{ij}, \qquad (7.136)$$

which is a generalization of equations (7.103).

7.3.2 Summary of equations

The stress-strait relations for a general amontopic medium are used also Chang. (1997).

As noted by Theorepoint and Well's (1980), such a hydronization of applace to the third points of Susume the first can section only by involutionatic stress. Cart (1) approach is, thus, strictly valid if non-coll the points noted set the output by pickets.

Pore pressure

$$p_f = M(\zeta - \alpha_f r_f^m),$$
 (7.131)

Total stress

$$\sigma_1 = c_{ff}^{(m)} c_{ff}^{(m)} = \alpha_1 p_f - (\sigma_0 - c_{gf}^{(m)} c_{ff}^{(m)} - \alpha_{ef} p_f), \qquad (7.142)$$

 $\sigma_{I} = c_{I,I}^{2} c_{I}^{2} = M c_{IS} - (\sigma_{0} - c_{0,I}^{2} c_{0}^{20} - M c_{0S}),$ (7.133)

Effective stress

$$\sigma_I' = \sigma_I - \sigma_I p_f, \qquad (7.144)$$

Skempton relation

 $p_f = -B_0 \sigma_{1i} - B_0 - M s_{iff}^2 \sigma_{1i}$ (7.135)

Undrained-modulus matrix

$$e_{II} = c_{IJ}^{(m)} - M \alpha_1 \alpha_3, \qquad (7.136)$$

$$M = \frac{K_s}{(1 - K_T/K_s) - \phi(1 - K_s/K_f)}$$
(7.437)

$$K^{*} = \frac{1}{9} \left[c_{(k)}^{(m)} + c_{(k)}^{(m)} + c_{(k)}^{(m)} + 2(c_{(k)}^{(m)} + c_{(k)}^{(m)} + c_{(k)}^{(m)}) \right],$$
(7.138)

$$\begin{aligned} \phi_{1} &= 1 - (c_{11}^{(m)} + c_{21}^{(m)} + c_{12}^{(m)})/(3K_{*}) \\ \phi_{2} &= 1 - (c_{21}^{(m)} + c_{22}^{(m)} + c_{23}^{(m)})/(3K_{*}) \\ \phi_{3} &= 1 - (c_{13}^{(m)} + c_{13}^{(m)} + c_{13}^{(m)})/(3K_{*}) \\ \phi_{3} &= -(c_{13}^{(m)} + c_{23}^{(m)} + c_{33}^{(m)})/(3K_{*}) \\ \phi_{5} &= -(c_{13}^{(m)} + c_{23}^{(m)} + c_{33}^{(m)})/(3K_{*}) \\ \phi_{5} &= -(c_{33}^{(m)} + c_{23}^{(m)} + c_{34}^{(m)})/(3K_{*}) \end{aligned}$$

(Note that J > I in the preceding equations)

7.3.3 Brown and Korringa's equations

An alternative derivation of the stress-strain relation for saturated parms anisotropic media is attributed to Brown and Kurringa (1975), who obtained expressions for the components of the multahed-compliance tensor.

$$s_{\alpha(1)}^{n} = s_{\alpha(0)}^{n} - \frac{1s_{\alpha(0)}^{n} - s_{\alpha(0)}^{n} (1s_{\alpha(0)}^{n} - s_{\alpha(0)}^{n})}{(s_{\alpha(0)}^{m} - C_{0}) + \phi(C_{f} - C_{0})},$$
(7.140)

256

7.4 Kinetic energy

where the superscripts $[n^{\prime\prime}]$, $[n^{\prime\prime}]$ and $[n^{\prime\prime}]$ denote undrained (saturated), matrix (dry reck), and solid (solid material of the frame), and C_{γ} and C_{γ} are the compressibilities of the limit and solid material, respectively (size equation (7.664)). Equations (7.136) and (7.140) are equivalent. They are the anisotropic versions of Gassmann's undrained modulus.

Transversely isotropic medium

In order to illustrate how to obtain the compliance components from the stiffness components and vice versa, we consider a transversely isotropic modum. The relation between the stiffness and compliance components are

$$\frac{s_{0}}{s_{1}} = \frac{s_{0}}{s_{1}} = \frac{1}{s_{1}} = \frac{s_{1}}{s_{1}} = \frac{s_{1}}{s_{1}} = \frac{s_{1}}{s_{1}} = \frac{s_{2}}{s_{1}} = \frac{1}{s_{1}} = 07130$$

where

(Auld, 1990a, §, 372). Moreover, $s_0 = 2(s_0 - s_0)$, Equations for converting C to S are obtained by interchangung all ϵ s and s s. The components of the corresponding undramed matrices transform in the same way. Let us consider the component $s_{1,0}^{*} = s_{10}^{*}$. Then, the different quantities in equation (7.140) are given by

$$s_{1+m}^{(m)} = s_{1}^{(m)} + s_{2}^{(m)} + s_{1n}^{(m)} + s_{1n}^{(m)} + 2s_{1n}^{(m)} + 2s_{1n}^{(m)} + s_{1+m}^{(m)} + s_{2nm}^{(m)} + \frac{C_{n}}{3},$$

$$s_{1m+m}^{(m)} = 2s_{1}^{(m)} + 2s_{2}^{(m)} + 4s_{1n}^{(m)} + s_{2n}^{(m)},$$
(7.132)

The value obtained for s_{ij} by substituting these quantities into equation (7.140) should coincide with the value obtained from equations (7.130) and (7.141). That is:

$$s_{11} = \frac{c_{11}^{2}}{c}, \quad c = c_{11}^{2} c_{12}^{2} + c_{13}^{2} c_{12}^{2} + 2c_{13}^{2} c_{13}^{2}$$
(7.15d)

7.4 Kinetic energy

Let us denote the macroscopic particle velocity by $v_i^{(p)} \to \partial_t u_i^{(p)}$, $p \mapsto m$ of f_i and the microscopic particle velocity by $w_i^{(p)}$. In the microscopic description, the kinetic energy is

$$T = \frac{1}{2} \int_{\Omega_1} \left[\rho_i w_i^{(m)} u_i^{(m)} d^{0,1} + \frac{1}{2} \int_{\Omega_2} \rho_j w_i^{(f)} w_i^{(f)} d^{0,1} \right]$$
(7.134)

where p_i and p_i are the densities of the fluid and solid material respectively, and Ω_{in} $(1 - \phi)\Omega_{in}/\Omega_{f} = \phi\Omega_{h}$ with Ω_{i} bring the volume of the elementary macroscopic and representative region of purons material. In the macroscopic discription, the kinetic energy cannot be obtained by the summation of two terms, since the involved particle velocities are not the time unicroscopic by chorines, but average velocities. We postulate a quadratic form with a coupling term, namely,

$$F = \frac{1}{2} \Theta_{k} [\rho_{11} v_{1}^{(0)} | v_{1}^{(0)} + 2\rho ([\mu_{1}^{(0)}] v_{1}^{(T)} + \rho_{12} ([\beta^{(T)} v_{1}^{(T)}])]$$
(7.135)

The solid reasonal is restroped. Note that a the oscillappenesses $s_{data} \in C_{2} = 1/K_{2}$ or deviation $90k_{12}$ This hypothesis assumes staristical isotropy. (In the arrisotropic case, terms of the form $c_i^{(n)} \in \mathcal{E}$, $i \neq j$ contradium to the kinetic energy.)

We need to find expressions for the density coefficients as a function of the densities of the constituents and properties of the frame. Assuming that the density of the solid and find constituents are constant in region Ω_{c1} the kinetic energy [7,111] becomes

$$T = \frac{1}{2} \rho_s \Omega_n \langle w_i^{(r)} | w_i^{(m)} \rangle_n = \frac{1}{2} \rho_f \Omega_f \langle w_i^{(f)} | w_i^{(f)} | f_s \rangle$$
(7.146)

where $\langle - \rangle$ denotes the average over the region compiled by the respective constituent Equating the microscopic and macroscopic expressions for the kinetic energy, we obtain

$$\mu_{12} v_{1}^{(m)} v_{2}^{(m)} + 2\mu_{12} v_{1}^{(m)} v_{1}^{(f)} + \mu_{22} v_{1}^{(f)} v_{1}^{(f)} = (1 \cdots n) \mu_{1} (w_{1}^{(m)} w_{1}^{(m)}) v_{1} + o \mu_{1} (w_{1}^{(f)} w_{1}^{(f)}) \mu_{2} / (7.147)$$

The linear momenta of the frame and the fluid are

$$c_{\mu}^{(i)} = \frac{\partial T}{\partial c_{\mu}^{(0)}} = \Omega_{k}(\mu_{\mu}c_{\mu}^{(0)} + \mu_{\mu}c_{\mu}^{(0)}), \qquad (7.148)$$

and

$$\pi_{i}^{(f)} = \frac{\partial T}{\partial v_{i}^{(f)}} + \Omega_{ii} p_{ij} v_{i}^{(f)} + p_{ij} v_{i}^{(m)} t_{i}$$
 (7.149)

respectively. The inertial forces acting on the frame and on the fluid are the cate of the respective linear momentum. An inertial interaction exists between the two phases. If, for fastance, a sphere is moving in a fluid, the interaction creates an apparent increase in the mass of the sphere. In this case, the induced mass is p_{12} . When no relative motion between solid and fluid scenars, there is no interaction. The material rows as a whole $(r_1^{(m)} - r_1^{(d)})$ and the matroscopic velocity is identical to the microscopic velocity. In this case, we obtain the average density from equation (7.147) and write it as

$$\rho = (1 - \phi_1 \rho_t + \phi_{P_1} + \rho_{11} + 2\rho_{12} + \rho_{22},$$
 (7.159)

The linear momenta of the frame and the fluid, from equation (7.146), are

$$\varepsilon_1^{(n)} = \frac{\partial T}{\partial v_1^{(m)}} = \frac{\Omega_0 (1 - \phi) \eta_1 v_1^{(m)}}{\partial v_1^{(m)}},$$
 (7.151)

and

$$\frac{d}{dt} = \frac{\partial T}{\partial w_i^2} = \Omega_{eq} q_0 w_i^2$$
(7.152)

A comparison of equations (7.148) and (7.129) with (7.151) and (7.152) yields

$$\rho_{11} - \rho_{12} = (1 - \phi)\rho_0$$
, (7.153)

and

$$p_{22} = p_{12} \approx \phi p_f,$$
 (7.154)

Substituting p_{11} and p_{22} in terms of p_{12} into equation (7.147), we obtain the following expression for the induced mass:

$$\frac{\rho_{12}}{\rho_{12}} = \frac{(1 - \phi(\rho_s)(w_i^{(m)}w_i^{(m)})_{ij} + v_i^{(m)}v_i^{(m)}) + \phi\rho_f)(w_i^{(f)}w_i^{(f)}w_i^{(f)})_{f} + v_i^{(f)}v_i^{(f)})}{(v_i^{(f)} + v_i^{(m)})(v_i^{(f)} + v_i^{(m)})} = \frac{(7.155)}{(2.155)}$$

7.4 Kinetic energy

(Nelson, 1988). Thus, the induced mass is given by the difference between the mean square particle velocities and the square of the corresponding macroscopic particle velocities, weighted by the constituent densities. Since $0 \le \phi \le 1$, the induced mass is always negative.

Afternatively, teatranging terms in equation (7,155), we obtain

$$\rho_{i2} = (1 - \phi) \rho_{i} \left[\frac{\langle (u_i)^m - v_i^{(f)} | (u_i)^m - v_i^{(f)} \rangle \rangle_m}{\langle v_i^{(f)} - v_i^{(m)} \rangle \langle (v_i^{(f)} - v_i^{(m)} \rangle \rangle_m)} \right] \\ + \partial_t \eta \left[\frac{\langle (u_i)^f - v_i^{(m)} \rangle \langle (u_i)^f - v_i^{(m)} \rangle \rangle_f}{\langle v_i)^f - v_i^{(m)} \rangle \langle v_i^{(f)} - v_i^{(m)} \rangle } \right],$$
(7.156)

We now define the cortrosities

$$T_m \sim \frac{c(w^{(m)} - v_0^{(f)})(w^{(m)} - c_0^{(f)})_m}{(v_0^{(f)} - v_0^{(h)})(v^{(f)} - c_0^{(h)})}$$
(7.157)

and

$$\mathcal{T} = \frac{\langle (w_i^{(t)} - v_i^{(t)}) | w_i^{(t)} - v_i^{(t)} \rangle_T}{\langle v_i^{(t)} - v_i^{(t)} | w_i^{(t)} - v_i^{(t)} \rangle},$$
(7.158)

and the induced mass can be expressed as

$$\rho_{i2} = -(1 - \phi)\rho_{i}(T_{0} - 1) - \phi_{II}\rho(T - 1).$$
 (7.159)

The torthosity of the solul is the mean square deviating of the uncrustrapic field of the solid from the fluid mean field, normalized by the square of the relative field between the fluid and solid constituents. The preceding statement is also rule if "fluid" is substituted for "solid" measury instance, and vice vetsa, but a nearly rigid percets frame, the microscopic field is approximately equal to the macroscopic field, $T_{m} \approx 1$, and

$$p_{10} = -2n g_0 T = 10.$$
 (7.169)

which is the expression given by Biot (1956a). If the ratio $w_i^{(I)}/v_i^{(I)} = J_i L_i$ where *l* is the torthous path length between two points and *L* is the straight line distance between those points, the torthowity (7.158) is shappy

$$T \leq \begin{pmatrix} T \\ L \end{pmatrix}^2$$
, (7.161)

where we assumed that the frame is acarly rigid. This assumption implies that the tormosity is related to the square of the relative path length.

A simple expression for the fortussity can be obtained if we interpret ρ_{ij} as the effective density of the solid moving in the fluid, namely,

$$p_{0} = (1 - \phi)(p_{0} + \epsilon)p_{1},$$
 (7.162)

where c_{PP} is the induced mass due to the oscillations of the solid particles in the fluid. Using equations (7.153), (7.160), and (7.162), we obtain:

$$T = 1 + \left(\frac{1}{\phi} - 1\right) c, \qquad (7.163)$$

where r = 1/2 for spheres moving in a fluid (Berryman, 1980).

7.4.1 Anisotropic media

Let us consider two different approaches to obtain the kinetic energy in anisotropic media. In the first approach, the general form of the kinetic energy is assumed to be

$$F = \frac{1}{2} \Omega_8(q_0 v_0^{(m)} v_0^{(m)} + 2v_0 v_0^{(m)} v_0^{(d)} + t_0 v_0^{(d)} v_0^{(d)})$$
(7.164)

where $\mathbf{Q}(q_{ij})$, $\mathbf{R}(r_{ij})$ and $\mathbf{T}(r_{ij})$ are 3×3 mass matrices, with \mathbf{R} being the induced mass matrix. Let its assume that the three matrices can be diagonalized in the same coordinate system, so that

$$\begin{aligned} \mathbf{Q} &= \operatorname{diag}(q_1, q_2, q_3) \\ \mathbf{R} &= \operatorname{diag}(r_3, r_2, r_3) \\ \mathbf{T} &= \operatorname{diag}(t_1, t_2, t_3). \end{aligned}$$
(7.165)

We shall see the implications of this assumption later. The kinetic energy in the microscopic description is given by equations (7,111) or (7,146). Equating the microscopic and macroscopic expressions of the kinetic energy, we obtain

$$g_{\ell} e_{\ell}^{m} e_{\ell}^{m} + 2r_{\ell} e_{\ell}^{m} e_{\ell}^{\ell} + h_{\ell} e_{\ell}^{\ell} e_{\ell}^{\ell} = 1 - \phi (p_{\ell} \langle w_{\ell}^{m} | w_{\ell}^{m} | p_{m} + \phi p_{\ell} \langle w_{\ell}^{\ell} | w_{\ell}^{\ell} \rangle)_{f} = 17.1661$$

The linear momenta of the frame and the fight are

$$\tau_{\mu}^{\mu} = \frac{\partial T}{\partial r_{\mu}^{\mu}} = \Omega_{\mu} \eta + r_{\mu}^{\mu\nu} + r_{\mu\nu} r_{\mu}^{\mu\nu} , \qquad (7.167)$$

and

$$v_{a}{}^{f} = \frac{\partial T}{\partial v_{a}^{af}} = \Omega_{b}(r_{a} v_{a}{}^{f} + r_{a}) v_{a}{}^{m}$$
). (7.168)

where the submatry (r) means that there is no implicit summation. As in the estimate case, to compute the relation between the different mass coefficients, we assume no relative mation between the frame and the fluid and equate the momenta (7.167) and (7.168) to the momenta (7.151) and (7.152). This gives

$$g_{1} + \eta_{1} = (1 - i t) g_{1}$$
 (7.169)
 $t_{1} + t_{1} = i g_{1} t$

Eliminating q_0 and f_1 in optation (7.166), we see that

$$c_{0}(v_{1}^{(d)} - v_{1}^{(m)})^{d} = -(1 - s)\mu_{0}(\langle w_{1}^{(m)} w_{1}^{(m)} \rangle_{m} - v_{1}^{(m)} v_{1}^{(m)}) - sg(r)(w_{1}^{(d)} w_{1}^{(d)}) r - v_{1}^{(d)} v_{1}^{(d)}) - (7.170)$$

which is the equivalent anisotropic relation of operation (7,155). With the use of equations (7,166), the kinetic energy (7,161) becomes

$$\mathcal{T} = \frac{1}{2} \Omega_0 [0 - \alpha(p_0)]^m r_0^{m_0} - r_0 (r_0^{m_0} - r_0^{(f)})^2 + \alpha_0 q r_0^{(f)} r_0^{(f)}], \qquad (7.171)$$

Note that in the absence of relative motion, the average density (7.150) is obtained. The induced mass coefficients r_0 , r_1 and r_2 are used as fitting parameters, as the fortuosity T in the isotropic case.

7.4 Kinetic energy

het us define the displacement of the fluid relative to the solid frame

such that the variation of fluid content (7.22) is

$$\zeta = div w_{c}$$
 (7.173)

The told variable

$$\dot{\mathbf{w}} = \partial_t \mathbf{w} = o(\mathbf{v}^{(t)} - \mathbf{v}^{(t)}) \tag{7.174}$$

is usually called the fitzation velocity, which plays an important role in Darcy's law. In terms of vector we the kinetic energy (7,171) can be rewritten as

$$T = \frac{1}{2} \Omega_0 (\rho e_i^{(m)} v^{(m)} + 2\rho_f e_i^{(m)} \partial_t w_i + 0 \partial_t w_i \partial_t w_i), \qquad (7.175)$$

where

$$m_{\rm e} = (cgg - r_{\rm e})/\delta^2,$$
 (7.176)

The second approach assumes that the relative uncrovelocity field of the fluid relative to the frame can be expressed as

$$v_{0} \sim w_{0} \partial_{t} w_{0} = (\mathbf{v} \sim \mathbf{u} \cdot \hat{\mathbf{w}}),$$
 (7.177)

where marrox a depends on the pore geometry (Biot, 1962).

The knowle energy is

$$T = \frac{1}{2} O_0 (1 - \phi_1) (v_1^{(\mu)}) v_2^{(\mu)} + \frac{1}{2} i g \int_{D_0} (v_1^{(\mu)} + v_1) (v_1^{(\mu)} + v_2) d\Omega, \qquad (7.178)$$

where the integration is taken on the fluid volume. The volume integral is

$$\int_{\Omega_0} (v_i^{(m)} + v_i) (v_i^{(m)} + v_i) d\Omega \le \int_{\Omega_0} (v_i^{(m)} v_i^{(m)} + 2v_i^{(m)} v_i + v_i v_i) d\Omega.$$
(7.159)

We have

$$\int_{\Omega_{t}} \left(v_{i}^{(m)} v_{i}^{(m)} - 2v_{i}^{(m)} v_{i} + v_{i} v_{i} d\Omega \le \Omega_{\delta} (c v_{i}^{(m)} v_{i}^{(m)} - 2v_{i}^{(m)} \partial_{t} w_{i}) - \int_{\Omega_{t}} c_{i} v_{i} d\Omega, \quad (7.180)$$

From the relation (7.177), we obtain

$$p_f \int_{\Omega_f} e_i e_i d\Omega = m_{ij} \partial_i w_i \partial_j w_j, \qquad m_{ij} = \frac{p_f}{\Omega_h} \int_{\Omega_f} u_k a_{ki} d\Omega.$$
(7.181)

After the substitution of equations (7/180) and (7/181), the kinetic energy (7/178) becomes

$$T = \frac{1}{2} \Omega_0 (\rho w_i^{(n)}) e_i^{(m)} + 2 \rho_f v_i^{(n)} (\partial_t w_i - m_g) \partial_t w_i \partial_t w_j), \qquad (7.182)$$

Equations (7.175) and (7.182) are equivalent if

$$m_{0} = m \lambda_{0}, \qquad (7.183)$$

e)

$$a_{0i} = a_0 \delta_{0i}$$
 (7.184)

that is, if the three Cortesian components of the flaid motion are uncoupled, or a_{ij} , $i \neq j$ are small compared to the diagonal components. This is a strong restriction. Alternatively, we may consider an includentialic median and choose the coordinate axis to lie in the planes of symmetry — recall that such a medium has three mutually orthogonal planes of mirror symmetry. In this case, the diagonalization is performed in the macroscopic domain (Biot, 1962)

7.5 Dissipation potential

Dissipation in michanical models, emissiving of springs and dashparts, is described by the constitutive equation of the dashparts, which relates the stress with the first time derivative of the strain. The strain energy is stored in the springs and websipation potential accounts for the dashparts. In Biot's theory, attenuation is caused by the relative motion between the frame and the fluid. Thus, the dissipation potential is written in terms of the particle vehicities as

$$\Phi_{1i} = \frac{1}{2} b(x_i^{(t)} - x_i^{(t)}) (x_i^{(m)} - x_i^{(t)}), \qquad (7.485)$$

where b is a friction (softment). A potential formulation, such as equation (7.185), is only justified in the vicinity of thermodynamic equilibrium. It also assumes that the fluid flow is of the Phisenille type, i.e., low Reymids number and low frequencies.

The coefficient b is obtained by comparing the classical Darry's law with the equation of the force derived from the dissipation potential. The dissipation forces are derived from a potential Ψ_{0} as

$$F_t = -\frac{\partial \Phi_0}{\partial u_t}, \quad (7.386)$$

$$\frac{\partial \Phi_B}{\partial a_i^f} = \frac{\partial \Phi_B}{\partial r_i^f}$$
(7.187)

Then.

$$F_{t} = \frac{\partial \Psi_{tr}}{\partial u_{t}} - b \left(e^{2\theta} \right)$$
(7.188)

Darce's law (Darce, 1856; Conssy, 1995, p. 71) relates the filtration velocity of the fluid, $\alpha(e_{i}^{(f)} - e_{i}^{(m)})$ (see equation (7.1744) to the pressure gradient, $\partial_{i}q_{i}$ as

$$\phi(r_i^{(R)} - r_i^{(n)}) = \partial_i w = -\frac{\kappa}{\eta} \partial_i p_f, \qquad (7.189)$$

where *n* is the global permetability and *q* is the viscosity of the fluid 1. Since *F*, is a finer per unit volume of fluid material, $F_{1}(\tau + \gamma)\partial_{z}p_{z}$. Comparing equations (7.188) and (7.189), we obtain the expression of the friction coefficient, namely

$$b = \phi_{\mu}^{(R)}$$
(7.190)

"Note that permeability is defined by κ and the magnitude of the real wavenumber vector is denired by κ

262

7.5.1 Anisotropic media

The next general form of the dissipation potential in anisotropic medians

$$\Phi_{II} = \frac{1}{2} b_{A} (x_{I}^{A}) = (z_{I}^{A}) 0 (z_{I}^{AB} - x_{I}^{AI}), \qquad (7.191)$$

where b_{i_1} are the components of a symmetric friction matrix **b**. Onsager's symmetry relations ensure the symmetry of **b**, and a positive definite quadratic potential (Biet, 1954) dr Grow and Mazur, 1963, p. 35, Nye, 1985, p. 207).

The patential (7,191) can be written in terms of the relative fluid displacement (7,172) as

$$\Phi_D = \frac{1}{2} g(\boldsymbol{\kappa}^{-1})_{ij} \partial_i a_j \partial_j a_j$$
(7.192)

where

$$\kappa = \alpha^2 \eta b^{-1}$$
(7.193)

is the permeability matrix (see equation (7.189).

Darcy's his rakes the form

$$d_{\mathbf{r}}\mathbf{w} \in \frac{1}{\eta} \mathbf{w} \cdot \operatorname{grad}(p_{f}).$$
 (7.194)

For orthorhotable media, the friction matrix can be recast in diagonal form in terms of three principal friction coefficients b_0 and, hence

$$\Phi_{ff} = \frac{1}{2} b_t (v_t^{(0)} - v_t^{(f)}) (v_t^{(r)} - v_t^{(f)}), \quad b_t = \phi_t^{(f)} \frac{\theta}{\mu_t}$$
(7.195)

The dissipation forces are derived as

$$F = -\frac{\partial b_0}{\partial c_i^{(f)}} - b_i c_i^{(f)} - c_i^{(m)} \qquad (7.196)$$

In terms of the three principal permeability components x_i and the filtration velocity (i|1i|i), we have

$$\Phi_{II} = \frac{1}{2} \frac{g}{c_1} \partial_t w_1 \partial_t w_1 = \frac{1}{2} \frac{g}{\kappa_1} \dot{w}_1 w_2, \qquad (7.197)$$

7.6 Lagrange's equations and equation of motion

The equation of motion can be obtained from Hamilton's principle. The Lagrangian density of a conservative system is defined as

$$L = T + V$$
 (7.198)

The motion of a conservative system can be described by Lagrange's equation, which is basist on Hamilton's principle of least action (Acherbach, 1984, p=61). The method can be extended to non-conservative systems if the dissipation fraces can be derived

from a potential as in equation (7.186). Lagrange's equations, with the displacements as generalized coordinates, can be written as

$$\partial_t \left(\frac{\partial L}{\partial r_t^{(p)}} \right) + \partial_t \left[\frac{\partial L}{\partial (\partial_t u^{(p)})} \right] - \frac{\partial L}{\partial a_t^{(p)}} + \frac{\partial \Phi_D}{\partial r_t^{(p)}} = 0, \qquad (7.195)$$

where p = m for the frame and p = f for the fluid. These equations are equivalent to Biotic classical approach

$$\partial_t \left(\frac{\partial F}{\partial e_s^{(p)}} \right) + \frac{\partial \Phi_D}{\partial e_s^{(p)}} + g_0^{(p)},$$
 (7.2004)

where g¹ are the generalized elastic forces, given by

$$-q_i^{(p)} = -\partial_i \left[\frac{\partial L}{\partial (\partial_i u_i^{(p)})} \right] = \partial_i \left[\frac{\partial b}{\partial (\partial_i u_i^{(p)})} \right], \qquad (7.201)$$

because *k* clors not depends explicitly on $u_i^{(n)}(\partial k/\partial u_i^{(n)} = 0)$ (Bot. (956b). Note that from equations (1.2) and (7.3)

$$\frac{\partial V}{\partial (\partial_j u^{(n)})} = \frac{\partial V}{\partial e_{ij}^{(n)}} = \sigma_{ij}^{(n)}, \qquad (7.202)$$

and

$$q_0^{(m)} = \partial_\mu \sigma_\mu^{(m)} = \operatorname{div} \sigma^{(m)}$$
(7.203)

According to equations (7.148) (7.149), the generalized finear minimum a pre-unit volume acting on the frame and on the fluid are

$$z_{\mu}^{(\mu)} = \frac{\partial T}{\partial z_{\mu}^{(\mu)}} = p_{\mu} x_{\mu}^{(\mu)} + p_{\mu} x_{\nu}^{(f)},$$
 (7.204)

and

$$\pi_{i}^{(I)} = \frac{\partial T}{\partial r_{i}^{(I)}} = p_{i} r_{i}^{(I)} + p_{i} r_{i}^{(I)} , \qquad (7.205)$$

Then, the equation of motion, from (7,200), is

$$(\partial_t \pi^{(t)} + T)^{(t)} = \operatorname{div} \sigma^{(t)}$$
, (7.209)

where $T^{(p)} = -\partial \Psi_D / \partial x_i^{(p)}$ are the dissipation forces.

From the expression (7.185) for the dissipation potential, we have, for isotropic media,

$$\partial_t \sigma_{0}^{(n)} = p_{11} \partial_0^2 u_1^{(m)} + n_{11} \partial_0^2 u_1^{(T)} + u(v_1^{(m)} - v_1^{(T)})$$
(7.207)

and

$$\partial \partial_t p_f = p_0 \partial_t^2 a_s^{(m)} + p_0 \partial_t^2 a_s^{(f)} - b e e_s^{(f)} - e_s^{(f)} b_s \qquad (7.208)$$

where the sign of the friction terms are chosen to ensure attenuated propagating waves. Equations (7/207) and (7/208) hold for constant periodity (Bort, 1956) he

7.6.1 The viscodynamic operator

Addwe equations (7.207) and (7.208) and using equations (7.21) and (7.20), we obtain

$$\partial_t \sigma_{ti} = (p_{ti} + p_{ti}) \partial_t^t a_t^{(m)} + (p_{ti} + p_{ti}) \partial_t^t a_t^{(m)}$$

$$(7.209)$$

Using equations (7.153) and (7.154), substituting the relative fluid displacement (7.172) into equations (7.208) and (7.209), and considering equation (7.195), we obtain the low-frequency equations of motion

$$\partial_t \sigma_{ij} = \rho \partial_{\mu} a_i^{\mu\nu} + \rho_j \partial_{\nu}^{\mu} a_{\mu}$$
(7.210)

and

$$\partial_t p_f = p_f \partial_{\mu}^{\mu} u_i^{\mu\nu} + m \partial_{\mu}^{\mu} w_i + \frac{h}{\epsilon} \partial_{\mu} w_i,$$
 (7.211)

where g is the average density (7.150), g, are the components of vector w, and

$$m = p_{D_0} \phi^2 - p_1 T \phi_0$$
 (7.212)

according to equations (7.154) and (7.160). Equations (7.210) and (7.211) hold for whomogeneous perosity (Ba), 1992). The demonstration and the appropriate expression of the stram-energy density is obtained in Section 7.8.

Equation (7.211) can be rewritten as

$$\partial_t p_t = p_f \partial_0^2 n_t^{(m)} + V * \partial_t w_t$$
(7.2)(3)

where

$$Y(t) = m\partial_t d(t) + \frac{b}{c} d(t), \qquad (7.2) 3)$$

is the low-frequency viscolynamic operator, with 5 brine Dirac's function.

In order to investivate the frequency range of validity of the viscoityname operator (7.214) and find an approximate operator for the high-frequency range, we evaluate the fraction force per unit volume in, say, the *i*-direction for a simple pore geometry. According to equations (7.186) and (7.187), the fraction or dissipation force is given by

$$F_1^{eff} = -i\theta \Phi_{tf} (de_1^{eff} - b_1^{eff} - e_1^{eff}) - be_1,$$
 (7.215)

where $\alpha_1 = \alpha_1^2 + \alpha_2^2$, is the average mattoscopic velocity of the fluid relative to the frame. Then the friction coefficient is given by

$$b = T_{\pm}^{(f)} r_{\pm}$$
(7.216)

i.e. it is the friction force per unit macroscopic velocity. In this cod, we solve the problem of theid flow between two parallel boundaries (see Figure 7.5).

7.6.2 Fluid flow in a plane slit

We consider that the fluid motion is in the collection and that the boundaries are located at q = z a, where a plays the role of the page radius. The displacement only depends on


Figure 7.51 (wordens asional flow between papade, walk,

the variable g and we neglect persone gradients and velocity components normal to the boundaries. The shear stress in the llubit is

$$\sigma_{12}^{(f)} = \eta \partial x_{12}^{(f)} = \eta \partial_2 k_{12}^{(f)}$$
, (7.2)7)

where ψ^{ij} is the intercompt particle velocity of the fluid. The viscons force is the divergence of the show stress. Then, higher's equation of notion for the viscons fluid is

$$\partial (p_f + g\partial_t \partial_t a_s^2) = p_0 \partial a_s^2$$
, (7.218)

Defining the microscopic relative fluid velocity.

$$v_i \equiv \dot{w}_i^{(I)} + v_i^{(i)}$$
, (7.219)

where $\phi^{(\alpha)} = \partial_{\mu}\phi^{(\alpha)}$ is the macroscopic particle velocity of the solid, we have

$$-\partial_{\tau} \mu_{l} + \rho_{l} \partial_{\tau} r_{1}^{(\nu)} + \eta \partial_{z} \partial_{z} r_{1} = \rho_{l} \partial_{t} r_{1}, \qquad (7.2.9)$$

where we have neglected the term $\eta d_i \partial_{ij} f^{\prime\prime\prime}$. If we consider that

$$\partial_t p_f = p_f \partial_t e^{i\theta \phi} - p_f k$$
 (7.221)

is equivalent to an external volume force, equation (7.220) becomes

$$\partial_t r_i = v \partial_t \partial_t r_i + F_i$$
 $\nu = \eta_i \rho_f,$ (7.222)

where ν is the dynamic viscosity. Assuming a harmonic wave with a time dependence exp(i ω), the solution to this equation is

$$c = \frac{F}{i\omega} \exp\left(\sqrt{\frac{2i\omega}{F}}g\right), \qquad (7.223)$$

requiring that the function c_{i} be symmetric in y_{i} . The condition $v_{i} = 0$ at the hogedaries allows us to determine the constant c_{i} . We obtain the solution

$$\phi = \frac{F}{i\omega} \begin{bmatrix} 1 & \cosh\left(\sqrt{i\omega/i\phi}\right) \\ 1 & -\cosh\left(\sqrt{i\omega/i\phi}\right) \end{bmatrix}, \qquad (7.224)$$

When $\omega \rightarrow 0$, equation (7.224) becomes

$$v_{\perp} = \frac{F}{2p_{\perp}} a^2 + y^2 t,$$
 (7.225)

and the velocity prolife is parabolic, corresponding to the Poisenith flow.

The average (filtration) vebriity n_1 (see reptation (7.174)) is

$$-w_{0} = \frac{1}{2g} \int_{-1}^{\infty} v_{0} dg = \frac{F}{i\omega} \left[1 - \frac{1}{g} \frac{\partial w}{\partial \omega} \tanh\left(a \sqrt{\frac{\partial \omega}{\nu}}\right) \right], \qquad (7.226)$$

since the averaging is performed in the fluid section (i.e., an effective provsity equal to 1).

Defining the dimensionless variable as

$$g = g \sqrt{\frac{i_{12}}{n}}$$
. (7.227)

the average vehicity becomes

$$\dot{\psi}_{1} = \frac{tgFa^{2}}{gq^{2}} \begin{bmatrix} 1 - \frac{1}{q} \operatorname{rand}(q) \\ 1 - \frac{1}{q} \end{bmatrix}$$
(7.328)

The combination of equations (7.221) and (7.228) yields

$$\partial_{i} \eta_{i} = \rho_{i} \partial_{i} r_{i}^{(r)} = \frac{\eta}{a^{2}} \begin{bmatrix} \eta^{2} \\ 1 + C_{i} \eta \left(\tanh(\eta) \right) \end{bmatrix} \dot{r}_{i}$$

$$(7.329)$$

A comparison of equations (7.213) and (7.229) reveals that the viscoelynamic operator of the plane shi for harmonic waves is

$$\tilde{Y} = \mathcal{F}[Y(q)] = \frac{\eta}{a^2} \begin{bmatrix} \eta^2 & 1\\ 1 & (1,q) \operatorname{tabl}(\eta) \end{bmatrix}$$
(7.230)

Using equation (7.224), the viscous stress at the walls is

$$z = \eta \partial_t v_1(y = \phi \phi) + \eta \partial_t v_1(y = -y) - 2\eta \partial_t v_1(y = -y) - \left(\frac{2\eta F}{1-y}\right) \eta \tanh(\eta) = (7.231)$$

A generalized b proportional to the viscous stress can be obtained. Since b should be equal to the total fraction force per unit average relative velocity and unit volume of bulk material (i.e., the porcenty ϕ' , the fraction force per unit area of the likel is obtained by multiplying the stress τ by $\phi_1(2a)$. Then

$$b = \frac{\alpha \tau}{2m^2} = \begin{pmatrix} 3\eta \phi \\ \eta^2 \end{pmatrix} F_{\rm e} = bF_{\rm e}, \qquad (7.232)$$

where

$$T_1(q) = \frac{1}{3} \begin{bmatrix} -q \tanh(q) \\ 1 + (1/q) \tanh(q) \end{bmatrix}$$
(7.233)

 $(T_1(0) = 1)^n$. At high frequencies, $T_1 = (q/3)$ that is,

$$F(\infty) = \frac{a}{3} \frac{i\omega}{e}$$
. (7.231)

and the friction furce increases as the square root of the frequency.

Consider the case of low frequencies. Expanding the expression (7.230) in powers of y, and limiting the expansion to the first term in y, gives

$$Y = \frac{3\eta}{w'} \left(\frac{2}{5} y' + 1 \right), \qquad (7.235)$$

Comparing the time Fourier transform of equation (7.214) with equation (7.235), we find that at inv frequencies.

$$m = 16/5 m_{fi}$$
, (7.236)

Now note that $Y/q^2 \to u/u^2$ is explation (7.230) at the high frequency limit, i.e., when $q \to \infty$. In this limit, the viscous contribution should vanish and the result should give the expression of the inertial term $\omega m_1 q^2$. Since $\omega / q^2 = c/u^2$, we obtain

$$m = \mu_f$$
 (7.237)

at high frequencies (Bhat, 1962).

The operator 17/230 can be recast as the sum of an mertial term $\gamma_{a}m$ and a viscous term η/n as

$$Y(\omega) = i\omega m(\omega) - \frac{\eta(\omega)}{\kappa}, \qquad (7.238)$$

who re-

$$\eta(\omega) = \eta F_{\beta}(\omega),$$
 (7.239)

and m depends on frequency.

It turns out that the viscodynamic operator for pores of a circular cross-section can be obtained from F by substituting only $3c_i$ h, where r is the radius of the tubes:

$$F_1(n) \rightarrow F_1\left(\frac{3r}{4}\right)$$
, 7.240:

Alternatively, this is equivalent to a scaling in the frequency $\omega \to 9\omega_0 16$. Dust for a general particle modium, we may write

$$F_1 = F_1 \cup I_+$$
 (7.244)

where β is a structural factor that depends on the geometry of the pores: $\beta = 1$ for slit-like pores, and $\beta = 9/16$ for a tube of a circular cross-section. The best value of β is obtained by futurg experimental data

Johnson, Koplik and Dashen (1987) obtain an expression for the dynamic torthosity $\mathcal{T}(\omega)$ which provides a good description of both the magnitude and phase of the exact

 $^{2}T_{\rm I}$ should not be exacted with the disspace forms defined as (7.186). The notation is exacted, with Ran (1956).

368

dynamic formosity of large networks formed from a distribution of random radii. The dynamic formosity and dynamic permeability are

$$\mathcal{T}(\omega) = \mathcal{T} + (\varepsilon I - \text{and} - \varepsilon (\omega)) - \frac{\partial g \phi}{\mathcal{T}_{\mathcal{S}}(g)} = \kappa_0 \left(F - \frac{\varepsilon \mathcal{T}}{\varepsilon}\right)^{-1}, \quad (7.2)2j$$

respectively, where

$$F(z) = \begin{cases} 1 & \frac{6F^2\kappa_0}{rN^2\phi}, \quad r = \frac{q\phi}{zN_0rg}, \quad (5.243) \end{cases}$$

In equation (7.243), κ_{c} is the global periordality, \mathcal{T} is the fortuosity defined in (7.158) and λ is a geometrical parameter, with $2/\lambda$ bring the surface-to-prior volume ratio of the processible interface. The following relation between $\mathcal{T}_{c}(\kappa_{c})$ and λ can be used:

$$\frac{\xi T \kappa_0}{\alpha N^2} = 1.$$
 (7.213)

where $\xi = 12$ for a set of cancel slabs of fould and $\xi = 8$ for a set of non-intersecting canted tubes. Function F plays the role of function F_1 in the previous analysis. Figure 7.6 compares the real (a) and integinary (b) parts of F and F_1 solid and dashed bass, respectively, versus the frequency f_2 for $\kappa_0 = 1$ Darcy (10⁻¹⁷ mb), $\kappa_0 = 1$ eP, $\mu_f = 1000$ kg/m³, $\phi = 0.23$, T = 2, n = 20 µm, $\xi = 2$ and $\beta = 0.6$. Since Johnson, Kuplik and Dashen (1987) use the opposite convention for the sign of the Fourier transform, we represent $F(-\omega)$.



Figure 7.6: A comparison of function F_1 for a plane shift with $\beta = 0.6$ and function |F| proposed by Julitania Koplik and Disloch (1987) to model the dynamic intrinsity

The viscodynamic operator (7/214) derived from the Lagrangian approach, is valid up to frequencies where the Prascolle flow breaks down . According to equation .7/224) the complex waveramber of the oscillations is

$$k = \sqrt{\frac{i\omega}{n}} = (1 + i)\sqrt{\frac{1}{2n}},$$
 (7.245)

The quarter wavelength of the boundary layer is

$$\Lambda_{c} = \frac{1 - 2\pi}{4 \operatorname{Re}(k)} = \pi \sqrt{\frac{n}{2\omega}}$$
(7.246)

If we assume that the Puisenille flux breaks drive when λ_i is of the order of the pole size 2n, the limit frequency is

If we consider that the permeability of slit-fike pores is

$$\epsilon = \frac{\cos^2}{3}$$
, (7.248)

to agreement with equations (7,190), (7,232) and (7,238), the limit frequency can be expressed as

$$-i = \begin{pmatrix} \pi^2 \\ 24 \end{pmatrix} \frac{\eta \alpha}{\kappa \rho_f}, \qquad (7.249)$$

In a general phone medium, we may assume that the transition nerms when inertial and viscous forces are of the same order, i.e., when $i_{\pm}m = y/r$ (see equation (7.2484). This relation defines another criterion, based on the limit frequency.

$$\omega_{t}^{t} \sim \frac{g}{m \epsilon}, \qquad (7.250)$$

Using equation (7.212), we rewrite equation (7.250) as

$$\omega_1^2 = \frac{\eta \phi}{T \kappa_0 q}, \quad (7.251)$$

These forquearies define the limit of validity of the low-forqueary Bint's theory.

7.6.3 Anisotropic media

The equation of motion for anisotropic media has the form (7,200)

$$\partial_t \pi^{(\mu)} = U^{(\mu)}_{\mu} = \text{div}(\sigma^{(\mu)}),$$
 (7.252)

considering the powe pressure and stress components (7,131) and (7,133), the hazar momenta (7,167) and (7,168), and the dissipation forces (7,196). We assume for simplicity, an orthonhombic medium, and that the elasticity, partneability and induced mass matrices are diagonal in the same coordinate system.

Explicitly, we obtain

$$\partial_t \sigma_{ij}^{(R)} = \left[(1 - \phi_i)_{ij} - r_{(i)} \partial_{ij}^{i} \sigma_{j}^{(m)} + r_{(i)} \partial_{ij}^{j} \sigma_{j}^{(l)} + b_i (r_{(m)}^{(m)} - r_{j}^{(l)}) \right]$$
(7.253)

and

$$\partial_t (qq_f) = r_0 |\partial_q^t u|^n + (qq_f - r_0) \partial_q^t u_i^T = b_i (r_i^m - r_i^T) \qquad (7.251)$$

In terms of the relative field displacement, these equations are similar in form to expections (7.210) and (7.221), namely,

$$\partial_{\mu}\sigma_{\mu\nu} = \rho \partial^{\mu}_{\mu} \sigma^{\mu\nu}_{\mu} + \rho m^{\mu}_{\mu} \sigma_{\mu\nu}$$
 (7.255)

and

$$-\partial_t p_f = \rho_f \partial_t^2 w_t^{(m)} + m_t \partial_t^2 w_t + \frac{\eta}{\kappa_t} \partial_t w_t. \tag{7.250}$$

where m_i is given in equation (7.176). Introducing the viscotynamic matrix

$$\mathbf{Y} = \begin{pmatrix} Y = 0 & 0 \\ 0 & Y_2 & 0 \\ 0 & 0 & Y_n \end{pmatrix}, \quad (7.357)$$

with components $Y_i = Y_{i+1} \delta_{i,j}$, we see that equation (7.256) becomes

$$\partial_t p_f = \rho_f \partial_t^{\dagger} a^{(1)} + Y_t + \partial_t a_{1t}$$
(7.258)

where

$$V_{i} = m_{i}d_{i}\delta + \frac{g}{\nu_{i}}\delta, \qquad (7.350)$$

Equations (7.255) and (7.256) hold for inhomogeneous precisity

7.7 Plane-wave analysis

The characteristics of waves propagating in a porous modimu can be obtained by "probing" the medium with plane waves. Because, in isotropic media, the compressional waves are decoupled from the shear waves, the respective reputions of motion can be obtained by taking divergence and carl in equations (7.267) and (7.208).

7.7.1 Compressional waves

Let us consider first the lossless case (k = 0 in orphytor (7.207) and (7.2081). Firstly, applying the divergence operation to equation (7.207) and assuming constant material properties, we obtain

$$\partial_i \partial_j \sigma_{ij}^{(m)} = \rho_{ij} \partial_{ij}^2 \vartheta_m + \rho_{ij} \partial_{j}^2 \vartheta_j,$$
 (7.260)

where $\theta_{m} = \partial_t a_t^{(m)}$ and $\theta_f = \partial_t a_t^{(f)}$. From equation (7.19) and using (1.15), we have

$$\partial_i \partial_i v_{ij}^{(m)} = 2\mu_n \partial_i \partial_j v_{ij}^{(m)} - \left(K - \frac{2}{3}\mu_n\right) \partial_i \partial_i \delta_m + C \partial_i \partial_j \delta_j, \qquad (7.261)$$

Because $2\partial_t\partial_t v_{i_1}^{(m)} = \partial_t\partial_t\partial_t a_i^{(m)} + \partial_t\partial_t\partial_t u_j^{(m)} = 2\partial_t\partial_t\partial_t a_i^{(m)} = 2\partial_t\partial_t\partial_{m_t}$ we obtain from (7.260).

$$\left(K + \frac{4}{3}\rho_{0a}\right)\partial_{t}\partial_{t}\theta_{aa} + C\partial_{t}\partial_{t}\partial_{f} = \rho_{0a}\partial_{g}^{2}\theta_{aa} + \rho_{0a}\partial_{a}^{2}\theta_{f}$$

$$(7.262)$$

Secondly, the divergence of oppation (7/208) and the use of (7/20) gives

$$C\partial_t \partial_s d_m + R\partial_t \partial_s d_f = p_{st} d_h^2 d_{ts} + p_{tt} d_h^2 d_f$$
 (7.263)

Let us consider, without loss of generality – because the noshing is not topic – propagation in the colligering and assume the plane waves

$$d_{\mu\nu} = \partial_{\mu\nu} \exp[(4\omega t - \kappa r)]$$
(7.264)

$$\partial f = \partial f \rho \exp[\phi^2 \omega t - \epsilon |t|],$$
 (7.265)

where κ is the real wavenumber. Substituting these expressions into equations (7.262) and (7.264), we find that

$$\mathbf{B} \cdot \boldsymbol{\vartheta} = c_p^T \mathbf{D} \cdot \boldsymbol{\vartheta},$$
 (7.2696)

where

$$v_{\mu} = \frac{v^{2}}{c}$$
(7.267)

is the phase velocity, and

$$\boldsymbol{\vartheta} = \begin{pmatrix} \theta_{m1} \\ \theta_{TT} \end{pmatrix}, \quad \mathbf{B} = \begin{pmatrix} K + 4\mu_{m1} | 3 \rangle | C \\ C \rangle | R \end{pmatrix}, \quad \mathbf{D} = \begin{pmatrix} \rho_{11} \rangle | \rho_{12} \\ \rho_{21} \rangle | \rho_{22} \rangle, \quad (5.268)$$

Equation (7.200) constitutes an eigenvalue/eigenvector problem whose characteristic equation is

$$de^{i}(\mathbf{D}^{-1} | \mathbf{B} - c_{p}^{2} | \mathbf{I}_{0}) = 0,$$
 (7.269)

The solution of this second-order equation in r_p^2 has two costs, corresponding to two compressional waves. Let us denote the respective velocities by r_p^2 , where the signs correspond to the signs of the square root resulting from the solution of equation (7.299). Now, let us consider the two eigenvectors $\boldsymbol{\theta}_{i}$ and $\boldsymbol{\theta}_{j}$ and the respective equations resulting from 17.266).

$$\mathbf{B} \cdot \boldsymbol{\vartheta}_{\pm} = (\vec{r} - \mathbf{D} \cdot \boldsymbol{\vartheta}_{\pm}) - \mathbf{B} \cdot \boldsymbol{\vartheta}_{\pm} = r_{\mu}^2 - \mathbf{D} \cdot \boldsymbol{\vartheta}_{\pm}, \quad (7.270)$$

Multiplying the first by ϑ , and the second by ϑ , from the felt-hand side, we get

$$\boldsymbol{\vartheta} = \mathbf{B} \cdot \boldsymbol{\vartheta}, \quad (\hat{\boldsymbol{\zeta}}, \boldsymbol{\vartheta}) \in \mathbf{D} \cdot \boldsymbol{\vartheta}, \quad \boldsymbol{\vartheta}, \quad \mathbf{B} \cdot \boldsymbol{\vartheta} = (\hat{\boldsymbol{\zeta}}, \boldsymbol{\vartheta}) \in \mathbf{D} \cdot \boldsymbol{\vartheta} \quad (7.271)$$

Since matrices **B** and **D** are symmetric and x_{μ} , $|f| < \mu$, we obtain two orthogonality conditions, namely

$$\partial = \mathbf{B} \cdot \partial = \mathbf{I} = \partial + \mathbf{D} \cdot \partial = 0,$$
 (7.272)

In explicit form, the first condition is

$$(K + 4\mu_{0})^{3}(d_{nd} - \theta_{ndb} + C(\theta_{nd} - \theta_{pd}) + d_{fb} - \theta_{ndb}) + R\theta_{fb} - d_{fb} = -0.$$
 (7.273)

Because the elasticity constants are positive, this relation shows that if the amplitudes for one mode, say $\theta_{a,b}$ and $\theta_{a,b}$ have the same sign, $\theta_{a,b}$ and $\theta_{a,b}$ have opposite signs. This means that there is a wave for which the solid and the fluid move in phase and another in which they are in counterphase. Moreover, the following relation holds:

$$\vartheta_{+} \cdot \mathbf{B} \cdot \vartheta_{+} = r_{\mu\nu}^{2} \vartheta_{+} \cdot \mathbf{D} \cdot \vartheta_{+} = \vartheta_{-} \cdot \mathbf{B} \cdot \vartheta_{-} = r_{\mu}^{2} \cdot \vartheta_{-} \cdot \mathbf{D} \cdot \vartheta_{-} = -7.274$$

implying

$$v_{i,i}^{2} = \frac{(K + 4\rho_{m}/3)\delta_{m,i}^{2} + 2C\delta_{i,i} + \delta\rho_{m} + R\delta_{j,i}^{2}}{\rho_{m}\delta_{m,i}^{2} + 2\delta_{m,i}\delta_{m,i} + \rho_{M}\delta_{j,i}^{2}}, \qquad (7.275)$$

Considering the relative signs between the components and that p_{12} is the only negative coefficient (see equation (7.160)), we deduce that the higher vehenty has amplitudes in phase and the lower velocity has amplitudes in opposite phase. The last wave is called the slow wave of the wave of the second kind (Biot, 1956a).

Let us consider new the lossy case, starting from equations (7.210) and (7.213), where the last equation is intended in general, i.e., describing both the high and the lowfrequency ranges. Applying the divergence operation to equation 17 200, using 17 172). and assuming constant material properties, we obtain

$$\partial_i \partial_j \sigma_{ij} = \rho \partial_a^2 v_m + i q v_f |\partial_a^2 v_m + \partial_b^2 v_f|,$$
 (7.276)

From equation (7.33) and using (1.15), we have

$$\partial_t \partial_t \sigma_0 = 2\mu_m \partial_t \partial_t e_0^{-m} + \left(K_0 - \frac{2}{3}\mu_0 - \phi \phi M\right) \partial_t \partial_t \partial_m + \phi \phi M \partial_t \partial_t \theta_0, \qquad (7.277)$$

and (7.276) becomes

$$\left(K_0 + \frac{4}{3}\rho_m - \phi\phi M\right)\partial_t\partial_t u_m + \phi\phi M\phi_t\partial_t (\phi - 1) + \phi(\rho,\partial_t^2)u_m + \phi_t q \partial_t^2 u_{tr} = (7.278)$$

where we used the relation $2\partial_t \partial_t e_0^{(1)} = 2\partial_t \partial_t \partial_m$ and equation (7.150). Now consider equation (7.213). The divergence of this equation and the use of (7.22) and (7.32) gross

$$\partial_t \partial_t p_f = M(\alpha - \alpha) \partial_t \partial_t d_m + M(\alpha) \partial_t \theta_f = \rho_f \partial_{\mu}^2 d_m + \alpha V * \partial_t \theta_{\mu} + \alpha V * \partial_t \theta * - (7.279)$$

Let us consider, without loss of generality, the plane waves

$$\vartheta_{\mu} = u_{\mu \mu} \exp(0(zt - kx))$$
 (7.280)

$$\theta_1 = \theta_{10} \exp[i(\omega t - kx)],$$
 (7.281)

where k is the complex wavenumber. Substituting the expressions (7.280) and (7.281)into oppartors (7.278) and (7.279), we obtain

$$\left[K_{1i} + \frac{4}{3}\mu_{m} - \cos M - c_{i}^{2}(1 - \phi)\rho_{i}\right]\theta_{mi} + \phi(\alpha M - \rho_{i}c_{i}^{2})\theta_{1i} = 0,$$
 (7.282)

$$\left[M(\alpha - \phi) - c_{i}^{2} \left(\rho_{f} + \frac{i}{\omega} \delta \hat{Y}\right)\right] \theta_{ad} + \phi \left(M + \frac{i}{\omega} \hat{Y} c_{i}^{2}\right) \delta f_{0} = 0, \quad (7.283)$$

where \hat{V} is given by equation (7.238) and

$$c = \frac{\omega}{k}$$
(7.284)

is the complex velocity. The dispension relation is obtained by taking the determinant of the system of equations (7.282) and (7.283) equal to zero; that is

$$\left[K_{t_{0}} + \frac{1}{3}\mu_{t_{0}} - \cos(M - c_{0}^{2}) 1 - \cos\rho_{s}\right] \left(M + \frac{1}{2}c_{0}^{2}\tilde{Y}\right)$$

Chapter 7. Biot's theory for parous media

$$\left(\alpha M - \rho \rho c^{2}\right) \left[M(\alpha - \alpha) - r^{2} \left(\rho \rho + \frac{1}{\omega} \alpha V\right)\right] = 0 \qquad (7.285)$$

Multiplying this equation by ω and taking the limit $\omega \to 0$, we get Gassmann's velocity, regardless of the value of the viscodynamic operator.

$$v_G = \sqrt{\frac{1}{\rho}} \left(K_G + \frac{3}{3} \mu_m \right). \tag{2.86}$$

Reordering terms in equation (7.285), we obtain

$$\left(p_{f}^{2} + \frac{1}{\omega} \hat{Y} p\right) c_{i}^{2} + \left[\frac{1}{\omega} \hat{Y} \left(K_{i} + \frac{1}{3} p_{i}^{2}\right) + M_{0} 2\alpha p_{f}^{2} - p \mathbf{I}\right] c_{i}^{2} + M \left(K_{i} + \frac{4}{3} p_{m}\right) - 0.$$
(7.287)

where equation -7.341 has been used. The solution of this second-order equation in Q has two roots (corresponding to the fast and slow compressional waves obtained earlier. Let us denote the respective complex velocities by $z_{1,0}$ where the signs encoupond to the signs of the square cont resulting from the solution of regration (7.287). The phase velocity z_p is equal to the angular frequency ω divided by the root part of the complex wavenumber k, that is,

$$r_{\mu} = -Rer c_{1}^{-1} r_{1}^{-1}$$
, (7.288)

and the attenuation for to consecute to minus the magnety part of the complex waverumher; that is

$$a_{2} = \sqrt{\ln(r_{1}^{-1})},$$
 (7.289)

The high frequency velocity, say r_{∞} , of the low frequency theory is obtained by taking the limit $\hat{W}/\omega \approx -m$ in equation (7.238) —this is equivalent to considering $\eta \approx 0$, since the itertial effects dominate over the viscosity effects. Equation (7.287) then becomes

$$(m\mu - p_T^2)c_s^2 = \left[m\left(K_0 + \frac{4}{3}\mu_m\right) - M(2m\eta - p)\right]c_s^2 + M\left(K_0 + \frac{4}{3}\mu_m\right) = 0, -7.290$$

where excits real valued. Using (7.34) and defining the dry rock P wave modulus as

$$T_{\mu} = K_{\mu} + \frac{1}{3} \rho_{\mu}$$
 (7.201)

we note that equation 17/2980 heromes

$$(mp - p_f^2)e_{\infty}^2 = [m(E_m + \alpha^2 M_f - M)2mp_f - p_f^2)e_{\infty}^2 + ME_m - 0,$$
 (7.292)

It can be verified that $r_{\infty} \gg r_G$

Relation with Terzaghi's law and the second P wave

Terzaghi's law, used in geotechnics (Terzaghi, 1925), can be obtained from Biot's theory if $\alpha = \alpha$, $K_f \in [K_m]$ and $T \to 1$. The result is a decoupling of the solid and Buol phases (Bourbié, Consectional Zuezner, 1987, p. 81). Let us consider the first two conditions

274

7.7 Plane-wave analysis

Then, $M = K a' \phi$ from equation (7.26), $E_{m} + \phi^{2} M \simeq E_{m}$, $M(2iq\phi - p') w^{2} E_{m}$ and the solution of equation (7.292) is

$$c_s^2 = \frac{mE_m + \sqrt{m^2 E_m^2 - 4(m\rho - \rho_s^2)ME_m}}{2(m\rho - \rho_s^2)},$$
 (7.293)

Due to the second condition $(M \otimes K_{i,i})$ and the second term inside the square rout is much smaller than the first term. The fast-wave vehicity is

$$\frac{E_{re}}{\sqrt{\rho - \phi \rho_F T}} = \frac{E_{re}}{\sqrt{\rho - \phi \rho_F T}}$$
(7.294)

and a Taylor expansion of the square cont in (7.202) gives

$$\gamma_{\infty} = \sqrt{\frac{K_f}{mT}}, \quad (7.295)$$

where equation (7.212) has been used. Note that Therzaghe's law requires $T \to 1$, which implies $\psi_{n+1} = \{L_{n+1}(1) - c(p_n)\}^{-1/2}$ and $\psi_{n+1} = \sqrt{K_f/p_f}$. Thus, the last wave travels in the skeleton and the slow wave in the fund. The latter has the fluid velocity divided by the factor $\sqrt{T} \to 1$, heranse of the tertions nature of the pure space. Use of the superfluid ⁴He, which is two orders of magnitude more compressible than water, makes repeation (7.295) very accurate (Johnson, 1986). In this case, the slow wave is identified with the learnt-sound phenomenon. Measurements of the lourth-sound velocity give us the tertuosity T_{\pm} .

To our knowledge, the first observation of the second (slow) P wave is attributed in Plona (1980). The asial water-saturated sintered glass bands (see Bourlin). Consist and Zinszner (1985), μ (88)). However, Onra (1952a b) measured the slow wave vehicity in show, and scenes to have grasped its nature before Biou's theoretical prediction in 1956 (Biot, 1956a). Onra states "... the sound wave is propagated mainly by air in snow and its by structure only interferes with the propagation. (See Johnson (1982) for an interpretation using Biou's theory.) Observations of the slow wave in natural rushia are reported by Nakagawa, Soga and Mitchell (1997) for granular soils and Webler and Smeables (1997) for Nivelsteiner sandstime (see Section 7.13).

Actually, the slow wave has been predicted by Birt before 1956. Biot (1952) obtained the velocity of the tube wave (Schulte wave) in a fluid-filled circular borehole. This velocity in the low-frequency limit is given by

$$r_{\infty} = \frac{r_f}{\sqrt{1 - K_f}/\mu_s}$$
 (7.296)

where K_f is the fluid leffk modulus, $v_f = \sqrt{K_{f_f}} \rho_f$ is the fluid sound velocity, ρ_f is the fluid density, and ρ_s is the formation shear modulus. Notris (1987) shows that the tube wave is a fluiding case of the slow wave when the hore is considered as an isolated pore in a homogeneous porous portugal. A typical borehole radius is 10 cm, and considering an acoustic logging frequency of 1 kHz and water, the viscous skin depth is on the order of 100 μ m. If the horehole is considered to be a pore, the rase of zero viscosity has to be considered, i.e., the viscosity effects are negligible compared to the intertial effects. The tiple wave follows from Biot's theory, by taking the limit of vanishing parasity, using a torthosity $\mathcal{T} = 1$ and a discretis functions $K_m = \mu_c$ where μ_c is the grain shear modulus in Biot's theory.

The diffusive slow mode

Let us consider equations (7.278) and (7.279) at very low frequencies, when terms proportional to $\pm i$ can be needed to interms containing second-order time derivatives). Using equation (7.214) and denoting the Laplacian $\partial_t \partial_t$ by Δ_t we can rewrite these equations as

$$\left(K_{t_1} + \frac{1}{3}y_1 = \cos(M)\right)\Delta \theta_m + \cos(M\Delta u_f = 0)$$
(7.297)

and

$$\Delta p_f = M(\alpha - \phi)\Delta \theta_{\mu} + M\phi\Delta d_f = -\frac{\phi \eta}{h}(\partial_t d_{\mu} - \partial_t d_f), \quad (7.298)$$

Eliminating θ_{m} and θ_{m} and defining $\mathcal{P} = \Delta p$, we obtain the diffusion equation

$$d | \Delta P = \alpha P$$
 (7.270)

where, using equation (7.34).

$$d = M \left(\frac{s}{n}\right) \left(\frac{K_0 + 4\mu_m/3}{K_0 + 4\mu_m/3}\right)$$
(7.305)

is the corresponding hydraulic diffusivity constant. Because we have neglected the acceleration terms, we have obtained the differential equation corresponding to the diffusive slow mode (Chaudler and Johnson, 1984). Shapiro, Audigate and Royer (1999) apply the atisotropic version of this theory for estimating the permeability tensor from induced increases in a botelede.

7.7.2 The shear wave

Before deriving the shear-wave properties, let us recall that the outh operation requires a vector product between the Cartesian unit vectors, that is, $\mathbf{e}_i \times \mathbf{e}_i = e_{ikk} \mathbf{e}_k$, where e_{ikk} is the Levi-Civita tensor. Then, the cart of a vector \mathbf{u} is $\mathbf{e}_i + \mathbf{e}_i + e_{ikk}\partial_i \sigma_i | \mathbf{e}_i - \mathbf{W}_i$ define

$$\Omega^{(\ell)} \sim \operatorname{curl} \mathfrak{n}^{(0)} = \Omega^{(\ell)} \sim \operatorname{curl} \mathfrak{n}^{(\ell)}$$
(7.301)

Applying the curl operator lits: roleguation (7.210), using equation (7.172), and assuming constant material properties, we obtain

$$\langle v_{1n}\partial_t d_i v_{ij} | v_1 \rangle = {}_i \partial_t^2 \Omega^{(m)} + c_i p_f (\partial_t^2 \Omega^{(m)} + \partial_t^2 \Omega^{(f)}),$$
 (7.302)

Using equations (7.33) and (1.15), we have

$$\mathcal{D}_{max}(\mathcal{A}\mathcal{A}\mathcal{A}_{m}^{(0)}) = \mathcal{D}_{max}^{(0)} \mathcal{D}_{max}^{(0)} = \mathcal{D}_{max}^{(0)} = \mathcal{D}_{max}^{(0)} \mathcalD_{max}^{(0)} = \mathcalD_{max}^{(0)} = \mathcalD_{max}^{(0)} = \mathcalD_{max}^{(0)} = \mathcalD_{max}^{(0)} = \mathcalD_{max}^{(0)} = \mathcalD_{max}^$$

where the terms containing the dilatations ϑ_m and ϑ_t disappear, because the rule of the gradient of a function is zero. Because $2\epsilon_{th}\partial_t\partial_t\epsilon_m^{(m)}|\mathbf{e}_k| = \epsilon_{th}\partial_t\partial_t(\vartheta_t u_t^{(m)} - \vartheta_t u_t^{(m)})|\mathbf{e}_t| = \epsilon_{th}\partial_t\partial_t(\vartheta_t u_t^{(m)} - \vartheta_t u_t^{(m)})|\mathbf{e}_t| = \epsilon_{th}\partial_t\partial_t(\vartheta_t u_t^{(m)} - \vartheta_t u_t^{(m)})|\mathbf{e}_t| = \epsilon_{th}\partial_t\partial_t(\vartheta_t u_t^{(m)})|\mathbf{e}_t| = \epsilon_{th}\partial_t\partial_t(\vartheta_t u_t^{(m)})|\mathbf{e}_t| = \epsilon_{th}\partial_t\partial_t(\vartheta_t u_t^{(m)})|\mathbf{e}_t| = \epsilon_{th}\partial_t\partial_t(\vartheta_t u_t^{(m)})|\mathbf{e}_t| = \epsilon_{th}\partial_t(\vartheta_t u_t^{(m)}$

$$p_{\nu} \partial_{\rho} \partial_{\rho} \Omega^{(m)} \sim (1 - \phi) \rho_{\nu} \partial_{\mu}^{2} \Omega^{(m)} + \phi \rho_{f} \partial_{\mu}^{2} \Omega^{(f)}$$
, (7.304)

where we used equations (7.150).

Consider now equation (7.213). The curl of this equation gives

$$(0 + \rho_f \partial_0^2 \Omega)^m = (0 + \partial_t \Omega)^m + (\partial_t \Sigma + \partial_t \Omega)^T$$
, (7.305)

Let us consider, without loss of generality, plane waves traveling in the *x* direction and polarized in the *y* direction; that is $\mathbf{\Omega}^{(p)} = (0,0,\partial_1 u_2^{(p)})$ (p = m or f). Let us define $\Omega^{(p)} = \partial_1 u_2^{(p)}$ and $\Omega^{(f)} = \partial_1 v_2^{(f)}$. Then

$$Ω^{+} = Ω_{ab} \exp[i(2\theta - kx)],$$
 (7.306)

$$\Theta^{(l)} = \Theta_{fn} \exp(k) (kr + kr)$$
(7.307)

where k is the complex wavenumber. Substituting these plane-wave expressions into equations (7.304) and (7.505), we obtain

$$[\mu_{m} - e_{cs}^{2}] + \phi[\rho_{s}] \Omega_{ms} - \phi\rho_{f} r_{c}^{2} \Omega_{fb} = 0.$$
(7.308)

$$\left(pr + \frac{1}{\omega}\hat{Y}\phi\right)\Omega_{\mu\nu} + \left(\frac{1}{\omega}\hat{Y}\phi\right)\Omega_{\mu} = 0, \qquad (7.309)$$

where $v_i = \omega_j k$ is the complex shear-wave velocity. The solution is easily obtained as

$$r_{0} \sim \frac{\mu_{co}}{\sqrt{\mu - 4 + \rho_{f}^{2} Y_{co}}}$$
(7.30)

The phase velocity c_0 is reput to the organization frequency ω divided by the real part of the complex wavenumber kythators.

$$c_{\mu} = |\mathbf{R}c_{\mu}c_{\mu}|^{-1}$$
(7.314)

and the attrantian factor is equal to the imaginary part of the complex wavemanher that is,

$$\alpha = \omega \ln(\alpha_{1}/\beta_{1}^{2})$$
(7.312)

At low frequencies, $\hat{V} = (\chi m + \eta/r)$ (see reputtors (7.2) for and opportuni (7.3) It becomes

$$\nabla = \frac{P_{20}}{\sqrt{p + p_f^2 (m + i\eta_s^2) + s_s^2}}$$
(7.313)

In the absence of dissipation $(\eta/\kappa > 0)$ or when $\omega \to \infty$.

$$v_{i} \sim \sqrt{\frac{\rho_{in}}{\rho_{i}\rho_{i}}} \sim \sqrt{\frac{\rho_{in}}{\rho_{i}\rho_{i}}} \sim \sqrt{\frac{\rho_{in}}{\rho_{i}\rho_{i}}}$$
(7.315)

and

$$\Omega_{fr} = \left(1 - \frac{\rho_f}{m_b}\right)\Omega_{r,f} + \left(1 - \frac{1}{T}\right)\Omega_{mr}, \qquad (7.315)$$

from equations (7.212) and (7.309). Since the quantity in parentheses is positive, because T > 1, the rotation of the solid and the fluid are in the same direction. At zero irrepresely $(x) \rightarrow 0$, $(x \rightarrow \sqrt{p_m}/p)$ and, from equation (7.309), $\Omega_{m} = \Omega_{f,n}$ and there is no relative motion between the solid and the fluid. Note that because (x = 0) the velocity (7.315) is higher than the average velocity $\sqrt{p_m}/p$.



Figure 7.7: Prase velocities were is frequenced if the fast P wave, shown involved slow P wave an wave submatrix sand-time. The medium properties are $K_{i} = 35$ GPa, $\mu = 2050$ kg/m⁴, $K_{\mu} = 1.7$ GPa, $\mu = 1.855$ GPa, $\kappa = 0.3$, $\kappa = 1.5$ Gea, T = 2, $K_{f} = 2.1$ GPa, $\mu = 1060$ kg/m⁴, and $\eta = 1.0$ P chargons, 1998 .

Figure 7.7 shows the phase velocities of the different wave modes (equations (7.288) and (7.411) as a function of frequency). The medium is water-saturated satisfience and the curves correspond to the low-frequency (file, $1 = 1.200 \pm 10/m_{\odot}$). The vertical dashed line is the frequency $f_{1}^{0} = \omega_{0}^{0} (2\pi)$ (equation (7.257)), which indicates the upper limit for the validity of low-frequency Biot's theory. The slow wave has a quasi-static character at low frequencies and becomes inverdamped due to the fluid viscosity. If we replace water by oil (say, $\eta = 260 \text{ eP}$), this behavior their corresponds to higher frequencies. This phenomenon predictes the observation of the slow wave at sensitic frequencies. The presence of elay particles in the pores is an additional cause of attendation of the slow wave (Kementos and McCours, 1988).

7.8 Strain energy for inhomogeneous porosity

The Lagrangian formulation developed by Biot in his 1956 paper (Biot, 1956)) holds for constant parasity. The uses the average displacements of the solid and the fluid as Lagrangian coordinates, $u_i^{(n)}$ and $u_i^{(f)}$, and the respective stress components $|\sigma\rangle_i^{(n)}$ and $|\sigma\rangle_i^{(f)}$, as conjugate variables. The equations for variable priority are developed in Biot

(1962) and compared in detail to the 1956 equations. In his 1962 work, he proposes, as generalized coordinates, the displacements of the solid matrix and the variation of fluid content ζ defined in equation (7.32). In this case, the corresponding conjugate variables are the total stress components, σ_{ij} , and the fluid pressure, p_j . It is shown in this section that the 1962 equations are the correct ones for describing propagation in an inhomogeneous points medium. They are consistent with Darry's law and the boundary conditions of interfaces separating media with different properties. Two approaches are developed in the following sections. The first is based on the complementary energy theorem under small variations of stress, and the second is based in volume-averaging methods. An alternative demonstration, one given here, has here diveloped by Badmicki (2000, persueal communication), in terms of the randynamic potentials.

7.8.1 Complementary energy theorem

Let us consider an elementary volume Ω_0 of powers material bounded by surface S_0 . Assome that Ω_N is initially in static equilibrium under the action of surface forces

$$f_i^{(\mu)} = \sigma_{i\nu}^{(\mu)} u_{i\nu} - f_{i\nu}^{(f)} = -iqq u_{i\nu}$$

$$(7.316)$$

where n_i are the components of the outward number of perpendicular to S. Assume that the system is perturbed by $\delta f_i^{(m)}$ and $\delta f_i^{(d)}$ and let $V(ef_i^{(d)}), ef_i^{(d)})$ by the strand-energy density, and

$$V^{s} = \int_{\Omega_{0}} V d\Omega + \int_{S} (f_{0}^{im} |u_{0}^{(m)} - f_{0}^{(f)} |u_{0}^{(f)}) dS$$
(7.317)

by the complementary energy. Strictly, V should be the complementary strain energy density, however, for 2mean stress-strain relations, V is equal to the strain energy density (Fung. 1965; p. 293) and 295). The complementary energy theorem states that of all sets of forces that satisfy the equations of equilibrium and boundary conditions, the network one that is consistent with the prescribed displacements is admined by completing the complementary convergence, 1065, p. 2011. Then

$$\delta V^{\mu} = 0 = \int_{\Omega^{\mu}} \delta V d\Omega + \int_{S} (\delta f_{\mu}^{(m)} a_{\nu}^{(m)} + \delta f_{\nu}^{(m)} a_{\nu}^{(f)}) dS$$
(7.318)

īυĭ

$$\int_{\Omega_N} \delta V d\Omega = \int_{S} \left(\delta f^{(r)} u^{(r)} + \delta f_i^{(r)} u_i^{(t)} \right) dS$$
(7.309)

We have

$$\delta f_i^{(m)} = (\beta \phi_0 + i \phi_0 \delta p_f m_f - \text{and} - \delta f_i^{(f)} - i \delta p_f m_f, \qquad (7.320)$$

where we have used equations (7.21) and (7.29). Equation (7.318) becomes

$$d\Gamma^{*} = 0 = \int_{\Omega} \left[\delta \Gamma \partial \Omega - \int_{S} (u)^{0} \left[\delta \sigma_{ij} u_{ij} - v_{i} \delta p_{j} u_{ij} dS \right] \right]$$
(7.321)

where w_i are the components of vector **w** defined in equation (7.172). Applying Green's theorem to the surface integral, we obtain

$$\delta V^{*} = 0 = \int_{\Omega} \left[\delta V d\Omega - \int_{\Omega} \left[\partial_{\theta} (u_{i}^{(m)} \delta \sigma_{ij}) - \partial_{\theta} (w_{i} \delta p_{ij}) d\Omega \right] \right]$$
(7.322)

Because the system is in equilibrium before and after the perturbation, and the fluid pressure is constant in Ω_0 the stress increments must satisfy

$$\partial_i(\delta\sigma_\beta) = 0$$
, and $\partial_i(\delta q_i) = 0$. (7.323)

and we can write

$$\delta V^{*} = 0 - \int_{\Omega_{0}} dV d\Theta - \int_{\Omega_{0}} \left(e_{0}^{(m)} \delta \sigma_{0} + \sqrt{\delta} p_{f} \right) d\Theta, \qquad (7.324)$$

where $\zeta = -\partial_t w$ is the variation of fluid content replation (7.221), and $e_0^{(m)}$ is of the form defined in equation (1.3). To obtain the fluid term $\beta \delta \mu_t$, we used the fact that the porosity is locally constant, i.e., w is constant in the elementary volume Ω_h , but it may vary point to point in the points include. Moreover, the symmetry of the stress tensor has here used to alread the relation $\partial_t w_t^{(m)} d\sigma_h = e_0^{(m)} \delta \sigma_h$.

We finally deduce from equation (7.321) that

$$\delta V \sim r_{1}^{00} \delta \sigma_{0} + \zeta \delta \eta g,$$
 (7.325)

where evidently 1, has the functional dependence $V(\sigma_{ij}, \rho_{j})$, because upon taking the total derivative, we obtain

$$dV = \frac{\partial V}{\partial \sigma_{ij}} d\sigma_{ij} + \frac{\partial V}{\partial \mu_i} d\mu_i$$
 (7.326)

Comparing the last two equations, we can identify the strain-stress relations

$$\frac{\partial V}{\partial r_0} = \frac{\partial V}{\partial \rho_0} \approx - \frac{\partial V}{\partial \rho_0} \approx - (7.325)$$

For limit stress-strain relations, we have

$$2V = v_{0}^{(0)} v_{0} + \sqrt{p_{0}}$$
 (7.328)

Similarly, under small variations of displacements $V > V(e_{ij}^{m}, \zeta)$ and

$$\delta V = \delta \zeta_{0}^{(d_{0})} \delta \eta_{0} + \delta \zeta_{0}^{(d_{0})} \eta_{0} \qquad (7.329)$$

The stress-strain relations are

$$\sigma_{ii} = \partial V [\partial e_{ij}]^{(a)} = \mu_f - \partial V [\partial \zeta]$$
(7.330)

These stress-strain relations, valid for non-uniform porosity, are given in operations (7–131) and (7,133); see Bird (1962) for an equivalent demonstration of equation (7.329) for small variations of displayements.

7.8.2 Volume-averaging method

We follow Pride and Berryman's approach (Pride and Berryman, 1998) to had the appropriate stransgraphy density. Consider a volume $\Omega_{0} = \Omega_{0} + \Omega_{2} + \Omega_{10}$ of periods usedium $= \Omega_{0} - \Omega_{2} + \Omega_{2}$ for a fully saturated medium – and the weight function $W(\mathbf{r} - \mathbf{r}^{*}) = 1$

for \mathbf{r}' raside $\Omega_{\mathbf{r}}$ and $W(\mathbf{r} = \mathbf{r}') = 0$ for \mathbf{r}' outside $\Omega_{\mathbf{A}}$, where \mathbf{r} is the position vector. The averages of a general field ψ for points in the solid material and in the fluid are defined as

$$\psi_{t} = \frac{1}{\Omega_{s}} \int_{\Omega_{t}} W(\mathbf{r} - \mathbf{r}') \psi_{s} d\Omega', \qquad (7.331)$$

and

$$c_{II} = \frac{1}{\alpha_{II}} \int_{\Omega_{II}} W(\mathbf{r} - \mathbf{r}') c_{II} d\Omega'.$$
(7.332)

respectively. We define the increase opic stress tensor in the solid material by $\tau_{i_1}^{(1)}$, and the nucleocopic fluid pressure by τ_{i_1} . The equilibrium conditions miply that the respective perturbations satisfy

$$\partial_i((\tau_{ij}^{-1}) = 0, \text{ and } \partial_i(\delta \tau_j) = 0$$
 (7.333)

Then, $\delta \tau_f = \delta \tau_f - \delta p_f$ or the fluid region. The region defined by Ω_s is bounded by surface S of which S_i corresponds to the solid material and S_f to the fluid part. Moreover, denote S as the solid material (fluid interface contained inside Ω_{ij} . Since $\partial_i r_f^j = \delta_i$ and $\langle \tau_{ij}^j | \mathbf{r} \rangle$ does not depend on \mathbf{r}'_i .

$$\partial_{1}^{2} (x_{0}^{2} x_{1}^{2}) = z_{0}^{2}$$
, (7.334)

where, for brevity, we omit, hereafter, the increment symbol τ on the stresses. Integrating this quantity over the region Ω_{s} , we obtain

$$\frac{1}{\Omega_0} \int_{\Omega_0} d_2^{\prime} e_{k0}^{\prime \prime} r_0^{\prime} (d\Omega) = \frac{1}{\Omega_0} \int_{\Omega_0} v_0^{\prime \prime} d\Omega' \Rightarrow (1 - \alpha) e_0^{\prime \prime \prime} \Rightarrow (1 - \alpha) e_0^{\prime \prime} , \qquad (7.335)$$

because $\Omega_{i} = (1 - \alpha)\Omega_{i}$.

On the other hand, the same quantity can be expressed in terms of surface integrals by using Green's theorem.

$$\frac{1}{\Omega_{h}} \int_{\Omega_{h}} \partial_{t_{1}} \tau_{1_{1}}^{(s)} v_{j}^{(s)} d\theta^{(t)} = \frac{1}{\Omega_{h}} \int_{S_{h}} u_{j}^{(s)} \tau_{1_{1}}^{(s)} v_{j}^{(s)} dS^{(s)} + \frac{1}{\Omega_{h}} \int_{S_{h}} u_{1}^{(s)} \tau_{n}^{(s)} v_{j}^{(s)} dS^{(s)}$$
(7.336)

where $n_1^{(n)}$ is the outward unit vector normal to the surfaces S_i and S_i . Now, in S_i , we have the boundary condition $n_i^{(n)} \tau_{k_i}^{(n)} := -p_i n_i^{(n)} - Using this fact and equation (7.335), we get$

$$(1 + c)\delta_{Q}^{(s)} = \frac{1}{\Omega_{0}} \int_{S_{0}} u_{s}^{(s)} \tau_{0s}^{(s)} v_{s}^{\prime} dS^{\prime} - \frac{Pf}{\Omega_{s}} \int_{S_{0}} u_{s}^{(s)} v_{s}^{\prime} dS^{\prime}$$
(7.337)

The equivalent relation for the fluid is

$$\partial p_i d_{ij} := \frac{R}{\Omega_i} \int_{S_i} n_i^{(f)} v_j^{(d)} S' + \frac{R_f}{\Omega_i} \int_{S_i} n_j^{(f)} v_j^{(d)} S'', \qquad (7.338)$$

where we used the fact that $w^{T_{cont}} = w_{cont}^{T_{cont}}$ on N., Adding equations (7.647) and (7.338), we get the total average stress σ_{co}

$$\sigma_{ij} = (1 - \alpha)\sigma_{ij}^{-1} - \alpha p_j \delta_{ij} = \frac{1}{\Omega_h} \int_{S_h} v_1^{s} \tau_{ij}^{s} \tau_j^{s} dS^s - \frac{\rho_i}{\Omega_h} \int_{S_h} v_j^{s} dS^s, \qquad (7.339)$$

To obtain the macroscopic strain energy, we will consider the jacketed experiment, in which a porces sample is scaled in a very their and flexible jacket and macrised in a reservan providing a spatially uniform containing stress $\tau^{(i)}$. As in the jacketed test illustrated in Figure 7.1, a small tabe connects the interior of the sample with an external fluid reservant pressure μ_{S} , Ω_{c} is the volume of the sample, $S = S_{s} + S_{f}$ is the external fluid reservant pressure μ_{S} , Ω_{c} is the volume of the sample, $S = S_{s} + S_{f}$ is the external surface with S_{s} and S_{f} denoting the solid and fluid parts of the sample's exterior surface, and S_{f} is the surface of the small rule (S_{f} modules S_{f}). Because the cross-section of the tube is regligible compared to S_{s} we may assume that $\tau^{(i)}$ is equal to the macroscopic stress σ . The variation of fluid content is the volume of fluid that restres (or leaves) the sample through the tube. Since the tube is moving with the jacket, the variation of fluid content is given by

$$\zeta = -\frac{1}{\Omega_0} \int_{S_0} a_0^{(I)} w_0^{(I)} dS'$$
(7.3.00)

where $n_s^{(1)}$ is the outword normal to the tube cross section and $n_s^{(1)}$ is the microscopic find displacement '). Note that, in principle, the application of Green's theorem leads to the more finalities equation (7.22), However, the identification $\zeta = -dx$ we according to equations (7.22) and (7.52) requires certain conditions. Prode and Berzyman (1998) demanstrate that equation (7.22) holds if the center of the grain distribution in the averaging volume emissions with the center of this volume. Biot (1956)) assumes that the surface prioxity access an arbitrary cross-section of a sample and the volume potosity α are the same. This assumption is almost equivalent to Price and Berzyman's conduction. For highly heterogeneous or highly axisotropic materials, it is possible that the above relation reads to be provided and an additional parameter, modeling the surface perosity, should be introduced (Prich and Berzyman, 1998).

The strain-tensor components of the frame are given by

$$E_{(j)}^{(m)} = \frac{1}{\Omega_{k}} \int_{S} w^{(l)} w_{j}^{(l)} dS^{\prime}, \qquad (7.341)$$

where $\alpha_j^{(\ell)}$ is the outward normal to the jacket surface and $w_j^{(\ell)}$ is the microscopic displacement of the jacket surface. Note that

$$\epsilon_{0}^{(m)} \simeq \frac{1}{2^{0} \sqrt{\int_{S}}} \int_{S} (w_{0}^{(d)} w_{0}^{(d)} + w_{0}^{(1)} w_{0}^{(d)}) dS',$$
 (7.342)

The strain energy density is the sum of the average solid material and fluid energy densities. We can express these densities by

$$2V_{\text{shef}} = \frac{1}{\Omega_0} \int_{\Omega_0} a_0^{(1)} \partial_t^2 a_0^{(2)} d\Omega^2$$
(7.3.13)

and

$$\mathcal{P}_{i,k1} = \frac{1}{\Omega_{b}} \int_{\Omega_{b}} \tau_{I} \partial_{i} \sigma_{j}^{-I} d\mathcal{P}_{i}$$

$$(7.344)$$

with

$$V \sim V_{obs} + V_{mat}$$
 (7.345)

⁴ H is notation is a disistent with that used in Section 7.4. The microsector displacement $m_1^{(1)}$ should not be call firsts with the matrice opticity placements from the left vector the solid optimus are 7.172.

Since the fluid pressure is uniform uside the jacket, this implies $\tau e = pq$, and using the branchest conditions $a_i^{(f)} w_i^{(f)} = -n_i^{(f)} w_i^{(m)}$ on S_i , we have

$$2V_{\text{out}} = -\frac{p_f}{\Omega_k} \left(\int_{S_k} w_i^{(f)} w_i^{(f)} dS' + \int_{S_k} w_i^{(f)} w_i^{(0)} dS' \right).$$
(7.316)

In the case of the solid, due to the equilibrium condition (7.333) $||v_j|| \partial_t v_j^{(m)} = \partial_t^2 \langle v_j^{(m)} v_j^{(m)} \rangle$ we have

$$2V_{\text{solid}} = \frac{1}{\Omega_{\nu}} \int_{S_{\nu}} u^{(1)} \tau_{ij}^{(2)} w_{i}^{(2)} dS^{i} = \frac{\theta_{i}}{\Omega_{\nu}} \int_{S_{\nu}} u_{i}^{(2)} w_{i}^{(2)} dS^{i}, \qquad (7.317)$$

where the boundary condition $a_i^{(n)} \pi_0^{(n)} = -n_i^{(n)} \rho_i$ was used on S_i . Adding equations (7.346) and (7.447) gives

$$21 = \frac{1}{\Omega_{t}} \left(\int_{S_{t}} u_{i}^{(t)} \tau_{ij}^{(t)} w_{j}^{(t)} dS^{t} - p_{f} \int_{S_{t}} u_{i}^{(t)} w_{i}^{(t)} dS^{t} \right), \qquad (7.318)$$

In hight of the jackited experiment, the second integral can be partitioned between an integral on the fluid surface plus an integral on the tube (toss section. Then, using (quarion (7.340), we have

$$21 = \frac{1}{\Omega_0} \left(\int_{S_0} w^2 |\tau_0^{(s)} w_j^{(s)} dS' - p_j \int_{S_0} w_j^{(t)} w_j^{(t)} dS' \right) < \varsigma(p_j)$$
(7.349)

where S_{ff} is the surface of fluid in contact with the jarket. The energy balance in the surface of the jacket implies that the list term of the right-hand side ration expressed as

$$\frac{1}{\Omega_{n}} \left(\int_{S_{n}} u_{i}^{(1)} \tau_{0}^{(1)} u_{i}^{(2)} dS^{n} - p_{f} \int_{S_{n}} u_{i}^{(f)} w_{i}^{(f)} dS^{n} \right) = \frac{1}{\Omega_{h}} \int_{S} u_{i}^{(f)} \tau_{0}^{(i)} u_{i}^{(f)} dS^{n} - \frac{i_{0}}{\Omega_{h}} \int_{S} u_{i}^{(f)} w_{i}^{(f)} dS^{n} - \frac{i_{0}}{\Omega_{h}} \int_{S} u_{i}^{(f)} u_{i}^{(f)} dS^{n} - \frac{i_{0}}{\Omega_{h}} \int_{S} u_{i}^{(f)} dS^{$$

because $v_0^{(1)}$ is spatially uniform. Using the fact that $v_0^{(1)} = \sigma_0$ and equation (7.341), we find that the strain energy (7.349) becomes

$$2V = E_0^{(m)} n_0 + \pi p_f, \qquad (7.351)$$

which can be conducted to equation (7.328) if we consider that the rotational part of the strain tensor $\mathbf{E}^{(m)}$, namely,

$$\frac{1}{2\Omega_{h}} \int_{\delta} (a_{j}^{(l)} a_{j}^{(l)} - a_{j}^{(l)} a_{j}^{(l)}) dS^{\prime}$$
(7.352)

does not contribute to the work required to deform the sample, since no stress moments are applied

7.9 Boundary conditions

The phenomena describing the reflection refraction and diffraction of waves are related to the presence of informageneities and interfaces. Knowledge of the corresponding boundary conditions is essential to concernly describe these phenomena. In fluid fluid contacts in potents materials, we should expect fluid flow across the interface when a wave passes, and as we have mode conversion from P to S energy in single-phase media, we may expect mode conversion from P to S energy in single-phase media, we may expect mode conversion form P to S energy in single-phase media, we may expect mode conversion between the three waves propagating on a portors monimum. In the developments that follow, we derive the appropriate boundary conductors for the different cases:

- 1. Priors medium/purous medium
- 2. Privous medium/viscuelastic (single phase) un dimit.
- 3. Parous medium/viscoacoustic medium clossy fluid).
- Free surface of a porous medium.

7.9.1 Interface between two porous media

We consider two different approaches used to drive the appropriate boundary conditions for an interface between two percons media. The first follows in part the demonstration of Deresiewlez and Skalak (1963) and the second is that developed by Gurevich and Scheenberg (1999), based on a method used to obtained the interface conditions in electromagnetism. The conditions, as given here, also hold for the anisotropic case

Deresiewicz and Skalak's derivation

Let us consider a volume Ω_{n} of porous material bounded by surface S_{n} and let us calculate the rate of change of the sum of the kinetic- and strain-energy densities, and dissipation potential: that is, $\partial (T + dV + \Phi_{D} + P)$, where P is the power input. First, note that the substitution of equations (7.131) and (7.133) into equation (7.338) gives

$$2V + r_{n,0}^{a} r_{n}^{(a)} r_{n}^{(a)} \sim 2M \alpha_{0} r_{0}^{(a)} \sim M_{S}^{(a)}$$
 (7.353)

and that

$$2\partial V \geq 2tr_{0,0}^{n}r_{0,1}^{m} = M\alpha_{0,s}(\partial_{t}r_{0,s}^{m} + 2M)_{s} - \alpha_{0}r_{0,s}^{m}(\partial_{t,s} + 2)\sigma_{0}(\partial_{t}r_{0,s}^{m} + p_{f}(\partial_{t,s}), (7.35))$$

where we have used $\phi_{ab} = \phi_{ab}^{a}$. Using these explations and the expressions for the kinetic and dissipated energies (7.1%) and (7.197) (premitt volume) and integrating fluse energy densities on Ω_{b} , we obtain

$$P = \partial_t F + \partial_t V + \Phi_0 = \int_0^{\infty} p \partial_t e_t^{(m)} + p p \partial_t a_t a_t^{(m)}$$
$$(p p \partial_t e_t^{(m)} + m_t \partial_t^2 e_t + (\eta (\kappa) \partial_t e_t) \partial_t e_t + (\sigma_t \partial_t)_t^{(m)} - p p \partial_t \partial_t e_t) d\theta = -(7.355)$$

7.9 Boundary conditions

where we have used the relation $\sigma_{ij}\partial_t \epsilon_{ij}^{(m)} = \sigma_{ij}\partial_j v_i^{(m)}$. Since $\sigma_{ij}\partial_j v_i^{(m)} = \partial_j(\sigma_{ij}v_i^{(m)}) - \partial_j\sigma_{ij}v_i^{(m)}$ and $p_f\partial_i\partial_t w_i = \partial_i(p_f\partial_t w_i) - \partial_i p_f\partial_t w_i$, we apply the divergence theorem to the terms in the last parentheses on the right and get

$$P = \int_{\Omega} [(\rho \partial_t v_i^{(m)} + \rho_f \partial_{tt}^2 w_i - \partial_j \sigma_{ij}) v_i^{(m)} + (\rho_f \partial_t v_i^{(m)} + m_i \partial_{tt}^2 w_i + (\eta/\kappa_i) \partial_t w_i + \partial_i \rho_f) \partial_t w_i] d\Omega + \int_S (\sigma_{ij} v_i^{(m)} - p_f \delta_{ij} \partial_t w_i) n_j dS, \quad (7.356)$$

where n_j are the components of the outer normal to S. In this equation, we can identify the differential equations of motion (7.255) and (7.256), such that the volume integral vanishes, and

$$P = \int_{S} (\sigma_{ij} v_i^{(m)} - p_f \delta_{ij} \partial_t w_i) n_j dS \qquad (7.357)$$

quantifies the rate of work done on the material by the forces acting on its surface.

We now consider two different porous media in contact with volumes Ω_1 and Ω_2 and bounding surfaces S_1 and S_2 , respectively, with a common boundary S_c , as shown in Figure 7.8. Let us define the power per unit area as

$$p_k = (\sigma_{ij}^{(k)} v_i^{(m)(k)} - p_f^{(k)} \delta_{ij} \partial_t w_i^{(k)}) n_j^{(k)}, \quad k = 1, 2.$$
(7.358)



Figure 7.8: Two different porous media in contact with volumes Ω_1 and Ω_2 and bounding surfaces S_1 and S_2 , respectively, with a common boundary S_c .

The respective power inputs are

$$P_1 = \int_{S_1} p_1 dS + \int_{S_e} p_1 dS, \qquad P_2 = \int_{S_2} p_2 dS + \int_{S_e} p_2 dS. \tag{7.359}$$

The power input of the combined system should satisfy

$$P = \int_{S_1} p_1 dS + \int_{S_2} p_2 dS. \tag{7.360}$$

Conservation of energy implies $P = P + P_0$, and, therefore,

$$\int_{S_1} p(dS + \int_{S_2} p(dS - 0))$$
(7.461)

or, using the last that in the contaton boundary $n_{i}^{(1)} = -n_{i}^{(1)} - n_{i}$

$$\int_{S} \left[\left(\sigma_{ij} | v_{ij}^{(i)} | - \rho_{j}^{(i)} \delta_{ij} \partial_{i} w_{ij} \right) - \left(\sigma_{ij}^{(i)} | v_{j}^{(ij)} | - \rho_{j}^{(i)} \delta_{ij} \partial_{i} w_{j}^{(i)} \right] \left[u_{ij} dS = 0, \quad (7.362) \right]$$

which can be satisfied if we require the continuity across the interface of the power input per unit area, namely.

$$(\sigma_0)_i^{(m)} = p_f \delta_{ij} \delta_f \phi_i (a_i)$$
(7.363)

This condition can be fulfilled by requiring the continuity of

$$r_0^{(i)} = \partial_0 w_0 u_0 + \sigma_0 u_0 + p_f$$
 (7.464)

that is eight houndary combitions.

The first condition requires that the two frames remain in contact at the interface. Note that continuity of $u_1^{(1)}$ or instead of $w_i u_i$ also guarantees the continuity of the power input per mit area. However, this is in contradiction with the conservation of fluid mass through the interface. The second condition (7.364) implies perfect fluid flow across the interface. If the interface is perpendicular to the tracks, equation (7.364) supplies continuity of

$$x_1^{(n)} = \partial_t w_{n} - \sigma_{nn} - p_0, \qquad (7.365)$$

If there is not perfect communication between the two works, fluid flow results in a pressure drup through the interface according to Darry's law

$$p_f^{(2)} - p_f^{(2)} = \frac{1}{\kappa_e} \partial_t w_t a_t.$$
 (7.306)

where c, is the hydraulic permeability (per unit length) of the interface, or

$$p_f^{(2)} - p_f^{(3)} = \frac{1}{\kappa_s} \partial_t w_0.$$
 (7.367)

for $n_1 = 3.3$

The second condition (7.364) is obtained for $\kappa_s \rightarrow \infty$. The chain $\kappa_s = 0$ encosponds to a scaled interface $(\partial_t w, \alpha_s \approx 0)$. A figurous justification of equation (7.366) can be obtained by invoking Handleon's principle (Bourblé, Consey and Zinszner, 1987, p. 246).

Gurevich and Schoenberg's derivation

Ginevich and Schoenberg (1996) derive the heatidary conditions directly from Birt's cutation of powerlasticity by replacing the discontinuity surface with a thin transition layer

in which the properties of the možium change topidly but continuously - and then taking the limit as the layer thickness tends to zero. The method considers the inhomogeneous equations of motion and assumes that the interface is described by a jump in the material propurities of the parameterizing - by $-\lambda$ be a parit on the discontinuity surface, and



Figure 7.9: Drasadar size at the interface herocontwo porors randor

consider a Carnesian system with its origin at point A and us x-axis perpendicular to the discontinuity surface (Figure 7.9)

I oblowing Deymann Leighton and Sands (1965, p. 34-4) to 33-7), we substitute the discontinuity by a thin transition layer of thickness d_{-12} which the material properties change rapidly but continuously. Fur thickness d_{-13} which the material properties derivatives with respect to x of the material properties are tunch larger than the derivatives with respect to x of the material properties are tunch larger than the derivatives with respect to y and z.

According to the arguments discussed in Section 7.8, Bor's differential equation for an inhomogeneous anisotropic poroclastic medium are given by equations (7.131), (7.134), (7.255) and (7.258). Denoting the particle velocities by $v_i^{(n)}$ and $\psi_i \approx dw_i$, and using equations (7.172) and (7.173), the equations of motion can be expressed in a particle velocity/stress form as

$$\begin{aligned} \partial_t \phi_{ij} &= -\rho \partial_t c_i^{(m)} + \rho_j \partial_t \dot{\psi}, \\ \partial_t \rho_j &= -\rho \partial_t c_i^{(m)} + \dot{Y}_i + \dot{\psi}_i \\ \partial_0 \rho_j &= \dot{W} (\partial_t w_i + \alpha_j c_j^{(m)}) \\ \partial_t \phi_j &= -c_j \phi_j^{(m)} + M \alpha_j \partial_t \dot{\phi}_j, \end{aligned}$$
(7.368)

where

$$\begin{aligned} \left(\frac{m}{r_{m}} - \partial_{1} \right) \frac{m}{r_{m}} \\ \left(\frac{m}{r_{m}} - \partial_{2} \right) \frac{m}{r_{m}} \\ \left(\frac{m}{r_{m}} - \partial_{2} \right) \frac{m}{r_{m}} \\ \left(\frac{m}{r_{m}} - \partial_{2} \right) \frac{m}{r_{m}} - \partial_{3} \left(\frac{m}{r_{m}} - \partial_{3} \right) \frac{m}{r_{m}} \\ \left(\frac{m}{r_{m}} - \partial_{3} \right) \frac{m}{r_{m}} - \partial_{3} \left(\frac{m}{r_{m}} - \partial_{3} \right) \frac{m}{r_{m}} \end{aligned}$$

$$(7.303)$$

We take the limit $d \to 0$ and neglect all the terms containing the derivatives ∂_t and ∂_t

We obtain the following eleven equations:

$$\begin{aligned} \partial_t \sigma &= \mathcal{O}(1), \quad \phi = 1, 2, 3, \\ -\partial_t \rho_0 &= \mathcal{O}(1), \\ M\partial_t \dot{w}_t + \alpha_t \partial_t v_t^{(m)} + \alpha_t \partial_t v_t^{(m)} + \alpha_t \partial_t v_t^{(m)} = \mathcal{O}(1), \\ v_{ij} \partial_t v_t^{(m)} + v_{ij} \partial_t v_t^{(m)} + v_{ij} d_t v_t^{(m)} + M\alpha_k (t v_t - \mathcal{O}(1), \quad k = 1, \dots, 6. \end{aligned}$$

$$(7.370)$$

Equations (7.370), and (7.370), are satisfied if and only if

 $\partial \langle v_i^{(n)} \rangle = O(1), \quad \partial_i w_i \in O(1)$ (7.371)

As indicated in Figure 7.9, the jump at the interface of a material property and held variable denoted by f is $f \to f^+$. Substituting each derivative $\partial_t f$ by the corresponding initialifficience value $(f^+ + f^-)/\partial_t$ multiplying both sides of each of the equations (7.370), (7.370), and (7.375) by ∂_t and taking the limit $d \to 0$, yields

$$\begin{aligned} \sigma_i^* &= \sigma_{ii} = 0, \quad i = 1, 2, 3, \\ \mu_i^* &= \mu_f = 0, \\ r_i^{(m)} &= r_i^{(m)} = 0, \quad i = 1, 2, 3, \\ w_i^* &= w_i = 0, \end{aligned}$$
(7.372)

that is, eight independent boundary conditions. These conditions are equivalent to the open-pore boundary conditions (7.654) of Deresiewicz and Skalak (1963). This means that Deresiewicz and Skalak's model of partially permeable contact when the pores of the two modia do not match at the interface is logidy unlikely to occur. A different interpretation is provided by Guervich and Scheroberg (1999). They emissible a thin pure lastic layer of thickness d with permeable to d_1 and open-pure boundary conditions at hoth sides of the layer. They show that the boundary conditions (7.366) holds for low frequencies, but for high frequencies the hydraulic permeability κ_1 should be frequency dependent. The solution of the reflection-transmission problem of plane waves for this boundary condition is green by Deresiewicz and Roc (1963). Dubba and Odé (1953) Source, Corberó (Kavazzoh and Hensley (1992), Derbeman Drukoningen and Wapenaar (2002) and Sharma (2004).

de la Craz and Spanos (1982) obtain an alternative set of homolacy conditions, based on volume-average arguments. They interpret the contact between the potous media as a transition region. An interesting discussion regarding these boundary conditions and those of Deresiewicz and Skalak (1963) is provided by Curevich (1994).

7.9.2 Interface between a porous medium and a viscoelastic medium

be this case, the visco-dastic medium, say, medium 2, is imperinciable $(\partial_t w_t = 0)$, and its points its should be set uspall to zero. These conditions require

$$dar_{s_0} = 0, \quad \text{in} \quad c_1^{(m)} = c_1^{(f_0)}$$
 (7.373)

and the continuity of

$$v_i^{(0)} = \sigma_{i,0} = (i - 1, 2, 3)$$
 (7.374)

The solution of the reflection-transmission problem of plane waves for the boundary condition is given by Sharma, Nauslak and Gogna (1990).

288

7.9.3 Interface between a porous medium and a viscoacoustic medium

The viscoaronstic medium, say, mediums 2, is a fluid, and therefore has porosity equal to 1. Hence there is free flew across the interface, $w_i = \infty$), and the filtration velocity of the fluid is $d_i w_i^{(2)} = v_i^{(2)} - v_i^{(2)}$. This requires

 $\sigma_{1}(\sigma_{1}^{(f)} = \rho_{1}^{(m-1)}) = \sigma_{1}^{(f)} = (\rho_{1}^{(m-1)} + \rho_{f}^{(f)} - \rho_{f}^{(f)} + \sigma_{11}^{(f)} - \rho_{f}^{(f)} - \sigma_{11}^{(f)} - 0, (7.375)$

An example is given in Santos, Cetheró, Ravazzoff and Hensley (1992).

7.9.4 Free surface of a porous medium

There are no constraints on the displacements since the medium is free to mucr. For this trasen, the stress components and porc pressure vanish. The natural conditions are

$$\sigma_{15}^{(1)} \approx \sigma_{15}^{(1)} \approx \sigma_{21}^{(1)} \approx 0, \quad p_1^{(1)} \approx 0. \quad (7.370)$$

The subtime of the reflection transmission problem of plane waves for this branchary condition is given by Deresiewicz and Rice (1962).

7.10 The mesoscopic loss mechanism. White model

A major cause of attennation in promis modul is wavesinduced fluid flow, which does at different spatial scales. The flow can be classified as macroscopic, mesoscopic and microscopic. The attennation mechanism predicted by Biot's theory has a macroscopic nature. It is the wavelength scale equilibration between the peaks and trenghts of the P wave. Geerstea and Smit (1964) showed that the dissipation factor 1/Q of the last P wave, obtained as $\lim_{t\to\infty} (1964)$ showed that the dissipation factor 1/Q of the last P wave, obtained as $\lim_{t\to\infty} (1964)$ in analogy with viscoelasticity (see equation (3.1284)), can be approximated by that of a Zener model for $Q \gtrsim 5$. They obtain the expression (2.175):

$$Q^{-1}(\omega) = \frac{\omega_{c}\tau_{c} - \tau_{c}1}{1 + \omega^{c}\tau_{c}\tau_{c}}, \quad z_{c} = \left(\frac{v_{C}}{v_{\infty}}\right)^{2}\tau_{c}, \quad v_{c} = \frac{\mathcal{X}e\rho}{\eta}, \quad (7.377)$$

where e_{X} , as the P-wave velocity at the high-frequency limit (see equation (7.292)), e_{H} is given by equation (7.286) and $\mathcal{X} = p_{T}T_{T}(po) = (p_{T}/p)^{2}$. We have seen in Section 2.4.3 that the location of the Zerer relaxition peak is $\omega_{R} = 1/\sqrt{\tau_{e,T}}$ (see equation (2.176)). Then, $f_{W} = \omega_{W}/2\pi$ and using (7.377) we get

$$f_h = \begin{pmatrix} r_{\infty} \\ \phi_f \end{pmatrix} \frac{\eta}{2\pi X \kappa p} \approx \frac{\eta}{2\pi X \kappa p} - \frac{\phi \eta \rho}{2\pi X \kappa p} = 2\pi \kappa \rho_f (\rho T - \phi \rho_f)$$
(7.378)

This equation shows that the relaxation peak moves towards the high frequencies with increasing viscosity and decreasing permeability. This means that, at low frequencies, attenuation decreases with increasing viscosity (or decreasing permeability). This is in contradiction with experimental data (e.g., dones, 1986). Another apparent drawback of Bot's theory is that the reperoscopic-flow mechanism underestimates the velocity dispersion and attenuation in nocks (e.g., Machizaki, 1982). Deorker, Maeko and Nuz, 1995; Another and Coreiner, 2001).

It is common to myoke mon-Biot, attenuation mechanisms to explain low-frequency twistic and sonial attemption in tooks. These mechanisms are the so-called local fluid flow, or "squirt" Row absorption mer hardstus, which have been extensively discussed in the literature (O'Connell and Budiwasky, 1974; Dyngkin, Mayko and Nur, 1995; Mayko, Mak erji and Dyorkin, 1998). In this mechanism, fleid-hlled microcracks respond with greater limil-pressure changes that the main pore space. The resulting flow at this inbroscopic level is the responsible for the energy loss. Thuse models have the proper dependence on viscosity with the center frequency of the attemption peak more sety proportional to third viscosity. However, it has been shown that this morehanism is incapable of describing the measured levels of dissipation at seismic frequencies (Diallo, Prasad and Appel, 2003). Profe. Berryman and Harris (2003) have shown that attenuation and volocity dispersion measurements can be explained by the combined effect of these copie-scale informogeneities and energy transfer between wave modes. We relet to this mechanism as mesoscope loss. The mesoscopic-scale length is intended to be much larger than the gram sizes but upult smaller than the wavelength of the pulse. For restonce, if the fluid compressibility varies significantly from point to point, diffusion of pure thail between different regions constitutes a mechanism that can be important at solshier frequencies. White (1975) and White. Mikhayleya and Lyakhovitskiy (1975) were the first roda roda e the mesoscopic loss mechanism based on approximations in the transports of Boet's theory. Drev considered gas pockets in a water-satigated porous realign and porous layers altereately saturated with water and gas, respectively. These are the first so-called spatche saturatum" models. Dutta and Dde (1975)a bitwal Datta and Scoff (1975) solved the problem exactly by using Bint's theory and confirmed the accuracy of White's results.¹¹

To illustrate the mesoscopic loss mechanism, we compute the P wave complex modulus of a layered mechan in the direction perpendientar to the layering. We follow the demonstration by White (1975) and White. Mikhaylow and Lyakhovitskiy (1975) and, in this case, the result is exact

Figure 7.00 shows alternating layers composed of two fluct-saturated porous media, where by symmetry, the elementary volume is enclosed by mollow boundaries. In order to obtain the **P** wave complex modulus, we apply a tension $\phi_i \exp(i_i t)$ on the top and layton of the elementary volume and compute the resulting strain $i_i \exp(i_i t)$. Then, the complex modulus is given by the ratio

$$s^{2} = \frac{\sigma_{0}}{\epsilon}$$
, 7.479)

The strain e is obtained in two steps by computing the strains *i*, and *ey* without and with fluid flow across the interfaces separating the two media.

If there is no fluid flow, the strain is

$$c_0 = \frac{\sigma_0}{\varepsilon_0}$$
, (7.480)

where the composite modulus is given by equation (1.179), as

$$\mathcal{E}_{tr} = \begin{pmatrix} p_{tr} + p_{z} \\ F_{trr} + F_{trr} \end{pmatrix} \quad . \tag{7.381}$$

⁴Darta and Scrift (1979) primbout war istoke in White (1975), where White uses the P wassemodulus assumed of the topk and physics tenter on the complex hulls contains.



highter 7/10; Alternating faces composed of two field saturative process residu

(see also equation (1.188) i), with $p_{ij} = d_{ij}(d_{ij} + d_{j}), l = 1, 2$.

$$E_0 = K_0 + \frac{1}{4}a_0$$
, $l = 1, 2$ (7.882)

where K_{th} are the Gassmann moduli of the potons media topiation (7.34)) and η_{th} are the respective dry-rock shear moduli.

The displacements in the κ -and *y*-directions are zero under the application of the normal stress σ_{00} and $\phi_{10}^{(0)} = \phi_{10}^{(0)} = 0$. Moreover, at low frequencies, fluid and solid move trigetion and $\zeta = 0$ in equation (7.33). Fuder these conditions and using equation (1.35) we have at each medium.

$$\sigma_{i} = \left(K_{ii} + \frac{4}{3}p_{iii}\right)r_{iii}^{(i)} = L_{ii}r_{iii}^{(i)}, \qquad (7.383)$$

On the other hand, equation (7.32) implies $-pq = \alpha M \theta_{\mu\nu}^{\alpha\nu} + \alpha M e_{\mu\nu}^{\alpha\nu}$, which combined with (7.383) gives

$$\frac{\eta}{\sigma_{0}} = \frac{\alpha M}{E_{0}} = r, \qquad (7.385)$$

where |W| and α are given by equations (7.24) and (7.25), respectively. Then, for each medium, it is

$$p_l = c_l \sigma_{ll}, \quad l = 1, 2, \quad (7.385)$$

where the physisign indicates that the fluid pressure is that of the fast compressional wave According to this equation, there is a fluid pressure difference at the interfaces, and this difference generates fluid flow and slow (diffusion) waves traveling into rach medium. As fluid flows, the matrix expands in the toditoction and $\sigma_{A} = 0$. In this case, ζ does not venish and equation (7.33) implies

$$0 = E_0 r_{18}^{\mu} = \alpha M_{s}$$
, (7.386)

Combining this equation with (7.32) gives the effective bulk modulus

$$\frac{p_f}{z} = M\left(1 - \frac{\alpha^2 M}{E_R}\right) - \frac{M E_m}{E_R} - K_F$$
(7.387)

where we have used repeations (7.34) and (7.291). Another result from equation (7.386) is the expansion coefficient

$$\frac{c_{A}^{(0)}}{E_{C}} = \frac{c_{A}M}{E_{C}} = c_{A} = -\frac{1}{2} \frac{c_{A}}{2} \frac{c_{A}}{E_{C}} = -\frac{1}{2} \frac$$

At low frequencies, when the acceleration terms can be neglected (equations (7.213) and (7.214) give Darcy's how

$$\dot{w}_{\lambda} = -\frac{\kappa}{\eta} \partial_{\lambda} q_{f}, \quad \text{or} \quad \partial_{0} \dot{\sigma}_{0}, \qquad \frac{\kappa}{\eta} \partial_{0}^{2} q_{f}, \qquad (7.389)$$

On the other hand, equations (7,173), (7,174) and (7,487) imply

$$\partial_4 \dot{w}_8 = -\frac{1}{K_f} \partial_t p_f. \tag{7.360}$$

The two preceding equations yield

$$\partial_s^2 p_f = -\frac{\hbar}{\kappa K_f} \partial_t p_f \qquad (7.391)$$

The solution is

$$p_I = Aexp(a_i) + Bexp(-a_i) exp(i_i t),$$
 (7.392)

where

$$w = \frac{6\pi h}{\kappa K_F}$$
, (7.393)

and, from optimon (7.389).

$$\dot{w}_{t} = -\frac{\kappa a}{\eta} \left[4 \exp(a_{t}) + B \exp(-a_{t}) \right] \exp(i_{t} t), \qquad (7.391)$$

Since by symmetry, there is no fluid flow across the center of any layer, $w_i = 0$ at $i = d_1/2$ ($i = d_2/2$) reprive that $B = 1 \exp(i d_1)$ ($B = 1 \exp(i d_2)$). Then, therefore between the fluid pressure p_1 (equation (7.392)) and the Effection velocity \hat{w}_i repution (7.394)) at i = 0 is

$$p_f = I_1 \delta_{(f_1)}$$
 and $p_{f_2} = I_2 \delta_{(f_2)}$ (7.395)

where

$$l_l = \frac{t_0}{s_l dr} \cosh\left(\frac{a_l d_l}{2}\right), \quad l = 1, 2.$$
(7.396)

are the unpedances looking neo medium 1 and medium 2 from the interface (with α given by (7.393)), and the superscript mians sign indicates that the fluid pressure corresponds to the diffusive mode.

According to equation (7.385), there is a fluid-pressure difference between the porous modia, but the total pressure $\mu_{f}^{2} + \mu_{f}$ and \dot{w}_{s} should be continuous at the interface Contributly of pow pressure is achieved by the generation of a slow P wave which diffuses away from the interface. These conditions, tagether with equations (7.385) and (7.395), imply that the fluid particle velocity at the interface is

$$e = i e_{\delta} - \begin{pmatrix} i_{I} & i_{I} \\ I_{I} + I_{I} \end{pmatrix} \phi_{e}, \quad i_{I} - \frac{i g M_{I}}{F_{eg}}, \quad l = 1, 2.$$
(7.397)

As fluid flows out of medium 1, for instance, the the kness of layer 1 decreases while that of medium 2 increases. According to equation (7.388), the matrix displacement due to the fluid flow is $w_3^{(n)} = -irr[\psi_1/(6\pi)]$. Therefore, the displacement fields related to this "unloading" and "hading" metions are

$$u_1 = v_1^{(m)} + \frac{r^{(n)}}{i\omega}$$
 and $u_2 = v_1^{(r)} v_2 = \frac{r^{(n)}}{i\omega}$. (7.398)

respectively. The sum of the displacements divided by the thickness of the elementary volumens the strain due to the fluid flow

$$\phi = 2 \begin{pmatrix} u_1 + u_2 \\ d_1 + d_2 \end{pmatrix} - \frac{2(r_2 - r_1)^2 \sigma_0}{(1/4 + d_2)(I_1 + I_2)}$$
(7.399)

where we have used equations (7.385), (7.397) and (7.398). The total strain is $r = \epsilon_0 + \epsilon_f$ and equations (7.379), (7.389) and (7.399) yield the P wave complex modulus

$$\mathcal{E} = \begin{bmatrix} 1 & -2(r_{i} - r_{i})^{2} \\ \mathcal{E}_{i} & -i_{\pm}(d_{i} + d_{i})(I_{i} + I_{i}) \end{bmatrix} \quad .$$
(7.400)

The approximate transition frequency separating the relaxed and unrelaxed states (i.e., the approximate location of the relaxation peaks is

$$f_{m} = \frac{8c/K_{T}}{(\eta/d')} \quad (7.401)$$

(Durta and Seriff 1979), where the subindex 1 refers to water for a layered medium alternately saturated with water and gas. At this reference frequency, the Hiot slowwave attenuation length equals the mean layer thickness or characteristic length of the inhomogeneities (Garevich and Loparnikov, 1995) (see next paragraph). Equation (7,101) indicates that the mesoscopic loss mechanism moves towards the low frequencies with increasing viscosity and decreasing permeability, i.e., the opposite behaviour of the Bost relaxation mechanism whose peak frequency is given by equation (7.378).

The mesoscopic loss mechanism is due to the pressure of the Biot slow wave and the diffusivity constant is $d = \pi K_F/\eta$, according to repeation (7.391). Note that the same result has been obtained in Section 7.7.1 (optation 7.300)). The critical fluid-diffusion relaxation length *L* is obtained by setting $|u\rangle = |uL| = 1$ in equation (7.391). It gives $L = \sqrt{d/\omega}$. The fluid pressures will be equilibrated if *L* is comparable to the period of the statification. For smaller diffusion lengths to *u*, higher frequences) the pressures will not be equilibrated, causing attemption and velocity dispersion. Notice that the reference frequency (7.401) is obtained when fin a diffusion length L = d/4.

Let us assume that the properties of the frame are the same in media 1 and 2. At enough law frequencies, the fluid pressure is uniform (isostress static) and the effective medials of the pore fluid is given by Word's equation (Word, 1955):

$$\frac{3}{K_{T}} = \frac{p_{0}}{K_{T}} + \frac{p_{0}}{K_{L}}, \quad (7.502)$$

It can be shown to \mathbf{g} = holoword, 2001) that $\mathcal{E}(\mathbf{x}) = 0$) is equal to Gassmann's modulus (7.33) for a fluid whose annihilds is K_T . On the other hand, at high frequencies, the pressure is not aniform but can be assumed to be constant within each phase. In such a situation Hill's theorem (Hill, 1964) gives the high frequency limit $\mathcal{E}(\omega) = \infty_0 + \mathcal{E}_{\omega}$. As an example, Figure 7.11 shows the phase velocity and dissipation factor as a function of frequency for a finely layered medium saturated with water and gas.



Figure 7.54: Phase velocity to the designation for the state function of frequency for a basis beyond wishing subtracted with which that gas. The finite is the same for media 1 and 2 with $\phi = 1.3$, $K_{m} = 1.3$ GPs, $\mu_{m} = 1.4$ GPs and $\kappa_{m} = 1.0$ are the gram productive media 1 and 2 with $\phi = 1.3$, $K_{m} = 1.3$ GPs, $\mu_{m} = 1.4$ GPs and $\kappa_{m} = 1.0$ are the gram productive media 4.5 GPs and $\rho_{m} = 2650$, $K_{M} = 1.0$ in Early proposities are $K_{M} = 2.2$ GPs, $\mu_{m} = 0.5$ Kg/m², $\eta_{0} = 1.15$ Kg = 0.16 eV, $\mu_{0} = 2.5$ kg/m² and $\eta_{0} = 0.15$ ePs (2 kg/m²) and $\eta_{0} = 0.15$ ePs

The misoscopic loss throw have been further refined by Norris (1993). Gurevich and Lupatalkov (1995). Gelinsky and Shapiro (1997). Johnson (2001). Pride, Berryman and Harris (2003) and Miller and Garevich (2005). Johnson (2001) developed a generalization of White model for parches of arbitrary shape. This model has two geometrical parameters, bosides the usual parameters of Boot's theory: the specific surface area and the size of the parches. Patche saturation effects on acoustic properties have been observed by Murphy (1982). Knight and Nulen-Hucksema (1990) and King, Mansden, and Demis (2000) Calmer. Marion and Zinszner (1995) investigated the phenomenes in the laboratory at the frequency range 1-500 kHz. Two different saturation in theds result in different fluid distributions and give two different values of velocity for the same saturation. Imbibilition by depresentization produces a very homogeneous saturation, while drainage by drying produces heterogeneous saturations at high water saturation, while drainage by drying produces heterogeneous saturations at high water saturation while drainage by drying produces heterogeneous saturations at high water saturation. While drainage by drying produces heterogeneous saturations at high water saturation and levels. In the latter rase, the experiments show considerably higher vehicities, as predicted by White model. Carcione, Helle and Pham (2003) performed memorical-modeling experiments based on Biot's equations of periodiasticity (Carcione and Holle, 1969) and White model of regularly distributed spherical gas inclusions. They showed that attraviation and velocity dispersion measurements can be explained by the combined effect of mesoscopic scale in homogenetics and energy transfer between wave modes. By using computerized tomography (C1) scales it is possible to visualize the Heid distribution and spatial heterogenetics in real rocks. (Coloret: Marion and Zeiszner, 1995). Fractal models, such as the von Karmin correlation function, calibrated by the C1 scales, were used to Helle, Pham and Carcione (2003) to model heterogeneous cork properties and perform numerical experiments based on Biot's reparings of priordasticity. These simulations show that Biot's theory gives encort attenuation levels when using heterogeneous models.

The mesoscopic loss mechanism indicates that information about the periodability of the rock – an important properties in hydrocarbon exploration –, is present in the seismic amplitudes the diffusion length is proportional to $\sqrt{20}$. Therefore, measurements of the quality factor at low pressuint frequencies new provide useful information about the structure of the bast reservoir tork.

7.11 Green's function for poro-viscoaconstic media

Green's functions for priordenstic media are studied by several authors: Deriviewicz and Rice (1962), Burridge and Vargas (1979), Nurris (1985, 1994), Burtin, Bouter and Bard (1987), Pride and Haartsen (1966) and Sahay (1999). Bourin, Bourier and Bard apply the theory of Auriault, Horne and Chambon (1985) and compute scatt-analytical transient solutions in a stratified medium. Bounet (1987) obtains a solution by applying the weakogy between the provelastic and thermoelastic equations (Norres (1993), and Kazi Aonal Bounet and Junania (1988) extend the solution of Burtin. Bounet and Bard (1987) to the transversely isotropic case.

Here, we obtain an analytical transient solution for propagation of compressional waves in a homogeneous porous dissipative we durin. The solution, based on a generalization of Brot's periodistic equations, holds for the low- and high-frequency ranges, and includes viscoelastic phenomeno of a general nature, heades Biot's relaxation mechanism. We consider the paradometic version of Biot's espatians, i.e., with the rightly of the mature equal to zero?". These equations may describe wave motion in a colloid that can be considered either an emulsion of a gel. On one hand, it is an emulsion since shear waves do not propagate. On the other hand, since the "frame" modulus is different from zero, the "solid" component provides a sufficient structural framework for rigidity, and, therefore, can be considered as a gel.

7.11.1 Field equations

The pore viscoaronstic model is dilatational, which implies no shear deformations. No shear deformations are obtained by setting $\mu_m = 0$ in reptation (7.19). Moreover, using equations (7.22) and (7.32), we can write the stress-strain relations as

$$p \ge H u_{m} = C_{wi}$$
 (7.403)

¹⁴ In principle, the rescharges an includization since the bijlk module should also constrain the case.

and

$$p_T = Co_m + M_{\chi}$$
(7.504)

where $p = -\sigma_n/3$ is the hulk pressure, H = K + R + 2C and $C = \sigma M$ with K, R and M defined in equations (7.16), (7.18) and (7.24), respectively. Equatines (7.163) and (7.364) can be seen as the stress strain relation in the frequency domain. Thus, invoking the correspondence principle (see Section 3.6), the stillnesses become complex and depend on the angular frequency π_n . Let us assume that H, C and M are appropriate complex moduli, describing viscondustic behavior, such that the expressions given in equations (7.16), (7.18) and (7.24) correspond to the high-frequency closeless) finit.

It is convenient to express reprations (7,403) and (7,404) in matrix form as

$$\begin{pmatrix} p \\ p_f \end{pmatrix} = \begin{pmatrix} -H - C \\ -C - M \end{pmatrix} \cdot \begin{pmatrix} d_{m} \\ s \end{pmatrix}$$
(7.405)

on in marpart notations

$$\mu = P \cdot e_{c}$$
 (7.309)

where P is the couplex stiffness matrix.

The dynamical equations (7.210) and (7.213), restricted to the viscoacoustic case and considering a general viscodynamic operator **b** , are

$$\nabla(p - s_0) = p \partial_0 \mathbf{u}^{(m)} + \partial p \partial_0^2 \mathbf{w}$$
 (7.407)

and

$$-\nabla(p_T + s_T) = p_T \partial_0^2 a^{(m)} + Y * \partial_t \mathbf{w}, \qquad (7.308)$$

where w^{m} is the average displacement of the solid, and w is the average displacement of the fluid relative to the solid (7.172). The quantities κ_{i} and κ_{j} are body forces acting on the bulk material and on the fluid phase, respectively.

For harmonic oscillations, opportunis: 7/407) and 17/408) can alternatively be written as

$$\nabla(\mathbf{p} - \mathbf{s}) = -\omega^2 \mathbf{T} \cdot \begin{pmatrix} \mathbf{u}^* \\ -\mathbf{w} \end{pmatrix}, \qquad (7.469)$$

where

$$s = (s_{\theta_{0}} | s_{f})$$
 (7.410)

and

$$\mathbf{r} \sim \begin{pmatrix} -\rho & \rho_f \\ -\rho_f & \mathcal{F}_f(\omega) \end{pmatrix}$$
(7.401)

is the viscodynamic matrix, and Y is the Fomier transform of Y.

7.11.2 The solution

Taking the divergence in equation (7.469) and assuming a homogeneous machine, we can write

$$\Delta(\mathbf{p} \otimes \mathbf{s}) \leq (\omega / \mathbf{\Gamma} \otimes \mathbf{e})$$
 (7.412)

where

$$\mathbf{e} \geq \begin{pmatrix} u_{n} \\ \ddots \end{pmatrix} \approx \operatorname{div} \begin{pmatrix} \mathbf{u}_{n} \\ -\mathbf{w} \end{pmatrix}$$
 (13)

and Δ is the haplacian operator. Substituting the constitutive law (7.406), we have that equation (7.412) becomes

$$\Delta(\mathbf{p} - \mathbf{s}) + \varphi^2 \mathbf{D} \cdot \mathbf{p} \leq 0,$$
 (7.13.3)

where

$$\mathbf{D} = \mathbf{\Gamma} (|\mathbf{P}||^{3})$$
 (7.415)

Note that **D** is a complex function of the frequency and does not depend on the position vector since the medium is homogeneous. This matrix may be decomposed as **D** $\mathbf{A} = \mathbf{A} = \mathbf{A}^{-1}$, where **A** is the coagonal matrix of the eigenvalues, and **A** is a centric whose rolumns are the right eigenvectors. Thus, substituting this decomposition into equation (7.411) and multiplying by $\mathbf{A} = 1$ from the left hand side, we get

$$\Delta(\mathbf{v} - \mathbf{f}) + \varphi' \mathbf{A} \cdot \mathbf{v} \ge 0, \quad (7.166)$$

where

$$\mathbf{x} = 0$$
, $p_0 = \mathbf{A}^{-1}$ (p. (7.117))

and

$$F = (f_{11} f_{12} - A^{-1}) s$$
 (7.138)

From (7.116), we get the following Helmholtz explations for the components of v.

$$(\Delta + \omega' \lambda_c) e_{\nu} = \Delta f_{c2} = 6 - 1.2.$$
 (7.119)

where λ_1 and λ_2 are the eigenvalues of **D**. They are related to the complex velocities of the fast and slow compressional waves. In fact, let us assume that a solution to explation (7/414) is of the form

$$\mathbf{p} = \mathbf{p}_{1}(\mathbf{x}\mathbf{p}_{1} + \mathbf{i}\mathbf{k} \cdot \mathbf{x}), \quad (5.120)$$

where \mathbf{x} is the position vector and \mathbf{k} is the complex wavevector. Putting this solution into equation (7.414) with zero body forces, and setting the differentiant to zero, we obtain the dispersion relation

$$\det \left[\mathbf{D} - \begin{pmatrix} k \\ -\ell \end{pmatrix}^T \mathbf{D} \right] = 0. \tag{7.421}$$

Since $\pm k < n$ is the complex velocity, the eigenvalues of **D** are $\lambda = 1/r_t^2$. Because **D** is a second-tank matrix, two modes, corresponding to the last and slow waves, propagate in the medium. A simplified expression for the eigenvalues is

$$\lambda \sim -\frac{1}{2\operatorname{dot}\mathbf{P}}\left(t \approx \sqrt{t}\,t = \operatorname{tdet}\mathbf{P}\operatorname{det}\mathbf{\Gamma}\right),$$
(7.122)

where

$$U = 2p(\delta) - pM = H(Y/(\omega))$$
 (7.423)

The phase velocities (7.285) calculated with the solutions (7.122) correspond to the solution of expansions (7.287) with $y_{m} = 0$, and equation (7.290) if, modelmone y = 0.

Considering that the solution for the Green function the , the right-hand side of (7,419) is a space delta function at, say, the origin, both equations have the form

$$(1A + z^2A)q = -89(x),$$
 (7.124)

where d is Dirac's function. The 24D solution (line source) of equation (7,124) is

$$q(r,\omega) = -2(H_0) \left[\omega_T \sqrt{\lambda(\omega)}\right]$$
(7.425)

(Pilant, 1979, p. 57), where $H_{i}^{(2)}$ is the Hackel function of the second kind, and

$$c_{ee}\sqrt{x^2 + d_e} = -7.4261$$
(7.426)

The 3-D substitut (point source) of (7,424) is

$$\langle g_z r, \omega \rangle = \frac{2}{\pi r} \exp\left[-i\omega r \sqrt{3} (\omega)\right]$$
 (7.427)

(Pilant, 1979, p. 64), where

$$\gamma = \sqrt{x^2 + y^2 + z^2}$$
 (7.428)

The solutions (7,125) and (7,127) as given by Phant (1979) hold only for real arguments. However, by invokum the correspondence principle (Section 3.6), complex, frequencydependent material properties can be considered. For instance, the poroelastic equations without the Biot mechanism (i.e. $\eta = -0$) have a real **D** matrix, whose eigenvalues are also real. The velocities are real and frequency independent, without dispersion effects. The introduction of the Bint mechanism, via the correspondence principle, implies the substitution $m \mapsto \cdots P(m)/i\omega$. In the same way, viscoelastic phenomena of a more general nature van be modeled.

The solution of equation (7.419), with the band-limited sources f_1 and f_2 , is then

$$c_r = f_r \Delta g(\lambda_r) = f_s G(\lambda_s),$$
 (7.429)

where

$$G(\Lambda_{0}) = \omega^{2} \Lambda_{0} g(\Lambda_{0}) + 850 \mathbf{x}_{10}$$
(7.430)

and equation (7,124) has been used. In equation (7,129), we introduced the source vector

$$f = (f_{+}, f_{f}) + \mathbf{A} = (\hat{s}, h(\varphi))$$
 (7.431)

where

is a constant vector and $h(\omega)$ is the frequency spectrum of the source.

The vector **p** is obtained from equation (7.417) and written as

$$\mathbf{p}(r,\omega) = \mathbf{A}(\omega)\mathbf{v}(r,\omega) \tag{7.333}$$

From the form of $\epsilon_{\rm c}$ and $\epsilon_{\rm c}$ in equation (7, 129) and using (7, 131), we can explicitly write the solution as

$$\mathbf{p} = \mathbf{A} \cdot \begin{pmatrix} G_1 \lambda_1 & 0 \\ 0 & G(\lambda_2) \end{pmatrix} \cdot \mathbf{A} \quad \text{s} \ h.$$
(7.431)

Using $\mathbf{D} = \mathbf{A} \cdot \mathbf{A} + \mathbf{A}^{\dagger}$, and from the theory of functions of matrices (Lamaster and Fishencetsky, 1985, p. 311), equation (7,134) hereovers

$$(p = G(D) \cdot sh)$$
 (7.435)

where $O(\mathbf{D})$ can be viewed as the evolution operator for Green's function) of the system. An effective numerical implementation of the evolution operator is obtained by decomposing it into its Lagrange interpolator chancaster and Tiszennetsky, 1985, p. 308). This yields

$$G(\mathbf{D}) = \frac{1}{\lambda_0 - \lambda_0} \left\{ G(\lambda_1) - G(\lambda_0) [\mathbf{D} + \lambda_1 G(\lambda_0) - \lambda_0 G(\lambda_0)] \mathbf{I}_0 \right\}$$
(7.036)

This expression avoids the calculations of the eigenvectors of **D** (i.e., of matrix **A**). Using equation (7.440) and the complex velocities $v_{ij} = 1/\sqrt{\lambda_i}$, v = 1, 2, we note that equation (7.446) becomes

$$G(\mathbf{D}) = \frac{w^2}{v_1^2 v_2^2} \left\{ p_1^2 \left[q(v_1) - r_1^2 g(r_2) \right] \mathbf{D} + q(v_2) - q(v_2) \mathbf{I}_2 \right\} - \frac{8\delta^2 \mathbf{x} (\mathbf{I}_2)}{\varepsilon^2} = (7.037) \right\}$$

In the absence of viscoelastic dissipation and with the Biot mechanism deactivated elect fluid viscosity), only the Green functions (7.425) and (7.427) are frequency degendent the eigenvalues of \mathbf{D} are real. Let us denote the phase velocities of the last and slow waves as ψ_{n-1} and ψ_{n-1} respectively (as in equation (7.290)). Then, the explicit frequency dependence of the evolution operator is

$$G(\mathbf{D}, \varphi) = \frac{G(r_{\infty}, \varphi\omega)}{(r_{\infty}, \varphi\omega)} \frac{\omega_{2}}{(1 - 1)} (r_{\infty}^{2}, \mathbf{D} - \mathbf{I}_{2}) - \frac{G(r_{\infty}, \varphi-1)}{1 - (r_{\infty}, \varphi\omega)} (r_{\infty}^{2}, \mathbf{D} - \mathbf{I}_{2}), \quad (5.638)$$

In this case, the solution can be obtained in closed form since the Green functions (7,125) and (7,127, can be Fourier transformed analytically to the time domain (Norris, 1985).

to ensure a time-domain real solution or the general viscoelastic case, we take

$$\mathbf{p}(r, \omega) = \mathbf{p}^{*}(r, -\omega),$$
 (7.439)

for $\omega < 0$, where the superscript * denotes the complex conjugate. Finally, the two domain solution is obtained by an inverse function transform

An example is shown in Figure 7.12 (see Carcione and Quiroga-Guode (1926) for details about the matricial properties and source characteristics). When the fluid is viscous chough, the slow wave appears as a quasi-static mode at the source location. This behavior is predicted by the analytical solution, where snapshots of the solid and fluid pressures due to a fluid volume injection are represented. The frequency hand corresponds to the source rate:

7.12 Green's function at a fluid/porous medium interface

The Rayleigh wave in a portous medium is composed of the fast P wave, the shear wave and the slow P wave. The physics has been studied by Denssowaz (1962), who found that the Rayleigh wave is dissipative and chapersive due to lasses by mode conversion to the slow wave (e.g., Buarbić, Cunssy and Zinszeier, 1987). Surface waves at liquid-purious randin interfaces classify into three kinds. A true surface wave that travels slower than all the wave velocities (the generalization of the Schulte wave), a pseudo Schulte wave that travels



Figure 7.12. Snapshow of the solid pressure on and find pressure do for a porote module saturated with a viscous fluid. The correct during interpreters is at the some range. The court of the source result must be some range. The court of the source result must be shown beaver, which behaves as a quasi state mode at these frequencies.

with a velocity between the shear-wave velocity and the slow-wave velocity (haking energy to the slow wave), and a pseudo Rayleigh wave, which becomes the classical Rayleigh wave if the liquid density goes to zero (Feng and Johnson, 1982a,b, Holland, 1984, Edebian and Wilmanski, 2002). For scaled-pore conductors the true surface wave exists for all values of material parameters. Nagy (1992) and Adler and Nagy (1994) observed this surface wave in alcohol-saturated porous surface glass and natural tooks. The conditions are a highly compressible third (e.g., air), a clusted surface iscaled pores due to surface truston in Nagy's experiments) and negligible viscoisty of the saturating theil.

According to equation (7.375), the boundary conditions at an interface between a potent medium and a fluid are

$$\phi_{i}(e_{1}^{(f)} - e_{2}^{(m)i}) = e_{1}^{(f)} - e_{1}^{(m)} \quad , \quad p_{f}^{(f)} - p_{f}^{(i)} = \frac{1}{h_{i}} d(w_{i}, -w_{i}) = -p_{f}^{(f)}, \quad \phi_{1i} = a_{21}^{(i)} = 0,$$
(7.110)

where we have considered the general case given by equation (7.367). The two limiting rases are equation (7.375) topen porest and $\epsilon_s = 0$, which corresponds to scaled pores. In this case, there is no relative flow arrows the interface and the hoppday conditions are

$$r_{4}^{(2)} = r_{1}^{(m-1)} + r_{2}^{(d-1)} + r_{3}^{(m-1)} + \sigma_{33} + p_{2}^{(d)} + \sigma_{33} + \sigma_{33}^{(1)} = 0,$$
 (7.141)

Fing and Johnson (1982b) obtained the high-frequency 2-D Green's function using the Cogmand-de Hoop technique. The source is a radial and uniform impulsive line source (a pressure source so in the fluid). If we assure that the upper medium is the fluid, the locations of the source and receiver are (0, z) and (z =) above the interface in the uverlying fluid fash space. The Green function is

$$G(r_{t},r_{t},t) = \begin{cases} 0, & \infty \in t < t_{k}, \\ \ln \left(\frac{R_{f}(s_{n})}{2\pi\sqrt{t_{k}^{2} - t^{2}}}\right), & t_{k} < t < t_{n}, \\ -R_{t}\left(\frac{R_{f}(s_{n})}{2\pi\sqrt{t_{k}^{2} - t^{2}}}\right), & t > t_{n}, \end{cases}$$
(7.142)

where R_{I} is the reflection coefficient,

$$s_{1}(t) = rac{xt}{x} - \frac{(z + z_{0})\sqrt{t(-t)}}{x - (z + z_{0})^{2}},$$
 (7.113)

$$s_{\lambda}(t) = \frac{xt - i(--z_{0})\sqrt{t^{2} - t_{0}^{2}}}{x^{2} - (z - t_{0})^{2}},$$
 (7.113)

$$E_0 \approx \frac{\pi}{c_{N,0}} + 1_{\gamma} \approx \pi i \sqrt{\frac{1}{c_{N,0}^2}} + \frac{1}{c_{N,0}^2},$$
 (5.145)

and

$$\sin \sqrt{n} + (1 + 2^{2} hy)$$
 (5.146)

 (v_I) is the wave-velocity of the fluid, $v_{\infty,i}$ is the high-frequency fast P-wave velocity, a solution of equation (7.292), s_0 and s_0 are slown sees encosponding to the head and holy wave, and t_0 and t_0 are the respective arrival times).


Figure 7.13. Calculated 2 D Green's functions for the water better saturated fitsed glass basis planar interface system, the source and receivers are ideally located on the surface (z = z = 0) and z = 10or , frictoris, $z = \infty$, corresponding to oper tools, and z = 0, is the specifing to scaled power. The resource properties on $K_{\pm} = 59.9$ GPs, $p_{\pm} = 2480$ kg/m², $K_{m} = 6.1$ GPa $p_{m} = 0.1$ GPa $\phi = 0.38$, T =0.38 Ky = 2.20 GPa and $p_{\pm} = 0.30$ kg/m². The black ends are the fact P wave (Table 1) the sound norm of the fact of Fhad") the stear cover of Slave 1. On grandor scale wave (128 schulter), the show P wave 5 Slow), and the mate surface wave of The studies in face 5.

The expression of the reflection coefficient is

$$R_f(s) = \frac{\Delta_R(s)}{\Delta_r(s)}, \qquad \Delta_R = \det \mathbf{N}, \qquad \Delta_h = \det (\mathbf{D}, -(7, 147))$$

and, using our notation, the components of matrix Distrate¹⁸

$$\begin{split} D_{1} &= 2p_{0s}s^{2} + (\alpha MT_{s} - E_{0})_{s}v_{s}^{2}, \\ D_{12} &= 2p_{0s}s^{2} + (\alpha MT_{s} - E_{0})_{s}v_{s}^{2}, \\ D_{13} &= p_{T} \\ D_{13} &= p_{T} \\ D_{14} &= \alpha(\gamma - /\kappa_{s} + M/r_{s}^{2})(T_{s} + \alpha M/r_{s}^{2}) \\ D_{15} &= \alpha(\gamma - /\kappa_{s} + M/r_{s}^{2})(T_{s} - \alpha M/r_{s}^{2}) \\ D_{16} &= \alpha(\gamma - /\kappa_{s} + M/r_{s}^{2})(T_{s} - \alpha M/r_{s}^{2}) \\ D_{16} &= \alpha(\gamma - 1)^{2} \\ D_{16} &= \alpha(\gamma - 1)^{2} \\ D_{36} &= -(1 - \alpha/T)^{3} \\ D_{46} &= s^{2} \\ D_{16} &= s^{2} \\ D_{16} &= s^{2} \\ D_{16} &= s^{2} \\ D_{16} &= 0_{1} \end{split}$$
(7.148)

where r_{2} is the high-frequency S-wave velocity (7.314) $|r_{\infty}|$ is the high-frequency slow P-wave velocity a solution of equation (7.292).

$$\gamma_f \approx \sqrt{1/r_f^2 - S_0^2} = 1 \, e^{-f_0} \left(\mathbf{x} \cdot \mathbf{1} - \mathbf{x} \right) - S_0$$
 (7.136)

$$I_{\pm} = \phi - \frac{(1 - \alpha)\rho_{e} + c\rho_{f}(T - 1)\psi_{n}^{+} - E_{m} - (\alpha - \alpha)^{*}M}{\rho_{f}(T - 1)\psi_{n}^{*} - (\alpha - c)M}$$

$$(7.150)$$

and

$$I = \alpha - \frac{[(1 - \alpha)\rho_s - \eta \eta (\mathcal{T} - 1)]r_s^*}{\rho_f(\mathcal{T} - 1)r_s^*} + \frac{F_{\sigma} - (\alpha - \alpha)^2 M}{\rho_f(\mathcal{T} - 1)r_s^*}, \quad (7.151)$$

The elements of N are the same of D except $N_{ij} = \varepsilon_f$.

Figure 7.13 shows the Green functions for open (a) and scaled (b) ports. Note the presence of the slow surface wave, observed by Nagy (1992) at approximately 1.1 µs in the scaled-port case.

7.13 Poro-viscoelasticity

Viscoclasticity can be introduced into Bior's percelastic explaneous for modeling attenuation mechanisms related to the strain energy (stiffness discipation) and the kinetic energy

 $^{(2^{17}}G_{12} + 2\mu_{c}s^{2}) = (0.MT_{c} + E_{c}s^{2})(\pi)$ for \log_{2} and $\log(129820)$. Instead, the sign 2^{12} the second transitional for (2114) . Inducing pressual containing to 2

twistedyname dissipation). In natural porous media such as such tones, discrepancies between Brat's theory and measurements are due to complex pure shapes and the presence of clay. This emiglexity gives use to a variety of relaxation mechanisers that contribute to the attenuation of the different wave modes. Stell and Bayan (1970) show that attenuation is controlled by both the anelasticity of the skeleton (friction at grain contacts) and by viscodynamic causes. Stillness dissipation is described in the stressestrain relation, and viscodynamic causes. Stillness dissipation is described in the stressestrain relation, and viscodynamic clauses. Hund and the solid matrix (Brot, 1956), Johnson, Koplik and Dashen, 1987).

Let us consider, as an example, the 2-D stress-strain relations for an isotroph purorlastic medium in the (x, z) plane. From equations (7.32) and (7.33), and using (7.22), we can rewrite the stress-strain relations as

$$\begin{aligned} \partial_t \sigma &\mapsto E_m \partial_t e_i^{(m)} &\mapsto (E_i - 2\mu_m) \partial_t e_n^{(m)} + (eM) &\mapsto \\ \partial_t \sigma_{M} &= (F_m - 2\mu_i) \partial_t e_i^{(m)} &\mapsto F_m \partial_t e_i^{(m)} + e(M) &\mapsto \\ \partial_t \sigma_{M} &= \mu_i (\partial_t e_i^{(m)} + \partial_t e_n^{(m)}) + s_{1n} \\ \partial_t \mu_i &= -Me + s_0 \\ e_i &= -\alpha (\partial_t e_i^{(m)} + \partial_t e_n^{(m)}) + \partial_t \theta_0 + + \partial_t \theta_0, \end{aligned}$$
(7.552)

where $s_i^{(\ell)}$ and $\dot{w}_i \in \partial m$, are the components of the particle velocities of the solid and flaid relative to the solid (see equation (7.172)), s_i , s_i , s_i , s_i , and s_f are external sources of stress for the solid and the fluid, respectively, and M_i is and E_m are given in equations (7.24), (7.25) and (7.291).

The 2-D povodastic equations of motion can be obtained from (7.210) and (7.211):

it Biot-Euler's dynamical equations:

$$\begin{aligned} \partial_t \sigma_1 &= \partial_x e_0 = \rho \partial_x e_0^{(0)} + \rho_1 \partial_t \dot{w}, \\ \partial_t \sigma_{13} &= \partial_t \sigma_{31} - \rho \partial_x e_0^{(1)} + \rho_1 \partial_t w_0. \end{aligned} \tag{7.553}$$

ii) Dynamical Darcy's law:

$$\frac{d(\mu) - \mu a h r_i^{(m)} + m h r_i + h * \partial_i r_i}{d_i a + \mu a h r_i^{(m)} + m h r_i + h * \partial_i r_i}$$
(7.51)

where $m = T_{PP}/\phi_0(\exp(arton(7.212)))$, with T denoting the corticosity, and b(t) a relaxation function $-\Lambda t$ low frequencies $b = H(t)\eta_0 \phi_0$ where H is Heaviside's function, and we obtain (7.351) (Corcions, 1998) Aratises and Corcions, 2006)

The stiffnesses E_{m} , μ_{m} and M are generalized to time dependent relaxation functions, which we denote, in general, by $r \sim R$. We assume that $r(0) = r_{0}$ equals the respective Biot modulus, i.e., we obtain Biot's poroclastic stress-strain relations at high frequencies. Assume, for example, that the relaxation functions are described by a single Zener model.

$$\psi(t) \ge \psi_t \begin{pmatrix} \phi_t \\ -\phi \end{pmatrix} \left[1 + \begin{pmatrix} \phi_t \\ -\phi \end{pmatrix} \exp(-t/\phi_t) \right] H(t)$$
 (3.35)

where π and π_{i} are relaxation times (see Sortion 2.1.3)

We introduce viscoelasticity by replacing the products of the elastic moduli and held variables in equations (7–552) with time-convolutions. For instance, in equations (7–552)

7.13 Pore-viscoelasticity

and (7.152)). these products are $E_{mn}\partial(x_1^m) + \partial_0 v_1^m$, $\mu_m \partial_0 v_n^m$ and M_0 . We replace then with $(|\cdot|, \partial_0)$, where is denotes the relaxation function corresponding to $F_{mn}(\mu_n)$ in M_0 and u denotes $\partial(v_1^m) + \partial_0 v_n^m$, $\mu_m v_1^m$ or v_0 . It is important to point out that this approach is prively phenomenological. As in the single phase viscoelastic case (see Section 2.7), we introduce memory variables to avoid the time convolutions. Then, the terms $v + \partial_0 u$ are substituted by $(v_n + v)$, where v is the memory variable. There are live stress memory variables related to the stress-strain relations, which satisfy the following differential equation:

$$\partial_t c \sim c_0 \left(\frac{1}{n} + \frac{1}{c_0}\right) u = \frac{c}{c_0},$$
(7.456)

I we additional memory variables are introduced via viscodynamic dissipation, due to the time-dependent relaxation function, b.(). Hence,

$$\delta(t) = \frac{g}{\pi} \left[1 + \left(\frac{\gamma}{\gamma_{1}} - 1 \right) (\operatorname{sp}(\cdot)t/\gamma_{1}) \right] H(t), \qquad (7.457)$$

the terms b + che are replaced by $50000 + c_1$ and the memory-variable equations have the form

$$\partial_t \epsilon = -\frac{1}{\gamma_s} \left[\frac{\eta}{s} \left(\frac{\eta}{\gamma_s} + 1 \right) \eta + \epsilon \right], \qquad (7.158)$$

In the frequency domain, the fine convolution $z \sim w$ is replaced by z = w obtain, from (7.155).

$$\psi = \psi_{\alpha} \left(\frac{\tau_{\alpha}}{\tau_{\alpha}} \right) \left(\frac{1 - i\omega_{\alpha} \tau_{\alpha}}{1 + i\omega_{\alpha} \tau_{\alpha}} \right)$$
(7.459)

and each complex modulus is denoted by $\vec{F}_{min}(\mu_{0})$ and \vec{W}

Each set of relaxation times can be expressed in trends of a Q factor Q_i and a reference frequency f_0 as

$$z_{i} = (2\pi f_{i} Q_{i}) - \left(\sqrt{Q_{0}^{2} + 1} + 1\right),$$

 $z_{i} = (2\pi f_{0} Q_{i}) - \left(\sqrt{Q_{i}^{2} + 1} - 1\right)$
(7.460)

On the other hand, the frequency domain viscolynamic operator has the form

$$\dot{h} = \frac{\eta}{\kappa} \left(\frac{1 + i\omega \pi}{1 + i\omega m} \right), \qquad (7.461)$$

The functional dependence of b on \pm is not that predicted by models of dyname fluid flow. Appropriate dynamic periodability functions are given in Section 7.6.2. Here, we intend to model the viscodynamic operator in a corrow band about the central frequency of the source. The advantage of using registion (7,061) is the easy implementation in time-domain momental modeling.

The results of a simulation with Biot's prior lastic theory are plotted in Figure 7.14b, and compared to the experimental inferoscismograms obtained by Kelder and Smerilders (1997), illustrated in Figure 7.14a. The discrepancies with the experimental results are due to the presence of non-Bot attenuation modernisms. Figure 7.14c shows the powvisculastic microscismograms. The relative amplitudes observed are in better agreement with the experiment than these predicted by Biot's theory without visculastic lasses



Figure 7.14: Microseismogram obtained by Kelder and Smeulders (1997) for Nivelsteiner sandstone as a function of the angle of incidence θ (top picture), and numerical microseismograms obtained from Biot's poroelastic theory (a) and Biot's poro-viscoelastic theory (b). The events are the fast compressional wave (FP), the shear wave (S), the first multiple reflection of the fast compressional wave (FFP) and the slow wave (SP).

7.14 Anisotropic pure-viscoclasticity

In many cases, the results obtained with Biot (two-phase) modeling are equal to these obtained with single-phase elastic modeling, mainly at seismic frequencies (Gurevich, 1996). A correct representation is ubtained with a viscodastic cheology that requires one relaxation peak for each Biot (P and S) mechanism. The standard viscoelastic model, that is based on the generalization of the compressibility and shear modelus to relaxation functions, is not appropriate for modeling Biot complex moduli, since Bier's attenuation is of a kinetic nature in e. it is not related to bulk deformations. The problem can be solved by associaving relaxation functions with each wave moduli. However, in a likelity inhomogeneous medium, single-phase viscoelastic modeling is not, in principle, equivalent to phonomedia modeling, due to substantial mode conversion from fast wave to quasistatic mode. For instance, if the fluid enumpressibility varies significantly from point to point, diffusion of precluing atoms different regions constitutes a mechanism that can be important at seisme frequences Dier Section 7.10).

7.14 Anisotropic poro-viscoelasticity

Anisotropic poroclasticity was introduced by Biot (1955, 1956) and Biot and Willis (1957) in terms of bulk parameters of total stress and strain. To our knowledge, Brown and Knimiga (1975) were the first to obtain the material coefficients in terms of the properties of the gram, porofluid and frame (see Section 7.3.3). Later, Carroll (1980). Budnicki (1985) and Thompson and Willis (1991) presented frether minimum hadrial analysis of the stress-strain telations. Chang (1997) related the Hookean constants to the engineering constants – obtained from laboratory measurements – including explicit relations for the orthorhorizon and transferse isotropy material symmetries. Chang's theory assumes that the solid constituent is isotropic and that anisotropy is due to the arrangements of the grains – i.e., the frame is anisotropic. Bereartly, Sahay, Spanos and de la Cruz (2000) used a volume averaging method to ubtain the stress strain relations. Then approach include a difficultiential equation for provsity, which describes the chances in parasity due to varying stress conditions.

Complete experimental data for anisotropic media is scarce. New propagation experiments on real rocks can be found in Lo. Coyrer and Toksöz (1986) and Aoki. Fan and Bramford (1993). Wave propagation in anisotropic predelastic rocks is investigated by Norris (1993). Ben-Menahem and Gibson (1993). Parea (1657) and Gelinsky and Shapiro (1997) and Gelinsky. Shapiro, Müller and Garovick (1998), who study plane layried systems and the effects of anisotropic preneability. Numerical simulations of wave propagation for the transversely isotropic case. (i) rocks and synthetic materials – are given in Carciane (1996b), and a complete analysis in terms of energy is given by Carciane (2001a). The developments in this section follow the last reference (i.e., Carciane, 2001a).

We have shown in Section 1.3.1 that in single-phase anisotropy via achiever racha, the phase vehicity is the projection of the energy-velocity vector into the propagation direction. We have also generalized other similar relations valid in the isotropic viscorlastic case. Here, these relations are further extended for anisotropic processorelastic media for the following reasons. Firstly, they provide a simple and useful means for evaluating the time-averaged kinetics, strains and dissipated-energy densities from the wavenumber, attenuation and energy-flow vectors. Secondly, they can be used to verify the kinematiand dynamic properties on terms of energy of complex porces materials. For instance, the above relation between phase and energy velocities has namediate implications for altrasonic experiments. If a pulse of acoustic energy is radiated by a phonowave transducer, the wave fruit travels along the wavenumber direction, which is mirrial to the transducer surface, but the wave garket much dation envelope travels in the direction of the energy velocity. This means that the receiving transducer must be offset in order to intercept the acoustic pulse, and the corresponding acade is the angle between the wavenumber and energy-velocity vectors. Although that relation between the velocities is well known for anisotropic basiless media (e.g., Auld, 1960), p. 222: equation [14111); it is not immediately evident that it holds for pore-viscoelastic and anisotropic media

In our example later in this chapter, we consider wave propagation in one of the planes of universymmetry of an arthochembic material (human ferroral bone). Bulk viscoelasticity is modeled by using the concept of eigenstrain (see Section 1.1) and the low-frequency viscoelynamic operator is used to model Biot-type dissignation.

7.14.1 Stress-strain relations

The stress-strain relations (7/131) and (7/133) can be rewritten in matrix form as

$$\sigma = C^{0} - e_{c}$$
 (7.362)

where

$$\sigma = (\sigma_1 \cup \sigma_{22}, \sigma_{33}, \sigma_{23}, \sigma_{33}, \sigma_{32}, -\mu_2)^T, \qquad (7.363)$$

is the stress array.

$$e = [e^{(m)}, e^{(m)}_{\mu}, e^{(m)}_{\lambda}, e^{(m)}_{\lambda}, e^{(m)}_{\lambda}, e^{(m)}_{\lambda}, e^{(m)}_{\lambda}, -\zeta)]$$
, (7.464)

is the strain array, with $e_f^{(n)}$ denoting the strain components of the potons frame and ζ the variation of fluid content.

$$\mathbf{C}^{*} = \begin{pmatrix} x_{1}^{0} & e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & M\alpha \\ e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & M\alpha \\ e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & M\alpha \\ e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & M\alpha \\ e_{1}^{0} & e_{1}^{0} & e_{2}^{0} & e_{1}^{0} & e_{1}^{0} & e_{1}^{0} & M\alpha \\ e_{1}^{0} & e_{2}^{0} & e_{2}^{0} & e_{1}^{0} & e_{1}^{0} & e_{2}^{0} & M\alpha \\ e_{1}^{0} & e_{2}^{0} & e_{2}^{0} & e_{1}^{0} & e_{1}^{0} & M\alpha \\ e_{1}^{0} & e_{2}^{0} & e_{2}^{0} & e_{1}^{0} & e_{1}^{0} & M\alpha \\ e_{1}^{0} & e_{2}^{0} & e_{2}^{0} & e_{2}^{0} & e_{1}^{0} & M\alpha \\ M\alpha & M\alpha_{1} & M\alpha_{2} & M\alpha_{3} & M\alpha \\ \end{pmatrix}, \qquad (7.36)$$

where $r_{f,i}^{*}$ are the components of the elasticity matrix of the undrained phones and inc. (7.136).

The magnate of the strain can be written as

$$\partial_{2} \theta = \nabla - (v_{1})$$
 (7.466)

where

$$\mathbf{v} = \left(v_1^{(m)}, v_1^{(m)}, v_2^{(m)}, w_1, w_2, w_3\right)$$
(7.567)

$$\mathbf{V} = \begin{pmatrix} \partial_1 & 0 & 0 & 0 & \partial_1 & \partial_2 & 0 \\ 0 & \partial_2 & 0 & \partial_3 & 0 & \partial_1 & 0 \\ 0 & 0 & \partial_4 & \partial_2 & \partial_1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 0 & \partial_1 \\ 0 & 0 & 0 & 0 & 0 & 0 & \partial_1 \\ 0 & 0 & 0 & 0 & 0 & 0 & \partial_1 \end{pmatrix}, \qquad (7.468)$$

The form (7.466), relating the particle velocities to the strain components, as well as the differential operator (7.468) are generalizations of those used by Aubl (1990a).

Biat (1956) developed a generalization of the stress-stram relations to the visco-lastic case by meaking the correspondence principle and using relaxation functions based on mechanical models of visco-lastic heliavior. Dissipation is due to a variety of an lastic mechanisms. An important mechanism is mesoscopic loss discussed in Section 7.10. Another mechanism is the separa-flow (Hiot, 1962, Dvorkin, Nolen-Hocksonia and Nut, 1994) by which a force applied to the area of contact between two grains produces a displacement of the suproceeding fluid in and out of this area. Since the fluid is viscous, the motion is not extactaneous and energy dissupation occurs. Other important attenuation mechanisms are discussed by Biar (1962). Using the correspondence principle (see Section 3.6), we generalize to relaxation functions the elements of matrix \mathbf{C}^* , and equation (7.462) becomes

$$\sigma = \Psi * o_t e_t$$
 (7.169)

where Ψ as the relaxation matrix. Matrix \mathbb{C}^{n} is obtained from Ψ when $\ell \to 0$ if we consider that Biet's periodistic theory corresponds to the numeric state.

7.14.2 Biot-Euler's equation

In matrix form, opportions (7/255) and (7/258) can be written as

$$\nabla \cdot \sigma = \mathbf{R} \cdot \partial_t \mathbf{v} - \mathbf{f},$$
 (7.170)

where

$$\mathbf{f} = (f_{ij}, f_{ij}, f_{ij}, 0, 0, 0, 0)$$
 (7.171)

is a body force array, and

$$\mathbf{R} = \begin{pmatrix} p & 0 & 0 & p_f & 0 & 0 \\ 0 & p & 0 & 0 & p_f & 0 \\ 0 & 0 & p & 0 & 0 & p_f \\ p_r & 0 & 0 & r_1 * & 0 & 0 \\ 0 & p_f & 0 & 0 & r_2 * & 0 \\ 0 & 0 & p_f & 0 & 0 & r_1 * \end{pmatrix}$$

$$(7.472)$$

is the density matrix operator. We refer to (7,470) as Biot-Euler's equation.

7.14.3 Time-harmonic fields

Let us consider a time-harmonic field exprime, where ω is the angular frequency. The stress-strain relation (7.469) becomes

$$\sigma = \mathbf{P} \cdot \mathbf{e} = -\mathbf{P} - \mathcal{F}(\partial_t \Psi),$$
 (7.153)

where \mathbf{P} is the complex and frequency-dependent stillness matrix, and the operator \mathcal{F} denotes time former transform. Equation (7,466) becomes

$$a_{0}v = \nabla - v$$
 (7.473)

Substituting equation (7,174) into (7,473), we obtain

$$i\omega\sigma = \mathbf{P} \cdot (\nabla \rightarrow \mathbf{v}),$$
 (7.375)

For the charmonic fields, Biot-Edict's equation (7,170) becomes

$$\nabla \cdot \boldsymbol{\sigma} = i_{\pi} \mathbf{R} \cdot \boldsymbol{v} + \mathbf{f},$$
 (7.476)

.

where

$$\mathbf{R} = \begin{pmatrix} p & 0 & 0 & \rho_I & 0 & 0 \\ 0 & p & 0 & 0 & \rho_I & 0 \\ 0 & 0 & p & 0 & 0 & \rho_I \\ \rho_I & 0 & 0 & Y_1/(1-1) & 0 & 0 \\ 0 & \rho_I & 0 & 0 & Y_0/(1-1) & 0 \\ 0 & 0 & \rho_I & 0 & 0 & Y_0/(1-1) \end{pmatrix}, \qquad (7.377)$$

.

and U are given in equation (7.238), provided that the currection (7.24) t is used for high frequencies, of the specific of ratio is obtained by experimental measurements.

The derivation of the energy-balance equation is straightforward when using complex notation. The procedure even in Section 1.3.1 for single-phase media is used here. The dot product of the complex conjugate of equation (7.474) with $-\sigma^2$ gives

$$\sigma = \nabla (\langle \mathbf{v}' \rangle - \mathbf{i}_{\perp} \sigma \rangle \langle \mathbf{e}' \rangle$$
 (7.478)

On the other hand, the dot product of $(\sqrt{2})$ with equation (7.476) is

$$-\mathbf{v}^* \otimes \nabla \otimes \mathbf{\sigma} = -i\omega \mathbf{v}^* \otimes \mathbf{R} \otimes \mathbf{v} + \mathbf{v}^* \otimes \mathbf{f},$$
 (7.479)

Adding operations (7.478) and (7.479), we get

$$\boldsymbol{\sigma}^{\dagger} = \nabla^{\dagger} \cdot \mathbf{v}^{\star} - \mathbf{v}^{\dagger} \cdot \nabla \cdot \boldsymbol{\sigma} - (\omega \boldsymbol{\sigma}^{\dagger} \cdot \mathbf{e}^{\star} - (\omega \mathbf{v}^{\star} - \mathbf{R} \cdot \mathbf{v} - \mathbf{v}^{\star})^{\dagger} \cdot \mathbf{f}$$
(7.580)

The left-hand side is simply

$$\sigma \rightarrow \nabla \rightarrow v^* \rightarrow v^* \rightarrow \nabla + \sigma = 2 \text{ div } p.$$
 (7.481)

where,

$$\mathbf{p} \leftarrow -\frac{1}{2} \begin{pmatrix} \sigma_{12} & \sigma_{12} & \sigma_{13} & p_1^2 & 0 & 0 \\ \sigma_{13} & \sigma_{32} & \sigma_{33} & 0 & p_1^2 & 0 \\ \sigma_{13} & \sigma_{33} & \sigma_{33} & 0 & 0 & p_2^2 \end{pmatrix} \cdot \mathbf{v}^2$$
(7.482)

is the complex Uniov-Dovaring vector, Using (7, 2011 and the stress-strain relation [7, 1731. we find that opportunit? 1800 gives

> 2 do to the (P et novi R v vi f. 17 (83)

where we used the fact that ${f P}$ is a symmetric factory. Equation (7.483) can be rewritten -15

$$\operatorname{div} \mathbf{p} \approx 2i\omega \left[\frac{1}{4} \operatorname{Re} \left[\mathbf{e}^{+} + \mathbf{P} \cdot \mathbf{e}^{*} \right] + \frac{1}{4} \operatorname{Re} \left[\mathbf{v}^{+} + \mathbf{R} \cdot \mathbf{v} \right] \right]$$
$$\approx 2\omega \left[-\frac{1}{4} \operatorname{Int} \mathbf{e}^{-} + \mathbf{P} \cdot \mathbf{e}^{*} \right] + \frac{1}{4} \operatorname{Int} \left[\mathbf{v}^{+} + \mathbf{R} \cdot \mathbf{v} \right] - \frac{1}{2} \mathbf{v}^{+} + \mathbf{f}, \qquad (7.584)$$

7.14 Anisotropic pure-viscoclasticity

The significance of this equation becomes clear when we recognize that each of us terms has a process physical meaning on a zone average basis. When using complex notation for plane waves, the field variables are obtained as this real part of the corresponding complex fields.

In the following derivation, we use the properties (1.105) and (1.105). Using these relations, we identify

$$\frac{1}{4} \operatorname{Re}(\mathbf{e}^{2} + \mathbf{P} + \mathbf{e}^{*}) = \frac{1}{2} \langle \operatorname{Re}(\mathbf{e}^{-1} + \operatorname{Re}(\mathbf{P}) + \operatorname{Re}(\mathbf{e})) - \langle U \rangle$$
(7.485)

as the strain-energy density.

$$\frac{1}{4} \operatorname{Re}(\mathbf{v}^* \to \mathbf{R} \cdot \mathbf{v}) = \frac{1}{2} \langle \operatorname{Re}(\mathbf{v}^*) + \operatorname{Re}(\mathbf{R}) \cdot \operatorname{Re}(\mathbf{v}^*) \rangle = \langle T \rangle$$
(7.486)

as the kinetic-energy density.

$$\frac{1}{2}$$
... hute $[> \mathbf{P} \cdot \mathbf{e}^*) > \frac{1}{2}$... hut $\mathbf{v}^* \ge \mathbf{R} \cdot \mathbf{v}$.

 $\omega(\operatorname{Rete}^{2}) + \operatorname{Im}(\mathbf{P}) \cdot \operatorname{Rete}) + \omega(\operatorname{Rete}^{2}) \cdot \operatorname{Im}(\mathbf{R}) \cdot \operatorname{Rete}^{2}) = \langle D_{0} \rangle - \langle D_{1} \rangle - \langle T(\mathbf{S}) \rangle$

as minus the rate of dissipated strans-curryy density $1 - (\hat{D}_k)$, the first (ever) minus the rate of dissipated kinetic energy density $(-D_k)$, the second term), and

$$\frac{1}{2}\mathbf{x}^* \ge \mathbf{f} = P, \quad (7.488)$$

as the complex power per unit volume supplied by the body lorges.

We may define the entresponding time-averaged energy densities $\langle D_1 \rangle$ and $\langle D_1 \rangle$ by the relations

$$\langle \hat{D}_V \rangle = \omega \langle D_V \rangle$$
 and $\langle \hat{D}_V \rangle = \omega \langle D_I \rangle$. (7.189)

Substituting the preceding expressions introducation (7,484), we obtain the energy balance equation:

$$d(\mathbf{v} | \mathbf{p} - 2(\omega_0 V) - \langle T \rangle) + \omega_0 D) \approx P_0.$$
 (7.490)

where

$$(D_1 = (D_1) + (D_1)$$
 (7.191)

is the total time averaged dissipated energy density.

The total stored energy density is

$$\langle F \rangle = \langle V \rangle + \langle T \rangle$$
 (7.492)

If there is no dissipation $(\langle D \rangle \simeq 0)$ and, since in the absence of sources $(P_i \approx 0)$ the net energy flow into or out of a given closed surface must vanish, div $\mathbf{p} = 0$. Thus, the average kinetic energy equals the average strain energy. As a consequence, the stored energy is two the strain energy

7.14.4 Inhomogeneous plane waves

A general plane-wave solution for the particle velocity (7.467) is

$$\mathbf{v} = \mathbf{v} \exp[i(\omega) - \mathbf{k} \cdot \mathbf{x})],$$
 (7.493)

where \mathbf{v}_i represents a constant complex vector and \mathbf{k} is the wavevector $-\mathbf{i}$ bis is, in general, complex and can be written as

$$\mathbf{k} = \mathbf{\kappa} - \mathbf{m} = (k_1, k_2, k_3),$$
 (7.394)

where κ and α are the real wavevector and attemption vector, respectively. They indicate the directions and magnetules of propagation and attemption. In general, these directions differ and the place wave is called inhomogeneous. For inhomogeneous viscorlastic plane waves, the operator (7, 9(8) takes the form

$$\nabla \rightarrow -iK$$
, (7.495)

where

$$\mathbf{K} = \begin{pmatrix} F & 0 & 0 & 0 & F_{2} & k_{2} & 0 \\ 0 & k_{2} & 0 & k_{3} & 0 & k & 0 \\ 0 & 0 & f_{1} & k_{2} & f_{1} & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & k_{1} \\ 0 & 0 & 0 & 0 & 0 & f_{1} \\ 0 & 0 & 0 & 0 & 0 & F_{1} \end{pmatrix}$$
(7.395)

When the operator is applied to a conjugated field, ∇ should be replaced by iK^{*}.

Substituting the differential operator interceptations (7,478) and (7,479) and assuming zeto body forces, we get

$$\sigma \rightarrow K^{*} \rightarrow v^{*} - \omega \sigma \rightarrow e^{*}$$
 (7.207)

and

$$-\mathbf{v}^{t} \rightarrow \mathbf{K} \cdot \boldsymbol{\sigma} \approx \omega \mathbf{v}^{t-1} \cdot \mathbf{R} \cdot \mathbf{v},$$
 (7.498)

respectively. The left-hand sides of equations (7,497) and (7,208) contain the complex Unive-Poyntine vector (7,482). In fact, by virtue of equation (7,494), equations (7,497) and (7,498) become

and

$$2\mathbf{k} \cdot \mathbf{p} = \pm \mathbf{v}^{*} + \mathbf{R} \cdot \mathbf{v},$$
 (7.50.0)

respectively. Adding (7,199) and (7,500), and using equation (7,394) $(\mathbf{k}^* + \mathbf{k} = 2\mathbf{\kappa})$, we obtain

$$\mathbf{i}\mathbf{x} \cdot \mathbf{p} = \pi \left(\boldsymbol{\sigma} \to \mathbf{e}^* + \mathbf{v}^* \to \mathbf{R} \cdot \mathbf{v} \right), \qquad (7.501)$$

Using equation (1.1065) the time average of the real Uniov-Bownting victor (7.482)

$$-\operatorname{Re} \begin{pmatrix} \sigma_{1} & \sigma_{2} & \sigma_{3} & -\rho_{f} & 0 & 0 \\ \sigma_{12} & \sigma_{21} & \sigma_{23} & 0 & -\rho_{f} & 0 \\ \sigma_{13} & \sigma_{24} & \sigma_{14} & 0 & 0 & -\rho_{f} \end{pmatrix} + \operatorname{Reev} (. (7.502))$$

7.11 Anisotropic pure-viscoclasticity

İ۶

which gives the overage power flow

As in the previous section, the time average of the strain-energy density

$$\langle V \rangle = \frac{1}{2} \operatorname{Re}(\boldsymbol{\sigma}_{+}) \cdot \operatorname{Re}(\boldsymbol{v})$$
 (7.504)

is.

$$\langle U \rangle = \frac{1}{4} R \mathbf{e} (\mathbf{a} + \mathbf{e}^*) = \frac{1}{4} R \mathbf{e} (\mathbf{e}^* + \mathbf{P} \cdot \mathbf{e})$$
 (7.505)

Similarly, the time-averaged kinetic-energy density is

$$\langle I \rangle = \frac{1}{4} \mathbf{Re} \left(\mathbf{v}^* \rightarrow \mathbf{R} \cdot \mathbf{v} \right),$$
 (7.506)

and the turns averaged strain and kinetic dissipated-energy densities are

$$(D_{VI} = \frac{1}{2} Im(\mathbf{e}^* \rightarrow \mathbf{P} \cdot \mathbf{e}), \qquad (7.507)$$

and

$$\langle D_1 \rangle = -\frac{1}{2} \ln \left(\mathbf{v}^* - \mathbf{R} \cdot \mathbf{v} \right),$$
 (7.508)

respectively. The last two quantities represent the energy has per unit volume due to viscoelastic and viscoelynamic effects, respectively. The thirds sign in equation (7.508) is due to the fact that $Im(V)/(\omega) \sim 0$ (see equation (7.1776). It can be shown that the dissipated energies should be defined with the opposite sign fract expendent (which is used This is the case for the dissipated kenetic energy in the work of Carologe (1996b).

Substituting equations (7.50%) (7.50%) and (7.50%) cuto the real part of equation (7.50%), we obtain

$$\kappa \cdot \langle p \rangle = \omega \langle V \rangle + \langle T \cdot (- \omega \langle F \rangle),$$
 (7.509)

where (E) is the stored energy density (7, 192). Furthermore, the imaginary part of equation (7, 501) gives

$$2 \kappa \ln p = \omega((D_0) - (D_i)),$$
 (7.510)

The wave surface is the locus of the end of the energy-velocity vector unitiplied by one quit of propagation time, with the energy velocity defined as the ratio of the average power-flow density $\langle p \rangle$ to the total energy density $\langle E \rangle$. Since this is equal to the sum of the average kinetics and strain-energy densities $\langle K \rangle$ and $\langle V \rangle$, the energy velocity is

$$\mathbf{v}_{i} = \frac{\left[\mathbf{p}\right]}{\left(T + V\right)}, \quad (7.511)$$

Dissipation is quantified by the quality factor, which can be defined as

$$Q = \frac{2(V)}{\langle D \rangle}$$
, (7.512)

Using the definition of the energy velocity and equation (7.509), we obtain

$$\boldsymbol{\kappa} = \boldsymbol{v}_{i} + (\boldsymbol{s}_{i} + \boldsymbol{v}_{i} - 1), \quad (7.513)$$

where $v_{\mu} = \omega/\epsilon$ is the phase velocity, and $\mathbf{s}_{H} = \mathbf{\kappa} (v_{\mu})$ is the slowness vector. Relation (7.543), as in a single-phase nucleum (see equation (1.78)), means that the phase velocity is simply the projectors of the energy velocity onto the propagation direction

Finally, solutracting equation (7.499) from (7.560) and using (7.494) yields the energy balance equation

$$2\alpha \cdot \mathbf{p} = 2(x_1(b)) - (I_0) - x(D),$$
 (7.514)

loking the real part of (7.514), we get

$$2\alpha_{\pm}(\mathbf{p}) = \pm_{i}D_{i},$$
 (7.515)

This equation is the generalization of equation (1.83) for viscodastic single-phase media stating that the time averaged dissipated energy can be obtained as the projection of the average power flow density onto the attendation direction.

7.14.5 Homogeneous plane waves

For homogeneous waves, the propagation and attenuation directions coincide and the wavevector can be written as

$$\mathbf{k} = (\kappa - in) \mathbf{k}_{\mathbf{k}_{1}} = k_{\mathbf{k}_{2}}$$
 (7.516)

where

$$\kappa = d_{12} d_{13} d_{13}$$
 (7.517)

defines the propagation direction through the directions cosmix t_{ij} i_{j} and l_{k} . For boundgeneous waves

$$\mathbf{K} \to k \mathbf{L} = k \begin{pmatrix} l_1 & 0 & 0 & l_2 & l_2 & 0 \\ 0 & l_2 & 0 & l_1 & 0 & l & 0 \\ 0 & 0 & l_1 & l_2 & l & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & l_1 \\ 0 & 0 & 0 & 0 & 0 & l_2 \\ 0 & 0 & 0 & 0 & 0 & l_3 \end{pmatrix},$$
(7.518)

where i is the complex wavenumber. Using (7, b5), we see that equations (7, 175) and (7, 176) give

$$(\mathbf{R}^{(4)} | \mathbf{\Gamma} = i_1^* \mathbf{L}) | \mathbf{v} = 0.$$
 (7.519)

where

$$\mathbf{F} = \mathbf{L} \cdot \mathbf{P} \cdot \mathbf{L} \qquad (7.5.9)$$

is the Kelvin-Christoffel matrix, and

 $\alpha_{i} = \frac{2}{k} \qquad (7.521)$

is the complex velocity

Making zero the differminant, equation (7,519) gives the following dispersion relation

$$\operatorname{der}(\mathbf{R}) = (\mathbf{\Gamma} + c_1^2 \mathbf{L}) = 0, \quad (7.522)$$

The eigensystem formed by equations (7.519) and (7.522) gives six eigenvalues and the corresponding eigenvectors. From of them correspond to the wave modes and the others equal zero, since it can be shown that two rows of the system matrix are linearly dependent. These modes correspond to the fast and slow quasi-compressional waves, and the two quasi-show waves.

The slavness and attranating victus for homogeneous waves can be expressed in trans of the complex velocity as

$$\mathbf{s}_{\theta} = \operatorname{Re}\left(\frac{1}{r_{\theta}}\right) \mathbf{\kappa} \tag{7.523}$$

sad

$$\alpha = -z \ln \left(\frac{1}{r_0}\right) \hat{\kappa}, \qquad (7.523)$$

respectively. (Note that $(1/r_i)$ is the recipiocal of the phase velocity.)

The average strain-energy density (7.505) can be written, using equations (7.474), (7.205) and (7.508)-(7.521), in terms of the density as matrix \mathbf{R}

$$\langle V \rangle \simeq \frac{1}{4} \epsilon_{\nu} - i \operatorname{Re}_{\nu} (\langle \mathbf{v}^{2} + \mathbf{R} + \mathbf{v}^{2}),$$
 (7.525)

where we used the fact that ${f R}$ and ${f \Gamma}$ are symmetric matrices.

Equation (7.525) is formally similar to the strain-energy density we arisotropic visroelastic media, where $\langle V \rangle = \langle \mu, v_e \rangle$ (Ref. () with the Section 4.344. In a single-phase medium, every particle vehicity component is regarily weighted by the density. Note that, when the medium is basics, v_e is trait and the average strainenergy density rights the average kinetic energy (7.506).

From equations (7.505) and (7.506) and using the property $\mathbf{v} \to \mathbf{R} \cdot \mathbf{v}^* \to \mathbf{R} \cdot \mathbf{v}$ (because \mathbf{R} is symmetric), we note that the stored energy density (7.492) becomes

$$\langle E \rangle = \frac{1}{4} \mathbf{R} c \left[\left(\mathbf{1} + \frac{c_s^2}{c_s^2} \right) \mathbf{v} + \mathbf{R} \cdot \mathbf{v}^* \right],$$
 (7.526)

When the medium is lossless, r_1 and \mathbf{R} are real, and $\langle E \rangle$ is equal to twice the average kinetic energy (7.506).

For exclutation purposes, the Uniov-Deputting vector (7/182) can be expressed in terms of the eigenvector \mathbf{x} and complex velocity v_i . The average power flow (7.503) can be written as

$$\langle \mathbf{p} \rangle = -\frac{1}{2} \operatorname{Re}\left[\mathbf{e}_{t} (\mathbf{U}^{t} \cdot \boldsymbol{\sigma}_{t}) (\cdot \mathbf{v}^{*})\right],$$
 (7.527)

where is as the unit Cartision vector and the Einstein convention for reprated indices is used. Up are 6 \approx 7 matrices with most of their elements equal to zero, except U_{11} , U_{22}^{*} , U_{23}^{*} , and U_{23}^{*} , which are equal to 1. Substitution of the stress-strain relation (7.173) into (7.527) and the use of equations (7.174), (7.195) and (7.518)-(7.521) yields the desired expression

$$\langle \mathbf{p} \rangle \leq \frac{1}{2} \operatorname{Be} \left[r_{1}^{-1} \mathbf{v} \rightarrow \mathbf{L} \cdot \mathbf{P} \cdot (\mathbf{e}_{1} \mathbf{U}^{t}_{-}) + \mathbf{v}^{t} \right]$$
 (7.528)

To obtain the quality factor (7.512), we follow the same steps that foll to equation (7.525) and note that the dissipated energy (7.421) can be written as

$$\langle D \rangle = \frac{1}{2} \text{Im} \left[\left(\frac{n}{n_{e}^2} - 1 \right) \mathbf{v} - \mathbf{R} \cdot \mathbf{v}^* \right]$$
 (7.529)

Using equation (7.525), we obtain

$$Q = \frac{2\langle V \rangle}{\langle D \rangle} = \frac{\operatorname{Retry}^{2} \left[|\mathbf{R} \cdot \mathbf{v}^{*} \rangle - |\mathbf{R} \cdot \mathbf{v}^{*} \rangle - (7.546) \right]}{2\operatorname{Im}(v, (\operatorname{Retry}^{2} + \mathbf{R} + \mathbf{v}^{*}))}$$
(7.546)

If there are no losses due to viscosity effects (**R** is real and $(D_1) = 0$), $\mathbf{v} \to \mathbf{R} \cdot \mathbf{v}^*$ is real and

$$Q = \frac{\text{Re}(c)}{\text{Im}(c)}, \quad (7.501)$$

as in the single-phase case (see Section 1.3.1).

7.14.6 Wave propagation in femoral bone

Let us consider propagation of homogeneous plane waves in human fermial home (intherhombic symmetry), investigated by Caroliner, Cavallini and Holbig (1998) using a single-phase theory for anisotropic viscoclastic taxia. (See Cowin (1999) for a survey of the application of poroclasticity in home mechanics.), A similar application for tooks is given by Caroline, Helbig and Helle (2003), where the effects of pore pressure and fluid saturation are also investigated.

To introduce viscoelastic attenuation, we use a stress-strain relation based on model 2 of Section 3.1. Each eigenvector (or eigenstrater) of the stiffness matrix defines a fundomental deformation state of the medium. The six eigenvalues (or eigenstriffnesses) represent the genuine elastic parameters. In the elastic case, the strain energy is uniquely parameterized by the six eigenstiffnesses. Eacso ideas data back to the middle of the 19th century when Lord Kelvin introduced the concept of "principal strain" (eigenstrain in modern terminology) to describe the deformation state of a medium (Kelvin, 1856).

We assume that the hone is saturated with watch of bulk modulus $K_f \approx 2.5$ GPa, density $p_f = 1000$ kg/m³ and viscosity q = 1 eP: the grain bulk modulus is $K_s = 28$ GPa. the grain density is $p_s = 1815$ kg/m³, the porosity is $\alpha = 0, 1$, the fortunistics are T = 2 $T_f = 3$ and $T_c = 3.6$, and the matrix permeablely oscilar $n_s = 1.2 \times 10^{-12}$ m³, $n_s = 0.8 \times 10^{-12}$ and $n_s = 0.7 \times 10^{-12}$ m³. The stiffness matrix of the denoised porous modium in Yoing's mutation ($r_{f,s}$ are Section 7.3) is

in GPa. The components of this matrix serve to calculate the elements of matrix C^* by using equation (7.136). This matrix corresponds to the high-frequency quirelexed: limit

whose components are

$$\mathbf{C}^{a} = \begin{pmatrix} 19.8 & 11.7 & 11.5 & 0 & 0 & 0 & 3.35 \\ 11.7 & 21.8 & 12.03 & 0 & 0 & 0 & 3.11 \\ 11.5 & 12.03 & 28.7 & 0 & 0 & 0 & 2.59 \\ 0 & 0 & 0 & 6.23 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 5.61 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1.01 & 0 \\ 3.35 & 3.14 & 2.59 & 0 & 0 & 0 & 6.12 \end{pmatrix}$$

in GPa. In order to apply belivin's formulation. Hooke's law has to be written in tensorial form – this implies multiplying the (41), 155, and (66) elements of matrix \mathbb{C}^{n} by a factor 2 (see equation (4.8)) and taking the leading principal submatrix of order 6 (the upper-left 6 \times 6 matrix). This can be done for the matrix and enchant for which the varieties of fluid content ζ is equal to zero telused system) (Concara). Helbig and Helbi, 2003). Let us call this new matrix stensort \mathbb{C}^{n} . This matrix can be diagonalized to obtain

$$C^{\mu} = \mathbf{Q} \cdot \mathbf{A} \cdot \mathbf{Q}$$
, (7.532)

where $\mathbf{A} = \operatorname{diag}(\lambda_1, \lambda_2, \Lambda_3, \Lambda_3, \Lambda_3, \Lambda_4)$ is the eigenvalue matrix, and \mathbf{Q} is the matrix formed with the eigenvectors of \mathbf{C}^n or more precisely, with the columns of the right (orthonormal) eigenvectors. (Note that the sciencity of \mathbf{C}^n implies $\mathbf{Q} = -\mathbf{Q}$.) Hence in accordance with the correspondence principle and its application to equation (7.532), we introduce the viscorlastic stiffness tensor

$$C = Q (A \to Q)$$
, (7.533)

where A 1 is a diagonal matrix with entries

$$X_{i}^{(r)}(\omega) = X_{i}M_{i}(\omega) = -I = 1, ..., 6$$

(7.533)

The quantities M_t are complex and frequency-dependent dimensionless moduli. We describe each of them by a Zener model, whose relaxation frequency is equal to ω (see equation (1.6)). In this case, we have

$$M_1 = \frac{\sqrt{Q_1^2 + 1} - 1 + iQ_2}{\sqrt{Q_1^2 + 1} + 1 + iQ_2},$$
 (7.535)

where Q_i is the quality factor associated with each modulus. (We note here that if an experiant) kernel is used, iQ_i should be replaced by $-iQ_i$, and the dissipated strain energy should be defined with the opposite sign (). In recover the Vog('s notation, we should divide the (11), (5)) and (6) references of matrix **C** by a factor 2. This gives the complex matrix **P**.

In orthorhondoic porous media, there are six distinct eigenvalues, and, therefore, six complex moduli. We assume that the dimensionless quality factors are defined as $Q_I \approx (\Lambda_{12}\Lambda_{6})Q_{6}$, I = 1, ..., 6, with $Q_I = 30$. This choice implies that the higher the stiffness, the higher the quality factor (i.e., the harder the medium, the lower the attenuation). Matrix **P** is then given by

п

ш

- 11 П

 $4.88 \times i0.13$

П

3/35

2.591 41

41

ŧI.

6.12

 $\mathbf{P} = \begin{bmatrix} 1.00 + 0.026 & 11.4 + 0.0002 & 11.4 + 0.0002 \\ 11.7 + 0.0002 & 21.5 + 0.26 & 12.0 + 0.0004 \\ 11.5 + 0.0001 & 12.0 + 0.0001 & 28.1 + 0.026 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \\ 3.35 & 3.14 & 2.59 \end{bmatrix}$

in GPa

 Polar representations of the attenuation factors (7.523) and energy velocities (7.511)
are shown in Figure 7 (5 and 7.16) respectively, for the (in 5) principal plane of the modum
(9, 10). Only one quarter of the curves are displayed because of symmetry considerations
The Cartosian planes of an orthorhombic medium are planes of symmetry, and, therefore
one of the shear waves, denoted by S. is a pure cross-plane mode. The tickmarks in Figure
7.16 indicate the polarization directions $(r_1, 0, r_4)$, with the points uniformly sampled as
a function of the phase angle . The curves are plottest for a frequency of $f=\omega/12\pi$
10 kHz, smaller than the characteristic frequency $f_{i} = g_{i} / (T_{i} p \phi_{A}) = 15$ kHz, which

 $11.7 \times 10.002 - 11.5 \times 10.004$

6

n

0

6.1 × i0.13

υ

υ

u

ά.

0

ίĽ

5.18 - 90.13

- 01

U

0

detrumus the upper limit of the low-frequency theory The strong dissipation of the slow qP wave is due to the Bint mechanism, i.e. the viscolynamic effect. On the other hand, it can be shown that $\langle D_V \rangle$ and $\langle D_V \rangle$ are rotu parable for the qP, qS and S waves. Anisotropic permeability affects the arregulation of the slow of wave. According to fliot's theory, the lower the terme ability, the higher the attenuation. In fact, the vertical attenuation factor is higher than the horizontal attenuation factor. Ansotropic fortuosity mainly affects the velocity of the slow **qP** wave. This is (approximately) inversely proportional to the square mot of the torthosity. Reme, the vertical vehicity is smaller than the horizontal vehicity.

The three faster waves propagating in the (7.5) plane of a single phase orthochembic medium have the following velocities along the coordinate axes.

$$\begin{aligned} v_{gs}(0) &= v_{gs}(90) = \sqrt{\mu_{gs}/\mu} \\ v_{gt}(0) &= \sqrt{\mu_{gs}/\mu} &= v_{gt}(90) = \sqrt{\mu_{gs}/\mu} \\ v_{s}(0) &= \sqrt{\mu_{gs}/\mu} &= v_{s}(90) = \sqrt{\mu_{gs}/\mu} \end{aligned} (7.54b)$$

where B corresponds to the travis and 90 to the travis, and car are the correspondentially stillnesses. The velocities (7.536) do not correspond exactly to the velocities in the potons case. since here the density is a matrix, not a scalar quantity. For instance, the densities corresponding to the S and qS waves along the traces are $\rho = \rho_{1}^{2}/\mathbf{R}_{0}$ and $\rho = \rho_{1}^{2}/\mathbf{R}_{0}$. where \mathbf{R}_{11} and \mathbf{R}_{12} are components of matrix \mathbf{R} defined in opportion (7.477). However, the



Figure 7.15: Polar representation of the attenuation factors in one of the planes of mirror symmetry of human femoral bone saturated with water, where (a) illustrates the fast quasi-compressional wave qP, the quasi-shear wave qS, and the pure cross-plane shear wave S, and (b) shows the slow quasi-compressional wave. The frequency is 10 kHz.



Figure 7.16: Polar representation of the energy velocities in one of the planes of mirror symmetry of human femoral bone saturated with water, where qP is the fast quasi-compressional wave, qS is the quasi-shear wave, S is the pure cross-plane shear wave, and slow qP is the slow quasi-compressional wave. The tickmarks indicate the polarization directions $(v_1, 0, v_3)$ for the qP, slow qP and qS waves, while the polarization of the S wave is (0,1,0). The curves correspond to a frequency of 10 kHz.

velocities (7.536) can be used to qualitatively verify the behavior of the energy-velocity curves. Ato the basis of these equations. Figure 7.16 is an agreement with the values indicated above for matrix \mathbf{P}

Chapter 8

The acoustic-electromagnetic analogy

Mathematical analysis is no extension as watno strelly it defines all periophile relations , measities track, spaces, funces, temperatures: this difficult science is formed slowly, but it preserves every principle, which it has not acquired, it grows and strengthens itself increasingly in the model of the noncy caractions and errors of the boundar model. Device fulliphile is chorressed has no models to express conflued mations. It brangs together previous real the most diverse, and discovery the helden analogies which units them.

Joseph Fourier (Fourier, 1822).

Many of the great scientists of the past have studied the theory of wave motion. Throughout this development there has been an interplay between the theory of light waves and the theory of material waves. In 1660 Robert Hooke formulated stress-strain relationships which established the elastic behavior of sulid bodies. Hooke helieved light to be a vibratory displacement of the medium, through which it propagates at finite speed. Significant experimental and mathematical advances came in the nine teenth century. Thomas Young was one of the first to consider shear as an elastic strain, and defined the elastic modulus that was later named Young's modulus. In 1809 Eliterne Louis Malus discovered polarization of light by reflection, which at the time David Brewster contectly described as "a memorable speech in the history of optics". In 1815 Brewster discovered the law that regulated the polarization of light. Augustus lean Freshel showed that if light were a transverse wave, then it would be possible to develop a theory accommodating the polarization of light. George Green (Green, 1838, 1842) made extensive use of the analogy between elastic waves and light waves, and an analysis of his developments illustrates the power of the use of mathematical analogies.

Later, in the second part of the ninetrenth contrast, hours Clerk Maxwell and Lord Kelvin used physical and mathematical analogies to study wave prenomena in clustic theory and electromagnetic equations is analogous to the elastic displacements. Maxwell into the electromagnetic equations is analogous to the elastic displacements. Maxwell assumed his equations were valid in an absolute system regarded as a reaching (ca²)of the effort that filled the whole of space. The other was to a state of stress and would only transmit transverse waves. With the advent of the theory of relativity, the emerge of the ether was aluminard. However the fact that electromagnetic waves are transverse waves is respontant. This situation is in contrast to a fluid, which can only transmitlongitudinal waves. A viscoelastic bady transmits bath longitudinal waves and transversiwaves. It is also possible to recast the viscoelastic equations into a form that closely parallels. Moreoull's expositions. To many cases this formal analogy becomes a complete mathematical equivalence such that the same equations can be used to solve problems in both disciplines.

In this chapter, it is shown that the 2-D Moxwell's equations describing propagation of the TFM mode in accordingle modia is completely analogous to the SH-wave oppation based on the Maxwell anisotropic viscoelastic solid. This equivalence was probably known to Maxwell, who was aware of the analogy between the process of conduction estatic induction through dielectrics) and viscosity relasticity). Actually, Maxwell's electron agenetic theory of high, including the conduction and displacement currents, was already completed in his paper 10m physical lines of forcell published in two parts in 1861 and 1862 (Headev, 1986). On the other hand, the viscoelastic model was proposed in 1867 (Maxwell, 1867, 1800). He seems to have arrived to the viscoelastic theology from a comparison with Thomson's tolography equations (Bland, 1988), which describe the pricess of conduction and dissipation, of electric energy through rables. We use this theory to obtain a complete mathematical analogy for the reflection-transmission problem.

furthemnore, the analogy can be used to get meight into the proper definition of energy, The concept of energy is important in a large number of applications where it is increasive to know how the energy transferred by the electromagnetic field is related to the strength of the field. This context involves the while electrical, radio, and optical engineering. where the medium can be assumed dielectrically and magnetically linear. Energy halance equations are important for characterizing the energy stored and the transport properties in a held. However, the delimition of stored threat energy and energy dissipation rate is controversial, both in electromagnetism (Oughstan and Sherman, 1984) and viscoe'asticky (Caviglia and Morro 1939). The problem is particularly intrigging in the time domain since different definitions may give the same time-average value for harmonic helds. This ambiguity is not present when the constitutive equation rate by described in terms of springs and dashpots. That is, when the system can be defined as terms of internal variables and the relaxation function has an exponential form. In Chapter 2 we gave a general expression of the viscoelastic energy densities which is consistent with the usochanical model description. In this chapter, the electric difference and magnetic energies an defined in terms of the viscoelastic expressions by using the analogy. The theory is applied to a simple dielectric relaxation process – the Debye model – that is mathe matically equivalent to the viscoelastic Zenet model. The Debye model has been applied to bio-elvertomagnetism in the analysis of the response of biological tissues (Roberts and Petropolous, 1996), and to geophysics in the singulation of ground-penetramic-nolar wave propagation through wet soils (Digmen and Siggins, 1994, Carmone, 1996).

The 3-D Maxwell's equations are generalized to describe realistic wave propagation by using mechanical viscoelastic models. A set of Zener elements describe several magnetic and dedictric-relaxation mechanisms, and a single Kelvin-Voigt element incorporates the out-obphase behaviour of the electric conductivity rany deviation from Ohm's laws. We assume that the mechanic has orthonhomoic symmetry, that the principal systems of the three material tensors concrde and that a different relaxation from tion is associated with each principal component. A brief divivation of the Kramers-Wronig dispersion relations by using the Couchy integral formula follows, and the opproximate with the acoustic case is shown. Moreover, the averaging methods use in clasticity (Backus, 1962) can be used in electromagnetism. We draive the constitutive equation for a layered anciham, where each single layer is anisotropic, honorzemons and this compared to the wavelength of the electromagnetic wave. Assuming that the layer interfaces are flat, we obtain the dielectricpermittivity and conductivity matrices of the composite mechanic. Other mathematical analogies include the high-frequency transaverage and CIGM equations, the recupiently principle, Babinet's principle and Alford extreme. Finally, a formal analogy can be established between the diffusion equation corresponding to the slow compressional wave described by Biot's theory (see Section 7.7.1) and Maxwell's equations at low frequencies A common analytical solution is obtained for both problems and a numerical furthed is outlined in Compute 9.

The use of mathematical analogies is extensively used in many fields of physics (e.g., Tonti, 1976). For instance, the haplace equation describes different physical processes such as thermal conduction, electric conduction and stationary instational flow in hydicalynamics. On the other hand, the static constitutive equations of porcelasticity and thermoelasticity are formally the same if we identify the proc fluid pressure with the temperature and the fluid compression with entropy (Norris, 1991, 1991).

The analogy can be exploited in several ways. In first place, existing acoustic modeling reales can be easily modeling to simulate electromagnetic propagation. Secondly, the set of solutions of the acoustic problem, obtained from the correspondence principle, can be used to test decromagnetic endes. Moreover, the theory of propagation of plane hazamair waves in acoustic media also applies to electromagnetic purposation.

8.1 Maxwell's equations

In 3-D vector mutation, Maxwell's equations are

$$\nabla \times \mathbf{E} = -\partial_t \mathbf{B} + \mathbf{M}$$
(8.1)

and

$$\nabla \times \mathbf{H} \approx \partial_t \mathbf{D} + \mathbf{J}^t$$
(8.2)

(Born and Wolf, 1963, p = 1), where E. H. D. B. J' and M are the electric vector, the magnetic vector, the electric displacement, the magnetic induction, the electric-carrient density (including an electric source current) and the magnetic source current density, respectively. In general, they depend on the Cartesian contributes (x, y, z) and the time variable t.

Additional constitutive equations are needed to relate \mathbf{D} and \mathbf{B} to the field vectors for consistropic lossy media including dielectric relayation and magnetic loss, \mathbf{D} and \mathbf{B} run be written as

$$\mathbf{D} = \hat{\epsilon} + \partial_t \mathbf{E}$$
 (S.3)

and

$$B = \mu + \partial_t H_c$$
 (8.3)

where $r(\mathbf{x}, t)$ is the diebetric-permittivity tensor and $\mu(\mathbf{x}, t)$ is the magnetic-permodality tensor. The electric-current density is given by the generalized Ω line slaw.

$$\mathbf{J}' = \boldsymbol{\sigma} + \partial_t \mathbf{E} + \mathbf{J}$$
 (8.5)

where $\sigma(\mathbf{x}, t)$ is the conductivity tensor, the convolution accounts for out-of-phase components of the conduction-current drosity with respect to the electric vector and \mathbf{J} is the electric-source current density¹. Substituting the constitutive equations (8.3) and (8.4) and the current density (equation (8.5)) into (8.1) and (8.2), and using properties of the convolution, gives

$$|\nabla \otimes \mathbf{E} = \phi \mathbf{\mu} + \partial_{\mu}^{\mu} \mathbf{H} \otimes \mathbf{M}$$

(8.6)

and

$$\nabla \times \mathbf{H} = \sigma * \partial_t \mathbf{E} + \mathbf{c} * \partial_0^2 \mathbf{E} + \mathbf{J}_c$$

(8.7)

which are a system of six scalar equations in six scalar unknowns.

The time-dependent tensors, which are symmetric and positive definite, describe various electromagnetic relaxation processes of the material, like dielectric relaxation and out-of-phase behavior of the conduction current at high frequencies. The time dependence is not arbitrary: it is assumed for each tensor that its eigenvectors are invariant in time, so that in a coordinate system coincident with these fixed eigenvectors, the time dependence of the tensor is hilly specified by three time functions on the main diagonal which serve as the time-dependent eigenvalues of the matrix. These equations also include paramagnetic losses through the time-dependent magnetic-permeability tensor μ .

In lossless media, the material tensors are replaced by

$$\begin{split} \mu(\mathbf{x},t) &= \langle \mu(\mathbf{x}(H)) \rangle \\ \sigma(\mathbf{x},t) &= \langle \sigma(\mathbf{x}(H)t) \rangle \\ \epsilon(\mathbf{x},t) &= \langle \epsilon(\mathbf{x}(H)t) \rangle . \end{split}$$
(8.8)

where H(t) is Heaviside's function, and the classical Maxwell's equations for anisotropic media are obtained from optations (8.6) and (8.7).

$$\nabla \times \mathbf{E} = -\mu \cdot \partial_t \mathbf{H} + \mathbf{M}$$
(8.9)

and

$$\nabla \times \mathbf{H} = \boldsymbol{\sigma} \cdot \mathbf{E} + \boldsymbol{\epsilon} \cdot \partial_t \mathbf{E} + \mathbf{J},$$
 (8.10)

In general, call of the 3 + 3 symmetric and positive definite tensors μ , ϵ and σ have a set of cantrally perpendicular eigenvectors. If there is no eigenvector in common for all three tensors, the medium is said to be tradinic. If there is a single eigenvector common to all three tensors, the medium is said to be monoclinic and has a mirror plane of symmetry perpendicular to the common eigenvector

8.2 The acoustic-electromagnetic analogy

In order to establish the mathematical analogy between electromagnetism and acoustics, we needs the accessic equations in the particle-velocity (stress formulation). The conservation equation (1.28) and use of (1.44) give

$$\nabla \cdot \boldsymbol{\sigma} \sim \mathbf{f} = pol_t \mathbf{v},$$
 (8.14)

³Note the difference between magnetic permeability due to responsible to the number with π_{12} contact a distinct conducts, strategies π_{12} cond π_{23} due to the previous diagrams.

and operations (1.26) and (1.14) combine to give the relation between strain and particle velocity

$$\nabla \rightarrow \mathbf{v} = \partial_t \mathbf{e},$$
 (8.12)

Auld (1990), p. 101) establishes the acoustic-destromagnetic analogy by using a 3-D. Kelvin-Viagi model:

$$\sigma = \mathbf{C} \cdot \mathbf{e} + \eta \cdot d_i \mathbf{e}, \quad (8.63)$$

where **C** and η are the elasticity and viscosity matrices, respectively. (Compare this relation to the 1-D Kelvin-Voig) stress-strain relation in equation (2.1594). Taking the first-order time derivative of (8.13), multiplying the result by **C** = , and using equation (8.12), we get

$$\nabla \rightarrow \mathbf{v} + \mathbf{C}^{-3} \cdot \boldsymbol{\eta} \cdot \nabla \rightarrow \partial_t \mathbf{v} = \mathbf{C} \rightarrow \partial_t \boldsymbol{\sigma}$$
 (8.14)

Auld establishes a formal analogy between (8.14) and (8.14) with Maxwell's equations (8.9) and (8.10), where σ corresponds to E and y corresponds to H.

A better correspondence can be obtained by introducing, instead of (8.13), a 3-D Maxwell constitutive equation:

$$\partial_t \sigma = \mathbf{C} \oplus \partial_t \sigma + \eta \oplus \sigma$$
 (8.15)

(Compare this relation to the 1-D Maxwell stress-strain relation (2.135)(1) Eliminating the strain, by using equation (8.12), gives an equation analogous to (8.10).

$$\nabla \rightarrow \mathbf{v} = \eta \rightarrow \sigma \oplus C \rightarrow \partial_t \sigma$$
. (8.16)

Defining the compliance matrix

$$S = C^{-1}$$
 (8.17)

and the fluidity matrix

$$\tau = \eta$$
 . (8.18)

equation (8.16) becomes

$$\nabla \rightarrow \mathbf{v} = \mathbf{\tau} \cdot \mathbf{\sigma} + \mathbf{S} \cdot \partial_t \mathbf{\sigma},$$
 (8.19)

In general, the analogy does not mean that the acoustic and electromagnetic equations represent the same mathematical problem. In fact, $\boldsymbol{\sigma}$ is a 6-D vector and **E** is a 3-D vector. Moreover, acoustics involves 6 × 6 matrices (for material properties) and electromagnetism 3 × 3 matrices. The complete equivalence can be established in the 2-D rase by using the Maxwell model, as run be seen in the following.

A realistic medium is described by symmetric dielectric permittivity and conductivity tensors. Assume an isotropic magnetic-permisability tensor

$$\mu = \mu h$$
 (8.20)

and

$$\mathbf{c} = \begin{pmatrix} \epsilon_{11} & 0 & \epsilon_{13} \\ 0 & \epsilon_{22} & 0 \\ \epsilon_{13} & 0 & \epsilon_{33} \end{pmatrix}$$
(8.21)

$$\boldsymbol{\sigma} = \begin{pmatrix} \phi & 0 & \sigma_{ik} \\ 0 & \sigma_{ik} & 0 \\ \sigma_{ik} & 0 & \sigma_{ik} \end{pmatrix}$$
(8.22)

where I_3 is the 3 × 3 identity matrix. There is (8.21) and (8.22) correspond to a monoclinic modium with the q-axis preparality to the plane of symmetry. There always exists a coordinate transformation that diagonalizes these symmetric matrices. This transformation is called the principal system of the medium, and gives the three principal components of these tensors. In order and isotropic media, the principal components are all equal. In retragonal and hexagonal materials, two of the three parameters are equal. In orthorhomline, monochnic, and tracher media, all three components are unequal.

Now, let us assume that the propagation is in the (x, z)-plane, and that the material properties are invariant in the g-direction. Then, E_{1} , E_{2} and H_{2} are decoupled from E_{1} , H_{1} and H_{2} . In the observe of electro-source currents, the first three fields obey the TM (transverse-magnetic) differential equations:

$$\partial (F_1 - \partial_i L_1 - \mu \partial_i H_1 + M_3)$$

(8.23)

$$(\partial_t H_i - \sigma_t | E) = \sigma_0 E_1 + \epsilon_0 (\partial_t E_1 + \epsilon_0) \partial_t E_0$$

(8.24)

$$\partial_{t}H_{2} = \sigma_{1s}E_{0} + \delta_{1s}E_{1} + \ell_{0}\rho_{b}F_{1} + c_{3s}\partial_{t}E_{1s}$$
(8.25)

where we have used equations (8.7) and (8.10). On the other hand, in acoustics, uniform properties in the galinection imply that one of the shear waves has its own ideoupled) diferential equation, known in the identities as the SII-wave equation (see Section 11.2.1). This is structly true in the plane of mirror symmetry of a monoclinic modium. Propagation in this plane implies pure cross-plane struit motion and it is the most general situation for which pure shear waves exist at all propagation angles. Pure shear wave propagation in hexagonal modia is a degenerate case. A set of parallel fractures embedded in a truts versely isotropic formation can be represented by a monoclinic modium. When the plane of unitor symmetry of this modium is writted, the pure cross-plane strain waves are SII waves. Moreover, monocline media include many other cases of higher symmetry. Weak tetrogonal media, strong trigonal media and orthorhombic modia are subsets of the set of momoclinic media.

In a monorlinic medium, the elasticity and vascustly matrices and then inverses have the firm (1.37). It is assumed that any kind of symmetry pussessed by the attranation follows the symmetry of the crystallographic form of the material. This startment, which has been used in Chapter 4, can be supported by an empirical law known as Neumann's principle (Neumann, 1885).

The SH-wave differential optations optivalent to equations (1.46), corresponding to the Maxwell viscoelastic model represented by equation (8.15), are

$$\partial_t \sigma_{12} + \partial_s \sigma_{23} = \mu \partial_t v_2 + f_2,$$
 (8.29)

$$d_{1}c_{2} = z_{1}\sigma_{2} = z_{2}\sigma_{2} = z_{1}\partial_{0}z_{2} = -z_{2}\partial_{0}z_{2} = -z_{2}\partial_{0}z_{2}$$
 (8.27)

$$\partial_{\tau} c_{\tau} = \tau_{0}\sigma_{10} + \tau_{0}\sigma_{10} + s_{0}ab\sigma_{0} + s_{0}b\sigma_{0},$$
 (8.28)

where

$$\gamma_{11} = \eta_{20}/\eta_{10} - \eta_{20} = \eta_{10}/\eta_{10} - \eta_{20} = -\eta_{20}/\eta_{10} - \eta = \eta_{11}\eta_{10} - \eta_{20}^2, \tag{8.29}$$

$$s_{11}$$
 vale s_{22} $c_{11}le$ s_{22} $c_{32}le$ e $c_{12}c_{22}$ c_{22}^{2} (\$30)

where the sufficiences x_{10} and the viscosities η_{10} , J, J = -1.60 are the components of matrices \mathbb{Q} and η_i respectively.

Equations (8.23)-(8.25) are converted into equations (8.26)-(8.28) and vice versal inder the following substitutions:

$$\mathbf{v} = \begin{pmatrix} e_2 \\ e_{21} \\ \sigma_{31} \end{pmatrix} \quad \Leftrightarrow \quad \begin{pmatrix} H_1 \\ -E \\ -F_1 \end{pmatrix} \tag{8.31}$$

$$(8.31)$$

$$\mathbf{S} = \begin{pmatrix} s_{12} & s_{23} \\ s_{24} & s_{35} \end{pmatrix} \quad \Leftrightarrow \quad \begin{pmatrix} -\ell_{11} & -\ell_{24} \\ -\ell_{23} & -\ell_{24} \end{pmatrix} = \ell^{1}$$
(8.33)

$$\mathbf{\vec{\tau}} = \begin{pmatrix} \tau_{11} & \tau_{12} \\ \tau_{22} & \tau_{22} \end{pmatrix} \quad s_{7} = \begin{pmatrix} -\sigma_{11} & -\sigma_{12} \\ -\sigma_{12} & -\sigma_{22} \end{pmatrix} + \mathbf{\vec{\sigma}}^{2}$$
(8.34)

$$\rho = 0 - \mu_c$$
 (8.35)

where S and τ are redefined here as 2 + 2 matrices for simplicity. Introducing the 2 + 2 stiffness and viscosity matrices

$$\mathbf{C} = \begin{pmatrix} c_{ij} & c_{jj} \\ c_{ijk} & c_{ij} \end{pmatrix} \quad \text{and} \quad \boldsymbol{\eta} = \begin{pmatrix} q_{ij} & q_{jk} \\ q_{jk} & q_{jk} \end{pmatrix}, \tag{8.36}$$

we obtain the 2-D identities $\mathbf{S} = \mathbf{C}^{-1}$ and $\mathbf{\tau} = \mathbf{\eta}^{-1}$, which are similar to the 3-D equations (S 17) and (S 18), respectively. Then, the SH-wave equation for anisotropic medial based as a Maxwell stress strain relation, is mathematically equivalent to the TM equations whose "forcing term" is a magnetic entrem.

The mathematical analogy also holds for the TE equations under exitain conditions. If we consider the dielectric perimitivity an scalar quantity, the conductivity tensor equal to zero, and the magnetic perimability a tensor, we obtain the following [15 differential registions:

$$- (\partial_{i}H_{s} + \partial_{j}H_{i}) = i\partial_{i}F_{2} + J_{2}, \qquad (8.37)$$

$$(\partial_1 F_T - \mu_1) \partial_2 H_1 + \mu_3 a h H_3,$$
 (8.38)

$$-\partial |E_z = \mu_{is}\partial_i H | - \mu_{s0}\partial_i H_s, \qquad (8.39)$$

Then, the TM equations (S.23)-(S.25) and the precoding TE equations are expected for the following correspondence: H_{2} ($z = F_{2}$, E_{1} ($z = H_{3}$, $E_{1} < H_{4}$, M_{2} ($z = J_{2}$, $\mu < z$) ($z = \Rightarrow |\mu_{1}|, |\nu_{\infty} \Rightarrow |\mu_{3}\rangle$ and ($z_{\infty} \Rightarrow |\mu_{3}\rangle$. In the frequency domain, the zero conductivity restriction can be relaxed and the encospondence for the properties becomes $\mu \Rightarrow \delta + i\omega \delta$, $\sigma + i\omega z_{0} < \phi > \sigma < \pi < i\omega z_{0} < \phi > \mu_{3}$, and $\sigma_{1} \Rightarrow i\omega < \phi > \mu_{4}$, where ω is the angular frequency

To get a more intuitive idea of the analogy, and to introduce the concept of quality factor, we develop the following considerations, which lead to hypers 8.1 and 8.2. For vastonic reportion (8.28) with $c_{2k} = q_{2k} = 0$ can be constructed from the model displayed in Figure 8.1, where v_{2k} and v_{2k} are the strains on the dashput and on the spring respectively. In fact,

$$\partial_t(z_i+z_i) = \partial_t r_i$$



Figure 8.1: Mowell viscoelastic model corresponding to the ω_0 component of the subsystemic on static conduction, with $\omega_0 = \omega_0 = 0$. The strains acting on the dislight and spring are ω_0 and ω_0 respectively.

imply (8.28); indeed, if $r_{ch} = \eta_{th} = 0$, $\Omega_{0,0} | s_{ch} = 1/r_{ch}$ and $r_{ch} = 1/\eta_{ch}$.

Obtaining a pintonial representation of the electromagnetic held equations is not so easy. However, if, instead of the distributed-parameter system (8.24) and (8.25), we consider the corresponding humped-parameter system (electric circum), then such an interpretation becomes straightforward. Indeed, if we consider, for example, equation (8.24) and assume for simplicity that $\sigma_{13} = \sigma_{14} = 0$ then its right-hand side becomes

$$\sigma_{11}F_1 + c_{11}\partial_0F$$

on internis of enour elements.

$$\frac{1}{R}V + C\frac{dV}{dt} = I_t + I_x - I,$$

which corresponds to a parallel contention of a capacitor and a resistor as shown in Figure 8.2, where R and C are the resistance and the capacitor, respectively, V is the voltage tile , the subgrad of the choices field, and I_2 and I_3 are the clostric currents W/R corresponds to σE .

An important parameter of the criticit represented in Figure 8.2 is the loss tangent of the capacitor. The circuit can be considered as a real capacitor whose losses are modeled by the resistor R. Under the action of a harmonic voltage of frequency ω , the total critical I is not in quadrance with the voltage, but makes an angle $\pi/2 - \delta$ with it (I_1) is in phase with V_1 while I_2 is in quadrantic). As a consequence, the loss tangent is given by

$$\tan \delta = \frac{I - I \cos(\pi/2 - \delta)}{I_2 - I \sin(\pi/2 - \delta)}$$
(8.40)

Multiplying and dividing (8, 69) by V gives the relation between the dissipated power in the resistor and the reactive power in the capacitor

$$\tan \delta = \frac{4|I \cos(\pi/2 + \delta) - V^*||R|}{|V I \sin(\pi/2 - \delta) - \omega CV|^2} = \omega CR$$
(8.11)

The quality factor of the circuit is the inverse of the loss tangent. In terms of divice the permittivity and conductivity it is given by

$$Q = \frac{\omega r}{\sigma} \qquad (8.02)$$



Figure 8.2: The the eigenst equivalent to the viscos basic model shown in Figure 8.1, where R and Care the resistance and connector, V is the voltage, and I_1 and I_2 are the electric corrects. The caralogy capties that the energy dissipated in the resistor is equivalent to the energy lass of the dashpart and the energy stated at the capacitor is equivalent to the potential energy stated at the spirit. The magnetic energy is equivalent to the classic kinetic care. g_1

The kinetics and strain-energy densities are associated with the magnetics and electricenergy densities, by terms of electric elements, the kinetic, strain and dissipated energies represent the energies stored in inductances, capacitors and the dissipative ohmic lesses, respectively: A such randomy, used by Maxwell, can be established between particle mechanics and energies (Hammanl, 1984).

8.2.1 Kinematics and energy considerations

The kmematic quantities describing wave motion are the slowness, and the physe-velocity and attenuation voctors. The analysis is carried out for the accessive case, and the electropogratic case is obtacted by applying the equivalence (8.31)-18.35). For a harmonic plane wave of angular frequency preparing 18.111 – in abstract of body forces – heromes

$$\nabla \cdot \sigma = i\omega \rho v = 0 \qquad (8.43)$$

On the other hand, the generalized Maxwell stressestratic relation (8.15) takes the form

$$\sigma \in \mathbf{P} \cdot \mathbf{e}_{i}$$
 (8.43)

where \mathbf{P} is the complex stiffness matrix given by

$$\mathbf{P} = \left(\mathbf{S} - \frac{1}{\omega} \mathbf{\tau}\right) \quad . \tag{8.15}$$

All the matrices in this equation have dimension six. However, since the SH modens pure, a similar equation can be obtained for matrices of dimension three. In this case, the stress and strain simplify to

$$\sigma = (\sigma_{12}, \sigma_{12})$$
 and $\phi = (\partial_1 a_2, \partial_2 a_2)$. (8.16)

respectively, where ay is the displacement field

The displacement associated to a homogeneous viscoelastic SII plane wave has the form 14 107):

$$\mathbf{u} = a_2 \mathbf{e}_3 - a_2 - U_3 \exp(i\omega t - \mathbf{k} \cdot \mathbf{x})^2$$
. (8.47)

where $\mathbf{x} = (x, z)$ is the position vector and

is the complex wavevector, with $\mathbf{\kappa} = (l_1, l_3)$, defining the propagation direction through the direction ensures l_1 and l_3 . Replacing the stress-strain reprotion [8–34) into equation (8.43) yields the dispersion relation

$$p_{00}t_1^t + 2p_{00}t_1^t + p_{10}t_0^t - p\left(\frac{\sigma}{h}\right)^2 = 0,$$
 (8.49)

which is equivalent to equation (0.5). The relation (8, 19) defines the complex velocity (see equation (1.28)).

$$c_{1} \geq \frac{\omega}{k} \geq \frac{p_{0}J_{1}^{2} + 2p_{0}J_{1}^{2} + p_{0}J_{1}^{2}}{p}$$
(8.50)

The phase-velocity, slowness and attenuation vectors can be expressed in terms of the complex velocity and are given by equations (1/29), (4/33) and (4/34), respectively. The energy velocity is obtained by the same proceedure used in Section 1/14. We obtain

$$\mathbf{v}_{e} = \frac{c_{e}}{\operatorname{Be}\left(r_{e}\right)} \left\{ \operatorname{Be}\left(\frac{1}{\rho c_{e}}\left[\left(p_{e} J_{1} + \rho_{e} J_{1}\right)\mathbf{e}\right] + \left(p_{e} J_{1} + \rho_{e} J_{1}\right)\mathbf{e}\right], \quad (8.51)$$

which generalizes equation (4.115). The quality factor is given by equation (4.92).

From equation (8.15), in virtue of the acoustic-destromagnetic equivalence (8.31)-(8.35), it follows that \mathbf{P} corresponds to the inverse of the complex dedectric-permittivity matrix \mathbf{c} , namely:

$$\mathbf{P} = \phi_{ij} \cdot \mathbf{e} \equiv \mathbf{e}^{i} + \frac{i}{\omega} \sigma^{i}$$
(8.52)

Then, the electromagnetic phase velocity, slowness, attenuation, energy velocity and quality factor can be obtained from equations (1.29), (1.33), (4.34), (8.51) and (1.92), respectively, by applying the analogy

In orthorhombic worth, the Discomponents wansh therefore the complex stuffress matrix is diagonal with components.

$$(\epsilon_0^{-1} - i\omega - i_0^{-1})$$
 (8.73)

in the acoustic case, where I = 4 or 6, and

$$(r_0 + i\omega_0 n_0)^{-1}$$
(8.54)

in the electromagnetic case, where $\tau \approx 1$ or 3. In isotropic media, where 41-components equal the fifscomponents, the complex velocity becomes

$$\eta = (n - i\omega / n^{-1} \eta)^{-1/2}$$
(8.55)

330

in the accustic case, and

$$c = (c + \omega_{c}) \sigma_{10} + \frac{12}{2}$$
, (S.56)

in the electromagnetic case, where was the shear modulus,

In the isotropic case. the acoustic and electromagnetic quality factors are

$$Q = \frac{\omega \eta}{0}$$
(8.57)

and equation (8.42) respectively of $\eta \to 0$ and $\sigma \to \infty$, then the halo viourus diffusive: while conditions $\eta \to \infty$ and $\sigma \to 0$ correspond to the lossless limit. Note that η/η and η/σ are the relaxation times of the respective wave processes.

The incology allows the use of the transient analytical solution obtavind in Section 4.6 for the electromagnetic case (Ursin, 1983). The most powerful application of the analogy is the use of the same computer ende to solve acoustic and electromagnetic propagation problems in general inhomogeneous media. The finite difference program shown in Section 9.9.2 can easily be adapted to simulate electromagnetic wave propagation based on the Debye model, as we shall see in Section 8.3.2. Examples of simulations using the acousticelectromagnetic analogy can be found in Carconic and Cavallin (1995b).

8.3 A viscoelastic form of the electromagnetic energy

The electromagnetic Univ-Poynting theorem can be resinterpreted in the light of the theory of viscoelasticity in order to define the stored and dissipated energy densities in the time domain. A simple dielectric-relaxation model equivalent to a viscoelastic nechanical model flustrates the analogy, that identities electric field with stress, electric displacement with strain, dielectric primittivity with reciprical helk models, and resistance with viscosity.

For three-harmonic fields with time dependence $\exp(\pi xt)$ equations (8.1) and (8.2) read

$$\nabla \propto \mathbf{E} = -6\pi \mathbf{B}$$
, (8.58)

$$\nabla \times \mathbf{H} = i\omega \mathbf{D} + \mathbf{J}'_{ij}$$
(8.59)

respectively, where \mathbf{E} , \mathbf{D} , \mathbf{H} and \mathbf{B} are the corresponding time-harmonic fields and we have neglected the magnetic source. For convenience, the field quantities, source and medium properties are denoted by the same signification in both, the time and the frequency domains.

For harmonic helds, the constitutive equations (8.3), 18-11 and (8.5) read

$$\mathbf{D} = \mathcal{F}[\partial_t \epsilon] \cdot \mathbf{E} + \epsilon \cdot \mathbf{E}, \qquad (8.60)$$

$$\mathbf{B} = \mathcal{F}[\partial_t \mu] \cdot \mathbf{H} + \mu \cdot \mathbf{H}, \quad (8.61)$$

and

$$\Gamma = \mathcal{F}[\partial_t \sigma] (\mathbf{E} + \mathbf{J} - \sigma - \mathbf{E} + \mathbf{J})$$
 (8.62)

where $\mathcal{F}_{i} = [$ is the Fourier-transform operator

8.3.1 Umov-Poynting's theorem for harmonic fields

For scalar product of the complex conjugate of equation (8.52) with \mathbf{E}_i as refidiv ($\mathbf{E} \times \mathbf{H}^*$) = $(\nabla \times \mathbf{E}) \cdot \mathbf{H}^* - \mathbf{E} \cdot (\nabla \times \mathbf{H}^*)$, and substitution of equation (8.58) gives Funce Phynicals theorem for harmonic fields

div
$$\mathbf{p} = \frac{1}{2} \mathbf{J}^{(2)} \cdot \mathbf{E} = 2i\omega \left(\frac{1}{4} \mathbf{E} \cdot \mathbf{D}^{2} - \frac{1}{4} \mathbf{B} \cdot \mathbf{H}^{2} \right).$$
 (8.63)

where

$$\mathbf{p} = \frac{1}{2} \mathbf{E} \times \mathbf{H}^{*} \qquad (8.64)$$

is the complex Unev-Poynting vector.

Without loss of generality regarding the energy problem, we consider an isotropic medium, for which, $\epsilon = (-\mathbf{l}_1, |\boldsymbol{\mu}| = p/\mathbf{l}_1)$ and $\boldsymbol{\sigma} = \pi/\mathbf{l}_3$. Then, substitution of the constitutive equations (3.66), (3.66), and (3.62) into equation (3.66) yields

div
$$\mathbf{p} = 2i\omega \left(\frac{1}{4} \mathbf{r}^2 |\mathbf{E}|^2 + \frac{1}{4} \mu |\mathbf{H}|^2\right).$$
 (8.65)

where:

$$\epsilon = i - \int_{-\infty}^{1} \sigma_i$$
 (8.6b)

and we have assumed $\mathbf{J} = 0$. Each term has a precise physical meaning on a time average basis:

$$\frac{1}{4} \operatorname{Ret}^{*}(\mathbf{E}^{(2)} | \mathbf{E}^{(2)} - \frac{1}{4} \operatorname{Ret}(\mathbf{E}^{(2)} - \langle E_{1}^{(2)} \rangle)$$
(8.67)

is the time-averaged electric energy density.

$$\frac{1}{2} \ln(\epsilon^*) \mathbf{E}^* = -\frac{1}{2} \ln(\epsilon) \mathbf{E}^* = (\hat{D}_i)$$
(8.68)

is the time-averaged rate of dissipated electric-energy density,

$$\frac{1}{4} \text{Re}(\mu) |\mathbf{H}|^2 \equiv \langle F_{\mu} \rangle \qquad (8.69)$$

is the time averaged magnetic energy density, and

$$\frac{\pi}{2} \ln (\mu) |\mathbf{H}|^2 = \langle \hat{D}_{m} \rangle \qquad (8.70)$$

is the time-averaged rate of dissipated magnetic-energy density. Substituting the privading expressions into equation (S fait, yields the energy-balance equation

div $\mathbf{p} = 2i\omega(\langle L_{0} \rangle - \langle E_{0} \rangle) + \langle \hat{D}_{0} \rangle + \langle \hat{D}_{0} \rangle = 0.$ (8.71)

This equation is equivalent to (4.57) for viscoelastic media, and, particularly, to (7.190) for pore-viscoelastic media, since the magnetic-energy loss is equivalent to the kineticenergy loss of Biot's theory. The immus sign in equation (8.70) and the condition that $\omega(D_{\mu}) = \langle \hat{D}_{\mu} \rangle \rightarrow 0$, where $\langle D_{\mu} \rangle$ is the time-overaged dissipated-energy density, explosible $\log \mu < 0$. Using (8.66), explation (8.71) can be rewritten in terms of the dielectric and conductive energies as

$$\dim \mathbf{p} = 2i\omega_{i}(\{E_{i} + E_{i}\}) - \{E_{in}\}\} + \{\hat{D}_{i}\} + \{\hat{D}_{in}\} + \{\hat{D}_{inn}\} = 0, \quad (8.72)$$

where

$$\frac{1}{4} \operatorname{Re}(\ell^{*}) |\mathbf{E}|^{2} = \frac{1}{4} \operatorname{Re}(\ell) |\mathbf{E}|^{2} + \langle E_{\ell} \rangle$$
(8.73)

is the time averaged dielectric energy density.

$$\frac{1}{2}\ln(c^*)|\mathbf{E}|^2 = -\frac{1}{2}\ln(c)|\mathbf{E}|^2 = \langle D_b \rangle$$
(8.73)

is the time-overaged rate of dissipated dirlectric-mergy density,

$$\frac{1}{12} \ln(\sigma) \mathbf{E}^{(l)} \equiv \langle E_{l'} \rangle \tag{8.75}$$

is the title-averaged conductive-energy density, and

$$\frac{1}{2} \operatorname{Re}(\sigma) \mathbf{E}^{(2)} = \langle \hat{D}_{r} \rangle \tag{8.76}$$

is the three-averaged rate of designated conductive-energy density, with

$$(E_i) = (E_i) + (E_i)$$
 and $(\hat{D}_i) = (\hat{D}_i) + (\hat{D}_i)$. (8.77)

The Finot-Privating theorem priordes a consistent formulation of energy Env, but this does not preclude the existence of alternative formulations. For instance - h freys (1993) gives an alternative energy balance, implying a new interpretation of the Univ-Poynting vector (see also the discussion in Robinson (1994)) and Jeffreys (1994)).

8.3.2 Uniov-Poynting's theorem for transient fields

As we have seen in the previous section, time averaged energies for harmonic fields are precisely defined. The definition of stored and dissipated energies is particularly controversial in the time domain (Oughstum and Sherman, 1984), since different definitions may give the same time-averaged value for harmonic fields (Cavigha and Morro, 1992). We present in this section a definition, based on viscorlasticity theory, where energy can be separated between stored and dissipated in the time domain. The energy expressions are consistent with the mechanical-model description of constitutive equations.

Let us consider an arbitrary time dependence and the difference between the scalar product of explation (8.1) with \mathbf{H} and (8.2) with \mathbf{E} . We obtain the Unov-Poynting theorem for transient fields

$$-div |\mathbf{p} = \mathbf{J}^{t} \cdot \mathbf{E} + \mathbf{E} \cdot \partial_{t} \mathbf{D} + \mathbf{H} \cdot \partial_{t} \mathbf{B}$$

(8.78)

Since dielectric energy is analogous to strain energy, let us consider a stored dielectric-(free-) energy density of the form (2.7),

$$F(t) = \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} K(t - \varepsilon_1, t - \varepsilon_2) d\varepsilon_1 \mathbf{D}(\varepsilon_1) \cdot \partial_{\varepsilon_1} \mathbf{D}(\varepsilon_2) d\varepsilon_2 d\varepsilon_2$$
 (8.49)

Note that the electric displacement **D** is explicitle to the strain field, since the electric field is explicitlent to the stress field and the dielectric permittivity is equivalent to the compliance (see explatings (8.31)/(8.35)). The underlying assumptions are that the dielectric properties of the medium do not vary with time from aging material), and, as in the lessless case, the energy density is quadratic in the electric field. Moreover, the expression includes a dependence on the kistory of the electric field.

Differentiating E. yields

$$\partial_t E_i = \partial_t \mathbf{D} \cdot \int_{-\infty}^t \mathbf{K} (t - z_0) \partial_t \mathbf{E} \cdot \mathbf{D} (z_0) dz_0$$
$$- \frac{1}{2} \int_{-\infty}^t \int_{-\infty}^t \partial_t K (t - z_0, t - z_0) \partial_t \cdot \mathbf{D} (z_0) + \partial_{z_0} \mathbf{D} (z_0) dz_0 dz_0.$$
(8.80)

The constitutive equation [8/3) for isotropic racho can be rewritten as

$$\mathbf{E} = \beta \ast \partial_t \mathbf{D}, \quad (8.81)$$

where 3.11 is the dielectric-importneability function, satisfying

$$\partial_t \phi \partial_t \beta = \phi_t \eta_t + \phi^{-1} \beta_{t_t} + \phi^{-1} \eta_t + 1, \quad (121)(120 - 1)$$
 (8.82)

with the sobindues ∞ and 0 corresponding to the limits $t \to 0$ and $t \to \infty$, respectively. If J(t) has the form

$$b(t) = K(t, 0)H(t),$$
 (8.83)

where H(t) is Heaviside's function. One,

$$\int_{-\infty}^{t} K(t - \varepsilon_{t}, 0) d\tau |\mathbf{D}(\varepsilon_{t}) d\varepsilon_{t} \Rightarrow \mathbf{E}(t).$$
(8.84)

and (S.S0) factures

$$\mathbf{E} \cdot \partial_t \mathbf{D} = \partial_t F_1 + D_1$$
 (8.85)

where

$$\hat{D}_{\epsilon}(t) = -\frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} dt K(t - \tau_{\epsilon}(t - \tau_{\epsilon})) \Theta_{\epsilon} \mathbf{D}(\tau_{\epsilon}) \cdot \partial_{\epsilon} \mathbf{D}(\tau_{\epsilon}) d\tau_{\epsilon} d\tau_{\epsilon}$$
(8.86)

is the rate of dissipated dielectric-energy density. Note that the relation (S.S.) does not determine the stored energy, i.e., this can not be obtained from the constitutive equation However, if we assume that

$$K(t, \tau_i) = ((t + \tau_i),$$
 (8.87)

such that 3 is defined by the relation

$$\delta(t) = 0 t(H(t))$$
 (8.88)

this choice will suffice to determine K, and

$$E_{\ell}(t) \geq \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} \eta(2t - \tau - \tau_{\ell}) \partial_{\eta} \mathbf{D}^{\ell}(\tau) \cdot \partial_{\tau} \mathbf{D}(\tau_{\ell}) d\tau_{\ell} d\tau_{\ell}, \qquad (8.89)$$

8.3 A viscoclastic form of the electromagnetic energy

$$\tilde{D}_{i}(t) = -\int_{-\infty}^{t} \int_{-\infty}^{t} \partial_{i} h(2t - v_{i} - v_{2}) \partial_{i} |\mathbf{D}(v_{i}) \cdot \partial_{i_{2}} \mathbf{D}(v_{2}) dv dv_{2}, \qquad (8.90)$$

where ϑ denotes differentiation with respect to the argument of the corresponding function. Equation (8.87) is emisistent with the theory implied by michanical models (Christenson, 1982). Brener and Onar (1964) discuss some realistic requirements from which $K(t, \tau)$ must have the reduced form $|0|t| < \tau$.

Let us calculate the time average of the stared energy density for harmonic fields using equation (8.89). Although $\mathbf{D} \in \infty$) does not vanish, the transient contained in (8.8%) vanishes for sufficiently large times, and this equation can be used to compute the average of time-harmonic fields. The change of variables $z_1 \rightarrow t - z_2$ and $z_2 \rightarrow t - z_2$ yields.

$$L_{0}(t) = \frac{1}{2} \int_{0}^{\infty} \int_{0}^{\infty} \beta(\tau_{0} + \tau_{0}) d\mathbf{D}(t - \tau_{0}) d\mathbf{D}(t - \tau_{0}) d\tau d\tau_{0}, \qquad (8.91)$$

Using (1.105), the time average of optation (8.91) is

$$\langle E_i \rangle = \frac{1}{4} \omega^2 \mathbf{D}^{(2)} \int_0^\infty \int_0^\infty |\theta(\tau_i + \tau_i) \cos(\omega(\tau - \tau_i)) d\tau| d\tau_i$$
 (8.92)

A new change of variables $y = \tau_0 + \tau_0$ and $x = \tau_0 - \tau_0$ gives

$$\langle I_{\alpha\beta} \rangle = \frac{1}{8} \omega^2 \mathbf{D}^{\alpha\beta} \int_{-\infty}^{\infty} f^{\alpha\beta}_{\alpha\beta} \theta(u) \cos(\omega v) dv dv = \frac{1}{4} \omega \mathbf{D}^{\beta\beta} \int_{0}^{\infty} |\langle u| \sin(\omega v) du\rangle = -(8.93)$$

From opaction (S.SS) and using integration by parts, we have that

$$\operatorname{Re}[\mathcal{F}(\partial_t \beta] = \operatorname{Re}[(0, \pi)] = \beta(\infty) + \omega \int_0^\infty |\beta(t) - 0(\infty)| \sin(\pi t) dt, \qquad (8.93)$$

Using the property

$$\star \int_{a}^{\infty} \sin(\omega t) dt = 1, \qquad (8.95)$$

we obtain

$$\operatorname{Re}[\beta(\omega) = \omega \int_{0}^{\infty} |\langle t \rangle \operatorname{sm}(\omega t) dt, \qquad (8.96)$$

Substituting (8.96) into equation (8.93), and using $\mathbf{E} = (0, \varepsilon) \mathbf{D} \in (0, \varepsilon) = \mathcal{F} \partial_t \delta(t)$, see equation (8.81)), and equation (8.82), we heatly get

$$\langle T_0 \rangle = \begin{bmatrix} \mathbf{D}^2 \operatorname{Re}(3) \varphi \\ - \end{bmatrix} \operatorname{Ree}(\mathbf{E}^2).$$
 (8.97)

which is the expression (8.73). A similar calculation shows that $\langle \hat{D}_{ij} \rangle$ is equal to the expression [8.74].

Similarly, the magnetic term on the right-hand side of equation (8.78) can be recasted as

$$\mathbf{H} \cdot \partial_t \mathbf{B} \approx \partial_t E_m + D_m$$

(8.98)

where

$$F_{00}(t) = \frac{1}{2} \int_{-\infty}^{t} \int_{-\infty}^{t} \gamma^{2} \mathcal{D} = \tau_{0} - \tau_{2} \partial \delta_{0} \mathbf{B}^{\dagger} \tau_{0} + \partial \epsilon_{0} \mathbf{B}(\tau_{2}) \partial \tau_{0} d\tau_{2}$$
(8.99)

$$\hat{D}_{in}(t) = -\int_{-\infty}^{t} \int_{-\infty}^{t} d\tau \left(2t - \tau_{0} - \tau_{0} \partial_{\tau} \mathbf{B}(\tau_{0}) + \partial_{\tau} \mathbf{B}(\tau_{0}) d\tau_{0} d\tau_{0}\right) = -\sqrt{8} |1000$$

are the stored magnetic-energy density and rate of dissipated magnetic-energy density, respectively, such that

$$\mathbf{H} = \gamma * \partial_t \mathbf{B}, \quad \gamma(t) = \gamma(t) H(t), \quad (8.101)$$

with a the magnetic-mapermeability function.

The rate of dissipated conductive-energy density can be defined as

$$\hat{D}_{\sigma}(t) = -\int_{-\infty}^{t} \int_{-\infty}^{t} \sigma(2t - \tau - \tau_{t}) \partial_{\tau_{t}} \mathbf{E}(\tau_{t}) \cdot \partial_{\tau_{t}} \mathbf{E}(\tau_{t}) d\tau d\tau_{t}, \qquad (8.102)$$

Formally, the stored energy density due to the electric currents not of phase with the electric field, $F_{\mu\nu}$ satisfies

$$\partial_t E_s = J^* \cdot E = \hat{D}_{s1}$$
 (8.103)

where

$$\Gamma = \sigma \star \partial_t \mathbf{E}, \qquad \sigma(t) = \sigma(t) H(t) \tag{8.101}$$

In terms of the energy densities, equation (8.78) becomes

$$-\operatorname{div} \mathbf{\mu} = \partial_t (E_i + E_s + E_m) + D_i + D_s + D_m, \quad (8.105)$$

which is analogous to equation (2.95). The correspondence with time-averaged quantumes are given in the previous section.

Note that $(\mathbf{J}^{0} \cdot \mathbf{E})$ is equal to the rate of dissipated energy density (\hat{D}_{n}) , and that

$$|\langle \phi_i E_i \rangle = 0$$
 (8.109)

The same property holds for the stored dectrics and magnetic-energy densities.

There are other alternative time domain expressions for the energy densities whose time-average values coincide with these given in Section 8.3.1, but fail to match the energy in the time domain. For instance, the following definition

$$L_{s}^{\prime} = \frac{1}{2} \mathbf{E} \cdot \mathbf{D}, \qquad (8.107)$$

as the stored dielectric-energy density, and

$$\partial_t^i = \frac{i}{2} \left(\mathbf{E} \cdot \partial_t \mathbf{D} - \mathbf{D} \cdot \partial_t \mathbf{E} \right)$$
(8.108)

as the rate of dissipation, satisfy equation (8.105) and $\langle E_i^* \rangle = \langle E_i \rangle$ and $\langle D_i^* \rangle = \langle D_i \rangle$. However, E_i^* is not equal to the energy stored in the capacitors for the Debye model given in the next section (see equations (8.120) and (8.122)). In the viscoelastic case (see Chapter 2), the definition of energy is consistent with the theory of mechanical models bucket promagnetism. On theory should be consistent with the theory of circuits, i.e., with the energy stored in the capacitors and the energy dissipated in the resistances

336

The Debye-Zener analogy

It is well known that the Debye model used to describe the behaviour of dielectric materials (Hippel, 1962) is mathematically equivalent to the Zener or standard-linear-solid model used in viscoelasticity (Zener, 1948). The following example uses this equivalence to diustrate the concepts presented in the previous section

Let us consider a capacitor C_{2} in parallel with a series connection between a capacitor C_{2} and a resistance R. This encode dows the following differential equation:

$$U + \tau_c \partial_t U = \frac{1}{C} (I + \tau_c \partial_t I),$$
 (8.109)

where $U = \partial V/\partial t$. F is the current, V is the voltage,

$$C = C + C_0 = \tau_0 - B \left(\frac{1}{C} + \frac{1}{C_2}\right)^{-1}, \quad \tau_t = C_t R_t$$
 (8.110)

From the paint of view of a pure diffective process, we identify l_{\perp} with E and l with D (see Figure 8.3).





Hence, the dirlectric relaxation model is

$$\mathbf{E} + \tau_t \partial_t \mathbf{E} = \frac{1}{r^0} \left(\mathbf{D} + \tau_t \partial_t \mathbf{D} \right), \qquad (8.111)$$

where

$$\vec{c} = c_1 + c_2, \quad \tau_2 = \frac{1}{\sigma} \left(\frac{1}{c_1} + \frac{1}{c_2} \right)^{-1}, \quad \tau_2 = c_2 \sigma, \quad (8.112)$$

with σ the conductivity. Note that ϕ^{α} is the starre (low-frequency) dielectric permittivity and $e^{\alpha} = e^{\alpha} \sigma/(\pi s) + e_{\sigma} < \psi^{\alpha}$ is the optical (high-frequency) dielectric permittivity.
We have that

$$e(t) = e^{t} \left[1 - \left(1 - \frac{\tau_{t}}{\tau_{t}^{s}} \right) \exp(-t/\tau_{t}) \right] H(t),$$
 (8.113)

$$u(t) = h(t)H(t) = \frac{1}{e^{\theta}} \left[1 - \left(1 - \frac{7\theta}{2g}\right) \left(\operatorname{sp}(-d_1 \tau_{\mathcal{C}}) \right] H(t) \right]$$
(8.114)

and

$$\epsilon(\omega) = i \cdot (\omega) - \epsilon^2 \left(\frac{1 - i\omega \gamma}{1 + i\omega \tau_C} \right)$$
(8.115)

Equation (8.155) can be rewritten as

$$\epsilon(\omega) = \epsilon^{\infty} + \frac{c^2 - \epsilon^{\infty}}{1 + i\omega_B}$$
(8.116)

The dielectric permittivity (8.116) describes the response of palar molecules, such as water, to the electromagnetic field (Delice, 95.9) former and Siggues (994).

Substituting (S.114) into equating (8.81) and defining the internal variable

$$\boldsymbol{\xi}(t) = \phi(\mathbf{x}|t) + t/\gamma_{\mathcal{F}}(H(t) + \mathbf{D}(t)), \quad \phi = \frac{1}{t^{n}\gamma_{\mathcal{F}}} \left(1 - \frac{\tau_{\mathcal{F}}}{cr}\right), \quad (8.117)$$

yi) lds

$$\mathbf{E} = \frac{1}{e^{N}} \mathbf{D} + \boldsymbol{\xi}, \qquad (8.118)$$

where $\boldsymbol{\xi}$ satisfies

$$\partial_t \boldsymbol{\xi} = \alpha \mathbf{D} - \frac{\boldsymbol{\xi}}{2\epsilon},$$
 (8.149)

The dielectric-storgy density is that stored in the capacitors:

$$F_{i} = \frac{2}{2r_{0}} \mathbf{D} + \mathbf{D}_{i} + \frac{1}{2r_{0}} \mathbf{D}_{i} + \mathbf{D}_{i}$$
 (8.125)

where \mathbf{D}_{i} and \mathbf{D}_{i} are the respective electric displacements. Since $\mathbf{D}_{i} = r_{i}\mathbf{E}$, $\mathbf{D} = \mathbf{D}_{i} - \mathbf{D}_{i}$ and $e^{i} = e_{i}$, we obtain

$$D_{1} = e^{\infty} \xi_{1}$$
 (8.121)

where equation (8.118) has been used. Note that the internal variable ξ is closely related to the electric field acting on the capacitor in series with the dissipation element. Substituting of \mathbf{D}_1 and \mathbf{D}_2 acts equation (8.120) and after some calculations withs

$$E_{n} = \frac{e^{\infty}}{2} \left[\left(\frac{e^{\infty}}{e^{n} + e^{\infty}} \right) \boldsymbol{\xi} \cdot \boldsymbol{\xi} + \mathbf{E} \cdot \mathbf{E} \right]$$
(8.122)

Let as verify that equation (8.122) is in agreement with equation (8.89). From equations (8.114) and (8.117) we have

$$\beta(t) = \frac{1}{t^0} - \phi_{TF} \exp(-t/t_F)$$
 (8.123)

338

Replacing (8,123) into equation (8.89) and after some algebra yields

$$E_{\theta} = \frac{1}{2\theta} \mathbf{D} \cdot \mathbf{D} - \frac{1}{2} (a_{S} \exp(-t_{e} z_{F}) H(t) + \partial_{\theta} \mathbf{D}(t))^{2}$$

$$(8.124)$$

where the exponent 2 means the scalar product of sing equations (8,117) and (8,119) gives

$$E_{i} = \frac{1}{2^{ij}} \mathbf{D} \cdot \mathbf{D} = \frac{1}{2\alpha^{ij}} \left(\alpha_{ij} \mathbf{D} - \boldsymbol{\xi} \right) \cdot \left(\alpha_{ij} \mathbf{D} - \boldsymbol{\xi} \right)$$
(8.125)

Since $(\gamma_{22} + e^{i}\gamma_{12})$ and a low calculations show that the expression in (8.125) is equal to the stored energy density (8.122). This expression of a dscale obtained by avoiding the use of internal variables. Bowever, the introduction of these variables is a requirement to obtain a complete differential formulation of the electromagnetic equations. This formulation is the basis of most simulation algorithms (Carriene (1996)) Nu and McMerban (1997).

The rate of dissipated dielectric-energy density is

$$\hat{D}_{c} = \frac{1}{\sigma} \partial_{t} \mathbf{D} + \partial_{t} \mathbf{D}$$
, (8.126)

which from equation (\$119) and (\$121) becomes

$$D_{t} = \frac{1}{\sigma} \left(\frac{e^{s}}{e_{t}} \right)^{2} (\phi \tau_{t} \mathbf{D} - \boldsymbol{\xi}) \cdot (\phi \tau_{t} \mathbf{D} - \boldsymbol{\xi}), \qquad (8.127)$$

Taking into account the previous calculations, it is easy to show that substitution of equation (8.123) into (8.90) gives equation (8.127)

The Zener model has been introduced in Sections 2.4.3 and 2.7.3. In this case, the free (stored) energy density can be uniquely determined (Cavallini and Carebrae, 1994). The relaxation function and complex modulus are given in equations (2.173) and (2.170), respectively.

$$\psi(t) = M_P \left[1 - \left(1 - \frac{\tau_e}{\tau_e}\right) \exp(-t/\tau_e) \right] H(t), \qquad (8.128)$$

and

$$M(\omega) = W_{W} \left(\frac{1 + i\omega \pi}{1 + i\omega \gamma} \right), \qquad (8.129)$$

where $M_{W_{c}}$, z and z_{c} are defined in equations (2.168) and (2.169), respectively. Equation (8.129) can be rewritten as

$$M^{-1}(\omega) = M_{t}^{-1} + \frac{M_{t}^{-1} - M_{t}}{1 + i\omega \pi}$$
(8.130)

where $M_{f} = M_{f} z_{a} z_{b}$. The manory variable is given in equation (2.283):

$$\zeta(t) = \varepsilon_0 \exp(-t_0 z_s) H(t) [+\epsilon(t), \quad \varepsilon_0 = \frac{M_R}{z_s} \left(1 - \frac{z_s}{z_s}\right), \quad (8.131)$$

and the field variables satisfy opportion (2.283).

$$\sigma = M_{eff} + \xi_{eff}$$
(8.132)

and equation (2.286).

$$\partial_t \xi = \rho_t (-\frac{\xi}{z_0})$$
(8.133)

Assuming that the strain energy is stored in the springs, we have that

$$V = \frac{1}{2} (k_{\perp} c_{\perp}^2 + k_{2} c_{2}^2) \qquad (8.134)$$

where c_i and c_j are the dilatations of the springs (see Figure 2.8), and k_j and k_j can be expressed from equations (2.168) and (2.169) as

$$k_i = M_l$$
 and $k_i = \frac{M_l M_h}{\varepsilon \sigma_s}$ (8.135)

Since $\sigma = k_1 \epsilon_1$ and $\epsilon = \epsilon_1 + \epsilon_2$, and using (2.283), we obtain

$$e^{-1}: e^{-1} = \frac{\xi}{M_0}$$
 and $e_2: e^{-1} = \frac{\xi}{M_0}$. (8.139)

Note that the memory variable f is closely related to the dilatation on the spring that is in parallel with the dashpot. Substitution of the dilatations into expection (8,134) yields

$$\mathbf{1} = -\frac{Q_I}{2} \left[\left(\epsilon + \frac{\xi}{M_I} \right)^2 - \frac{Q_R}{\varphi_0 \sigma_c} \left(\frac{\xi}{M_I} \right)^2 \right], \qquad (8.137)$$

which, after some calculations, can be rewritten as

$$V = \frac{1}{2}M_{B}t^{2} - \frac{1}{2\varphi_{\sigma}\sigma_{\sigma}}(\varphi_{\sigma}z_{\sigma}t - \xi)^{2}$$
(8.138)

On the other hand, the rare of energy density dissipated in the dashpot of viscosity q is

$$D = \eta \left(\frac{\partial b_{\beta}}{\partial t}\right)^{2} \qquad (8.139)$$

which from equations (8,133) and (8,136) becomes

$$D = \frac{a}{(\gamma_{0})^{2}} (\varphi_{0} \tau_{0} \epsilon - \xi)^{2}, \quad (8.140)$$

The mathematics of the viscoelastic problem is the same as for the chelectric relaxation model previously introduced, since equations $(8,114) \cdot (8,119)$ are expressed to $(8,128) \cdot (8,143)$ and equations (8,125) and (8,127) are equivalent to (8,138) and (8,140), respectively. The mathematical equivalence identities electric vector \mathbf{E} with stress σ and electric displacement \mathbf{D} with strain c. The complete correspondence between the dielectric and the viscoelastic models is

	Fields		Properties			
Е	< :	σ.	· '	0.5	M_{μ}^{\pm}	
D	< >	· · ·	15	< >	M_{c}^{+1}	
F.	⇔	σ_{i}	10	⇔		(x 11) i
Е,	< >	σ_{i}	77	< >	7.4	(0.1111
\mathbf{D}_{1}	4.8	11		<>	<i>k</i> . 1	
\mathbf{D}_{2}	4.5		ϕ_{ij}	< >	k. 1	
ξ	C >	5	.7	c >	9	

where some of the symbols can be identified in Figures 2.8 and 8.3.

3 III

The Cole-Cole model

Equation (8.116) can be generalized as

$$\epsilon(\pm 1 \dots)^{N_{n-1}} \le \frac{\epsilon^{n} - \epsilon^{N_{n-1}}}{1 + (i_{n-1} + i_{n})},$$
 (8.142)

where y = m/n, with *m* and *n* positive, integer and prime, and m < n. This model has been introduced by Cole and Cole (1941). The corresponding frequency- and timedomains constitutive equations are

$$\mathbf{D} = \begin{bmatrix} e^{\mathbf{u}} + e^{\mathbf{x}} (\mathbf{i}_{x} \cdot \mathbf{p})^{\mathbf{v}} \\ -1 + (\mathbf{i}_{x} \cdot \mathbf{p})^{\mathbf{v}} \end{bmatrix} \mathbf{E}$$
(8.1.03)

and

$$\mathbf{E} = [\hat{j}\frac{\partial^{2}\mathbf{E}}{\partial m} = \frac{1}{c}\left(\mathbf{D} - i\hat{j}_{i}\frac{\partial^{2}\mathbf{D}}{\partial m}\right), \qquad (8.113)$$

where $\beta_{ij} = e^{i t} \beta_{j} r^{\mu}$, and $i \theta_{j} d\theta_{j}$ is the fractional derivative of order η (see Section 2.5.2). Equation (8.144) is a generalization of (8.144).

The rational power of the imaginary unit (i) in equations (8.142) and (8.143) is a multi-valued function and implies a number n of different physically accepted values of the dielectric permittivity. As a consequence, a time-harmonic wave is split into a set of waves with the same beguency and slightly different wavelengths which interfere and disperse (Caputo, 1998, Beffore and Caputo, 2000). The expression (8.142) is also called the generalized Debye form of the delectric permittivity, and the Debye-Zeper opology (8.141) can also be applied to the CabsCole model.

The fractional derivative is a generalization of the derivative of natural order by using Cauchy's well-known formula. For a given function f(t), the fractional derivative is given by

$$\frac{\partial^{2} f}{\partial t^{2}} = f(t) + \Phi_{-i}(t), \quad \text{where} \quad \Phi_{i}(t) = \frac{t^{2}}{1/(q)}, \quad t^{i-1} = \begin{cases} t^{i-1}, & t > 0, \\ 0, & t < 0, \end{cases}$$
(8.145)

and F is Euler's Gamma function (Caputo and Mainardi, 1971). If $\Phi_{\pm j} = e^{ij}(t)$, j = 0, 1, 2, ..., where i is Dirac's function, equation (8.145) gives the p-order derivative of f(t). Caputo and Mainardi (1971) have shown that

$$\epsilon(t) = \{e^{0} + (e^{0} - e^{0})E_{0}^{(1)}(t/\tau_{0})^{\theta}\}H(t).$$
 (8.146)

where

$$L_{q}(\tau) = \sum_{k=0}^{\infty} \frac{\tau^{k}}{\Gamma(qk+1)}$$
(8.147)

is the Mittag-Leffler function of molet q_i introduced by Gösta Mittag-Leffler in 1904 (note the similarity with the Wright function (4.212)). It is a generalization of the exponential function, with $E_i(\tau) = \exp(\tau)$ (e.g., Podlubry, 1999). Equation (8.146) becomes equation (8.143) for q = 1.

8.4 The analogy for reflection and transmission

be this section, we obtain a complete parallelism for the selfection and refraction (transmission) problem, considering the most general situation, that is the presence of anisotropy and attendation. A viscosity in the acoustic case and conductivity in the electromagnetic case if (an ione and Robinson, 2002). The analysis of the electro-solid theory of cellection applied by George Green to high waves (Green, 1842) and a brief historical review of wave propagation through the other. Further illustrate the analogy,

Let us assume that the incident, reflected and reflected waves are identified by the superscripts I, R and T. The boundary separates two linear viscoelastic and monoclinic media. The upper median is defined by the suffices are provided the complex permittivities c_0 and magnetic permeability p. The lower medium is defined by the concesponding primed quantities. Let us denote by R and β the propagation and attravation angles, and by γ the Upper fit of the control by γ the Upper fit of the Upper fit of the transfer permeability p and β the propagation and attravation angles, and by γ the Upper fit of the propagation and referse fit.

The analogy can be extended to the boundary conditions at a surface of discontinuity, say, the Letter-plane, because according to equation [8,11] continuity of

in the acoustic case, is equivalent to continuity of

$$F = \text{and} (-H_0)$$
 (8.149)

in the electromagnetic case. The field variables in [8,149] are precisely the tangential components of the electric and magnetic vectors. In the absence of surface current densities at the interface, the boundary conditions mapose the community of those components (Born and Wolf, 1964, p. 4).

The SII reflection-transmission problem is given in Section 6.1, where the Zenri model is used to describe the attenuation properties. In the case of an incident inhomogeneous plane wave and a gravital stiffness matrix \mathbf{P}_{i} the relevant reputities are summarized in the following section:

8.4.1 Reflection and refraction coefficients

The particle velocities of the reflected and refracted waves are given by

$$e_{I}^{R} \rightarrow i_{s} R \exp[i\omega(t + s/r + s_{1}^{R})]$$

(8.150)

and

$$(1 - i\omega T \exp(\omega tt - s_0 x - s_0^2 x))$$
 (8.151)

respectively, and the reflection and refraction (transmission) coefficients are

$$R = \frac{Z^{T} - Z^{T}}{Z^{T} + Z^{T}} = T = \frac{2Z^{T}}{Z^{T} + Z^{T}}$$
(8.152)

where

$$Z^{t} = p_{0,S_{1}} + p_{11}s_{1}^{t} = Z^{t} - p_{0,S_{1}}^{t} + p_{1}^{t}a_{1}^{t}, \qquad (8.153)$$

with

$$s^R = s^1 - s^I_1 - s - (Suell's law),$$
 (8.153)

$$-s_{\pm}^{q} = \left(s_{\pm}^{p} + \frac{2p_{2q}}{p_{11}}s_{1}\right), \qquad (8.155)$$

830

$$s_{0}^{T} = \frac{1}{p_{c0}^{T}} \left(-p_{c0}^{T} s_{0}^{T} + p s_{V} p_{c0}^{T} - p_{c}^{T} s_{0}^{T} \right), \qquad (8.156)$$

with

$$p^{tT} \Rightarrow p_{ty}^{t} p_{ty}^{t} - p_{ty}^{t} p_{ty}^{t}$$
 (8.157)

(For the principal value, the argument of the square root has between $-\pi/2$ and $-\pi/2$). As indicated by Krebes (1984), special case is needed when choosing the sign, since a wrong choice may lead to discontinuities of the vertical wavenumber as a function of the incidence angle.

Propagation, attenuation and ray angles

Fig each plane ways.

$$\tan \theta = \frac{\operatorname{Re}(s_0)}{\operatorname{Re}(s_0)}, \quad \tan \phi = \frac{\operatorname{Im}(s_0)}{\operatorname{Im}(s_0)}, \quad \tan \phi = \frac{\operatorname{Re}(X)}{\operatorname{Re}(Z)}, \quad (8.158)$$

where

$$\begin{aligned} X^{I} &= \rho_{0} s^{-1} + \rho_{0} s^{I}_{0} \\ X^{R} &= \rho_{0} s^{-1} + \rho_{0} s^{R}_{0} \\ X^{I} &= p_{0}^{I} s^{-1} + p_{0}^{I} s^{I}_{0} \end{aligned} \tag{8.159}$$

The ray angle denotes the direction of the power-flow vector Retp), where **p** is the Uniov-Powering vector (6.9).

Energy-flux balance

The balance of energy for involves the contributy of the normal component of the Unice-Powning vector across the interface. This is a consequence of the homodory conditions that impose continuity of normal stress $\sigma_{n'}$ and particle velocity v_n . The balance of power flow at the interface, on a time average basis, is given in Section 6.1.7. The equation are

$$\langle p^{\ell}\rangle + \langle p^{\ell}\rangle + \langle p^{\ell''}\rangle - \langle p^{\ell'}\rangle,$$
 (8.160)

where

$$\langle p^{I} \rangle = -\frac{1}{2} \operatorname{Re}(\sigma_{12}^{I} c_{12}^{I}) = \frac{1}{2} \omega^{2} \operatorname{Re}(Z^{I}) \exp(2\omega \operatorname{Im}(s)) x$$
 (8.161)

is the incident flux.

$$\langle g^{R} \rangle = -\frac{1}{2} \operatorname{Ret} \sigma^{R}_{A} v^{R^{*}}_{A} v - \frac{1}{2} \omega^{T} B^{*} \operatorname{Ret} Z^{R}_{+} \exp 2\omega \operatorname{Int} s_{0} \phi$$
 (8.162)

is the reflected flux.

$$\langle p^{IR} \rangle = -\frac{1}{2} Rr (\sigma_{ij}^I r_j^{R'} + \sigma_{ij}^R r_j^{R'}) = \mathscr{A} \ln(R \operatorname{dm}(Z^I)) \exp 2 \mathscr{A} \operatorname{dm}(s_1(r))$$
(8.163)

is the interference between the modent and reflected normal fluxes, and

$$\langle p^{T} \rangle = -\frac{1}{2} \operatorname{Re}(\sigma_{R}^{T} r_{s}^{T}) - \frac{1}{2} \omega^{2} F^{T} \operatorname{Re}(Z^{T}) \exp(2\omega \operatorname{Im}(s_{s}) a)$$
 (8.164)

is the refracted this. In this lossless case, Z⁴ is real and the interference flux vanishes

8.4.2 Application of the analogy

On the basis of the solution of the SH-wave problem, we use the analogy to find the solution in the electromagnetic case. For every electromagnetic phenomenon — using the electromagnetic terminology — we analyze its corresponding mathematical and physical concategory in the accessing case. Maxwell (1891, p. 65), who used this approach writes. The analogy between the action of electromatics intersisting in producing the displacement of an elastic leady is so elemans that I have contained to call the ratio of electromatics interval to call the ratio of electromatics interval to the corresponding electric displacement the neighbor of electromatics in and of electric displacement the medium.

Refraction index and Fresnel's formulae

Let us assume a bisdess, isotropic medium. Isotropy implies $v_{ij} = v_{ij} = \mu$ and $v_{ij} = 0$ and $e_{ij} = e_{1j} = e_i$ and $e_{1j} = 0$. It is easy to show that, in this case, the reflection and refraction coefficients (8,152) reduce to

$$B = \frac{\sqrt{\rho}\rho\cos\theta^{\dagger} - \sqrt{\rho}\rho^{\dagger}\cos\theta^{\dagger}}{\sqrt{\rho}\rho\cos\theta^{\dagger} - \sqrt{\rho}\rho^{\dagger}\cos\theta^{\dagger}} \quad \text{and} \quad T = \frac{2\sqrt{\rho}\rho\cos\theta^{\dagger}}{\sqrt{\rho}\rho\cos\theta^{\dagger} - \sqrt{\rho}\rho^{\dagger}\cos\theta^{\dagger}}.$$
 (8.165)

respectively. From the analogy (optation (8.33)) and equation (8.30) we have

$$|\mu|^{\perp} \Leftrightarrow |i\rangle$$
 (8.165)

The refraction index is defined as the velocity of light in vacuum n_0 divided by the phase velocity in the median, where the phase velocity is the required of the real slowness. For lossless, isotropic media, the refraction index is

$$u = sr_0 = \sqrt{\frac{m}{p_{\rm eff}}}$$
 (8.167)

where $s = \sqrt{p}e$ is the slowness, and $c_{i} = 1/\sqrt{p_{i}x_{i}}$, with $r_{0} = 8.85$ (0) ± 12 in and $p_{0} \neq 4$. (0) ± 14 m, the dielectric premittivity and magnetic premerbility of fere space. In acoustic media there is not a limit velocity, but using the analogy we can define a refraction index

$$u_n = \nu \sqrt{\frac{p}{p}}.$$
(8.168)

where p is a constant with the dimensions of velocity. Assuming $p \in [p]$ in (8.165) and using (8.166), the electromagnetic coefficients are

$$H = \frac{\sqrt{c^2 \cos \theta^2}}{\sqrt{c^2 \cos \theta^2}} \frac{\sqrt{c^2 \cos \theta^2}}{\sin \theta} = \frac{2\sqrt{c^2 \cos \theta^2}}{\sqrt{c^2 \cos \theta^2}} \frac{\sqrt{c^2 \cos \theta^2}}{\sqrt{c^2 \cos \theta^2}} \frac{\sqrt{c^$$

In terms of the refraction index (8.167) we have

$$R = \frac{n^2 \cos \theta^I}{n^2 \cos \theta^I} - \frac{n \cos \theta^I}{n \cos \theta^I} \quad \text{and} \quad I = \frac{2n \cos \theta^I}{n^2 \cos \theta^I} - \frac{(8.170)}{n \cos \theta^I}$$

Equations (8.170) are Freshel's formulae, corresponding to the electric vector in the phase of incidence (Born and Wolf, 1963, p. 40). Bence, Freshel's formulae are mathematically equivalent to the SH-wave reflection and transmission coefficients for lossless, isotropic media, with no density contrast at the interface.

Brewster (polarizing) angle

Frisnel's formulae can be written in an alternative firm, which may be obtained from (S (50) by using Shell's law

$$\frac{\sin\theta^{d}}{\sin\theta^{d}} = \sqrt{\frac{\mu}{\mu^{\prime}}} + \frac{n_{\mu}^{\prime}}{n_{\mu}} + \sqrt{\frac{n^{\prime}}{r}} + \frac{n^{\prime}}{n}, \qquad (8.171)$$

It yields

$$R = \frac{\tan(\theta^{i} - \theta^{i})}{\tan(\theta^{i} + \theta^{i})} \quad \text{and} \quad T = \frac{2\sin(\theta^{i})\cos(\theta^{i})}{\sin(\theta^{i} + \theta^{i})\cos(\theta^{i} - \theta^{i})}.$$
(8.172)

The denominator in (8.172) as finite, except when $b^{i} + b^{i} = \pi/2$. In this case the followed and refracted cases are perpendicular to each other and R = 0. It follows from Suell's law that the incidence angle, $\theta_{B} = \theta^{i}$, satisfies

$$(a_0, b_0) = \cos(b^2) - \frac{p}{\sqrt{p^2}} - \frac{u_n^2}{u_n} - \frac{e^2}{\sqrt{e^2}} - \frac{u^2}{u}$$
 (8.173)

The angle θ_R is called the Brewster angle, first noted by Literine Malus and David Brewster (Brewster (1815)) see Section 6.1.5(10). It follows that the Brewster angle in elasticity ran he obtained when the medium is lossless and batropic, and the density is constant across the interface. This angle is also called polarizing angle, heranse, as Brewster states. When a polarised ray is incident at any angle upon a transpire at lody, in a plane at right angles to the plane of its primitive polarisation, a particular of the ray will lose us property of bring reflected, and will entropy ponetrie the transpire at lody. This partition of highly, which has host its effected day, merganes as the angle of methods to the polarisity planetation of the ray of highly the base has the property angle, when it becomes a merganeous as the angle of methods and polarizing angle, the electric vector of the reflected wave has no components in the plane of mergence.

The restriction about the density can be removed and the Brewster angle is given by

$$\tan \theta_0 = \sqrt{\frac{\rho c / \rho' - \rho'}{\rho' - \rho \rho' / \rho'}}$$
(8.173)

but $\theta^1 \mapsto \theta^2 \neq \pi_1 2$ in this case. The analogies (8.34) and (8.35) imply

$$\tan q_{H} = \frac{\mu \epsilon'/\epsilon - \mu'}{\mu' - \mu/\epsilon'}$$
(8.175)

in the electromagnetic case.

In the anisotropic and Jossless case, the angle is obtained from

$$\cot \theta_R = (-b + \sqrt{b^2 - 4ac})/(2a)$$
 (8.176)

where

$$a \in c_{11}(pc_{11} + pU_{11}^{*})/p_{\gamma} = b \oplus 2ac_{10}/c_{10},$$
 (S.177)

and

$$c = c_{10}^2 - c_{10}^2 - c_{11}^2 (p'c_{10} - m'c_{10}) c_{10}$$
 (8.178)

(see Section 6.1.5). If $c_{3n} \sim c_{2n}^2 = 0$, we obtain

$$\cos\theta_{B} = \frac{c_{11}(w_{12} - dc_{11}')}{\sqrt{c_{11}'(dc_{22} - dc_{22}')}}$$
(8.179)

or, using the analogy,

$$c_{i_1} \iff i_{i_2}$$

 $c_{i_2} \iff i_{i_2}$
(8.180)
 $p \iff p_i$

the Brewster angle is given by

$$\tan \theta_0 = \frac{1}{(-\sqrt{-\mu^2 c_0^2})^{\mu \nu^2 + (-\mu^2 c_0^2)}} = \frac{1}{(+\sqrt{-\mu^2 c_0^2 + (\mu^2 c_0^2)})}$$
(8.181)

In the lossy case $\tan \theta_B$ is complex, in general and there is no Brewster angle. However, let as consider equation (8,175) and incident homogeneous plane waves. According to the correspondence (8,52), its extension to the lossy case is

$$\gamma_{01} \theta_R = \sqrt{\frac{\mu \epsilon'/\epsilon - \mu'}{\mu' - \mu_{\perp} \epsilon'}} \tag{8.182}$$

The Brewster angle exists if c' is proportional to c, for instance, if the conductivity of the refraction medium satisfies $\sigma' = (c'/c)\sigma'(\eta' - (\mu'/\mu)\eta)$ in the acoustic case). This situation is unlikely to accur in reality nulless the interface is designed for this purpose

Critical angle. Total reflection

he isotropic lossless media, total reflection occurs when Shell's law

 $\sin\theta^{\mu} = \sqrt{\frac{\rho \theta^{\prime}}{\rho^{\prime} \mu}} \sin\theta^{\mu} = \sqrt{\frac{\rho^{\mu}}{\rho^{\prime} e^{\prime}}} \sin\theta^{\mu} \qquad (8.183)$

does not give a real value for the refraction angle θ^{i} . When the angle of incidence exceeds the critical angle θ_{ij} defined by

$$\sin\theta^{0} = \sin\theta_{0} = \sqrt{\frac{t^{2}\mu}{t^{2}\theta^{2}}} = \frac{n_{\mu}^{2}}{n_{\mu}} = \sqrt{\frac{\mu^{2}t^{2}}{\mu}} = \frac{n^{2}}{n}.$$
 (S.184)

346

all the nuclear wave is reflected back into the incidence medium (Born and Wolf, 1964, p = 47). Note from equations (S473) and (S453) that $\tan \theta_R = \sin \theta_0$ when $\eta' = p$ and p' = p.

For critical angle is defined as the angle of incidence bound which the refracted Uniov-Poynting vector is parallel to the interface. The condition $\operatorname{Ret} Z^1$ (=0 (see Section 6.1.5) yields the critical angle θ_{12} (for the anisotropic, lossless case, with $e_0 = e'_{01} = 0$, we obtain

$$\tan\theta_{t^{*}} = \frac{p^{t}\cos^{2}\phi}{\sqrt{p^{t}\cos^{2}\phi} - p^{t}\cos^{2}\phi}} = \frac{p^{t}\cos^{2}\phi}{\sqrt{r_{t^{*}}}p_{t^{*}}} = \frac{p^{t}\cos^{2}\phi}{\sqrt{r_{t^{*}}}}$$
(8.185)

where we have used the carrospinitence (\$180).

In the isotropic and lossy case we have

$$\lim_{t \to 0} \theta_{t^*} = \frac{\mu^{t_{t^*}}}{\left(\mu^* - \mu^{t_{t^*}}\right)}$$
(8.186)

The critical angle exists if e' is proportional to e, i.e., when the conductivity of the refraction medium satisfies $\theta' = 1e'/(10)$

Example. The acoustic properties of the incidence and refraction media are

$$v_{01} = 9.68$$
 GPa, $v_{02} = 12.5$ GPa, $\eta_{13} = 20 |v_{131}|^2$, $\eta_{13} = \eta_{13}$, $\rho = 2000$ kg m⁴

and

$$\vec{r}_{11}^{\prime} = 25.6~{
m GPa}^{\prime} / \vec{r}_{10}^{\prime} - \vec{r}_{10}^{\prime} - \vec{\eta}_{10}^{\prime} - \vec{\chi}_{10}^{\prime} - \chi^{\prime}$$
 (2.500 kg/m³

respectively, where $\omega = 2\pi f_s$ with f = 25 Hz. The refraction medium is isotropic and lassless. The absolute value of the acoustic reflection and refraction coefficients – solid and dashed lines – are shown in Figure 8.4 for the hissbess (a) and hissy the cases respectively. The Browster and critical angles are $\theta_R = 42.61^{\circ}$ and $\theta_c = 47.76^{\circ}$ (see Figure 8.4a), which can be verified from equations (8.179) and (8.185), respectively.

The electromagnetic properties of the maidence and refraction media are

$$\epsilon_{11} = 3 | \hat{\epsilon}_{12} - \epsilon_{13} = 7 | \epsilon_{01} - \sigma_{11} = \sigma_{88} = 0.15 | S_1 m_e / \mu = 2 \mu_e$$

aral

وشيار والمتوقية وفاست وعنائه

respectively, where we consider a frequency of 1 GHz. The refraction medium is vacuum. We apply the analogy

$$\begin{array}{l} r_{12}^{-1} \iff r \\ r_{13}^{-1} \iff r \\ \eta_{11} \iff \sigma \\ \eta_{21} \iff \sigma \\ \eta_{22} \iff \sigma \\ \eta_{23} \iff \sigma \\ \mu \iff \mu_{23} \end{array}$$
(S.187)

and use the same computer code used to obtain the accessic reflection and refraction coefficients. The absolute value of the electromagnetic reflection and refraction coefficients

solid and dashed lines – are shown in Figure 8.5 for the lossless (a) and lossy (b) cases, respectively. The Brewster and critical angles are $b_{12} = 13.75^{\circ}$ and $\theta_{13} = 22.96^{\circ}$ which can be verticel from equations (8.181) and (8.185), respectively.



Figure N.R. Relivents and transmission coefficients evolution divided function for cluster and a conlossless case and the lossy case.



Figure 8.5). In the two and transmission continuous conditional dashed lines. Succedulation appetic resolution destruction appetic resolution destruction appetic resolution.

Reflectivity and transmissivity

Equation (8.169) is the balance of energy flux across the interface. After substitution of the fluxes (8.161):18.165), we obtain

$$\operatorname{Ret} Z' = -\operatorname{Re} (Z'') B \to \operatorname{Re} (Z') T' = 2\operatorname{Im} (Z') \operatorname{Im} (B), \qquad (8.188)$$

Let us consider the isotropic and lossy case and an incident homogeneous plane wave $Has, p_{00} = 0, p_{01} = p_{00} = p_0$ where p is complex, and equations (6.10), and (6.28) imply $Z = \sqrt{pp} \cos \theta_0$ (hen, equation (8.188) becomes

$$\operatorname{Be}(\sqrt{p\mu})\cos\theta^{i} = |R|^{i}\operatorname{Be}(\sqrt{p\mu})\cos\theta^{i} + |T|^{i}\operatorname{Be}\left[\operatorname{pv}\left(\sqrt{\rho^{i}\mu^{i}}|\chi|^{1 + \frac{i\eta\mu^{i}}{d^{i}\mu}}\sin^{i}\theta^{i}\right)\right]$$
$$-2\operatorname{fra}(R)\operatorname{Im}(\sqrt{\eta}i)\cos\theta^{i},\qquad(8.189)$$

where we have used equations (6.47) and (8.153) [8,157). For lossless media, the interference flux — the last term on the right-hand side — vanishes, because μ is real. Moreover, using Snell's law (8,183) we obtain:

$$1 = R + T$$
, (8.196)

where

$$\mathcal{R} = R^{(i)}$$
 and $\mathcal{T} = \sqrt{\frac{d\mu^{i}\cos\theta^{j}}{d\mu^{i}\cos\theta^{j}}}T^{(i)}$ (8.191)

are called the reflectivity and transmissivity, respectively. Using the analogy (8,166) and assuming $\rho = \rho'$ and $\rho' = \mu'$ we obtain

$$\mathcal{T} = \frac{u_i^{\dagger} \cos \theta^T}{u_i \cos \theta^T} \frac{T^{-2}}{T} - \frac{u_i^{\dagger} \cos \theta^T}{u_i \cos \theta^T} T^{-2}$$
(8.192)

(Born and Wolf, 1964, p. 41), where wand at an defined in equations (8.167) and (8.168), respectively

Dual fields

The reflection and reflection coefficients that we have obtained above correspond to the particle-velocity held or to be more process to the displacement field (due to the factor is on equations (8.150) and (8.1511). In order to obtain the reflection coefficients for the stress components, we should make use of the constitutive equations, which for the plane wave are

$$\sigma_{12} = X v_2 \mod \sigma_{12} = Z v_2$$
(8.193)

(see equations (6.4)), where Z and X are defined in equations (8.153) and (8.159), respectively. Let us consider the reflected wave. Combining equation (8.150) and (8.193) we obtain

$$\begin{aligned} \sigma_{ij}^{R} &= R_{ij} \exp\left(i\omega(t + s_{ij} + s_{j}^{R})\right), \\ \sigma_{ij}^{R} &= R_{ij} \exp\left(i\omega(t + s_{j} - s_{j}^{R})\right), \end{aligned}$$
(8.193)

where

$$R_{12} = -\mathbf{i}_x |\mathbf{X}^d R |$$
 and $||B|_{12} = -\mathbf{i}_x Z^d R$ (8.195)

are the stress reflection coefficients.

In isotropic and lossless media, we have

$$R_{ij} = -i\omega\sqrt{\rho_P}\sin\theta^T R$$
 and $R_{sj} = i\omega\sqrt{\rho_P}\cos\theta^T R$ (8.195)

where we have used equations $(8.153)_{11}(8.159)_{128} = -\sin\theta^{0}\sqrt{p/p}$ (see equation (6.2714), and $Z^{h} = -Z^{0}$ (see equation (6.361).

The analogies (8.31), (8.35) and [8.166] imply

$$E_{\infty} = \beta \sqrt{\frac{\mu}{r}} \sin \theta^{T} R \quad \text{and} \quad F_{1} = -i\omega \sqrt{\frac{\mu}{r}} \cos \theta^{T} R \qquad (8.197)$$

(Born and Wolf, 1954, p. 39)

Sound waves

There is a mathematical analogy between the TM equations and a modified version of the acoustic wave equation for fluids. Denoting the pressure field by p, the modified acoustic equations can be written as

$$\partial_{t}\phi + \partial_{t}f_{d} = e_{f}\partial_{t}p_{t}$$
(8.198)

$$d_1 p = \gamma c + \rho \partial c$$
 (8.199)

$$-\partial_0 \rho = \gamma r_0 + \rho \partial_0 r_0$$
 (8.203)

where wais the find compressibility, and $\gamma=0$ yields the standard accosety expansion motion. Equations (8:198)-(8:2001 correspond to a generalized density of the form

$$p(t) = \gamma I(t) + pII(t).$$
 (8.201)

where H(t) is Hraviside's function and I(t) is the integral operator. The acceleration term for, say, the ascomputent is

Equations 18,1986(8,260) are mathematically analogous to the isotropic version of the electromagnetic equations (8,23)-(8,25) for the following correspondence

TM:	4.5	Flmd	
H_{\pm}	::	<i>p</i>	
F.	< :		
E_{\pm}	<>	<i>e</i> 1	- 8 203
1	43	ρ	
-1	÷H·	·.	
11	< 5	67.	

where $M_t = 0$ has here assumed. Let us assume a lossless electromagnetic rection, and consider Suell's law (8.183) and the analogy between the SH and TM waves. That is

3.41

transform equation (8.165) to the 1 M equations by using the analogies $\mu = \infty$, ϵ and $\mu \ll \mu$. In order to apply the mathematical analogies correctly, we need to wrast the reflection coefficients as a function of the material properties and incidence angle. We obtain

$$-R = \left(\sqrt{\frac{p}{c}} \cos\theta^{t} - \sqrt{\frac{p'}{c'}} \sqrt{1 - \frac{p_{c}}{p'c'}} \sin^{t}\theta^{t} \right) \left(\sqrt{\frac{p}{c}} \cos\theta^{t} + \sqrt{\frac{p'}{c'}} \sqrt{1 - \frac{p_{c}}{p'c'}} \sin^{t}\theta^{t} \right)$$
(8.204)

If $\kappa_f = -\rho c_i^2$ where c is the sound-wave velocity, application of the analogy (8.203) to equation (8.203) implies

$$R = \frac{p^{tet}\cos\theta^{t}}{p^{tet}\cos\theta^{t}} + \frac{p^{ter}\cos\theta^{t}}{p^{tet}\cos\theta^{t}}, \qquad (8.205)$$

where we have used Snell's law for accustic media

$$\frac{\sin \theta^{T}}{c} = \frac{\sin \theta^{T}}{c}$$
(8.206)

If we assume $\rho = p'$ and use Srell's law again, we obtain

$$R = \frac{\sin(\theta^T - \theta^T)}{\sin(\theta^T + \theta^T)}$$
(8.207)

which is the reflection coefficient for light polarized perpendicular to the plane of incidence (the electric vector perpendicular to the plane of incidence), as we shall see in the next section. Note that we storted with the TM equation corresponding to the electric vector lying in the plane of incidence.

8.4.3 The analogy between TM and TE waves

The TE (transverse-electric) differential equations for an isotropic and lossless medium are

$$\partial_3 H \rightarrow \partial_1 H_1 = (\partial_i F_i),$$
 (8.208)

$$(\partial_{\beta} E_{\beta} \otimes \mu \partial_{t} H)$$
, (8.209)

$$\cdots \partial |E_i = \mu \partial_i H_{\infty} \qquad (8.210)$$

The isotropic version of (quations (8.23) (8.25) and (8.208) (8.240) are mathematically analogous for the following correspondence

From equation (8.204), and using the analogy (8.214) and Suell's law (8.183), the TE reflection coefficient is

$$B = \left(\frac{1}{\sqrt{p}} \cos \theta^{I} - \sqrt{\frac{r^{2}}{p^{2}}} \cos \theta^{I} \right) \left(\frac{1}{\sqrt{p}} \cos \theta^{I} + \sqrt{\frac{r^{2}}{p^{2}}} \cos \theta^{I} \right) = . \tag{8.2(2)}$$

Assuming p' = p and using again Sneff's law, we obtain

$$R = \frac{\sin(\theta^{t} - \theta^{t})}{\sin(\theta^{t} + \theta^{t})}, \quad (8.213)$$

This is the reflection coefficient for the electric vectors component L_{20} i.e., light polarized perpendicular to the plane of moderne. Note that R for H_2 requiring (8.1721) and R for F_2 requiring (8.233), have different functional dependences in terms of the moderne and refraction angles.

From equation (8.175) and using the analogy (8.211), the TE Brewster angle is

$$\tan \theta_R = \frac{(p'/\mu - \epsilon')}{\sqrt{r' - \epsilon \mu/p'}}$$
(S.214)

In the case of non-magnetic media, $\mu \sim \mu^* = 1$, there is no TE Brewster angle.

Green's analogies

On December 11, 1837, Green read two papers to the Cambridge PhEosophical Society. The first paper (Green, 1838) makes the analogy between sound waves and light waves polarized in the plane of meidence. To obtain his analogy, we establish the following correspondence between the accustic equations (8,1985) (8,200) and the 11, equations (8,2085) (8,210):

where we have assumed that $\beta = 0$. Using Suell's law (S 183), the TE reflection coefficient (8.212) can be rewritten as

$$R \sim \left(\sqrt{\frac{\epsilon}{\mu}} \cos \theta^{T} \sim \sqrt{\frac{\epsilon^{T}}{\mu^{T}}} \frac{1 - \frac{\mu}{\mu^{T}}}{\sin^{T}} \sin^{T} \theta^{T} \right) \left(\sqrt{\frac{\epsilon}{\mu}} \cos \theta^{T} - \sqrt{\frac{\epsilon^{T}}{\mu^{T}}} \frac{1 - \frac{\mu}{\mu^{T}}}{\mu^{T}} \sin^{T} \theta^{T} \right)^{-1}.$$
(8.216)

If we apply the analogy (8.215) (certific equation and Stoff's law (8.206), we obtain equation (8.205). Given obtained the reflection coefficient for the potential field, and assumed $w_{\ell} = w'_{\ell}$ or

$$\frac{p^{\epsilon}}{p^{\epsilon}} = \frac{c^{\epsilon}}{c^{\epsilon}}$$
(8.217)

Using this condition. Shell's law (S 206) and equation (S 205), we obtain

$$R = \frac{\sin \theta^{i} \cos \theta^{i}}{\sin \theta^{i} \cos \theta^{i}} + \frac{\sin \theta^{i} \cos \theta^{i}}{\sin \theta^{i} \cos \theta^{i}} + \frac{\tan \theta^{i} + \theta^{i}}{\sin \theta^{i} + \theta^{i}}$$
(8.218)

which is the same ratio as for high polarized in the plane of incidence. Green (1838) has the opposite convention for describing the polarization direction, i.e., his convention is to denote R as given by equation (8.218) as the fellection coefficient for light polarized perpendicular to the phase of vanishings

Conversely, he considers the reflection coefficient (S 207) to correspond to light polarized in the plane of incidence. This is a convention dictated probably by the experiments performed by Malus, Brewster (1815) and Faraday, since Green did not know that light is a phenomenon related to the electric and magnetic fields — a relation that was discovered by Maxwell nearly 30 years later (Maxwell, 1865). Note that different assumptions lead to the different electromagnetic reflection coefficients. Assuming $p = p^2$, we obtain the reflection coefficient for light polarized perpendicular to the plane of periodence (oppation (8 207)), and assuming $r_f = r_f^2$, we obtain the reflection coefficient for light polarized in the plane of incidence tequation (8.218)).

Green's second paper (Green, 1842) is an attempt to obtain the electromagnetic reffertion coefficients by using the equations of elasticity (isotropic case). Firstly, he considers the SII-wave equation (Green's equations (7) and (8)) and the boundary conditions for the case $\mu = p'$ (five equation (9)). He obtains equation (8,165), for the displacement reflection coefficient. If we use the condition (8,217) and Suell's law (8,206), we obtain precisely equation (8,207), i.e., the reflection coefficient for light polarized prependicular to the plane of in idence – in the plane of incidence arcording to Green.

Secondly, Green considers the P-SV repretation in mation in terms of the potential fields (Green's equations [14]) and (16)), and makes the following assumptions

$$\rho v_{f_{1}}^{2} = \rho v_{f_{1}}^{2} , \quad \rho v_{g}^{2} = \rho v_{g}^{2} , \quad (8.299)$$

that is, the P- and S-wave moduli are the same for both media. This condition implies

$$\frac{c_F}{c_8} = \frac{c_F}{\beta_8}, \qquad (8.220)$$

which means that both media have the same Poisson ratio. Conversely, relating (8,220) implies that the P- and S-wave velocity contrasts are similar.

$$\frac{c_R}{d_L} = \frac{c_S}{d_S}$$
, a_s (8.221)
 $\frac{c_R}{d_S} = \frac{c_S}{d_S}$

Given is aware — on the basis of experiments — that light waves with polarization perpendicular to the wave front were ant observed experimentally. He writes: But in the transmission of light through a prism, through the one which is propagated by normal eductions acro, meanwhile dualified affecting the eye, yet it would be expedie of gaung resto an ardinary where of light propagated by transferrer enterthrons... Their then, constrained to assume that $e_{\rm P}$ the est. that is, according to less own words, that is then constrained enter the relative of transmission of waves propagated by normal relations, as very quarcompared with that of andmary light. The implications of this constraint will be clear below.

The reflection coefficient obtained by Green (1832), for the shear putertial and an incident shear way: has the following expression using our rotation:

$$R^{r} = \frac{r}{r} \, , \quad r = -(rr^{2} + 1)^{r} \left(x^{2} \pm \frac{s_{ds}^{2}}{s_{ds}^{2}} \right)^{2} + (rr^{2} - 1)^{r} \frac{s_{ds}^{2}}{s_{ds}^{2}} \, . \tag{8.222}$$

(Green's equation (26)), where s_{is}^{1} and s_{is}^{2} are the vertical components of the slowness vector corresponding to the S wave. On the basis of the condition $c_{0} \leq c_{s}$, Green assumed that the vertical components of the slowness vector corresponding to the incident to flected and refracted P waves satisfy

$$is_{10}^{I} \rightarrow is_{10}^{I} - is_{10}^{I} - s_{10}$$
 (8.223)

Ense relations can be obtained from the dispersion relation $s_1^* + s_2^* = \omega/c_p^*$, of each wave assuming $c_P \rightarrow \infty$. This assumption gives an imcompressible in diam and inhomogeneous P waves confined at the interface. The complete expression for the Si infection coefficients are given, for instance, in Pulant (1979, p. 137) f. He defines $\alpha = e_{SP}c_P$ and $\alpha = e_{SP}c_P^*$. Green's solution (8.222) is obtained for $\alpha = e_{SP} = 0$.

The vertical components of the shear slowness vertor are given by

$$s_{18}^{0} = \sqrt{\frac{1}{r_{1}^{2}} + s_{18}^{0}} + s_{18}^{0} + \sqrt{\frac{1}{r_{18}^{2}} + s^{2}}$$
 (8.221)

However, equation (8.222) is not freshel's reflection coefficient. To obtain this expansion, Green assumed that $w \approx 1$; in his own words: When the reflective power in passing from the upper to the barry madrim is not very quark in (quasking los notation) does not differ much from 1. The result of applying this approximation to expansion (8.222) is

$$R = \left(w^{3} - \frac{s_{48}^{3}}{s_{62}^{3}}\right) \left(w^{2} + \frac{s_{48}^{3}}{s_{48}^{3}}\right)^{-1}, \qquad (8.225)$$

If θ^{i} is the nucleur mate of the shear wave and θ^{i} is the angle of the refracted shear wave, equation (8.221). Shell's for and the relation

$$\frac{s_{sy}^{0}}{s_{sy}^{0}} = \frac{\cos\theta}{(\phi)\theta^{0}}$$
(8.226)

(which can be obtained by using equation (3.224) and Shell's law), yield

$$R = \left(\frac{\sin^2\theta^T}{\sin^2\theta^T} - \frac{\cos\theta^T}{\cot\theta^T}\right) \left(\frac{\sin^2\theta^T}{\sin^2\theta^T} - \frac{\cos^2\theta^T}{\cot\theta^T}\right)^{-1} - \frac{\sin^22\theta^T}{\sin^2\theta^T + \sin^22\theta^T} - \frac{\tan^2\theta^T}{\tan^2\theta^T} - \frac{\tan^2\theta^T}{\tan^2\theta^T} - \frac{\cos^2\theta^T}{\tan^2\theta^T} - \frac{\cos^$$

which is the reflection coefficient for light polarized \mathbf{n} , the plane of incidence. Green considers that equation (8.227) is an approximation of the observed reflection coefficients. We claims, on the basis of experimental data, that the intensity of the inflicted light income because absolutely hold, but attacks a nonnegatively. Moreover, he calculates the minimum value of the reflection coefficient and obtains.

$$B_{\rm are}^2 = \frac{(w^2 - 1)^2}{4w^2(w^2 + 1)^2 + (w^2 - 1)^2},$$
(8.228)

which using the approximation $w \approx 1$ gives zero reflection coefficient. This minimum value corresponds to the Browster angle when using the Freshel coefficient (8.227). Green

⁻²Note a possible in Pilant's equation (12.21). The other coefficient of matrix Δ_0 of cubic be $2\sin\theta_{s1}\sqrt{c^2} + \sin^2\theta_{s1}$ (b) d ensured of $-2\sin\theta_{s1}\sqrt{c^2} + \sin^2\theta_{s1}/(c^2)$

assumed w = 1.5 for the air-water interface. The absolute values of the reflection coefficient R given by equations [8,222] and (8,227) are shown in Figure 8.6. The dashed burcorrespond to equation (8,222). We have assumed $c_{\chi} = 30$ cm/ms and $c_{\chi} = c_{\chi}/w$. At the Brewster angle ($\theta = atantwore$). Green obtained a minimum value $R_{tow} = 0.08138$.



Figure 8.65. Groups reference coefficient for light polarized in the plane of moderne edashed lines and more spondary frequencies allocation coefficient (so fid have).

The non-existence of the Brewster angle term reflecting medicante, can be explained by the presence of dissipation, hole conductivity effects), as can be seen in Figure 8.5. Green attributes this to the fact that the term incohom is highly refracting. Quoting him: This remains take $[R_{tot}]$ increases supply, as the rade *x* of refriction discretions, and thus the quantity of light reflected at the polarizing Brewster larger, because considcrathe for highly refracting substances, a fact which has been long known to experimental phalosophers (Green, 1842). For metance, firsh water is almost lessless and is a less refracting medicin them soft water, which has a higher conductivity.

8.4.4 Brief historical review

We have seen in the previous section that Green's theory of refraction does not provide an exact parallel with the phenomenon of light propagation. MarCullagh (Tracs - Rue, high: Acad., xii, 1848; Whittaker (1987, p. 141) presented an alternative approach to the Royal Irisih Academy in 1839. He devised an isotropic medium, whose potential current is only based or rotation of the volume elements, thus ignoring pure dilatations from the beginning. The result is a rotationally elastic other and the wave equation for show waves. The curresponding reflection and refraction coefficients contride with Fresnel's formulae

Green (1842) assumed the P-wave velocity to be entirate and distansish a zero P-wave velocity on the basis that the medique would be costable (the potential energy must be positive). Canchy (Comptes Rendus in (25 Nov (1830), p. 676, and (2 Dec. (1830), p. 726) Whittaker, (1987), p. (115) reglecting this fact, considered that P waves have zero velocity, and obtained the sine law and tangent law of Freshel. He assumed the shear meablus to be the same for both micha. Canchy's effect is known as the contractile or both without the sine law and tangent law of Freshel. He assumed the shear meablus to be the same for both micha. Canchy's effect is known as the contractile or both without the sine law and tangent law of Freshel. He assumed the shear meablus to be the same for both micha. Canchy's effect is known as the contractile or both without the the same for both michae. Canchy's effect is known as the contractile or both with a shear of the propagation direction of the compressional waves to be normal to the interface. The energy cancel away by the P waves is negligible since no work is required to generate a dilatational displacement, due to the negative value of the compressibility. If we assume the shear modulus of both micha to be the same the differences depend on density characters and the inference. Firshell's formulae. The advantage of the labile effect is that it overcomes the difficulty of requiring continuity of the normal component of the displacement at the motiface. Light waves do not satisfy this condition, but light waves physicalitational vibrations, taken together, do satisfy the condition.

8.5 3-D electromagnetic theory and the analogy

We cannot establish a complete mathematical analogy in three dimensional space, but we can extend Maxwell's equations to include magnetic and didlet tric-relaxation processes and out-of-phase electric currents using viscoelastic models. The approach, based on the introduction of memory or bidder variables, uses the analogy between the Zener and Delive models (see Section 8.3.2) and a single Kelvin-Voigt element to describe the out of phase behaviour of the electric cumbertivity tany deviation from Ohn's law). We assume that the mediane is orthorhombler i.e., that the principal systems of the three material tensors coincide and that a different relaxation function is associated with each principal component. The physics is investigated by probing the mediani with a *maferice* thronogeneoust plane wave. This analysis gives the expressions of measurable quantities, like the energy velocity and the quality factor as a function of propagation direction and frequency.

In orthorhomher media $\hat{\mu}$, \hat{i} and $\hat{\sigma}$ have considerit eigenvectors. Rotating to a coordinate system defined by those contrion eigenvectors, allows the tensors to be written as

$$\hat{\boldsymbol{\mu}} = \begin{pmatrix} \mu_1 & 0 & 0 \\ 0 & \mu_2 & 0 \\ 0 & 0 & \mu_1 \end{pmatrix} = \boldsymbol{\epsilon} = \begin{pmatrix} \epsilon_1 & 0 & 0 \\ 0 & \epsilon_2 & 0 \\ 0 & 0 & \epsilon_3 \end{pmatrix} \quad \text{and} \quad \hat{\boldsymbol{\sigma}} = \begin{pmatrix} \sigma & 0 & 0 \\ 0 & \sigma_2 & 0 \\ 0 & 0 & \sigma_4 \end{pmatrix} = -(8.2/9)$$

The following symmetries are embraced by the term arthoropy(z) (*Orthorhombic*, for which there are no two eigendirections for which all three tensors have opeal eigenvalues. Crystals of this kind are said to be optically *harmic* iii) *Transverse isotropy* for which there are two eigendirections, and only two, for which all three tensors have equal eigenvalues, e.g., if the two directions are the *i*-s and *q*-directions, then $\mu = \mu_2$, $\nu_{\pm} = i_2$ and $\sigma_1 = \sigma_1$. This electromagnetic symmetry includes that of hexagonal regional and trigonal crystals. These are said to be optically *non-outling* for which all three tensors between which the tensors have equal eigenvalues, e.g., these are said to be optically *non-outling*. For which all three tensors have eigendident symmetry includes that of hexagonal regional and trigonal crystals. These are said to be optically *non-outling*. For which all three tensors have eigenvalues, we three equal eigenvalues, it is they are all isotropic tensors. Crystals of other symmetry are derivatively isotropic.

For the sole of surplicity in the evaluation of the final opinitions, we consider a Cartesian system that exincules with the principal system of the medium. The electromagnetic repratimes (8.6) and (8.7) in Cartesian components are

$$\begin{aligned} \partial_{\theta} F_{I} &= \partial_{\theta} L_{3} = p_{1} + \partial_{\theta}^{2} H_{1} + M_{1} \\ \partial_{t} E_{1} &= \partial_{\theta} F_{1} = p_{2} + \partial_{\theta}^{2} H_{2} + M_{2} \\ \partial_{x} E_{1} &= \partial_{z} F_{2} = p_{1} + \partial_{\theta}^{2} H_{3} + M_{3} \\ \partial_{x} H_{1} &= \partial_{z} H_{2} = \sigma_{3} + \partial_{z} E_{z} + v_{z} + \partial_{\theta}^{2} E_{1} + J_{1} \\ \partial_{1} H_{1} &= \partial_{z} H_{z} = \sigma_{3} + \partial_{z} E_{z} + v_{z} + \partial_{\theta}^{2} E_{1} + J_{1} \\ \partial_{t} H_{1} &= \partial_{t} H_{z} = \sigma_{3} + \partial_{z} E_{z} + v_{z} + \partial_{\theta}^{2} E_{z} + J_{z} \\ \partial_{z} H_{z} &= \partial_{t} H_{z} = \sigma_{3} + \partial_{z} E_{z} + v_{z} + \partial_{\theta}^{2} E_{z} + J_{z} \end{aligned}$$
(8.230)

8.5.1 The form of the tensor components

The principal components of the dielectric-pertaitivity tensor can be expressed as

$$\epsilon_i(t) = \epsilon_i^n \left[1 - \frac{1}{I_n} \sum_{i=1}^{I_n} \left(1 - \frac{\lambda_0}{\tau_n} \right) \exp(-t/\tau_n) \right] H(t), \quad i = 1, \dots, 3, \quad (8.231)$$

where Q is the state deflection termittivity, λ_0 and τ_0 are relaxation times $(\lambda_0 < \tau_0)$, and L_0 is the number of Debye relaxation mechanisms. The condition $\lambda_0 < \tau_0$ makes the relaxation function (S 231) analogous to the viscor lastic energy function corresponding to Zener elements connected in series (see Social 2.1.5 and Casula and Carcione (1992)). The optical for high-frequency) diebetic permittivity.

$$e^{N} = \frac{r_{1}^{2}}{L_{0}} \sum_{n=0}^{L_{0}} \frac{\lambda_{n}}{z_{n}}$$
(8.232)

is obtained as $t \rightarrow 0$. Note that $s_i^{\infty} > t_i^{0}$

Similarly, the principal components of the magnetic-perturability tensor can be written as

$$\mu_{\theta}(t) = \mu_{\theta}^{0} \left[1 - \frac{1}{N_{\theta}} \sum_{y=1}^{N} \left(1 - \frac{\varepsilon_{yy}}{\theta_{yy}} \right) \exp\left(-\theta/\theta_{yy}\right) \right] H(t), \quad y = 1, \dots, 3, \quad (8.233)$$

where p_1^* is the static permutability, γ_n and θ_0 are relaxation times ($\gamma_0 \in \theta_0$), and N_i is the number of Debye relaxation mechanisms

On the other hand, the conductivity components are represented by a Kelvin-Voigt mechanical model (see Section 2.1.2):

$$c_i(t) = c_i^T [H(t) + \chi_i \delta(t)], \quad i = 1 \dots 3,$$
 (8.234)

where σ_i^{β} is the static conductivity, χ_i is a relaxation time and $\delta \theta_i^{\beta}$ is Dirac s function. The out-of-phase component of the conduction current is quantified by the relaxation time χ_i . This choice implies a component of the conduction current 90° out-of-phase with respect to the electric field.

8.5.2 Electromagnetic equations in differential form

Equations (8.230) could be the basis for a numerical solution algorithm. However, the anmarical evaluation of the convolution integrals is probabilize when solving the differential equations with grid methods and explicit then evolution tochniques. The conductivity terms pose no problems, since conductivity does not involve time convolution. To eitcunvent the convolutions in the dashe tro-permittivity and magnetic-permitability components, a new set of field variables as introduced, following the same approach as in Section 3.9.

The dielectric internal (hidden) variables, which are analogous to the memory variables of viscorlastic media, are defined as

$$\epsilon_{ll} = -\frac{1}{\tau_l} \phi_l * E_l, \qquad l = 1, \dots, L_l.$$
(8.235)

where i 1, 3, and

$$\phi_0(t) = \frac{H(t)}{L_c \gamma_l} \left(1 + \frac{\lambda_d}{\eta_l}\right) \exp\left(-t/\tau_d\right), \qquad l = 1, \dots, L_c.$$
(8.236)

Similarly the magnetic holders variables are

$$d_{\mu} = -\frac{1}{R_{\mu\nu}}\varphi_{\mu\nu} * H_{\nu} = I - 1, \dots, N$$
, (8.237)

where

$$\varphi_{in}(t) = \frac{H(t)}{\nabla_{t}\theta_{m}} \left(1 - \frac{\gamma_{m}}{\theta_{in}}\right) \exp(i(-t/\theta_{m}), \qquad n = 1, \dots, N.$$
(8.238)

(there is no implicit summation in equations (8.235)-(8.238))

Following the same procedure as in Section 3.9, the electromagnetic reputions in differential form become

$$\begin{split} \partial_{t}E_{2} &= \partial_{2}F_{3} = \mu^{\infty}\partial_{t}H_{1} + \mu_{1}^{0} \left[\Psi_{1}H_{1} + \sum_{i=1}^{N_{1}} \left[d_{in} \right] + M_{1} \right] \\ \partial_{i}F_{3} &= \partial_{3}F_{1} = \mu_{1}^{\infty}\partial_{t}H_{2} + \mu_{0}^{0} \left[\Psi_{2}H_{2} + \sum_{i=1}^{N_{1}} \left[d_{2n} \right] + M_{2} \right] \\ \partial_{i}F_{2} &= \partial_{i}F_{2} = \mu_{1}^{+}\partial_{i}H_{3} - \mu_{3}^{0} \left[\Psi_{3}H_{3} + \sum_{i=1}^{N_{1}} \left[d_{3n} \right] + M_{3} \right] \\ &= \partial_{i}H_{3} = \partial_{i}H_{2} - \sigma_{3}^{+}F_{1} + c_{1}^{\infty}\partial_{i}F_{3} + c_{1}^{\infty}\sum_{i=1}^{N_{1}} \left[c_{2i} + J_{1} \right] \\ &= \partial_{i}H_{1} - \partial_{i}H_{3} - \sigma_{3}^{+}F_{2} + c_{1}^{\infty}\partial_{i}F_{3} + c_{1}^{+}\sum_{i=1}^{N_{1}} \left[c_{2i} + J_{2} \right] \\ &= \partial_{i}H_{2} - \partial_{i}H_{3} - \sigma_{3}^{+}F_{2} + c_{1}^{\infty}\partial_{i}F_{3} + c_{1}^{+}\sum_{i=1}^{N_{1}} \left[c_{2i} + J_{2} \right] \\ &= \partial_{i}H_{2} - \partial_{i}H_{3} - \sigma_{3}^{+}F_{4} + c_{3}^{+}\partial_{i}F_{3} + c_{1}^{+}\sum_{i=1}^{N_{1}} \left[c_{2i} + J_{2} \right] \\ &= \partial_{i}H_{2} - \partial_{i}H_{3} - \sigma_{3}^{+}F_{4} + c_{3}^{+}\partial_{i}F_{3} + c_{1}^{+}\sum_{i=1}^{N_{1}} \left[c_{2i} + J_{2} \right] \\ &= \partial_{i}H_{2} - \partial_{i}H_{3} - \sigma_{3}^{+}F_{4} + c_{3}^{+}\partial_{i}F_{3} + c_{1}^{+}\sum_{i=1}^{N_{1}} \left[c_{2i} + J_{2} \right] \\ &= \partial_{i}H_{2} - \partial_{i}H_{3} - \sigma_{3}^{+}F_{4} + c_{3}^{+}\partial_{i}F_{3} + c_{1}^{+}\sum_{i=1}^{N_{1}} \left[c_{2i} + J_{2} \right] \\ &= \partial_{i}H_{2} - \partial_{i}H_{3} - \sigma_{3}^{+}F_{4} + c_{3}^{+}\partial_{i}F_{3} + c_{3}^{+}\partial_{i}F_{3} + c_{3}^{+}\partial_{i}F_{3} \\ &= \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} + c_{3}^{+}\partial_{i}H_{3} \\ &= \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} + c_{3}^{+}\partial_{i}H_{3} \\ &= \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} + c_{3}^{+}\partial_{i}H_{3} \\ &= \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} - \partial_{i}H_{3} + d_{i}H_{3} - \partial_{i}H_{3} \\ &= \partial_{i}H_{3} - \partial$$

where

and

$$\tau_i^* = \sigma_i^* + \epsilon_i^* \Phi_i \qquad (8.241)$$

are the effective optical dielectric-permittivity and conductivity components, respectively, with

$$\Psi_{i} = \sum_{n=0}^{N} \varepsilon_{in}(0) \quad \text{and} \quad \Psi_{i} = \sum_{n=0}^{L_{i}} \phi_{in}(0). \tag{8.242}$$

The first two terms or the right side of the last three of equations (8.230) correspond to the instantaneous response of the machine, as can be inferred from the relaxation functions (8.231) and (8.234). Note that the terms containing the conductivity relaxation time χ_i are in phase with the instantaneous polarization response. The third terms in each equation involve the relaxation processes through the hidden variables.

The set of equations is completed with the differential equations corresponding to the hidden variables. Time deferentiation of equations (8.235) and (8.237), and the use of convolution properties, yield

$$\partial_t e_{tt} = -\frac{1}{\tau_1} [e_{tt} + \phi_{u,0} 0] E_{tt} = (-1, \dots, L_n)$$
(8.213)

and

$$\partial_t d_{tr} = -\frac{1}{n_{th}} \left[d_{th} + \frac{1}{r_{th}} \left[0 \left(H_{t_1}^{\dagger} \right) - a + 1, \dots, N_{tr} \right] \right]$$
 (8.214)

Equations (8,239), (8,243) and (8,244) give the electromagnetic response of a conducting axisotropic medium with magnetic and delectric-relaxation behaviour and outinf-phase conduction currents. These equations are the basis of minimized algorithms for obtaining the anknown vector field.

$$\mathbf{v} = [H_1, H_2, H_3, E_1, F_2, E_3, \{v_n\}, \{d_n\}], \quad i = 1, \dots, 3 = l = 1, \dots, L_n = n - 1, \dots, N_n$$
(8.245)

8.6 Plane-wave theory

The plan-wave analysis gives the expressions of measurable quantities, such as the slowness vector, the energy-velocity vector and the quality lactor as a function of frequency. Assume *new-andylarm* (inhomogeneous) fratmonic plane waves with a phase factor.

$$\exp[i_{\pm}(t - s_{\pm}x)]$$
, (8.246)

where **s** is the complex slowness vector. We use the following engespendences between time and frequency domains.

$$\nabla s \rightarrow -i\omega s \times -$$
 and $-\partial_t \rightarrow i\omega$, (8.247)

Substituting the plane wave (8/246) into Maxwell's equations (8.6) and (8.7), in the absence of supress and using (8,247) gives

$$s \propto E = \mu \cdot H$$
 (8.248)

and

$$\mathbf{s} \propto \mathbf{H} = \mathbf{e} \cdot \mathbf{E}$$
 (8.349)

where

$$F[0;\mu] = \mu$$
 (8.250)

and

$$\mathcal{F}[\partial \hat{x}] \sim \frac{1}{\omega} \mathcal{F}[\partial_t \sigma^* + \hat{x} + \frac{1}{\omega} \sigma^* + \hat{x}]$$
 (8.251)

For convenience, the medium properties are denoted by the same symbols, which the time and frequency domains

Note that e can alternatively be written as

 $\epsilon = \epsilon_0 = \frac{1}{\omega} \sigma_{cc}$ (8.252)

$$\epsilon_{i} = \operatorname{Re}(\epsilon) + \frac{1}{\omega} \operatorname{Im}(\boldsymbol{\pi})$$
(8.253)

and

$$\sigma_i = \operatorname{Re}(\sigma) = \operatorname{sln}(i)$$
 (8.254)

are the real effective dielectric-permuttivity and conductivity matrices, respectively. The components of ϵ and σ from oppotnois (8.234) and (8.234) are

$$\epsilon_i = \mathcal{F}(\partial x_i) = \frac{v_i^2}{L_i} \sum_{j=1}^{d_i} \frac{1 + i\omega \lambda_i}{1 + i\omega \tau_i}$$
(8.255)

and

$$\sigma_i = \mathcal{F}(\partial_i \sigma_i) = \sigma_i^{\dagger}(1 + i \omega_{ij}),$$
 (8.256)

The dielectric-periodynyity component (8:255) can be rewritten, as equation (8:116,

$$\epsilon_{\rm c} = \epsilon_{\rm c}^{(n)} + \frac{1}{L_{\rm c}} \sum_{i=1}^{n} \frac{\epsilon_{\rm c}^{(n)} - \epsilon_{\rm c}^{(n)}}{1 + i\omega \gamma_{\rm c}}, \qquad (8.257)$$

where $c_i \Sigma = c_i^n \lambda_{ni} z_n$ is the infinite-frequency (upth-ai) dielectric parametrizity of the *l*-th relaxation mechanism. A similar expression is used in bro-electromagnetism (Perropertos, 1995).

Similarly, from equation (8:233).

$$p_{i} = \mathcal{F}_{i}(\partial_{i}p_{i}) = \frac{p_{i}^{T}}{N} \sum_{n=1}^{N-1} \frac{1 + i\omega \gamma_{n}}{1 + i\omega \theta_{m}}$$
(8.258)

Since $\lambda_0 \leq \varepsilon_0$ implies $\operatorname{Im}(q) \leq 0$ and $\operatorname{Retrat} \geq 0$, the two terms on the right side of equation (S.254) have the same sign and the wave propagation is always dissipative. The importance of the effective matrices \boldsymbol{e}_i and $\boldsymbol{\sigma}_i$ is that their components are the quantumes that are measured in experiments. The coefficients multiplying the electric field and the time derivative of the electric field in equations (S.239) correspond to the components of $\boldsymbol{\sigma}_i^{(n)}$ and $\boldsymbol{e}_i^{(n)}$, respectively.

Taking the vector product of equation (8/248) with so gives

$$\mathbf{x} = \mathbf{i}_{\mathbf{p}} + \mathbf{x} + \mathbf{E} \mathbf{i}_{\mathbf{p}} + \mathbf{x} \times \mathbf{H}_{\mathbf{p}}$$
 (8.259)

which, with equation (8.249), becomes

$$s \propto (\mu^{-1} \cdot s \propto E) + \epsilon \cdot E = 0,$$
 (8.260)

360

for three equations for the components of \mathbf{E} . Alternatively, the vector product of equation (8.249) with s and use of (8.248) yields

$$|\mathbf{s} \times [\mathbf{c}]| = (\mathbf{s} \times \mathbf{H}] + \mu (\mathbf{H} = 0).$$
 (8.261)

for three equations for the components of H.

From equation (8.200), the equivalent of the 3 × 3 Kelvin-Christoffel equations (see Sections 1.3 and 4.2), for the electric-vector components, are

$$(e_0, s_0 \hat{h}_{01})_{abc} s_b + e_a (E_1 - 0) = 1, ..., 3.$$
 (8.262)

where ϕ_{ik} are the components of the Levis-Civita tensor.

Scuilarly, the equations for the magnetic-vector components are

$$(r_{ijk}s_j(a_1))^{-1}(r_{ijk}s_k + p_{ijk})H_i = 0, \quad i = 1, ..., 3.$$
 (8.263)

Both dispersion relations (8.262) and (8.263) are identical. Uniting one relation from the other implies an interchange of ϵ_0 and $\dot{\mu}_0$ and vice versa

So fay, the dispersion relations correspond to a general triclicit medium. Consider the orthonhombic case given by equations (8.229). Then, the analogue of the Kelvia-Christollel equation for the electric vector is

$$\Gamma \cdot E = 0,$$
 (8.264)

where the Kelvar-Christoffel matrix is

$$\Gamma = \begin{pmatrix} e_1 & \begin{pmatrix} s_1^{*} & s_1^{*} \\ \mu_{N} & \mu_{N} \end{pmatrix} & & s_1 s_2 & & s_1 s_1 \\ & & \mu_{N} & & \mu_{N} \\ & & s_1 s_2 & & \\ & & \mu_{N} & & c_2 & \begin{pmatrix} s_1^{*} & s_1^{*} \\ \mu_{N} & \mu_{N} \end{pmatrix} & & \frac{\mu_{N}}{s_2 s_1} \\ & & & p_1 \\ & & s_2 s_1 & & \\ & & & s_1 s_1 & & \\ & & & \mu_{N} & & & e_3 + \begin{pmatrix} s_1^{*} & s_2^{*} \\ \mu_{N} & + & \mu_{N} \end{pmatrix} \end{pmatrix},$$
 (8.265)

After defining

$$\eta_{i} = e_{i}p_{i}, \quad \zeta_{i} = e_{i}q_{i} + e_{i}p_{j}, \quad j \neq l \neq i,$$
 (8.266)

the 3-D dispersion relation (i.e., the vanishing of the determinant of the Kelvin-Christoffel matrix), becomes,

$$(e_1s_1^2 + e_1s_2^2 + e_3s_1^2)(\mu(s_1^2 + \mu)s_2^2 + \mu)s_3^2) = (\eta(s_1s_1^2 + \eta)s_3s_2^2 + \eta(s_1s_1^2) + \eta(\eta(\eta_3) - 0, (18.267)))$$

There are only quarter and quadratic terms of the slowness components in the dispersion relation of an orthorhombic medium.

8.6.1 Slowness, phase velocity and attenuation

The showness vector \mathbf{x} can be split intrineal and imaginary vectors such that $\langle Rr(t | \mathbf{x}) | \mathbf{x} \rangle$ is the phase and $-\Box \ln(\mathbf{x}) | \mathbf{x} \rangle$ is the attenuation. Assume that propagatime and attenuation directions coim ide to produce a aniform plane wave, which is equivalent to a homogeneous plane wave in visco lasticity. The slowness vector can be expressed as

$$s = s(l_1, l_2, l_4) = ss.$$
 (8.268)

where s is the complex slowness and $\mathbf{s} = (l_1, l_2, l_3)$ inside real unit vector, with l_i the direction cosines. We obtain the real wavenumber vector and the real attemption vector as

$$\mathbf{s}_{\theta} = \operatorname{Be}(\mathbf{s}) \quad \text{and} \quad \mathbf{o} = -\Box \operatorname{In}(\mathbf{s}), \quad (8.269)$$

respectively. Substituting reparator (8/268) into the dispersion relation [8/267] yields

$$As^{4} \in Bs^{4} + \eta(\eta_{t}\eta_{t}) = 0.$$
 (8.270)

where

$$A = (e_1 b_1^2 + e_2 b_2^2 + e_3 b_4^2) (\mu_1 b_1^2 + \mu_2 b_3^2 + \mu_3 b_4^2)$$

and

$$B \sim \eta |_{\infty} F + \eta_{2\infty} J_{\eta}^{2} + \eta_{3\infty} F_{s}$$

In terms of the complex velocity $c_{ij} = 1$, s, the magnitudes of the phase velocity and attemption vectors are

$$v_{i} = \left[\operatorname{Re}\left(\frac{1}{v_{i}}\right)\right]^{-1} \quad \text{and} \quad v_{i} = -\operatorname{Iee}\left(\frac{1}{v_{i}}\right).$$
(8.271)

responsively.

Assume, for instance, propagation in the (x, y)-plane. Then, G = 0 and the dispersion relation (8.270) is factorizable, giving

$$[s^{2}(x)^{2} + (y^{2}_{y}) - (y_{y}\mu_{y})^{2} + \mu_{y}\mu_{y}^{2} + \mu_{y}\mu_{y}^{2} + (x_{y})(\mu_{y})^{2} + 0), \qquad (8.272)$$

These factors give the TM and TL modes with complex velocities

$$v_{i}(\mathrm{TM}) = \frac{1}{\sqrt{\rho_{i}}} \begin{pmatrix} l_{i} \\ l_{j} + l_{j}^{\prime} \\ l_{j} \end{pmatrix}$$
(8.273)

and

$$v_{\rm e}({\rm TE}) \leq \frac{1}{\sqrt{c_s}} \left(\frac{H}{\mu_s} + \frac{E}{\mu_s} \right).$$
 (8.274)

In the TM (TE) case the magnetic (electric) vector is perpendicular to the propagation plane. For obtaining the slowness and complex velocities for the other planes, simply make the following subordey substitutions:

from the (x, y) -plane to the (x, γ) -plane $(1, 2, 3) \rightarrow (3, 3, 2)$, from the (x, y) -plane to the (y, γ) -plane $(1, 2, 3) \rightarrow (2, 3, 1)$. (8.275)

The analysis of all three planes of symmetry gives the slowness sections represented in Figure 8.7, where the values on the axes refer to the square of the complex slowness. There exists a single ranked point given by the intersection of the TE and TM males, as can be seen in the (r_{-1}) -plane of symmetry. The heation of the remical point depends on the values of the material properties. At the orthogonal planes, the waves are termed *cohmony* renclet and *extremological* polymetry. For the latter, the magnitude of the slowness vector is a function of the propagation direction. The result of two waves propagating at different velocities is called birefringence or double refraction (e.g., Kong, 1986). This phenomenon is analogous to shear-wave splitting in elastic wave propagation, (see Section 1.1.1).



Figure 5.7: increased on of the slowness such as with the principal planes. The verte speeding waves are either transverse electric. The of transverse in epoch (TM). The values of the axes refer to the square of the complex slowness.

8.6.2 Energy velocity and quality factor

The scalar product of the complex conjugate of equation (8.249) with **E** use of the relation $2\lim_{t\to\infty} (\mathbf{E} \times \mathbf{H}^*) = (\mathbf{s} \times \mathbf{E}) \cdot \mathbf{H}^* + \mathbf{E} \cdot (\mathbf{s} \times \mathbf{H})^*$ (that can be deduced from div ($\mathbf{E} \times \mathbf{H}^*$) ($\nabla \times \mathbf{E} / \cdot \mathbf{H}^* = \mathbf{E} \cdot (\nabla \times \mathbf{H}^*)$ and equation (8.248), gives Units, Powning's theorem for plane waves

$$-2Im(\mathbf{s}) \cdot \mathbf{p} = 2i(\langle E_{\mathbf{r}} \rangle + \langle E_{\mathbf{m}} \rangle) + \langle D_{\mathbf{r}} \rangle - \langle D_{\mathbf{m}} \rangle \qquad (8.276)$$

where

$$\mathbf{p} = \frac{1}{2} \mathbf{E} \times \mathbf{H}^{*} \qquad (8.277)$$

is the complex Uniov-Poynting vector,

$$(E_{\sigma}) = \begin{cases} \operatorname{Re} \mathbf{E} \cdot (\mathbf{r} - \mathbf{E})^{\sigma} \end{cases}$$
 (8.278)

is the time-averaged electric-energy density.

$$\langle \hat{D}_{\epsilon} \rangle = \frac{\omega}{2} \ln[\mathbf{E} \cdot (\mathbf{c} \cdot \mathbf{E})^{*}]$$
(8.279)

is the time-averaged rate of dissipated electric-energy density

$$\langle F_{\mu\nu}\rangle = \frac{1}{4} \text{Ite}\left[\left(\boldsymbol{\mu} \cdot \mathbf{H}\right) \cdot \mathbf{H}^{2}\right]$$
(8.280)

is the time-averaged magnetic-energy density and

$$(D_{\mu\nu}) = -\frac{1}{2} \operatorname{Im}^{2} \boldsymbol{\mu} \cdot \mathbf{H} \cdot \mathbf{H}^{2} - (8.281)$$

is the time-averaged rate of dissipated magnetic-energy density. These expressions are generalizations to the anisotropy class of the equations given in Section 8.3.1

The energy-velocity vector \mathbf{v}_{i} is given by the energy power flow. Re(p) divided by the total stored energy density.

$$\mathbf{v}_{i} = \frac{\mathrm{Re}(\mathbf{p})}{(E_{i} + E_{i})^{2}} \qquad [8|282]$$

As in the acoustic case, the relation (4.78) holds, i.e., $\mathbf{x} \in \mathbf{v}_{i} = v_{i}$, where \mathbf{x} and v_{i} are defined in equations (8.988) and (8.971) , respectively.

The quality factor quantifies energy dissipation in matter from the electric-entrent standpoint. As stated by Harrington (1964, p = 28), the quality factor is defined as the magnitude of reactive entrent density to the magnitude of dissipative entrent density. In visco-clostodynamics, a common definition of quality factor is that it is twice the ratio between the averaged strain energy density and the dissipated energy density. The kinetic and strain energy densities are associated with the magnetics and electric energy densities. Accordingly, and using the aroustic-electromagnetic analogy the quality factor is defined here as twice the time-averaged electric-energy density divided by the timeaveraged dissipated electric-energy density, where we consider the dissipation due to the magnetic permeability, as in potoclasticity we consider the dissipation due to the kinetic energy (see Sections 7.14.3) and 7.14.3). Then,

$$Q = \frac{2\langle E_r \rangle}{\langle D_r \rangle + \langle D_m \rangle}, \qquad (8.283)$$

where

$$D_{ij} = \omega^{-1} \langle \hat{D}_i \rangle$$
 and $\langle D_{ig} \rangle = \omega^{-1} \langle \hat{D}_{ig} \rangle$ (8.284)

are the time-averaged electric and magnetic dissipated-energy densities respectively

Consider the TF mode propagating in the (in graphane. Then

$$\mathbf{E} = E_0(0, 0, 1) - \exp(-i\mathbf{s} \cdot \mathbf{x}),$$
 (8.285)

where E_0 is a complex amplitude. By equation (8.248),

$$\mathbf{H} = \boldsymbol{\mu}^{-1} \cdot \mathbf{s} \times \mathbf{E} = -E_0 \left(\frac{l_2}{\mu_1}, -\frac{l_1}{\mu_2}, 0 \right) \cdot \exp(-i\mathbf{s} \cdot \mathbf{x}), \qquad (8.286)$$

where we have assumed uniform plane waves. Substituting the electric and magnetic vectors into the energy densities (8,278)-(8/281) yields.

$$(E_t) \geq \frac{1}{2} \operatorname{Re}(\epsilon_0^* | E_t^{-1} \exp(-2\alpha \cdot \mathbf{x}))$$
(8.287)

$$\langle E_{0a} \rangle = \frac{1}{4} \operatorname{Re}\left(\epsilon_{a} \frac{\epsilon_{a}}{\epsilon_{a}^{2}}\right) \left[E_{a} \left[2\mathbf{\alpha} \cdot \mathbf{x} \right] \right],$$
 (8.258)

$$\langle D_i \rangle = -\frac{\pi^2}{2} \ln(c_3) \left[F_0^{-2} \exp(-2\alpha_i \cdot \mathbf{x}) - (8.289) \right]$$

and

$$\langle \hat{D}_{0s} \rangle = \frac{\delta}{2} \ln \left(\epsilon_x \frac{\epsilon_y}{c_s^2} \right) \left[E_x^{-1} \exp(-2\alpha \cdot \mathbf{x}) \right],$$
 (8.290)

where the complex velocity v_i is given by equation (8.274).

Summing the electric and magnetic energies gives the total stored energy

$$\langle E_r + E_m \rangle = \frac{1}{2} \operatorname{Re}\left(e_1 \frac{c_1}{c_p}\right) \left| E_n^{-2} \exp((\cdot \cdot 2\mathbf{\alpha} \cdot \mathbf{x})) - (8.291) \right|$$

where v_{μ} is the phase velocity (8.271). The TE power-flow vector is

$$\operatorname{Re}(\mathbf{p}) = \frac{1}{2} \operatorname{Re}\left[\frac{1}{r_{1}} \left(\mathbf{e} \left[\frac{t}{p_{2}} + \mathbf{e}_{2} \frac{t_{1}}{\mu}\right]\right] |E_{0}|^{2} \exp(-2\boldsymbol{\alpha} \cdot \mathbf{x}), \quad (8.292)$$

From equations (8.291) and (8.292), we obtain the energy velocity for 11, waves propagating in the (x,y)-plane as

$$\mathbf{v}_{i}(\text{TE}) = \frac{c_{i}}{\text{Rel}(z_{i}, 1)} \left[l_{i} \text{Re}\left(\frac{1}{c_{i} p_{2}}\right) \mathbf{e}_{i} + l_{i} \text{Re}\left(\frac{1}{c_{i} p_{2}}\right) \mathbf{e}_{j} \right]$$
(8.393)

Performing similar rabulations, the energy densities, power flow vector (8.277) and energy velocity for TM waves propagating in the (x, y)-plane are

$$\langle L_{\nu} \rangle = \frac{1}{4} \operatorname{Re} \left(\mu_{N} \frac{\alpha_{\nu}}{\alpha_{\nu}^{2}} \right) \left[H_{0} \right]^{2} \exp \left[-2\alpha - \mathbf{x} \right], \qquad (8.294)$$

$$\langle F_{\nu} \rangle = \frac{1}{4} \operatorname{Re}(\mu_{1}) (H_{\nu}^{-2} \exp(-2\alpha \cdot \mathbf{x}))$$
 (8.295)

$$\langle \hat{D}_{\ell} \rangle = \frac{\kappa}{2} \ln \left(\mu_{\lambda} \frac{v_{\ell}}{v_{\ell}^{2}} \right) \left| H_{\ell}^{-\ell} \exp\left(-2\alpha \cdot \mathbf{x}\right) \right|$$
(8.296)

and

$$\langle \hat{D}_{m} \rangle = -\frac{2}{2} \ln(\mu_{3}) |B_{0}|^{2} \exp(-2\alpha \cdot \mathbf{x}),$$
 (8.297)

where the complex velocity v_i is given by equation (8.273).

The TM total stored energy and power-flow vector are

$$\langle E_r + E_m \rangle \simeq \frac{1}{2} \operatorname{Re} \left(\mu_1 \frac{c_1}{c_p} \right) \left| H_0^{-2} \exp((\cdot \cdot 2\alpha \cdot \mathbf{x}) - (8.298) \right|$$

and

$$\operatorname{Re}(\mathbf{p}) = \frac{1}{2} \operatorname{Re}\left[\frac{1}{v_{0}} \left(\mathbf{e}_{1} \frac{b_{1}}{v_{0}} + \mathbf{e}_{2} \frac{b_{2}}{v_{0}}\right)\right] \left(H_{0}\right)^{2} \exp\left(-2\mathbf{\alpha} \cdot \mathbf{x}\right), \quad (8.299)$$

and the energy volucity is

$$\mathbf{v}_{i}\left(1M\right) = \frac{c_{k}}{\operatorname{Re}(k_{i}t_{i})} \left[t_{i}\operatorname{Re}\left(\frac{1}{c_{i}c_{k}}\right) \mathbf{e}_{i} + t_{k}\operatorname{Re}\left(\frac{1}{c_{i}c_{i}}\right) \mathbf{e}_{i} \right]$$

$$(8.300)$$

Calculation of the total time-averaged rate of dissipated energy for the 14 and TM waves vields.

$$\langle \hat{D}_{i} + \hat{D}_{in} \rangle (W) = \frac{1}{2} \operatorname{Re}(\epsilon_{0} \alpha_{i}) \langle F_{i} \rangle^{2} \exp(-2\alpha_{i} |\mathbf{x}|)$$
 (8.301)

and

$$\langle \hat{D}_{\mathbf{r}} + \hat{D}_{\mathbf{m}} \rangle (4|\mathbf{M}_{I}| - \frac{1}{2} \operatorname{Re}\left(\mu_{J}(\alpha_{s}) \mid H_{a}^{-1}\exp\left(-2\alpha_{s}|\mathbf{x}\right)\right)$$
 (8.402)

where we have used equations (8/2877-08/290) and (8/291)-(8/297).

Let us consider the TF made – Substitution of (8.287) and (8.301) into equations (8.283) and (8.284) gives

$$Q = \frac{\text{Reeq}}{\text{adm(ca.)}}$$
 (8.303)

If we neglect the magnetic bases, for instance, by assuming that μ is real, we obtain

$$Q = - \frac{\text{Ree}(z)}{\text{Inst}(z)} - \frac{\text{Ree}(z)}{\text{Inst}(z)} + \frac{\text{Ree}(z)}{$$

which is the viscoelastic expression (e.g. secondation (1.920).

Another definition of quality factor, which considers the total energy, is a generalization of equation (2.124).

$$Q \sim \frac{(E_c + E_m)}{(D_c + D_m)}$$
, (8.305)

In this case, the quality factor takes the simple form

$$Q = \frac{\pi}{\beta_{0}r_{f}}, \quad (8.369)$$

The form (8,306) coincides with the relation between quality factor and attemption for low-loss media (see equation (2,123)), although we did not invoke such a restriction here.

The quality factor (8.283) for TM waves is

$$Q = \frac{\omega \text{Rel}(\phi_{c}(c))}{\alpha \text{Rel}(\phi_{c}(c))}, \quad (8.307)$$

and Ω has the same form (8.406) but using the phase velocity and attentiation factor corresponding to the 1M wave.

An application of this theory to ground-penetrating-radar wave propagation is given in Concore and Schoenberg (2000)

8.7 Analytical solution for anisotropic media

We can derive a closed form frequency domain analytical solution for electromagnetic waves propagating in a 3-D iossy orthorhombic medium, for which the dielectric-permittivity tensor is proportional to the magnetic-permeability tensor. Although this solution has limited practical value, it can be used to test simulation algorithms.

Maxwell's equations (8.6) and (8.7) for a transharmonic magnetic field propagating in an inhomogeneous anisotropic medium can be written as

$$\nabla \times (e^{-1} \cdot \nabla \times \mathbf{H}) = \omega^{2} \mu \cdot \mathbf{H} = \nabla \times (e^{-1} \cdot \mathbf{J})$$
, (8.308)

366

where the dielectric-permittivity tensor \mathbf{c} is given by equation (8.251). Maxwell's equations are symmetric by intercharging \mathbf{H} and \mathbf{E} . The equivalence — or duality — is given by

$$\mathbf{H} \Rightarrow \mathbf{E} = \mathbf{J} \Leftrightarrow -\mathbf{M}, \ \epsilon \Rightarrow -c\mu, \ \mu \Rightarrow -\epsilon$$
 (8.309)

The equivalent of the vector equation (8.308) is

$$\nabla + (\mu - \nabla \times \mathbf{E}) - \omega^2 \mathbf{c} \cdot \mathbf{E} - \nabla + (\mu^{-1} \cdot \mathbf{M})$$
, (8.310)

We assume now that the modifina is homogeneous. However, note that even in this situation, the tensors \mathbf{c}^{-1} and $\mathbf{\mu}^{-1}$ do not commute with the carl interator. We further assume that the medium is orthorhombic and that its principal system coincides with the Cartesian system where the problem is suboil. In orthorhombic media, the eigenvectors of the material tensors cuim ide, allowing these tensors to have a diagonal form (see equation (S 229)). In Cartesian coordinates, the vector term $\nabla \times (\mathbf{r}^{-1} \cdot \nabla \times \mathbf{H})$ consists of three solar terms.

$$(\phi_{ij})^{(1)}(\partial_{ij}H_{ij} - \partial_{ij}^{2}H_{ij}) = (\phi_{ij})^{(1)}(\partial_{ij}^{2}H_{ij} - \partial_{ij}\partial_{ij}H_{ij}),$$
 (8.314)

$$(\epsilon_1)^{-1}(\partial_t d_t H_V - \partial_x^t H_s) = (\epsilon_x)^{-1}(\partial_t^t H_s - \partial_t d_t H_s)$$

(8.312)

and

$$|\langle \psi_{i}\rangle|^{-1} (\partial_{i}\partial_{4}H_{1} - \partial_{i}^{2}H_{4}) - \langle \psi_{i}\rangle\rangle^{-1} (\partial_{i}^{2}H_{4} - \partial_{i}\partial_{4}H_{4}),$$
 (8.313)

In the absence of magnetic-current densities, we have $N \cdot \mathbf{B} = 0$, where $\mathbf{B} = \mu \cdot \mathbf{H}$, and then

$$p_{1}\partial_{t}H_{1} + p_{2}\partial_{t}H_{2} + p_{3}\partial_{t}H_{3} = 0$$
 (8.363)

Using (8.311)-(8.314) and an introlying the three components of (8.308) by $-\epsilon_{200} = \epsilon_{100}$ and $-\epsilon_{100}$ respectively, yields

$$\frac{\partial}{\partial x}\epsilon_2\partial_x^2 H_1 + \epsilon_2\partial_x^2 H_1 + \epsilon_3\partial_x^2 H_3 + \left(\epsilon_1 - \frac{\partial}{\partial x}\epsilon_2\right)\partial_x\partial_x H_3 + \omega^2 \rho(\epsilon_2\epsilon_3 H_1) + \epsilon_3\partial_y J_2 + \epsilon_2\partial_x J_3.$$
(8.315)

$$\epsilon |\partial_t^2 H_2 + \frac{p_1}{p_1} \epsilon |\partial_1^2 H_2 + \epsilon_0 \partial_s^2 H_2 - \left(\epsilon_0 + \frac{n_0}{n_0} \epsilon_1\right) \partial_2 \partial_0 H_1 + \omega^2 \mu_2 \epsilon_0 \epsilon_0 H_2 :: \epsilon_0 \partial_1 J_1 + \epsilon_0 \partial_0 J_2,$$
(8.316)

$$\epsilon \left[\partial_t^2 H_1 + c_i \partial_x^2 H_1 + \frac{\mu_0}{\mu} c_i \partial_x^2 H_3 - \left(c_f - \frac{\mu_0}{\mu_1} c_i\right) \partial_x \partial_y H_1 + \omega^2 \mu_0 c_i c_f H_1 - \epsilon_i \partial_f J_1 - \epsilon_i \partial_f J_2,$$
(8.317)

The system of equations (8.315)-08/317) can be solved in closed form by assuming that the general dielectro-percentrizity tensor is proportional to the magnetic-permeability tensor.

$$\epsilon \propto \mu_c$$
 (8.318)

This particular class of arthorhombic media satisfies

$$\mu_{12} = \mu_{21}, \quad \mu_{13} = \mu_{12}, \quad \mu_{23} = \mu_{32}, \quad (8.319)$$

This assumption is similar to one proposed by bindell and Obsloger (1997). Using Onse relations, explations (8.315) (8.317) become three Helmhultz equations.

$$\Delta_s H_s + \omega \gamma_h H_b = \epsilon_h d_h d_h - \epsilon_h d_h d_h$$
 (8.320)

Chapter 8. The acoustic-electromagnetic analogy

$$\Delta_i H_i + \omega^i \eta H_i = c_i \partial_i J_1 - c_i \partial_i J_3$$

(8.321)

$$\Delta_i H_4 + \omega^2 \eta H_4 = c_2 \partial_x J + c_3 \partial_z J_2,$$
 (8.3.22)

where

$$\eta = \rho_{12} \sigma_{3}$$
(8.323)

and

$$\Delta_i = e_i \partial_i^2 + e_i \partial_i^2 + e_i \partial_{ij}^2$$

(8.324)

The registrons for the electric-vector comprehensions has obtained from equations (8.320) of 8.323] using the duality (8.309):

$$\Delta_{\mu}L_{\nu} + \omega^{2}\chi E_{\nu} = \rho_{\mu}\partial_{\nu}M_{\lambda} - \rho_{\mu}d_{\mu}M_{\nu}, \qquad (8.325)$$

$$\Delta_{\mu}E_{2} + \omega^{2}\chi E_{2} \geq \mu_{s}\partial_{1}M_{s} + \mu_{1}\partial_{3}M_{0}, \qquad (8.326)$$

$$\Delta_{\mu}F_{\mu} + \omega^{2}\chi F_{\mu} = \mu_{0}\partial_{\mu}M_{\nu} - p_{\mu}\partial_{\nu}M_{\nu}, \qquad (8.327)$$

where

$$\Delta_{\mu} = \mu_{i}\partial_{i}^{2} + \mu_{j}\partial_{i}^{2} + \mu_{j}\partial_{i}^{2} - (8.328)$$

and

$$\chi = \epsilon_1 \hat{\mu}_2 \mu_3$$
, (8.3.9)

Note that the relations (8.319) are not modified by chadny,

8.7.1 The solution

The following change of coordinates

$$x \to \alpha \sqrt{\alpha_0}$$
, $y \to \beta \sqrt{\alpha_0}$, $(-\gamma \sqrt{\alpha_0})$ (8.300)

transforms Δ_{γ} into a pure Laplacian differential operator. Using equation (8.430)₃, equation (8.320) becomes

$$\Delta H_0 + \omega^2 \eta H_0 = \sqrt{(g0)J_0} - \sqrt{(g0)J_0}$$

(8.331)

where

$$\Delta = \partial_0^2 + \partial_1^2 + \partial_1^2 \qquad (8.332)$$

and analogously for equations (8.321) and (8.322)

Consider equation (\$331) for the Green function

$$(\Delta + \omega' \eta)\eta = -\delta(\rho),$$
 (8.533)

whose solution is

$$g(p) = \frac{1}{4\pi p} \exp(-i_{\pm}p\sqrt{y}),$$
 (8.331)

where

$$\dot{p} = \sqrt{\alpha^2 + (6 + 1)} \tag{8.335}$$

(Pilant, 1979, p. 64). The spatial derivatives of the electric currents in (8.331) maply the differentiation of the Green function. Assume, for distance, that the electric currents J_{A} and J_{A} are delta functions: $J_{A} = \mathcal{J}_{A}(q)$ and $J_{A} = \mathcal{J}_{A}(q)$. Since the solution of (8.331) is

368

the convolution of the Green function with the source term, it can be obtained as the β spatial derivative of the Green function. Then, for impulsive electric currents, the solution is

$$H_{\alpha} \simeq -\left(\sqrt{\epsilon_{\alpha} I} |\hat{\sigma}| y + \sqrt{\epsilon_{\alpha} I} |\hat{\sigma}| y\right), \qquad (8.336)$$

We have that

$$\partial_{\alpha}q = \begin{pmatrix} f \\ \rho \end{pmatrix} \partial_{\mu}g, \quad \partial_{\beta}g = \begin{pmatrix} e \\ \rho \end{pmatrix} \partial_{\mu}g \qquad (8.337)$$

where

$$d_i q = -\left(\frac{1}{p} + \mathrm{i}_S \sqrt{y}\right) q, \qquad (8.338)$$

In terms of Carresian coordinates, the solution is

$$H_{\perp} = \frac{1}{4\pi \rho^2} \left(\left| \mathcal{J}_z - g \mathcal{J}_1 \right| \left(\frac{1}{\rho} + \mathrm{i} \omega \sqrt{\eta} \right) \exp(-\mathrm{i} \omega \rho \sqrt{\eta}), \qquad (8.339)$$

where

$$p = \sqrt{\frac{c^2}{c_1} + \frac{a^2}{c_2} + \frac{c^2}{c_1}}$$
(8.340)

Similarly, the other components are given by

$$H_{2} = \frac{1}{4\pi\rho^{2}} \left(i\sqrt{t}_{1} + i\sqrt{t}_{1} \right) \left(\frac{1}{\rho} + i\omega\sqrt{\eta} \right) \exp\left(-i\omega\rho\sqrt{\eta} \right)$$
(8.311)

and

$$H_1 = \frac{1}{1 \pi \rho^2} \left(\eta \mathcal{J}_1 - v \mathcal{J}_1 \right) \left(\frac{1}{\rho} + (\omega \sqrt{\eta}) \exp(-(\omega \rho \sqrt{\eta})) \right)$$
(8.342)

The three components of the magnetic vector are not functionally independent, since they must satisfy explation (8.314). When solving the problem with a limited-band wavilet source f(t), the frequency-domain solution is multiplied by the Fourier transform $F(\pm)$. To ensure a real transformain solution, we consider an Hermitian frequency-domain solution. Evally, the transformain solution is obtained by on inverse transform. Examples illustrating this analytical solution can be found in Carrience and Cavallicit (2001).

8.8 Finely layered media

The electromagnetic properties of healy plane-hypered media can be obtained by using the same approach used in Section 1.5 for elastic media. Let us consider a plane-hypered medium, where each layer is humagemons, anisotropic and thin compared to the wave length of the electromagnetic wave. If the layer interfaces are parallel to the (x, y)-plane, the properties are independent of x and y and may vary with z.

We follow Backus's approach (Backus, 1962) to obtain the properties of a finely layer of medium. Let $n(t_i)$ be a continuous weighting function that overages over a length d. This function has the following properties:

$$\begin{split} w(z) &= 0 \\ w(z) &= 0 \\ \frac{1}{2} \int_{-\infty}^{\infty} w(z') d(z'-1) \\ \int_{-\infty}^{\infty} w(z') d(z'-0) \\ \int_{-\infty}^{\infty} w(z') d(z'-0) \\ \int_{-\infty}^{\infty} w' w(z') d(z'-d^2). \end{split}$$
(8.343)

Then, the average of a function if over the length of around the location it is

$$\langle f_{1}^{*}|\rangle = \int_{-\infty}^{\infty} u (s^{2} + \beta) f^{*,2} (d\beta^{2}, \beta)$$
 (8.344)

The averaging removes the wavelengths of f which are smaller than ϑ . An important approximation in this context is

$$(f_{fb}) = f(g),$$
 (8.345)

where f is nearly constant over the distance ϑ and g may have an arbitrary dependence as a function of z_i

Let us consider first the dielectric-permittivity properties. The explicit form of the frequency-domain constitutive equation is obtained from equation (8.50).

$$\begin{pmatrix} D \\ D_2 \\ D_3 \end{pmatrix} = \begin{pmatrix} \epsilon_1 & \epsilon_2 & \epsilon_3 \\ \epsilon_2 & \epsilon_2 & \epsilon_3 \\ \epsilon_4 & \epsilon_4 & \epsilon_4 \end{pmatrix} \cdot \begin{pmatrix} E_1 \\ E_2 \\ E_4 \end{pmatrix},$$
(8.346)

where the dielectric-permittivity components are complex and frequency dependent. The boundary conditions at the single interfaces impose the continuity of the following field components.

$$D_{50}(E_{10})$$
 and E_{10} (8.347)

(Born and Wolf, 1964, p. 4), which vary very slowly with z. On the contrary, D, D_s and L_s vary rapidly from layer to layer. We express the rapidly varying fields in terms of the slowly varying fields. This gives

$$D_{i} = \left(e_{i} + \frac{e_{i}^{2}}{e_{i}s}\right) E_{i} + \left(e_{i}e_{i} + \frac{e_{i}^{2}e_{i}}{e_{i}}\right) E_{i} + \frac{e_{i}^{2}}{e_{i}} D_{i}, \qquad (8.3)8)$$

$$D_{2} = \left(e_{12} - \frac{e_{12}e_{13}}{e_{13}}\right) E_{1} = \left(e_{12} - \frac{e_{13}^{2}}{e_{13}}\right) E_{1} + \frac{e_{23}}{e_{13}} D_{1} = (8.349)$$

and

$$F_{t} = \frac{1}{\epsilon_{33}} \left(D_3 - \epsilon_{13} F_4 - \epsilon_{23} F_2 \right)$$
(8.356)

These equations contain no products of a rapidly varying field and a rapidly variable chelectric-permittivity component. Lien, the average of equations 18.3486(8.350) over the length d can be performed by using equation (8.345). We obtain

$$\langle D_{1}\rangle = \left\langle c_{11} - \frac{c_{11}'}{c_{12}} \right\rangle F_{1} + \left\langle c_{22} - \frac{c_{13}C_{15}}{c_{15}} \right\rangle F_{2} - \left\langle \frac{c_{15}}{c_{15}} \right\rangle D_{5} = -\langle 8.351 \rangle$$

$$\langle D_i \rangle = \left\langle \epsilon_{ij} - \frac{\epsilon_{ij} \delta_{ij}}{\epsilon_{ij}} \right\rangle F_i + \left\langle \epsilon_{ij} - \frac{\delta_{ij}}{\epsilon_{ij}} \right\rangle F_j + \left\langle \frac{\delta_{ij}}{\epsilon_{ij}} \right\rangle D_j = (8.352)$$

and

$$\langle E_0 \rangle = \left\langle \frac{1}{\epsilon_0} \right\rangle H_0 - \left\langle \frac{\epsilon_0}{\epsilon_0} \right\rangle L_0 - \left\langle \frac{\epsilon_0}{\epsilon_0} \right\rangle E_0$$
 (8.353)

370

Expressing the average destric-displacement components in terms of the averaged electricvector isourpoments gives the constitutive equations of the medium,

$$\begin{pmatrix} \langle D_{\alpha} \rangle \\ \langle D_{\beta} \rangle \\ \langle D_{\beta} \rangle \\ \langle D_{\beta} \rangle \end{pmatrix} = \begin{pmatrix} \gamma_{1} & \gamma_{2} & \alpha_{1} \\ \gamma_{2} & \gamma_{3} & \alpha_{1} \end{pmatrix} \cdot \begin{pmatrix} E \\ E_{\beta} \\ \langle E_{\beta} \rangle \\ \langle E_{\beta} \rangle \end{pmatrix},$$
(8.354)

where

$$a_{\rm el} = \left\langle \epsilon_{\rm el} - \frac{d_{\rm s}}{\epsilon_{\rm el}} \right\rangle + \left\langle \frac{\epsilon_{\rm s}}{\epsilon_{\rm el}} \right\rangle^2 \left\langle \frac{1}{\epsilon_{\rm s}} \right\rangle^2 , \qquad (8.355)$$

$$_{0} = \left\langle e_{1}, -\frac{e_{13}e_{23}}{e_{13}} \right\rangle + \left\langle \frac{e_{43}}{e_{13}} \right\rangle \left\langle \frac{e_{43}}{e_{33}} \right\rangle \left\langle \frac{1}{e_{33}} \right\rangle^{-1}, \qquad (8.356)$$

$$_{13} = \left\langle \frac{\epsilon_{13}}{\epsilon_{13}} \right\rangle \left\langle \frac{1}{\epsilon_{33}} \right\rangle \quad . \tag{8.357}$$

$$_{E} \geq \left\langle \epsilon_{E} - \frac{\ell_{D}^{2}}{\epsilon_{D}} \right\rangle + \left\langle \frac{\ell_{D}}{\epsilon_{D}} \right\rangle^{2} \left\langle \frac{1}{\epsilon_{D}} \right\rangle^{2} \left\langle \frac{1}{\epsilon_{D}} \right\rangle^{2} \left\langle \frac{1}{\epsilon_{D}} \right\rangle^{2} , \tag{8.3.8}$$

$$a = \left\langle \frac{e_{ee}}{e_{ee}} \right\rangle \left\langle \frac{1}{e_{ee}} \right\rangle$$
(8.359)

and

$$\gamma_{11} \approx \left\langle \frac{4}{\gamma_{13}} \right\rangle^{-1}, \qquad (8.360)$$

For isotropic layers, $e_{12} = e_{13} = e_{13} = 0$, $e_1 = e_{22} = e_{33} = e$, and we have

$$_{1} = |_{M_{1}} = \{0\},$$
 (8.361)

$$\omega = \left\langle \frac{1}{\epsilon} \right\rangle \tag{8.362}$$

and the second of the

The acoustic-electromagnetic analogy between the 1M and SH cases is $r \ll \mu^{-1}$ (see equation [8–341], where μ is the shear modulus. Using the precedum equations, we obtain the following stiffness constants

$$e_{11} = \left\langle \frac{1}{\rho} \right\rangle \tag{8.363}$$

and

$$\sigma_{ijk} = (\mu),$$
 (8.304)

respectively. These equations are equivalent to equations (1.188), and (1.188), for isotropic layers, respectively.

The same functional form is obtained for the magnetic-permeability and conductivity tensors of a budy layered medium if we apply the same procedure to explane (8.61) and (8.62), with J = 0. In this case, containity of $B_{d,1}, H_1, H_2$ and J_d, E_1, E_2 is required, respectively.

8.9 The time-average and CRIM equations

The acoustic and electromagnetic wave velocities of rocks depends strongly on the tock composition. Assume a stratilied model of n different media, each having a thickness h_i and a wave velocity r_n . Lie transit time t for a wave through the rock is the sum of the partial transit times:

$$t = \frac{b}{c} = \sum_{i=1}^{n} \frac{b_i}{c_i},$$
(8.465)

where $b \geq \sum_{i=1}^{n} b_i$ and v is the average velocity. Defitting the material proportions as $v_b = b_b/b_i$ the average velocity is

$$c = \left(\sum_{i=1}^{n} \frac{\alpha_i}{\alpha_i}\right)^{-1} \quad [8.496]$$

For a rock saturated with a single fluid, we obtain the time-average equation:

$$v = \left(\frac{\alpha}{v_f} + \frac{1 - \alpha}{v_s}\right)^{-1}, \qquad (8.467)$$

where α is the porosity (, β) is the fluid wave velocity and α is the wave velocity in the numeral oggregate (Wyllie) Gregory and Gardner (1956)

The electromagnetic version of the time average equation is the CRIM equations (complex reflaction index model). If $\mu_i(\pm)$ and $i(\pm)$ are the magnetic permutability and dividentic permutativity of the single phases, the respective slownesses are given by $1/v_i = \sqrt{n_{el}}$. Using equation 18.366), the equivalent electromagnetic equation is

$$\sqrt{p_{\ell}} = \sum_{i=1}^{n} \phi_i \sqrt{p_i} e_{i\ell} \qquad (8.468)$$

where a and state the average permetablicy and permittivity, respectively. The URDM equation is obtained for creation magnetic permeability. That is

$$-\left(\sum_{i=1}^{n} \alpha_i \sqrt{\alpha_i}\right)^2 = \alpha_i - \frac{1}{\omega}\sigma_i, \qquad (8.369)$$

where $r_{\rm e}$ and $\sigma_{\rm e}$ are the real-valued effective perioritivity and conductivity, respectively (see equation [8.252)). A useful generalization is the high use key-Rother formula:

$$\epsilon = \left(\sum_{i=1}^{n} \phi_i(\epsilon_i)^{(i)}\right)^2, \qquad (8.370)$$

where t is a fitting parameter (e.g., Gnégurn and Pahianskas, 1994).

ı

While Backus averaging yields the low frequency clasticity constants, the time average and CRIM equations are a ligh-frequency approximation, i.e., the limit known as geometerod optics.

¹Note that with the linear parameter which compute the volume portoint, m_{1} for planar pares of cracks. For three interacting, notically perpendicular, planar cracks, the relation is $m_{1} = 1 - 1 + 51^{2}$, with $m_{1} \approx 30$ for $m \approx 1$.

8.10 The Kramers-Kronig dispersion relations

The Kramers-Kronig dispersion relations obtained in Section 2.2.4 for and astic metric were first derived as a relation between the real and imaginary parts of the frequency dependent dielectric permittivity function (Kramers, 1927; Kronig, 1926). Artially, the relations are applied to the electric susceptibility of the material.

$$\chi(z) = c[z] + c_0 = c(z) + ic_2(z) + c_0.$$
 (8.371)

where ϵ_{i} and ϵ_{j} are the real and imaginary parts of the chelsettic permittivity and here, ϵ_{i} is the differt in permittivity of free space. Fuder rectain conditions, the linear response of a medium car be expressed by the electric polarization vector $\mathbf{P}(t)$.

$$\mathbf{P}(t) = \chi + \partial_t \mathbf{E} = \int_{-\infty}^{\infty} \chi(t - t') \partial \mathbf{E}(t') dt'$$
(8.372)

(Born and Wulf, 1964, p. 56 and 54), where of denotes the derivative with respect to the argument. A Kentier transform to the frequency domain gives

$$\mathbf{P}(\varphi) \simeq \chi(\varphi) \mathbf{E}(\varphi)$$
(8.3)(3)

where $\chi(\omega)$ stands for $\mathcal{F}(\partial_t \chi(t))$ to simplify the notation. The electric displacement vector is

$$\mathbf{D}(\omega) \simeq c_0 \mathbf{E}(\omega) + \mathbf{P}(\omega) \simeq c_0 + \chi(\omega) [\mathbf{E}(\omega) \approx \delta(\omega) \mathbf{E}(\omega)],$$
 (8.374)

according to equation (8.371). The electric susceptibility $\chi(\omega)$ is analytic and bounded in the lower half-plane of the complex frequency argument. This is a consequence of the causality condition, i.e., $\chi(t - t') = 0$ for t < t' (Golden and Graham, 1988, p. 18).

An alternative derivation of the Kramers Krunig relations is based on Cauchy's integral formula applied to the electric susceptibility. Since this is analytic in the lower half plane, we have

$$\chi(\omega) = \frac{1}{4\pi} [m \int_{-\infty}^{\infty} \frac{\chi(\omega')}{\omega - \omega} d\omega', \qquad (8.375)$$

where pv is the principal value. Separating real and imaginary parts and using equation (8.371), we obtain the Krowners-Mrong relations.

$$\psi_{0}(\omega) = e_{0} + \frac{1}{\omega} p_{0} \int_{-\infty}^{\infty} \frac{r_{0}(\omega')}{\omega - \omega'} d\omega'$$
 (8.376)

and

$$\phi_{\ell}(\omega) = -\frac{1}{\pi} \operatorname{pc} \int_{-\infty}^{\infty} \frac{\psi_{\ell}(\omega') - \psi_{\ell}}{-\omega \omega'} d\omega', \qquad (8.377)$$

The acoustic-dectromagnetic analogy (8.33) implies the mathematical equivalence between the dielectric permittivity and the complex creep compliance defined in equation (2.43), i. $n, r \Leftrightarrow J \neq J \rightarrow iJ_{tr}$. Hence, we obtain

$$J_{i}(z_{i}) = \frac{1}{\pi} g_{i} \int_{-\infty}^{\infty} \frac{I_{i}(\omega')}{\omega' - \omega'} d\omega'$$
(8.378)
and

$$J_{2}(\omega) = -\frac{1}{\pi} \left[w \int_{-\infty}^{\infty} \frac{J_{1}(\omega^{2})}{\omega^{2}} d\omega^{2} \right]$$
(8.379)

which are confidematically equivalent to the Kramers-Kronig relations (2.70) and (2.72) corresponding to the viscoalistic romplex modulus. The term equivalent to r_{i} , is zero in the constituence, since there is not an upper limit velocity equivalent to the velocity of light (M and J can be infinite and zero, respectively). Atalogeus Kramers Kronig relations apply to the complex magnetic-permeability function.

8.11 The reciprocity principle

The reciprocity principle for acoustic waves is illustrated in detail in Chapter 5. In this section, we obtain the principle for electromognetic waves in acusotropic lossy media.

We suppose that the source currents \mathbf{J}_1 and \mathbf{J}_2 give ruse to fields \mathbf{H}_1 and \mathbf{R}_2 , respectively. These fields satisfy equation (8.308):

$$\nabla \times (\mathbf{c}^{-1} \cdot \nabla \times \mathbf{H}_{1}) = \omega^{2} \hat{\boldsymbol{\mu}} \cdot \mathbf{H}_{1} - \nabla \times (\mathbf{c}^{-1} \cdot \mathbf{J}_{1})$$

(8.380)

and

$$\nabla \times \left(\mathbf{c} \to \nabla \times \mathbf{H}_{2} \right) = \mathbb{A} \left[\mathbf{\mu} \cdot \mathbf{H}_{2} - \nabla + \left(\mathbf{c}^{-1} \cdot \mathbf{J}_{2} \right) \right]$$
(8.381)

The following scalar products are valid.

$$\mathbf{H}_{2} \cdot \nabla \times \left(\boldsymbol{\ell} \to \nabla \times \mathbf{H}_{1} \right) \cdots \mathbb{I} \left[\mathbf{H}_{2} \cdot \boldsymbol{\mu} \cdot \mathbf{H}_{1} = \mathbf{H}_{2} \cdot \nabla \times \left(\boldsymbol{\ell} \to \mathbf{J}_{1} \right) \right]$$
(8.382)

and

$$\mathbf{H} + \nabla \otimes \left(\mathbf{c} \to \nabla + \mathbf{H}_2 \right) + \omega^2 \mathbf{H} \oplus \mathbf{\mu} \oplus \mathbf{H}_2 = \mathbf{H} \oplus \nabla \times \left(\mathbf{c}^{-3} \oplus \mathbf{J}_2 \right), \quad (8.383)$$

The second terms on the left-hand-side of equations (8.382) and (8.383) are equal if the magnetic-permeability tensor is symmetric, i.e., if $\boldsymbol{\mu} = |\boldsymbol{\mu}| = \text{The first terms can be$ $rewritten using the vector identity <math>\mathbf{B} \cdot \nabla \times \mathbf{A} = \nabla \cdot (\mathbf{A} \times \mathbf{B}) + \mathbf{A} \cdot (\nabla \times \mathbf{B})$. For instance, $\mathbf{H} \cdot \nabla \times (\mathbf{e}^{-1} \cdot \nabla + \mathbf{H}_{i}) = \nabla \cdot (\mathbf{e}^{-1} \cdot \nabla \times \mathbf{H}_{i}) \times (\mathbf{E}^{-1} \cdot \nabla \times \mathbf{H}_{i}) \cdot (\nabla \times \mathbf{H})$. Integrating this quantity over a volume Ω bounded by surface S_{i} and using Gauss's theorem, we obtain

$$\int_{S} [(\mathbf{c} - (\nabla \times \mathbf{H}_{1}) \times \mathbf{H}_{1}] \cdot \mathbf{n} \, dS + \int_{\Omega} (\mathbf{c}^{-1} \cdot \nabla \times \mathbf{H}_{2}) \cdot (\nabla \times \mathbf{H}_{2}) \, d\Omega \qquad (8.38))$$

where **n** is a unit vector directed along the outward normal to S. The second term on the right-hand side of equation (8.384) is symmetric by interchanging **H**, and **H**₂ if $\mathbf{c} = \mathbf{c}$. Regarding the first term, we assume that the incluming isotropic and homogeneous when $S \to \infty$, with a dielectric permittivity could to c. Furthermore, the wave helds are plane waves in the finite field so that $\nabla \to -\mathbf{k}$ where **k** is the complex wavevector. Moreover, the plane wave assumption implies $\mathbf{k} \times \mathbf{H} = 0$. Hence

$$(\mathbf{r} = (\nabla \times \mathbf{H}_{1}) \times \mathbf{H}_{2} = i\mathbf{k} (\mathbf{r}) - (\mathbf{H}_{2} \cdot \mathbf{H}_{1})$$
 (8.385)

(Chew, 1990). Thus, also the first term on the right-hand side of conation (8.385) is symmetric by interchanging \mathbf{H}_2 and \mathbf{H}_3 .

374

Consequently, a volume integration and subtraction of equations (8.382) and (8.383) yields

$$\int_{\Omega_{1}} \left[\mathbf{H}_{\mathbf{z}} \cdot \nabla \times \left(\mathbf{e}^{-1} \cdot \mathbf{J}_{1} \right) - \mathbf{H}_{1} \cdot \nabla \times \left[\mathbf{e}^{-1} \cdot \mathbf{J}_{1} \right]_{1} d\Omega = 0, \quad (8.386)$$

Using the vector identity indicated above, with $\mathbf{A} = \mathbf{c} - \langle \mathbf{J} |$ and $\mathbf{B} = \mathbf{H}$, and using Maxwell's equation $\nabla [z, \mathbf{H}] = z_0 \mathbf{c} + \mathbf{E}$, we obtain $\mathbf{H} + \nabla [z] (\mathbf{c} \to \mathbf{J}) = z_0 \mathbf{E} + \mathbf{J}$. Hence equation (8.386) becomes

$$\int_{\Omega} \left(\mathbf{E}_{2} \cdot \mathbf{J}_{3} + \mathbf{E}_{3} \cdot \mathbf{J}_{3} \right) d\Omega \geq 0.$$
(8.387)

This equation is equivalent to the aroustic version of the reciprocity (equation 5.3)). It states that the field generated by \mathbf{J}_{+} invasticed by \mathbf{J}_{+} is the same held generated by \mathbf{J}_{+} measured by \mathbf{J}_{+} . Note that the principle holds if the magnetic permeability and dielectric permeability are symmetric tensors.

8.12 Babinet's principle

Balmotis principle was originally used to relate the diffracted light fields he complementary thin screens (hours, 1986). In obscionagnetism, Babinet's principle for infinitely thin perfectly conducting complementary screens implies that the sam, beyond the screen plane, of the electric and the magnetic fields (adjusting physical dimensions) equals the incident tuns recard) electric field. A complementary screen is a plane screen with opagite areas where the original plane screen had transparent areas. Roughly speaking, the prinriple states that behind the diffracting plane, the sum of the fields associated with a screen and with its complementary screen is just the field that would exist in the absence of our screen that is, the diffracted fields from the two complementary screens are the treative of each other and cancel when summed. The principle is also valid for electromagnetic fields and perfectly conducting plane screens or diffractory thores (1986).

Consider a screen S and its complementary screen C and assume that the total held in the presence of S is v_s and that related to C is \mathbf{v}_i . Balanct's principle states that the total fields on the opposite solve of the screens from the source satisfy

$$v_{s} + v_{t'} - v_{ts}$$
 (8.388)

where \mathbf{v}_{i} is the field in the absence of any screen. Equation (8.388) states that the diffraction fields for the complementary screens will be the negative of each other. Moreover the total fields on the source side must satisfy

$$v_{S} + v_{0} = 2v_{0} + v_{R}$$
 (8.389)

where \mathbf{v}_{H} is the reflected field by a screen composed of S and C.

Carcione and Gaugi (1998) have investigated Rabinot's principle for acoustic waves by using a numerical simulation reducipte. In classodynamics, the principle holds for the same field (particle velocity or stress), but for complementary screens satisfying different types of boundary conditions, i.e. if the original screen is work (stress-free condition), the complementary screen must be ugid. On the other hand, if this original screen is rigid, the complementary screen must be weak Balmet's principle holds for screens embedded in anisotropic media, both for SII and qP-qS waves. The simulations indicate that Babinet's principle is satisfied also in the case of shear wave triplications (qS waves). Almenter, the numerical experiments show that Babinet's principle holds for the near and far fields, and for an arbitrary pulse waveform and frequency spectrum. However, as expected, lateral and interface waves (e.g., Bayleigh waves) do not satisfy the principle.

Balmet's principle is of value since it allow us to obtain the solution of the complementary problem from the solution of the original problem without any additional effort. Minowire, it provides a clock of the solutions for problems that the self-complementary (e.g., the problem of a plane wave minimally incident on a half plane). Finally, it adds to our knowledge of the complex phenomena of elastic wave diffraction.

8.13 Alford rotation

The atalogy between acoustic and electromagnetic waves also applies to multi-component data acquisition of seismic and ground-penetrating-radar (GPR) surveys. Aford (1986) developed a method, subsequently referred as to "Alford rotation", to determine the rears axis of sub-orface sessiic anisotropy. Alford considered four seismic sections acquired by using two intrinuital (orthogonal) sources and two orthogonal horizontal receivers. If we denote someward receiver by S and R and in line and cross-line by I and C. respectively, the four scismic sections can be denoted by, S_IR_I , S_IR_I , S_CR_I and S_CR_C . where "line" refers to the orientation of the seismic section. Allord observed that the seismic events in the cross-component sections (S_1R_1) and (S_2R_1) were better than those of the principal components sections (S_tR_t) and S_tR_t . The reason for this behavior is shear-way splitting which occurs in azricutbally anisotropic media (see Section 1.4.4) for instance, a transversely isotropic medium whose axis of symmetry is horizontal and makes an angle $\pi/2 - \theta$ with the direction of the scistuic line. If $\theta = 0$, the scistuic energy in the cross-component sections should be minimum. Thus, Alford's method consist in a rotation of the data to minimize the energy in the cross-component surveys, obtaining nothes way the orientation of the symmetry axis of the medvin. An application is to ful the orientation of a set of vertical fractures, whose planes are perpendicular to the symmetry axis. In addition, Alford intation allows us to obtain the reflection amplitudes for every angle of orientations of transmitter and involved without having to collect data for all configurations.

The equivalent acquisition configurations in GPR surveys are shown in Figure 8.8, where the *i.u.*, *i.g.*, *qx* and *qq*-configurations correspond to the seismic surveys S_iR_i , S_iR_i , S_iR_i , S_iR_i , and S_iR_i , respectively. In theory, the *i.u.* and *qx*-configurations should give the same result because of recursority.

We consider the i-D equations along the vertical – direction and a hissless transversely isotropic medium whose axis of symmetry is parallel to the surface (t, y) plane). In this case, the slower and faster sloar waves S1 and S2 waves, whose velocities are $\sqrt{e_{N}/p}$ (S1) and $\sqrt{e_{N}/p}$ (S2), are analogous to the TM and TE waves, whose velocities are $1/\sqrt{p_{P}}$ (TM) and $1/\sqrt{p}$ (e) (TE) (see Section 1.3.1 for the acoustic case and the (x, z)-plane of Figure 8.5 for the lossless electromagnetic case. A tigorous science theory filteration, the physics involved or V5ad isotropic signer (398)

When the source radiation directivities (seismin shear vibrators or dipole automas)



Figure 8.8: Different GPR transmitter received antenna configurations, where S is transmitted and Ris receiver. The survey largest oriented along the x show that

are aligned with the principal axes of the medium, the propagation equations can be written as

$$\begin{pmatrix} L_{ij} & 0 \\ 0 & L_{ji} \end{pmatrix} = \begin{pmatrix} a & 0 \\ 0 & y_{ii} \end{pmatrix} = \begin{pmatrix} \delta_{ij}(f(t) & 0 \\ 0 & -\delta_{ij}(f(t)) \end{pmatrix} \cdot \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}, \qquad (8.390)$$

where the L_0 are differential propagation operators, the u_0 are the recorded wave fields, and f(t) is the source time lastory. The source term in the right-hand side of equation (S 350), defines a set of two orthogonal sources aligned along the principal coordinates axes of the medium, such that the solutions u_0 and u_D correspond to the seismic sochars S_1R_1 and S_2R_2 or to the GPB configurations v_0 and y_0 respectively.

Equation (8.390, car be expressed in matrix form as

$$L + U = S + I_f = S.$$
 (8.391)

The rotation matrix is given by

$$\mathbf{R} \simeq \begin{pmatrix} -\cos\theta & \sin\theta \\ -\sin\theta & \cos\theta \end{pmatrix}, \quad \mathbf{R} \cdot \mathbf{R} \simeq \mathbf{I}_{2}, \quad (8.392)$$

High transitionary equation (8.391) by ${\bf R}$ rotates the same estromater checkwise through an angle θ_1

$$L \in U \setminus R[\rightarrow S \setminus R]$$
 (8.393)

The term is square brackets is the solution for the new sources in the principal coordinates of the medium. The following operation corresponds to a counter-clockwise rotation of the receivers through an angle θ .

$$L + R + R = (U + R) = S + R.$$
 (8.393)

where we have used equation (8,492)

Denoting by primed quantities the matrices in the acquisition coordinate system, equation (8/391) reads

$$L^{0}(U) = S^{0}$$
 (8.375)

where $\mathbf{L}' = \mathbf{L} \cdot \mathbf{R}$, $\mathbf{S}' = \mathbf{S} \cdot \mathbf{R}$, and

$$U' = \mathbf{R} \rightarrow U + \mathbf{R}$$
, (8.396)

This equating allows the computation of the solutions in the principal system in terms of the solutions in the acquisition system.

$$\mathbf{U} = \begin{pmatrix} u_{11} & u_{12} \\ u_{21} & u_{22} \end{pmatrix} \in \mathbf{R} \cdot \mathbf{U}^{t} \cdot \mathbf{R} \quad . \tag{8.397}$$

where

$$\begin{aligned} u_{11} &= u_{11}^{*} \cos^{2}\theta + u_{21}^{*} \sin^{2}\theta + 0.5(u_{21}^{*} + u_{12}^{*}) \sin 2\theta \\ u_{22} &= u_{12}^{*} \cos^{2}\theta + u_{12}^{*} \sin^{2}\theta + 0.5(u_{21}^{*} + u_{21}^{*}) \sin 2\theta \\ u_{22} &= u_{21}^{*} \cos^{2}\theta + u_{11}^{*} \sin^{2}\theta + 0.5(u_{21}^{*} + u_{21}^{*}) \sin 2\theta \\ u_{22} &= u_{22}^{*} \cos^{2}\theta + u_{12}^{*} \sin^{2}\theta = 0.5(u_{21}^{*} + u_{22}^{*}) \sin 2\theta \end{aligned}$$
(8.398)

Minimizing the energy in the off-diagonal sections to g and u_{23} cas a function of the angle of rotation, we obtain the main orientation of the axis of symmetry. An example of application of Alford rotation to GPR data can be found in Man Gestei and Stoffs (2001).

8.14 Poro-acoustic and electromagnetic diffusion

Diffusion equations are obtained in partor lasticity and electromagnetism at law frequencies and under certain conditions, by which the merical terms and displacement currents are respectively in glected. In this section, we derive the equations from the general theories, study the physics and obtain analytical solutions.

8.14.1 Poro-acoustic equations

The quasi-static limit of Biot's poroclastic equations: to describe the diffusion of the second (slow) compressional mode, is obtained by neglecting the acclerations terms in the equations of momentum conservation (7,210) and (7,211), and considering the constructive equations (7,131) and (7,132), and Darcy's law (7,194). We obtain

$$\eta = M(s - \alpha_0 \epsilon_0^{-\alpha_0}), \qquad (8.399)$$

$$\sigma_{0} = e_{1,0}^{(m)} \frac{\omega_{0}}{M} = (\phi_{0} p), \qquad (8.40.0)$$

$$\partial p_f = \frac{\eta}{\kappa_0} \partial_0 v_0 \qquad (8.401)$$

and

$$\partial_{j}\sigma_{ij} = 0$$
 (8.302)

 $3'_{1}8$

373

where i, j = 1, ..., 3, and the parenthesis in (8,101) indicates that there is no implicit summation. Using $\zeta = -\partial_t c_t$ (see equation (7.1734), doing the operation α on (8.399), substituting (8.401) into the resulting equation, and combining (8,400) and (8.407) gives

$$\frac{1}{M} \frac{\partial q p_f + \alpha_0}{\partial r_0} \frac{\partial q}{\partial r_0} = \partial_t \partial_t \left(\frac{\hbar}{\eta} p_f\right)$$
(8.003)

and

$$\partial_t (c_{jk}^{(n)} t_j^{(n)} + \alpha_{ij} p_f) = 0.$$
 (8.404)

These equations and the strain displacement relations $e_{\alpha}^{(n)} = (\partial_t u_{\alpha}^{(n)} - \partial_t u_{\alpha}^{(n)})/2$ is a set of four partial differential equations for $u_{\alpha}^{(n)}$, $u_{\alpha}^{(n)}$, $u_{\alpha}^{(n)}$ and p_{α} . Equation (8, 104) can be differentiated, and summing the three equations, we obtain

$$\partial_0 \partial_j (e_{ij}^{(m)} e_{ij}^{(m)} - e_{ij} e_j) = 0$$
 (8.405)

In the sotropic case, equations '8 403; and 18 105) because

$$\frac{1}{M}\partial_t p_T + \alpha \partial_t \theta_m = \Delta \left(\frac{\kappa}{\eta} q_f\right)$$
(8.106)

a.201

$$\Delta[\lambda_{\mu}\theta_{\mu} - \alpha\rho\sigma] + 2\partial_{\mu}\partial_{\nu}[\rho_{\mu\sigma}r_{\mu}^{(0)}] = 0, \qquad (8.407)$$

where

$$\beta_m = K_m - \frac{2}{3}p_m,$$
 (8.408)

and we have used reputing (7.28), $\Delta = \partial_t \partial_t, \ \theta_m = e_1^{(0)}$, $\kappa_s = \kappa_s$ and $\alpha_m = \alpha \alpha_m$. If we assume an homogeneous medium and use the property $\partial_t \partial_t e_0^{(0)} = \theta_m$, we obtain

$$\frac{1}{\Omega} \partial_t p_l + \alpha \partial_t \theta_n = \frac{c}{\eta} \Delta p_l \qquad (8.109)$$

and

$$E_{\alpha} \Delta \theta_{\mu} + \alpha \Delta p_f = 0. \qquad (8.140)$$

where E_{cc} is given by equation (7.291). Equation (7.299) is obtained if we take the Laplacian of equation (8, 100) and combine the result with (8, 110). An alternative diffusion equation can be obtained by doing a linear combination of equations (8, 40), and (8, 110).

$$d_t \left(\frac{1}{M}p_f + \alpha \theta_m\right) = d\Delta \left(\frac{1}{M}p_f + \alpha \theta_m\right), \qquad (8.101)$$

where d is the hydraulic diffusivity constant defined in equation (7.300). Then, it is the quartity $M^{-1}pr + \alpha \theta_{0}$ and not the fluid pressure, which satisfies the diffusion equation in Biot's paroclastic theory.

8.14.2 Electromagnetic equations

Maxwell's equations (8.6) and (8.10), neglecting the displacement-currents term $\hat{\epsilon} + \partial_t \mathbf{E}$ and redefining the source terms, can be written as

$$\nabla \times \mathbf{E} \rightarrow \mu (\partial_0 (\mathbf{H} + \mathbf{M}))$$
(8.412)

and

$$|\nabla \times \mathbf{H} - \hat{\sigma}| (\mathbf{E} + \mathbf{J}),$$
 (8.313)

These equations can be expressed in terms of the electric vector or in terms of the magnetic vector as

$$(0, \mathbf{E}) := \sigma^{-1} \cdot \nabla + (\mu^{-1} \cdot \nabla \times \mathbf{E}) + \sigma^{-1} \cdot \partial_t (\nabla + \mathbf{M}) + \partial_t \mathbf{J},$$
 (8.414)

and

$$\partial_t \mathbf{H} = -\mathbf{\mu} - (\nabla \times (\boldsymbol{\sigma}^{-1}) | \nabla \times \mathbf{H}) - \partial_t \mathbf{M} + \mathbf{\mu}^{-1} (| \nabla \times \mathbf{J})$$
 (8.415)

respectively. Assuming a homogeneous and isotropy medium, equation (8.414) carrier rewritten as

$$\partial_t \mathbf{E} = -(\boldsymbol{\mu} \cdot \boldsymbol{\sigma}) - \nabla (\nabla \cdot \mathbf{E}) + \Delta \mathbf{E} [-\partial_t \mathbf{J} = (\boldsymbol{\mu} \cdot \boldsymbol{\sigma})^{-1} \Delta \mathbf{E} + \partial_t \mathbf{J},$$
 (8.416)

where we have considered a region free of charges $(\nabla \cdot \mathbf{E} - 0)$ and have neglected the magnetic source. Note that only for an isotropic mechanic the censors μ^{-1} and σ commute with the outloperator. In this case, $(\mu^{-}\sigma)^{-1} = (p\sigma)^{-1} \mathbf{I}_{a}$.

Similarly repatron (8/115) ran be written as

$$\partial_t \mathbf{H} = (\hat{\boldsymbol{\mu}} \cdot \hat{\boldsymbol{\sigma}})^{-1} \Delta \mathbf{H} + |\boldsymbol{\mu}|^{-1} \cdot \nabla \times \mathbf{J},$$
 (8.4)7)

Equation (8,117) is a diffusion equation for **11**, which is analogous to equation (8,111).

The TM and TE equations

If the material properties and the sources are invariant in the y direction, the propagation can be described in the (x, 1)-plane, and F_1 , E_1 and H_2 are decoupled from E_2 , H_1 and H_3 , corresponding to the TM and TL equations, respectively.

Writing equations (8.114) and (8.415) in explicit Cartesian form, we obtain the TM equations

$$\frac{\sigma\partial_t \left(\begin{array}{c} E_1\\ E_2\end{array}\right) - \left(\begin{array}{c} -\partial_s \mu^{-1}\partial_s & -\partial_1 \mu^{-1}\partial_z\\ -\partial_t \mu^{-1}\partial_s & -\partial_t \mu^{-1}\partial_z\end{array}\right) \cdot \left(\begin{array}{c} E\\ E_s\end{array}\right) - \partial_t \left(\begin{array}{c} -\partial_t M_2\\ -\partial_t M_2\end{array}\right) - \sigma\partial_t \left(\begin{array}{c} J_1\\ J_1\end{array}\right) - \frac{\sigma}{8.4181}$$

and

$$pcsH_{1} = \partial_{t}(\sigma - \partial_{t}H_{2}) + \partial_{t}(\sigma - ^{2}d_{t}H_{2}) = pcsM_{2} + (\partial_{t}A_{-} - \partial_{t}J_{2}), \qquad (8.319)$$

The respective TE equations are

$$\sigma\partial_t E_2 = \partial_t \left(p^{-3} \partial_t E_1 \right) + \partial_t \left(\hat{p} - \partial_1 F_2 \right) - \partial_t \left(\partial_x M_1 - \partial_t M_2 \right) - \sigma \partial_t J_2$$
(8.420)

and

$$\dot{\mu}\partial_{\mu}\left(\begin{array}{c}H_{1}\\H_{3}\end{array}\right) = \left(\begin{array}{c}-\partial_{\mu}\sigma & \partial_{\mu} & -\partial_{\mu}\sigma & \partial_{\mu}\\-\partial_{\mu}\sigma & \partial_{\mu} & -\partial_{\mu}\sigma & \partial_{\mu}\end{array}\right) \cdot \left(\begin{array}{c}H_{1}\\H_{2}\end{array}\right) - \mu\partial_{\mu}\left(\begin{array}{c}M_{1}\\M_{3}\end{array}\right) - \left(\begin{array}{c}-\partial_{\mu}J_{1}\\-\partial_{\mu}J_{1}\end{array}\right), \quad (8.321)$$

Phase velocity, attenuation factor and skin depth

Let us consider an homogeneous isotropic medium. Then, the Green function corresponding to repracting 18, 11(r) and a survive current

$$J_1(x, y, z, t) = t \delta(x) \delta(y(y(z))) - H(t)^2$$
, (8.122)

is the solution of

$$\partial_t \mathbf{E} = a \Delta \mathbf{E} + \phi(x) \phi(y) \phi(z) \phi(t),$$
 (8.423)

where a defines the direction and the strength of the source, and

$$a = \frac{1}{\mu\sigma}$$
(8.124)

In the frequency domain, the diffusion equation can then be written as a Helmholtz equation

$$\Delta \mathbf{E} = \left(\frac{\pi}{r}\right)^{T} \mathbf{E} = -(r/a)\theta(r)\theta(g)\delta(z), \qquad (8.125)$$

where

$$v_i = \sqrt{\frac{as}{2}(1 + i)}$$
(8.126)

is the complex velocity. The same kinematic concepts used in wave propagation (acoustics and electromagnetism) are useful in this analysis. The phase velocity and attendation factor can be obtained from the complex velocity as

$$v_{\mu} \sim [\operatorname{Br}(v_{\mu}))]$$
 and $\alpha \approx -2 \operatorname{Im}(v_{\mu}^{-1})$. (8.427)

respectively. The skin depth is the distance d for which expt. (ad) = 1/c, where it is Napier's number, i.e., the effective distance of penetration of the signal. Using equation (8.426) yields

$$v_p = 2 : fd$$
, and $\alpha = 1, d$. (8.428)

$$u = \sqrt{\frac{u}{z_f}}$$
(8.429)

where $f = \pi_1 2\pi$ is the frequency.

Analytical solutions

Equation (S 423) has the following solution (Green's function)

$$\mathbf{E}(r, t) = \left(\frac{t}{4\pi at}\right) \exp[-r^2/(4at)], \qquad (8.430)$$

where

$$(|z||\sqrt{z^2 + g^2 + z^2})$$
 (8.131)

(Carshew and Jacger, 1959; Polyann and Zartsev, 2004). The func-domain solution for a source $F(t) = c_{\pm}$ equation (2.233), is obtained by a incriminal time convolution between the expression (8.430) and F(t).

Equation (8.323) corresponding to the initial-value problem is

$$\partial_t \mathbf{E} = a \mathbf{A} \mathbf{E}$$
 (8.33)

Assume for each component E_i the initial condition $E_{ij} = E_i(x, y, z, 0) = d_i r(x'y) 0^{-1}$. A transform of (8, 132) to the haplace and wavenumber domains yields

$$F_1(k_x, k_y, k_x, p) = \frac{1}{p + n(k_x^2 + k_y^2 + k_y^2)}$$
(8.433)

where p is the Laplace variable, and the properties $\partial_t F_t \rightarrow pE_t \sim E_t(k_1, k_2, k_3)$ and $E_{tr}(k_1, k_2, k_3) \Rightarrow 1$ have been used.

To obtain $E(k_1, k_2, k_3, t)$, we compute the inverse happace transform of (8, 135).

$$E_{\theta}(k_{0}, k_{0}, k_{0}, \ell_{0}, \ell) = \frac{1}{2\pi i} \int_{\ell_{0}}^{\ell_{0}+\ell_{0}} \frac{\exp(p\ell)dp}{p + \kappa(k\ell - k_{0}^{2} - k_{0}^{2})}$$
(8.431)

where *i* is 0. Three is one pole.

$$\mu_0 = -u(k_0^2 + k_0^2 + k_0^2),$$
 (8.435)

Use of the residue theorem gives the solution

$$F_1(k_1, k_2, h_3, t) = \exp[-a(k_1 + k_1^2 + k_1^2)t]H(t),$$
 (8.436)

The solution for a general initial condition $E_{0}(k \mid k_{2} \mid k_{1})$ is given by

$$E_i(k_1, k_2, k_3, \ell) = E_{\theta_i}(k_3, k_2, \ell_3) \exp (-u(\ell^2 + k_2^2 + k_3^2)\ell_1^2 H/\ell).$$
 (8.437)

where we have used equation 15,136). In the space domain the solution is the spatial convolution between the expression (8,136) and the initial condition. The effect of the exponential on the right-hand side is to filter the higher wavenumbers. The solution in the space domain is obtained by a discrete inverse Fourier transform, using the first [fourier transform. The three components of the objective vector are not functionally independent, since they must satisfy $\nabla \cdot \mathbf{E} = 0$ the aregion free of electric charges). These analytical solutions also describe the diffusion of the slow compressional mode, since equations (8,11) and (8,432) are mathematically equivalent.

8.15 Electro-seismic wave theory

The according to Obris's faw — and a body force on the excess charge of the efficient of the store of the state of the store of the observation of the store of t

382

8.15 Electro-seismic wave theory

There is experimental evidence that controllake (triggering is associated with fluid pressures gradients (Mizictani, Ishido, Yokukura and Ohuishi, 1976). The related fluid flow produces the nation of the fluid electrolyte and creates an electric held (*distribution tic* effect) (Sill, 1983, Pride and Morgan, 1981). It has been reported that accordons electropagnetic emissions were observed hours before the occurrence of major carthopakes and volcanic emptions (Yaxuala, Masuda and Mizictani, 1989). Similarly, the electrokinetic phenomenon may play an important role in predicting rock fracturing in must, and boundary water and oil reservoirs (Wirmstich and Morgan, 1994).

The basic electroscismic theory involves the empling between Maxwell's equations and Biot's equations of dynamical prior lasticity. Frenkel (1911) was the first to have developed a theory to describe the phenomenon. Pride and Garambois (2005) analyzed brenkel's equations and point out an error in developing his effective compressibility coefficients, preventing him to obtain a correct expression for Gassmann's modulus. A complete theory is given by Pride (1993), who obtained the coupled electromagnetic and provelastic equations from first prioriples.

The general equations describing the empling between mass and electric current flows are obtained by imbaiing coupling terms in Darcy's and Ohm's laws (7.194) and (8.5), respectively. We obtain

$$\partial_t \mathbf{w} = -\frac{1}{\eta} \mathbf{\kappa} * \partial_t [\operatorname{grad}(p_f)] - \mathbf{L} * \partial_t \mathbf{E}, \qquad (8.138)$$

$$\mathbf{J}' = -\mathbf{I} + \partial_t [g(\operatorname{ad}(p_f)) + \boldsymbol{\sigma} + \partial_t \mathbf{E} + \mathbf{J}],$$
 (8.139)

where p_i is the pore pressure, \mathbf{E} is the electric field, η is the fluid viscosity, $\mathbf{\kappa}(i)$ is the global-permeability matrix, $\sigma(i)$ is the time-dependent conductivity matrix, \mathbf{J} is an external electric source, and $\mathbf{L}(i)$ is the time-dependent electrokarctic coupling matrix. We have considered time-dependent transport properties (Pride (1994)) equations (7.194) and (S.5) are obtained by substitution of $\mathbf{\kappa}(i)$ and $\mathbf{L}(i)$ with $\mathbf{\kappa}H(i)$ and $\mathbf{L}H(i)$, where the time-dependence is only in the Heaviside step function. For an electric flow deriving from a streaming potential U (Sili, 1983, Warmsteh and Morgan, 1994), $\mathbf{E} = -$ gradif, 1, and the electromagnetic equations (where to quasistatic equations similar to those describing piezoelectric wave propagation, i.e., the acoustic field is coupled with a quasi-static electric field.

The complete time domain differential equations for anisotropic (orthorhordic) media are given by the purorlastic equations (7.255) and (7.255), and the electromagnetic equations (8.1)((8.5)), including the coupling terms according to equations (8,438) and (8,439). We obtain

$$\partial_t \sigma_{ij} = \rho \partial_0^2 \sigma_i^{(m)} + \rho_j \partial_0^2 \sigma_{ij}$$
(8.100)

$$(d_t p_t - \eta_t d_t^t w_t^{ter} + m_t \partial_t^t w_t + \eta_{\lambda t} + \rho_t m = (\mathbf{L} + \partial_t \mathbf{E})_{ter}$$
 (8.141)

$$\nabla \propto \mathbf{E} = \partial_t \mathbf{B} + \mathbf{M}$$
 (8.142)

ant

$$|\nabla \times \mathbf{H} - \partial_t \mathbf{D} - \mathbf{L} * \partial_t \operatorname{grav}(p_t) + \sigma * \partial_t \mathbf{E} * \mathbf{J},$$
 (8.113)

where there is no implicit summation in the last term of equation (8.441). Note the property $d\mu_i(t) + \partial_i\chi_i(t) = \delta(t)$, according to equation (2.11). Pride (1993) obtained analytical expressions for the transport coefficients as well as for the electric rouchu fivity as a function of frequency.

This page intentionally left blank

Chapter 9

Numerical methods

In these process of scientific' apparatus I see out only derives to make the forces of nature science encode in new ways, no. I more then with mark greater respect: I down say that I see in them the true derives for succeiving the essence of things.

Ladwig Boltzmann (1886) (commenting on Lord Kelvin's idea to found a mathematical institute for computations (Broda, 1983)).

Seismic numerical modeling is a technique for simulating wave propagation in the earth. The objective is to predict the seismogram that a set of sensors would record, given an assured structure of the subsurface. This technique is a valuable tool for seismic interpretation and an essential part of seismic viversion algorithms.

In sub- the equation of mation by direct methods, the geological model is approximated by a numerical mesh: that is, the model is discretized in a linkle numbers of points. These techniques are also called grid methods and full-wave equation methods, since the solution implicitly gives the bulk wave field. Direct methods do not have restrictions on the material variability and can be very accurate when a sufficiently line grid is used. Although they are more expressive than analytical and ray worthods in terms of computer time, the technique can easily bundle the implementation of different theologies. Moreover, the generating of supplicits can be an important and in interpretation.

Finite-differences (FD), pseudospectral (PS) and linite-chement (FE) methods are considered in this chapter. The main aspects of the modeling are introduced as follows: (a) time integration: (b) calculation of spatial derivatives: (c) source implementation, (b) boundary condutions, and (c) absorbing boundaries. All these aspects are discussed and dilustrated in the next sections, using the acoustic and SII equations of motion.

9.1 Equation of motion

Consider the bissless aroustic and SH equations of motion which describe propagation of compressional and proceshear waves, respectively.

The pressure formulation for inhomogeneous media can be written as

$$-L^{2}\rho - f = \partial_{\mu}^{2}\rho, \quad -L^{2} = \rho c^{2}\partial_{\nu}(\rho - \partial_{\nu})$$
(9.1)

where v_i , i = 1, 2, 3 are Cartesian modelinates $p(x_i)$ is the pressure, $r(x_i)$ is the vehicity

of the compressional wave, $\rho(x_i)$ is the density and $f(x_i, t)$ is the body force. Repeated natives imply summation over the number of spatial dimensions

The propagation of SH waves is a two-dimensional phenomenon, with the particle velocity, say v_2 , prepriedicular to the plane of propagation. Euler's equation and Hocke's law yield the particle-velocity, stress for anilation of the SH equation of motion.

$$\partial_t \mathbf{v} = \mathbf{H} \cdot \mathbf{v} + \mathbf{f}_t$$
(9.2)

where

$$\mathbf{v} = (v_i, \sigma_{ij}, \sigma_{ij}) = -\mathbf{f} = (f |0| 0)^T$$

(9.3)

$$\mathbf{H} \cdot \mathbf{v} = \mathbf{A} \cdot \partial |\mathbf{v} - \mathbf{B} \cdot \partial \mathbf{v}$$
, (9.4)

$$\mathbf{A} = \begin{pmatrix} 0 & 0 & \mu \\ 0 & 0 & 0 \\ \mu & 0 & 0 \end{pmatrix}, \quad \mathbf{B} = \begin{pmatrix} 0 & \mu & 0 \\ \mu & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}, \quad (9.5)$$

 σ denotes stress and μ is the shear modulus (see Chapter 1). The form (9.2) is representative of most of the equations of motion used in seismic wave propagation, regardless of the stress-strom relation. The solution to equation (9.2) subject to the mutual condition with $-\mathbf{v}_{i}$ is formally given by

$$\mathbf{v}(t) = \exp(t\mathbf{H}) \cdot \mathbf{v}_{n} + \int_{0}^{0} \exp(\tau\mathbf{H}) \cdot \mathbf{f}(t - \tau) d\tau, \qquad (9.6)$$

where exp(f11) is called the evolution operator, because application of this operator to the mutual condition, array for to the source array) yields the solution at tune t. We refer to **H** as the propagation matrix. The SH and acoustic differential equations are hyperbolic than, $BSU(p = 2\pi t)$ for the 1985, p = 1, since the field has a finite velocity.

The standard variational formulation used in finite element methods is written in terms of the pressure. Fourbrain the variational formulation consider a volume Ω bounded by a surface S. The surface S is divided into S_{μ} , where pressure boundary conditions are defined, and S_{μ} , where normal accelerations (or pressure fluxes) are given. Assume a small pressure variation by that is consistent with the boundary conditions. The variational principle is obtained by multiplying reputier (9.1) by $d\mu$ and integrating over the volume Ω and by parts using the divergence theorem).

$$\int_{\Omega(D)}^{-1} d\delta p \partial_t p \, d\Theta = -\int_{\Omega(D)}^{-\delta p} \partial_0^2 p \, d\Theta + \int_{\Omega(D)}^{-1} \frac{f \delta p}{p e^{-\delta}} \, d\Theta + \int_{\Omega_{\Omega}(D)}^{-\delta p} \partial_t q m_e \, dS, \qquad (9.7)$$

where a_i are the components of the normal to the surface S. This formulation is equivalent to a Galerkin procedure (Zienkiewicz, 1977, p. 70; Hughes, 1987, p. 7).

9.2 Time integration

The numerical solution of the equation of function requires the discretization of the time variable using finite differences. (An exception to this is the spectral methods discussed later). The basic idea underlying FD eacthods is to replace the partial derivatives by approximations basic on Taylor-series expansions of functions near the point of interest

356

Forward and backward difference approximations of the time derivatives (Smith, 1985) p 7) load to explicit and implicit FD schemes, respectively. Explicit means that the wave field at a present time is computed from the wave field at past times. On the other hand, in implicit methods the present values depend on past and finite values. Unlike explicit methods, neglicit methods are uncoralitionally stable, but lead to extensive computations due to the need to convolut large matrix inversions. In general, the differential formalation of the equation of mution is solved with explicit algorithms, since the time step is determined by accuracy criteria rather than he stability criteria (Fourierman, Schmid) and Stephen, 2982)

Equations of motion used in scismic exploration and scismology can be expressed as $\partial_t \mathbf{v} = \mathbf{H} \cdot \mathbf{v}$, where **H** is the propagation matrix containing the material properties and spatial derivatives (e.g., equations (1.47) and (9.20). Assume constant material properties and a plane-wave kernel of the form exprikt $\mathbf{x} = (a, b)$, where \mathbf{k} is the real wavenumber vector, \mathbf{x} is the position vector and z_i is a complex frequency. Substitution of the plane-wave kernel into the equation of motion vectors of motion, real-science equations for the equation of motion vectors of motion, these eigenvalues ($\lambda = -i_{12}$). For the aroustic and SH equations of motion, these eigenvalues corresponding to equation (9.2) are $\lambda = -i_{13}$, where c is the shear-wave velocity.

In solution modeling, there are other explations of interest in which eigenvalues might lip in the left-hand λ-plane. We describe some of these below. Consider an anelastic medium described by a vaccoclastic stress-stram relation. Wave attenuation is governed by material relaxation times, which quantify the response time of the nucleum to a perturbation (Lessless relation solid materials respond instantaneously) i.e., the relaxation time is zero. For a viscorlastic medium with moderate attenuation, the eigenvalues have a small negative real part, meaning that the waves are attenuated, the addition, when solving the equations in the time domain, there are eigenvalues with a large negative part and close to the real axis that are approximately given by means the recurred of the relaxation times as a "Lishapentise" factorization mechanism. Then, the domain of the eigenvalues have a "Lishapentise" factorization mechanism. Then, the domain of the eigenvalues have relaxation peaks is close to the source frequency band, or requivalently, if the related eigenvalues are close to the imaginary axis of the λ -plane, an explicit scheme performs very efficiently.

In order to determine the efficiency of an explicit scheme applied to portons media in is critical to understand the roles of the eigenvalues. For portons media, the eigenvalue corresponding to the slow wave at seismic frequencies to quasi-static model has a very large negative part, which is related to the location of the Bjot relaxation peaks, usually beyond the scare band for prior fluids like water and rul (Carcione and Quiroga-Goade, 1996). When the methods of the eigenvalues is very large compared to the inverse of the maximum propagation time, the differential equation is said to be still (claim 1984, p. 72). Smith, 1985, p. 1980. Although the best algorithm would be an implicit method, the problem can still be solved with explicit methods use below).

Denote the discrete time by $\ell = ndt$, where $d\ell$ is the time step, and n is a non-negative integer. There and space discretization of the equation of motion with an explicit a form = forward time difference only - leads to an equation of the form $\mathbf{v}^{n-1} = \mathbf{G} \cdot \mathbf{v}$, where \mathbf{G} is called the amplitudina matrix. The Neumann condition for stability requires max $y_i = -1$, where y_i are the eigenvalues of \mathbf{G} (take, 1984), p_i (H8). This condition does

not hold for all dl when exploit schemes are used, and we note that implicit schemes do not have new restrictions on the time step. For instance, explort for theorder Taylor and Itingo-Nuttic methods require $dt \lambda_{-n} = 2\sqrt{2}$ (Jain, 1981) p. 711, implying very small time steps for very large eigenvalues. Implicit methods are λ stable than, 1981, p. 1184, meaning that the domain of convergence is the left open-hall λ -plane. However, stability does not mean a cursely and, therefore, the time step must comply with certain requirements.

9.2.1 Classical finite differences

Evaluating the second time derivative in equation (9.11 of 10.5) M_{c}^{2} and (n = 160) by a balance expansion, and summing both expressions, we obtain

$$\partial_{\alpha}^{\beta} p^{\beta} = \frac{1}{dt^{\beta}} \left[p^{\alpha} - p^{\alpha} - 2p^{\alpha} - 2\sum_{l=2}^{l} \frac{dt^{2l}}{l! l!} \frac{dt^{l} p^{\alpha}}{dt^{\beta}} \right], \qquad (9.8)$$

The wave equation (9.1) provides the high order time derivatives, using the following recursion relation

$$\frac{\partial^{2}p^{\mu}}{\partial t^{2}} = L \frac{\partial^{2-2}p^{\mu}}{\partial t^{\mu-\eta}} + \frac{\partial^{2-2}f^{\mu}}{\partial t^{\mu-\eta}}, \qquad (9.9)$$

This algorithm, where high-order time derivatives are replaced by spatial derivatives, is often referred to as the Lax Wendrolf scheme (Jain, 1984, $p_{\rm e}$ 415; Smith, 1985, $p_{\rm e}$ 181, Dablain, 1986, Blanch and Robertsson, 1997). A Taylor expansion of the evolution operator $\exp(dt|\mathbf{H})$ is equivalent to a Lax-Wendroff scheme.

The dispersion relation connects the frequency with the wavenumber and allows the calculation of the phase velocity corresponding to each Fourier component. Time discretization implies an approximation of the dispersion relation, which in the continuous case is $\omega = ch$ for equations (9.4) and (9.2). Assuming constant material properties and a 1-D wave solution of the form equification the following dispersion relation the FD angular inspiracy, we obtain the following dispersion relation

$$\frac{2}{dt}\sin\left(\frac{\omega dt}{2}\right) = ck\sqrt{1 - 2\sum_{k=0}^{\infty}(-1)^{k}\frac{(\omega ddt)^{k-1}}{(2t)!}},$$
(9.10)

The FD approximation to the phase velocity as $v \in \sqrt{k}$. Using (9.10) with second order accuracy tracefort $O(dt^2)$ terms), we find that the FD phase velocity is

$$v = \frac{v}{\operatorname{suc}(\theta)}, \quad \theta = fdt.$$
 (9.11)

where $\omega = 2\pi f$ and sinct $\theta = \sin(\pi\theta)/(\pi\theta)$. Equation (9.14) indicates that the FD velocity is steater than the tear phase velocity. Since ω should be a real quantity, thus avoiding exponentially growing solutions, the value of the sine function in (9.10) must be between -1 and 1. This constitutes the stability criterion, for instance, for second-order time untigration this means $ckdt/2 \leq 1$. The maximum phase velocity c_{max} and the maximum wavenumber (i.e. the Nyipist wavenumber π/dr_{max}) must be considered. Then, the condition is

$$dt \le s \left(\frac{dx_{\rm max}}{c_{\rm max}}\right), \quad s = \frac{2}{\tau}.$$
(9.12)

A rigorups demonstration, based on the amplituation factor, is given by Smith (1985) p. 70, see also Celia and Gray, 1992, p. 232). In *n*-D space, $s = 2/(\pi\sqrt{a})$, and for a furth-order approximation (1, (2) in 1-D space, $s = 2\sqrt{3}/\pi$. Equation (9.12) indicates that stability is governed by the minimum grid spacing and the higher velocities

Let us consider the presence of attenuation. As we have seen in previous chapters, time-domain modeling in lossy media described by viscoelastic stress-strain relations requires the use of memory variables, one for each relaxation mechanism. The introduction of additional differential equations for these field variables avoids the numerical compatation of the viscoelastic convolution integrals. The differential equation for a memory variable cut viscoelastic modeling has the form

$$\frac{\partial b}{\partial t} = a - b + b + 0,$$
 (9.13)

(see Section 2.7), where ϵ is a field variable, for instance, the dilutation, and v and b are material properties – b is approximately the central angular frequency of the relaxation peak. Equation (9.13) can be described by using the central differences operator for the time derivative $(dt(b_{i}/\partial t)) \rightarrow e^{i(-1)t} e^{i(-1)t}$ and the mean value operator for the memory variable $(2e^{i(-1)t}) = e^{i(-1)t} + e^{i(-1)t}$. The approximations are used in the Crank Niculson scheme (Smith, 1985, p. 19). This approach leads to an explicit algorithm

$$e^{n_{e}} = \frac{2dta}{2+bdt}e^{n} + \left(\frac{2-bdt}{2+bdt}\right)e^{n_{e}+t/2}$$
(9.14)

(Enumeric), and Korn. 1987). This method is colust in terms of stability, since the coefficient of $e^{6/3}$), related to the viscoelastic eigenvalue of the amplification matrix, is less than 1 for any value of the time step dt. The same method performs equally well for wave propagation in porous media (Carcione and Qarroga-Goode, 1996).

9.2.2 Splitting methods

There integration can also be performed using the method of dimensional splitting, also ralled Strang's scheme (by)a. 1984, p. 114; Bayloss, Jordan and LeMesurier, 1986; Mufth, 1985; Vafulis, Alminovici and Kanasevich (1992). Let us consider equation (9.2). The 1-D equations $\partial_t \mathbf{v} = \mathbf{A} \cdot \partial_t \mathbf{v}$ and $\partial_t \mathbf{v} = \mathbf{B} \cdot \partial_t \mathbf{v}$ are solved by means of one dimensional difference operators \mathbf{L}_i and \mathbf{L}_i , respectively. For instance, Baylos, Jordan and LeMesurier (1986) use a fourth-order accurate predictor-corrector scheme and the splitting algorithm $\mathbf{v}^{(i)} = \mathbf{L}_i + \mathbf{L}_i + \mathbf{L}_i + \mathbf{v}^{(i)}$, where each operator advances the solution by a half-step. The reasonemic allowed time step is larger than for wasplit schemes, since the stability properties are determined by the 1-D schemes.

Splitting is also useful when the system of differential equations is stiff. For instance Biot's provelastic equations can be partitioned into a stiff part and a non-stiff part, such that the evolution operator can be expressed as $\exp(\mathbf{H}_{i} \sim \mathbf{H}_{i})$, when i indicates the regular matrix and since stiff matrix. The product formulas $\exp(\mathbf{H}_{i}) \sim \exp(\mathbf{H}_{i})$ and $\exp([\mathbf{H}_{i})] \exp(\mathbf{H}_{i})/\exp([\mathbf{H}_{i}))$ are first- and second-order accurate, respectively. The stiff part can be solved analytically and the non-stiff part with a standard explicit method (Carciane and Quiraga-Goude (1996) Carciane and Seciani, 2001). Strong's scheme can be shown to be equivalent to the splitting of the evolution operator for solving the puror lastic equations.

9.2.3 Predictor-corrector methods

Predictor-cuencetor schemas of different orders find wide application in scismic modeling (Barliss, Jordan and LeMesmier, 2086; Mufri, 1985; Vatidis, Abramovici and Kanasewich, 1992; Dat. Validis and Kanasewich, 1995). Consider equation (9.2) and the first-order approximation

$$v^{\mu\nu} = v^{\mu} + dt \mathbf{H} / v^{\mu}$$
, (9.15)

known as the forward lixler scheme. This solution is given by the intersection point between the tangent of v at t = odt and the line t = (n + 1)dt. A second-order approximation can be obtained by averaging this tangent with the predicted nue. Then the encoder is

$$\mathbf{v}^{n,i} = \mathbf{v}^n + \frac{d^i}{2} (\mathbf{H}_{\perp} \mathbf{v}^n + \mathbf{H}_{\perp} \mathbf{v}^{n-i}_{\perp}),$$
 (9.16)

This algorithm is the simplest predictor-corrector scheme (Celurard Gray, 1992) $p_{\rm c}(64) = \Lambda$ predictor-corrector MacCormack schemet scroud-order in time and fourth-order in space, is used by Vafidis. Abrammeri and Karasewich (1992) to solve the elastodynamic equations.

The Runge-Kutta method

The Runge-Kutta method is popular because of its simplicity and efficiency. It is one of the most powerful prefictor-correctors methods, following the form of a single predictor step and one or more corrector steps. The fourth-order Runge-Kutta approximation for the solution of equation (9.2) is given by

$$\mathbf{v}^{(-)} = \mathbf{v}^{*} + \frac{dt}{6} \left(\Delta_{t} + 2\Delta_{t} + 2\Delta_{s} + \Delta_{t} \right)$$

$$(9.17)$$

where

$$\begin{split} \boldsymbol{\Delta}_{1} &= \mathbf{H}\mathbf{v}^{(n)} + \mathbf{f}^{n} \\ \boldsymbol{\Delta}_{2} &= \mathbf{H}\left(\mathbf{v}^{n} + \frac{d^{n}}{2}\boldsymbol{\Delta}_{n}\right) + \mathbf{f}^{(n-2)} \\ \boldsymbol{\Delta}_{1} &\mapsto \mathbf{H}\left(\mathbf{v}^{n} + \frac{d^{n}}{2}\boldsymbol{\Delta}_{2}\right) + \mathbf{f}^{(n-2)} \\ \boldsymbol{\Delta}_{1} &= \mathbf{H}(\mathbf{v}^{n} + dt\boldsymbol{\Delta}_{3}) + \mathbf{f}^{n-3} \end{split}$$

The stability region extends to $\lambda_{max} = -2.78$ on the negative real axis and $\lambda_{max} = i(2\sqrt{2})$ on the magnetov axis, where λ_{max} are the eigenvalues of matrix **H** than

1984, p. 71). Hence, the time step is determined by the relation $dt \Lambda_{max} < 2\sqrt{2}$.

....

9.2.4 Spectral methods

As mentioned before to Toylor expansion of the evolution operator $\exp(dt\mathbf{H})$ is equivalent to a Lax Weinhoff scheme. Increasing the number of terms in equation (9.8), allows one the use of a larger time step with high accuracy. However, Taylor expansions and Bunge-Kutta methods are not the best in terms of accuracy. The evolution operator in equation (9.6) can be expanded in terms of Chebyshev polynomials as

$$\mathbf{v}(t) \simeq \sum_{k=0}^{N} C_{0} I_{k}(tR) Q_{0} \left(\frac{\mathbf{H}}{R}\right) \cdot \mathbf{v}_{0},$$
 (9.18)

where $C_0 = 1$ and $C_s = 2$ for $k \neq 0$, J_s is the Bessel function of order k, and Q_1 are modified Chebyshev polynomials. R should be chosen larger than the absolute value of the eigenvalues of \mathbf{H} -(labelie). Kooloff and Koren, 1987). This rechnique allows the calculation of the wave field with large time steps. Chebyshev expansions are optimal since they require the manimum number of terms. The most time consuming part of a modeling algorithm is the evaluation of the terms. The most time consuming part of a modeling algorithm is the evaluation of the terms -L/p in equation (9.1) or $\mathbf{H} + \mathbf{v}$ in equation (9.2), due to the computation of the spatial derivatives. A Taylor-expansion algorithm needs $\mathbf{V} = t_{max}/dt$ of such evaluations to compute the solution at time t_{max} . On the other hand, the annular of evaluations using equation (9.18) is equal to the number of terms in the Chebyshev expansion. Numerical tests indicate that M is comparable to \mathbf{N} for second-order finite differencing, but the error of the Chebyshev operator is practically negligible for single-precision programming that-fizer. Kostoff and Koren, 1987). This means that there is no many reliable to the time integration

When the wave equation is second-order in time as in equation (9.1), the REM method (rapid-exponence method) is twice as efficient since the expansion contains only even order Chebyshev functions (Koshuf) Quenoz Filho, Tessmer and Belde, 1989). A similar algorithm for the viscoelastic wave equation is developed by Tal Elser. Carriene and Koshuf (1990).

The Carlasskiv expansion can also be used for solving parabolic equations (Tal Elect. 1989). Let us consider the 2-D electromagnetic diffusion equation (8, 119). This equation has the form (9.2) with $\mathbf{v} = H_{i}$, $\mathbf{H} = \mu^{-1}(\partial_{i}\sigma^{-1}\partial_{j} + \partial_{j}\sigma^{-1}\partial_{j})$ and $\mathbf{f} = -\mu\partial_{i}M_{i}^{2} + (\partial_{i}J_{i})^{2}$ $\partial_{i}J_{j}$). The eigenvalue equation in the complex λ -domain ($\lambda = -\infty$), corresponding to matrix \mathbf{H}_{i} is

$$\lambda \left(\lambda - \frac{k_{\perp}^2 - k_{\perp}^2}{\mu \sigma} \right) = 0. \tag{9.19}$$

The eigenvalues are therefore zero and real and negative, and the maximum (Nyquist) wavenumber components are $k_1 = \pi/dx$ and $k_3 = \pi/d$; for the grid spacings dx and dz

The evolution operator in equation (8,119) can be expanded in terms of Chebyshev polynomials as

$$\mathbf{v}(t) = \sum_{1=0}^{M} C_{t} \exp(-b) (R_{t} Q_{t} \cdot \mathbf{F}) \cdot \mathbf{v}_{t}, \qquad (9.20)$$

where

$$\mathbf{F} = \frac{1}{b}(\mathbf{H} + b\mathbf{1}),$$
 (9.21)

bis the absolute value of the largest eigenvalue of **H**, and $I_{\rm e}$ is the modified Bessel function of order $k_{\rm e}$. The value of b is equal to $(z^2/\mu)(14dx^2 + 14dz^2)$. As Tablezer (1989) has shown the polynomial order should be $O(\sqrt{b}t)$ (his equation (1.13)). It can be shown that $M = 6\sqrt{b}t$ is enough to obtain stability and accuracy (Carcione, 2006). The main order (Fortran 77) for solving equation (8,119) is given in the appendix (Section 9.9.1). The spatial derivatives are experiments, the staggered Fourier method (see Section 9.3.2). The complete computer program can be downhaded from http://software.seg.org (Carcione, 2006)

These methods are said to have spectral accuracy, in the sense that the error of the approximation tends exponentially to zero when the degree of the approximating polynomial increases.

9.2.5 Algorithms for finite-element methods

In the FF worthood, the field variables are evaluated by interpolation from nodal values. For a second and α is uparametric method (Zienkiewicz, 1977) μ = 178; Hughus, 1987, μ 118), the interpolation can be written as

$$p(x_0) = \Phi \rightarrow \mathbf{p},$$
 (9.22)

where \mathbf{p} is a column array of the values $p(r_1)$ at the nodes and $\mathbf{\Psi}_{-}$ is a row stray of spatial interpolation functions: also referred to as shape and basis functions. The approximation to (9.7) is obtained by considering variations by according to the interpolation (9.22). Since $dp = \mathbf{\Phi}_{-} \otimes \mathbf{p}_1$ and $d\mathbf{p}$ is infimitive the result is next of ordinary differential equations at the model pressures \mathbf{p} (Zienkiewicz, 1977, \mathbf{p}_1 , 53): Highes (1987, \mathbf{p}_1 , 505):

$$\mathbf{K} \cdot \mathbf{p} + \mathbf{M} \cdot \partial_t^2 \mathbf{p} + \mathbf{S} \approx 0.$$
 (9.23)

where **K** is the sufficess matrix. **M** is the mass matrix, and **S** is the generalized source matrix. These matrices concare volume integrals that are evaluated immerically. The matrix **M** is often replaced by a diagonal hunged mass matrix, such that each entry equals the sum of all entries in the same raw of **M** (Zienkiewitz, 1977, p. 535). In this way, the solution can be obtained with an explicit time integration method, such as the rentral difference method, Serön, Sanz, Kindelan and Badal, 1990). This technique can be used with low-order interpolation functions, for which the error introduced by the algorithm is relatively low. When high-order polynomends – including Chebyshev polynomials – are used as interpolation functions, the system of equations (9.25) is generally solved with implicit algorithms. In this case, the most pupular algorithm is the Newmark method (Hagles, 1987, p. 460, Padovani, Priolo and Seriani, 1994, Serón, Badal and Sabadell 1996).

Finally, numerical modeling can be performed in the frequency domain. The method is very accurate but expensive when using differential formulations, since it involves the solution of many Heliofoltz equations (i.e. Shin and Sight 1996). It is more often used in FF algorithms (Marfin), 1984; Santus, Douglas, Mudey and Lovera, 1988; Kelly and Marfin), 1990).

9.3 Calculation of spatial derivatives

The algorithm used to compute the spatial derivatives usually gives its name to the modeling method. The following sections hereby review these algorithms

9.3.1 Finite differences

Finite differences methods use the so-called characgeneous and beterogeneous formulations to solve the equation of motion. In the first case, the motion in each homogeneous region is described by the equation of motion with constant acoustic parameters. For this method, boundary conditions across all interfaces must be satisfied explicitly. The heterogeneous formulation supplicitly enorporates the boundary conditions by constructing indicadifference representations using the equation of motion for bettrogeneous media The homogeneous formulation is of functed used, since it can only be used efficiently for simple geometries. Conversely, the hypergeneous formulation makes it possible to assign different accessive properties to every grid point, providing the flexibility to simulate a variety of complex subsurface models, e.g., random media, vehicity gradients, etc.

In general, staggeted grids are used in heterogeneous formulations to obtain stable schemes for large variations of Poisson ratio (Virieux, 1986). In staggeted grids, groups of field variables and material properties are defined on different resches separated by half the grid spacing (Founderg, 1996, p. 91). The rewly computed variables are centered between the idd variables. Staggering effectively divides the grid spacing in half thereby increasing the accuracy of the approximation.

Seistnic modeling in inhomogeneous mudia requires the calculation of first derivatives. Consider the following approximation with an odd number of points, suitable for staggered grads:

$$\frac{\partial p_i}{\partial x_i} = a_0(p_{1i} - p_{1i}) + \dots + w_0(p_{ij}) + p_{ij} \in \mathbb{N}.$$
(9.24)

with *l* weighting coefficients w_l . The obtiseconstruction form guarantees that the derivative is zero for even process of $r_{\rm c}$ bet us test the spatial derivative approximation for $p = r_{\rm c}$ and $p = x^3$. Requiring that equation (9.21) be accurate for all polynomials up to order 2, we had the approximation $p \mapsto p = pdr$, while for fourth-order accuracy (the boding error term is $O(dr^3)$) the weights are obtained from $w_l \mapsto dw_l = 1/dr$ and $w_l \in 27w = 0$, giving $w_0 = 9/(8dr)$, and $w_l = -1/(2dr)$ (for the graph of p = 91).

To obtain the value of the derivative at r = jdr, substitute substruct 0 with $j, l \in \frac{1}{2}$ with $j \in l \in \frac{1}{2}$ and $|l| = \frac{1}{2}$ with $j = l = \frac{1}{2}$. For observed (1998), p = 1 (provides an algorithm for computing the weights of first and second spatial derivatives for the general case, i.e., approximations which need not be evaluated at a grid point such as contend and one-sided derivatives. He also shows that the FD coefficients m_j in equation (9.24) are equivalent to these of the Fourier PS method when $l \in 1$ approaches the number of grid points (Fourberg, 1996) p = 3.)

Let us now study the accuracy of the approximation by considering the dispersion relation. Assuming constant material properties and a 1-D wave solution of the form expectively $-i\omega t$, the second order approximation gives the following FD dispersion relation and phase velocity:

$$\omega^2 = r^2 k^2 \sin^2(\omega)$$
, $r = c \sin c(r)$, $r = K dr$ (9.25)

where $k = 2\pi K$. The spatial dispersion acts in the sense opposite to temporal dispersion (see equation (9.11)). Thus, the FD velocity is smaller than the true phase velocity

Staggered grads improve accuracy and stability, and eliminate non-causal artifacts (Madariaga, 1976; Virieux, 1986; Levender, 1988; Ozderovar and McMerlan, 1997; Carrione and Helle, 1999). Staggered grid operators are more a carate than central differences operators in the vicinity of the Nyquist wavenumber (e.g., Kneib and Kerner, 1993). The particle-velocity/stress formulation in stagg (ed grids constitutes a flexible modeling technique, since it allows as to freely impose boundary conditions and is able to directly yield all the keld variables (Karrenbach, 1998).

However, there is a disadvaritage in using staggered grafs for anisotropic media of symmetry lower than orthorhoridae. Staggering implies that the off-dragonal stress and strain components are not defined at the same location. When evaluating the stress-strain relation, it is necessary to superover a linear combination of the elastraity constants $\{x_i\}$, $I, J \in [1, -, 1]$, and multiplied by the strain components. Hence, some terms of the stress components have to be interpolated to the locations where the diagonal components are defined (Mora, 1989). The elasticity constants associated with this interpolation, providure are $(x_i, k = 1, 2, 3, 3)$, where (x_i) and (y_i) .

A physical criteriou is to compare the weights w_i in equation (9.24) by mainizing the relative error in the components of the group volucity $v_i = \partial v/\partial k_i$. This procedure combined with grid stargering and a convolutional scheme, yields an optimal differential operator for wave equations (Holberg, 1987). The method is problem dependent, since it depends on the type of equation of motion. Eq.(Mora and Rodler (1995) obtain high a curve with operators of small length (eight points) in the anisotropic case. The treatment of the U-SV case and more details about the limite-difference approximation can be fining in Lexander (1989).

The modeline algorithm can be made more efficient by using hybrid techniques, for instance, combining fraite differences with faster algorithms such as tay tracing methods (Robertsson, Levander and Holliner, 1998) and integral-contracticols (Kummer, Belde and Donan, 1987). In this way modeling of the full wave held can be restricted to the target (e.g., the reservoir) and propagation in the rest of the model (e.g., the overburder) can be simulated with faster methods.

Integrilat itterfaces and variable grid spacing are easily handled by FE methods, since, in principle, grid cells can have any arbitrary shape. When using FD and PS algorithms, an averaging method can be used to reduce spinous diffractions arising from an mappropriate modeling of enryed and dypping interfaces (the so-called staticose effect). Muir Dellinger, Ergen and Nichols (1992) use effective in dia theory based on Backus averaging to find the elasticity constants at the four grid paints of the cell. The modeling requires an anisotropic theological equation. Zeng and West (1999) obtain satisfactory results with a spatially weighted averaging of the model properties. Similarly, algorithms based on rectangular cells of waving size allow the reduction of both stancase diffractions and the number of grid points (Morzo, 1986; Opisal and Zahradok, 1999). When the grid points are not thus in a genue trically regular wav, combinations of 4 D. Taylin series ramot be used and 2 D. Taylor series must be applied (Colia and Gray, 1992, p. 93).

A finite-differences code (Fortza, 77) for solving the SH-wave equation of motion for anisotropic-viscoe'astic media is given in the appendix (Section 9.9.2) and a program for solving Maxwell's equations is given in Section 9.9.3. The latter is based on the acoustic-electromagnetic analogy. Both codes use a fourth-order staggreed approximation for computing the spatial derivatives. The error of this approximation is $3/dx^2/640$, empared to $dx^2/60$ for the approximation on a registargived (Fourberg, 1996, p. 91).

9.3.2 Pseudospectral methods

The pseudo-spectral methods used in forward modeling of scismic waves are malaly based on the Fonder and Chebyshev differential operators. Gazdag (1981), first, and Koslelf and colleagues, later, applied the technique to scismic exploration problems (e.g., Koslelf and Baysal, 1982; Roshef, Koslelf, Edwards and Hsimog, 1988). Mikhailenko (1985) combined transform methods with FD and analytical techniques

9.3 Calculation of spatial derivatives

The sampling points of the bounce method are $x_1 = x_{\max}$, $j = 0, ..., N_i$, where x_{\max} is the maximum distance and N_i is the number of grid points. For a given function f(x), with Fourier transform $f(k_i)$, the first and second derivatives are computed as

$$\partial_t I = ikI_s = \partial_t \partial_t f = -k^2 I_s$$
 (9.26)

where $\hat{\kappa}$ is the discrete wavemander. The transform \hat{f} to the wavemmber domain and the transform back to the space domain are calculated by the fast Former transform (FF1). Staggered operators that evaluate first derivatives between grid points are given by

$$D^{i}f = \sum_{k=n}^{k=N_{i}} \mathrm{i}k \exp(\pm \mathrm{i}k dx/2) \hat{f}(k) \exp(\mathrm{i}kx), \qquad (9.27)$$

where $k(N_{i}) = 1/dr$ is the Nymost wavenumber. The standard differential operator is given by the same expression, without the phase shift term $\exp(z)kdx/2$. The standard operator requires the use of ordebased FFT's, i.e., N_{i} should be an odd number. This is because even transforms have a Nymust component which does not posses, the Hermitian property of the derivative (Koshoff and Kessler, 1989). When f(z) is real |f(k)| is Hermitian frequency is real part is even and maginary part is odd. If N_{i} is odd, the discrete form of k is an odd function: therefore, h(f(t)) is also thermitian and the derivative is real (see the appricity (Section 9.9.11).

On the other hand, the first derivative computed with the staggered differential operator is evaluated between grid points and uses even-based Fourier transforms. The approximation (9.27) is accurate up to the Nymest waverender. If the source spectrum is negligible beyond the Nymist waverender, we can consider that there is no significant numerical dispersion due to the spatial discritization. Hence, the dispersion relation is given by equation (9.10), which for a second order time integration rate by written as

$$\omega = \frac{2}{dt}\sin^{-1}\left(\frac{dtdt}{2}\right), \qquad (9.28)$$

Because k should be real to word exponentially growing solutions, the argument of the inverse sine const be less than one. This implies that the stability condition $k_{inv}(dt/2) \le 1$ leads to $\alpha = -edt/dt \le 2/\pi$, since $k_{max} = \pi/dr$ to is ralled the Contant members. Generally, a criterion $\alpha < 0.2$ is used to choose the time step (Kusloff and Baysa) (1982). The Fermier method has periodic properties. In terms of wave propagation, this means that a wave implicing on the left boundary of the grid will terms from the right boundary (the manerical artifact called wraparound). The Fourier method is discussed in detail in the appendix (Section 9.9.1).

The Chebyshev method is morely used in the particle-velocity/stress formulation to model free-surface, regid and non-reflecting buundary conditions at the boundaries of the mesh. Chebyshev transforms are generally computed with the FFT, with a height twice that used by the Fourier method (Gortfieb and Otszag, 1977, p. 117). Since the sampling points are very dense at the edges of the mesh, the Chebyshev method requires a one-dimensional stretching transformation to avoid very small time steps (see equation (9.12)). Because the grid refly are rectangular, mapping transformations are also used to model curved interfaces to obtain an optimal distribution of grid points (Fornberg, 1988) Carcione, 1994b) and model surface topography (bessmer and lyosholl, 1994b). The bounder and Chelwshov worthouls are accurate up to the maximum wavenumber of the much that corresponds to a spatial wavelength of two grid points – at maximum grid spacing for the Chelwshov operator. This fact makes these methods very efficient in terms of computer storage – mainly in 3.D space – and makes Chebyshev technique highly accurate for simulating Neumann and Dirichlet boundary conditions, such as stress-like and right conditions (Carcione, 1994b) – Lyamples of its use in domayn decomposition is given in Carcione (1996a) and Carcione and Helle (2004) to model wave propagator across fractures and at the acroin bottom, respectively. The Chebyshev method is discussed in detail in the appendix (Section 9955).

9.3.3 The finite-element method

The FE in third has two advantages over FD and PS methods, namely, its flexibility in handling boundary conditions and hregular interfaces. On the basis of equation (9.22), consider the 1-D case, with uniform grid spacing dx_i and an element whose coordinates are Λ_i and $X_2_i(X_i - X_i - dx)$ and whose nodal pressures are P_i and P_2 . This element is mapped into the interval [-1, 1] in a simplified coordinate system (the prference Z-system). Denote the physical variable by x_i and the new variable by x_i . The linear interpolation functions are

$$\phi_1 = \frac{1}{2}(1 - z), \quad \phi_2 = \frac{1}{2}(1 + z)$$
(9.29)

If the field variable and the independent (physical) variable are computed using the same interpolation functions, one has the so-called isoparametric method (Hughes, 1987, p. 20). That is,

$$p = \phi_1 P + \phi_2 P$$
 (9.30)
(9.30)

Assembling the coartilations of all the elements of the stiffness matrix results in a central second-order differencing operator if the density is constant. When the density is variable, the stiffness matrix is equivalent to a staggered FD operator (Kesloff and Kessler, 1989).

EX to the ds have been used to solve problems in seismology, "a particular, propagation of how and Raylogh waves in the presence of surface topography (hysner and Drake 1972) Schlie, 1979). EE applications for seismic exploration require in principle, nonnermory and computer time than the study of surface waves (soil structure interaction) in fact, the problem of propagation of seismic waves from the surface to the target (the teservolic involves the storage of large matrices and much computer time. During the 708 and the 808, choirs were made to render existing low-order EE to during the subset that proposing new algorithms. In the 908, Serier, Sanz, Ku delay and Badal (1990) and Serier, Badal and Schadell (1996) further developed the computational aspects of low order EE to make them more efficient for seismic exploration problems.

When high order FE methods are used, we must be aware that besides the physical propagation modes, there are parasitic modes (Kelly and Marflutt, 1950). These parasitic modes are non-physical solutions of the discrete dispersion relation obtained from the Neumann stability analysis. For instance, for a 2D cubic element grid, there are ten modes of propagation – two corresponding to the P and SV waves, and eight parasitic modes of propagation. High-order FF methods because more obtened with the advent of the spectral

element method (SPEM) (Seriani, Priolo, Carcione and Padowani, 1992) Padowani, Priolo and Seriani, 1994; Priolo, Carcione and Seriani, 1994; Kornatusch and Vikatte, 1998; Kureatusch, Barnes and Tramp, 2000). In this method, the approximation functional space is based on high order orthogonal polynomials having spectral accuracy, that is, the rate of convergence is exponential with respect to the polynomial order. Consider the 2-D case and the acoustic wave equation. The physical domain is decomposed into fouroverlapping quadrilateral elements. On each element, the pressure field $p(z_1, z_2)$, defined on the square interval [1,1,1] \times [1,1] in the reference system Z, is approximated by the following product

$$u_i(z - z_y) = \sum_{j=0}^{N} \sum_{k=0}^{N} P_{ij}\phi_j(z_k)\phi_j(z_y), \qquad (9.31)$$

where P_{i_k} are the radial pressures, and ϕ_i are hag rangian interpolarity satisfying the relation $\phi_i(\varphi_i) = \phi_i(\varphi_i)$ within the interval [-1,1] and identically zero matshir. Here $\phi_i(\varphi_i)$ hences the Kronecker delta and φ stands for [-1 and $[\varphi_i]$. The Lagrangian interpolarity are given by

$$\phi_r(\zeta) = \frac{2}{N} \sum_{i=1}^{N} \frac{1}{v_i e_n} T_r(\zeta_i) (I_n(\zeta_i)).$$
 (9.32)

where L_{i} are Chelwshev polynomials, ζ_{i} are the Gauss-Lobarro quadrature points, and $\alpha_{i} = \alpha_{i} = 0, \ \alpha_{i} = 1$ for $1 \le n \le N$. The Chelyshev functions are also used for the mapping transformation between the physical world X and the local system Z. Seman, Priolo, Carciner and Padouar (1992) use Chelwshev polynamials from eighth-arder to differenth order. This allow up to these paints per minimum wavelength without generating parasitie or spinlous mides. As a result, computational efficiency is improved by about our order of magnified compared to low order FE in the ds. If the misling of a geological structure is as regular as possible (i.e., with a reasonable aspect ratio for the elements), the matrices are well conditioned and an iterative method such as the conjugate grachent uses less their regular densities to solve the implicit system of oppartures.

9.4 Source implementation

The basic sensure sources are a directional force, a pressure source, and a shear source, simulating, for instance, a vertical vibrator, on explosion for a shear vibrator. Complex sources, such as carthquakes sources, can be represented by a set of directional forces origin a double couple (Aki and Richards, 1980, p. 82)).

Consider the so-called elastic formulation of the equation of motion, that is, P and S wave propagation (Kosloff Reshoff and Lorwenthal, (984)). A directional force vector has components $f_{i} = a(x_i)b(t)a_{ini}$, where a is a spatial function possible a Gaussian b(t) is the time history, a denotes the Krunicki radelta function, and w is the source direction. A pressure source can be obtained from a potential of the form $\alpha = a_i x_i b(t)$ as $f_i = a(\alpha - A)$, shear source is of the form f_{--} and \mathbf{A} , where \mathbf{A} is a vector potential. In the tary teplate $\mathbf{A} = (0, 0, A)$ with $A = a_i x_i b(t)$, by particle-velocity/stress formulations, the source can be introduced above or \mathbf{n} the stress-strain relations, such that a pressure source mights equal above or \mathbf{n} the stress-strain relations, such that a pressure source mights equal above or σ_{12} , σ_{22} and σ_{33} of the source location and shear

sources result from a stress tensor with zero trace reigh Bayliss, Jordan and LeMesurer, 1986)

Introducing the source in a homogeneous region by imposing the values of the analytical solution should handle the singularity at the snarre point. More FD techniques (Kelly-Ward, Treitel and Alford, 1976; Virieux, 1986) are based on the approach of Alternian and Karal (1968). The numerical alife ulties present in the vicinity of the source point are solved by subtracting the held due to the source from the total held due to reflection, refraction and diffractions in a region surrounding the source point. This processing unserts the source on the boundary of a rectangular region. The direct source held is computed analytically.

On the other hand, when solving the particle vehicity is tress formulation with pair despectral (PS) algorithms and high order FD methods (Bayliss, Jordan and LeMesnriet, 1986), the source can be implemented in one grid point in view of the accuracy of the differential operators. Numerically (in 1-D) space and maiform grid spacing), the strength of a discrete delta function in the space domain is $1, dx_i$ where dx is the grid size, since each spatial sample is represented by a sinc function with argument x/dx_i . (The spatial integration of this function is precisely dx_i). The introduction of the discrete delta (with alias the wavenumbers beyond the Nyquist (z, dx_i) to the lower wavenumbers. However, if the source time function h(t) is band limited with cut off frequency f_{max} , the wavenumbers greater than $h_{max} = 2\pi f_{max}(t_{ibb})$ will be filtered, where t_{max} is the mitimum wave velocity in the mesh. Moreover, since the equation of motion is linear beismograms with different time histories can be implemented by convolving h(t) with only one surplation using 40t as a source – a discrete delta with strength 1/dt

The computation of synthetic seismonrous for simulating zero-offset (starked) seismon sections requires the use of the exploding reflecting concept (Lorwenthal, Lo. Roberson and Sherwood, 1976) and the so-called correctiveing wave equation (Baysal, Kesloff and Sherwood, 1984). A source proportional to the reflection coefficients is placed on the interfaces and is immated at time zero. All the velocities must be divided by two to get the correct arrival times. The non-reflecting condition ruphes a constant impedance model to avoid multiple reflections, which are in principle, absent from starked sections and constitute anyanted artifacts in impedances.

9.5 Boundary conditions

Free surface boundary conditions are the most originator in seismic exploration and seismology. They also play an important role in the field of non-destructive evaluation for the accurate sizing of surface breaking cracks (Saffar) and Bond, 1987). While in FE methods the implementation of traction-free boundary conditions is natural – simply do not impose any constraint at the surface rades – FD and PS methods require a special boundary treatment.

Some restrictions arise in FE and FD modeling when large values of the Puisson ratio occur at a free surface. Consider first the free surface boundary conditions. The classical algorithm used in FD methods (e.g., Kelly, Ward, Treitel and Allord, 1976) includes a hetitions line of grid points above the surface, uses one-sided differences to approximate manual derivatives, and employs central differences to approximate tangential derivatives. This simple low-order scheme has an upper limit of $c_1/c_2 \sim 0.035$, where c_2 and c_3 are the P-wave and S-wave velocities. The use of a stoggered differential operator and iorhation conditions of the paraxial type (see below) is effective for any variation of Poisson ratio (Virieux, 1986).

The fraction free condition at the surface of the earth can be achieved by using the Fourier PS method and including a wide zone on the lower part of the mesh containing zero values of the stillnesses – the so-called zero-padding technique (Kosloll, Reshef and Loewencha) (1984). While for small angles of varience this approximation vields acceptable results, for larger angles of unidence, it introduces inconencial errors. Free-surface and solid-solid biumdary conditions can be implemented in numerical modeling with non-periodic PS operators by asing a biumdary treatment based on characteristics variables (Koslolf, Kessler, Querioz Filho, Tessmer, Behle and Strahilevitz, 1990, Kessler and Koslolf, 1991; Carcione, 1994; Lessmer, Kessler, Kosloff and Behle, 1992; Igel, 1990). This method is proposed by Bayhs, Jordan and LeMesuner (1986) to molef free-surface and hear-reflecting boundary conditions. The method is summarized below (Tessmer, Kessler, Kessler

Consider the algorithm for the SH equation of motion (9.2). Must explicit time in trigtation schemes compute the operation $\mathbf{H} \cdot \mathbf{v} = (\mathbf{v})^{\mathrm{del}}$ where \mathbf{H} is defined in equation (9.2). The array $(\mathbf{v})^{\mathrm{del}}$ is then optiated to give a new array $(\mathbf{v})^{\mathrm{del}}$ that takes the bound any conditions have accente. Consider the boundary (=0) (e.g., the surface) and that the wave is incident on this boundary from the half-space (=0). Compute the eigenvalues of matrix $\mathbf{B} := \sqrt{\mu/\rho} = +\epsilon c$ and 0 (see equation (9.4)). Compute the right eigenvectors of matrix \mathbf{B} , such that the columns of a matrix \mathbf{R} . Then $\mathbf{B} = \mathbf{R} \cdot \mathbf{A} \cdot \mathbf{R}^{-1}$, with \mathbf{A} being the diagonal matrix of the eigenvalues. If we define the characteristics array as $\mathbf{c} = \mathbf{R} = (\mathbf{v})$ and consider equation (9.2) corresponding to the schemateria.

$$\partial_t \mathbf{c} = \mathbf{A} \cdot \partial_t \mathbf{c}_t$$
 (9.33)

the incoming and outgoing waves are decoupled. How of the characteristics variables, romponents of array \mathbf{c}_i are $c_i + \sigma_{ij}/I$ and $c_i = \sigma_{ij}/I$, with $I = j\mathbf{e}$. The first variable is the incoming wave and the second variable is the outgoing wave. Equating the new and all outgoing characteristics and assuming stress first boundary conditions $(\sigma_{ij} = 0)$, the update of the free-surface grid points is

$$\begin{pmatrix} c \\ \sigma_{12} \\ \sigma_{22} \\ \sigma_{32} \end{pmatrix}^{16W} = \begin{pmatrix} 1 & 0 & f^{-1} \\ 0 & 1 & 0 \\ 0 & 0 & 0 \end{pmatrix} \cdot \begin{pmatrix} c \\ \sigma_{22} \\ \sigma_{32} \end{pmatrix}^{16W}$$
(9.34)

It can be shown that this application of the method of characteristics is explicitlent to a paravial approximation (Clayton and Eugens), 1977) in one spatial dimension. Robertssice (1996) presents a FD method that does not only in must have a staincase shape. The free surface condition is based on the method of images introduced by Levandre (1986). This method is accurate and stable for high values of the Poisson ratio. An efficient solution to the staincase problem is given by Mooze, Bystricky, Kristek, Carcione, and Bouchon (1997), who propose a hybrid scheme based on the discrete-wavenumber, I D and FF methods. These modeling algorithes include attemption based on memory-variable equations (Frumerich and Norm, 1987) Carcione, Noshoff and Koshiff, (1988d)

9.6 Absorbing boundaries

The boundaries of the manifical mesh may produce non-physical artifacts that distants the physical events. These artifacts are reflections from the boundaries or wrapszonids as in the case in the Fourier method. There are two main techniques used in seismimodeling to avoid these artifacts: the sporge method and the paraxid approximation

The classical springe method uses a strip along the homehoies of the annecical mesh, where the field is attenuated (Cerjan, Kesheff, Kesheff and Reshef, 1985). Kesheff and Kesheff, 1986). Considering the pressure formulation, we can write optation (9.1) as a system of coupled optations as

$$\partial_t \left(\frac{y}{y} \right) = \left(-\frac{\xi}{L^2} - \frac{1}{\xi} \right) \cdot \left(\frac{y}{y} \right) + \left(\frac{0}{f} \right), \qquad (9.45)$$

where ξ is an absorbing parameter. The solution to this equation is a wave traveling with out dispersion, but whose amplitude decreases with distance at a frequency independent rate. A traveling pulse will, thus, diminish in amplitude without a change of shape. An improved version of the sponge method is the perfectly matched-lawer method or PML method used no electromagnetism and enterpreted by they and har (1996) as a coordinate stretching. It is based on a - non-plexical - modification of the wave equation inside the absorbing strips, such that the reflection coefficient at the strip/model boundary is zero. The improvement implies a reduction of math 75 % in the strip thickness compared to the classical method.

The sponge method can be implemented in FE modeling by including a damping matrix **D** in equation (9.23).

$$\mathbf{K} \cdot \mathbf{p} + \mathbf{D} \cdot \partial_t \mathbf{p} + \mathbf{M} \cdot \partial_0^2 \mathbf{p} + \mathbf{S} = 0.$$
 (9.35)

with $\mathbf{D} = \alpha \mathbf{M} + \beta \mathbf{K}$, where α and β are the damping parameters to \mathbf{c} . Sarma, Mallick and Gadhinglajkar (1968)

The paraxial approximation method is another technique used to avoid undescuble non-physical artifacts. One was equations and the method based on characteristics variables discussed in the previous section are particular cases. For approximations based on the one-way wave equation (paraxial) concept, consider the accustic wave equation on the domain $\phi > 0$. At the boundary $\phi = 0$, the absorbing boundary condition has the general form

$$\left\{\prod_{i=1}^{n} \left[(\cos \phi_i) \partial_i - \phi_i \right] \right\} p = 0.$$
(25.37)

where $|\phi_{1}| \ll \pi/2$ for all y (Higdon (1991), Lepizotion (9.37) provides a general representation of absorbing boundary conditions (Kees, 1985); Randall, 1988). The reason for the success of repretion (9.37) can be explained as follows: Suppose that a plane wave is hitting the boundary at an angle ϕ and a velocity ϕ_{1} in 2 D space, such a wave can be written as $p(x \cos \phi + z)$ since (x). When an operator of the form (cos $(d_{1} + d_{2}))$ is applied to this plane wave, the result is zero. The angles ϕ_{1} are closen to take advantage of a priori information about directions from which waves are expected to reach the boundary.

Consider now the opproach based on characteristics variables and apply it to the SII equation of motion (9.2) in the plane z = 0. The ontgoing characteristic variable is

 $\sigma_1 = \sigma_1/J$ (see the previous section). This mode is left unchanged (new = old), while the incomenty variable $\psi_2 + \sigma_{s_2}/J$ is set to zero (new = 0). Then, the update of the boundary grid points is

$$\begin{pmatrix} -r \\ \sigma_{12} \\ \sigma_{12} \\ \sigma_{12} \end{pmatrix}^{6.7} = \frac{1}{2} \begin{pmatrix} 1 & 0 & I^{-1} \\ 0 & 2 & 0 \\ I & 0 & 1 \end{pmatrix} + \begin{pmatrix} -r \\ \sigma_{12} \\ \sigma_{12} \\ \sigma_{12} \end{pmatrix}^{61} ,$$
 (9.38)

These exptations are exact in one dimension, i.e., for waves incident at right angles. Approximations for the 2-D case are provided by Clayton and Engquist (1977).

9.7 Model and modeling design – Seismic modeling

Modeling synthetic sciencegrams may have different purposes — for instance, to design a seismet experiment R followar. McMischan and Chanov, 1996), to provide for structural interpretation (Figure 1992) or to perform a sensitivity analysis related to the detectablety of a petrophysical variable, such as percesity fluid type, fluid saturation, etc. Modeling algorithms can also be part of inversion and migration algorithms.

Designing a model requires the joint collaboration of geologists, geophysicists and loganalysts when there is well information about the study area. The geological modeling procedure generally involves the generation of a seismic-coherence volume to define the main reservoir units and the incorporation of fault data of the study area. Seismic data require the standard processing sequence and pre-stack depth indication supported by proper inversion algorithms when possible. A further improvement is achieved by including well-logging fronties and density (log) information. Since the logs have a high degree of detail, averaging factleds are used to obtain the velocity and density field at the levels of seismic resolution.

In planning the modeling with direct worthods. On following steps are to be followed:

 From the maximum source frequency and minimum velocity, find the constraint on the grid spacing, amounty

$$\frac{dx}{2f_{max}} = \frac{c_{max}}{2f_{max}}, \quad (9.39)$$

The explat sign implies the maximum allowed spacing to avoid allosing: that is, two points per wavelength. The actual grid spacing depends on the particular scheme. For instance, O(2/1) UD schemes require 5 to 8 grid points per maining inwavelength.

- 2. Find the number of grid points from the size of the model.
- 3. Allocate additional wavelengths for each absorbing strip at the sides, top and borroun of the model. For instance, the standard spenge method repares four wavelengths, where the wavelength is $\lambda_d = -2r_{max}/f_d$ and f_d is the dominant frequency of the science symple.
- Choose the time step according to the stability condition (9.12) and accuracy criteria. Moreover, when possible, test the modeling algorithm against the analytical solutions and perform seisme-reciprocity tests to verify its correct performance.
- 5. Define the source-receiver configuration



Figure 9.1. Gaslag calls sold

Consider the model shown in Figure 9.1 with the properties indicated in Table 9.1 the low velocities and low quality factors of medium 7 singla is a saufstone subjected to an excess pow pressure. All the media have a Poissue ratio equal to 0.2, except medium 7 which has a Poissue ratio of 0.3, corresponding to an overpressured condition The modeling algorithm (Carcion), 1992at is based on a fourth-order Runge-Kurra timeintegration, scheme and the Fourier and Chebyshey methods, which are used to compute the spatial derivatives along the horizontal and vertical directions, respectively. This allows the modeling of free-surface houndary conditions. Since the most is exarge theor points per manimum wavelength). Zeng and West's overaging method (Zeng and West 1990) is applied to the slownesses to avail diffractions due to the staincase effect TLE density and the relaxation times are arithmetically averaged. The mesh has 135×129 points, with a horizontal grid spacing of 20 ns and a vertical dimension of 2181 m with a maximum vertical grid spacing of 20 m. Stress-free and non-reflecting boundary conditions of the type 19.31; and 19.38, are applied at the top and bottom boundaries, respectively. In addition, absorbing houndaries of the type 19.3% of length 18 grid points are implemented at the side and but tous houndaries. The source is a vertical face, (a Ricker wavelet) applied at 30 m depth, with a maximum frequency of 10 Hz. The wave held is computed by using a time step of 1 ms with a maximum time of 1 s - the total walls lock time is 120 s in an Origin 2000 with 1 CPU's. The seismogram recorded at the surface is shown in Figure 9.2. where the main event, is the Rayleigh wave (ground-roll) travelying with velocities between the shear velocities of media, I and 2, approximately. The reflection event corresponding to the acticlual structure can be clearly seen between 0.6 s and 0.8 s

¹ Modoum	- ege -		Q_{0}	Q_{s}	·
	(km/s)	(kmys)			2.00
· 1	- <u>2</u> 6	l li	'su'	141	' 21 '
2	3.2	1 96	1181	76	2.3
:1	3.7	2.26	110	N5	2.3
I I	•	2,45	115	96 F	2.1
5	13	2.63	120	92	2.5
h	1.5	2.75	125	95	2.6
ï	1.2	1.7	30	25	2.3
4	1.6	2.52	150	115	2.6
9	1.8	2.94	160	120	2.7
. 10	. h.t. ,	1.3	. 220 .	170	2.8

Table 9.1. Material properties



Figure 9.2: Sessing non-of-the vestical particle relativy

Forward numerical modeling is a powerful method to aid in the interpretation of seismic surveys. Carcion: Fitterti and Gei (2004) use my tracing, the non-reflecting wave equation and the explositing-reflector approach to interpret low signal-to-noise ratio deep crust seismic sortions. Synthetic seismograms are useful to recognize patterns associated with deferred types of structures, and predicting some of the drawbacks when caterpreting relignated and quantynated sections of a given complex structure.

Another useful application is seismic characterization. Cardone and Ger (2003) use took physics, seismic theory and momenical modeling of wave propagation to analyze the seismic response of an autoretic subglarial lake. Optimal seismic surveys can be planned on the basis of this type of investigations.

9.8 Concluding remarks

The direct methods docussed in this chapter (finite difference, pseudospectral methods and linite-element methods) do not impose restrictions on the type of stress-strain relation, boundary conditions or source-type. In addition, they allow general material variability. For a stance, the numerical solution of wave propagation in an according poro-viscoelastic medium – appropriate for assessive environments – is not particularly difficult in comparison with simple cases, such as the accessive wave equation describing the propagation of dilatational waves. Many of the complex stress strain relations handled by direct methods cannot be solved by integral equations or asymptotic methods without snaphilying assumptions. However, direct methods for solving these equations are obtainly more expensive in terms of computer tone and storage.

Finite differences are simple to program and are efficient when compared to alternative methods, make fairly mild arcmacy requirements. In this sense, a goal choice can be a second order in time, fourth order in space FD algorithm. Pseudospretral methods can be more expensive at some cases, but guaranter higher arcmacy and relatively lower background noise when staggeted differential operators are used. These operators are also simulde when large variations of Poisson ratio are present in the model to go a fluid/solid otterface). In three dimensions, pseudospectral methods require a number of group points compared to finite differences, and can be the best choice when limited computer storagis available. How yet, if a drose grad is required for physical means (e.g., fin, hypering scattering informageneities, or it the FD algorithm can be more convenient.

Without a doubt, the hest advorithm to model surface topography and curved interlaces is the finite-element method. With the use of spectral interpolators, this algorithm concompute with earlier techniques with respect to accuracy and scalibly. However, this approach new prove to be justable for large equations of the Poisson paper. Finite-element methods are best suited for engineering problems where interfaces have well defined genmetrical features, in contrast with geological interfaces. Moreover, model meshing is not intensively trupited as is the case in seismic interfaces. Moreover, model meshing is not intensively trupited as is the case in seismic interfaces algorithms. Use of our certangular grids, mainly in deD space, is one of the main disadvantages of limite-element methods, hereause of the topological problems to be solved when constructing the model. Uniteelement methods are, however, preferred for sensine problems involving the propagation of surface waves in semations of complex topography.

9.9 Appendix

9.9.1 Electromagnetic-diffusion code

The following Fortrain 77 computer program implements the simulation of the initial value problem corresponding to the TM equation (8,119) using the expansion (9,20) and the Fourier PS method (equation (9,27)). The same program can be used for the TE equation (8,420) if the conductivity is interchanged with the magnetic permeability and vice versa-

The model is homogeneous, but the properties are defined as arrays, so the program ran be used for a general informogeneous machine. The first-order spatial derivative computed with the staggered differential operator uses even-based Former transforms. The spectral coefficients of the Former reparsing are computed by the East Former Transform (FFT) using the algorithm introduced by Transform (1988), requiring the number of grid priors be composed of prime factors formed with 2, 3, 4, 5, 7, 8, 0, 11, 13, 16 and 17.

The routine for the moduled Bessel functions is taken from Zhang and Jir (1996), who provide a floppy disk with the program. This code exceeds the dynamic range of the computer (Origin 300) for arguments larger than 500, but a small modification allows the radiofation of exp($-\Theta(I_{1})bt$), which poses no difficulties. The main equations describing the algorithm are indicated in the comments.

I Electromagnetic diffusion optation. Magnetic liefd r Section 8,10.2 : Differential equation (8/419) Section 8,10.2 : Analytical solution requation (8,137)) r Section 9.2.1 - Time integration (equation (9.204) r Section 9.3.2 : Spatial derivatives loquation (9.27)) parameter bixt, 120 nzt, 120 nbes, 2000 na, 200 dimension Hyperxt nzt (Hylenstenzt) (Hyterstenzt) (Barension Expectatize) (E2)(x)(12) (dennesia blanbes) dimension ifaxee III jakstnet prostnet (isistnet) dominision ifaxz-00.ak/tuzto.co/tuzt.si/tuzt) real*8-bid.1kr0mbes(.4kpe0.nbes),Kk00mbes(.Kkp00mbes) real negative averables dimension sigma(nxt.nzt.twx)(a)(wz)(a) cummon/tre/lifae data pi/3 14150265/ r open idih = [SNP]e INPUT DATA Number of grid points (These numbers should be even and composed of primes factors) nx=nxt 1.2 = 0.71 Grad spacings dx 10 ショム Initial-condition parameters kbar 111

```
Dk 0.5*kba
c Mudel
e Reference magnetic permeability and conductivity
       annul l'pric 7
       50-00001
       atrix=0.
      do Li Lux
       do t j= taz
          mulitie-anen
          SCHOOLD SO
          a=1./tumei.j('sigmati.jt)
Т
          atrix-attax1 (attxtabs1a1)
       a lams

    Propagation time

       t = 3 \mod b

    If ideal apply absorbing boundaries

      ia's
Ċ,
eINITIAL CONDITION
       s0=0.5fas1ds
       20-0.5° (2.5) s
      do 5 i - Lux
       ຟດວ່າງ ໄປຂ
          x=(i+1)bix
          z (g. 1)tdz
arg = 0.257DktDkt/18 (s061726)z (200112)
          argl = kbar((x - x0))
          arg2_kbar*17_200
'n,
          Hyperg ("costary hitestary?")
U,
e Wavenenber components and phase shifts for staggering
       nall wielaky invisitentedyj
       rall wielakzenzisizarzeż}
e Vertse-FLT factors
       call spfat7(ifaxoury)
       rall splat 7(ibsz az)
+ ABSORBING BOUNDARIES
       il(iaboq.1) then
         nab 18
         gaun=Less fil
         alp=0.1

    Weights for the absorbing strips.

         rall wgrowenah gamaljó
         call wgttwznab.gam.alpt
c Define properties of the hottom strip.
         do 10 r. Los
         do 10 j=nz-nab+0.nz
           muniti nutulo
10
           signaria) signati D
       rudif
۰.
       do II n. Lux
       do 11 j=1.uz
          Hylagr 9.
ш
          Hernin 0
```

```
to a priprint, blx/dx -1, blz/dz)
                       le bit
1
« EXPANSION COLFFICIENTS
                       M 6 serubu
                       lod-br
                      exp(= br) 1k(br)
call IKN \(M1cd,NM1k 1kp,Kk Kkp)

    Jk

                       16(1) \rightarrow 1k_0(1)
                       58(2) 2 *l8(1)
                       da(15|\mathbf{k}|||3|\mathbf{M}|)
                                 kl⇒k−1
1.5
                                6s(k) 2.*1k(k1)
÷
4 TIME EVOLUTION

    First two terms

                       Éar I
                                 call chebeHv.Hv.t.mu.sigma.ux.uz.
                  ¥.
                                                                 ilaxs.aks.cox.sis.
                                                                 ifayz akz noz siz
                  ٨÷
                  k
                                                                 www.coabliab1
4
                      do 20 i - Uny
         du 20 j⇒1.az
Isq. (9.20) **
 ••
20
                               Hereigh Hough Skitt Herei (1802) Herigi
1
    Jerus 2...M
4
                       for 2
                       do 25 k=3.M+1
                                 call cheb(Hv1.Hv.mu.sigma.ux.uz.
                  ٨÷
                                                                 ifays,aks.coy.six
                  k
                                                                 ifasztakztroztsiz.
                  ¥.
                                                                 wx.wz.acob.iab1
ł,
                                 do 30 i=t.ox
                                 (5) 30 (1.1a)
                                          Hyfrige Hyfrige Skik("Hyfrige
Sh-Hyffige
                                           Hyltigio Hy(ig)
30
                                          llyn, polifi.
r.
25
                               e of this for
÷
4 COMPUTE ELECTRIC FIELD
e Spatial derivatives: Dyt (1) and Dz1(3)
                       call differHyt Exts 1.00 faxs also consistenting)
                       call officially allows and a second size of the sec
                       da 35 i - Env
                       da 35 j⇒taz
                                 Ev(ij) -- Ev(ij)/signatij)
Ez(ij) - Ez(ij)/sepsatoj)
35

    WRITE SNAPSHOTS

                       writer for the nz
```

```
do 105 - Lus
          writer (17) Hwrgigh Exter) (Estiggg - Case
K) (
ı.
       write(6.1000,*1.e+6
       formatt1x."Propagation time: ".F1.0." microser.")
100
       write 6.101160
       formattix, Argament of Bessel featuris, bt: - F540
100
       write/fi.1021M
102
       formattEx, Chebyshev polynomial degree, M: 7.11;
       wnie(6.10315k)Mr/5k/1r
       formatt(s. Statio bM/.bfb/.F157)
103
ı
       stop
       ruÍ
e End of main program
۰.
» SUBBOUTINES
Ċ,
« SPATIAL DERIVATIVES AND RECEIPTION FOUNDAND
e.
       subjoutine chebeHv.Hv1.mu.sigma.us.uz.
     æ
                       ifaxe.ake.roe.six
     ġ,
                       ifaxz akzuvzsiz
     s,
                       wx.wz.isabdabi
       dimension Hypexity (Hypexitz)
       dimension attrixing(a2) group (a3) group)
       dimension (fasx110).aks/nx().cox(nx).six(nx)
       dimension (faszi 100/wsztuczztaczznz), szenze
       dimension signa (axea) we trade, w/trade)
       real antifuxing [
       communication (b.fac
e Spatial defivatives. Dx > ) and Dz(+1) *** Eq. (9.27) **
       call difxtHv.at. (10) faxsaks/cox/sis/nvn//
       call dif/(Hyay > 1,0),fasz akz roz sizurenz).
c.
       do 5 i - Lux
       ຟດວ່າງ ໄປຂ
          alti.jt=al(i.jt/sigmati.jt)
5
          c Spatial derivatives: Dx = c and Do(=) [131] Eq. (9.27) [31]
       call difference3, 1.0. (faxstakstows sixtux nz)
       rall difztag a3 = 1, trifasz akz rez, siz ny az r
c(a) = o(1/sigma) Hype(x) + (1)/sigma) Hype(x)
r'
       do 10 n. Lux
       do 10 j=1.0z
Ъu
          siliji stručnotiji
e Apply absorbing boundaries
       if(along1) then
         rall alco3.11v.wv,wz,coh av az r
       rudif
e Recursion equation
       do Dafi Lux
```

```
do 15 % 1.52
          GN=a3 hj)
** Lap. (9/21) **
          FN=(GN+h<sup>a</sup>Hyri,j)), b
15
          Hyfrijt fae'EN İlvhijt
ı.
       D----11-11
       end
ı
r.
 Subcontines
ı.
ı
 Modified Hessel functions (Zhang and Jin, 1996).
ı.
     subjointing IKNA/mx/min.bidi.bk.dkt
ı
ı
e Wavenumber components and phase shifts for staggering
     subjoriting wayak, cost a dif-
r.

    x-derivative

     subrouting difvial agaisging tifaxy also receivers as
r.
 zəletiyatiya
ı
     subjoutine difzja La2.isc.com (fasztakztrozisiz)(x.i.z.)

    Vector FFT (Temperion, 1988)

     subjourne sulat 70 fax.ed.
r.
     subroutine plat7(add2axine pumpa, intdsignder))
ı
r Absorbing homolaries (Kosloff and Kosloff 1986).
     subtomine wertwinals.gam.alpr
ı
     subsorting abtacting wx.wz nab.mx.mz]
ı
ı

    The complete computer program can be downloaded
```

```
n from http://software.see.org/
```

9.9.2 Finite-differences code for the SH-wave equation of motion

The following Forman program solves the inhomogeneous anisotropic and viscoclastic SHwave equation of morror, which is given in Section 1.5.3. The time discretization of Fuler's reprating (1.46), has second index accuracy, and it is based on equation (9.8) (the first three terms on the right-hand side ()

$$u_{\mu}^{0/3} \le 2a_{\mu}^{0} - u_{\mu}^{0} + dv_{\mu}^{2}\rho_{\mu}(D_{\mu}a_{\mu} + D_{\mu}a_{\mu}) + f_{\mu\mu}^{2}$$
(9.40)

where $\partial \sigma_{i} = c_{i} / \sigma_{0} = \sigma_{i}$ and $\sigma_{i} = \sigma_{i}$. The strain components are obtained as

$$|x_3| \le D_s u_c$$
 and $|x_b| \le D_s^+ u_c$. (9.41)

where D_{-} and D_{-} represent staggered spatial durivarive operators of order 4. The different signs imply a shift of half the grid size, to obtain the acceleration at the same points of the displacement (Carcione, 1999c).

The discretization of the momory-variable operations (11180), and (11180), is based on
equation (0.14). For example, the fast equation is

$$e_{21}^{n} = \frac{2}{2} \left(\frac{2}{2} \frac{dt}{dt} \frac{d^2}{dt^2} \frac{d^2}{dt} \right) e_{3}^{n} + \left(\frac{2i_{2}^{12} - dt}{2i_{2}^{12} + dt} \right) e_{23}^{2} = \frac{2}{2}, \tag{9.12}$$

where ϵ_{j_1} denotes the memory variable, and $\varphi_j = (\omega^2 + (\omega^2 + \omega_j^2))^2$. On a regular goal, the field components and centered properties are represented at the same grid points. On a staggered grid, variables and material properties are defined at half-grid points, as shown by Carvione (1999e). Material properties at half-grid points should be computed by averaging the values defined of regular points (not implemented in this program. The averaging is chosen in such a way to reduce the error between the numerical solution corresponding to an interface aligned with the numerical grid and the equivalent solution obtained with a regular grid. Minimum ringing amplitudes are obtained for the authmatic average of the density and relaxation times, and the genmetric average of the shear moduli.

In particular, the program solves the reflection-transmission problem of Section 6.1 for a source of 25 Hz control frequency. The mesh has 120 × 120 points and a grid spacing of 10 m. A snapshot of the displacement weat 250 ms is shown in Figure 9.3.



Figure 9.3: Supplier of the SIL wave displacement, or aspending to the reflection that says comproblem studied in Section 6.1. The star indicates the location of the section

The comments in the program indicate the different rotations used in the significant

c e Amsorropic, viscoclastic SH-wave propagation c c Section 4.4 (c) Plane wave analysis e Section 4.5.4. Differential constitutes Section 4.6 - Analytical solution e Section 6.1 - : Reflection transmission problem e Section 8.2.1 : Time integration c Section 8.3.1 · Spatial derivatives

```
c O(2.4) finite-difference scheme
4
       parameter (ax) [120, nzt [120, nstep] [520)]
a tield variables

    u2. displacement

e el and etic strant comporents
nisk and still stress components

    e23 and e12: memory variables

       denier sion utilistance) universities linetance), sfitustance)
       dimension, e23(met.mzt ) of 2(met.mzt)

    material properties

       drama sion (s201x) avtophi2/axtor/tits1(axtor/tiple40/x00/t)
       dimension, e 14 metanzi 5 (66) astanzi (re46 metanzi orba) netanzi (
4
       dynn son seistiktu steptt
       dimension also 302
       dimension, hebboor
÷
       opene DLEIG-TSNAP")
       open(15.02+"SEIS")
         dx 10.
         dz = 10.
         11-10.001
         nx 120
         nz 120
         ustep=250
         pi 3.14159265
a snapshots every risp steps
       csp = rstrp
e source location and central frequency
       ix fill
       i_2 = 1i0
       fteq - 25
a MODEL
e central frequency of relaxation peaks
       fet 25
       ton 1972 (pi19))
       dn = 1 \pi x
       dog = 1.az
e upper laver
<sup>11</sup> Eqs. (6.22) (6.24) 12
       rko(i.j) 2000.
       e14(ij) - 9486+9
       ena(ij)=12.5e+9
       -16(ij) = 0.5(sor)/13(ij)(66(ij))
a loss in vertical direction
       Q_{2-30}
** Lg - 0620 0 **
         (a) (a) (0) (Q2) (sort(Q2)Q2 (1) -1).
         tr 2=0 au/Q210 sqrt(Q21Q2 - 1.1 - 1.)
         phi2(i.j) 1/(n/2 1./082.i.j)
a loss in horizontal direction
       Q = H
         (3):10.0703/18qc(Q1'Q1+1.) 1.0
         (i) Data Q1C sq(1)Q1*Q3*11*1;
```

```
phillips 1,001 1 /ostriju
e lower laver
                        if(japo.80) theat
\begin{array}{c} 0(1,20,500,1063)\\ \hline 10(6,22)(6(2,3),15)\\ + 0(6,61)(-25,00)\\ + 11(6,1)(-125,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-25,06)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(-9)\\ + 686(1,1)(
e loss in vertical direction.
                        Q2 = 60.
                              iseria) = (tam/O207) sqr(1O22O2 - 1.) - 1.)
                              te2=fran/Q2(fisar)(Q2(Q2)Q2+).)+1.)
                              phi2mp 1,002 1,08201.jp
e loss in lanizantal direction
                        Q4, 80.
                              (strij) (ar/Q1) (sq0(Q1)Q1)(1) 1.)
                              (a) (ba)/QU [sort(QU)] (1) (1)
                              philippel/ml-1/istriju
                        endit
                        end do
                        rud do
r,
                        do i Lux
                        do j=Luz
                        u2(i,j)=0.
                        uti,ji li
                        not do
                        end do
                        doug Lustep
                        f(a) = 0
                        cut do
cubsorbing parameters
                        r 0.997
                        nab 12
                        do i=Luah
                        aboit r**i
                        abai) 1
                        rud do
e someés waydet
 14 Eq. (2/233) 14
                        rall wavelet(Efregany.dt)
e huite-differences weights
 17 Eq. (9/21) 17
                        x1=9.7(8/1)fx1
                        x2 1.7(21.50x)
                        71 9 /(8 1dz)
                        /2 = -1.7(24.2)(z)
c TIME STEPPING
                        do 10 nº 1 notep
                        if(ano5)a.10 ceq.0 (print 1.a
c apply absorbing boundaries.
e horizontal strips
                        do 11 j=1.uali
                       (2-j-2
(3-12-j-1
```

```
sale abtuales E 💥
                    dn | 11 i = 3.nx - 2
                    (2002) 02002)*sab
 11
                    n2(i,j4(=n2(i,j3)*sab

    vertical strips

                   du 12 i⇒t.nab
                    12 1 - 2
                    3240x - i - 1
                    sale abjudes (-i)
                    do 12 j. -3.nz - 2
                    a2(a)j(=a)i2(j)*sab
 12
                    a2(id.j) /02(i3.j) sab
ar.
                    do 13 i 53.ng/2
                    do 13 j 3 nz 2
e straits
 ** Eqs. (9/24) and (9,41) **
 n i - 3/2 - 5 i - 2
(i) 1.2 (> i) 1
i i < 1/2 - 5 ×</p>
i:i:3/2 \rightarrow i+1
                    et 21 (c2(q) (c2(q - 1))) / 22 (c2(q) (1) (c2(q - 2))
                    et=x1*,a2(),j++a2(i+1,j)+x2*(a2(i+1,j)+a2,i+2,j))
a mercor complete equations
                        f)=2.0s2(hj)+dt
                         N 211825 (162
                        (c) (233).0
 17 Feys 104 1491, and 19 421 17
                    (23) iji (2.50) (s2) iji (ph2) iji (c1) (1) (20) iji (22)
                    e23(i,j)=0.5*(e23(i,j)-ee)
                         (1 + 2.7 \text{ s}(5))) dt
                         12-2.0 s1(hj) - dt
                        aer 1420.0
 <sup>14</sup> Eqs. (4.1494, and (9.42) <sup>14</sup>
                    erzaju uz til testa juplati juta - fiterzi ju 72
                    el2(ii) 0.5"(el2(ii) - eet
e ser cosserva
   1 Eq. (1.150) 11
                    stuji a Hogi tel se230 po setoriji eri
                    stiritj1=r6britg(*teti ve12fitj1) - c46ritj(*e4
 13
                    continue
                    do(14) = 3.0x - 2
                   do 144 3 nz 2
 11 Eq. (9.24) 1
(i) 3.2 (N) 1
ci 1/2 0 m
+i+1/2 > 5+1
e) (3/2) > 5 (2)
                   ds_4 = z 1^{+} (s l(i, j + 1) + s l(i, j)) + z 2^{+} (s l(i, j + 2) + s l(i, j + 1))
                    dati wertaati dige aanige wertaite zijn atte dige

    according to the second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second se
 second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second second sec
                   per als todsti fatataji
1 SURFICE
                    source (0
```

if(nde aw mobile quistanchi e quizt source of ete Euler's entation * Eqs. (1,16) and 9,401 utility 2 ha2(n) of the dt dt face source 11 routhing e update of displacement do 154-3.6x-2 do 15 j. 3.nz. 2 nn=u2(i,j) 02(i,j) = 0(i,j)nti.jj=nn 15 1010300 • e write snapshot if(mod)mnsprospf0_there print "l'write snapshut'.n write, 10.* instazalxalz do i=t.ux write 10.* fru2ritjt.? - 1.az1 real do **cudif** e load seismogram at p. 25 do istany seisti, io - 6210251 not do Iu. 10.00530.00 e write seismogram. write, 15,* invustepidx.du dour Lux write, 15, *0 seisticitej (denstep) real do close(10). rlose(15) stope P1.4 e WAVELET subconting wavelet (f.O. my.dt) dimension (CLW) ** For (2.233) ** pi=3.14159265 ան շորքի r0=6./(5.16) Dec. 0.5 %h nw = 2.710 / dtdour 1.1 w $i = (n - 1)^{-1} di$ J) r 40. $f(a) \rightarrow xp(-Dw^*Dw^*D^*D)^*V$ ("cus'wb"D) end do irtur end

9.9.3 Finite-differences code for the SH-wave and Maxwell's equations

The following Furthan program can be used to solve the SH wave and Maxwell's equations in italion ogeneous media. The SH-wave differential equations for isotropic media, based on Maxwell's viscodastic model, can be rewritten from equations 18.266(8.28) or the particle-velocity, stress formulation

$$\partial_t v_t = \frac{\gamma}{p} \left(\partial_t \sigma_{1t} + \partial_t \sigma_{2s} + f_t \right)
\partial_t \sigma_{1t} = p \left(\partial_t v_t - \frac{1}{q} \sigma_{2t} \right)
\partial_t \sigma_{1t} = p \left(\partial_t v_t - \frac{1}{q} \sigma_{2t} \right).$$
(9.16)

where $c_0 = c_0 = \mu$ is the shear modulus $z_0 = z_0 = 1/\eta$, and μ is the shear viscosity. On the other hand, the TM Maxwell's equations are

$$\partial_t H_2 = \frac{1}{\mu} [\partial_t F_s + \partial_0 (-F_s) - M_2]$$

$$\partial_t - E_s \left(-\frac{1}{2} [\partial_1 H_2 - \sigma (-E_s)] - (9.15) \right)$$

$$\partial_t E_1 = \frac{1}{2} (\partial_t H_2 - \sigma E_1) ,$$

where $c_{12} = \epsilon_{13} = \epsilon$ is the deficitive primitivity, and $\phi_{13} = \phi_{13} = \phi$ is the conductivity. Legislations (9.13) and (9.14) are mathematically analogous for the following correspondence

The program is written by using the field variables and material properties of the SR-wave equation. Maxwell's equation can easily be solved by using the correspondence 0.451. The time discretization has fourth-order accuracy, and it is based on the Runge-Kutta approximation (9,17), while the spatial derivatives are computed with the fourthinduc staggined operator (9,31). In terms of the staggered operators, repathors (9,13) because

$$\begin{split} \partial_{0} \psi_{i} &= \frac{1}{p} \left(D_{i} |\sigma_{ij} + D_{j} |\sigma_{j1} + f_{j} \right) \\ \partial_{0} \sigma_{ij} &= p \left(D_{i}^{*} \psi_{i} - \frac{1}{\eta} |\sigma_{jj} \right) \\ \partial_{0} \sigma_{ij} &= p \left(D_{i} |\psi_{j} - \frac{1}{\eta} |\sigma_{ij} \right) , \end{split}$$

$$(9.16)$$

where D_{-} and D_{-} represent staggered spatial-derivative operators of order 4. The differrat signs imply a shift of half the grid size to obtain the acceleration at the same points of the particle velocity (Carcione, 1998) of The averaging of the material properties is performed as indicated in Unicione (1999) (.

The program solves the isotropic version of the reflection transmission problem illustrated in Section 6.1. The mesh has 120×120 points and a grid spacing of 10 m. A supplied of the particle velocity is at 250 ms is shown in Figure 9.3.



Figure 9.4: Suspear of the SR coverparate velocity corresponding in the reflection transmission problem studied in Section 0.1. The star industry the location of the source.

```
e Isotropic viscodastic SH-wave propagation
Sortion 1.4. : Philosware analysis
Section 1.5.3 : Differential equations
Section 1.6 . : Analytical solution
Section 6.1 - Reflection-transmission problem
Socion 9.2.3 : Time integrations
Sortion 9.3.1 : Spanal derivatives
c
e TM Maxwell's equations
Section S 2.1 : Plano-wave analysis
Section 8.2 . : Differential equations
Sortion 1.6 . : Analytical solution
Section 8.4 - : Reflection-transmission problem
Socion 9.2.3 : Time integration
Sortion 9.3.1 : Spatial domatives
c
e Acorstic-decromagnetic analogy
17 Eq. (9,15) 17
e v2/c 1/112
c $23=$32 · · · ·
                ET
est2 < 124
care of a surveyse of the permittivity
e those de magnetic permability
clera succeiverse of the conductivity
c
```

```
    Floritomognetic example cumits; cm and ust

J.

    Velocity: 20 cm/ns (light velocity = 30 cm/ns)

    Dielectric permittyvity (vacuum) 8.85 f.e. 12
    Magnetic permeability (vacuum): 1 pi 1.0 - 23

    Conductivity (0.001715) - 21 (0.001 S/m).

c Frequency | 0.2 (200 MHz)

    ib (10 cm)

    obv 10 (10 cm)

÷

    O(4.1) fittite-difference scheme

a:
parameter 10xt = 120, nzt = 120 (astop) = 5001

    tielú variables

a v2: porticle velocity
4 512 and $23; stress commonents.
       dimension v2(axraz)(s12)axraz(1.832)axraz()
       dram story v2al ost nzt ) s12a not nzt (s32a(nxt.521)
       dimension v2((instanzt)).si2((instanzt)).si2((instanzt))
e material properties
e nue shear modulus

    rime density

e eta: Maxwell viscosity
       dama store muticatural) a loci estinat (reportstruat) -
       rammon/fil-weights/witx2.21.72
       1057-1091
÷
       dimmission abr301
       dimension, frustept (
upon(10.00+SNAP))
         dx = 10.
         dz 10.
         41 111111
         ns = 120
         67-120
         ustep: 250
         pi=3.14159265
cosmpolious every hopotepis
        psto ristep

    source location and control frequency

       ix=60
       17 60
        freq - 25

    MODEL

       do i linx
       d_0 = 1.92
e oppes layer
** Laps 106 221-(6.24)
       (ba)(j) 2000
        matij(=9.68e+9

    quality factor at source central frequency

<sup>11</sup> Eqs. (8.42) and (8.57) 14
       O=5.
       etsister Q"ungissistification
a lower lover
```

```
if(rge 80) then
         rho(1)=2500
         trup(1))=19.6c--9
         Q 10.
         rtati.jt=Q*enti.jt; pi*inp
         rudif
         cul do
         real do
e
         do r. Lus
         do j=1.0z
\sqrt{2}(i,j)=0.
         st2fi.j)=0.
         -320 ĴU -0
         nul du
         end do
         do o Lostept
         f(a) = 0.

    addo

<u>،</u>
calsorbity parameters
         r 0.99
         nab 12
         iko i⇔Luah
         abui: r**i
abui: 1
         nul du
e somer's wavelet
14 Eq. (20233) 14
         rall wavelet(f.forg.mv.dt)
c indusdifferences weights
17 Eq. (9.24) 17
         \begin{array}{l} x_1 = 0, & (S^{-1}(x)) \\ x_2 = -1, & (2^{-1}(x)) \\ x_1 = 0, & (8^{-1}(x)) \\ x_1 = 0, & (8^{-1}(x)) \end{array}
         72 = -(1.7)(24.7)(2)
e TIME STEPPING
         do 10 n. Unstep
         if(and)a.10.eq.0 (print ) at
c apply absorbaic boundaries.
e horizontal strips
         do 11 j=1.uali
         2 j -2
3 n/ j -1
         sab = ab(nab + 1 + j)
         do II n. dans. 2
         v2(i_{0}^{2}) = v2(i_{1}^{2})^{*}sah
v2(i_{1}^{2}) = v2(i_{1}^{2})^{*}sah
11
e venual strips
         do 12 i - Luali
         i2=i-2
         i3 nx i 1
         sale abjuates 1 (r)
         do 12 j=3.02+2
         v2(i2.j) v2(i2.j) sab
v2(i3.j) v2(i3.) sab
12
```

```
4
c Russe-Kutta zurthal.
  ** Eq. (9.17) **
                         do 13 i - Lux
                         do 13 j=1 az
                          \sqrt{2t}(\mathbf{i}_{ij}) = \sqrt{2t}(\mathbf{i}_{ij})
                         $120(i.); $12(i.j);
$320(i.); $320(j);
                          \sqrt{2}a(i,j) = \sqrt{2}(i,j)
                         s12((i)) s12((j))
s32a((i)) s32((j))
13
                          continger
 ** Eq. (9.4) **
                          call H1x2a s12a/s32a/mitchildta nx nz t

    D1

                          da Di Env
                          do 14 j=1.67
                         \begin{array}{l} \chi_{2t}(i,j) = \chi_{2t}(i,j) + (i_1^{-1}\chi_{2t}^{-1}(i,j) + 0, \\ \pm 12t(i,j) = \pm 12t(i,j) + (j_1^{-1} \pm 12t(i,j)) + 0 \\ \pm 32t(i,j) = \pm 32t(i,j) + (0 + 532t(i,j)) + 0. \end{array}
                          v2ab.p / v2b.p (0.5 d) v2adia.
                          stadin striggenfahrstraffi
                          s32a(i,j) = s32(i,j) - 0.5^{1}d(1s32a(i,j))
ι.
                          continue
                          if to locing to here
                          \sqrt{2}t \cdot (x, iz) = \sqrt{2}t \cdot (x, iz) + dt^2 f(u)/6.
                          v2ahstiz) v2ahstizi 0.5 d0 ftati
                         endif
1
                          call 11y2a/s12a/s32a/murboora/axe/zi/i
÷

    1)2

                          do 15 i - Lux
                          da tā jo 1 nz
                          v2t(i,j) = v2t(i,j) + (it) v2a(i,j)/3.
                         s120(12) s120(1); ol ($12a(1))/a,
s320(12) s320(12); ol ($32a(12))/3
                          x_{2a}(i,j) = x_{2}(i,j) + 0.5^{a}(i)^{a}x_{2a}(i,j)
                         §12a(i), §12(i)) (0.5*d(*s12a)(i)
§32a(i), §32(i)) (0.5*d(*s12a)(i))
 65
                          cuation
                          difficiency then
                          (2) (waz) w20 (waz) of 200 (173)
                          \sqrt{2}a((x,iz) + \sqrt{2}a((x,iz) + 0.5)d(2f(a+1))
                          endil
÷
                          call H1v2a s12a.s32a.cmu huota axiazt
4
c D3
                          do Di i=Lox
                          do Diji Laz
                          (2)(a) (2)(a) of (2a)(p/3)
                         \begin{array}{l} s(2(i,j) = s(2(i,j) + d(1s)2a(i,j), 3, \\ s(2(i,j) = s(2(i,j) + d(1s)2a(i,j), 3, \\ s(2(i,j) + s(2(i,j)), 3, \\ s(2(i,j) + s(2(i,j)), 3, \\ s(2(i,j)) = s(2(i,j), 3, \\ s(2(i,j)) = s(2(i,j)), \\ s
                          (250.0) v20.0) dt v250.0
```

```
st2a(ij) st2(ij) (d0'st2a(ij)
       s32.nij(=s32(ij)--d)*s32nij)
Ro.
       routinte
       if(n.lenw) then
       v2((iv.iz)=v2)(iv.iz)~d(*f(u--1)/3)
       v2atixiz(=v2atixiz) -alt (f, a-.1)
       eralif
• *
       call Hty2as12as32a.muthoeta.tiv.tz)
r,
e D1
       do 175° Lux
       do 17 j. Luz
       v2((ij)=v2((ij)+d)*v5((ij)/6.
       s12((i,j)=s12(j),j)~d(*s12a(i,j)/).
       s3200.jp s3200.p+o0*s32a0.jp/6.
121
       routinue
       if(n.le.av.) then
       v20fix.iz) v2(fix.iz) off*fun-21/6.
       rralif.
e
       do 185° Luis
       do 15 j. Luz
       52(i,j)=52((i,j))
       st20.j) st20.0
       <32(i) - <32(i))</p>
15
       rontinne
r,
e write snapshot
       if(mod)mnspireq.01 then
       print "Cwate supplier in
       write 10,2 juxaizaly dz
       do i=Lux
       write, 10.* (iv2(i,j), j=1,nz)
       real dou
       rudif
Ju.
       1046300
       rdose(10)
       stup
       r tal
r'
submittine Hty2.s12.s32.mu.chi.eta.ux.nz)
       dimension v2(ns.nz)/s12(ns.nz)/s32(ns.nz)
       dimension muturs, or the out (iz) eta hy arti
       dimension v2acus azt
       common. Id-weights/x1.x2.21.22
       real ang
••
       do Li Lux
       46.14, 1.02
       v2ati.j1=0.
      continue
L
14 Eq. (9.11 **
ı.
       do 23, 3 us. 2
```

```
do 2 j - 3.nz - 2
       \sqrt{2}a(i,j) = \sqrt{2}(i,j)
a momentum conservation.
** Fqs. (9.24) and (9.46) **
i \in 3/2 \rightarrow 1 1
(i) 172 (b);
e i < 1/2 (3 t < 1
i = 3/2 \rightarrow 3-2
       ds3 - 71*18325 p+10 - 8320.p0 + 72*08320.p+21 - 8320.p+1+
       ds(i=x(1)s(12)(i-1)i) = s(2(5)i)(i-x(2^{*})s(12)(1-2)i) = s(2(i-1)i))
e soch tation
** Eq. (9) 166 **
       v2(i,)(=)(is i+ ds6)/(ho(i,j))
-2
      continue
ŧ.
       do 3 i - 3.68 - 2
       do 3 j 3 nz 2
e strains and stresses
11 Las (19/24), web (9,46), and (9,16), 11
ri 3/2 on 2
i \in 1/2 \rightarrow i = 1
+ i + 1/2 -> 5
+ i = 3/2 = - i = 1
       el vl (v2a).j) v2a(i,j=1)) v2 (v2a(i,j=1) v2a(i,s=2))
       eli xt*tx2a(cj) x2a(i Lj)(x2*tx2a(i Lj) x2a(i 2j))
       s32(ij)=mu(ij)*(e1/s32(ij)/etatij))
       <12(c) mm(c)*06 s12(c)/eb(c)11
3
     continue
4
       101/1011
       end
+ WAVELET
       subnorrine wavelet (fill-awal))
       dimension from (
<sup>14</sup> Eq. (2.233) <sup>14</sup>
       [6] 344159265
       wb=2.1pi1fb
       01 - 6, 35 * flo
       Dw-051wb
       EW 2.110.dr
       data Luns
       1-11-11-01
       D I-III
       Interespit Dw*Dw*D*D*D/1.(Tros(wh*D)
       cuel do
       iotana.
       end
```

9.9.4 Pseudospectral Fourier Method

The Fermici PS method is a collocation technique in which a continuum function u(x) is approximated by a truncated series

$$u_{N}(x) \mapsto \sum_{t=0}^{N} \hat{\sigma}_{t} \phi_{t}(x)$$
 (9.47)

of known expansion functions $\phi_{i,j}$ where h the spectral texpansion coefficients are chosen such that the approximate solution $\phi_{i,j}$ considers with the solution u(x) at a discrete set $\phi_{i,j}$, $v_{i,j}$, $v_{i,j}$, $v_{i,j}$ of sampling or collocation points,

$$u_N(x_j) = u(x_j), \quad j = 0, ..., N - 1.$$
 (9.18)

The collocation points are defined by optidistant sampling points.

$$x_f \in dt_{c}$$
 (9.49)

where do is the grid spacing. The expansion functions are defined by

with

$$h_r = \frac{2\pi i}{Ndr}, \quad r = 0, ..., N - 1$$
 (9.51)

heighthe wavenumber. Thus

$$\phi_i(x_i) = \exp(2\pi i x_i / N).$$
 (9.52)

Since the functions come terms by the Foguer PS method is appropriate for problems with periodic boundary conditions – for example, a wave which exits the grid or nor side, and corners it on the opposite side. The coefficients \hat{y}_i are implicitly defined by

$$u(x_j) = \sum_{\tau=0}^{N-1} h_t \exp\{2, ix_j/N\} \quad j = 0, ... N - 1.$$
 (9.53)

The sequence of $u(x_i)$ is the inverse discrete Franici transform of the sequence of u_i . This set of equations is equivalent to

$$\dot{u}_{i} = \frac{1}{N} \sum_{j=0}^{N-1} u(r_{j}) \exp(2\pi i r_{j} / N) \qquad r = 0, \dots, N-1$$
(9.54)

The computation of differential operators by the Former in the d-conveniently reduces to a set of multiplications of the different coefficients δ_{t} , with factors ik, since

$$\partial_{t}(q_{1}|r) = \partial_{q}(q_{1}|r),$$
 (9.55)

so that

$$\partial_t u_{\infty}(x) = \sum_{i=1}^{\infty} (k, k, \phi_i(x)).$$
(9.56)

The spectral coefficients \dot{a}_i are computed by the East Fourier Transform (FF1). Examples of efficient algorithms are the mixed-adax FFT (Temperton, 1983) and the prime factor F1 U themperton, 1988_{ij} . The steps of the calculation of the first partial derivative are as follows:

$$(a(x_i)) \mapsto FFT + (a_i) = (A_i, a_i) \mapsto FFT (1 + (bar(x_i)))$$

$$(9.57)$$

The method is infinitely accurate up to the Nequist wavenumber, which corresponds to a spatial wavelength of two grid points. This means that if the source is band-limited, the absorbtion is free of imprecial dispersion provided that the grid spating is chosen $dr \leq c_{eq}/(2f_{max})$ with f_{max} hence the ent off frequency of the source and c_{eq} the continuum phase velocity in the mesh. The wavenumber can be expressed in the more convenient frace.

$$k_{\nu} = \begin{cases} \langle \langle k_{Neg} \nu | \text{ for } \nu = 0, \dots, \rangle \rangle, \\ - \frac{2}{N} k_{Neg} \langle N - \nu \rangle | \text{ for } \nu = \frac{N}{T} + 1, \dots, N - 1, \end{cases}$$
(9.58)

where for N odd, N/2 represents truncation to the closest integer, and $k_{Ne1} = z_T dx$ is the Nyquist wavenumber. For example, $N \simeq 5$ has wavenumbers

$$\left(0, \frac{2}{5}, \frac{1}{5}, -\frac{4}{5}, -\frac{2}{5}\right) k_{\rm Nup}, \tag{9.59}$$

and X = 6 has wavenumbers

$$\left(0, \frac{2}{6}, \frac{1}{6}, 1, -\frac{1}{6}, -\frac{2}{6}\right) k_{NEP}$$
 (9.60)

We see that when X is even the wavenumber operator contains the Nyipeist wavenamber beace, k_i is an odd function in the particulic sense only for N odd, since $k_n = -k_N$. When N is even, the Nyipest wavenumber breaks the antisymmetry

We shall see any that when computing hist indiciderivatives, the number of grid points must be odd. Indeed, it is well known that when add t is real, its continuous Former transform eqk) is Hermitian treats real part is even and its imaginary part is add (Brarewell, 1965, p. 16), and vice virsa, if *atk* its Hermitian its inverse transform is real. Similar properties hold for discrete Former transform, Indeed, for A sold.

Theo,

$$\Im h(k) = i \operatorname{add} - i \operatorname{yea}$$

$$(9.62)$$

is also Hermitian, and $\theta(y)$ is real. Conversely, when X is even $M^{1}(qk)$ is not Hermitian by anse of the Nyquist wavenumber.

We now give some momencial tricks when using the UET for computing partial derivatives.

1. It is possible to compute the derivatives of two real functions $\partial_t f$ and $\partial_t g$ by two complex FFT's write following wave put f into the real part and g into the imaginary part and compute the direct FFT at k_i .

$$\sum_{ij} [f_{ij}^{i} + f_{ij}^{i} + i(y_{ij}^{i} + y_{ij}^{in})] \cos \theta_{ji} = i \sin \theta_{ji}), \qquad (9.63)$$

where summations go from 0 to N = 1. The functions have been split into even and oder parts α and σ , respectively), and θ_{α} is an althresiation of $k_{\alpha} x_{\alpha}$. Terms like $\sum f_{\alpha}^{\alpha} \cos \theta_{\beta}$ vanish some supernation of an odd function is zero – note that the cosine is even and the sine is odd. Then, equation (9.03) reduces to

$$\sum_{ij} f_{ij}^{ij} \cos \theta_{ji} - q_{jj}^{ij} \sin \theta_{jj} - \mathrm{i} \left(g_{ij}^{ij} \cos \theta_{ij} - f_{jj}^{ij} \sin \theta_{ij} \right), \qquad (9.64)$$

Now, multiply by 5%, and transform back to the space domain. At point 26, this gives

$$\sum_{\sigma} \sum_{i,j} \left[ik_i - f_i^{\sigma} \cos \theta_{\sigma} + g_j^{\sigma} \sin \theta_{\sigma} + i \left(g_j^{\sigma} \cos \theta_{\sigma} - f_j^{\sigma} \sin \theta_{\sigma} \right) \right] \left[\cos \theta_{\sigma} + i \sin \theta_{\sigma} \right], \quad (2.65)$$

Since many of the terms vanish, the result is

$$\sum_{i} \sum_{i} k_i \left(f_i^* \sin \theta_i, \cos \theta_0 + f_i^* \sin \theta_0 \cos \theta_0 \right) + \epsilon i k_i \left(g_i^* \sin \theta_i, \cos \theta_0 + g_i^* \sin \theta_0 \cos \theta_0 \right).$$
(9.69)

By applying the same arguments to each single function, it can be easily shown that the real and imagency parts of 12.66; are the derivatives of T and u at v_i respectively.

It is possible to compute two FEU's from one complex FFT, where, by real and imaginary FFT's, we must

$$f_R = \sum_{ij} f_j \cos \theta_{jj}, \qquad (966)$$

and

$$\hat{f}_{l} = \sum_{n} f_{i} \sin \theta_{ii}, \qquad (9.68)$$

As before, we take a complex FFT of F = f - iq, which gives

$$F = f_E + \hat{y}_I + i\left(\hat{y}_E - f_I\right), \qquad (9.69)$$

Since f and q are real functions, then transforms are Hermitian, hence,

$$\hat{f}_{\theta}(k) = \hat{f}_{\theta}(\neg k)$$
 $\hat{q}_{\theta}(k) = \hat{q}_{\theta}(\neg k)$ (9.70)

$$\hat{f}_{i}(t) = \hat{f}_{i}(-k) = \hat{g}_{i}(k) = \hat{g}_{i}(-k)$$
(9.71)

Using these properties, we note that

$$\frac{1}{2} \left[F_{R}(-k) + \tilde{F}_{R}(k) \right] = f_{R}, \qquad (9.52)$$

and

$$\frac{1}{2} \left[F_I(-k) - F_I(k) \right] = \hat{q}_R.$$
(9.73)

i.e. the two desired real transforms,

9.9.5 Pseudospectral Chebyshev Method

Where a function is not periodic, the Function method is not conversion for implementing free-surface and rigid boundary conditions. The reason is that the basis functions of the Fourier expansion are periodic. Satisfactory results are obtained with orthogonal polynomials, such as Chebyshev in Legendre pulynomials. We consider the Chebyshev basis, because, as we shall see later, the derivative can be computed by using the ITT routine. The function $u(z_0, -1) \le z \le 1$ is expanded into Chebyshev polynomials $E(\chi)$ as

$$u(s_0) = \sum_{r=0}^{\infty} u_r T_r(s_0)$$
 (9.74)

where

$$T_{n,(s_{1})} = cos(\theta_{1})$$
 (9.75)

with

$$\zeta_j = \cos \theta_j, \qquad \theta_i = \frac{\beta_i}{N}, \qquad j = 0, \dots, N.$$
 (9.76)

denoting the Gauss-Lobarto collocation points $\sum_{i=1}^{n}$ halves the first and last terms. The partial derivative of order q is given by

$$\frac{\partial(a_{n,k})}{\partial z^{k}} = \sum_{n=0}^{N} a_{n}^{(k)} T_{n,k} \,, \qquad (9.77)$$

(Gorflieb and Oiszag, 1977, p. 117), where

$$c_{n-1}a_n^{(0)} = a_{n-1}^{(0)} = 2m e^{\theta} \quad , \qquad 0 > 1.$$
 (9.78)

with $a_0 = 2$, $a_0 = 1$ (n > 0). Bence defining $a_0 = a_0^2$ and $b_1 = a_0^2$, the first-order derivative is regard to

$$\frac{\partial u}{\partial z} = \sum_{n=0}^{\infty} b_n T_n e_{\infty}$$
(9.79)

where

$$b_{n-1} \dots b_{n-1} + 2na_n$$
, $n \dots N, \dots, 2, b_N \dots , b_N \dots 0,$ (9.80)

We consider the domain $[0, z_{max}]$ and want to interpolate $\sigma(z)$ in this domain. The transformation

$$z_{i} \sim \frac{5ms}{2} (\zeta_{i} + 1)$$
 (9.81)

maps the domain [-(4, 4)] onto the physical domain $[0, ..., n_n]$. The Gauss habitto points have maximum spacing to the center of the numerical grid, with

$$dz_{\rm hav} = \frac{z_{\rm max}}{2} \left\{ \cos\left(\frac{N}{2}\frac{\pi}{N}\right) - \cos\left[\left(\frac{N}{2}+1\right)\frac{\pi}{N}\right] \right\} - \frac{z_{\rm max}}{2}\sin\left(\frac{\pi}{N}\right), \quad (9.82)$$

Note that $d_{smax} = \sin(\pi/N)$. In wave problems, we determine the maximum grid spacing according to the Nyquist catering, $d^{1/2} \approx c_{em}/(2f_{max})$. The spatial derivative is

$$\frac{\partial u}{\partial t} = \frac{\partial u}{\partial s} \frac{\partial s}{\partial s} = -\frac{2}{\pi s} \frac{\partial u}{\partial s} = -\frac{1}{d \gamma_{\max}} \sin\left(\frac{\varepsilon}{N}\right) \frac{\partial u}{\partial s}.$$
(9.83)

This is a transformation from the physical domain to the Chebyshev domain.

Now, let us see how to calculate dr/d_{s} . The expansion of $r' \in t$ and its coefficients can be written as

$$u(s_0) = \sum_{n=0}^{N-1} a_n \cos\left(\frac{i(nj)}{N}\right)$$
(9.84)

and

$$u_{i} = \frac{2}{\chi} \sum_{j=0}^{\chi_{i}} u(\zeta_{i}) \cos\left(\frac{\pi nj}{\chi}\right), \qquad (9.85)$$

The coefficients α_n can be evaluated by using a UTT routine. Let us define $N^2 = 2N$ and $w(z_0) = 0$ for $y = 1 + N^2/2$, $z \in N^2 - 1$. Then

$$a_{0} = \frac{1}{N'} \sum_{j=0}^{N} a_{0,k_{1}} \cos\left(\frac{2^{-}a_{j}}{N'}\right).$$
(9.86)

is a real fourier transform, that can be calculated by complex FF I's as described in the previous section. Afterwards, we get the b_0 's from the a_0 's by using the recursion equation (9.78) and again, the calculation of (9.79) is carried out with a real Fornici transform. However, the Uhebyshev method, as presented so far, is impractical, because the grad spacing at the extremes of the dimain is very fine. When the number of grid prints is doubled, the grid spacing decreases by a factor of two. Hence, when solving the problem with an explicit two marching scheme, the conventional Chebyshev differential operator requires time steps of the order $O(N^{-1})$. A new algorithm diveloped by Koshiff and Fall Free (1979), based on a emodiant transformation, allows time steps of index $O(N^{-1})$, which are those required also by the Fourier method. The new N sampling points are defined by

$$z_{i} = z_{max} \begin{bmatrix} q(z_{i}) - q(-1) \\ g(1) - q(-1) \end{bmatrix}, \quad j \in [0, \dots, N],$$
(9.87)

where $q_{i,k}^{*}$ is a grid stretching function that stretches the very four Chebyshev grid mean the boundary in order to have a meaning grid size of the order $O(N^{-1})$, thus respiring a less severe stability condition. A suitable stretching function is

$$g(\zeta) = -\frac{1}{\sqrt{\mu}} \arctan\left(\frac{2\rho\zeta + q}{\sqrt{q^2 - 4\rho}}\right). \tag{2.88}$$

where $p = 0.5\alpha^{-1}(0)^2 + 1$ (1, and $q = 0.5\alpha^{-1}(3)^2 - 1$). Since

$$\frac{dg}{d_{\rm s}} \simeq \frac{1}{\sqrt{1 - q_{\rm s}^2 - \rho_{\rm s}^2}},\tag{9.89}$$

it can be seen that the amount of grid stretching at $\zeta = -1$ is $dy/d\zeta = \alpha$, and that the stretching at $z = \zeta$ is $dy/d\zeta = \alpha\beta$. The spatial derivative is

$$\frac{\partial u}{\partial z} = \frac{\partial u}{\partial \zeta} \frac{\partial z}{\partial z} = \left[\frac{g(1) - g(-1)}{z_{\max}} \right] \sqrt{1 + g_{\zeta}} + p_{\zeta} \frac{\partial u}{\partial \zeta}, \tag{9.260}$$

he many cases, we need to sample the function at the optidistant points.

$$z_0^{(i)} = \alpha N_i$$
 $\Delta z = \frac{5 m s}{N}$ (9.90)

The corresponding prants in the Chebyshev domain are from (2.8%)

$$z_{1}^{\prime} = y - \left\{ \begin{bmatrix} g(1) + q(-1) \\ -\lambda_{\max} \end{bmatrix} :_{0} + q(-1) \right\},$$
(9.92)

The values of the function at equidistant points in the physical space are given by

$$m_{ij}^{(i)} = w_{ij}^{(i)},$$
 (9993)

To obtain these values, we compute the spectral coefficients u_{ij} of $u(z_i)$, and then

$$u\left(\gamma_{r}^{\prime}\right) = \sum_{i=1}^{N} u_{i} T_{r}\left(\varphi_{i}^{\prime}\right), \qquad (9.94)$$

Examinations

The careful study of the previse answers to the following questions will prove helpful in preparing for examination of the subjects developed in this book. Numbers or parentheses refer to payes on which particlent information can be finand.

- Describe the contants crystal symmetries of geological systems. How many independent elasticity constants are there in each case? Provide the interpretation in terms of fractures, cracks and fine layering (2.346).
- Consider a transversely isotropic modulin whose symmetry axis is horizontal and makes an arg'e θ with the x-axis. If equivelences the elasticity constants in the principal system and c_{ij} are the elasticity constants in the system of coordinates express c_{ij} in terms of the c₁ is (9,10).
- Which is the relation between the energy-velocity vector and the slowness vector? (16)(1113,151,343)
- Discuss the conditions by which the group velocity is equal to the energy velocity (19)1570.
- Discuss the relation between slowness surface and charge-velocity vector and slowness vector and ray surface (24,157–159).
- Give the features of waves in planes of mirror symmetry (7.12-11.168-169).
- What is a cusp? When is represent? Which type of waves have cusps? (22:23:223)
- What is the shape of the slowness curve for SB waves propagating via plane of symmetry? for the group velocity curve? (12.21.157).
- Consider have bygging in the long-wavelength limit. Explain the physics and comment on the location of the cusps (25-29.132.393-371).
- 10. Consistent $r_{constraints}$ in a long-wavelength requivalent medium of a layered medium" (28)
- What is anomalous pularization? Explain (29)37).
- Explain why the pubrizations are orthogonal in anisotrupic clastic media? (11.15)
- Describe the method to obtain the best isotropic approximation of a general anisotropic medium (38-40)
- Define critical angle (14/49,1955)
- Is the strain (dicleratio energy unique in anelastic telectromagnetic) media? [52] 51.70.433-340).
- Is the relaxation tersor symmetric? (55)

- How are strain and dissipated energies related to complex modulus? (57)
- Explain the physical meaning of the Kramers-Krong relations Express them in encentringal form 6.8-60.373-3737
- 19. List the properties of the relaxation function and complex modulus (60).
- How are the energy and phase velocities related in 1-D anelastic media? (58).
- Explain the concept of centrovelocity (88-92).
- What is a memory (hidden) variable? Explain (92-96)124-125.162-166.338.358-359.4104.
- Explain the properties of the Zener model: relaxation function and phase velocity and quality factor versus frequency. How do you obtain a nearly constant-Qmedium? (71-77.89-82).
- Is there a period constant-Q model. What is the corresponding equation of motion? Comment on the phase velocity versus frequency (83-87)
- Is the energy vehicity equal to the phase velocity in isotropic visioclastic media? (113)
- How many Bacleigh waves may propagate in a viscoelastic andomi? What can you say about the propagation velocity? (110) 1210.
- Given the slowness vector and the time-averaged energy-flow vector, is it possible to compute the time-averaged energy density? (1112:152,313)
- 28 Consider hand/s problem; a dilatation source (or explosion for metanice), and a receiver measuring the vertical component of the particle velocity. Discuss the recuprocal experiment (178).
- What is the Rayleigh weather? (230-231)
- Explain the properties of an inhomogeneous budy wave, and the physics involved in wave propagation in an anclastic occur bottom (110-151).
- Which requirements are necessary to have forbidden bands? (161-162).
- Describe the priorization of inhomogeneous holy waves (1.4).
- 33. What is the relation among the phase, group, energy and envelope velocities in the following theologies, i) isotropic elastic, n_i isotropic weekstic, m_i arisotropic elastic and iveranisatropic inelastic. Consider the distinction between homogeneous and inhomogeneous waves (19,20.04),155-157).
- What happens with the Brewster and entited angles in viscoelastic medial (124,195-(99))
- How many relaxation functions are there in isotropic media? How many, at most, in trichnic media? (112-115).
- Explain the physics of the slow wave. When is it present as a wave, and why? (274-278).
- Describe Lord Kelvin's approach for attisettopic clastic media and its extension to describe viscoelastic behavior (142-141.316-320)

- How do you compute the Green function for viscoelastic media from the elastic Green's function? (126-129,168-469)
- What is the direction of the attoination certor with respect to the interface when the incidence medium is classic? (225).
- 10. What are the interference Brixes? (201-204.223).
- Describe the three experiments used to obtain the expression of the ponodastic moduli (237-240).
- Can the transmitted cay be parallel to the interface when the incidence medium is loss'ess and the transmission medium is lossy? (192).
- Describe the boundary conditions of a fracture? Explain the physics (129-138)
- What is the nature of Biot's attemption mechanism? (262-263).
- Common on the stiffness of Biot's differential equations and its physical mason (389).
- What are contining, hydrostatic and pore pressures? When is there overpressure? (242-244).
- What are the nain causes of overpressure? Commut on its effects on the acoustic and transport properties of the rock? (212-246).
- Explane Saell's law in viscoelastic media (114-115).
- Explane the correspondence principle (116).
- Discuss the brendary conditions at interfaces separating proofs media (284-289/299-303).
- Bow many wave modes are there in an axisotropic porous medium? Does the number present depend on frequency" (318-320)
- 52. Represent the Burgers viscoelastic model, obtain its croep function and thus of the Maxwell, Kelvin-Veig) and Zetter models as limiting cases (75-79).
- 53. Describe the nature of the mesoscope base mechanism (289-295).
- 54 How more surface waves propagate on the surface of a porous medium, with openpute and scaled-pore homeony conditions? (209-303).
- Fstablish the mathematical analogy between Maxwell's equations and the elastic wave equation (324-329).
- 56 Industry herelectromagnetic analogue of the elastic kinetic and strain energies (329).
- 57. Write the Delive chelicitic permittivity by using the analogy with the Zener relaxation function (337/340).
- Indicate the acoustic-electromagnetic analogy for the boundary condition at an interface, and the analogy between TM and 145 waves in postropic media (342.351-352).
- Find a mathematical analogy betweet, the TM equations and a modified equation for sound waves (350-351).

- Explain how George Green obtained breshel's reflection coefficient from the explations describing wave propagation in an elastic medium (352-355).
- Indicate how the analogy can be used in 4-D space. Design the electromagnetic slowness curves across the three symmetry planes of an orthorhombic coslugio (356-363).
- Write the accustic-electronagnetic analogy for Backus averaging of isotropic layers (302) 3715
- List other possible mathematical analogies between acoustic and electromagnetic waves (379-378)
- 64. Write the diffusion equation in terms of the electric vector (380).
- 65. What is a direct method in numerical modeling of wave propagation? (385).
- 66. How do you plan a numerical modeling simulation? (1801)

Chronology of main discoveries

2. I have assumed, as applicable to the huminferrors (the increment) the known equations of motion of an elastic median, such as an elastic solid. These equations contain her arbitrary constants, depending upon the nature of the medians. The angument which Green has employed to show show that the huminferrors (ther must be regarded as secondary operations which constraints layer by the metal of the median of the real operations of the second start of the metal operation of the second start of the metal operation of the metal operation of the second start of the metal operation of the metal operation of the second start of the second start of the metal operation of the metal operation of the metal operation of metal operations of the metal operation of metal operation of the metal operation of metal operations of the metal operation of metal operations of the metal operation of metal operations of the metal operation of metal operations of the metal operation of metal operations of the metal operation of metal operations of the metal operation of metal operations of the metal operation operation operation operation of metal operations of the metal operation operation of metal operations of the metal operation operation operation operation operation operation operations of the metal operation operation operation operation operation operations operatio

George Gabriel Stokes (Stokes, 1856).

As early as the 17th century it was known that light waves and acoustic waves are of a similar nature. Honor balleved light to be a vibratory displacement of a medium (for ether), through which it propagates at finite speed, have, in the 19th century. Maywell and Lind Kelvin made extensive use of physical and mathematical analogues to study wave phenomena in acceptics and electromagnetism. In many cases, this formal analogy becomes a complete mathematical equivalence such that the problems in both fields can be sulved by nearly the same analytical for numerical in thoulding. Green (1842) made the analogy between clustic waves in an incompressible solid (the ether) and light waves. One of the most remarkable analogies is the equivalence between electric displacements and elastic displacements (hooke's law used by Maxwell to obtain his famous cheeromagnetic equivales. Therefore, the study of acceptie wave propagation and field propagation are intimately related, and this fact is reflected in the comes of scientific research.

For task of describing the principal achievements in the field of wave propagation is a difficult one since many important scientists have been involved in the subject, contributing from difficient fields of research. Dates reveal connections and parallels, they famish us with a basis for comparisons, which make historical studies meaningful and exciting. The following characlegical table catends to give a basis glumpse of "evolution" and "causes and results" of the main scientific developments and ideas .

Senses - Cajore (1920). Love 1941. Asinow (1972). Caldstrate (1977). Ben Menalama and Single (1981): Carmon and Doctrowsky (1981): Process (1981): Reviewley (1964): the web sinework harteneous can buyer (as a apping and www.catagy.rag/light/fiberes and the web sine of the linewessity of St. Andrees, Scotland owwer history.cas st. analogs as uk/history.)

600 BC, ca.	Thates of Milerus discovers that amber (elektron in Greek) rubbed with hit
1	attracts light males.
istrative, ra	instructional and the second sec
	The length of vibuling strings to the pilot
326 BC . (a.	Eachd describes the law of reflection in $las Oplica.$
fill, ca.	Hermi writes his Cologdrica, when the states that high gays travel with
	indiante velocity.
1391. 641	Proletzy measures angles of the identice and refraction, and attanges them
	in tables. He found these angles to be proportional (sight-angle
	approximation)
990, c.i.	al Hay fair writes his Optics. He shows that Pfulency was more and
1.1.	refers for the first time to the "camina discura".
1210. ra.	Grossereste withes the Valuar Locotran and the Irele.
1268, Ga.	Bacon Willes, The Communic Submitteen and The Communic Mathematicae .
	The attributes the rainbow to the reflection of single from single radid ops.
1369, ca.	Petrus Pereguncis writes Apelaida de Magia fe
1270. ra	John Peckham [dud] 1299] writes the treatise in optics Perspectica Communi-
1250. ra.	Witele writes <i>Despectivenia</i> , <i>Educ</i> , where he interprets the rambow as
	reflection and other for of light.
1301.1.4	Diction of Fierberg gives the first accurate explanation of the randow.
1.980	Loonardo da Vinet makes the analogy between BLID waves and sound.
1558	Defa Porta publishes Mayor Saturatis, where he analyzes anguetism,
1560 га	Magnalycijs writes Phalisina de Fanone al Lindre, abagi pladoijjetry
1.581	V. Galder (Galders father) studies sound waves and vibrating strings
1600	Gilbert writes De Abigmete, and shows that the Earth is a magnet.
1008	hipparishey constituets a telescope with a converging objective lens and a
	(i) write ingression of the second s
1611	De Donnas explains the decomposition of colors of the minimum and the fides.
11.11	weptic pitch-liss his Draphical where he presents in empirical expression of
	the law of refraction. The discovers total internal reflection
11i20. (a.	San II obtains experimentally the law of reflaction, although the
1	discourry is attributed to flarmof.
10/29	Cabeo Writes Thatosophia Mogacteo, where he investigates electrical reptilsion.
10:05	Metschile Ollubsias ats <i>Inducate</i> (1900) (9), containing the first correct
	account of the violations of strings, and the first determination of the
	tregneney of an applied one (81 Hz).
11611	Descarses publishes from as have in first A Triaggraphy, without functioning Stell-
1638	Stableo publishis Philosofi Philosoficcioni Malchaluche, informa a dia Serve-
1.1.11	[5] I. B. G. HERBERG & OSCIDENCING THE VIOLATION OF DOTIES. [5] K. K. K. K. K. K. K. K. K. K. K. K. K.
1041	Kitcher Writes Maynes, Oc. 379 Wagnebra, Basolitants the first fiscal the
	Define cost to -market site is the second cost of the second site is a first second for the second site of the second site is a first second for the second site of the second site is a first second for the second site of the second site o
11.4%	DECEMBER (2016) ALL CARLES AND AND AND AND AND AND AND AND AND AND
11.45	FERREZ INFORMATING THE MEAN AND DEPENDENT AND A A DEPENDENT AND A DEPENDENT
Паб	Botenii and variane measure the smanl velocity in an angloblam 340 m/s .
1000	Boyle in monstrates from vacuum experiments that sound propagates in arr.

1660	Hooke states his law. Ut tracto so ers (The Power of any Spring is in the
	same proportion with the Tension Detector, published in 1678.
1661, 420,	Fermat demonstrates Snell's low using the propende of least type
1665	Those publishes his <i>Wringraphic</i> , where he proposes a theory of light as
	a transcense vibrational motion, making an analogy with water waves.
	(Marintte connectates the same law independently in (680.)
Etatici	Gritialdi discovers the phenomenou of diffraction (in <i>Physics Mathesis</i>) of <i>Commut</i>
LCCC	ng annonen. Nauran na farms bis avrarennuts an rha untura of light, arren stira urbita.
,	light into a band of colours - rol, orange voltor, greet blue, and violet
	In uses the corpuscitian assumption to explain the phenomenon.
116621	Batholius observes double refraction in behand spin.
1075	Newton is against the assumption that light is a vibration of the other.
1675	Boyle writes Experiments and Notes about the Mechanical Organ or Production of Electricity.
1675	Newton develops the fluory of junte differences and interpolation
	previously introduced by Harriot and Briggs
1675	Newton argues that double refraction rules out light bridg other waves
1676	Römer measures the snewl of light by studying Jupiter's refuses of its
	for larger sitellites
1678	Huweets treations for more relative of helt males. Controls to Foreity
1.11.	(inspires property to 100m). The second solution of the site of the site of the second solution.
	provide providence in the experimental and contractions in the contraction of the second s
	(A) a marchaeour of fight had been recorded and for her Anna and Darden ?
LCT &	1. Swave merally of nymenae more perposed earned in suggesting cancers, and the same as which a the section has in fact hadde colourly as
11072 LLAST	Provide the second Relation of the Association of t
1082	Forthe study phonesics instal optimple.
1054	sewton publishes his <i>Principle</i> . He provides a theoretical deviction for
	the velocity of sound in air, and Ends 298 m/s. The relation wavelength
-	figures troppeners repeat velocity is given
1700, ası.	Sarvent introduces the terms "nodes", "farmeric tree", "finidamental
	vibration", and suggests the name "acoustics" for the science of sound
1704	Newtran publishes his Opticks.
17)3	"Taylor obtains a dynamic solution for the vibrating string (Philosophical Process team).
1727	Edder proposes a linear relation between stress and strain.
1728	Bradley discours the phonon provident abritating
1729	Gravishnys that electricity can be transferred with conducting wires
1730	Biancomi shows that the velocity of sound in air increases with temperature.
17.33	d'Alembert publishes his Tradi de Diaranome.
17:1	Effortiation the concept of strain energy per mit length for a loam.
17 (4-51	D. Berrous,"S and Enter obtain the differential contribution and the dispersion
	relation for lateral vibrations of hois
1745	Nullet writes Essar our l'Electricité des Curps
1232	d'Alembert derives the one dimensional wave equation for the case of a
	vibrating string, and its solution for plane waves.

1750	Michell writes A Treatest un Artificial Maquets.
1752	Eth) introduces the idea of compressive normal stress as the pressure in
	a llnet.
1755	D. Bernoull' processes the principle of Process-tence of small oscillations
	the superior tearnedet.
1759	Ended derives the wave regention for sound. He develops the method of
1750	Activity with the second s
1750	Su priors priorities in a stability of the only of the transmission of the transmission of the transmission of the transmission.
1700	Faiguring: Sonsystems provided of the structuring sectory for the sectory interesting the first sectory of a fair scheme structure of the sector.
1.002	Lapado introduces de instantação de tra cuvergence mosterir cuver-
1.1.2.1.1	Contraction by Gauss III 1815.
1,62	Californ demonstratos transcriter is compressinge,
1761	built inferrors the "Bessel replaciate" in an incides solution of
	mensbranes.
1772	Cavendish writes An attempt to explain some of the Penneupal Phononenna
	of Electricity by means of an Electric Fluid
1773-79	Coulomb applies the concept of shear stress to failure of soils and
	fuctorial slip
1726	Ether publishes the so-called "Euler's equation of motion" in its general form
1776	Subher calculates the diffection of light by the sur (0.85 are seconds).
	rederived later by Cavendish and Enstein.
1777	Lagrange introduces the concept of scalar potential for gravitational helds.
1782	Luplace derives the so-called Laplace equation
1785	Coulomb uses the torsion balance to verify that the electric force law is
	(DOV160 SQUATE)
1787	Chladii visualizes - experimentally - the nucles of vibrating plates.
1788	Lastance mblishes his Withouten Analyticae.
1799	Latshan tutblishes his. Event da Wearnawe Céleste
1799	Volta intents the electric bettery
LXIII	Ritha algeorges the inferience calendary.
1801	Normal surveys the energy (Large of high). The interclasses the second of
1.5011	the figure in some some tradition figure. The discontines the further pro-
Lens	CHRENE CONTRACTOR AND AND AND AND AND AND AND AND AND AND
1502	 Constant probability in the second state of the secon
1802	Chladin nArshigatis bogundinal and forsional vibrations of bars
	experimentally.
1806	Young defines his modulus of elasticity and considers shear as an elastic
	strain
1808	J. B. Biot measures the velocity of sound up usin
1808	Chluchi studies for vibrations of strings and plates and horgetudinal and
	tursional valuations in ruds.
1808	Laplace purposes a corposular theory of double refraction.
1808	Malus discovers polarization of light.
1808	Poisson publishes his memory on the theory of sound
1809	Young proposes a dynamic evayor theory of light merestals
1811	Paisson publishes his Toute de Micanague
1811	Arago shows that some crystals after the polarization of light.

- 1812 J. B. Biot shows that some crystals rotate the plane of polarization of light.
- 1813 Poisson derives the so-called "Poisson equation" as a relation between gravitational potential and mass density.
- 1814 Fraudoder discovers the dark line sportrug. Light waves reveal the presence of specific elements in celestral bodies (Kerddooff and Barsen's paper, 1859).
- 1815 Brewster investigates the "Brewster angle" on the basis of his experiments and those of Malus.
- 1816 Frestel establishes the basis for the "Freshel-Kirel-Jaoff theory of diffraction".
- 1816 Laplace shows that the adiabatic clasticit constant should be used to calculate the sound velocity on air.
- 1816 Young suggests the transversality of the cibratings of light basic on the fart that light of differing publication cannot interfree. This solves many of the differities of the wave theray.
- 1820 Poisson solves the problem of propagation of compressional waves in a three-dimensional hild medium.
- 1820. Oersted notes the relation between electricity and magnetism.
- 1820 Ampère models magnets in terms of molecular electric currents telectrodynamics).
- 1820 Bint and Savart dochere the formula for the magnetic strength generated by a segment of wire excrying electric current.
- 1821 Daty shows that the resistance of a long conductor is purportional to its length and inversely proportional to its cross-sectional area.
- 1821 Treshel interprets the interference of polarized field in terms of transverse vibrations.
- 1821 Navire derives the differented equations of the theory of elasticity in transof a single elasticity constant.
- 1822 Seclarck discusses the thermoelectric effect.
- 1822 Cauchy introduces the notion of stress (strain) by means of six component stresses (strains). He also obtains an equation of motion in terms of the displacements and two elasticity constants.
- 1822 Fourier publishes his "badytool Theory of Heat, where he introduces the vibrate series of sines and cosines (mathematical firms of the superposition principle).
- 1823 Freshel ultrains his formular for reflection and refraction of light.
- 1824 Batuilton publishes his first paper On Constics.
- 1825 Ampère publishes his law, known also as Stokes theorem",
- 1825 Weber publishes his book Wellenlehre.
- 1826 Airy publishes his Mathematical Tracts on Physical Astronomy,
- 1826 Uolladon and Sturm measure the speed of sometim water obtaining 1335 ar/s
- 1826 Bamilton publishes his *Theory of Systems of Roys*. The introduces the characteristic function for optics.
- 1827 Ohm obtains the relation between electric current and resistance.
- 1828 Cauchy extends his theory to the general case of acolotropy, and hads 21 clasticity constants. Theorem are type elasticity constants (the "ran-constant" theory).
- 1828 Green introduces the concept of potential in the mathematical theory of clerificity and magnetism. He derives "Green's theorem".

- 1825 Poisson predicts the existence of compressional and shear elastic waves. His theory predicts a ratio of the wave velocities equal to √3/1, and Poisson ratio equal to 1/4.
- 1830 Cauchy investigates the propagation of plane waves in crystalline media.
- 1840 Savart measures the minimum and maximum audible frequencies (8 and 24000 voluments per second, respectively).
- 1831 Foraday shows that varying currents in one circuit induce a current in a mighboring circuit.
- 1832 Henry independently discovers the induced encours effect.
- 1832 Gauss independently states Green's theorem.
- 1833 Hamilton introduces the concept of "enkonal equation", the term eikonal being introduced into optics by Brazis.
- 1833 Harolton develops the basic geometric corrects of slowness surfaces for anisotropic media. If: predicts emiral refraction, that is verified experimentally by Lloyd .
- 1831 Hamiltun publishes his On a General Method in Dynamics. The Hamiltonian concept for dynamics is introduced.
- 1815 Gauss formulates "Gauss law".
- 1835 MacCullage and Neumann generalize Cauchy's theory to anisotropic media
- 1836 Any calculates the diffusion pattern produced by a coordar aperture
- 1837 Green discovers the boundary conditions of a solid/solid interface
- 1837 Green derives the equations of elasticity from the principle of conservation of energy. He defines the strain energy, and finds 21 elasticity constants in the case of acolor ropy (the "nuclino onstant" theory).
- 1847 Haraday nutroduces the concept of the dielectric permittivity.
- 1838 Foraday explains electromagnetic induction, showing that magnetic and electric induction are analogous.
- 1838 Viry develops the theory of caustics.
- 1838 Green solves the reflection-reflection problems for a fluid/fluid boundary and for a solid/solid boundary (the other), and applies the results to light propagation.
- 1839 Cauchy proposes an elastic other of negative compressibility.
- 1839 Green, like Cauchy in 1830, investigates crystalline modia and obtains the oppations for the propagation velocities in terms of the propagation direction.
- 1839 MarCollag2e proposes an elastic other without longitudinal waves, based on the rotation of the volume elements.
- 1839 Lord Kolym Ends a mechanical-model analogue of MacCullagh's erbor.
- 1842 Doppler discovers the Dopplet effect.
- 1842 Mayer states that work and heat are operadent. His paper is rejected in the Dotalen der Physik.
- 1842. Ford Kelvin usis the theory of heat to obtain the continuity equation of electricity.
- 1841 Scott Bussell discovers the solitary wave.
- 1845 Finaday discovers the magnetic rotation of light. He introduces the endrept of held,
- 1845 Neumann narodines the vector potential. The next year, hord Kelvin shows that the magnetic field can be obtained from this vector.

- 1845 Stokes identifies the modulus of compression and the modulus of rigidity, as corresponding to resistance to compression and resistance to shearing, respectively.
- 1836 Founday publishes Thoughts on Ray Vibrations in Philosophical Magazine III suggests the electromagnetic nature of light.
- 1846 Weber combines electrostatics, electrodynamics and induction, and proposes an electromagnetic theory.
- 1847 Helmholtz writes a memoir about the conservation of energy. The paper is rejected for publication in the Annalaz dev Physik.
- 1848 Kuchhulf generalizes Ohm's law to three dimensions
- 1849 Mearcl invents the trilephone.
- 1849 Stokes shows that Poissin's two waves correspond to irrotational dilatation and equivoliminal distortion.
- 1849 Fizena confirms Presnel's results using interferometry.
- 1850 Foreault measures the velocity of light in vater to be less than in an Newton's emission theory – which predicts the opposite – is aluminoid.
- 1850 Stukes introduces a twring) concept of anisotropic in (tia to explain wave propagation in (rystals.)
- 1850 Lord Kelvin states Stokes's theorem without proof and Stokes provides a demonstration.
- 1854 Lord Kelvargives the theory of the RLC circuit.
- 1854 Lord Kelver derives the telegraphy registron without the enductance is diffusion registron?
- 1855 Lord Kelvin justifies Green's strain energy function on the basis of the first and second laws of thermodynamics.
- 1855 Palmieri devises the hist seismograph.
- (855) Writev and Kukhrausch find an electromagnetic velocity equal to √2 the light velocity.
- 1850 Lord Kelvin introduces the concepts of eigenstrain ("principal strain") and eigenstiffness 1 principal elasticity").
- 1857 Airchkoff derives the telegraphy equation including the inductance. He hads a vehicity close to the velocity of light.
- 1861 Riemann modulus Weber's electromagnetic theory.
- 1861 Kirchhulf derives the theory of the black budy.
- 1863 Th hubility introduces the end opt of "point source".
- 1863 Helmholtz publishes his Lehn conden Two mfindgragen about the theory of harmony.
- 1864 Maxwell obtains the equations of electromagnetism. The electromagnetic nature of light is demonstrated.
- 1867 Maxwell intruduces the "Maxwell midel" to discribe the dynamics of gases
- 1867 Loring develops the electromagnetic theory in terms of retained patentials.
- 1870 Christiansen discovers animalous dispersion of light in solutions.
- 1870 Helmholtz shows that Weber's theory is not consistent with the conservation of energy.

- 1870 Helmholtz derives the laws of reflection and refraction from Maxwell's equations, which were the subject of Lorentz's thesis in 1875.
- 1871 Rankup publishes equations to describe shock waves. Date: also published by Hagmint in 1889).
- 1871 Rayleigh publishes the so-called "Rayleigh scattering theory", which provides the first correct explanation of why the sky is blue.
- 1872 Betti states the occipionity theorem for static fields.
- 1873 Maxwell publishes his Theater on Electricity and Magnetism
- 1873 Rayleigh derives the reciprocity theorem for vibrating bodies.
- 1871 Boltzmann has the foundations of broeditory mechanics ("Boltzmann's superposition primiple").
- 1871 Corner introduces the "Corner spiral" for the solution of diffraction, problems,
- 1874. Oskar Emil Meyer introduces the "Voigt solid",
- 1871. Uniov introduces the vector of the density of energy flux.
- 1875 Kerr discovers the "Kerr effect". A dielectric modulu subject to a strong electric field becomes birefringent.
- 1876 Puchhammer studies the axial vibrations of cylinders
- 1877 Christoffel investigates the propagation of surfaces of discontinuity in anisotropic media.
- 1877 Rayleigh publishes The Theory of Sound.
- 1879 Hall discovers the "Hall effect 1
- 1880 There and Jacques Curie discover piezoelectracity.
- 1880 Knult discovers anomalous dispersion in the vapor of sodium
- 1881 Michelson larging his experiments to driver) the other
- 1884 Pownting establishes from Maxwell's equations that energy flows and can be localized.
- 1885 Lamb and Heaviside Escover the concept of skin depth.
- 1885 Somithana obtains solutions for a wide class of sources and boundary conditions
- 1885 I ord Rayleigh products the existence of the "Baylingh surface waves"
- 1887 Vuigt performs experiments on anisotropic samples (beryl and necksile). The "multi-constant" theory - based on energy considerations - is enafirmed. The "tari-constant" theory - based on the noderular hypothesis - is dismissed.
- 1887 Heaviside writes Maxwell's equations in vector form. He invents the modern vector calculus notation, including the gradient, divergence and curl of a vector.
- 1887 Vogt, investigating the Doppler effort in the other obtains a first version of the "Lorentz transformations"
- 1888 There generates callic waves, confirming the electromagnetic theory. He discovers the photoelectric effect and predicts a finite gravitational velocity.
- 1889 Fitzgetald suggests that the speed of light is an upper bound.
- 1889 Reuber-Paschwitz detects P waves in Potsdam generated by an earchiptake in Japan. Global seismology is born.
- 1890 Hertz replaces potential by held voctors and deduces Ohm's. Kirchhoff's and Coulomb's laws.
- 1893 Puckels discovers the "Pockels effect", similar to the Kerr effect,

1891	Korroweg and de Vibes obtain the equation for the solitary wave
1894-901	Burge and Kutta develop the Burge-Kutta algorithm.
1895	Lorentz gives the "Lorentz transformations" to first order in the normalized
1896	Ballzki andles the theory of anisotency to seismic wave perpayation.
1897	Material's first windows tole raphy patient
1899	Knott derives the contations for the reflection and transmission of classic
	idane ways at plane interfaces
1959)	Marconi's second wireless-telegraphy patent.
1902	Powertung and Thomson introduce the "standard linear solid" model.
-	referred to here as the Zenne model.
1963	Live develops the theory of point services in an unbounded elastic space
1901	Lands obtains the Green function for surface Rayleigh waves.
1901	Volterra publishes his theory of dislocations based on Somigliana's solution.
1994	Volterra netroduces the integro-differential conations for hereditary problems
1905	Einstein (avestigates the photoe) of the effort and states that high as discrete
	electroningnetic admition
1906	Oldham (1906) discovers the Farth's care by using P-wave amplitudes.
1908	Mich velops the "Mie scattering" theory, describing scattering of spherical
	particles.
19-0	Cosserat publishes his theory of micropolar elasticity (Cosserat and
	Cosserat, 1909).
1909	Multanziene discovers the "Multa" discontignate on the basis of sensatic waves -
1941	Debye introduces the ray series in "Debye expansion".
1911	Live discovers the "lane surface waves"
1912	I. F. Richardson parents the first version of sonar.
1912	Somacrield introduces the "Sommerfeld radiation condition"
1915	Galerkin bublishes his functe-element method
1919	Mintrop discovers the sustanchinal wave
1989-37	The WKBJ (Wrotze) Kramers Brillmain Juffreys) approximation is
	introduced in several branches of physics.
1923	de Broghe proposes the model by which thry particles of matter, such as
	electrons, display the characteristics of waves.
1921	Stendey (1924) publishes his paper about - Stondey interface waves '.
1925	Walter Elsasser describes electron diffraction as a wave property of matter.
1926	Born develops the "Born approximation" for the scattering of atomic
	particles.
1926	hilfreys establishes that the onter Farth's cure is liquid by using S waves.
1926	Scholdinger works out the mathematical description of the atom called
	"wave mechanics", based on Hamilton's principle.
1926	Klene-Fock-Gordon equation: a relativistic version of the Schrödinger wave
	equation
1997	Paul Dirar presents a method to represent the electromagnetic field as quanti-
1928/35	Graffi studies hereditary and hysteretic phenomena based or
	Volterrais theory.

- 1928 Syquist nerodates the sampling theorem.
- 1928 Sokolov proposes an ultrasonic technique to detect flavs in metals.
- 1932 Debter and Series observe the diffraction of light by gltrasonic waves
- 1934 Frenzel and Schiftes (1934) discover similarities center "Born and Will, 1964, p. 594).
- 1935 Richter and Gutenberg incent the Richter magnitude scale
- 1936 Lehmann discovers the Earth's inner rote on the basis of P waves generated by the 1929 New-Zeaherd earthquide.
- 1937 Bruggeman shows that finely layered media behave as anisotropo media.
- 1938 S. M. Rytox develops the ray theory for electromagnetic waves
- 1939 Walter Elsusser states that oddy currents in the liquid core, due to the Earth's rotation, generate the observed magnetic field.
- 1939 Cagnizol (1939) publishes his method for solving transient elastic wave propagation.
- 1939 Grafficestends the reciprocal theorem of Bérri to dynamic fields, although the concept dates back to Helmholtz (1860) and Rayleigh (1973).
- 1940 Firestone develops an ultrasonol pulse-echo antial-flaw detector,
- 1941 Biot publishes the theory of consolidation
- 1941 K. F. Dussik makes the first attempt at medical imaging with ultrasment.
- 1941 Kosten and Zwikker (1941) propose a scalar theory, predicting the existence of two compressional waves.
- 1944 Terzaghi publishes his Throw treal Soft Mechanics.
- 1941 Trenket publishes his paper on the dynamics of porous media and the seismoelectric effect. The requiring are marky identical to BioUs poroclastic registrons.
- 1944 Pishkov observes second (thermal) sound in liquid helium II.
- 1947 Schulte identifies the interface wave traveling at liquid solid interfaces.
- 1948 Feynman develops the path-integral formulation.
- 1948 Galor describes the principle of wave-from reconstruction, the basis of holography.
- 1949 Kvanie (1949) publishes his theory about waves in precordectric crystals
- 1939 Mindhe publishes the Hertz-Mindlin model to obtain the rock moduli as a function of differential pressure.
- 1951 Gassmann derives the Gassmann modulos" for a saturated purons medium.
- 1952 Lighthill (1952) publishes the acroacoustics replation.
- 1954 Haskel (1953) publishes his matrix method for wave propagation.
- 1953 Kombauser (1953) publishes the ray theory for moving fluids
- 1956 Biot publishes the dynamic theory of purious media and predicts the slow compressional wave.
- 1958 (in Thiop develops the Cagniard de Huop technique,
- 1958 McDonal, Angona, Milss, Scipletsh, var. Nostrand, and White publish field experiments indicating constant Q in the seistable frequency band.
- 1959 Knopoll and Gaugi develop the reciprocity principle for anisotropic media.
- 1962 Backus obtain the transversely-isotropic operadent medium of a finely lowered medium.
- 1963 Deresiewicz and Skalak obtain the boundary conditions at an interface between porcos media.

- 1963 Hashin and Shtrikman obtain bounds for the elastic bulk and shear moduli of a composite.
- 1961 Brutshert presents a theory for wave propagation in partially saturated suils. The theory predicts three P waves.
- 1964. Hess (1964) provides evidence of the seismic anisotropy of the uppermost mantle.
- 1965 Shipuy and Rudnik (1965) observe fourth sound in helenn IL.
- 1966 dr. Hoop develops the reciprority principle for axisotropic anelastic media
- 1966 King performs laboratory experiments on partially-saturated rocks
- 1968 Mitermati and Karal use fittite differences to compute synthetic scisonograms.
- 1968 McAllister (1965) invents the Sodar.
- 1969 Waterman (1969) introduces the 1-matrix formulation for acoustic scattering.
- 1971 Buchen investigates the properties of plane waves in visco-dastic media
- [971] First observational evidence that the inner rope is solid (Dzirwonski and Gilbert, 1971).
- 1971 O'Dakerty and Anstey ideam their formula to discribe stratigraphic filtering
- 1972 Berker and Richardson explain the "Rayleigh window" phenomenon using viscorlastic waves.
- 1972 Lysner and Drake simulate seismic surface waves with finite-elements methods.
- 1975 Brown and Korringa obtain the elasticity tensor for anisotropic porous media.
- 1975. White develops the theory describing the autoscopic loss enchanism
- 1977 Currie, Haves and O'Leary predict additional Rayleigh waves in viscoelastic media.
- 1977 Domenico performs laboratory experiments on meonsolidated reservoir sands.
- 1979 Allow M. Cormack and Godfrey N. Bounsheld receive the Nobel Prize for developing computer axial romography (CA14.
- 1929 Burndge and Vargas obtain the Green function for periodisticity
- 1980 Plana observes the slaw compressional wave in synthetic media.
- 1984 Gazdag introduces the English pseudospectral method to compute synthetic scistnograms
- 1981 Masters and Gilbert (1981) observe spheroidal mode splitting in the inner code, indicating anisotropy.
- 1982 Feng and dolmson predict a new surface wave at a fluid/porous medium interface.
- 1984 Dav and Minster use internal variables (memory variables) to model anelastic waves .
- 1990 Santos, Donylas, Corbrid and Luvira generalize Birt's theory to the case of one rock matrix and two saturating fluids. The theory predicts a second slow P wave.
- 1994 Leebare, Cohen-Friondji and Aguirre-Puents generable Bior's theory to the case of two rock matrices and one saturating kind. The theory predicts two additional slow P waves and a slow S wave.
- 1994 Hellog introduces Kelvad's theory of eigenstratios in scismic applications.
- 2004 Prof. Borryman and Harris show that the mesoscopic loss is the dominant mechanism in third filled rocks at scismic frequencies.

This page intentionally left blank

Leonardo's manuscripts

Leonardo da Vinci (1452-1519)²

"Decreated perceived intuitively and used effectively the right experimental method a century before brancis Bacon philosophised about it inadequately, and Galileo put it into practice (Dampier, 1961)

Description of wave propagation unterference and Havgens' principle (1678):

Everything in the cosines is propagated by means of vares. (Manuscip: B vit, butto de Vouce, Paper of sure of gue theory two small stores of the sure time on a short of normalies water at some distance from each other, you will abserve that around the two per assume name events sequente exclusioner formed: these will meet us they exceeded in size and the neperstrate and intersect one conthere all the white maintaining as there respective centres the spats struck by the stones. And the reason for this is that water, although apparently many does not leave its original particula because the concerners made by the shines close again monodout by. There fire the mation predated by the quick opening and classing of the water has consect only a shock which may be descended as treacy with c than marchield. To order to understand better about Lyrano, watch the blades of strain that because of these lightness float on the water and abserve how they do not depart from their compand positions in space of the mains malerizanth them consol by the memory of the endos. The machine of the whiter being in the nature of termor wither them more word, the cardes cannot been, one another on accenting, and us the water is of the same grafity all the way through, its parts transmit the transmitter are another without champing position. (Manuscript A, 166) It storat de France, Paris

Description of the effort discovered by Doppler in 1832;

If a stone is flow with matural sometry, its carbox will be equilation from the orienter. But if the stream is maxing, these circles will be cloughted, egg-shaped, and will travel with their centre stream from the spot where they were exected, following the stream. (Maxweege 1,87, heritar de Yrane, that's c

Description of Newton's prism experiment (1600).

If you phase a glass full of water on the windowsill so that the saw's rays will strake it from the other sole, you will see the afon such colours formed in the impression work by the set second that have penetroned through that glass and failers in the durk of the foot of a window and since the eye is not used for e-no many with full certainty say that these colours are not in any way due to the eye. (Codes takeson, 2010) Wead throws Window to

[&]quot;Sciences: White (2000) http://www.entertheory.com

Leonardo's scientific approach to investigate the refraction of Ref.()

Here two trays made parallel to each other. And let one by $\frac{3}{2}/5$ smaller than the other, and of equal length. There each on an the other and grant the consider and leave an another entry of specific devices on a the other and parallel to consider, and leave another opening the set of a leave do there are a for simplify pass there coming from another opening in worder. Then see whether or not the ray passing in the water contrast for the set of the standilet opening in worder. Then see whether or not the ray passing in the water contrast before a the two transferences is the standilet opening in the water contrast for the set of the se

Description of atmospheric refraction, discovered by Brake in the 16th century:

for see how the study rays proctant this concentre of the sphere of the sur, have two glass spheres much, one twee the second the other, as round as can be. Then cut there in half and put one assolution of the other and class the fronts and fill with water and have the ray of similarly pass as you ded above here is referring to his explore simpler references experiment). And we whether the ray is beat. And thus you can make an infinite marker of experiments. And form gove rate. (Manascire 5, 38, 10 - 100) for the trans.

Explanation of the blue sky, before Tyndall's 1869 experiments and Rayleigh's 1871 theory.

I say that the film which is seen to the admissible is not previous own colour, but is caused by the boated measture barring evaporated into the most remote and improvphile particles, which the beams of the solar rays altitud and cause to see a hummous against the deep indexes darkness of the region of fire that factor a covering among them. Over-Decision A Royal Lingar, Windson

Statem at about light having a finite velocity, before Römer's conclusive measurement in 1676

It is unpossible that the eye should project the cusoul power from doubled person rays, since, as soon as it opens, that front of the eyel which would get use to this constraints would have to go forth to the object, and this it could not do without time. And this being so, it could not travel as high as the sam in a month's time when the eye could be set it. (N-informing 1.6.11, Biblioutlesgie Nationale, Parse).

Description of the principle of the telescope

It is possible to find means by which the eye shall not see remain objects as much dimmedial as an induced perspective π . [Maximum F. 58, Instituted France Parise. The further gain place the eyephics from the eye, the larger the objects appear in them, when they die for pressure liftly genes odd. And if the eye sees the equid objects in comparison, one catisate of the glass and the other within the field, the one to the glass will seem large and the other small. But the things seen could be 200 ells petitle over 200 m from the eye... (Monology 7, 12) Institute de France Parise Construct glasses to see the Muon tanguiffed. (Codex Maximum 1996) e Mathematica. Maxim

A statement anticipating Newton's third law of motion (1666):

As note by pressances exercised by the adject against the arc as better are against the leady. [Codes Atlanticus, 381 Ambro-sana Intern. The principle of least action, stated before Fermat in 1657 and Hamilton in 1834;

Every action in output takes place to the shortest possible way. (Qeadequ, IV) (in s

Leonardo described lossil shells as the remains of ancient organisms and put forward a mass/mertra theory to describe scaled and continent ups and down-lifting as monitories ended elsewhere on the planut. The evolution and age of the Farth and loging contines, preceding Genrye Cuvier (1804) and Charles LyeD (1863), and plane tectorics, anticipating Wegener (1905):

That is the drifts, arrang one and mother, there are still to be found the traces of the second which exacted aport there where they were need yet dry. And all marine chars still cantain shells, and the shells are petitled together with the char. From their transies and unity some persons will have it that these animals were carved up to places remain from the second persons will have it that these animals were carved up to places remain from the second persons will have it that these animals were carved up to places remain from the second of the delaye. Another sect of animal persons declare that Nature or Heaven created there in these places by released caffirmers, as if in these places we dot not also find the houses of fishes which have taken a long time to grow, and as if, we could not count, in the shells of coeffee and smalls, the genes and months of their tip, we we do to the houses of holls and area, and in the heavetax of plants that hore never here end on the line houses of holls and area, and in the heavetax of plants that hore never here end in any part.

And within the limits of the sequence strate of works they are found, for in number and in price like these which were left by the sea, larved alore in the word, which subsequently decid up and, in time, was peticified.

Great means always can tacked, being released by the certh, which is strived by the function of their waters at the battain and on their shares and this wearing distarts the face of the startin mode by the layers of shells, which is an ite surface of the matrix mode and which were produced there when the soft waters covered them; and these startin weak entered over much from time to true, with oral of carries the bases, or covered down to the second over much from time to true, with oral of carries the bases, or covered down to the second over much from time to true, with oral of carries the bases, or covered down to the second process and flowly of more or bases iterate and thus these happens of und became mused to such a bright, that they came up from the battain to the own. At the present time these battains are so high that they form halls in high availants, and the creases which wear away the soles of these momentains, uncover the starts of these shells, and they the softened sole of the cardia continually trees and the antipoles stark closer to the centre of the cardia, and the antipole of the sees have become meaning ridges .

Fin centre of the sphere of centers is the true centre of the yields of our world, which is composed of neuro and earth, having the shape of a sphere. But, if gave word to just the centre of the element of the could, this is placed at a point equidistant from the surface of the ocean and not equidistant from the surface of the earth; for it is evident that this yields of earth has nearline any perfect totaching, excepting in places where the second coconsists of other static any perfect totaching, excepting in places where the second coconsists of other static static scale energy part of the earth that rules above the write is farther from the centre (Codex Leberster, Royal Labora).

The theory of evolution, stated hefere Macquertuis (1745) and Charles Darwin (1859).

Notare, being accordant and taking pleasure in construction and anikant constantly new large and forms, because she prices that her terrestrial poderadis becaute thereby ang-
mented, is more ready and more swift in her creating than time in his destruction... (Codex Leicester, Royal Library, Windsor.)



The coffee cup caustic. The bright line seen in a coffee cup on a sunny day is a caustic. Consider the Sun as a point source of light and constructs rays according to geometrical optics. Parallel rays reflected in the inner surface generate a curved surface (caustic), which is the envelope of the rays. The caustic has a cusp at its center (paraxial focus). Note that the surface is brighter below the caustic (e.g., Nye, 1999). This phenomenon has been described by Bernoulli (1692) and Holditch (1858). Leonardo has predicted the phenomenon. He is arguing that in concave mirrors of equal diameter, the one which has a shallower curve will concentrate the highest number of reflected rays on to a focal point, and as a consequence, it will kindle a fire with greater rapidity and force (Codex Arundel, MS 263, L86v-87, British Library, London). Seismic reflections from a geological syncline produce these types of caustics.

A list of scientists

L'in nomente commence à Gablée, Bogle et Descartes, les fondateurs de la Philosophie expérimentale, tous tous consucrent leur cre à méditer sur la mature de la familie, des contenes et des forces - Gablée jette les bases de la Méranique, et, avec le télescope à référention, cettes de l'Astronomie physique : Boyle perfectionne l'experimentations quant à Descartes, d'embasse d'une van périetante l'ensemble de la Philosophy mitardh

Affied Corna (Corna, 1900)

The following scientists have contributed to the indepstancing of wave propagation from different fields – optics, negate theology, electronognetism, acoustics, ray and field theory, differential calculus, seismology, etc. This list includes scientists been during and before the light rearray's.

Thates of Miletus	ea 634 BC	ea. 546 BC	Gener
Pythagoras	(a) 560 BC.	(a) 186 BC	Gpoor
Aristotle	on 384 BC	on 322 BC	Crewe
Euclid of Alexandria	(a) 325 BC	ea - 265 BC	Egypt
Chrysippus of Sull	[ca] 279 BC	ca. 207 BC	Greene
Vittorvius	ca. 25 BC		Rome
Heron of Alexandria	ra. 14	La. 75	Egypt
Ptolemy, Cheolius	(a. 85	on, 165	Lgypr
Boethius, America Maglius Secerans	ra, 480	ea. 525	Rome
Hm al Haythan	ra, 965	ca - 1040	Loaq
al Giazzali, Alta Hamid Muhammadi	1058	1111	lean.
Grossereste, Robert	1168	1253	England
Bacon, Roger	1214	1294	England
Petrus Peregrinus	 (a) 1220 	Ga. 1270	France
Witelo	< a. (1239)	ca. 1275	Poland
Dietrich of Freiberg	1250	1310	Highnid
Buridan, Joan	en 1295	1358	Figland
Pacinik Luca	1415	1514	Italy

The sources are the Directary of Sciencife Biography, Gölspiel, C. C. Fell, Charle-Sechers, Suis, 1972), the web strend the University of St. Ardrews, Sectional aways historytansistematicwalatik, historyte, the rock strend Era. Weisstein's Treasure Tawe of Scienthe Regraphy covariational traveletion of the web size of the University of Hormon, have covariable contribution traveletion at the web size of the University of Hormon, have even and conduct contribution traveletion of the Weisstein's first trayer, Germany active covariable contribution traveletion at the web size of the University of Bertrayer, Germany active applysized at the web support first first on Berght, Spars, it was stress in bold lent appear at the domology. The plane of both is a drated

Leonardo da Vinci		1452		1519	ltaly
Agricola, Georgius Baner		1460		1555	Germany
Maurolycus, Franciscus		1491		1.57.5	baly
Galilei Vincenzo		1520		1591	Italy
Cardano, Guolamo		1501		1576	Italy
Della Porta, Giambattista		1535		1615	Italy
Risner, Friedrich			Cit.	1580	Germano
Gilbert, William		1511		1603	England
Brale, Tycho		1546		[60]	Sourcher.
De Dominis, Marco Automo		1560		1624	baly
Harriot. Thomas		1560		1621	England
Baron Francis		1561		1626	England
Briggs Honey		1561		1630	England
Galilei. Galilen		1561		1632	ltab
Lippershey, Hans		1570		1619	The Notherlands
Kepler, Johannes		1571		1630	Germany
Schemer, Christoph	ca.	1573		1650	Germany
Such on Rosen (Suellius) Willebrook		1580		1626	The Netherlands
Cabro, Nirolo		1585		1650	Italy
Mersenne, Marin		1588		1638	Frances
Gassendi, Pirrre		1592		1655	France
Descartes, René		1566		1650	France
Cavalieri, Bonavettura		1598		1647	Italy
Fermat, Pierre de		1601		1665	France
Gipeneke, Otto you		1602		1686	Germany
Kircher, Athanasus	ca	1603		1680	Germany
Browne, Thomas		1605		1682	England
Borelli, Giovani		1608		1677	Italy
Divai, Eustachio		1610		1685	baly
Wallis, John		1646		1703	England
Grimaldi, Francisco Morio		1618		1663	ltalv
Marintte Edm	ca	16:50		1684	France
Pirard. Jean	ca.	1620		1682	France
Viviani, Vincenza		1622		1703	Italy
Bartholinus, Erasmus		1625		1698	Dennark
Cassini, Giovanni Domena o		1625		1712	Dalv
Modulet, Samuel		1625		1615	England
Boyle Robert		1627		1621	Indaud
Huygens, Christiaaa		1629		1625	The Netherlands
Hooke, Bulant		0635		1702	England
Pardies, Ignae, Caston		1646		1673	Frator
Gregory, James		1648		1675	England
Ango, Piene		1640		[131]	France
Newton, Isaac		642		1727	England
Römer Olaf		1641		1710	Deamark
Flamsteed, John		06.06		1719	England

Leihniz, Gottfried Wilhelm	16.06	1716	Germany
Eschirdonisch, Ehrenfried Walther	1651	1708	Germany
Sanveur, Joseph	1653	1716	France
Halley, Dimmad	1656	1742	England
Hankshoe, Francis	1666	17.Wi	England
Bernoulli, Jukaan	Hitiz	1748	Switzerland
Gray, Stephen	1670	1736	Figland
Hermann, Jakob	1678	1733	Switzerland
Taylor, Brook	1685	1731	Lugland
Masschenbrock, Pieter van	1692	1791	The Netherlands
Bradley, James	1693	1762	England
Bongner Pierre	11218	1208	France
Cisternay du Eay, Charles-François de	11218	1739	Enner
Manpertuis, Pierre Louis Morran	11298	:759	Franci
Bernoulli, Danel	1700	1782	The Netherlands
Kleist, hwald fürgen von	1700	1715	Germany
Nollet, Jean Autome	1700	1770	France
Celsus, Anders	1701	1711	Swalen
La Condunante Charles Marie de	1500	:271	Filmer
Cumer, Grabriel	170.5	:752	Switzerland
Franklin, Benamin	1706	17:80	153
Euler Loonard	1707	1783	Switzerland
Beservich, Ruggiero Guserne	1711	1787	ltaly
Lemenessy, Mikleat	1711	1765	Itussia
Watson William	1715	1757	Frieland
Bianconi, Giovanni Ludovato	1217	1781	ltalv
d'Alembert, hat le Rond	1717	:783	Franci
Canton, John	1718	1772	Lugland
Mayer, Johann Toblas	1723	1762	Germany
Michell, John	1724	1793	Endand
Arrings Funz Meta Dusdosus	1723	1802	Germany
Lamberts, Johann Berrich	1728	: 777	Germany
Sudlanza 3 Lazzaro	1729	17:20	lish
Cavoudish Heres	1731	*sta	Fuelend
Wilder Johannes	1732	1796	Swaha
Priestlet Joseph	1733	1801	Looland
Coulomb Charles Arenstu de	1736	1806	Datas
Lagranders & second carls	1736	1811	ltahr
fisheast huisi	1537	- 205	li she
Malto Messindus Gresosuri Antonio Accestorio.	1745	- 4 27	lush
Lanland, Piece Siere	17:00	10.17	L'a dana
Lagrandez, Adria, Maria	17.54	1021	là an a
Bring and Bring and Bring and Bring and	17.52	1200	1.81
Chladad Fredd Marga Kandad	1756	1907	Contents
Normanni (1918) (1800) Stephenson D'Asar - Henrowski Williador Marchalos	11-1-0	1921	cormany r!
CALIFORNIA THE CALIFORNIA CONTRACTOR AND AND AND AND AND AND AND AND AND AND	1708	- 2 10 - 2 12	CIPILIAIR
renner, acan napusu acsepu	1.115	10.201	L. 17116.1

Nicol, William	1768	1851	Sentland
Seeheck, Thomas	1770	1831	Estoria
Young, Thomas	1774	1829	England
Biot, Jean Baptiste	1774	1862	France
Ampère: André Marie	1775	$^{+836}$	France
Mahus Eticane Louis	L'initi	1812	France
Germain, Suphie	1776	1831	France
Ritter, Johann Wilhelm	1770	1810	Gormany
Soldner, Johann Georg von	1776	1833	Germany
Gauss, Carl Friedrich	1777	1855	Germany
Oersted, Haus Cleristian	1777	1851	Dennark
Davy, Humothey	1778	1820	England
Brewster, David	1781	1865	Sectional
Prissur, Simón Denis	1781	18.00	France
Storgron, William	1783	1850	England
Ressel. Friedrich Wilhelm	1781	18.06	Germany
Hausteen, Christopher	1781	1873	Norway
Navier, Claude Louis Marie Henri	1785	1836	France
Poltier Juan Charles Athanase	1785	1845	France
Arago, Donnique François	1786	1853	France
Framhofer, Juseph you	1787	1826	Germany
Fresnel, Augustin, Joan	1788	1827	France
Cauchy, Augustin Louis	1789	1857	France
Olan, Georg Simon	1789	1851	Germany
Faraday, Michael	1791	1867	England
Mossitti, Ottavimo Fabrizio	1791	1863	Baly
Piela, Galizio	1791	1850	Italy
Savart, Félix	1791	18.01	France
Herschel, John Frederik William	1792	1871	England
Green George	1794	18.01	England
Habjuet Jacques	1794	1872	France
Lamé, Gabriel	1795	1870	France
Henry, dosuit	1797	1878	USA
Puisenille Jean Lémard Marie	1797	1869	France
Saint Venant, Adhémai Jean Claude Barré de	1797	1886	France
Melloni, Macedonio	1798	1851	halv
Neumana: Franz Ernst	1798	1895	Czech Republic
Clapevron, Benoit Buildraule	1799	1864	France
Lloyd, Buushney	1500	1881	hehad
Airy, Genree Biddell	1501	1802	England
Feature, Gustay Finodo:	1801	1887	Gormany
Colladon, Jean Daniei	1802	1893	Switzerland
Sturm Jacques Charles	1802	1855	Switzerland
Doppler, Caristia, Andreas	1803	1853	Austria
Jacob, Carl Gistay Jacob	15114	1851	Contractory
Legz, Heimich Friedzich Emil	1804	1865	Germany

Wehrer Williahn Edward	15111	1891	Germany
Dirichler, Gustav Peter Lejonn	1805	1859	Germany
Hamilton, William Rowan	1805	1865	Irclazed
Yohr, Friestuch	1.5106	1879	Germany
Pahnieri, Lugi	1807	1896	Daly
Mencei, Antonio	1808	1896	lialy
Scott Russell, John	1808	1882	Scotland
Lionville, hseph	1809	1882	Frater
Mar Cullagh, James	1809	1847	Inclosed
Menabroa, Foderigo	1809	1896	Italy
Vallet, Robert	1810	1881	Indand
Brusen, Robert Wilhelm Eberhaud von	1811	1899	Germany
Grove William Robert	1811	1896	Walts
Augstrum, Anders Jüns	1811	1874	Swehn
Mayer Julius Robert	1814	1878	Germany
Sylvester, James Joseph	1511	1897	hughand
Jonle, James Prescott	1818	1889	England
Fizeau Augand	1819	1896	Dance
Fourault, Jean Lixon	1819	1868	Filmer
Stukes, Groupe Galmiel	1819	1903	fieland
Rankine, William John Macquorti	1820	1872	Scotland
Tyndall, John	1820	1894	helazol
Chebashey, Pafnuty Lyoyich	1821	1891	Russia
Helmholtz, Hermann von	1821	$[\times 1]$	Germany
Cerchi, Edippo	1822	1587	halv
Clausius Right? Julius Emmanuel	1892	1888	Germany
Galton, Francis	1822	1911	England
Hernáte, Charles	1822	1901	Futer
Krönig, A. K.	1822	1879	Germany
Lissajous, Jules Autoine	1822	[X80]	Prance
Betti Fariro	1823	1892	balv
Kennecker, Luopuld	1823	1891	Poland
Kirchhoff, Gastav Robert	1821	1887	Russia
Kerr. John	1821	1907	Scottand
Thomson, Wilham (Baron Kelvin of Largs)	1821	1907	Inclazed
Beer, August	1825	1863	Germany
Riemann, Goorg Friedryd, Bernhaud	1826	1866	Germany
Christoffel, Elven Brano	1829	1900	Germany
Lorenz, Luciwig	18/29	1891	Dremark
Maxwell, James Clerk	1831	1879	Scotland
Tait, Pere: Guthrie	1831	1901	Scotland
Crookes, William	1832	1919	England
Notanian, Carl Gottfried	1832	1925	Itussia
Clebsen, Rudolf Friedrich, Africa	1833	1872	Germany
Meyer (Oskar Find)	1834	1909	Germany
Beltrami, Engenio	1835	1900	Italy

Newcomb, Simm	1835	1909	USA
Stefan, Josef	1845	1893	Austria
Mascart, Éléuthere, Élie Nicolas	1847	1908	France
van der Wards, Johannes Dieferik	1847	1923	The Netherlands
Mach Erns)	1838	1956	Slovakia
Morley Edward William	1838	1923	USA
Hankel, Hermann	1830	1873	Germany
Kundt, August Adolf	1830	1894	Germany
Abbe, Ernst Karl	1840	1905	Germany
Kohlrausch, Friedrich	1810	1910	Germany
Corm. Marie Allred	1811	1902	Indaud
Pochhammer 1.69 August	1841	1920	Germany
Boussingsq. Valentin desepte	1842	1929	France
Lie. Marins Sophus	1842	1899	Nieway
Reynolds, Osburne	1842	1912	England
Strutt, John William (Thiof Baron Rayleigh)	1842	1919	England
Christiansen, Christian	1843	1917	Austria
Boltzmann, Ludwig	1811	1906	Austria
Branly, Edoquid Fagene Désiré	1841	1910	France
Lippusana, Galariel	1845	1921	France
Burgton, Whilhem Contact	1845	1923	Germany
Umoy, Nikolai Alekseevira	1846	1915	Russia
Mittag-Leffler, Gösta Magnus	1846	1927	Sweden
Castighano, Carlo Alberto	1847	1884	haty
Formet, Gaston	1847	1920	France
Bruns, Eras, Heinrich	1845	1919	Corrections
Korteweg, Diederik Julganes,	1848	1911	The Netherlands
Rowland Henry Augustus	1848	1900	USA
Hopkinson, John	1849	1898	England
Lamb. Horace	1819	1934	Lagland
Cerruti, Valentino	1850	1909	Itals
Goldstein Engen	1850	1939	Poland
Gray, Thomas	1850	1908	Seathand
Heaviside, Olivier	1850	1925	England
Milne, John	1850	1913	England
Voigt, Woldeman	1850	1919	Germany
Bartoh, Adolfo	1851	1896	halv
Fitzgerald, George Francis	1851	1901	heland
Huganiat Pierre Here:	1851	1887	France
Lodge, Oliver Juseph	1850	1210	England
Michelson, Mbrit	1852	1931	Gormany
Poynting, John	1852	1914	England
Lorentz, Hendrik Aatsool	1853	1928	The Netherlands
Meyer Max WEndm	1853	1910	Germany
Pum are, Jules Henry	1851	1912	France
Curie Jamus	1855	1211	France

Ewing, James Affred	1855	1935	Scuthard
Hall, Diwin Herbert	1855	1938	153
Sekiya, Selži	1855	1896	եւթու
Knott, Cargill Gilston	1856	11:22	Scotland
Runge, Carl David Tolmé	1856	1927	Germany
Thurison, Joseph John	1856	1940	England
Hertz, Ihrinrich Budulf	1857	1894	Germany
Latuor, Joseph	1857	1842	In land
Mohorovičić, Andrija	1857	1936	Cioatia
Oldham, Richard Dixon	1858	19:36	Inland
Planck, Max	1858	1947	Germany
Ceshro, Equisio	1859	1906	ltaly
Cipie Pierre	1859	1906	France
Reid, Harry Fielding	1859	1941	115.3
Chies, Charles	1860	1928	England
Sonigliana, Carlo	1860	1955	Italy
Volterra, Vito	1860	1940	ludy
Kennelly, Arthur Edwin	1860	1939	Inder
Reuber-Paschwitz Frost voc	1861	1895	Lithuana
Wirehest Emil	1861	1928	Lithuanéa
Hilbert, David	1862	1943	Germany
Lenard, Phillipp	1862	1947	Huigary
Rudzki, Manyey Pus	1862	1916	Poland
Wiener, Otto Henrich	1862	1927	Germany
Love, Augustus Edward Hough	18433	1940	England
Michell John Henry	18433	1940	Vastraha
Pérus, Iran Bajitiste Alfred	1863	1925	France
Mittkowski, Hermann	1861	1909	Germany
Wien, Willom Carl Werton Ono Fritz Franz-	1861	1928	Germany
Hadamard, Jacques Salomon	1865	1963	France
Pockels Friedich Carl Alway	1865	1963	Italy
Zeeman, Pirte:	1865	1943	The Netherlands
Cosserat, Engline Manrice Pierre	1866	1931	France
de Vries, Gastav	1866	1934	The Netherlands
Fabry, Marie Paul Auguste Charles	1867	1945	France
Kolosov, Gury	1867	1946	Bussia
Kutta, Willelm	1867	1941	Germany
Hale, George Ellery	1868	1938	1.83
Mie, Gustav	1868	1957	Generativ
Millikan, Robert Andrews	1868	1953	1183
Omora Eusakiehi	1868	1923	Japan
Sabine, Wallace Clement	1868	1919	1.83
Sommerfeld, Arnold Johannes	1868	1951	Bussia
Galerkin, Bous Grigonevich	1873	1945	Bussa
Rytherford Fruest	1871	1937	New Zealand
Langivin, Paul	1872	1946	France
Tranges and the second	12012	13140	1.211-1.

Levi-Civita, Inflao	1874	19.11	Italy
Whittaker, Lemmal Taylor	1873	1956	England
Marconi Gughelmo	1874	19.37	Pain
Praudtl Ludwig	1805	1953	Germany
Augenheister, Gustav	1878	1945	Germany
Freehett, Manufree René	1878	1973	France
Fitnoshet.ku, Stephet.	1878	1972	En t-citt-r
Mintrop. Indger	1880	1956	Germany
Wegener Affred	1880	1930	Germany
Einstein Albert	1879	1955	Gormany
Heightz, Gustav	1881	1953	Austria
Zneppritz Karl	1881	1908	Germany
Bateman, Harry	1882	19.16	England
Born, Max	1882	1970	Polatel
Geiger, Ludwig Carl	1882	1966	SwitzerLand
Macelwane Jaynes Bornard	1883	1956	USA
Misrs, Richard con	1883	1953	USA
Terzaghi, Karl you	1883	1963	Czech Republic
Debye, Prier Juseph William	1884	1966	The VetLerlands
Weyl, Hermann Klans Hugo	1885	1955	Germany
Taylor, Geoffrey Ingtain	1896	1975	England
Loomis, Affred Lee	1887	1975	USA
Rodors Johann	1887	1956	 Czoch Republic
Schrödinger, Erwin	1887	1964	Austria
Courset, Richard	1888	1972	Poland
Lohmann, higi	1888	1953	Domaak
Ranani, Chandrosekhara Venkata-	1888	1970	India
Brillouin, Léon	1889	1.050	Russia
Gutenberg Bruz	1889	19950	Germany
Hubble, Edwin Powell	1889	1953	USA
Nyquist, Harry	1889	1976	Sweden
Jeffreys, Harnhl	1891	1989	England
de Broglie, Louis Vietor	1892	1987	France
Watson-Watt, Robert	1892	1973	France
Gordon, Walter	1893	19140	Germany
Knadsen, Vern Oliver	1893	1:171	USA (
Laarzus, Cornollius	1893	1974	Hurngary
Frenkel, Yaraw Illich	1894	1952	Itussia
Klein, Oskar	1894	1977	Sweden
Kramers, Headrik Anthony	1894	1952	The Netherlands
Stoneley, Robert	1891	1476	England
Wager Nothert	1391	1:151	USA
Burgers, Johannes Martinus	1895	1981	The Netherlands
Hand, Friedrich	1896	19.97	Germany
Blackott, Patrick Maynard Struct-	1897	1974	England

Sokolov, Sergei	1897	1971	Russia
Firestone, Floyd	1898	1986	1°SA
Fuck, Vladimir Meksamhovich-	1898	1974	Russia
Sears, Francis Weston	1868	1975	USA
Wentzel, Gregor	1898	1978	Germany
Richter, Clubles Frances	1960	1985	1.83

Multicational back an effect an instance constant indeptioning of her forces, if her does not know the matural back in terms of measurement and non-need relations. There also has the strength of the matural catelling need which measures and deerworks according to such knowledge. Knowledge and compactic usion are the joy and postification of homometry they are parts of the matural needby, often a replacement for these materials that watter has all too specify dispersed. These very peoples who are there in the endered that watter has all too specify dispersed. These very peoples who are there in formed industrial activity in application of mechanics and technical to material to general industrial activity in application of mechanics and technical to material does not permute all classes, will meritably define in prospective all the non-section meriphering states, in which sciences and the industrial articles are not reclationship, proposes with people's come and the industrial articles are not reclationship, proposes with people's come and the industrial arts have an active intervalue and people science and the industrial activities.

Mexander von Humboldt (Kosmos, I-1845, 30).

This page intentionally left blank

Bibliography

- Aboučí, J., 1991. Mechanics of composite materials, a unified micromechanical approach. Studies in Applied Mechanics, 29, Elsevier Science Publ. Co. Inc.
- Achenbach, J. D., 1984, Wave propagation in elastic solids, North Holland Publ. Co.
- Viller, I., and Nagy P. B. (1994) Measurements of acoustic surface waves on fluid-filled porous recks: J. Graphys. Res., 99, 17863 17869.
- Aki, K., and Richards, P. G., 1980, Quantitative seismology, W. H. Freemar, & Co.
- Alekseev, G. N., 1980, Energy and entropy, Mir Publishers
- MIord, R. M., 1986. Shear data in the presence of azumuthal axis tropy: Difley, Texas: 59th Ann. Internat. Mtg. Sov. Expl. Graphys., Explanded Abstracts, 176–179
- [Mlard, J. F., 1993, Propagation of sound in porous media, Elsevier Science Publ. Coline
- Alterman, Z., and Karal, Jul. F. C. 1998, Propagation of elastic waves in layered media by finite chference methods: Bull Scism. Soc. Am., 58(1), 367-398.
- Aoki, T., Tan, C. P., and Riamford, W. E. 1993, Effects of deformation and strength anisotropy on borelades fidlures in saturated rocks: Internat. J. Rock Mech. Min. Sci., 30, 1031-1034.
- Arntsen, R., and Carcione, J. M., 2000. A new insight into the reciprocity principle, Geophysics, 65, 1604-1612.
- Arntsen, R., and Careinar, J. M., 2001. Numerical simulation of the Bint slow wave in water saturated Nivelsteiner sandstone: Geophysics, 66, 800–806.
- VUS, B. J., 1993, A study of general anisotropic elasticity in rocks by wave propagation theoretical and experimental aspects: Ph.D. thesis, Paris University.
- Atts, R. J., Rasolotosion, P. N. J., and Zuiszner, B. U. 1992, Experimental determination of the complete anisotropy viscoelastic tensor in rocks: 62nd Ann. Internat. Mug. Soc. Expl. Grophys., Expanded Abstracts, 636-639.
- Asmoy, L. 1972. Asimov's guide to science. Penguin books.
- Anhl, B. A., 1990a, Acoustic fields and waves in solids, Vol.1, Krieger Publ. Cu

- And, R. A., 1990b, Acoustic fields and waves in solids, Vol II, Kneger Publ. Co.
- Aurasht, J. L., Borne, L., and Chambon, R., 1985. Dynamics of process saturated media checking of the generalized law of Darry: J. Acoust. Soc. Aus. 77, 1641-1650.
- Backus, G. E. 1962. Long-wave clastic anisotropy produced by horizontal layering. J. Geophys. Res. 67, 1427-1110
- Backus, G. E., 2070, A geometrical picture of an software closure tensors: Revs. Geophys. Space Phys., 8 (633-67).
- Bayley, R. L., and Forvik, P. J. 1986. On the fractional calculus model of viscoelastic behavior. J. Rheology, 30, 133-155.
- Bakulin, A., and Moloktov, L., 1997. Generalized anisotropic Biot model as an effective mailed of stratified pornolastic mediume 50th Ann. Internat. Mag. Europ. Assoc. Expl. Genghys., Expanded Abstracts, 1965.
- Bano, M., 2004. Modelling of GPR waves for lossy media obeying a complex power law of frequency for dielectric permutivity. Geophys. Prosp. 52, 11-26.
- Baste, S., and Andorr, R., 1991, On internal variables in anisotropic domage. Fur. J. Mich. A/Solids 10, 587-606.
- Batzle, M., and Wang, Z., 1992, Scismic properties of pure fluids: Geophysics, 57, 1396-1408.
- Bayhss, A., Jozdan, K. F., LeMesquer, B. J., and Tipkel, F., 1986, A fourth-order accurate finite difference scheme for the computation of clastic waves: Bull. Seism Suc. Am., 76, 1115–1132.
- Baysal, E., Kosloff, D., and Shewood, J. W. C., 1984, A two-way nonreflecting wave equation: Geophysics, 49, 132-141
- Breker, F. L., and Richardson, R. I. (1972). http://www.ofunatorial.properties on Rayleighcritical.org/denoffectivity. J. Acoust. Soc. Am., 51, 1609-1617.
- Belliore, I., and Caputo, M., 2000. The experimental set-valued index of refraction of thelectric and anelastic media: Annalish Geolisica, 43, 207-216.
- Beltzer, A.J., 1988, Acoustics of solids, Springer-Verlag,
- Ben Monahem, A., and Singh, S. G., 1981, Scistule waves and sources, Springer Verlag,
- Ben Menahem, A., and Gibson, R. L. 1993, Directional attenuation of SH-waves in anisotropic porcelastic media: J. Acoust. Soc. Am., 93 (20)7-3065.
- Birmoulli, J. (1632, Lectiones mathematicae de methodo integraliani ahisque conscriptar în usum ill. Marchionis Hospitalii, Opera otniba, Paris, III, 486-558
- Berryman, J. G., 1979. Long-wave elastic anisotropy in transversely isotropic medic Geophysics, 44, 896-957.

Berryman, I. G., 1980, Conbringtion of Biot's theory: Appl. Phys. Lett. 37, 482-384.

- Berryman, J. G., 1992. Effective stress for transport properties of inhomogeneous parms mck. J. Geophys. Res. 97, 17409 17424.
- Bortyman, J. G., Milton, G. W., 1991. Exact results for generalized Gassuram's oppation in composite pursus media with two constituents: Gouphysics, 56 (1950)1960.
- Berryman, J. G., and Wang, H. F., 2000. Electre wave propagation and attenuation in a doubly-percessive dual-permeability medium: Internat. J. Rock. Medi. & Min. Sci., 37, 63-78.
- Bétti, I., 1872. Icoria dell'elasticità. Il Nuovo Cimento, Sezione 6(7-8), 87-97
- Biot, M. V. 1952. Propagation of elastic waves in a cylindrical base containing a fluid: J. Appl. Phys. 23, 997-1005
- Biot, M. A., 1954. Theory of stress-strain relations in anisotropic viscoelasticity and relaxation phenomena, J. Yapl. Phys. 25, 1385-1391.
- Biot, M. A., 1955. Theory of elasticity and consolidation for a potens anisotropic solid. J. Appl. Phys., 26, 182-185.
- Biot, M. A., 1956a. Freezy of propagation of elastic waves in a finite-saturated points solid. I. Low-frequency tanget J. Acoust. Soc. Ann. 28, 108-178.
- Biot, M. A., 1956b. Theory of propagation of elastic waves in a dipid-saturated paragsulid. II. High-frequency range J. Acoust. Soc. Am., 28, 179-193.
- Biot, M. A., 1956c. Theory of deformation of a porous viscoelastic anisotropic solid: J. Appl. Phys., 27, 159-167.
- Biot, M. A. 1992, Mechanics of difformation and acoustic propagation in porcets matha, 4 (Appl. Phys., 33,)482-1498.
- Biot, M. A., and Willis, D. G., 1957. The elastic coefficients of the theory of consolidation: J. Appl. Mech., 24, 594-604
- Blanch, J. O., and Robertsson, J. O. A., 1997, A moduled Lax-Werdroff correction for wave propagation or media described by Zener elements: Grophys. J. Internat-111, 381–386
- Bland, D. R., 1960. The theory of linear viscoelasticity, Pergamon Press Inc.
- Bland, D. R., 1988, Wave theory and applications, Clarendon Press,
- Bleistein, N., 1984, Mathematical methods for wave phenomena. Academic Press Inc.
- Bohatski, N. N., 1983. Generalized reaction principles and reciprocity theorems for the wave equation, and the relationship between the time-advanced and time-tetatlod fields: J. Acoust. Soc. Am., 74, 281-285.

- Boltzmann, L., 1874. Zur theorie der clastischen nachwirkung, Sitzungsber, Kaiserlich, Akad. Wiss. Wien Math.-Naturwoss., K1 70, 275-306 (also Yol. 4, p. 167 of his Collected Papers (1999)).
- Bonnet, G., 1987, Basic singular solutions for a potoelastic medium in the dynamic range 4, Acoust. Soc. Ann. 82, 1758-1762.
- Bucker, H. G. 1992. Energy in Flectromagnetism. IFF Electromagnetic Waves Series 13.
- Borcherdt, R. D., 1973, Raylergh-type on face wave on a linear viscositistic half-space J. Acoust. Suc. Am. 54, 1651-1653.
- Borcherdt, R. D., 1977. Reflection and reflacmon of type-II/S waves in elastic and anelastic media: Bull. Seism. Soc. Am. 67, 43-67.
- Burcherdt, R. D., 1982. Reflection-refraction of general P- and typed Sowaves in elastic and anelastic solids: Geophys. J. Roy. Astr. Soc., 70, 621-648.
- Burchenft, B. D., Glassmaver, G., and Weimerberg, L. (1986). Influence of welded boundaries in anelastic media memory. Boy: and characteristics of *P. S.* I and *S.* II waves: observational evidence for inhomogeneous loady waves in low-loss solids. J. Geophys. Res., **91**, 11501-11118.
- Burcherdt, R. D., and Wennerberg, L., 1985, General P. type-I S. and type-II S waves in anclastic solids: Inhomogeneous wave fields in low less solids. Bull. Science Soc. Ann. 75, 1729-1763.
- Burejku, P., and Ziegler, F., 1988. Surface waves in an isotropic viscorlastic half-space. The method of generalized cays. in Packer D. F., and Mangia, G. A., Eds., Berent developments in surface accustic waves. Springer-Verlag, 299-308.
- Burn: M., and Walf, F., 1964. Principles of Optics: Pergamon Press Inc.
- Bourbié, T. Coussy, O., and Zinszner, B., 1987 Acoustics of porous media, Editions Jochnup
- Binitin, C., Bornet, G., and Bard, P. Y., 1987. Green's functions and associated sources in infinite and stratified percolastic media: Geophys. J. Roy. Astr. Soc. 90, 521-550.
- Bracewell, R., 1995, The Fourier transform and its applications, McGraw-Hill Bonk Co.
- Brand, L., 1957, Vector analysis, John Wiley & Sons
- Brekhovskikh, L. M., 1960, Waves in layered media. Academic Press Inc.
- Broger, S. and Onat, E. T. 1964, On the determination of free energy in linear viscoelastic solids: Z. Angew. Math. Phys. 15, 184-190.
- Brewster, D., 1815, On the laws that regulate the polarisation of light by reflexion from transparent bodies: Phil. Trans. Roy. Soc. London, Part I, 125-159.

Brillonin, L., 1969, Wave propagation and group velocity. Academic Press,

- Bioda, E., 1984, Lidwig Belizmana, man. physicist, philosopher, Ox How Press,
- Brown, B., and Korringa, L. 1975. On the dependence of the clastic properties of a potons rock on the compressibility of the nore finid: Geophysics, 40, 628-616
- Binggenan, D. A. G., 1937, Derechmängen der verschiedener physikalischen Konstanten von heterogenen Substanzen. III: Die elastischen Konstanten der unasinsotropen Mischkraper aus isatropen Substanzen. Aufalen der Physik 29, 160-178.
- Brugger, K., 1965, Pure modes for elastic waves an revstal: J. Appl. Phys. 36, 759-768.
- Brutsbert, W., 1964. The propagation of elastic waves in unconsolidated uncotinated granular medium: J. Grophys. Res., 69, 243-257
- Buchen, P. W., 1977a, Plane waves in linear viscoelastic media: Grouples / J. Roy, Astr Soc., 23, 531-542
- Buchen, P. W., 1971b. Reflection, transmission and diffraction of SH-waves in linear viscuelastic solids: Grophys. J. Rey. Astr. Suc., 25, 97-113.
- Burneige, R., Chadwick, P., and Norris, A., 1993. Fundamental clastudynamic solutions for anisotropic media with ellipsoidal slowness surfaces. Proc. Roy. Soc. London, Ser. A, 440, 655-681.
- Burndie, R., de Hoop, M. Y., Le, L. H. T., and Norris, A., 1993. Waves in stratified viscoelastic mesha with microstructure J. Access. Soc. Am. 94, 2884-2895.
- Burndge, R., and Keller, J. B., 1985, Powelasticity equations derived from uncrostructure J. Access. Soc. Am., 105, 626-632
- Bignidge, R., and Vorgas, U. A., 1979. The fundamental solution in dynamic poroclastimity: Geophys. J. Roy. Astr. Soc., 58, 61-90.
- Cadoret, T., Marian, D., and Zinszner, B., 1995. Influence of frequency and third distribution on clastic wave velocities in partially saturated linestones: J. Geophys. Res., 100, 9789–9803.
- Cagniard, L., 1939. Reflexing et rifraction des males sitismiques progressives. Gauthier Villats.
- Cajori, F., 1929, A history of physics, McMillan Co.
- Caloi, P., 1918, Comportamento delle onde di Rayh igh in un mazzo fittuo elastico in definito: Annali di Geobsica, I, 550-567.
- Cannon, J. T., and Dostrovsky, S., 1981. The evolution of dynamics (vibration theory from 1687 to 1742). Studies in the History of Mathematics and Physical Sciencies 6. Springer-Verlag.

- Capito, M. 1998. The set valued unified model of dispersion and attenuation for wave propagation in dedectro- and anelastic functia: Ani.ali di Golisica, 41, 653-656.
- Unperto, M., and Mamardi, F. (1974) Linear models of dissipation in anelastic solids: Rev. Nuovo Cimento (Ser. II) 1, 161-198
- Carcinen, J. M. 1990. Wave propagations in anisotropic linear viscoelastic media: theory and simulated wavelields: Grophys. J. Internat., 101, 739-750. Erratum: 1992. 111, 191.
- Carcione, J. M., 1991, Domain decomposition for wave propagation problems: J. Sci. Comput., 6, 453-472.
- Carcinge, J. M., 1992a, Modeling anelastic singular surface waves in the earth: Conphysics, 57, 781-792
- Carcinni, J. M., 1992b. Rayleigh waves in isotropic viscolastic media: Geophys. J. Internat., 108, 453-454. Erratum, 1992, 114, 191.
- Catelone, J. M., 1992e, Anisotropic Q and velocity dispersion of linely layered media Geophys. Prosp., 40, 761-784.
- Carcione, J. M., 1994a, Wavefronts in dissipative anisotropic and act Grophysics, 59, 6414657
- Carcium, J. M., 1994b. The wave equation in generalized coordinates. Grophysics, 59, 1914–1915.
- Carcione, J. M., 1995, Constitutive model and wave equations for ibnear, viscoelastic anisotropic media: Geophysics, 60, 537-548.
- Farcone, J. M., 1996a, Rostodynamos of a non-adval interface. Application to crack and fracture scattering: J. Geophys. Res. 101, 28177-28188.
- Carcinne, J. M., 1996b, Wave propagation in anisotropic, saturated porms media: plane wave theory and minorical simulation. J. Woost, Soc. Am. 99, 2655–2665.
- Carcinar, J. M., 1996). Ground perirtrating radat: Wave theory and immerical similar tions in lossy anisotropic media. Geophysics. 61, 1063-1677.
- Carcione, J. M., 1997a, Reflection and refraction of anti-plane shear waves at a plane boundary between viscoelastic anisotropic media: Proc. Roy. Soc. London. Ser. A, 453, 919-932.
- Carcinne, J. M., 1997b. Reflection and transmission of qP-qS plane waves at a plane homolary between viscoelastic transversely isotropic media. Grophys. J. Internat., 129, 169-680.
- Carcione, I. M., 1998, Associastic effective rheologies for modeling wave propagation in porous metha: Geophys. Presp., 46, 249-270

- Caralone, J. M., 1999a, On energy definition in electromagnetism: an analogy with viscoelysticity J. Acoust. Soc. Am. 105, 626-632.
- Carcione, J. M., 1999b. The effects of vector attenuation on AVO of aff short reflections Geophysics, 64, 815-819.
- Carciano, J. M., 1999c. Staggood mesh fig the anisotropic and viscoelistic wave equation: Geophysics, 64, 1863-1866.
- Carcione, J. M., 2001a, Energy balance and fundamental relations in dynamic anisotropic parasynance/asticity. Proc. Roy. Soc. London, Sci. A, 457, 33i-338
- Caraone, J. M., 2001b. Amplitude variations with offset of pressure-scal reflections Grouphysics, 66, 283-293.
- Carciano, J. M., 2006. A sportral numerical method for electromagnetic diffusion. Geophysics, 71, 11-19.
- Carciane, J. M., and Cavallini, F. (1963, Fuergy balance and fundamental relations in anisotropic-viscorlastic media: Wave Motion, 18, 11-20.
- Cardione, J. M., and Cavalhai, E., 1994a. A semi-analytical solution for the propagation of pure shear waves in disservative nonoclean media: Acoustics Letters 17, 72-76
- Categore, J. M., and Cavallia', F., 1994b. A theological model for anelastic anisotropic media with applications to seismic wave propagation. Geophys. J. Internat., 119, 338-348.
- Categoro, J. M., and Cavallini, F., 1995a. Forhidden directions for informogeneous proshear waves in dissipative anisotropic media. Grophysics, 60, 522-540.
- Carciano, J. M., and Cavallini, F. (1995h) On the secretic electromagnetic analogy: Wave motion, 21, 149–162
- Carcione, J. M., and Cavallini, I., 1995c. The generalized SII-wave oppation: Geophysics 60, 549-555.
- Concone, J. M., and Covallim, F., 1995d. Attentiation and quality factor surfaces in anisotropic-viscoclastic media: Mech. of Mat., 19, 411-327
- Carcione, J. M., and Cavallui, F., 1997. Forbidd rudirections for TEM waves in anisotrupic conducting media. IEEE Trans. Antennas and Propagat., 45, 133-139.
- Carenone, J. M., and Cavalluci, F. 2001, A semi-analytical solution for the propagation of electromagnetic waves in 3-D lossy orthorripic media. Geophysics. 66, 1141-1148
- Carcione, J. M., Cavallini, F., and Helleg, K., 1998, Ansotropic attenuation and material sympetry: Acustica, 84, 495-502
- Carcione, J. M., Cavallini, F., Mainardi, F., and Hanyga, A. 2002. Fine-domain modeling of constant-Q seismic waves using tractional derivatives: Pure Appl. Geophys. 159.

- Carcione, J. M., Cavallini, F., Santos, J. E., Ravazzoli, C. L. and Gauzellino, P. M., 2004, Wave propagation in partially-saturated porous media. Signification of a second slow wave: Wave Motion, 39, 227-240.
- Carcinni, J. M., Finetti, L. and Gri, D., 2003. Seismir modeling study of the Farth's deep crust. Grophysics, 68, 656–664.
- Carcione, J. M., and Gaugi, A. F., 1998, Habinet's principle for elastic waves: wnumerical test. J. Access. Soc. Am., 105, 1485-1402.
- Carciner, J. M., and Grany, A., 2000a. Non-republiciting compacting and abnormal pore-Built pressures: effects on ruck properties. Grophys. Prosp. 48, 521-537.
- Catelone, J. M., and Gaugi, A., 2000b. Gas generation and overpressure: effects on seismic attributes: Geophysics, 65, 1769-1779.
- Concrone, J. M., and Gei, D. (2003) A session modeling study of a subglacial lake, Geophys. Prosp. 51, 501-515
- Carcinar, J. M., Gurevich, B., and Cavallini, F., 2000, A generalized Birt Gassmann mulel for the acoustic properties of shaley samistones. Grophys. Prosp. 48, 539– 557.
- Carcione, J. M., and Helleg, K., 2001, Wave polarization in transversely-isotropic and orthorhorith media: in Hoad, J., Ed., Advances in Anisotropy Scherol Theory, Mudeling, and Case Studies. Society of Exploration Grophysicists, Open File Publication 5, 289–322.
- Carcione, J. M., Hellog, K., and Helle, H. B., 2003. Effects of pressure and saturating fluid on wave velocity and attenuation in anisotropic rocks: Int. J. Rock Misch, Min. Sci., 40, 389-403.
- Carcium, J. M., and Hello, H. R., 1999, Numerical solution of the poroviscoelastic wave equation on a staggered mesh. J. Comput. Phys., 454, 520-527.
- Carcione, J. M., and Helle, H. B., 2004. On the physics and simulation of wave propagation at the ocean bottom: Grophysics 69, 825-839.
- Carcinny, J. M., Helle, H. B., and Gauge A. F., 2006. Theory of barehule stability when drilling through salt formations: Geophysics.
- Carcion , J. M., Helle, H. B., and Pham, N. H., 2003. White's model for wave propagation in partially sangated rocks. Comparison with poroclastic numerical experiments: Geophysics, 68, 1389-3398
- Carciner, J. M., Helle, H. B., Santos, J. E., Ravazzoli, C. L. 2005, A constitutive equation and generalized Gassmann modulus for multimineral pouries modia: Genphysics, 70, N17-N26.
- Cantione, J. M., Kosloll, D., and Behle, A., 1991, Long wave automopy in stratilied media: a numerical test: Gosphysics, 56, 245-251.

- Caratone, J. M., Kosloll, D., Beide, A., and Seriani, G., 1992, A spectral scheme for wave propagation simulation in 3-D elastic-anisotropic metra: Geophysics, 57, 1593-1607.
- Carcione, J. M., Kosloff, D., and Kosloff, R. (1988). Wave propagation simulation in an anisotropic (transversely isotropic) medium: Q. J. Mech. Appl. Math. 41, 320-345.
- Carcione, J. M., Kosloff, D., and Kosloff, R., 1988b, Wave propagation simulation in a linear viscoacoustic mediance Grophys. J. Roy. Astr. Soc. 93, 303-407. Environ-1988, 95, 612.
- Carciono, J. M., Kusloff, D., and Kosluff, R., 1988c, Wave propagation simulation in a linear viscoelastic medium. Geophys. J. Roy. Astr. Soc., 95, 597-611.
- Categore, J. M., Kosloff, D., and Kosloff, R., 1988d. Viscoacoustic wave propagation simulation in the earth. Grophysics, 53, 769-777.
- Categore, J. M., and Poletto, F., 2000. Simulation of stress waves in attenuating drift strings, including plezoelectric sources and sensors: J. Acoust. Soc. Am., 108, 53-64
- Carcione, J. M., Poletto, L., and Gei, D., 2004, 3-D wave simulation in and astic media using the Kelvin-Voigt constitutive equation: J. Comput. Phys. 196, 282-297
- Carenone, J. M., and Qairogu-Gende, G., 1996, Some aspects of the physics and manufield modeling of Biot compressional waves: J. Comput. Acoust., 3, 261–280.
- Carciene, J. M., Qairuga Gonde, G., and Cavallini, F. (1966). Wavefronts in dissiparive anisotropic media: comparison of the plane wave theory with manerical modeling: Geophysics, 61, 857-861.
- Carcione, J. M., and Rebinson, E. A., 2002. On the acoustic-electromagnetic analogy for the reflection-refraction problem: Studia Geoph. et Geod., 46, 321-345
- Carcione, J. M., Santos, J. E., Ravazzoh, C. I., and Helle, H. B., 2003, Wave simulation in partially forzen porors modia with fractal freezing conditions. J. Appl. Phys. 94, 0830-7847
- Carcione, J. M., and Seriano, G. (200). Wave simulation in fozzen sediments: J. Comput. Phys., 170, 1-20.
- Carciany, J. M., and Schernberg, M., 2000, 3-D ground-penetrating radia simulation and plane wave theory. Grouphysics, 65, 1527 1541.
- Carroll, M. M., 1979, An effective stress law for anisotropic elastic deformation: J. Grophys. Res., 84, 7510-7512.
- Carroll, M. M., 1980, Mechanical response of fluid-saturated potous materials: or Rimrott, I., P. J., and Tabarrok, B., Iols, Theoretical and Applied Mechanics, 15th Int. Cong. Theoretical & Appl. Mech., 251-262.
- Carslaw, H. S., and Jaeger, J. C., 1959. Conduction of heat on solids: Clatendon Press.

- Casula, G., and Carcione, J. M., 1992, Generalized mechanical model analogies of linear viscoelastic behaviour: Boll. Geofis. Lear. Appl., 34 (235):256.
- Cavalhui, F., 1999. The best isotropic approximation of an anisotropic elasticity reason. Boll. Geoix. Teor. Appl. 40, 1-18.
- Cava@en, F., and Carrione, J. M., 1994, Energy balance and inhomogeneous plane-wave analysis of a class of anisotropic viscoelastic constitutive laws, or Rimmo, S., and Rungeri, F., Eds., Waves and Staluhty in Continuous Media, World Scientific, 17-53.
- Caviglia, G., and Morro, A., 1992. Inhomogeneous waves in solids and thirds. World Scientific
- Caviglia, G., Miario, A., and Pagani, E. (1989). Reflection and refraction at elastic viscoelastic interfaces: II Nuovo Cimento, 12, 399-413.
- Cederbaum, G., Li, L., and Schulgasser, K., 2000. Puroclastic structures. Elsevier Science Ltd.
- Celia, M. A., and Gray, W. G., 1992. Numerical methods for differential equations: Englanced al concepts for scientific and engineering applications. Prentice-Hall Inc.
- Cerjan, C., Kosloff, D., Kosloff, R., and Reshef, M., 1985, A nonreflecting boundary condition for discrete accorstic and elastic wave expansions. Grephysics, 50, 705-708.
- Cervený, V., 2001, Seismie nay theory, Cambridge Univ. Press.
- Cervergi, V., and Psencik, L. 2005a, Plane waves in viscoelastic anisotropic media. Part 19 Heavy: Graphys. J. Internat. 164, 197-212
- Cervený, V., and Psenaik, I. (2005). Pione waves in viscoelastic ansotropic media. Part 2: Numerical examples: Coopless. J. Internat. 161, 213-229.
- Chadwick, P. (1989) Wave propagation in transcensely isotropic elastic wider. II Surface waves: Proc. Roy. Soc. London. Soc. A, 422, 65-101
- Chandler, R. N., and Johnson, D. J. (1984) The equivalence of quasi-static flow in fluidsaturated parons media and Biot's slow wave in the limit of zero frequency. J. Appl. Phys., 52, 3391-3395.
- Chapman, C. H., 1993. Reflection/transmission coefficients reciprocities in anisotropic media: Geophys. J. Internet. 116, 198-501.
- Cheng, A. H. D., 1997, Material coefficients of anisotropic porcelasticity: Int. J. Rock Mech. Min. Sci., 34, 199-205.
- Chew, W. C., 1990, Waves and fields in infomogeneous media, van Nostraud Reinhold,
- Chew, W. C., and Luc, Q. H., 1996. Perfortly matched layers for classosynamics: a new absorbing boundary condition. J. Comput. Acoust. 4, 341-359.

- Chasti, S., and Krebes, E. S., 2000, Exact and approximate formulas for P = SU reflection and transmission coefficients for a nonvelocid contact interface: J. Geophys. Res. 105, 28045-28054.
- Chu, R. C. Y., 1980. Wave propagation in viscorlastic media on Dziewonski, A. M., and Boschi, L., Eds., Physics of the Earth's Interior, Proceedings of the International School of Physics of mice Fermi 7, Course 78, 213-236.
- Christensen, R. M., 1982. Theory of viscoolasticity: Academic Press Inc.
- Clayton, R., and Engquist, B. (1977) Absorbing boundary conditions for acoustic and elastic wave equations: Bull. Seism. Soc. Ana. 67, 1529-1540.
- Cole, K. S., and Cole, R. H., 1911, Dispersion and absorption in dielectrics. I. Alternating current el atacteristics: J. Chem. Phys. 9, 341-351.
- Comper, H. F., Jul 1967. Reflection and transmission of oblique plane waves at a plane interfacet 5, Acoust. Soc. Am., 42, 1064-1069.
- Coraprioght, M. Y., and Tenray, K., 1996. Propagation of waves in preons media: in Coraprioght, M. Y., Ed., Advances in Pormis Media, 3, Elsevier Science Publ. Cu-Inc., 361-110.
- Cornu, A., 1899, La théore des ondes humanises son influence sur la physique moderne (The Rude Lecture, June 18) 1899): Tracs. Cambridge Phil. Soc., 18, xvii-xvvii.
- Cosserat, E., and Cosserat, F., 1939. Theorie des corps deformables, Hermann et Fils.
- Conssy, O.: 1995. Mechanics of porcus media, John Wiley & Sons,
- Cowin, S. C., 1999. Bone porcelasticity: dormal of Biomerbanics, 32, 217-238
- Crampin, S., 1981. Vacview of wave motion in anisotropic and cracked clastic-media. Wave Motion. 3, 413-381
- Cristeson, N., 1986, Ruck thendogy, Kluwer Academic Publ.
- Currie, P. K. 1979, Viscoelastic surface waves on a standard linear solid: Quart. Appl. Math., 37, 332-336
- Currie, P. K., Hayes, M. A., and O'Leary, P. M., 1977, Viscoelastic Ravleigh waves, Quary Appl. Mach., 35, 35-53.
- Currie, P. K., and O'Leary, P. M., 1978, Viscoelastic Rayleigh waves 11: Quart. Appl. Math., 35, 445–151.
- Dablam, M. A., 1986, The application of high-order differencem to the scalar wave equation: Geophysics, 51, 54-66.
- Dan, N., Vafidis, A., and Kanasewich, E. R., 1995, Wave propagation in heterogeneous porous media: A velocity-stress, mate-difference method: Geophysics, 60, 327-340.

- Daley, P. E., and Hron, F., 1977. Reflection and transmission coefficients for transversely isotropic media: Bull. Sesua. Soc. Am., 67, 661-675.
- Dampier, W. C. (1964) A history of science and its relations with phylosophy and religion Cambridge Univ. Press.
- Daniels, D. L. 1996. Surface-petierrating radat. IEE radat, sonar, navigation and avionles series, 6.
- Darcy II 1856, Les fontaines publiques de la ville de Dijon, Dahnout, Paris
- Dattuli, G., Torre, A., and Mazzarumiti, G., 1998, An alternative point of view to the theory of fractional Fourier (tausform: J. Appl. Math., 60, 215-224.
- Day, S. M., 1998. Efficient simulation of constant Q-using coarse-gradued memory variables: Bull. Seisn. Soc. Am., 88, 1051-1062.
- Day, S. M., and Minster, J. B., 1984. Numerical simulation of attennated wavefields using a Padi approximant method: Geophys. J. Roy. Astr. Soc., 78, 105 (18).
- Debye, P., 1929, Polar molecules, Dover, New York
- de Groot, S. R., and Mazur, P., 1963. Thermodynamics of inteversible processes. North Holland Publ. Co.
- de Huop, A. F., 1966. An elastudynamic reciprocity theorem for linear, vascoelastic media: Appl. Sci. Res., 16, 39-45.
- de Hoop, A. L. 1995, Handbook of radiation and scattering of waves. Academic Press fur.
- de Huop, A. T., and Stam, H. J. (1988) Fine domain reciprocity thron us fin elastody namb, wave fields in solids with relaxation and their application to inverse problems. Wave Motion, 10, 179-489.
- de la Cruz, V., and Spanos, J. J., I., (980) Seismic boundary conditions for poivors media: J. Grophys. Res., 94, 3025-3025.
- Dellinger, J., and Vermk, L., 1992, Docore simple measurements record group or phase velocityl: 62nd Ann. Internat. Mtg. Soc. Expl. Grophys., Expanded Abstracts, 662-665.
- Demension A. J. M., Drijkoningen G. G., Smeulders, D. M. J., and Wapenaar, K., 2002. Reflection and transmission of waves at a fluid, param-medium interface. Geophysics, 67, 282-291.
- Decesiawicz, H., and Rice, J. T., 1962. The effect of boundaries on wave propagation in a hquid-filled porous solid: III. Beffection of plane waves at a free plane boundary (general case). Bull. Sesna, Soc. Ann. 52, 595-625.

- Deresiewicz, H., and Rice, J. T., 1964. The effect of houndaries on wave propagation in a hipped-filled porous solid: V. Transmission across a plane interface: Bull. Seism. Soc. Am., 54, 409-116
- Deorstewicz, H., and Skalak, R., 1963. On uniqueness in dynamic porcelasticity: Bull. Scient. Soc. Atta. 53, 784-788.
- Derks, G., and van Groesen, E., 1992, Energy Propagation in dissipative systems, Par-II: Controvelocity for goaling a systems: Wave Mation 15, 159-172
- Diallo, M. S., Prasad, M., Appel, F., 2003, Comparison between experimental results and theoretical predictions for P-wave velocity and attenuation at altrasonic frequencies Wave Motion, 37, 1-16.
- Domenico S, N. 1977 Elastic properties of unconsolidated porces sand reservous: Grophysics 42, 1339-1368
- Dung, Z., and McMechan, G. A. (1995) 3 D viscoelastic anisotropic modeling of data from a multicomponent, multi-azimuth scismic experiment in northeast Texas. Geophysics, 60, 1128-1138.
- Dutta, N. C., and Odé, H. (979a, Attraction and dispersion of compressional waves in fluid lifled params rocks with partial gas saturation (White model). Part E Bint theory. Geophysics, 44, 1777-1788.
- Dutta, N. C., and Odé, H., 1979b. Attenuation and dispersion of compressional waves in fluid-filled porous rocks with partial gas saturation (White model). Part II: Results: Grophysics 44, 1789-1805.
- Dutta, N. C., and Odó, H., 1983. Scismic reflections from a gas-water contact. Grophysics 48, 14-32.
- Dytta, N. C. and Scriff, V. J., 1979. On White's model of attempation in nocks with partial saturation: Genphysics, 44 (1806)1812.
- Dvorkin, J., Mayko, G., and Nur, A. 1995. Sight flow in fully saturated rocks. Grophysics 60, 97-107.
- Dvarkin, J., Nolea-Horksema, R., and Nur. A., 1991. The sepirt-flav mechanism: Macroscopic description: Graphysics. 59, 428-438.
- Dziewonski, A. M., and Gilbert, F., 1974. Solidity of the huner core of the Earth inferred from normal mode observations: Name, 234, 465-466.
- Fason G. Fulton, J. and Sunddon, I. N. 1956. The generation of waves in an infinite clustic solid by variable budy forces. Phil. Frans. Roy. Soc. London. Soc. A, 248, 575-607.
- Eckart, C. 1948. The approximate solution of one-dimensional wave equations: Reviews of Modern Physics 20, 399-417.

- Edelman, I., and Wilmanski, K., 2002. Asymptotic analysis of surface waves at vacmun/porous medium and fiquid/porous medium interfaces: Cont. Mech. Thermociyn., 14, 25-44.
- Edelstein, W. S., and Gurtin, M. E. 1965, A generalization of the hand and Somiglian stress functions for the dynamic linear theory of viscoelastic solids. Int. J. Eng. Sci., 3, 109-117.
- Emmerich, H., and Korn, M., 1987. Incorporation of attentiation into time domain computations of sciencia wave helds: Geophysics, 52, 1252-1264.
- Funserman, S. H., Schmidt, W., and Stephen, B. A., 1982. An implicit finite-difference formulation of the elastic wave equation: Genphysics, 47, 1521 4526.
- Fabrizzo, M., and Morro, A., 1992. Mathematical problems in linear viscoelasticity SIAM, Studies in Applied Mathematics 12.
- Fagin, S. W., 1992. Seismic modeling of geological structures: applications to exploration problems: Geophysical Development Series, 2, Society of Exploration Geophysicists.
- Fednow, F. J., 1968, Theory of elastic waves in crystals, Plenum Press
- Felsen, I. P., and Marcuvitz, N. (1973) Radiation and sewtering of waves. Prentice-Hall Inc.
- Fenavi, D., and Rocca, F., 1984. Scisme reciprocity field tests from the Italian Peninsula: Geophysics, 49, 1690-1760.
- Fong, S., and Julinson, D. L., 1983a, High frequency acoustic properties of a fluid (porous solid interface). I. New surface model: J. Acoust. Soc. Ann. 74, 906-914.
- Feng, S., and Inhuson, D. L., 1983b. High-frequency acoustic properties of a fluid (porous solid interface) II. The 2D reflection Green's function. J. Aroust. Soc. Am., 74, 915-924.
- Feynman, R. P., Leighton, R. B., and Swids, M. (1964) The Feynman loctures on physics, vol. 2, Addismi-Wesley Publ. Ca.
- Fokkema, J. L. and van den Berg, P. M., 1993, Seistab Applications of Acoustic Resprority, Flower Science Publ. Co. Inc.
- Fornberg, B., 1988. The pseudospretral methods are unaterrepresentation of interfaces in clastic wave calculations. Geophysics, 53, 625-667.
- Fornherg, B., 1996. A practical guide to pseudospectral methods. Cambridge Univ. Press
- Fourier, J., 1822, The analytical theory of heat (Paris), traps: A Freeman (Cambridge 1878).
- Frenkel, J., 1944. On the theory of sensible and seismode trie phenomena in a moist solit J. 1968, (USSIC: 8, 230-241.)

- Brenzel, H., and Schultes, H., 1934. Luminoszenz im ultraschallbeschickten wasser: Z. Phys. Chem., 1927, 421-424.
- Fung, V. C., 1965, Solid reschances, Prentice-Hall Inc.
- Gujevski, D., and Pseudy, J. (1992) Vector wavelields for weakly attempting anisotropic media by the ray method: Grophysics, 57, 27-38.
- Ganyi, A. F. 1970, A derivation of the seismic representation theorem using seismic reciprocity. J. Geophys. Res., 75, 2088-2095.
- Gangi, A. E., 1980a, Theoretical basis of seismic reciprocity (50th Ann. Internat. Mtg. Soc. Expl. Graphys. Research Workshop II., Seismic Reciprocity, Expanded Abstracts, 3625-3646.
- Gangi, A. F., 1980b. Elastic-wave reciprocity: model experiments, 50th Ann. Internat-Mty. Suc. Expl. Genphys., Expanded Abstracts, Research Workshop II., Seismic Reciprocity, 3657-3670.
- Gangi, A. F., and Carlson, R. L., 1996. An asperity-deformation model for effective pressure: hypenoplaymes, 256, 241-251.
- Gorret, C. G. B., and McCumber, D. F. (1970, Propagation of a Goyssian light pulse through an anomaliers dispersion medianic Phys. Rev. A, 1 (305-313).
- Gassmann, F., 1951. Uber die elastizität puolsen nachen: Viruteljahuesschrift der Natur fürschenden Grsellsr Laft in Zurich, 96, 4/23.
- Gazdag, J., 1981. Modeling the acoustic wave equation with transform methods. Grophysics, 54, 195-206.
- Gerrisma, J., and Smit, D. C., 1961, Some asports of elastic wave propagation in Endsaturated parons solids. Geoplesics, 26, 169-181.
- Gelinsky, S., and Shapiru, S. A., 1997, Porodastic Barkus-averaging for anisotropic, layered finid and gas sutmated sediments. Grophysics, 62, 1867–1878.
- Golinsky, S., Slaujiro, S. A., Müller, F., and Garevich, B., 1998. Dynamic poroclasticity of thirdy layered structures: Internat. J. Solids Structures, 35, 1739-1751.
- Golden, J. M., and Graham, G. A. C., 1988, Houndary value problems in fuoar visco-dastruty, Springer-Veilag.
- Goldstone, H. H., 1977. A history of improvided analysis from the 16th through the 19th century. Studies in the History of Mathematics and Physical Sciencies 2. Springer-Verlag.
- Gortlieb, D., and Orszag, S., 1977. Numerical analysis of spectral methods. Theory and applications, CBMS Regional Conference Series in Applied Mathematics 26, Society for Industrial and Applied Mathematics, SIAM.

- Graebner, M., 1992, Plane-wave reflection and transmission coefficients for a transversely isotropic solid: Geophysics, 57, 1512-1519.
- Graffi, D., 1928. Sai problemi dell'i reditacietà lun arr: Niovo Cimento A. 5, 53-71
- Braffi, D., 1939. Sur teorem di reorganettà nei fenomeni dipendenti dal tempo. Annali di Matematica, 745-18, 173-200.
- Graffi, D., 1954. Über den Reziprozitäitsatz in der dynamik der elastischen körper: Ingemenn Archy., 22, 15-16
- Graffi, D., 1963, Sui temetni di reciprocità nei fenomento menestazionari. Atti della Accademia delle Scienze dell'Istituto di Bologna, Classe di Scienze Fisiche, (Ser. 11), 10, 33-10
- Graffi, D., and Fabrizio, M., 1982. Nuc unhitta dell'energia Ebera per i materiali visenclastici. Arti Arcad. Naz. Lineri, 83, 209-214.
- Grien, G. 1838, On the reflexion and reflection of sounds (tracs: Combridge Phil: Soc. 6, Part III - 103-113.
- Green, G., 1842. On the laws of the reflection and refraction of fight at the common surface of two non-revealbord methac lams. Cambridge Phil. Soc. 7, Part I, 1-24.
- Guégien, Y., and Paleiauskas, V., 1994. Introduction to the physics of tocks, Princeton Univ. Press.
- Gurevich, B., 1993. Discussion of "Reflection and transmission of seismic waves at the homedaries of porous media". Wave Motion, 18, 303-304. Response by de la Cruz, V., Hube, J. O., and Spanos, T. J. T.: Wave Motion, 18, 305.
- Gurevich, B., 1996, Discussion nu, "Wave propagation in heterogeneous, purmis media: A velocity stress, finite difference method,", Dai, N., Vafidis, A., and Kanasewich, E. R., authors: Geophysics, 61, 1230-1232.
- Gurevich, B., and Carcione, J. M., 2000, Cassmann modeling of acoustic properties of sand.clay.mixtures: Pure Appl. Grophys., 157, 811-827
- Gurevich, D., and Loparnikov, S. L., 1995, Velocity and attenuation of elastic waves in hucly layered porous rocks: Goophys. J. Internat., 121, 933-945.
- Gurevich, B., and Schoenberg, M., 1999, Interface houndary conditions for Biot's equations of poroclasticity. J. Acoust. Soc. Ann. 105, 2585-2589.
- Gratin, M. E., 1981. An introduction to continuous mechanics. Academic Press Inc.
- Guiwed J., 2001. On the pulse velocity in absorbing and non-linear media and parallels with the quantum mechanics: Progress in electromagnetic research, PIER, 33, 69-96.
- Gutenberg, B., 1945, Energy ratio of reflected and retracted seismic waves: Duff. Seism. Soc. Ann. 34, 85-102.

Hammond, P., 1981, Thergy methods in electromagnetism, Clatendon Press,

Hanyga, A. (Ed.), 1985. Scismic wave propagation in the Earth, Elsevier,

- Hanyka, A., and Seredyńska, M., 1999. Some effects of the manory kernel singularity on wave propagation and giversion in poroclastic media. I: Forward modeling: Geophys. J. Internat., 437, 319-335
- Hardrwig, E., 1943. Cher the welleransbreitung in extens viscoelastischen medium: Z Groph., 18, 1-20
- Harryngton, B. F. (1961, Tene harmonic electromagnetic fields, McGraw-Hill Book Co-
- Haskell, N. A., 1953, The dispersion of surface waves in multilayered media: Bull. Scism. Soc. Amer. 43, 17-34.
- Havrs, M. V. and Biylin, R. S. 1973, Plane waves in linear viscoelastic materials: Quart Appl. Math., 32, 113-121.
- Hendry, J., 1986. Jam's Clerk Maxwell and the theory of the electromagnetic field. Adam Hilger 1.61
- Hellig, K., 1991. Fuundations of anisotropy for exploration seismics, Handbook of Grophysical Exploration, Pergamon Press Inc.
- Helbig, K., and Schornberg, M., 1987, Anomalous pubarization of elastic waves in transversely isotropic media: J. Acoust. Soc. Am., 81, 1235-1245.
- Helle, H. B., Pham. N. H., and Carcione, J. M., 2003, Velocity and attenuation in partially saturated rocks. Poroclastic numerical experiments: Greephys. Prosp., 51, 551-596
- Hennicke H. F. G., 1971. Reflection-refraction of a stress wave at a phase boundary between anisotropic media: J. Acoust. Soc. Am., 51 (200-217).
- Hess, H., 1964, Seisme anisotropy of the apprimest monthe under the oceans: Nature 203, 629-634
- Highm, R. L. (1991) Absorbing boundary conditions for elastic waves: Geophysics, 56 234–244.
- Hill, R., 1964, Theory of mechanical properties of tilicostrengthenod materials: J. Mech. Phys. Solids, 11, 357-372.
- Holberg, O., 1987. Computational aspects of the choice of operator and sampling interval for numerical differentiation in large-scale simulation of wave phenomena. Geophys. Prosp., 35, 629-655.
- Holdireh, H., 1858. On the n-th constitution from a circle. Quarterly Journal of Mathematics, 2, 301-322

- Holland, R., 1967. Representation of dielectric classific and piezoelectric losses by complex coefficients: IEEE trans. Sonies and Ultrasonies, 14, 18-20.
- Holland, C. W., 1991. Surface waves in puro-viscorlastic marine sedments: or Hovem, J. M. et al. (eds), Shear Waves in Marine Sediments, 13-20.
- Hooke, R., 1678, De Petentia Restitutiva, or of Springs (Philosophical itansactions of the Royal Society, and Cluterian lectures).
- Hongan, C. O., 1995. Anti-plane shear definitiations in linear and num-linear sulid mechanics: SIAM Review, 37, 53-81.
- Horron, C. W., 1953, On the propagation of Rayleigh waves on the surface of a viscoclastic solid: Goophysics 18, 70-74.
- Hosten, B., Deschamps, M., and Tittmann, B. R., 1987, Inhomogeneous wave generation and propagation in host anisotropic solids: Application to the characterization of viscorlastic composite materials: J. Acoust. Soc. Am., 82, 1763–1770.
- Hughes, J. J. R., 1987. The finite element method, Prentice-Hall Inc.
- Higher, S. C., 1983, Merhann's of continuous modul, John Wiley & Sons
- Egel, H., 1999. Wave propagation, in three dimensional spherical sections by the Cheby slow spectral methods Geophys. J. Internat., 139, 559-566.
- Igel, H., Mora, P., and Rodlet, B. (1915) Anisotropic wave propagation through initedifference gends: Geophysics 60, 1203-1216
- Jahn, M. K., 1984, Numerical solutions of differential equations, Wiley Eastern Ltd.
- [b]Breys, C., 1993. A new conservation law for classical electrodynamics: SIAM review, 34, 386-405.
- Juffreys, C., 1994, Response to a commentary by U. N. H. Robinson: SIAM review, 36, 638-641.
- [felliwys, 11, 1926] The rigidity of the Karth's central core: Mon. Not. R. Astron. Soc. Geophys. Suppl., 1, 371-386.
- Ju. C.-H., Shin, C., and Suh, J. H., 1996. An optimal 9-point finite-difference: frequencyspace, 2-D scalar wave extrapolating Grophysics, 54, 522-537.
- Johnson, D. L., 1986, Recent developments in the acoustic properties of porces media, in Sette, D., Ed., Frontiers in Physical Acoustics, Proceedings of the International School of Physics, "Enrice Ferric", Course 93, 255-200
- Juhnson, D. L., 2001. Theory of frequency dependent acoustics in patche-saturated porous media: J. Acoust. Soc. Mat. 110, 082-693
- Johnson, D. L., Koplik, L. and Dasher, R., 1987. Theory of dynamic permodility and torthosity in fluid-saturated porors media: J. Fluid Mech., 176 (379-402).

- Johnson, J. B., 1982. On the application of BioUs theory to acoustic wave propagation in snow. Cold Regions Science and Technology, 6 (19-60).
- Johnston, D. H., 1987, Physical properties of shale at temperature and pressure: Geophysics 52, 1391-1401
- Jones, D. S., 1986, Acoustic and electromagnetic waves, Chrenolog Press
- Jones, T. D., 1986, Poresfluids and frequency dependent-wave propagation tocks: Geophysics 51, 1939-1953
- Kang, I. B. and McMochaer G. A. 1993, Viscoelastic seismic responses of 2D reservoir models: Graphys. Prosp. 41, 149-163
- Karrenhach, M. 1998. Fall wave form modelling in complex media: 68th Ann. Internat. Mtg. Suc. Expl. Geophys., Expanded Abstracts, 1444-1447.
- Kazi-Aonal, M. N., Bunnet, G., and Jonanna, P. (1988, Green's functions in an infinite transversely isotrophe saturated porcelastic medium. J. Aroust. Soc. Ann. 84, 1883-1889.
- Keith, C. M., and Crampin, S., 1977. Scismic body waves in anisotropic methat tefficition and refraction at a plane interface. Geophys. J. Roy. Astr. Soc., 49, 181-208.
- Kelder, O., and Smeulders, D. M. L. 1997, Observation of the Dior slow wave in watersaturated Nivelsteiner sandstonet Geophysics, 62, 1794-1796.
- Kelly, K. R., and Marbirt, K. J., Eds., 1990, Numerical modeling of seismic wave propagation: Geophysical reputat Series No. 13.
- Kelly, K. R., Ward, R. W., Trovel, S., and Alford, R. M., 1976. Synthetic sersinograms: A finite-difference approach: Genphysics, 41, 2-27.
- Kelver, Lard (Thousan, W.), 1856, Elements of a mathematical theory of elasticity Phil. Trans. Roy. Soc. Lumdon, 146, 181-498.
- Kelvin Lund (Thunsan, W.), 1875 Math. and Phys. Papers. Cambridge Univ. Press 3, 27.
- Kessler, D., and Kosloff, D. 1991. Flastic wave purpagation using cylindrical enoulinates: Geophysics, 56, 2080-2089.
- Keys, R. G., 1985. Algorithmy boundary conditions for acoustle media. Geophysics, 50, 892-962.
- King, M. S., 1966. Wave velocities in rocks as a function of changes in overburden pressure and pore fluid sammanes: Geophysics, 31, 50-73.
- Iving, M. S., 2005, Rock-physics developments in scismic exploration: A personal 50-year perspective: Geophysics, 70 (3ND-8ND)

- King, M. S., Marsden, J. R., and Dennis, J. W., 2000. Biot dispersion for Ps and Swayes colorities in partially and fully saturated sandstones: Geophys. Prosp. 48, 1075-1089.
- Kjattansson, E., 1979, Constant Q-wave propagation and attennation: J. Geophys. Res., 84, 4737-4748
- Klausner, Y., 1991, Jundamentals of contineum mechanics of soils, Springer-Verlag,
- Klimentos, T., and McCann, C., 1988, Why is the Biot slow compressional wave not observed in real rocks?: Geophysics, 53, 1605-1609
- Kneib, G., and Kerner, C., 1993. Accurate and efficient seismic modeling in random media. Geophysics, 58, 570-588.
- Knight, R., and Nohra-Horksema, R., 19890, A laboratory study of the dependence of elastic wave velocities on pore scale fluid distribution: Grophys. Res. Lett., 17, 1529-1532.
- Knopaff, L., and Gangi, A. F., 1959. Soismic reciprocity: Geophysics, 24, 681 (91).
- Kolsky, H., 1953, Stress waves in solids, Chrendon Pross
- Komaritsch, D., and Vilotte, J. P. (1998) Die spectral element method: An elbenent tool to simulate the science response of 2D and 3D geological structures: Bull. Scien-Soc. Am., 88, 368-392.
- Komatitsch, D., Bornes, C., and Tromp, J., 2000. Similation of anisotropic wave propagation based upon a spectral element method: Grophysics, 55, 1251-1260.
- Kong, J. A., 1986. Electromagnetic wave theory. Wiley-Intriscience,
- Kornhauser, E. T., 1953, Ray theory for moving fluids. J. Acoust. Soc. Am., 25, 945-949
- Kosloff, D., and Baysal, U., 1982. Forward modeling by the Fourier method. Geophysics, 47, 1502-1412
- Kosloff, D., and Kessler, D., 1989. Seismie mimerical modeling, or Tarantola, A., et al., Eds. Geophysical Lonography, North Holland Publ. Co., 249-312.
- Kosloff, D., Kessler, D., Queitoz Filho, A., Tessmer, E., Belle, A., and Strahibvitz, 1990. Solution of the equation of dynamic elasticity by a Chebyelev spectral method: Geophysics, 55, 734-748.
- Kosloff, D., and Kosloff, R., 1986, Absorbing boundaries for wave propagation problems, J. Comput. Phys., 63, 463-376.
- Kosloff, D., Queiroz Filho, A., Tessmer, E., and Behle, A., 1989. Numerical solution of the acoustic and elastic wave equations by a new tapid expansion method: Geophys. Prosp., 37, 483-301

- Kosloff, D., Reshel, M., and Loewonthal, D., 1984. Elastic wave calculations by the Fourier methody Bull. Seism. Soc. Am., 74, 875-895
- Isostoff, D., and Tal-Lizer, H., 1993, A moduled Chebyshev pseudospectral method with an O(N=2) time step restriction. J. Comp. Phys., 104 (157-169)
- Kosten, U. W., and Zwikker, C., 1915. Extended theory of the absorption of sound by compressible wall coverings: Physica (Amsterdam), 8, 968-978.
- Kynners, H. A., 1937. La diffusion de la lumière par les atomes: Atti Congr. Intern Fisien, Cuma 2, 545-557.
- Krebes, F. S. (1983). Discrepancies in energy calculations for informageneous waves: Grophys. J. Rev. Astr. Suc., 75, 839 846.
- Krebes, E. S., 1983b. The viscoelastic reflection (transmission problem: two special cases Bull. Seistin, Soc. Am., 73, 1673-1683.
- Krobry, E. S., 1984. On the reflection and transmission of viscorlastic waves. Some manerical results. Grophysics, 49, 1374-1380.
- Krebry, E. S., and Le, I. H. T., 1994, Inhomogeneous plane waves and cylinchical waves in anisotropic anelastic media: J. Geophys. Res., 99, 23899-23919.
- Krebes, E. S., and Slawinski, M. A., 1991, On taytracing in an elastic anelastic medium: Bull. Seism. Soc. Am. 81, 697-686
- Krong, R. de L., 1926. On the theory of the dispersion of X-rays: J. Opt. Soc. Am., 12, 547-557
- Kummer, B., Behle, A., and Doran, F., 1987. Hybrid modeling of elastic-wave propagation in two-dimensional laterally inhomogeneous modia: Geophysics, 52, 535-577.
- Kvame J. J., 1949, Wave propagation in precordentric crystals: J. Acoust. Soc. Am., 21, 159-167.
- Lamb H., 1888, On reciprocal fluorents in dynamics: Proc. Fondon Math. Soc. 19 144-151.
- Lamb, H., 1904. On the propagation of termors over the surface of an elastic solid: Phil. Trans. Rev. Soc. London. 203, 1542.
- Lamb, L. and Richter, J. (1966) Anisotropic aconstic attenuation with new measurements for quartz of room temperatures: Proc. Roy. Soc. London, Sci. A. 293, 179-192.
- Lancaster, P., and Tismenetsky, M., 1985. The theory of matrices, second edition, with applications. Academic Press Inc.
- Le, L. H. J., 1994, Or. Cagniard's problem for a qSH line source in transversely-isotropic media. Bull. Seisna, Soc. Ann. 83, 529-541

- Le, L. H. T., Krebes, E. S., and Quiroga-Goode, G. E., 1994. Synthetic seismograms for *SH* waves in anelastic transversely isotropic media: Geophys. J. Internat., 116, 508-604
- Loctane, P., Cohen-Téhoudji, F., and Aggerre-Phente, J. (1994). Extension of Boot's theory of wave propagation to frazen porous media: J. Acoust. Soc. Am., 96, 3753-3768.
- Leitman, M. J. and Fisher, M. C., 1983. The linear theory of viscoelasticity, in Truesdell, C., Ed., Michanics of solids, Vol. 111. Theory of viscoelasticity, plasticity, elastic wayes, and elastic stability. Springer-Verlag.
- Leonardo da Vinci, 1923. Del more e misma dell'acqua. Cansú, E., and Favaro, A., Eds., Bologna, N. Zamchelli.
- Levander, A. R., 1988, Fourth-order Inite-difference P-SV seismograms: Goophysics, 53, 1425-1436
- Levander, A., 1989. Funte-difference forward modeling or seismology, in: The Encyclopedia of Solid Farth Grophysics, James, D. F., Ed., 405-431
- Lighthill, J., 1652. On sound generated accordynamically: I. General theory: Proc. Roy. Sur. London. Ser. A, 211, 561-587.
- Lighthill, L. 1984. Group velocity: J. Just. Maths. Applics., 1, 1/28
- Lighthill, L. 1978, Waves in Ihids, Cambridge Univ. Press,
- Limbell, I. V., and Olyslager, F., 1997. Analytic Green dyadic for a class of nonceriprocal anisotropic media. PIERS97.
- Lin, H. P., Anderson, D. L., and Kanamora H., 1970. Velocity dispersion due to anelasticity implications for seismology and manife composition. Geophys. J. Roy. Astr. Soc., 47, 41-58.
- Lo, I. W., Coyner, K. B., and Joksoz, M. N., 1986. https://annoisof.com/annois annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/annoisof.com/a
- Loewenthal, D., Lu, L., Roberson, R., and Sherwood, J. W. C., 1976. The wave equation applied to migration: Graphys. Prosp., 24, 380-399.
- Lumhn, R., 1970. The propagation of electromagnetic energy through an absorbing medium: J. Phys. A, 3, 233–245.
- Luce, A. E. H., 1941, A treatise on the mathematical theory of elasticity, Cambridge Univ. Press.
- Lysmer, J., and Drake, L. A., 1972, A finite element method for seismology, in Alder, B., Fernbach, S., and Bolt, B. A., Eds., Methods in Computational Physics II, Seismology (Chapter 6), Wademic Press Inc.

- Modariaga, R., 1976, Dynamics of an expanding circular fault. Bull. Seista, Soc. Am., 66, 639-696
- Mainardi, F. 1983, Signal velocity for transient waves in linear dissipative media. Wave Motion, 5, 64-41.
- Mainardi, F., 1987, Energy velocity for hyperbolic dispersive waves: Wave Motion, 9, 201-208.
- Mainardi, F., and Tomborri, M., 1997. Scismic prise propagation with constant Q and stable probability distributions: Annali di Geofisica, 40, 1311-1328.
- Mann, D. M., and Mackenzie, A. S., 1990, Prediction of pore fluid pressures in sedimentary bashes. Marine and Perrolema Geology, 7, 55-65.
- Marfurt, K. J. (1984) Accuracy of finite-difference and finite-characterized ment modeling of the scalar and elastic wave equations: Geophysics, 49, 533–549.
- Maris II., J., 1983. Effect of hnite phonon wavelength on phonon facusing: Phys. Rev. B 28, 7033-7037.
- Masters, G., and Gilbert, F., 1984. Structure of the inner core inferred from ubservations of its spheroidal shear mades: Genphys. Res. Lett., 8, 569–571.
- Mayku, G., and Mijkerji, T. (1995). Seismic processare compressibility and Gassmann's relation: Grophysics, 60, 1743-1749.
- Macko, G., Mukerp, T. and Dvorku, J., 1998. The tack physics handbook: Tools for seismic analysis in potents media. Cambridge Prov. Press.
- Maxwell, J. C., 1865, A dynamical theory of the electromagnetic field: Phil. Trans. Rev. Soc. London, 155, 479-512.
- Maxwell, J. C., (867, On Subdynamical theory of gases: 194) Trans. Roy. Soc. London 157, 29-88 (Scientific papers, 2 – 36-78 (1890)).
- Maxwell, J. C., 1890. The Scientific Papers of James Clerk Maxwell, Niven, W. D., Ed. 2 volumes, Cambridge Univ. Press.
- McAllister, I. G., 1968, Accorstic sounding of the lower troposphere: J. Ataros Ierr Phys., 30 (1994)1440.
- McDonal, F. J., Angona, F. A., Milss, R. L., Sengbush, R. L., van Nostrand, R. G., and White, J. E., 1958, Arteneiation of show and compressional waves in Pietre shale: Grophysics, 23, 421-339
- McTigue, D. F., 1986. Thermoclastic response of fluid-saturated porous took: J. Geophys. Res., 94, 9533-9542.
- Mehrabidi, M. M., and Cowin, S. C., 1990. Eigentensors of linear anisotropic elastic materials: Q. J. Mech. Appl. Math. 43, 15-11.

- Melrose, D. R., and McPhoiran, R. C. 1991, Electromagnetic processes in dispersive media, Cambridge Univ. Press.
- Meyer, O. E., 1874, Theorie der clastischen nachwirkung: Ann. Physik u. Chemio, 1, 108-118.
- Michailenko, B. G., 1985. Numerical experiment in seismic investigation: J. Geophys., 58, 101-124.
- Mimillin, R. D., 1949, Compliance of elastic bodies in contact. J. Appl. Mech., 16, 259-268.
- Minster, J. B., 1980, Wave propagation in viscorlastic media, in Dziewarski, A. M., and Boself, E., Eds., Physics of the Earth's Interior. Proceedings of the International School of Physics – Entrico berm?", Course 78, 152-212.
- Mitter, R., and Hokstad, K., 1995. Transforming walk away VSP data into reverse VSP data: Geophysics, 60, 968-977.
- Mizmani, H., Ishidu, F., Yokokma, T., and Ohnishi, S., 1976. Electrokinetic phenomena associated with carthquakes: Geophys. Res. Lett., 3, 365-368.
- Muchizuki, S., 1982 Attenuation in partially saturated nucks: J. Grophys. Res., 87, 8598(80)4.
- Mucza, P., Bystrický, E., Kristek, J., Carcione, J. M., and Bunchma, M., 1997. Hybrid multilling of P + SV seismic motion at inhomogeneous viscoelastic topographic structures: Bull. Seism. Soc. Am., 87, 1305-1423.
- Mura, P., 1989, Mudoling anisotropic science waves in 3 D, 59th Ann. Internat. Mitg. Soc. Expl. Geophys., Expanded Abstracts, 2, 1039-1043.
- Minues, A., and Vianelles, M., 1990. Minimal and maximal free energy for materials with memory. Boll. Un. Mat. Ital A, 4, 15-55.
- Miniso, P. M., and Foshhach, H., 1953, Methods of Theoretical Physics (McGraw-Hill Book Co.
- MuPi I, R., 1985, Scisme modeling in the implicit model Gequitys. Prosp., 33, 619-656
- Muir, F., Dellinger, J., Digen, J., and Nichols, D., 1992. Modeling elastic waveleds across groupher boundaries: Geophysics 57, 1189-1193.
- Mukerji, T., and Mayko, G., 5994. Fore fluid effects on seismic velocity in anisotropic torks: Geophysics, 59, 233-234.
- Müller, G., 1983. Rhoological properties and velocity dispersion of a meetima with powerlaw dependence of Q on frequency of Graphys., 54 (2042).
- Müller, J., and Gurevich, B., 2005. Wavesindneed Buil Bow in random porous media: Attentiation in dispersion of elastic waves: J. Acoust. Soc. Am., 117, 2732-2741.

- Murphy, W. U. 1982. Effects of microstructure and pore fluids on the acoustic properties of granular solumentary materials: Ph.D. these, Stanford University.
- Musgrave, M. J. P., 1960. Reflexion and refraction of plane elastic waves at a plane boundary between architempic media: Grophys. J. Roy. Astr. Suc., 3, 106–418.
- Musgrave, M. J. P., 1970, Crystal acoustics, Holden-Day,
- Nagy, P. B., 1992, Observation of a new surface mode on a fluid-saturated permeable solid: Appl. Phys. Lett. 60, 2735–2737.
- Nakagawa, K., Soga, K., and Mitchell, J. K., 1997, Observation of Biot compressional wave of the second kind in granular soils: Géotechnique, 47, 133-147
- Nelson, J. T. (1988) Acoustic emission in a fluid saturated hyperogeneous purous layer with application to hydraulic fracture: Ph.D. thesis, Lawrence Berkeley Laboratory, University of California.
- Nemagan, F. L. 1885, Vorlesungen über die Theorie der Flastizität, Joubaer, Leipzig.
- Nurris, A. N., 1985, Radiating fram a point source and scattering theory in a fluid saturated perons sulid [4, Aroust, Suc. Att., 77, 2012-2023]
- Norris, A. N., 1987. The time wave as a Biot slow wave: Geophysics, 52, 694-696.
- Norris, A. N., 1991, On the correspondence between princlasticity and thermaclasticity: J. Appl. Phys., 71, 1138-1141.
- Norris, A. N., 1992, Dispersive plane wave propagation in periodically layered anisotropic media. Proc. R. Ir. Acad. **92** VOI: 49-67
- Norris, A. N. 1993, Low frequency dispersion and attenuating in partially saturated rocks, J. Acoust. Soc. Ann. 94, 359-370.
- Norres, A. N., 1994. Dynamic Green's functions in anisotropic previolitetric, chermicalistic and purcelastic solids: Proc. Roy. Soc. London, Sep. A, 447, 175-188.
- Novacki, W., 1986, Theory of asymmetric elasticity. Pergamm Press Inc.
- Nussenzyeig, H. M., 1972. Causalay and dispersion relations. Academic Press Inc.
- Nutting, P. G., 2921, A new general law of deformations J. Franklin Just., 191, 679-685.
- Nye, J. P., 1985, Physical properties of crystals Their representation by tensors and matrices, Clarendon Press.
- Nyr. J. F. (1999) Natural focusing and fine structure of light, canstics and wave dislocatimes. Institute of Physics Publishing Ltd.
- Nyutai, I. F., Bron, F., and Razavy, M., 1996. Sensitiv reciprocity revisited: 66th Ann-Internat. Mtg. Soc. Expl. Grophys., Expanded Abstracts, 1947-1950.
- O'Conaelli, R., and Hudhansky, B., 1974. Seismie velocities in dry and saturated chacked solids: J. Geophys. Res., 79, 5412-5426.
- Ohaman, V., Suyder, T. M., and Carcione, J. M., 1997, Mesawerde and Greenriver shale anisotropies by wavefront folds and interference payterns: 67th Anni Internat. Vtg. Soc. Expl. Grouphys. Expanded Abstracts, 937-940.
- Didbam, R. D. (1996) Constitution of the Forth as revealed by contheprakes: Q. J. God Sur. 62, 456-475.
- Opisal, I., and Zabračnik, J., 1999. Elastic finite-difference method for integralar grids: Geophysics, 64, 240–250.
- Oughstim, K. E., and Sherman, G. C. 1993. Electromagnetic pulse propagation in causal dielectrics. Springer-Verlag.
- Onca, H., 1952a, Sound velocity in the snow cover: Low Temperature Science, 9, 171-178.
- Oma, H. 1952b. Reflection of sound at snow surface and mechanism of sound propagation in snow. Low "hemperature Science, 9, 179-186.
- Ozdenvar, T., and McMechan, G., 1997. Mgouthn's for staggerid-gibl computations for procelastic clastic, acoustic, and scalar wave equations: Geophys. Prosp. 45, 403-420
- Ozdenvar, T., McMochan, G., and Chaney, P., 1996. Simulation of complete seismesurveys for evaluation of experiment design and processing: Graphysics 61, 196-'908.
- Padovani, F., Prinlo, F., and Seriani, G. (1994) how- and hyp-order finite element method: Experience in scismic modeling: J. Comput. Accest., 2 (37):422.
- Parker D. F., and Maugin, G. A. Eils., 1988. Recent developments in surface acoustic waves. Springer Verlag.
- Parra, J. O., 1997. The transversely isotropic periodistic wave equation including the Biot and the squirt incluations: Theory and application. Geophysics, 62, 369–318.
- Payton, R. G. 1983. Elastic wave propagation in transversely isotropic media. Martinus Nijhoff Publ.
- Petropenles, P. G. 1995. The wave hierarchy for propagation in relaxing dielectrics. Wave Motion, 21, 253-262.
- Pierce, A. D., 1981, Acoustics, an introduction to its physical principles and apple arions Methaw-Rill Book Un.
- Pilant, W. L. (1979) Elastic waves in the earth, Elsevier Science Publ. Co. Inc.
- Pipkin, A. C., 1952, Lectures on viscoelasticity theory, Springer Publ. Co. Inc.

- Pipkin, A. C., 1976. Constraints in linearly clastic materials (dominal of blasticity, 6, 179-193).
- Plana T (1980) Observation of a served helk compressional wave in a parents medium at ultrasonic frequencies: Appl. Phys. Lett., 36, 259(2)4
- Podlubnyd , 1999, Fractional differential equations, Academic Press Inc.
- Polymen, A. D., and Zaetsev, V. F., 2003, Handbook of nonlinear partial differential equations, Chapman & Hall/CRC Press.
- Postma, G. W., 1955, Wave propagation in a stratilied medium. Grophysics, 20, 789 806.
- Pownting, J. H., and Thomson, J. J., 1962, Properties of matter, C. Griffer and Co.
- Prosad, M., and Manghuani, M. H. (1997, Fiffects of pure and differential pressure in compressional wave velocity and quality factor in Borra and Michigan sandstones Geophysics, 62, 1163-1176.
- Profe, S. R., 1994. Governing equations for the complext electromagnetics and acoustics of porces media: Phys. Rev. B, 50, 15678-15696
- Pricht, S. R., and Berryman, J. G., 1998. Connecting theory to experiments in puror lasticity: J. Mech. Phys. Sulids 46, 719-747.
- Prade, S. R., Berryman, J. G., and Harris, J. M., 2004. Seisma attenuation due to wave-induced flow. J. of Geophy. Res. 109, B01201, 1-19.
- Pride, S. R. Gaugi, A. F. and Morgan, F. D. (1992) Deriving the equations of mution for params isotropic media. J. Acoust. Soc. Ann. 92, 3278 4290.
- Profe, S. R. and Garambois, S., 2005, Electroscismic wave theory of Frenkel and more recent developments: J. Fug. Moch., 131, 898-907
- Pridi, S. R., and Haartsen, M. W., 1996, Electrospismic wave propagation: J. Acoust Soc. Am., 100, 1307-1315.
- Pride, S. R. and Morgan, P. D., 1991, Electrokinetic dissipation induced by scismic waves: Geophysics, 56, 914-925.
- Prinlo, F., Carrione, J. M., and Seriani, G., 1994. Numerical simulation of interface waves by high order spectral modeling techniques: J. Acoust. Soc. Atm. 95, 681–693.
- Präss, J., 1993, Evolutionary integral equations and applications, Birkhäuser.
- Psoneik, I. and Vovevénik, V.: 1998. Weak contrast PD wave displacement IQT coeffirients in weakly anisotropic classic media. Proc Appl. Genphys., 151, 699-718.
- Pyrak-Nofre, L. J., Myer, I. R., and Cook, N. G. W., 1990. Anisotropy in seismic velocities and amphrados from multiple parallel fractures. J. Geophys. Res. 95, 11415-11358.

Rabotnov, Y. N., 1980, Elements of hereditary solid mechanics, Mir Publ.

- Bandall, C. J., 1988, Absorbing brendary condition for the elastic wave equation: Genphysics, 53, 644-624.
- Rosolofosaon, P. N. J., 1991, Plane acoustic waves in linear discoelastic porous media: Friency, particle displacement and physical interpretation: J. Acoust. Soc. Am. 89, 1532-1550.
- Bayleigh, Lord (Strutt. J. W.), 1873, Some general theorems related to vibrations: Proc. London Math. Soc. 4, 366-368.
- Rayleigh, Lord (Strutt, J. W.), 1885. On waves propagated along the plane surface of an elastic solid: Proc. London Math. Soc., 17, 4-11.
- Bayleigh Lund (Strutt J. W.), 1899a On the application of the principle of recipiority to accustics: Scientific papers, 1(art. 11), 305-309.
- Rayleigh [Jord (Strutt J. W.), (899b, On purious backes in relation to sound: Scientific papers, 11a(1, 103) (221-225).
- Raylogh, Lord (Strint, J. W.), 1915. Theory of sound, Dover Publ.
- Roshef, M., Kosluff, D., Edwards, M., and Hsiung, C. 1988. Frare dimensional elastic modeling by the Fourier method. Grophysics, 53, 1184-1193.
- Bace, J. B. and Cleary M. P., 1976. Some basic stress diffusion solutions for fluidsaturated elastic porous media with compressible coefficients. Rev. Genphys. 14, 227–241.
- Bachards, P. G. (1984) On wave fronts and interfaces in an lastic mislion Bill. Sersen Sur. Am. 74, 2157-2165
- Biznichenko, Y. V., 1949, Seismie quasi-anisotropy Bull. Acad. Sci. USSR, Geograph, Geophys. Serv., 13, 518-514
- Roberts, T. M., and Petropoudos, P. G. 1999. Asymptotics and energy estimates for electromagnetic pulses in dispersive media. J. Opt. Soc. Am., 13, 1201-1217.
- Bohertsson, J. O. A., 1996. A memorical free-surface condition for elastic viscoelastic finite difference modeling in the presence of topography: Graphysics, 61, 1921–1934.
- Robertsson, J. O. A., and Coares, R. T., 1997. Funte-difference modeling of Q for qPand q8-waves in anisotropic model: 67th Ann. Internat. Mtg. Soc. Expl. Geophys., Expanded Abstracts, 1836-1849.
- Robertsson, J. O. A. Levander, A., and Holliger, K., 1996, A hybrid wave propagation simulation technique for occup aroustic problems: J. Geophys. Res. 101, 11225-11231.
- Robuson, F. N. H., 1994, Poynting's vector: Comments on a recent papers by Clark deficeys: SIAM review, 36, 633-637.

- Rokl.5n, S. L. Bolland, T. K., and Adler, L. 1989. Reflection-refraction of clastic waves on a plane interface between two generally anisotropic media: J. Acoust. Soc. Am., 79, 906-918
- Romeo, M., 1993, Inhomogeneous waves in anisotroph dissipative solids: Q. I. Mech Appl. Math., 47, 182–491.
- Rudaicki, J. W., 1985. Effect of pore fluid diffusion on deformation and failure of rock, us Bazant, Z., Ed. Mechanics of Geomaterials, Wiley Intersectate.
- Righticki, J. W. (2000) On the form of the potential for porms madia, presonal communication (10 pages).
- Saffari, N., and Hond, I. J., 1987, Body to Rayleigh wave mode-conversion at steps and slots [4, of Nondestri, Eval., 6, 1-22]
- Sahay, P. N., 1999, Groen's function in dynamic poio-dasticity: 69th Anni. Internat Mig. Soc. Expl. Geophys. Expanded Abstracts (1801-1804).
- Sahay, P. N., Spanos, T. J. T. and de la Cruz, V., 2000. Macroscopic constitutive regtations of an inhomogeneous and anisotropic porous medium by volume averaging approach. 70th Ann. Internat. Mug. Soc. Expl. Geophys., Expanded Abstracts, 1834-1837.
- Santamarina, J. C., Klem, K. A., and Fam, M. A. (2000) Soils and Waves: Participlate materials behavior, characterization and process monitoring. John Wiley & Suns
- Santos, J. E., Corberó, I. M., Ravazzoli, C. L., and Hensley, J. L. 1992, Reflection and transmission coefficients in fluid-saturated porous media: J. Acoust. Soc. Am., 91, 1911-1923.
- Santos, J. E. Dunglas, E. J. and Corbiró, J. M. 1990. Static and dynamic behaviors of a porous solid saturated by a two-phase fluid: J. Aronst. Soc. Ann. 87, 1128-1438.
- Santos, J. E., Douglas, JE. E. Corbero, J. M., and Lovera, O. M., 1990, A model for wave propagation in a perious medium saturated by a two-phase fluid: J. Acoust Soc. Am., 87 (1439-1448).
- Santos, J. F., Douglas, Jr., J., Murley, M. F., and Lovera, O. M. (1988). Fmitterelement methods for a model for full waveform accurate logging: J. of Numer Analys., 8, 115–433.
- Sarma, G. S., Mallick, K., and Gadhinglajkar, V. R., 1998, Nonzellecting boundary condition in limit-element formulation for an elastic wave equation. Graphysics 63, 1006-1016.
- Schlue, J. W., 1979. Finite element matners for seismic surface waves in three dimensional structures. Bull. Scisto: Soc. Mat. 69, 1125-1138.
- Schoenberg, M., 1974. Transmission and reflection of plane waves at an elastic-visco elastic interface: Geophys. J. Roy. Astr. Soc., 25, 35-47.

- Schoenberg, M., 1980, Elastic wave behavior across linear slip interfaces: J. Acoust. Soc. Am., 68 (1516)1521.
- Schornberg, M., and Costa, J. (1991). The monositivity of reflected SH waves to anisotropy in an underlying layered medium: Genphys. Presp. 39 (285)1003.
- Schoenberg, M., and Muit, F., 1989, A calendas for finely layered media. Geophysics, 54, 581-589.
- Schornberg, M., and Protazio, J., 1993. Zooppratz rationalized and generalized to anisotropy: J. Scis. Expl. 4, 125-141.
- Scholte, I. G., 1947, On Rayleigh waves in visco-clastic media, Physica, 13, 245-250.
- Scott Blan, G.W., 1999, Survey of general and applied theology, Pitman,
- Seriadi, G., Prioh, F., Carcinur, J. M., and Padovani, E., 1992. High-order spectral channel method for elastic wave modeling: 62nd Ann. Internat. Mtg. Soc. Expl. Geophys. Expanded Abstracts, 1285-1288.
- Serón, F. J., Badal, J. L. and Sabadell, P. J., 1996. A numerical laboratory for simulation and visualization of seismic wavefields: Graphys. Prosp., 44, 603-612
- Serún, F. J., Sanz, F. J., Kindelan, M., and Badal, J. L. 1990. Finite element method for elastic wave propagation. Communications in Applied Numerical Methods, 6, 459-368.
- Shapara, K. A., and Rudnik, I., 1965, Experimental determination of fronth sound velocity in Julium II, Phys. Rev. A 137, 1383-1391
- Shaphor, S. A., Andigam, P., and Royer, J. J. 1999, Large scale in site permeability of tooks from induced microscismicity: Geophys. J. Internat., 137, 207-213.
- Shapoo, S. A., and Hubbal, P. (1999) Elastic waves in random media, Springer.
- Sharma, M. D., 2004, 3 D wave propagating in a general anisotropic poroelastic medium, Bellection and refraction at an interface with Iluid. Geophys. J. Internat. 157, 947-958
- Sharma, M. D., Kouslok, V. P., and Gogna, M. I., 1990. Reflection and refraction of phase waves at an interface between liquid-saturated porous solid and viscoelastic solid: Q. J. Merk. Appl. Math., 13, 127-348.
- Sill, W. R., 1983, Self-potential medeling from primary flows: Geophysics, 48, 76-86,
- Silva, M. A. G., 1991, Pass and stop bands in composite humanes: Acustica, 75, 62-68.
- Skempton, V. W. (1974) The prio-pressure medicients A and B: Gentrehnique, 4: 133-152
- Slawinski, M. A., 2003, Seismic waves and rays in elastic media. Handbook of Geophysical Exploration. Perganaon Press. Inc.

- Smith, G. D., 1985. Numerical solution of partial differential equations: Finite difference methods. Clarendom Press.
- Smith, R. I., 1970. The velocity of light: Amer. J. Physics, 38, 978-984
- Steinberg, A. M., and Chao, R. Y., 1994. Dispersionless, highly superluminal propagation in a medium with a gain doublet. Phys. Rev. A , 49, 2071-2075.
- Stokes, G. G. 1856, Or. the dynamical theory of diffraction: Dans Cambridge Phil. Soc. 9 (1462)
- Stuff, R. D. (1989, Sodiment acceptive, Springer-Verlag
- Stoff, R. D., and Bryan, G. M., (970, Wave attenuation in saturated soluments: J Acoust. Sur. Am., 47, 1140-1447.
- Straidev, R., 1924. Flastic waves at the surface of separation of two solids: Proc. Roy Sar. Lundon, Sri. A, 106, 116-128
- Stiwas, A., and Fishi, B., 2001. Second-order approximations of the reflection and transmission coefficients between two visco elastic isotropic media. J. Seis, Expl., 9, 223-233.
- [14] Flott, H., Carolone, J. M., and Kosloff, D., 1990. An accurate and efficient scheme for wave propagation in Invary viscoelastic media: Geophysics, 55, 1366-1379.
- Tal-Der, H., Kosloff, D., and Koten, Z., 1987, An accurate scheme for seismic loward modeling: Graphys. Prosp., 35, 479-490.
- Tal-Uzer, H., 1989, Spectral methods in time for parabolic problems: SIAM J. Numer, Anal., 26, 1-11
- Temperton C. 1983, Fast mixed radix real Former transforms: J. Comput. Phys. 52, 340-350
- Temperton, C., 1988. Implementatives of a prime factor ITT algorithm on CRAV-1: Parallel Comput., 6, 99-108.
- 'berzagh: K., 1995, Erdbarnnerhauik auf hodersphysikalischer grundlage, Leipzig, F. Deutieke.
- Berzaglé, K., 1936. The shearing resistance of saturated soils and the angle between the plates of shear, in: Proceedings of the International Conference on Soil Michanics and Foundation Flagmeeting, Harvard University Press, Cambridge, 1, 51-56.
- Terzaghi, K., 1943. Theoretical soil mechanics, John Wiley & Sons.
- Tessmer, E., Kesshii, D., Kosloff, D., and Bihle, A., 1992. Multi-domain Chelyshevlocation method for the solution of the equations of motion of dynamic elasticity: J. Comput. Phys., 100, 355-363

- Tessmer, I., and Kosloff, D., 1991, 3-D elastic modeling with surface topography by a Chebyelley spectral method: Geophysics, 59, 464-473
- Primes, J. F., Abousleman, A., Cheng, A. H. D., Conssy, O., and Detommay, E., 1998, Collected papers of M. A. Bior (CD-ROM).
- Frompson, M., and Wills, J. R. (1991). A reformulation of the oppations of anisotropic porodasticity: J. Appl. Morb., ASMI:, 58, 612-616.
- Thomsen J. (1988) Reflection seismology over azimuthally mosotropic wiedro. Geophysics, 53, 304-313.
- Dimmoni, W. T., 1950. Transmission of clastic waves through a stratefied solid material J. Appl. Phys., 21, 89–93
- Fing, T. C. T., 1996, Anisotropy clusticity, theory and applications, Oxford Univ. Press Int.
- Toksoz, M. N. and Johnston, D. H., Eds., 1981. Seismic wave attenuation. Gouphysical reprint series.
- Fonti, E. 1976. The mason for mathematical analogies between physical theories. Appl. Math. Modelling, 1, 37-50.
- Esvankin, L. 2001. Seismic signatures and analysis of reflection data in anisotropic media. Handbook of Geophysical Exploration, Elsevier Science Publ. Co. Inc.
- Dirmer, G., and Seggins, A. F., 1994. Constant Q attenuation of subsurface radar pulses: Geophysics, 59, 1192-5200
- UISII, B., 1983. Review of elastic and electron suggeture wave propagation in horizontally layered media: Grophysics 48, 4063-1081
- Ursin, B. and Hangen, G.U., 1996. Weak-contrast approximation of the elastic scattering matrix in anisotropic media: Print Appl. Genphys., 148, 685-724.
- Unsin, B., and Stovas A., 2002. Reflection and transmission responses of a layered isotropic viscoelastic mediane Geophysics, 67, 307–323.
- Vatidas, V., Abramovici, F., and Kanasowich, F. R., 1992. Elastic wave propagation using hully vectorized high order linite-difference algorithms. Geophysics, 57, 218-252.
- Vainshtein, L. A., 1957, Group velocity of damped waves: Soviet: Phys. Jechn. Phys., 2, 2426-2428
- Van Gestel, J.-P., and Stoffa, P. L., 2001, Application of Allord rotation to groundpenetrating radar data. Geophysics, 66, 1781-1792.
- van Groesen, E., and Mainardi, F., 1989, Energy Propagation in dissipative systems, Part I: Centrowelocity for linear systems: Wave Motion, 11, 201-209

- Virieux, J., 1986, P-SV wave propagation in heterogeneous media. Velocity-stress limitdifference method: Geophysics, 51, 888-901.
- Vorgt, W. (1892). Cher numero reducing fester knoper insbestimbere der metallis. Ann Physik u. Chemin. 47, 671-623.
- Voherra, V., 1909. Sulle equazioni integro-differenziali della elasticità nel caso della esotropia: A04 Reale Accad. Naz. Lincei, 18, 275-301
- Vultoria, V., 1940. Energia nei fenomeni cueditari: Acta Pontulicia Arcad. Scient. 4, 145-128.
- Hippel, A. R. von, 1962, Dielectries and waves, John Wiley & Sons,
- Wang, H. F., 2000. Theory of lanear percelasticity, with applications to geomechanics and hydrogenlogy. Principlan Univ. Press.
- Wang, L. J., Kuzmith, A., and Dogarin, A. 2000. Gain assisted superhuminal light propagation. Nature, 397, 277-279.
- Waterman, P. C. 1969, New formulation of acoustic scattering: J. Acoust. Soc. Am., 45, 1117-1329
- Weaver, R. L., and Pao, Y. H. 1981, Dispersion relations for linear wave propagation in homogeneous and inhomogeneous media: J. Math. Phys., 22 (1909)1918.
- Wennerberg, L., 1985, Snell's law for viscoelastic materials: Geophys. J. Roy. Astr. Suc. 81, 13-18.
- White, J. E., 1960. Use of reciprocity theorem for computation of low frequency radiation patterns: Geophysics, 25, 613-624.
- White, J. E., 1975, Computed seismic speeds and attentiation in rocks with partial gas subration: Gosphysics 40, 224-232
- White, J. E., Mikhaylova, N. G., and Lyakhovitskiy, F. M., 1975, Low Inqueary seismic waves in fluid saturated layered rocks. *Lyestija Academy of Sciences USSR*, PLVS, Solid Earth, 11, 654-659.
- White, M., 2000, Leonardo, the first scientist: Abacus
- Whittaker, E. T. 1987, A lastary of the themnes for aether and electricity, I. The classical theories. The history of Modern Physics, 7, Tomash Publishers, Los Angeles, American Institute of Physics.
- Winterstein, D. F., 1987, Vector attenuation: Some implications for glane waves in andastic layered modia: Geophysics, 52, 810-813.
- Winterstein, D. F. 1990, Vehicity anisotropy: "hernihalizzy for geophysicists: Geophysics, 55, 1070-1088.
- Wood, A. W., 1955, A textbook of sound MarMillan Co

- Wright, J., 1987. The effects of transverse isotropy on reflection amplitude versus offset: Geophysics, 52, 564-567.
- Wirnesteile, B. and Morgan, F. D., 1991. Modeling of streaming patiential responses caused by oil well-principling. Geophysics, 59, 16:56
- Wyllie, M. R. J., Gregory, A. R., and Gardner, L. W., 1956. Elastic wave velocities in heterogeneous and porons media: Geophysics, 21, 11-70.
- Xu, T., and McMechan, G. A., 1995. Composite memory variables for viscorlastic senthetic science/grams: Geophys. J. Internat. 121, 634-639.
- Xu, T., and McMechan, G. A., 1997, GPR attenuation and its aumerical simulation in 2.5 dimensions. Geophysics, 62, 103-414.
- Yamada, I., Masuda, K., and Mizutani, H. (1989). Electromagnetic and accorsio-emission associated with eack feature. Phys. Earth Planet. Inter. 57 (157-168)
- Yin H. 1993 Acoustic velocity and attenuation of rocks: Isotropy, intrinsic anasotropy, and stress induced anisotropy: Ph.D. thesis, Stanford University.
- Zhang, S., and Jin, J., 1996, Computation of special functions, John Wiley & Sons,
- Zeper, C., 1948. Flasticity and onelasticity of metals, Fuw- of Chicago Press
- Zrug, X., and West, G. F., 1996. Reducing sparing adiffractions in elastic waveleful calculations, Geophysics, 61, 1436-1435.
- Zenkiewicz, O. C., 1977. The fight element method, Ourl edition: McGraw-Hill Book Cu
- Zimmerman, R. W. 1991, Compressibility of sandstones, Elsevier Science Publ. Co. Int.
- Zwikker, C., and Kosten, C. W., 1949, Sound absorbing materials, Elsevier Science Publ. Co. Inc.

Name index

Aldos: 152 Aboudi, I. 157 Abousleman, 236, 488 Miramocici, 389, 390, 488 Achenhach, 263 (157 Adlor, 12, 197, 198, 301, 457, 485 Aepimes, 134, 149 Agricola, 418 Aguirre-Puente, 236, 141, 178 Airy, 135, 136, 150 Also 135 171 175 397 457 Mekseev, 15, 157 Alford, xix, 323, 376, 378, 398, 157, 475 al Ghazzali, 147 al Havihan, 132, 117 Allard, 235, 157 Alterican, 398, 411, 457 Ampère, 135, 150 Anderson 70, 80, 82 (178) Augenbeister 174 Augo, 443, 448 Augona, 80, 86, 87, 140, 479 Angström, 451 Anstey, 411 Yoki 307, 157 Annol. 230, 489 Arago, 134, 150 Aristotic, 147 Amtsen, 181, 289, 304, 457 Arts, 40, 549, 154, 457 Asimes, 541, 557 Aud.(200) 276 486 Vu@ue 140 158 And six 1, 2, 1, 5 9, 115, 140, 141, 149 197, 257, 308, 309, 325, 457, 458 Antianh, 295, 458 Baldnet, xiv. 323, 375, 376, 459, Backas, xix, 28, 29, 38, 140, 142, 323, 369 391 439 440 458 Bacum, F = 143 - 148 Bacue, R., 132, 117 Badal, 392, 396, 186 Bagley, 86, 158

Bakulin, 29, 458 Bano, 86, 158 Bard, 295, 460 Barnes 397 176 Bartholiums, 433, 448 Bartoli, 452 Baste, 140, 158 Bateriani, 151 Barzle, 245, 458 Bayliss, 389, 390, 398, 399, 458 Baysai, 394, 395, 398, 158, 476 Berker 230 235, 141 158 Berr, 451 Behle, 28, 29, 12, 391, 394, 399, 464, 465, 176, 177, 187 Beltiore, 341, 458 Beltrandi, 451 Beltzen, I. 97, 458 Bon MegaLem, 82, 87, 149, 307, 134, 158 Bermulli, D., 133, 434, 449 Bermeilli, J., 146, 149, 158 Berryman, 20, 235, 236, 240, 241, 250, 259 280 282 290, 294, 141, 158, 159, 183 Bossel, 128, 391, 434, 450 Bore, 17: 218 438 440 451 459 Banconi 433 449 Biol. J. B., 451, 455, 450. Biot. M. A., xvi. xviii, six, 87, 235 237. 239, 240, 242, 243, 246, 250, 259, 281 285 268, 270 273 276 278, 2811 283, 286, 287, 289, 290, 293 205 (208, 299, 303 340, 318 (323, 432 378, 379, 383, 387, 389, 129, 110, 111, 159 Blackett, 451 Blanch, 388, 159 Blund: 83, 97, 322, 459 Bleistein, 126–153 Boothus 117 Beharsky, 171, 159 Bolland, 12, 197, 198, 485 Boltzmann, wii, 51, 52, 54, 69, 92, 93,

95, 124, 164, 165, 085, 418, 452, Hitk Hond, L. J., 498, 185 Hotsl, W. L., 9 Bonnet 295, 160, 175 Horskey, Di 160 Borcheidt, 97, 105, 110, 116, 121, 124, 184, 198, 199, 201, 230, 231, 460 Bosejko, 117, 160 Borelli, 132, 118 Horn, xix, 76, 195, 197, 323, 342, 345, 347, 349, 350, 370, 374, 449, 440, 154, 160 Hotae, 295, 158 Hoscovich, 119 Houchers, 3967, 4801 Hougher, 149 Barn bio 235, 274, 275, 286, 209, 460 Brussinesq. 452 Bertin, 295, 460 Bawle, 132, 133, 147, 148 Branewell, 59, 89, 423, 460 Bradley 433 449 Brahe, 154, 158 Branford 307, 457 Brand, 112, 460 Diserby, 152 Brekhovskikh, 192, 230, 244, 460 Brener, 345, 460 Brewstor, 183, 195, 196, 199, 321, 345, 354, 128, 135, 150, 160 Briggs, 133, 118 Brillouin, 87, 159, 439, 454, 461 Broda, 385, 161 Borwar, 240, 256, 307, 441, 461 Browne, 432, 448 Bruggeman, 25 (10) Rit Brugger, 13, 161 Bruns, 436, 452 Britsmit, 235, 341, 361 Bryan, 301 187 Buchen: 65, 97, 105, 106, 110, 111, 121, 141. 161 Budiansky, 290, 182 Bunsen, 435–451 Burgers, xis, 77, 79, 129, 154

Buridan, 147 Burridge, 20, 10, 235, 295, 441, 381 Bestnek(, 399, 480 Calon, 132, 148 Cadoret, 294, 295, 461 Cagmaid, 301, 110, 101 Catori, 131, 161 Cales, 196, 161 Camuer, 431, 461 Canton, 131, 449 Capturo, 83, 86, 341, 458, 462 Carcione, 16, 22, 28, 29, 35, 42, 43, 52, 54, 62, 77, 79, 84, 87, 94, 94, 96, 119, 120, 135, 138, 140, 142, 144 145, 147, 148, 159, 161, 165, 166 169, 171, 181, 184, 191, 197, 228, 234 236, 240, 241, 246, 278, 289, 294, 295, 295, 304, 307, 313, 316, 317, 322, 341, 349, 342, 357, 366, 369, 375, 387, 389, 391, 393, 396, 397, 399 [02] [04, 109, 416, 415 457. R.2. 469, 172, 153, 480, 482 183, 486, 487 Cardano, 143, 148 Carlson, 243, 471 Carnell, 254, 255, 307, 465 Carslaw, 381, 465, Cassine 118 Castigliance 152 Casula, 79, 357, 466 Cap.Lv. 335, 336, 156, Cavadieri, 148 CavaDoni, 16, 38, 40, 54, 87, 540, 142, 144 145, 147, 159, 161, 166, 169, 191 197, 235, 236, 240, 241, 346, 331, 339, 369, 463, 466 Cavene3sie, 134, 149 Caviglia, 54, 62, 97, 100, 104, 140, 114, 121, 124, 322, 333, 466 Costla, 151 Cide-barm, 235, 460 Celia, xix, 389, 390, 394, 466 Celsius, 1497 Cerjan, 100, 166 Cerruti, 452 Cerver, v. 1, 466

Cosám, 153 Closelwick, 10, 116, 161, 166 Chambred, 295, 358 Chandler 276, 166 Chapper 101, 182 Channian, 33, 466 Chebyslaw, 91 (390) 392, 391 397, 402, \$21 \$26 \$41 Chang, 236, 255, 307, 466, 488 Chew 374, 100 466 Chao, 161, 187 Closen, 135, 167 Chen. 97, 467 Caladhi, 411, 449 Cagos: 153 Christensen, 52, 55, 57, 97, 126, 435, 467 Christhausen, 437, 452 C'aristoffei, 5, 10, 11, 14, 18, 29, 31, 33, F.B. 153, 154, 206, 414, 461, 138. 151 Cirrysippers of Soli, 117 Cisternay dn Eay, 119 Clapeyton, 50 Clausuis, 151 Clavion, 395, 301, 167 Cleary, 235, 484 Clobsch: 451 Contes, 134, 184 Cohon Ténnadji, 236, 441, 478 Colo, K. S., 86, 311, 467 Colo, R. H., 86, 341, 467 Celladon, 135, 150 Cook, 130, 135, 483 Cooper, 121 - 367 Compringlu, 235, 467 Corbero, 235, 288, 289, 141, 485 Connack, 111 Cornu. 448, 447, 452, 467 Cossetar, E., 139, 153, 167 Cosserar, F., 449, 467 Costa, 43, 47, 186 Conlomb, 134, 449 Comant. 395, 454 Conssy. 235, 236, 262, 274, 275, 286, 299, 460, 467, 488 Covin, 39, 142, 143, 316, 467, 479

Covnet 307, 178 Conner, 149 Compute 1, 12, 167, 175 Crank, 389 Custescu, 235-367 Croakes [14] Capite, J., 438, 452 Case, P., 139, 138, 153 Currie, 117, 118, 120, 441, 467 Cuvier, 145. Dablam 388, 467 Dat 390 467 472 d'Alembert, 133, 149 Dates, 12, 168 Dampiet, 113, 168 Daniels, 87, 468 Dasey, 261 (263, 279, 286, 304, 378, 458, 168 Datavin, 115 Dashen, 268, 269, 304, 474 Datteli, 85, 468 Davy, 435, 450 Day, 95, 125, 111, 108 de Brazlie, 139-154 D) Dominis, 132, 148 de Graot, 263, 468 de Huop, A. T., 171 (173, 301) 440, 141. 168de Hong, M. V., 29, 461 de la Cruz 288, 307, 468, 472, 485 de Vries 139, 153 Delves, vor. 338 (356) 429 (139) 440 (154) 468 Dolla Porta (132, 118) Dellinger 22, 304, 468, 480 Dememan, 288, 468 Dennes 204, 176 Deresiewa 2, 284, 288, 289, 295, 299, 140, 468 169 Darks, 88, 469 Descartes, 132, 147, 148 Deschamps, 139, 140, 174 Denominav. 236, 488 Dis25, 290, 469 District of Freiberg, 132, 117

Dirac, 168, 174, 174, 265, 298, 341, 357, 139 Duichlet, 172, 396, 351 Divini, 148 Dogarm, 161, 489 Domenico, 141, 1181 Done, 166, 169. Doppler, 536, 443, 450 Dorau, 394, 477 Destroysky, 131, 161 Douglas, 245, 392, 441, 485 Diake, 396, 141, 178 Drijkoningen, 285, 468 Dussik, 440. Datto, 201, 288, 290, 293, 161 Dvorkm: 1 (235, 246, 289, 260, 309, 469, 1791 Dziewoński, 111, 109 Eason, 128, 469 Eckart: 87, 88, 469 Efelman, 300, 470 Féelstein 100 470 Edwards, 394, 484 Einstein, 315, 334, 139, 154 Elsasson, 439, 440 Emmerich, 125, 389, 399, 470. Emacriman, 387, 470 Engquist, 399, 401, 467 1.0gen, 394, 180 Endid of Mexandrea, 132, 147 Eicher, 4, 84, 146, 125, 146, 168, 236, 249. 250, 266, 304, 309, 340, 386, 390, 109, 114, 133, 134, 149 Ewing, 152 Labrizio, 52, 55, 97, 470, 472 Edge 153 Eigin, 001, 170 Enn. 235–485 Enaday 353 136, 137 150 Forliner, 450 Fedorov, 1, 38, 470 Felsen, 159, 170 Fenari, 172, 470 Fing. 301, 303, 411, 470 Fermity 433, 445, 448

Feshbach, 126, 127, 480 Feynian, 287 (110, 170 Furetti, 403, 464 Firestants, 140, 474 Fisher, 97, 478 Fitzgerald, 438, 452 Fizoan, 137, 151 Flamstered, 118 Floquet, 52 Fork 439, 435 Fokkoma, 172, 470 Fornberg, 393–395, 170 Foreault 137, 151 Fourier, xvi. sxn. 55, 59, 69, 72, 75, 85. 86, 89, 90, 94, 99, 146, 120, 126 129, 151, 163, 168, 169, 173, 220 268, 269, 266, 290, 309, 321, 331, 369, 373, 382, 388, 393, 396, 400, 912, 422, 424, 126, 135, 149, 170, Franklin, 349 Fraudoter, 135, 150 Frechet, 151 Freukel, 383, 440, 454, 470 Frenzel 140 471 Freshel, 321, 345, 354, 430, 435, 450 Fulton, 128, 169 Fm.e. 171, 172, 218, 279, 171 Gabin, 140 Gailhinglajkar, 100, 185 Gajewski, 110, 174 Galerkin, 386, 439, 453 Galilei, G., 532, 143, 147, 148 Gabler, V., 432, 148 Calture 451 Galvani, 439 Gangi, 77, 79, 171, 173, 181, 235, 243, 246, 375, 440, 464, 474, 476, 484 Garandools, 483, 483 Gardner, 372, 190 Garriet, 161, 471 Gassendt 118 Gassmann, xviii, 241, 248, 253, 257, 274, 383, 430, 459, 461, 471 Gauss, 374, 397, 425, 434, 436, 450 Gauzellino, 245, 464 Gazdag, 394, 441, 471

Greetsma, 289, 471 Gri, 94, 103, 104, 104, 165 Griget, 154 Gelinsky, 29, 294, 307, 471 Germann, 150 Gibson, 307, 458 Gibert, F., 141, 569, 479 Giban, W., 132, 138 Glassmover, 97, 121, 230, 231, 160 Gogua, 288, 186 Golden, 52, 53, 55, 56, 58, 59, 116, 373, 171 Grédszein, 452 Gubbstine, 131, 171 Gooim, 139, 151 Gordieb, 395, 425, 474 Graebuer, 12, 472. Graff, 51, 52, 171, 139, 140, 172 Graham, 52, 53, 55, 56, 58, 59, 146, 373 171 Gray, S., 433, 129 Grav. T., 452 Grav. W. G., xix, 389, 390, 394, 466 Grosar, xviii, 40/42, 97, 126, 127, 129, 168 160 171 173 279 281 282 295, 297, 209, 301, 321, 342, 352 355, 368, 369, 381, 129, 131, 435 447, 449, 411, 450, 472 Gregoev, A. R., 372, 490 Gregory, J., 118 Gregabli, 133, 148 Grassoviste, 132, 147 Grave, 451 Guéguen, 372, 472 Guericke, 148 Guievich, 87, 236, 240, 241, 284, 286, 288, 293, 294, 407, 464, 471, 472, 1.0 Guetin, 38, 100 - 170, 472 Grawach, 88 (20) 472 Gunedbirg, 212, 440, 454, 472 Baartsen, 236, 295, 483 Badamard, 453 Hale, 153 Hall, 438, 452 Halley, 449

Hamilton, xviii, 235–263, 286, 435, 436, 139. 115. 151 Hammond, 329-473 Hankel, 126, 127, 129, 168, 298, 152 Hanstein, 150 Hanyga, 1, 87, 463, 473 Hardtwig, 116–120, 173 Harring on 364 473 Harriot, 132, 133, 118 Harris 200/204/111/183 Hashin, 441 Haskell, 440, 473 Hattern, 13, 188 Hattkshov, 449 Haves, 147, 120, 184, 141, 167, 473 Heaviside, 11, 42, 52, 53, 126, 130, 304. J21, J31, J50, J83, 138, 152 Hellóg, J. 2, 7, 29, 35, 38, 44, 142, 143. 145, 181, 198, 316, 317, 111, 163, 164, 173 Helle, 77, 79, 236, 241, 296, 316, 317, 393, 396 [61] \$65, 173 Heinzholtz, 92, 90, 127, 171, 297, 367. 381 392 437 438 440 451 Heidev, 322, 473 Hermeke II, 42, 197, 198, 173 Henry, 436, 450 Hensbey, 288, 289, 185 Heiglory, 154 Hermann, 119 Hermite, 451 Heron of Alexandria, 132, 147 Herschol, 150 Herrz, 138, 140, 153 Hess. 441, 473 Higdon, 400, 473 Hilbert, 32, 60, 453 Hill, 294, 473 Hippel 337–489 Hokstad, (72, 480 Holberg, 391, 373 Hoblich, \$16, \$73 Holland, C. W., 301, 174 Hollard, R., 148, 174 Holliger, 394, 184 Horskei, E. J. 51, 55, 141, 161, 407, 317.

321, 386, 131, 134, 138, 174, Hopkinson, 152 Horgan, 184, 474 Horton, 116, 174 Hosten, 139, 140, 474 Houssteld 111 Hum 12 172 178 168 181 Hsming, 391–481 Hubble 153 Hubral, 29, 486 Hughes, 386, 392, 396, 474 Hagmabat, 138, 152 1bmd, 154 Ihumes, 52, 54, 62, 474 Thiygens, 133, 113, 118 Igel, 394, 399, 474 Ishido, 383, 180 Jarobi, 150 Janger, 381, 465 Jain, xix, 385, 389, 474 leffreys, C., 333, 474 Infrays, H., 409, 474, 474 lin, 105, 109, 190 Jo. 492, 471 Johnson, D. L., 235, 268, 269, 275, 276. 294, 301, 303, 304, 441, 466, 470, 1774 Johnson, J. B. 275, 475 Johnston, 86, 139–186, 175–188 Jones, D. S., 76, 375, 475 Jones, T. D., 289, 175 Jordan, 389, 390, 398, 399, 458 Jouanna, 295, 475. Joule, 155 Kanamori, 79-82, 478 Kanasowich, 389, 390, 467, 472, 488 Kang, 125, 475 Karal, 398, 441, 457 Katmán, 295 Karrenbach, 393, 175 Kaushik, 288, 486 Kazi-Aonal, 295, 475 Keta, 42, 475

Kelder, 275, 305, 306, 475

Keller, 245, 461 Kelly, 392, 396, 398, 475 Kelvin, 77 Kelvin, Lucé, 5, 10, 11, 14, 18, 29 31, 33. 38, 51, 68, 71, 74, 76, 78, 79, 86, 94, 117, 139 143, 146, 154, 154, 205, 314, 316, 317, 321, 322, 325, 356, 357, 361, 385, 428, 429, 431 \$6. \$7. \$1. \$6. \$75 Kennelly, 453 Kender, 432, 148 Kenner, 393-476 Ken, 138, 151 Kessler, 495, 496, 489, 475, 476, 487 Keys, 100, 175 Kindelan, 392, 396, 186 King, 235, 294, 141, 175, 176 Kircher, 132, 138 Kirchhoff, 235, 435, 437, 451 Kjartansson, 83 85, 476 Klausne: 1, 77, 476 Klem, K. A., 235, 185. Klem O 139 154 Kleist, 149 Klumentos, 278, 476 Kneb, 393, 476 Knight, 294, 176 Knopoff, 174, 173, 140, 176 Knott, 449, 453 Knudsen, 154 Kohliausch, 437, 452 Kolusov, 153 Kolsky, xix, 176 Komatitseb 397 176 Korg, 362, 476 Kophk 268 269, 301 471 Koren, 391, 187 Korn, 125, 389, 399, 470 Kornhauser, 140, 176 Korringa, 240, 256, 407, 441, 461 horroweg, 439, 452 Koshdf, D., 22 (28, 29, 12, 84, 93, 96, 387. 391, 394, 100, 109, 126, 458, 464 466, 475, 477, 184, 187, 188 Kosloff, R. 22, 12, 81, 94, 96, 399, 400, 109, 465, 466, 476

Kasten, 235, 340, 477, 490 Kramus, viz. 59, 86, 373, 128, 439, 151 177 Krebes, 97, 140, 113, 121 (123, 135, 161) 165, 169, 191, 193, 199, 291, 214, 221, 228, 343, 365, 375, 478 Kristok, 399-480 Krusneker, 3, 173, 174, 397, 151 Kronig, R. de L., xix, 59, 86, 373, 428. 177 Kronig, A. K., 151 Kummer, 394, 477 Kundt 438, 452 Kutta, 220, 388, 396, 402, 439, 453 Kuzmirk, 161, 489 Kyame, 440, 477 La Condonssie, 139 Lagrange, 171 (236, 263, 264, 296, 134 119 Lamb. H., 171, 172, 178, 181, 438, 439, 152, 177 Lamb. J., 140, 177 Lamberts 419 Lumé, 3, 55, 98 (146, 117, 142, 230, 233) 1.1.1 Laneaster, 298, 299, 477 Langues, 341 Langèvin, 153 Laplace, 323, 382, 434, 435, 449 Larger, 453 Lax: 388 Lo. 29, 161, 166, 169, 191, 193, 161, 177, 475 Loclaire, 235, 111, 178 Legendre, 124, 149 Lehmonry 110, 153 Leibniz, 432, 449 Leighton, 287, 470 Leitman, 97, 478 LeMesurier, 389, 390, 398, 399, 158 Lenard, 453 Lenz 150 Leonardo da Venci, 132, 113, 116, 118 175 Lonamier, 393, 394, 399, 478, 484 Levi-Civita, xxii, 99, 276, 361, 453

Li. 235, 466 Liehturekov, 372 Lie: 152 Lighthill, 18, 140, 178 Ludell, 367, 478 Lionville, 151 Lippershey, 132, 148 Lippmann, 452 Lissajous, 351 Lin, H. P., 79, 80, 82, 178 Liu, Q. H., 400, 466 Llood, 436, 450 Lo. 307, 448 Loharta, 397, 425 Loder, 352 Loowenthal, 397–399, 477, 478 Lonon soy, 149 Loomus 454 Lopatinkov, 87, 203, 204, 472 Lorentz, 87, 159, 161, 138, 139, 152 Lonenz, \$37, Vil. London, 101, 178 Love, xiz. 1 (1) 20, 396, 441, 139, 453, 478 Lovera, 392, 141, 185 Lo. 398, 178 Lvakhneitskiv, 290, 294, 389 1.yeQ, 115 Lysner, 396, 141, 178 MacCorgark, 390 MacCullagh, 436, 454 Macelwant, 274 Mach. 152 Mackenzie, 211, 479 Madariaga, 393, 479 Mainardi, 83, 87, 88, 159, 341, 462, 463. 479 188 Mallet 151 Mallick, 100, 185 Malus, 195, 321, 353, 434, 435, 450 Manghuané, 211–183 Manu, 214, 179 Marcont, 139, 154 Marcawitz, 159, 470 Marfurt, 397, 396, 475, 479 Marion, 294, 295, 461 Mariotte, 433, 148

Maris, 22, 479 Maysden, 294, 476 Mascart 152 Masters, 111, 179 Masuda, 383, 490 Mangin, 116, 460, 482 Manpertnis, 145, 149 Maurilycus, 132, 118 Mayko, 1, 235 246, 248, 289 290, 469, 179E 180 Maxwell, xviii, xix, 51, 62, 68, 72, 76, 79, 92, 94, 117, 130, 321, 329, 341, 353, 356, 359, 366, 367, 375, 380, 129, 131, 137, 138, 151, 179, Mayer, J. R., 136, 151 Mayer, J. F. 119 Mazin, 263, 468 Mazzacurath, 85, 468 McMfister, 441, 479 Mr.Cam. 278, 476 Mr.Cumber, 161, 171 McDonal, St. 86, 87, 440, 479 MrMichan, 125 (66, 339, 393 (001, 469, 175, 182, 190 McPhishian, 55, 480 Mi Figur, 236, 479. Michrahoet 39, 142, 143, 479. Mellon, 150 Meliose, 55, 480 Menabusa 151 Mersenne, 432, 148 Menoci, 137, 151 Meyer, M. W. 152 Meyer, O. E. 68, 138, 151, 180 Miclo II, 134, 149, 153 Mirhelson, 438, 452 Mir. 139, 153 MikLadonko, 394, 480 Mikhaylova, 290, 294, 189 Millikar, 153 Milne 1-2 Milss, 80, 86, 87, 140, 179 Milma 240, 241, 459 Mindlus 140, 180 Minkowski, 473 Minster, 95, 97, 441, 468, 480

Min10p. 139, 154 Mises, 454. Mord ell, 275, 481 MB tog-Lellin, 341, 452 Mistor, 172, 480 Mizmani, 383, 180, 190 Mochizuki, 289, 180 Moczo, 494, 499, 180 Molonovice, 539, 453 Maler 151 Muloktuv, 29, 158 Muta, 394, 474, 480 Murgan, 235, 383, 183, 196 Morland, 118 Modey, E. W., 452 Modes, M. E., 392, 485 Marro, 52, 54, 55, 62, 95, 400, 104, 110. 111, 121, 121, 322, 333, 466, 470, (NO Morse, 126, 127, 180 Mossoffi, 459 Mare: 389-300, 180 Mair. 28, 394, 480, 486 Makerji, 1, 235, 246, 248, 290, 459, 180 Müller, G., 83, 86, 180 Müller, F. 293, 307, 471, 480 Murphy, 291, 481 Musgrave, 1, 42, 14, 187 Masselupbrock, 449 Merry 130, 135, 483 Nagy, 301, 304, 157, 181 Nakagawa, 275, 181 Napier, 481 Navier, 135, 150 Volsini, 259-481 Nonmania, C. G., 172, 387, 396, 151 Noumann, F. F., 139, 208, 326, 136, 150. 1×1 Newcomb, 151 Newmark, 392 Newton 88, 4, 433, 437, 133–144, 148 Nichols, 394, 480 Nicu2 (150) Nicolson, 389 Nolea-Hocksema, 294, 309, 469, 476 Noller, 133, 149

Norris, 29, 40, 275, 294, 295, 299, 307 323, 461, 481 Nowas Et. 1, 181 Nut. 246, 289, 290, 309, 169 Nussenzo(a, 161, 481) Nutring, 83, 181 Nve. 1, 139 (208, 253, 446, 481 Netrai, 172, 178, 181 Nyquist, 388, 393, 395, 398, 423, 425, 140 151 O'Connell, 296, 182 0.548, 201, 288, 290, 469 O'Doherty, 411 Oeisted [35] 150 Ohanian, 22, 182 Ohm 356, 382, 383 435 437 450 Ohnislå 383–180 Olbris, 329. Ohlham, 139, 153, 182 O'Lesev, 117, 120, 411, 467 Olyslager, 367, 478 Omori, 453 Onat, 335, 469 Onsager, 55, 263 Opesal 394, 482 Orszag, 395, 425, 471 Oughstun, 54, 459, 461, 322, 333, 482 Oma, 275, 482 Ozdenvar, 393, 401, 482 Pacioli, 147 Padowing 392, 397, 182, 186, Pagani, 124, 466 Paleiauskas, 372, 172 Palmieri, 437, 451 Pao, 59, 60, 189 Pardies, 133, 148 Parket, 116, 460, 482 Paris, 307, 182 Passeal, 81 Pactor, 1, 22, 40, 42, 582 Perkham, 332 Politica, Mo. Perot. 153 Poslikin 1410 Petropoulos, 360, 482, 484

Petrus Peregrinus, 132, 147 Pham. 295, 461, 473. Picatel 148 Pictor, 431, 482 Pilsar, xix. 88, 128, 134, 174, 298, 354. 368 182 Prola, 150 Pipka: 5, 97, 182, 183 Plauck, \$53 Phina 275 111 183 Porhhammer, 438, 452 Packels, 138, 153 Podlubny, 128, 341, 184 Poincaré, 452 PoisemHe, 262, 267, 269, 270, 450 Poisson, 116, 120, 136, 393, 398, 399, 102. 404 131 437, 450 Poletto 94, 161, 465 Polyanin, 381, 483 Postma, E0, 25, 28, 142, 183 Poynting, xvi, xviii, 15, 16, 21, 13, 14, 16, 62, 63, 66, 74, 107, 114, 124, 147, 148, 150, 156, 185, 189, 191, 192, 194, 195, 297, 201, 203, 206, 210, 213 215, 310, 312, 315, 331 333, 342 343 347 363 438 439 452 183 Proudtl. 454 Prasad, 244, 290, 169, 183 Pride, 235, 246, 280, 282, 290, 294, 295, 483, 111, 183 Pricetley, 119 Priolo, 392, 397, 182, 183, 186 Protozo, 12, 186 Prüss 85, 183 Psencik, 43, 116, 166, 171, 183 Prolemy, 432, 447 Pyrak Nobe, 130, 135, 183 Pythagoras, 10, 132, 147 Queiroz I ilho, 391, 399, 476 Quiroga-Goode, 159, 166, 169, 299, 387, 389 165 478 Rabitana, 52, 184 Radon, 154 Raman, 151

Randall, 200, 484 Barkule, 438, 451 Resolutionari, 114, 139, 457, 484 Ravazzoff, 245, 236, 241, 288, 289, 464, 165, 185 Rayleigh, Lord. 116 118, 120, 121, 136. 138, 171, 230, 231, 235, 248, 299, 376, 396 102 128, 431, 438 441, 114, 152, 184 Razavy, 172, 178, 481 Reid 153 Reshof 204, 207, 399, 400, 466, 477, 484 Renher-Paschwitz, 438, 153 Reynolds, 262, 452 Rios, 235, 288, 289, 295, 468, 469, 484 Richards, 122, 135, 174, 175, 397, 457, 1×4 Ridendson, L. F. 439 Richardson, R. L., 230, 231, 441, 458 Richter, 140, 140, 155, 177 Ricker, 220, 402 Riemana, 437, 451 RioBer, 394, 474 Risner, 148 Ritten, 434, 450 Rivlin, 184, 473 Rizmehenko 25, 484 Roberson, 398, 478 Roberts 392, 184 Robertsson, 143, 388, 394, 399, 159, 183 Robinson, E. A. 342, 465 Robinson F. N. H., 184 Rocca 172, 470 Rokhim, 12, 197, 198, 485 Romoo, 192, 485 Romet, K&, 144, 148 Rometen, 452 Rocker, 372 Rowland, 152 Rover, 276, 486 Rodnicki, 279, 307, 485 Rudnik, 111, 186 Rodzki, 439, 453 Rumford, 149 Runge, 220, 388, 390, 402, 439, 453 Righerford, 453

Rytow, 110 Sahadoll, 392, 396, 486 Salúae, 154 Sallari, 398, 485 Sahay, 295, 307, 485 Sant Venant, 150 Souds: 287, 470 Santamagina 235, 185 Santus, 235, 236, 241–288, 289, 352, 441, 164, 365, 185, Sanz, 392, 396, 486 Sarma, 100, 185 Souvear, 433–449 Sevart, 135, 136, 150 Scheiner, 148 Schlue, 396, 485 Schmidt, 387, 470 Scheenberg, 28, 29, 42, 43, 47, 121, 140. 284, 286, 288, 366, 365, 472, 473, 185, 486 Scholte, 116, 275, 140, 186 Schrödinger, 139, 151 Schulgasser, 235, 166 Schultes, 140, 371 Scott Blan, 83, 486 Scott Bussell, 336, 351 Sears, 110, 155 Socherk, 135, 150 Sekiya, 453 Sengbush, 80, 86, 87, 410, 479 Servetyńska, 87, 473 Seriavo, 12, 236, 389, 392, 397, 165, 482, 183, 486 Scriff, 220, 223, 469 Seria: 392, 396, 486 Shapiro, K. A., 411, 486 Shapiro, S. A., 20, 276, 294, 307, 471, 486 Shatta, 288, 186 Sherman, 54, 159 (16), 322 (333) 182 Showood, 398, 458, 478 Shin, 392, 474 Shttikaaa, 111 Siegins, 322, 338, 188 Still, 483, 486 Séva, 165, 186 Seeple 82, 87, 129, 431, 458

Skalak, 284, 288, 140, 169 Skempton, 243, 252, 254, 256, 486 Slawinski, six, 121, 214, 177, 186 Smenklers, 275, 288, 305, 306, 498, 475 Smit. 289, 471 Smith G D, 386 389, 387 Smith R. L., 88, 89, 487 Sucidon, 128, 469 Snell, vv6, 13, 14, 16, 115, 122, 124, 132, 133, 188, 191, 196, 198, 199, 208, 209, 224 226, 228, 350 354, 429, 432 433 415 Suvder 22 482 Soga, 275, 181 Sokidov, 140, 474 Soldner, 134, 150 Somigliana, 448, 453 Sommerfeld, 87 (159, 139) 153 Spallanzora, 449 Spinos, 288, 307, 468, 472, 485 Stam, 171, 468 Stefan, 452 Stendorg, 161, 487 Stephen, 387, 470 Staffa 378 188 Stokes, 431, 435, 437, 451, 487 Stoll, 235, 364, 487 Stoneley, 139, 154, 187 Stowas, 121, 244, 487, 488 Strahilevitz, 399, 476 Strang, 389 Striptt, J. W. so, Rayleigh, Lord Sturgeon, 150 Sturm, 135 150 Sol., 392, 474 Sylvester, 451 Tait, Md Tab.Ezer, 387, 391, 426, 477, 487 Tan. 307, 457 Taylor, B., 65, 153, 275, 386, 388, 390, 391, 391, 433, 449 Taylor, G. L. 454 Temperton, 305, 409, 322, 423, 487 Terzaghi, 240, 243, 274, 440, 454, 487 Tessmin, 391, 396, 399, 476, 487, 488 Thales of Milenes, 132, 147

Thumus, 236, 488 Thompson: 252 (255) 307 (188 Thomson, 9, 376–488 Theanson, J. J., 73, 439, 453, 483 Thomson, W., so: Kelvin, Lord Thomson, W. T., 234, 488 Tunoshenko, 154 Fing, 1, 2, 5, 188 Lismen-tsky, 298, 299, 477 Tirmani, 139, 140, 474 Toksőz, 86, 139, 307, 478, 488 Tomicotti, 84, 179 Tonti, 323, 188 Torn. 85, 468 Toryik, 86, 158 Trend 398, 475 Trongs, 397, 476 Techind ausen, 349 Tsymkin, 1, 188 Image, 235, 167 Turner, 322, 338, 488 Tyndall, 114, 451 Umov. xvi. xviii, 15, 16, 21, 43, 44, 46, 62, 63, 66, 107, 111, 124, 147, 148, 150, 156, 185, 189, 191, 192, 194 195 (197) 201, 203, 206, 210, 213, 215 310, 312, 315, 331 333 342, 343, 347, 363, 438, 452 Ursin, 43, 121, 234, 331, 187, 188 Validis, 389, 390, 467, 472, 488 Vainshtein, 88, 488 van den Berg, 172, 470 van der Waals, 452 Van Gestel, 378, 488 van Groesen 88 / 1120 / 188 yan Nestrand, MI, 86 87, 440, 479 Vargas, 215, 441, 461 Vavivén, k. 43, 483 Vernik, 22, 468 Vianello, 52, 180 Vilotte, 397, 476

- Virienx, 394, 398, 399, 189
- Vitrovius, 417
- Viviani, 132, 118

Voigt, xxn. 2, 38, 40, 51, 60, 68, 71, 74, 76 79 86 94, 106 117, 149 144 144, 181, 316, 317, 322, 325, 356, 357, 329, 438, 452, 489 Volta, 434, 449 Volterra, 51, 139, 153, 189 Wallis, 148 Wang, H. F., 235, 236, 459, 489 Wang, L. J., 164, 489 Wang, Z., 215, 458 Wapenar, 288, 168 Word, 398, 475 Waterman 411 489 Watson, 449. Watson Watt, 154 Weaver, 59, 60, 189 Weber, 33, 37, 341 Wegener, 115, 154 Weisstein, 147 Wendroff, 388 Wennerberg, 97, 140, 121, 124, 230, 231, 160, 489 Wentzel, 339, 555 West, 491, 402, 490 Weyl, 154 White, J. F., sviii, six, 80, 86, 87, 181 236, 290, 291, 295, 440, 441, 470, 189 White: M., 189 Whittaker, 455 (456) 453, 489 Washerr, Kid Waen, 453. Wrener, N., 153 Whener (O., 153) WErko, 119 Wills, D. G., 237, 239, 307, 159 WYEN, J. R., 252, 255, 307, 488 Wilmerski, 301, 470 Wraterstein 1, 6, 221, 489 Weeks 432, 447 Well, vix, 76, 195, 197, 323, 342, 345, 347, 349, 350, 370, 374, 140, 160, Wood, 291 189 Winglet, 11, M., 128, 341 Wingle J., 226, 200 Wigraystich: 383–490

Wellie, 372, 490 Nu 125 339 190 Yamada, 383, 490 Yin, 140, 490 Yokokuta, 383, 480 Young, 321, 434, 435, 450 Zahradnik, 394, 182 Zansey, 381, 183 Zorman, 153 Zener, xv. xix, 60, 62, 67, 74, 77, 79, 82 84, 90, 96, 141, 141, 145, 159, 167, 169, 186, 221, 304, 317, 337, 356 \$28, \$29, \$39, \$30 Zeng, 394, 402, 490 Zhang, 105, 109, 190 Zieghar, 117, 460 Zienkiewiez, 486, 492, 490 Ziannemaan, 245, 247, 249, 400 Zenszmer, 139, 235, 274, 275, 286, 294 295, 299, 457, 460, 461 Zneppritz, 154 Zwikker, 245, 410, 477, 490

Subject index

Astalohty, 388 (Osorbing boundary, 100, 102 adr-filled porous media, 235 Alford totation, 376 amplification Lotor, 489 matrix, d87 anaboyy. acousticados tromagnetic, 161, 196, 197, .111 boundary conditions, 342 Delive-Zener, 337, 340 electric circuit, 328 portoclastic selectromagnetics 378 nonoclastic-thermoelastic, 295 reflection-transmission problem, 442 SIL and TM equations, 327 FE and sound-wave equations, 352 I'M and sound-wave equations, 350 FW and 44, equations, 327, 354 aztisotzopé electromagnetic media, 356 conductivity, 457 dielectric permittivity, 357 magnetic permeability, 357 Maxwell's equations, d57 plane-wave theory, 359 poro-viscoelastic media, 307 homogeneous wave, dd l inhomogeneous wave, 412 (incellarmont) lield, dott porous media. dissipation potential, 263 effective stress, 254, 256 effective-stress coefficient, 256 equation of motion, 270 kinetic chergy, 260 pore pression, 255. Skempton relation, 256 smain energy, 250 total stress, 256 moltained-modulus matrix, 256 anomalous polarization

monorlinic moha, 33 orthorhomble media, 33 polarization vector, 41 stability constraints, 32 apatite custs, 22, 42 Greats function, 12 erono velocity, 22 sloten ss. 22 attendation factor constant O naulol, 84 electromagnetic diffusion, 381 homogeneitis wave. HH anisotropic electromagnetic un dia. (11.9)atisotrupii Visendastie nodia, 146 isotruph viscuolastic media, 143 one dimensional lassy media, 61, 65 memes media, 274, 277 Bayleigh wave, 119 reflection transmission problem qP qSV waves [210] SH wave, 2001 attendation victor anisotropic poro viscoclastic nu dia, 315 anisotroper viscoebistic modia, 153 homogeneous qP-qSV scares, 207 inhomogeneous qP/q8V/wavis, 225 isotropic viscoelastic media, 360 Bayleigh wave, 119 WO, 224, 234 Balanet's praciple, 375

Backus averaging elastic media, 29, 140, 394 electromagnetic media, 369 basis functions, 392 best isotropic approximation, 38 anisotropy index, 40 bulk and shear moduli, 39 Bérti-Rayleigh reciprocal theorem, 248 Bior relaxation peaks, 387 Bior Euler's equation, 404, 409

birchingence, 362, 376 lower for, 97. Holtzmann operation: 52, 69, 92, 93, 95, 124, 164, 165 Bultzmann's law, 51, 54 Bond matches, 9 borekole stalofity, 79. wave, 276 houndary conditions [114] 209 [229, 230. 3918 for surface, 117, 302 lavered media, 233 non-ideal montheer, 130, 131 open-pore, 288, 301 power halance, 285 sealed interface, 286, 301 Hoewster angle, 195, 196, 199, 345, 352 Brown and Korninga's equations, 256 feilk modulus, generalized, 165 Burgers model, 77 Iorial rate, 234 Cardiand-de Hump tochtique, 301 causahiw condition, 58, 161, 373 centroid, 89 contravolacity, 87, 89, 90 chalk, 226 characteristics variables, 399 Chebyslu v expansion hyperbolic equations, 391 parabolic equations, 391 an thial, 395, 402 strutching function, 426 Cole-Cole model, 86, 341 collocation points Chebyshev method, 425 Fourier method, 422 complementary energy theorem, 279 complex moduli dilatational, 142 shoat, 142 complex velocity ausotropic por existo elastic media, 314

amsofropic viscoclastic media, 146 constant-O-model, S4 electromagnotic dillusion, 381 figrial, 230 isotropic norms media, 273 isotropic viscoolastic media, 99 one dimensional lossy modia, 61, 65 poro visroaronstic media, 297 Rayloigh wheel 148 SH wave: 155, 188 TE wave 262. [] M. wave, 362. compliance tensor, 254–256 compressibility, 244, 247, 248 compressional wave isotiopic viscoelastic media, 108 norsons meshal 271 conductivity, 324 effortive, 458, 460 contral refraction, 11 configure-gradient method, 397 constant Q, 60, 83 nearly, 80, 82 constitution equations for objection agnotic media: 323 correspondence principle, 116, 127, 129, 142, 143, 145, 168, 169, 206, 298, 309 conside double without moment, 177, 178 single, 174, 178 with moment, 177 without moment, 177 Cogan number, 395 erack 46 (129, 135, 136, 183 cracking, 212, 216 Crank Nirolson scheme 389 comprompliance, 56, 78 crossi fanctaus 68 Burgers model, 78. Kolvin-Voigt middl, 72 Maxwell model, 69 Zener model, 75 CRDJ equation, 372 cristical angle anisotropic elastic modia, 43

attisotrupii. Visenolastie uredia qP-qSV waves: 215, 224, 226 SII wave, 195, 197, 199, 201, 203 electromagnetic media, 346 isotropic viscoelastic media, 123, 397 mossiplane shear wave (SH wave). anisotrupir elastic modia, 7, 15 attisotropic viscoolastic media, 155, 168 isotropic viscoelastic media, 104 reflection-transmission problem, 15, 121. 132 181 custal axes its ers stabs himisl. 356 milaxial, 356 nstis, 22, 220. effective anisotropy, 28 evlindnes) ports, 268 damping matrix, 400 Datew's law, 261 (263, 279, 286, 383) dynamical, 292, 394 density matrix operator, 309 deviator 3 diebetrie. impermentility, 334, 338 permittivity, 324 Cole-Cole model, 311 complex, 330 Dobye model: 338 officiates 358 3441 diffusion equation, 276 electromagnic nextia, 380 porous media, 378 14: wave, 380 I.M. wave, 380 diffusion length (293 diffusive slow model 276 dilatation, 3 recipiocity of: 177, 179, 181 dimensional splitting, 389 direction cosme, 10 directional former 397 discrete delta function, 398 disequilibrium compaction, 242, 244, 246 distrension relation anisotroph clastic media, 11, 18, 13,

anisotropic electromagnetic nuclia, 362 anisotropic poro-viscoelastic media, 314 anisotropic viscoelastic media, 146, 155, 330 constant-Q-model 85 isatuqin viscuelastic media. Diti mmerical methods, 388, 393, 395 one dimensional lussy media, 61, 65 orthorhombic media, 13 porteviscoacioisto mecha, 297 porous media, 273 Hayleigh wayn, 218 transversely isotropic media. 12 displacement. anisotropic viscoelastic media, 340 discontingly model, 130 forgulation, 165 vior" or anisotrunic clastic media. 10 anisotrupii Aisenelastie modia, 155 Bayleigh wave, 147, 148 dissipated energy, so contrydissipation factor 68 Kelvin-Virst model [72, 73] Maxwell miniel, 71 nearly constant O, 82 one-dimensional lossy facdia, 64 force, 262–265 potential anisotropic promis media, 193 isotrupic purcus media 262 double porosity, 236 downgoine wave, 232 dry-rock bulk modulus, 248, 241, 246 elasticity constants, 252 P-wave modulus 273 shear madalas, 237, 236 dual fields, 349 dynamic viscosity, 266 rarthquake 116-129-174 effective consolvopy clastic media, 26 popuelastic media, 29, 307 viscoulastic facilia. 142

effective-stress coefficient ausotropic media, 255 isotropic media, 242 permeability, 250 500 sity, 249 velocity, 213 eigenstillness, 5, 142, 144, 346 eigenstrain, 5, 142, 316 Suchorit, 144, 144 elasticity. constants, 2, 10, 22 effective anisotropy, 28 unrelayed. This mature, i 11 anisatione nero-visiorlastic media. 305 monoclinic gardia, fi nuthurkunskie nastia, ti transversily isotropic media, 6 "misor, 2, 143 dictue displanement 323 vector, 323 electric suscentibility, 373 electric-polarization vortor, 373electro-seismic wave propagation electro-liftration, 382 cleatro-osmosis, 482 electrokinetic coupling, 383 electro-seismic wave theory, 236, 382 electromagnetic duality, 487 eter av attisotropic elastic media, 16 aulsotropic porcevisco dastic media, 311 anishtropic viscorlastic media, 149, 152 coefficients 202 complementary, 279 roadactive isotropic dectricitação tie media, 333 dirlectric. Deliver model, 338 isotropic electromognetic media, 333 discongeness Lift dissipated anisotropic electromagnetic media. 364

an sotropic poro-visco dastic media 311, 313ansotropic viscoelastic media, 153 (softopic electromagnetic media, 332 (somopic viscoelastic media, 113 one-dimensional lossy modia, tid, 66 electede anisotropic electromagnetic media. 3064 isotropic electromagnetic media (132) TERLOR 364 TM wave, 365 lumingements waves and soft opic poro-visco dastic media. 315 satropic viscolastic media, 113 isotropic electromagnetic andra 331 isotropic viscodastar aurha, 111 knothe ansotropic elastic nasha, 16, 17 anisotropic poweriscoelastic media. 311, 313 an softopic porous media, 260, 261 ansotiopic viscoelastic media, 148 isomopic clastic media, 111 isomopic porous media, 257 isotropic viscoclastic media, 107 une-dimensional lossy modia, 62, 63, titi. P way , 108 magnetic anisotropic electromagnetic media. 361 Isotropic obstrunguotic media: 332. 336TE ways, 304 TM wave, 365 one dimensional lossy modia, 63 rate of dissingted and set ropic poro-visco dastic media. 311 anisotropic viscuelastic media, 53, 148, 152 Debye madel 339 humingeneous SH waves 156 sotiopic distrophysicir media, 332

331, 336 isotropic viscoolastic media, 108 one-dimensional lussy media, 63 Passee, 110 S waves, 114 IF wave 365, 366 [M serve, 365, 366] Zener model: 340 S verves, isotropic ciscoelastic medici, 111 strain anisotropic elastic media, 16, 17 and softwork porcession districting the 311 313 unisotropic porous media, 250 anisotropic visco-lastic media, 53, 118 elastic, 1 homogeneous SH wave 156 homogeneous waves, 315 infromomentous porosity, 278 isotropic clastic media, 111 isotropic media, d isotropic porous media, 247 isottopia viscoelastie media, 107 Maxwell model, 70. manaclinii media 2 one dira usional lossy media, 57, 62, 63, 66 orthorhombic media, 2 P wave, 110 pornus media, 280, 283 transversely isn'ropic navlia, 2 mitpreness 52/54 Zener model: 340 energy balance anisotropic elastic media, 15, 36 anisotropic electromagnetic mistra, 363 anisotropic porcevisco dastic media, 314 anisotropic viscoelastic media, 147, 149 isotropic electromagnetic modia, 331. 333isotropic viscoelastic media, 107 one-dimensional lossy media, 62, 66 Powaye, viscoelastic media, 108 Rayleigh wave, 119

reflection transmission problem qP/qSV waves: 210 SH wave, 201, 343 mergy mefficient, 211 mercy Envittow) sne Unite-Pownting voctor, 133 energy has non-ideal interface P-SV waves (135) SIL waves 133. energy velocity anisotropo elastic media, 16, 17-19, :90 anisotropo electromagnetic media, 364 anisotropo por existor lastic media, 313anisotropo viscoelastic media, 151 homogeneous SII wave, 155, 156 isotropic elastic media, 113 one-dimensional lossy media, 64, 66, 87 reflection-transmission problem qP-qSV wayes, 212 SII wave, 203 relaxed, 151 111 wave, 365 FM wave: 365 uniclased, 16, 151 envelope vehicity. anisotropic clastic media, 20 anisotroph viscoelastic modia, 147 homogeneuus SII wave, 155, 157 reportion of mution sre Enfer's equations, 4 anisotropic proces media (270) isotropic viscoelastic profia, 124 Maxwell model, 93 plane wave, 15 рото-унуссорсовуто днудур, 296 porous media, 265 Puler sel ente, 390 Fubri's equations (1, 525, 386) evolution operator, 8, 386 expansion functions Chebyshev method, 424 isourier method, 122 exploding-tellector concept, 398, 403

SUBJECT INDEX

fading menaory hypothesis, 51 ferenard brens 308, 316. FIFL 122, 326 Identification for ity, 264, 263, 267, 280 hardy layered an Sia, 183, 208 clastic monta, 6, 25 electromagnetic media, 469 pozous anedia, 29, 307 viscorlastic modia, 142 Initialifficience method: 385 exolicit, 387 haplien, 387 Indicodement method, d85 fhéd (solid miter lace, 228 forbidden directions, 161 Contrast cordes. andsutropic viscorlastic SH-waverengiation, 410 diffusion contation, 105 isotropic viscoelastic SH-scave equarion, 415 Maxwell's constion, 415 Fornie: method, 220, 394, 402, 422 fractional derivative, 85 electromagnetic media, 311 postous modia, 87 fracture, 129-135, 250 free energy 52. frequency counder, 65 Freshol's formular, 314, 351 liketion coefficient, 262, 263, 265 Irictional contact, 129 frozen pororis media, 236 Galerkin procedure, 386 Gassmann's compressibility, 249 contation, 211 modulus, 241, 248, 257, 291, 383 velocity, 274 gel 295 grothermal gradient, 241 Green's analogies, 352.

Green's analogies, 352
Green's femation anasotropic clostromagnetic media, 366 constant-Q media, 127 diffusion oppation, 381

one-dimensional, 88 poro-ciscoacoustic media, 295, 299 SIL wave amsonopic clastic media, 169 athsottopic viscoclastic media, 169 surface waves in potous media, 299thus alituetsional atúsotropic elastic media: 42 anisotropic el etranarnotic modia 369 two-dimensional atúsotropic elastic nuclia. W viscoaronstic media, 126 viscorlastic media, 128 ground-penotating radar, 87, 369 ground-roll 402 group velocity amsotropic elastic media, 18/20 unisotropic viscolastic nustro, 147 anomalous polarization, 35 Longenoous SH wave, 157 Legative, 161 numerical modeling, 394 one-dimensional lossy media, 61, 87 Hamilton's principle, 235, 263, 286 Handtwig solid (120) head wave 224 Howside-type function, 52 Helmholtz equation, 99, 392, 398 constant-Q media, 127 electromagnery diffusion, 381 electromagnero miedra, 367 poro-viscon oustic media, 297 heterogeneous formulation, 392 homogeneous formulation, 492 homogeneous wave, 97, 100, 461 homogenization theory, 235 Hould's law, 317, 386 anisotropic elastic media. 4 anisotropic viscoolastic andra, 141 one duminisional lossless media, 55 hybrid modeling scheme, 399 hydrauho diffusivity constant, 276 hydronarban, 211 hyperbolic differential equation, 386

iteraming wave, 399 incompressible solid, 120, 351 induced mass, 258, 259, informagencity angle, 100, 102 informogenoous wave, 97, 109, 359 interference coello ieut, 212 interference flux, 124, 199, 201, 202, 214, 311 internal fluidden) yn iable, 335, 355 interpolation functions, 392 isoparam trie method, 392, 396 polacted experiment, 237, 247, 252, 282 Kelvin's notation, 38, 141, 143 Kelvin-Christollel differential-operator marris, 5 essension relation, 11, 30 equation ansotiopic clastic needia. 11 ansotione decromagnetic media. 361transversely isotropic media, 11 read tix. andsotropic clastic media, 11, 18 anisotropic electromagnotic media 361 anisotropic poro-viscoclastic media. 314antsotropic viscoelastic media. 146 Kelvin-Voogt model: 60, 68, 71, 94, 141, 395 kinetic curray, so, concerve Kjartansson model, 83 Krammis-Krang dispersion relations, all constant/Q model, 60, 86 oh etromagnotic media, 373 Kelvin Voigt model, 60 Maxwell model, 601 Zener model, 60 Lagrangian, 235, 263, 269, 278 Lamb's problem, 172, 178, 181 Landeronstatus, 98 complex, 146, 120, 142, 230, 234 elastic media, 3, 6, 55 Lay-Wendroff scheme, 388

Lichtarcker Rother formula, 372 limestone, 139, 224 limit frequency, 270 linear momenta amsotropic porous media, 260 isotropic pozors media, 258 Loss wave, 396 low loss media, 65, 81, 83, 86, 123, 153, 157 181 MacCormark scheme 200 magnitic innermentality, 336 induction, 323 pernosability, 323, 360 vector, 323 mass matrix 392 Maxwell model 60, 62, 68, 92, 325 nomideal interface, 130 Maxwell's exprations, 323 isotroph media, 115. nemory reason dilatational, 164 show, 164 strain, 161 deviatorie. Hit memory variable, 92 anisotropic viscoelastic media, 162 computer storage, 125 differential equation. anisotrapic viscoelastic nu dui, 165 Enity difference approximation, 389. 1141 isotropic viscoelastic media, 125 Zener model, 95. dilatational mechanisms, 125 equation of motion pomovisi orlastic media, 305 generalized Zener model, 95 isotroph viscoclastic novila, 124 Kelvin-Voigt model, 91 portes visco dastar arrefat. 305 show merbowsn's 125 strain Maxwell model, 93 Zenet model, 95, 339 stress, 305.

Maxwell model, 93 mesoscopic loss morbarism, 289 manuscismogram, 305 modulus condex, 56, 141–186 analyticity, 59 conditions, 111 constant-O-model, 84 generalized Zener model, 79 isotropic viscoelastic media, 100 Kelvia-Voigt model, 72 Maxwell model, 69 nearly constant O, 83non-ideal interface, 131 properties, fill Zener model, 339 clastic, 55. loss, 56, 57 relaxed, 60 ceneralized Zener model, 79 Kelvin-Voigt model, 71 Zener model, 73. storages 56, 57 umehood, 60 ceneralized Zenet model, 80 Maxwell model, (is, 70) Zener model, 75. monopole, 174, 178 Neumann stability analysis, 396 Neumann's principle, 139, 208 Newmark method, 392 tion, aging material, 52 non-ideal interface, 129 non-reflecting wave oppation, 398, 105 non-midorm plane wave, 359 numerical modeling, potous mecha, 305, 384Nyonist criterion, 125 waverminher/ 388, 395, 398, 423 count bortons, 221, 228 Ohne's law, 323, 383 one-way wayse or parton. [101 Gusager's relations, 55, 263

orthogonal transformation regions, 8

onteone ways, 394 overnessure 242, 402 P-SV waves, non-ideal interface, 133 natabolic differential contation, 85 nanasitie modes, 395 paraxial approximation, 349 Parsecal's theorem, 89, particle motion. anisotropic viscorlastic moßa, 154 elliptical, 103, 106 usotropic viscoclastic motia-P wave: 102 S wave, 104. particle-velocity/stress for implation anisotropic viscorlastic ano£a, 110 concellair clastic media, 7 symmetry plane, 7 peak value, 147, 149 permoability, 262 dynamic 269 matrix, 263 perpendicularity properties atusottopic clastic media, 24 anisotropic viscoelastic media, 157, 158, 240 phase velocity unisotropic elastic miglia, 11 anisotropic viscorlastic moßa, 151 complex frequency, 66 complex wavenumber, 01 constant-O-model, S4 electromagneric diffusion, 381 traite difference, 388 humagemons waves. anisotropic electromagnetic media. 362anisotropic viscoclastic media, 146 sotropic viscodastic medla, 102 mhormeneous waves lisetroper visrodastic rundia, 102 isotropic portus media, 274 Kolvin-Voigt modol, 72 Maxwell model, 71 tosuly constant Q. 82 one-dynamicated lossy media, 61–87. orthorhombic elastic media, 13

porntis media, 277 Rayleigh wave [119] reflection transmission problem. aP/aSV waves, 210 SIL wave: 1991 ferzuchi's approximation, 275 transversely isotropic media, 12 Zener madel 26 Piece shale, Sti plane of symmetry, 7 plane slit, 265 plane wave anisotropic elastic media, 40 anisotropic electromagnetic archa, 359 anisotropic porcevisco dastic media, 312 anisotropic viscoelastic media, 149, 205 isotropic porous media, 277 isotropic viscoclastic media, 100 one-dimensional lossy media, 61, 65 normers media, 273 Rayleigh wave, 117 PML method. 4.3) Poisreille flow, 262, 267, 269, 270 Doission ratin 353, 393, 399, 402, 404 solid, 120, 136, 353. polarization, 229 anisotrupic viscuolastic media. Li l anomalous, 29, 40 elliptical, 154 isotrupii visenelastie media, 106 orthoganality, 14 transversely isotropic media, 12 pure compressibility, 245–247 pore-volume balance, 214, 245 processing the equations (375) porosvjecovljetjertv amsotroper media, 307 isotropic media, 114, 403 porous media anisotropic por systematicity, 307 boundary conditions, 284 compressibilities, 246 compressional wave, 271 dissipation (scential, 262 effortive stress, 240

Green's Innetina, 295, 299 isolated experiment, 237 kitorik carergy, 257 Lagrange's originations, 263 mmercel nodeling 305 shear ways, 276 strain meroy 237, 250, 278 stress strain relations, 237, 250 uniaskened «speriment, 238 viscodynamic operator, 265 vision/astur, 303 power flaw (flag), we Unice-Preseting vortor profertie concerto: scheme, 390 bussion containe, 242 differential, 232 elfortive, 230, 242, 246 fluid see price pressure hydrostatic, 242 lithestatic 242 pore, 240, 242 pressure logarithation, 385 pressure source, 497 pressno-seal reflections, 234 principal axes, 8 propagation matrix, 386, 387 osciolocritical angle, 197 osciolospectual method, 385 pure mode direction, 12 ul' wave, elastic un dia, 13 qP-qSV waves attentistion angle, 209 attentiation vector, 205 conflox-slowness voctor, 205 dispersion relation, 205 correctancle, 209 homogenenus, 207 polarization, 205 propagation angle, 200 reformentransmission problem, 205 shown secontor 1964 Uniov-Privating vertor, 206 qPapSV, equation of motion, 166 uS wave, clastic media, 13 quality factor, 186

ausonopic electromagnetic media. 464 presotropic porceviscor/astic media, 313 ardsattonic viscodastic ruglia, 152 **Biot** relaxation, 289 constant Q-model, 81 homogeneous wave: 101, 207, 316 anisotropic viscoelastic media, 153 isotropic viscoglastic media, 113, 341 esotropic electronogueta media, 328 isutropic viscorlastic media, 113 Kolvin Voigt mudol, 72 Maxwell model, 71 one-dimensional lossy media, 61 rellection-transmission problem qP-qSV waves, 213 SH waves 204. TE wave, 366 TM wave, 366 Zeno i model, 76, 289 quasi-static mode, 278–299, 300, 307, 387 rate of dissipated energy is a energy (a) tracing, 121, 166 Rayleigh wave, 116, 396, 102 non-ideal interface, 136 tionous media, 301 quasi-elastic, 117, 118, 120 viscordastic 117/118/120 Rayleigh window, 230 reality condition, 55, 58 rectino ity borehole seismics, 181 hydrophones, 181 of strong 174 of stress, 179 recipitority principle clastodynamics, 171 else tromagnetism, 374 referrion coefficient enhomogeneous wave: 226 Ravleigh window, 231 reflection matrix, 8 reflection-transmission problem atilsotropic clastic media: 12 anisotropic viscorlastic media, 183, 342 fluid-solid interface, 229. isotropic viscodastic media, 121

lavered media, 231 non-ideal introface, 132-133 solid (Incidenterface, 225) reflectivity, 349 reflectivity method, 234 refraction index, 314 relaxation frequency Biot relayation prak, 289 White model, 293 function, fi8, 98 Bingers model, 78 conditions, 58 constant O. Si generalized Zeney model, 79. satropic viscodastic media, 124 Kolyme-Voigt middl, 72 Maxwell medel, 69 nearly constant O. 82 P-wave, 99 properties, 60 Zener model, 75, 339 matrix and subropic poro viscorlastic media. 309 anisotropic viscoelastic media, 163 symmetry, 55 peak, Zener model, 76 tensor 198 symmetrics, 55 times, 81, 167, 186, 305, 387 generalized Zener model, 79. Kelvm-Voigt model, 72 Maxwell model 69 Zener model, 75. REM method, 391 representation theorem, 173 response function, 92 Maxwell model, 93 rigidity, constalized, 165 rotation potrix 8, 9 RampsKutta method 220/388/390/402 S waves, viscoelastic media type-1 (SV wave), 104 type-II (SH novel, 194 sandstone 139/236, 250/278, 306/102

stattering, 135, 104 Scholtr wave in porms media, 299 scismic explanation, 224, 232, 394, 396 seismor pedset 327 scismogram, 402 SH waveamsotropic elastic media energy velocity 21 group voluence 21 reflectorationsmission problem, 43 slowness surface, 24 Uniov-Pointine voctor, 21 wave surface, 24 anisotropic viscoelastic media attentiation angle, 190 attentiation on tor. 181 dissipated energy, 186–191 energy angle, 192 energy velocity, 156, 330 envelope velocity, 157 group velocity 157 kinetic energy, 156 propagation angle 190 reflectoristransmission problem, 181, 185, 342 slowness relation, 185 slowness vector, 184 strain energy, 186 stress-strang relation, 184 Uniov-Poynting voctor, 185, 191, 193 195. ebistic nasha, 12 equation of motion anisotropic viscoelastic media, 166, 100 isotropic clastic media, 386 isotropic viscoelastic modia, 115 Maxwell model, 326 Instrugenticions. anisottopia viscoclastic media, 155, 181isotropic viscoelasme archa, 122 nen-ideal interface, 132 reflection-transmission problem, 45, 121. 132, 181 shab., 86, 186, 224, 226, 236, 250

shaley sandstone (236 shape functions, 392 shirar source, 397 shirar wayo, parmis media, 276 shear-wave splitting, 22, 392, 376 Skringstin, coefficient, 243, 252, 253 skin depth, 381 Sine waves 273 (275) 278 (293) 318 (387) Sourcess come: SH wave: 157 showness surface, orthorhombic glustic mečni 13. slaveness vector anisotroper elastic media, 11 anisotropo poro-viscoelasticumenta, 315 anisotropa viscoelastic media, 146 inhomozenosus wave, 225 isotropic visco-dastic media, 102 qP-qSV ways, 207 SH wave: 155 stowness, one-dimensional lossy modia, 61 SucI's law, 188, 191, 199, 208, 228, 351 anisotropic clastic media, 13 non-ideal interface, 133 SH wave, 122 viscoolastic gardia, 111 storrife stilfaces, 130 sporific viscosity, 130 sportial an index, 391 spectral element method, 396 spingermethod, 400 sumu-flow merhanism, 309 stability condition anisotropu elastic media, 5, 6 anisotropa viscoebstic modia, 148 nomencal modeling, 358 Runge-Nutta method, 390 stoggened grid, 393 spatial derivative, 391, 394, 395, 399, IGE 109 115 stancase effect, 394, 402 standard linear solid model, 74 stoff differential equation, 387 stiffaces complex, 142, 224 constants

omborkondsie media. 111 uniclased, 147 metros, 141, 327, 329 finite-element methad, 392 miclayed, 214 stop bands, Dill SHain anisotropic poro-viscoelastic media, 408 components, 2 CHARGES STATEMENTERS humingeneous SH waves 155 invariants, 3 madrie, 4 reciprosity of 174 strain-displacement relations. 1 stram-spass relation, 56 anisatropic elastic media. A Strang's scheme, 389 stress atisotropic poro-visco lastic media, 308 components, d deviatoric components, 98-1114 offective 240,242,253,255 humingoneous SH water, 156 isotropic porots media, 237 man, 144 restiprosity of 179 -mail: 240 stress-street relation. anishtropic paro-viscorlastic media, 309 anisotropic porces media, 251, 308 aufsutropic viscoelastic im dia, 53, 54, 140, 161 constant-O-model, 85 generalized Zener model, 79 esotropic porors needed, 237, 239 isatropic viscodastic media, 98 Kolvin Voigt mudol, 71 Maxwell model, 68 one-dimensional lossy media, 55 portous mecha: 280 Zener model, 73, 75 structural factor, 268 superfluid, 275 surface topography, 395, 396

hydor method, 388

IE (mansy) (seeded (i)) equations, 427 terzachi's equation, 242 Lau, 274 thermal-equipsion coefficient, 211 time average over a cycle. 15 time-average equation, 372 FM (transverse magnetic) reputions (326) fortuosity, 259, 304, 318 dynamic, 268 transducers, 24 transmissivity, 319 tiplet wayse, 275 Unary-Poynting theorem, see energy halincut et Uniov Posiating vertor atusofropic clastic modia, 15, 16 anisotropic do tromagnetic media, 363 anisotropic poro-viscoelastic media, 310, 312anisotropic viscorlastic aurita, 148 obsetches TF waves 3655 humogeneous SH wave, 156 longagemons waves, 113, 315 isottopic clastic media, 112 isotropic viscoclastic modia, 107 one-finensional lossy media, 62, 66 P wave, 111 S wavas, 111 [] M. waves, 365. undrahurd-modulus matrix 253 miform plane wave: 361 unjacki ted experiment, 238, 252 mirelaxed volucity, 226 upcoing wave, 232 variation of Bud content, 240, 249, 251 261, 279, 280, 282, 308 variational formulation, 386 viscolynamic matrix, 271, 296 operator, 265, 267–269 low frequency, 265 viscoelastic fluid, 70 viscosity, 262 natus, 141, 327

SUBJECT INDEX

visemas lesses, 130 Volet's notation, 111 volume-averaging method, 245, 280 wave econssion), 49, 198, 201, 203, 235 from Di attisottoph Viscoelastie media, 151 homogeneous, 100 inhomogeneous, 97, 224, 225 clastic 402 surface. anisotrupir clastic media, 16 anisotropic prenorisonolastic media. 313 anisotropic viscoelastic media, 151 traveline, 17 wave equation, constant-Q (mode). So wave-manifest. conndex, 61 honorgeneous waves. 101, 314 Planet 99 S wave, 99 and, homogeneous wavas, 1011 white anisotrupir clastic media, 10 anisotropic porous media, 312 attisottopii viseneiastie media, 150 isotropic viscoelastic media, 100 weighting coefficients, 393 White padel: 289 Powerse complex modules, 223 Wood's equation, 294 waqanound, 395 Zener model, 60, 74, 95, 167, 317, 339 generalized: 79, 95 zen-padding trebinque 399

This page intentionally left blank